THE FEDERAL DEMOCRATIC REPUBLIC OF ETHIOPIA CENTRAL STATISTICAL AGENCY

STATISTICAL REPORT ON THE 2015 URBAN EMPLOYMENT UNEMPLOYMENT SURVEY

Addis Ababa October 2015

STATISTICAL BULLETIN

TABLE OF CONTENTS

	tents	
	ical Tables	
Abbreviations	S	vii
Obantan I	Deplement of the Company	
Chapter I	Background of the SurveyIntroduction	1
1.1		
1.2	Objectives of the Survey	
1.3	Content and Organization of the Survey Questionnaire	
1.4	Training of Field Staff	5
1.5	Organization of the Field Work	5
1.6	Data Processing	
1.7	Definitions of Terms and Concepts	7
1.8	Organization of the Report	9
Chapter II	Survey Methodology	11
2.1	Coverage	11
2.2	Sampling Frame	
2.3	Sample Design	
2.4	Sample Size and Selection Scheme	
Chapter III	Socio - Demographic Characteristics of	
Chapter in	Urban Population	15
3.1	Introduction	1C
3.1		18
3.2	Age- Sex Structure of Urban Population and	
	Other Socio - Demographic Data	15
01 (1) (
Chapter IV	Economic Activity Status	
4.1	Introduction	51
4.2	Data Collection Approaches of the Economic Activity Status of	
	Urban Population	53
	4.2.1 Current Activity Status Approach	53
	4.2.2 Usual Activity Status Approach	53
4.3	Current Activity Status	
4.4	Economically Active and Non-Active Population and Activity Ra	
Chapter V	Characteristics of the currently urban Employed Population	- 75
5.1	Introduction	
5.1 5.2	Employment to Population Ratio	
5.2 5.3	·	
	Occupation and Industry	
5.4	Status in Employment	
5.5	Number of Hours Worked	
5.6	Earning from Paid Employment	
5.7	Absence from Work	
5.8	Availability to Work Additional Hours	
5.9	Formal and Informal Sector of the Economy	81

Chapter V I	Unemployment in Urban Areas	219
6.1	Introduction	219
6.2	Measurement of Unemployment	220
6.3	Size and Rate of Urban Unemployment	222
Annex I	Survey Questionnaire	273
Annex II	Estimation Procedures of Total, Ratio and Sampling Errors	279
Annex III	Estimates and CV's for Selected Tables	283
References		286

LIST OF STATISTICAL TABLES

Pag	ges
-----	-----

Table 3.1	Estimated Total Population of Urban Areas by Age Group, Sex and Region: 201517
Table 3.2	Distribution of Urban Population, Number of Households and Average Household Size by Region: 201519
Table 3.3	Distribution of Urban Population by Region, Sex and Relationship to the Head of Household: 2015
Table 3.4	Distribution of Urban Population Aged Five Years and Above by Age Group, Sex and School Attendance- COUNTRY TOTAL: 2015
Table 3.5	Distribution of Urban Population Aged Five Years and Above by Region, Sex and School Attendance: 201523
Table 3.6	Distribution of Urban Population by Ten Years and above by Region, Sex and Literacy Status: 201524
Table 3.7	Distribution of Urban Population Aged Ten Years and Above by Age Group, Sex, and Educational Attainment - COUNTRY TOTAL: 201525
Table 3.8	Distribution of Urban Population Aged Ten Years and Above by Region, Training Status, Sex and Types of Training - COUNTRY TOTAL: 201527
Table 3.9	Distribution of Urban Population by Five Years Age Group, Sex and Sex Ratio - COUNTRY TOTAL: 201550
Table 4.1.	Urban population Aged Ten Years and Above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach) – COUNTRY TOTAL: 2015
Table 4.1.1	Urban population Aged Ten Years and Above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach) – TIGRAY REGION: 2015
Table 4.1.2	Urban population Aged Ten Years and Above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach) – AFFAR REGION: 2015
Table 4.1.3	Urban population Aged Ten Years and Above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach) – AMHARA REGION: 2015
Table 4.1.4	Urban population Aged Ten Years and Above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach) – OROMIA REGION: 2015
Table 4.1.5	Urban population Aged Ten Years and Above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach) – SOMALI REGION: 2015
Table 4.1.6	Urban population Aged Ten Years and Above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach) – BENISHANGUL GUMUZ REGION: 2015
Table 4.1.7	Urban population Aged Ten Years and Above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach) – SNNP REGION: 2015
Table 4.1.8	Urban population Aged Ten Years and Above r by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach) – GAMBELLA REGION: 2015
Table 4.1.9	Urban population Aged Ten Years and Above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach) - HARARI REGION: 2015

Table 4.1.10	Urban population Aged Ten Years and O Above by Age Group, Sex, Activity Status and Rate during the Last Seven Days
	(Current Status Approach)- ADDIS ABABA CITY ADMINISTRATION: 2015
Table 4.1.11	Urban population Aged Ten Years and Above by Age Group,
14016 4.1.11	Sex, Activity Status and Rate during the Last Seven Days
	(Current Status Approach) – DIRE DAWA ADMINISTRATION: : 2015
Table 4.2	Population of Major Towns Aged Ten Years and Above by Sex,
Table 4.2	Activity Status and Rate during the Last Seven Days (Current Status Approach):
	2015
Table 4.3	Economically Non - Active Population of Urban Areas Aged Ten Years and Above
14510 4.0	by Age Group, Sex and Reasons For Not Being Active During the Last Seven Days
	(Current Status Approach) - COUNTRY TOTAL: 201571
Table 4.4	Economically Non-active Population of Urban Areas Aged Ten Years and Above by
	Region, Sex and Reasons for Not Being Active During the Last Seven days
	(Current Status Approach):201573
Table 5.1	Urban Population Aged Ten Years and Above by Region, Age Group,
	Sex and Employment to Population Ratio - COUNTRY TOTAL: 201583
Table 5.2	Urban Population of Major Towns Aged Ten Years and Above by Age Group, Sex
	and Employment to Population Ratio: 201587
Table 5.3	Employed Population of Urban Areas Aged Ten Years and Above by Major
	Occupational Groups, Age Group and Sex - COUNTRY TOTAL: 201593
Table 5.4	Employed Population of Urban Areas Aged Ten Years and Above by Major
	Occupational Groups, Region and Sex: 201595
Table 5.5	Employed Population of Major Towns Aged Ten Years and Above by Major
	Occupational Groups and Sex: 201597
Table 5.6	Employed Population of Urban Areas Aged Ten Years and Above by Minor
	Occupational Groups, Region and Sex: 201599
Table 5.7	Employed Population of Urban Areas Aged Ten Years and Above by Major
T	Industrial Divisions, Age Group and Sex - COUNTRY TOTAL: 2015120
Table 5.8	Employed Population of Urban Areas Aged Ten Years and Above by Major
Table F O	Industrial Divisions, Region and Sex-COUNTRY TOTAL: 2015
Table 5.9	Employed Population of Major Towns Aged Ten Years and Above by Major Industrial Divisions and Sex: 2015126
Table 5.10	Employed Population of Urban Areas Aged Ten Years and Above by Minor
14016 3.10	Industrial Divisions, Region, and Sex - COUNTRY TOTAL: 2015
Table 5.11	Employed Population of Urban Areas Aged Ten Years and Above by Status in
14510 0111	Employment, Age Group and Sex - COUNTRY TOTAL: 2015
Table 5.12	Employed Population of Urban Areas Aged Ten Years and Above by Status in
	Employment, Region and Sex: 2015171
Table 5.13	Employed Population of Major Towns Aged Ten Years and Above by Status in
	Employment and Sex: 2015
Table 5.14	Employed Population of Urban Areas Aged Ten Years and Above by Major
	Industrial Divisions and Major Occupation Groups and Sex
	- COUNTRY TOTAL: 2015 175
Table 5.15	Employed Population of Urban Areas Aged Ten Years and Above by Major
	Industrial Divisions and Status in Employment and Sex –
	COUNTRY TOTAL: 2015
Table 5.16	Employed Population of Urban Areas Aged Ten Years and Above by Major
	Occupational Groups and Status in Employment and Sex -
Table 5 47	COUNTRY TOTAL: 2015
Table 5.17	Employed Population of Urban Areas Aged Ten Years and Above by Mean Number
	of Hours Worked during the last Seven days, Age Group and Sex, - COUNTRY TOTAL: 2015
Table 5.18	Employed Population of Urban Areas Aged Ten Years and Above by Mean Number
. 45.0 0.10	of Hours Worked Per Week, Region, and Sex : 2015

Table 5.19	Employed Population of Urban Areas Aged Ten Years and Above by Mean Number of Hours Worked Per Week, Major Industrial Divisions, and Sex - COUNTRY TOTAL: 2015
Table 5.20	Employed Population of Urban Areas Aged Ten Years and Above by Mean Number of Hours Worked Per Week, Major Occupational Groups and Sex - COUNTRY TOTAL: 2015
Table 5.21	Employed Population Urban Areas Aged Ten Years and Above by Mean Number of Hours Worked Per Week, , Status in Employment and Sex – COUNTRY TOTAL: 2015192
Table 5.22	Paid Employees of Urban Population Aged Ten Years and Above by Type of Payment, Region and Sex: 2015194
Table 5.23	Paid Employees of Urban Population Aged Ten Years and Above by Period of Payment and Sex- COUNTRY TOTAL: 2015196
Table 5.24	Paid Employees of Urban Population Aged Ten Years and Above by Period of Payment and Amount of Total Payment and Sex, – COUNTRY TOTAL: 2015197
Table 5. 25	Paid Employees of Urban Population Aged Ten Years and Above by Major Industrial Divisions, Sex and Amount of Payment per month – COUNTRY TOTAL: 2015
Table 5.26	Employed Population of Urban Areas Aged Ten Years and Above who were not at Work by Reason for Absence, Region and Sex: 2015201
Table 5.27	Employed Population of Urban Areas Aged Ten Years and Above Whether Available and Ready to Work Additional Hours by Percent Available and Ready to Work Additional Hours Age Group and Sex, – COUNTRY TOTAL: 2015202
Table 5.28	Employed Population of Urban Areas Aged Ten Years and Above Who Sought or Available to Work Additional Hours and Number of Hours Worked Per Week by Age Group and Sex – COUNTRY TOTAL: 2015
Table 5.29	Employed Population of Urban Areas Aged Ten Years and Above Who were willing and Available to Work Additional Hours and Type of Work Sought by Region and Sex: 2015
Table 5.30	Employed Population of Urban Areas Aged Ten Years and Above by Sector of Economy, Percentage Share of Informal, Age Group and Sex, - COUNTRY TOTAL I: 2015
Table 5.31	Employed Population of Urban Areas Aged Ten Years and Above by Sector of Economy and Percentage Share of Informal, Region and Sex: 2015207
Table 5.32	Employed Populations of Major Towns Aged Ten Years and Above by Sex, Sector of Economy and Percentage Share of Informal and Sex: 2015
Table 5.33	Distribution of Urban Population aged ten years and Above Working in the Informal Sector by Major Occupational Group, Region and Sex: 2015210
Table 5.34	Distribution of Urban Population aged ten years and Above who were Working in the Informal Sector by Major Branches of Industry, Region, and Sex: 2015 212
Table 5.35	Distribution of Urban Population aged ten years and Above who were Working in the Informal Sector by Literacy Status and Educational Level Region and Sex : 2015213
Table 5.36	Distribution of Urban Population Aged ten years and above who were working in the Informal Sector by Type of Working Place, Region and Sex,: 2015215
Table 5.37	Employed Population Urban Areas aged ten years and above by Mean Number of Hours Worked Per Week, Major Branches of Industry, Region and Sex: : 2015
Table 6.1	Unemployed Populations of Urban Areas Aged Ten Years and Above by Region, Unemployment Rate, Age Group, Sex and: 2015224
Table 6.2	Unemployed Populations of Major Towns Aged Ten Years and Above by Unemployment Rate, Age Group and Sex: 2015228

Table 6.3	Unemployed Population of Urban Areas Aged Ten Years	
	and Above by Literacy Status and Educational Attainment, Age group, and Sex	ζ,
		234
Table 6.4	Unemployed Populations of Urban Areas Aged Ten Years	
	and Above by Region, Educational Attainment and Sex: 2015	236
Table 6.5	Unemployed Population of Urban Areas Aged Ten Years	
	and Above by Region, Sex, Training Status and Types of Training: 2015	. 238
Table 6.6	Unemployed Population of Urban Areas Aged Ten Years	
	and Above by Marital Status, Age Group and Sex- COUNTRY TOTAL: 2015	262
Table 6.7	Unemployed Population of Urban Areas Aged Ten Years	
	and Above by Region, Marital Status and Sex: 2015	.263
Table 6.8	Unemployed Population of Urban Areas Aged Ten Years	
	and Above Who Were Seeking and not Seeking Work by Region, Steps	
	Taken to Seek Work and Sex: 2015	.264
Table 6.9	Unemployed Population of Urban Areas Aged ten Years	
	and Above by Region, Type of job looking for and Sex: 2015	266
Table 6.10	Unemployed Population of Urban Areas Aged Ten Years	
	and Above by Previous Work Experience, Duration of Unemployment, Age Ground	
	and Sex -COUNTRY TOTAL: 2015	
Table 6.11	Unemployed Population of Urban Areas Aged Ten Years and Above by Region,	
	Previous Work Experience, Duration of Unemployment and Sex	
	(In Months): 2015	.269
Table 6.12	Unemployed Population of Urban Areas Aged Ten Years and Above by Region,	
	Types of Problem Faced to Establish Own Business and Sex,: 2015	271

ABBEREVATIONS

CSA- Central Statistical Agency

EA - Enumeration Area

HH- Households

ICSE- International Classification of Status in Employment

ILO- International Labour Organization

CSPro- Census and Survey Processing system

ISCO- International Standard Classification of Occupation

ISIC- International Standard Industrial Classification of all economic activities

ICLS- International Conference on Labour Statistics

KILM- Key Indicators of Labour Market

GTP-Growth Transformation Plan

NIHSP- National Integrated Household Survey Program

NLFS - National Labour Force Survey

NS- Not Stated

NOIC- National Occupation and Industrial Classification

PSU- Primary Sampling Unit

RLFS- Rural Labour Force Survey

SDGs-Sustainable Development Goals

SNA- System of National Account

SNNP- Southern Nations, Nationalities and Peoples

TOT – Training of Trainers

TVET- Technical Vocational Educational Training

UEUS - Urban Employment and Unemployment Survey

CAHAPTER I

BACKGROUND OF THE SURVEY

1.1 Introduction

Statistical information on all aspects of the population is vital for the design, implementation, monitoring and evaluation of economic and social development plan and policy issues. Labour force survey is one of the most important sources of data for assessing the role of the population of the country in the economic and social development process. It is useful to indicate the extent of available and unutilized human resources that must be absorbed by the national economy to ensure full employment and economic well being of the population. Statistics on the labour force further present the economic activity status and its relationship to other social and economic characteristics of the population. Seasonal and other variations as well as changes over time in the size, distribution and characteristics of employed and unemployed population can be monitored using up-to-date information from labour force surveys. It serves as an input for assessing the achievements of the post 2015 Sustainable Development Goals (SDGs). Furthermore, labour force data is also useful as a springboard for monitoring and evaluation of the second five years growth and transformation plan/GTP- 2/ of the country.

Despite the significance of the labour force data, the availability of reliable and timely labour force data were inadequate. The lack of reliable and timely data on different aspects of the population hinders the monitoring and evaluation of changes of developmental activities.

The Central Statistical Agency (CSA) has been providing labour force and related data at different levels with various contents and details in order to fill the gap in data requirement for the purpose of planning, monitoring and evaluation. These include the 1976 Addis Ababa Manpower and Housing Sample Survey, the 1978 Survey on Population and Housing Characteristics of Seventeen Major Towns, the 1980/81 and 1987/88 Rural Labour Force Surveys (RLFS). The 1984, 1994 and 2007 Population

and Housing Censuses and the 1999, 2005 and 2013 National Labour Force Surveys provided a comprehensive national labour force data representing both urban and rural areas.

Considering the development process of the country and dynamic nature of the sectors and fulfilling the demands of different data users, the CSA has been launched Urban Employment Unemployment Survey program (UEUS) since 2003. The Agency conducted UEUS in seven rounds. The statistical and analytical report of the October 2003, April 2004, and April 2006 surveys were published together in bulletin number 301, 319, 373, respectively. Whereas the May 2009, May 2010, March 2011 and March 2012 surveys, the statistical and analytical results were published separately. The statistical report found in bulletin number 469, 499 and 521, while the analytical report in bulletin number 477, 504, 530, 555 and 576, respectively.

The 2015 Urban Employment and Unemployment Survey conducted from February 26 - March 27, 2014 is the ninth series. Unlike the previous surveys the May 2009, May 2010, March 2011, and March 2012 the recent survey of April 2015 survey have collected information from selected major towns with population size 100,000 and above and including region capital cities regardless of their population size.

In addition, data on economic activities of children were also collected to measure child labour in urban areas. For this purpose, the former minimum age limit 10 years was lower down to 5 years since May 2009. The data in each subsequent survey was also collected from those persons aged five years and above this is because children in rural and urban areas used to work at their early age such as collection of fire wood, looking after cattle, shoeshine, street vendor, petty trading...etc.

The survey primarily provides data on the size, distribution and other sociodemographic characteristics of employed and unemployed population. It also provide data on active population and activity status and employment in the informal sector, and its distribution in the various sectors of the economy during the survey reference period. For the purpose of measuring the economic activity status based on Ethiopian situation, the lower age limit was fixed in to ten years. This statistical report is therefore mainly aimed at provide information on the economic characteristics of the population aged ten years and above. Furthermore, the 2015 UEUS provide data on employment in the informal sector, their spatial distribution and at various sectors.

This statistical report presents a brief introduction of each chapter including basic concepts based on ILO standard definitions together with statistical tables.

1.2. Objectives of the Survey

The general objective of the 2015 Urban Employment and Unemployment Survey is to provide statistical data on the characteristics and size of the economic activity status i.e. employed, unemployed population of the country at urban levels on annual basis. The specific objectives of the survey are to:

- collect statistical data on the potential manpower and those who are available to take part in various socio-economic activities;
- update the data and determine the size and distribution of the labour force participation and the status of economic activity for different sub-groups of the population at different levels of the country; and also to study the socioeconomic and demographic characteristics of these groups;
- identify the size, distribution and characteristics of employed population i.e. working in the formal or informal employment sector of the economy and earnings from paid employees and its distribution by occupation and Industry...etc;
- provide data on the size, characteristics and distribution of unemployed population and rate of unemployment;
- provide data that can be used to assess the situation of women's employment or the participation of women in the labour force;
- generated time series data to trace changes over time;

1.3. Content and Organization of Survey Questionnaire

The questionnaire was organized in to six sections;

- **Section 1:** Area identification of the selected household: this section dealt with area identification of the respondents such as region, zone, wereda, etc.
- Section 2: Socio- demographic characteristics of households: it consisted of the general socio-demographic characteristics of the population such as age, sex, education, status and type of migration, disability, literacy status, educational Attainment, types of training and marital status.
- Section 3: Economic activities during the last seven days: this section dealt with a range of questions which helps to see the status and characteristics of employed persons in a current status approach such as hours of work in productive activities, occupation, industry, status in employment, earnings from employment, job mobility, service year for paid employees employment in the formal and informal sector and time related under employment.
- **Section 4:** Unemployment and characteristics of unemployed persons: this section focused on the size, rate and characteristics of the unemployed population.
- Section 5: Economic activities during the last twelve months: this section consists of the usual economic activity status refereeing to the long reference period i.e. engaged in productive activities during most of the last twelve months, reason for not being active, status in employment, main occupation and industry with two digit codes.

A structured questionnaire was used to solicit the required data in the survey. The draft questionnaire was tested by undertaking a pretest in selected kebeles (lower administrative unit) in Addis Ababa. Based on the pretest, the content, logical flow, layout and presentation of the questionnaire was amended. The questionnaire used in the field for data collection was prepared in Amharic language. Most questions have pre coded answers and column numbers were assigned for each question. A

copy of the questionnaire translated in English language is attached as Annex I of this report for reference.

1.4. Training of Field Staff

The training program of enumerators, supervisors, statistician and other field staffs will be conducted from February 16 - 23 /2015, at four training centers in Ambo, Hawassa, Dessie and Harar towns for nine days. The trainees were selected from CSA head and branch offices and from those who have long time experience in data collection. The training will be given by head office senior professionals.

The training was guided by an enumerator and supervisor manuals, which consists of detailed explanation of concepts, ideas and instructions on how to fill each entry in the questionnaire.

The training consisted of theoretical classroom discussions on concepts, definitions and techniques of completing the questionnaire as well as mock interview. The class room presentation and discussions were intended to exchange experiences among participants and pinpoint the areas of the survey questions that need more care and attention. The objectives of mock and practical interviewing of households were twofold. First, it enabled to assess how well the classroom theoretical discussions were understood by all participants to convey the same message at all levels. The second objective was to examine the difficulty, which would likely be encountered during actual field work. At the end of the training, one day field practice was carried out to familiarize on the contents and structure of the questionnaire through interviewing of household members followed by general discussion.

1.5. Organization of the Fieldwork

In order to carry out this survey with the desired level of quality, there was a need to organize a large staff that performs the various survey activities. The 25 Branch Statistical Offices of the Agency were responsible to carry out the data collection operation. These offices have permanent and contract enumerators stationed in the

selected enumeration areas. The data collection operation of the survey involved a total of about 520 of which 357 were enumerators, 45 statisticians and 118 field supervisors with an average supervisor to enumerator ratio of 1: 3.

The interviews have been done by canvassing from house-to-house. Information was collected directly from the head of the household or from other responsible household member. The enumerators were assigned to two sampled enumeration area, where he/she could easily converse with the respondent's dialect. However, in some cases whenever enumerator faces difficulty to converse with the respondent's dialect interpreters were used to facilitate the data collection.

Data collection from the field took place from Feb 26 – March 27, 2015. During the survey data collection operation, close and regular supervision was undertaken at various levels. Immediately after the commencement of the data collection exercise the supervisors had made spot checking, re-interviewing and a thorough scrutiny of filled-in questionnaires to ensure that the data collection activities are taking place according to the given instruction. The branch statistical office heads, coordinators and statisticians were also made supervision of the data collection operation. Moreover, senior staff members from the CSA head office made field supervision activities in seven branches for a week at different towns.

All branch statistical offices of the CSA participated in the survey process undertaken, that is, in organizing the field survey in deploying the field staff to their respective sites, and retrieving completed questionnaires and submitting them to the head office. They will also responsible in administering the financial and logistic aspect of the survey as well as field supervision within the areas of their assignment.

1.6. Data Processing

The filled-in questionnaires that were retrieved from the field were first subjected to manual editing and coding. During the fieldwork, field supervisors and statisticians of the head and branch statistical offices have checked the filled-in questionnaires and carried out some editing. However, the major editing and coding operation was carried out at the head office. All the edited questionnaires were again fully verified and checked for consistency before they were submitted to the data entry by the subject matter experts.

Using the computer edit specifications prepared earlier for this purpose, the entered data were checked for consistencies and then computer editing or data cleaning was made by referring back to the filled-in questionnaire. This is an important part of data processing operation to maintain the quality of the data. Consistency checks and rechecks were also made based on frequency and tabulation results. This was done by senior programmers using CSPro software in collaboration with the senior subject matter experts from Manpower Statistics Team of the CSA.

1.7. Definitions of Concepts and Terms

Some of the major operational definitions of basic concepts are given below. Moreover, other concepts and definitions related to the economic activity status are provided under respective chapters.

<u>Urban Center:</u> is defined as a locality with 2000 or more inhabitants. In this survey, however, for practical purposes an urban center includes the following regardless of the number of inhabitants.

- a) All administrative capitals
 - i) Regional capitals
 - ii) Zonal capitals not included in (i)
 - iii) Wereda capitals not included in (i) and (ii)
 - iv) Localities with Urban Dweller's Association (UDAs) not included in (I-iii).
- b) Municipal town not included in item (a) above.
- c) All localities which are not included either in item (a) or (b) above, having a population of 1000 or more persons, and whose inhabitants are primarily engaged in non-agricultural activities. Note that localities with population less than 1000 persons should be considered as rural.

<u>Major Urban Centers</u>: For the purpose of this study, major urban center include those selected urban centers with population 100,000 and above. It also includes region capitals irrespective of their population size.

<u>Urban Kebele</u>: is the lowest administrative unit in an urban center with its own jurisdiction. It is an association of urban dwellers (commonly known as kebele) formed by the inhabitants.

Enumeration Area (EA): is a unit of land delineated for the purpose of enumerating population and housing units without omission and duplication. An EA in rural areas usually consists of 150-200 households. On the other hand, an EA in urban areas constitutes 150-200 housing units.

<u>Household:</u> Consists of a person or group of persons, irrespective of whether related or not, who normally live together in the same household and housing units and have common cooking and eating arrangements.

Housing Unit: is defined as a separate and independent place of abode either intended for habitation or not intended for habitation but was occupied as a living quarter by a household during the survey period. Although intended for habitation by one household, a housing unit may, at the survey be occupied by one or more households or may be used partly for living and partly for establishment.

<u>Head of Household</u>: is a person who provides economic supports or manages the households. The head of the household is selected by household members for some reasons of his age or respect regardless of their sex.

<u>Usual Member of a Household</u>: a person is considered as usual members of a household if he or she is:-

- a) a person who continuously live/reside at least for six months and have a common cooking and eating arrangements with the household; or
- b) a person who is temporarily absent from the household at the time of the survey but his absence has not violated the six months criterion.
- c) House maids, guards, baby sitters, etc...with no other dwelling and who were staying with the household at the time of the survey.
- d) Persons who plan to live more than six months due to searching for job and transfer from job ...etc.

1.8. Organization of the Report

This statistical report contains seven chapters. The first chapter covers background and objectives of the survey, where the objectives and the contents of the survey questionnaire, training of field staff, organization of field work, concepts and definitions of terms and concepts. Chapter II presents methodology of the survey including scope, coverage, sample size, covered households and response rate. Chapter III describes briefly the socio-demographic characteristics of the urban population such as the size and distribution of urban population by age group and sex. Chapter IV presents data on the economic activity status of the population aged ten years and above using the current status approach. Chapter V explores the size, characteristics and distribution of the employed population aged ten years and above. Chapter VI presents the size, characteristics and distribution of the unemployed population and unemployment rate aged ten years and above.

Annex I provides survey questionnaire English version, while estimation procedures of total, ratio and sampling errors; and estimates and CV's are also presented in Annex II and Annex III, respectively.

CHAPTER II

SURVEY METHODOLOGY

2.1 COVERAGE

The 2015 Urban Employment and Unemployment Survey (UEUS) covered all urban parts of the country except three zones of Afar, Six zones of Somali, where the residents are pastoralists. This survey follows household approach and covers households residing in conventional households, thus population residing in the collective quarters such as universities/colleges, hotel/hostel, monasteries and homeless population etc are not covered by this survey.

It was initially planned to cover 660 EAs and 19,800 households in the survey, but ultimately 19,764 (99.8%) of households were successfully covered.

2.2 SAMPLING FRAME

The list of households obtained from the 2007 population and housing census is used to select EAs. A fresh list of households from each EA was prepared at the beginning of the survey period. The list was then used as a frame in order to select 30 households from sample EAs.

2.3 SAMPLE DESIGN

For the purpose of this survey, the country was divided into two broad categories. That is major urban center and other urban center categories.

Category I:- Major urban centers:- In this category all regional capitals and five other major urban centers that have had a high population size as compared to others were included. Each urban center in this category was considered as a reporting level. The category has a total of 16 reporting levels. In this category, in order to select the sample, a stratified two-stage cluster sample design was implemented. The primary sampling units were EAs of each reporting level. From each sample EA 30 households were then selected as a Second Stage Unit (SSU).

Category II: - Other urban centers: Urban centers in the country other than those under category I were grouped into this category. A domain of other urban centers is formed for each region. Consequently 8 reporting levels were formed in this category. Harari, Addis Ababa and Dire Dawa do not have urban centers other than that grouped in category I. Hence, no domain was formed for these regions under this category.

A stratified three stage cluster sample design was also adopted to select samples from this category. The primary sampling units were urban centers and the second stage sampling units were EAs. From each EA 30 households were selected at the third stage and the survey questionnaires administered for all of them.

2.4 SAMPLE SIZE AND SELECTION SCHEME

Category I: - In this category 393 EAs and 11,790 households were selected from 16 major urban centers. Sample EAs from each reporting level in this category were selected using probability proportional to size /PPS/ with systematic sampling method; size being number of households obtained from the 2007 population and housing census. From the fresh list of households prepared at the beginning of the survey, 30 households per EA were selected using systematic random sampling techniques /SRS/.

Category II:- consists of 86 urban centers, 267 EAs and 8,010 households were selected in this category. Urban centers from each domain and EAs from each urban center were selected using probability proportional to size /PPS/ with systematic sampling method; size being number of households obtained from the 2007 Population and housing census. From the listing of households in each EA then 30 households were selected using systematic random sampling techniques /SRS/ and the survey questionnaire was administered for the 30 sample households.

The distribution of planned and covered EAs and households and the Estimation procedures are given in the appendix.

Appendix I: Number of planned and actually covered sampling units (EAs and households) of the 2015 (2007 E.C) Urban Employment and Unemployment Survey (UEUS).

	Stratum	Enumerat	ion Areas	Households			
	Major Urban/	Planned	Covered	Planned	Covered		
	Other Urban						
Tigray	Mekele	23	23	690	690		
	Tigray other urban	28	28	840	840		
Affar	Asayita	15	15	450	448		
	Affar other urban	14	14	420	419		
Amhara	Bahir dar	23	23	690	690		
	Gonder	23	23	690	690		
	Dessie	23	23	690	688		
	Amhara other urban	54	54	1620	1618		
Oromiya	Debrezeit/Bishoftu/	22	22	660	659		
	Nazreth/Adama/	23	23	690	689		
	Jimma	23	23	690	689		
	Shashmene	23	23	690	690		
	Oromiya other urban	83	83	2490	2487		
Somalie	Jijiga	20	20	600	600		
	Somalie other urban	15	15	450	450		
Benishangul -	Asosa	15	15	450	448		
Gumuz	Benishangul -Gumuz	13	13	390	390		
	other urban						
S.N.N.P.	Awassa	23	23	690	690		
	S.N.N.P other urban	51	51	1530	1529		
Gambela	Gambella	14	15	450	449		
	Gambella other urban	9	8	240	240		
Harari	Hareri	23	23	690	689		
Addis Ababa City	Addis ababa	75	75	2250	2,247		
Admini.							
Dire Dawa Admini.	Diredawa	25	25	750	735		
Total		660	660	19,800	19,764		

CHAPTER III

SOCIO-DEMOGRAPHIC CHARACTERISTICS OF URBAN POPULATION

3.1 Introduction

The figure in this chapter refers to the dejure population residing in conventional households. The dejure population comprises all persons who belong to a given area at a given time by virtue of usual residence. Thus, excluded are visitors, persons residing in collective quarters (hotel/hostel, boarding schools, prisons...etc.) as well as homeless persons.

This chapter presents the basic socio-demographic characteristics of urban population of the country which includes estimates of population size; distribution of population by age and sex, sex ratio, relationship to the head of household, Number of households and household size, literacy status, attending formal or informal education, educational attainment, training status and type of training.

3.2. Age-Sex Structure of Urban Population and Other Socio-Demographic Data

This section portrays a set of tables that depict the main socio-demographic characteristics of the urban population as of . The size and distribution of estimated total population of urban areas by age group, sex and region is shown in Table 3.1. The total urban population of the country estimated to be 16,598,199 in Feb. 2015. Table 3.2 presents the distribution of urban population, number of households and average household size by region. The average household size is computed from the total population to that of the total households. Data on household size and distribution of persons per household can sometimes be used as a proxy measure of crowdedness of population and has a great implication to health and housing problem. The distribution of urban population by region, sex and the relationship of the respondents to the head of household is presented in Table 3.3.

Table 3.4 and Table 3.5 show the status of school attendance for those aged five years and above by age group and sex at national and regional levels. The literacy status and educational attainment of urban population aged ten years and above has shown in Table 3.6 and Table 3.7, respectively. As indicated in these tables secondary education refers to those persons who did not complete high school education, that is, grades 9 - 11 in old curriculum and grade 9 in new curriculum, while secondary education completed covers those persons who completed high school education which includes grade 12 in old curriculum and grade 10 in new curriculum.

The 2015 Urban Employment and Unemployment Survey also collected data on status and types of training of urban population aged ten years and above. Tables 3.8 reveal the status and type of training received by region and sex.

Besides, the distribution of urban population by five-year age group, sex and sex ratio is presented in Table 3.9. Sex ratio is defined as the number of males per 100 females in a given population. The sex composition has an effect on economic activities through changing the relative size of the working population.

Table 3.1: Estimated Total Population of Urban Areas by Age Group, Sex and Region: 2015

Age Group						Regi						
and Sex						_	Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
All Ages												
Total	16,598,199	1,217,967	283,362	3,350,976	4,937,029	414,693	196,030	2,364,569	124,936	130,433	3,295,206	282,999
Male	7,839,679	548,252	140,426	1,552,640	2,381,342	203,978	96,265	1,139,522	60,086	61,083	1,519,949	136,137
Female	8,758,520	669,715	142,936	1,798,336	2,555,687	210,715	99,764	1,225,046	64,850	69,351	1,775,257	146,862
0-4												
Total	1,704,779	147,209	35,494	348,985	555,717	73,421	20,028	229,852	15,268	11,506	241,477	25,823
Male	875,713	79,645	17,995	175,374	280,604	38,285	9,695	123,437	8,015	5,535	125,196	11,934
Female	829,066	67,564	17,499	173,611	275,113	35,137	10,333	106,415	7,254	5,971	116,281	13,888
5-9												
Total	1,512,044	118,388	30,885	279,405	483,146	67,535	18,292	247,024	14,722	12,431	209,334	30,883
Male	740,230	57,024	13,659	136,274	237,646	33,933	9,487	118,665	6,267	5,729	104,833	16,712
Female	771,814	61,363	17,226	143,131	245,501	33,601	8,804	128,359	8,455	6,702	104,501	14,170
10-14												
Total	1,790,429	116,810	29,479	352,195	563,599	52,965	22,907	301,593	16,286	14,752	288,827	31,017
Male	858,124	56,402	15,835	176,748	272,782	27,439	9,561	134,093	7,967	8,160	132,572	16,564
Female	932,305	60,408	13,643	175,448	290,817	25,526	13,346	167,500	8,319	6,592	156,254	14,452
15-19												
Total	2,041,908	154,538	28,316	410,971	632,697	47,750	23,054	320,875	15,992	13,807	358,193	35,714
Male	853,760	67,758	11,399	164,308	275,662	22,099	11,884	132,561	7,729	6,411	136,563	17,386
Female	1,188,148	86,780	16,918	246,663	357,035	25,651	11,170	188,314	8,264	7,396	221,629	18,328
20-24												
Total	2,011,484	146,474	30,313	425,178	602,264	35,088	26,866	277,103	16,983	11,449	410,305	29,463
Male	836,505	44,307	16,846	172,718	255,222	15,543	10,504	125,632	7,264	4,494	170,339	13,637
Female	1,174,979	102,167	13,466	252,460	347,041	19,546	16,361	151, 4 71	9,719	6,955	239,966	15,826
25-29												
Total	2,024,794	143,397	36,477	412,707	608,648	34,163	27,224	279,538	15,908	13,804	423,669	29,258
Male	978,477	63,153	15,061	203,310	302,293	16,071	15,121	139,390	7,715	6,887	195,870	13,607
Female	1,046,317	80,244	21,416	209,397	306,355	18,093	12,103	140,148	8,193	6,917	227,799	15,652
30-34												
Total	1,276,614	101,199	21,738	252,372	349,526	22,497	14,432	149,010	10,160	9,960	327,199	18,521
Male	663,030	52,212	12,072	127,941	191,280	10,907	7,513	79,612	4,981	4,839	162,345	9,328
Female	613,583	48,987	9,666	124,431	158,246	11,590	6,919	69,398	5,179	5,120	164,854	9,193

Table 3.1: Cont'd

Age Group						Reg						
and Sex							Benishangul-				Addis	Dire
and ocx	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
35-39												
Total	1,174,449	71,373	22,868	225,093	334,301	21,200	13,439	163,312	6,677	11,642	284,287	20,257
Male	559,061	30,622	12,303	106,123	159,482	9,202	7,116	76,667	3,941	4,275	140,316	9,013
Female	615,388	40,751	10,565	118,969	174,819	11,998	6,323	86,645	2,735	7,367	143,971	11,244
40-44												
Total	713,737	39,684	12,518	129,669	210,778	18,841	7,539	109,822	3,861	7,203	160,599	13,224
Male	397,066	22,539	6,660	74,582	121,589	8,729	3,468	62,386	2,091	3,901	85,122	5,997
Female	316,672	17,144	5,858	55,087	89,189	10,112	4,070	47,436	1,770	3,302	75,476	7,227
45-49												
Total	561,559	35,537	10,638	109,362	163,564	12,348	7,049	75,533	3,342	6,005	122,300	15,880
Male	278,068	16,500	5,117	50,160	79,937	6,989	4,169	42,846	1,344	3,201	60,128	7,677
Female	283,491	19,038	5,522	59,202	83,627	5,359	2,880	32,687	1,997	2,805	62,172	8,203
50-54												
Total	422,014	31,595	8,602	92,623	95,344	8,798	4,966	56,464	2,213	5,044	108,325	8,041
Male	195,936	17,425	4,382	34,403	50,811	4,762	2,842	29,101	1,201	2,573	45,443	2,990
Female	226,078	14,170	4,219	58,220	44,533	4,036	2,124	27,362	1,012	2,470	62,882	5,051
55-59												
Total	345,943	21,438	6,466	79,419	87,204	3,218	3,918	39,865	1,206	3,020	92,115	8,074
Male	154,230	7,149	2,348	35,936	39,523	1,885	1,936	22,323	744	1,297	37,064	4,024
Female	191,713	14,289	4,118	43,483	47,681	1,333	1,982	17,542	462	1,723	55,051	4,050
60-64												
Total	310,206	22,037	5,062	60,732	87,711	6,185	1,778	34,855	870	3,292	82,066	5,618
Male	126,910	8,081	3,307	25,448	37,757	3,231	759	10,038	375	1,531	33,743	2,639
Female	183,297	13,957	1,755	35,285	49,954	2,954	1,018	24,817	495	1,761	48,323	2,979
65+												
Total	708,239	68,288	4,506	172,264	162,531	10,685	4,539	79,724	1,448	6,518	186,510	11,227
Male	322,571	25,436	3,439	69,316	76,752	4,905	2,209	42,771	453	2,249	90,413	4,628
Female	385,668	42,852	1,066	102,948	85,778	5,780	2,330	36,953	995	4,268	96,097	6,599

Table 3.2 Distribution of Urban Population of Regions by Number of Households and Average Household Size: 2015

Table 3.2 Distribution	of Urban Po _l	pulation (or Regions	ру митре			Average Ho	ousenoia S	IZe: 2015	1		Λικοποσο
			:1	1	Hou	sehold size					40.0	Average
Region	T - 1 - 1		0	0	4	_	0	7	0	0	10 &	Household
_	Total	1	2	3	4	5	6	7	8	9	above	Size
COUNTRY TOTAL												
Household	4,629,670	800,114	831,684	869,970	771,822	547,421	385,586	203,234	106,371	59,358	54,110	
Population	16,598,199	800,114	1,663,367	2,609,909	3,087,288	2,737,107	2,313,518	1,422,639	850,970	534,222	579,066	
TIGRAY REGION												
Household	368,415	66,848	81,823	73,069	57,872	42,176	21,884	12,217	7,734	2,292	2,499	3.3
Population	1,217,967	66,848	163,647	219,207	231,489	210,882	131,301	85,521	61,871	20,631	26,569	
AFFAR REGION												
Household	87,044	20,642	14,817	17,112	13,410	9,372	6,576	2,245	911	1,018	942	3.3
Population	283,362	20,642	29,634	51,337	53,639	46,861	39,458	15,715	7,288	9,158	9,630	
AMHARA REGION												
Household	1,029,373	204,325	224,152	193,478	162,277	114,214	73,083	29,700	14,872	5,150	8,121	3.3
Population	3,350,976	204,325	448,304	580,435	649,108	571,072	438,500	207,897	118,979	46,349	86,007	
OROMIA REGION												
Household	1,361,779	216,424	234,530	271,426	244,392	155,537	114,129	60,407	31,756	20,290	12,889	3.6
Population	4,937,029	216,424	469,061	814,277	977,569	777,685	684,772	422,847	254,046	182,606	137,741	
SOMALI REGION												
Household	88,199	12,573	10,573	11,223	11,728	8,678	9,676	8,847	6,293	2,879	5,730	4.7
Population	414,693	12,573	21,145	33,670	46,912	43,390	58,053	61,927	50,345	25,911	60,765	
BENISHANGUL-GUMUZ RE	EGION											
Household	65,608	17,642	13,863	11,736	8,082	7,021	4,361	1,947	585	209	163	3.0
Population	196,030	17,642	27,726	35,209	32,327	35,104	26,164	13,627	4,683	1,879	1,669	
S.N.N.P REGION												
Household	655,420	142,951	100,940	102,532	97,113	81,735	62,356	32,144	18,085	10,380	7,183	3.6
Population	2,364,569	142,951	201,880	307,596	388,454	408,675	374,136	225,011	144,679	93,422	77,765	
GAMBELLA REGION	. ,	,	,	ŕ	•	•	•	•	,		•	
Household	34,667	7,499	6,217	5,709	4,745	3,706	2,827	1,617	1,084	374	889	3.6
Population	124,936	7,499	12,434	17,127	18,979	18,531	16,960	11,322	8,674	3,366	10,045	
HARARI REGION	,	,	,	,	•	•	•	•	,	,	,	
Household	39,194	7,316	7,439	9,164	6,644	3,649	2,044	1,385	1,004	88	461	3.3
Population	130,433	7,316	14,878	27,492	26,574	18,247	12,263	9,698	8,033	793	5,139	
ADDIS ABABA CITY ADMIN	•	,	,	, -	-,-	-,	,	-,	-,		-,	
Household	827,364	92,256	125,356	160,503	154,575	112,850	83,241	49,140	21,695	14,899	12,847	4.0
Population	3,295,206	92,256	250,713	481,508	618,301	564,252	499,449	343,977	173,562	134,093	137,095	
DIRE DAWA ADMINISTRAT		5=,=50	_00,0	.5.,500	0.0,001	33.,202	.55, . 10	0.0,011	0,002	,	70.,000	
Household	72,607	11,639	11,973	14,017	10,984	8,481	5,410	3,585	2,351	1,779	2,387	3.9
Population	282,999	11,639	23,946	42,050	43,936	42,407	32,461	25,096	18,810	16,015	26,640	

Table 3.3 Distribution of Urban Population of Regions by Relationship to the Head of Household and Sex: 2015

		Relationship to the head of household											
Region and Sex	Total	Head of Household	Spouse	Son/ daughter of head and spouse	Son/ Daughter of head	Son/ daughter of spouse	Mother/ father of head/ spouse	Sister/ brother of head/ spouse	Domestic workers (include guard, driver)	Other- relatives	Non- relatives		
COUNTRY TOTA	\L	•	•	•		•	•	•	, , ,				
Total	16,598,199	4,629,670	2,403,555	4,838,739	2,114,329	81,733	126,519	593,531	230,559	1,354,498	225,067		
Male	7,839,679	2,984,381	194,148	2,537,332	1,100,529	41,102	22,729	258,060	23,969	571,779	105,650		
Female	8,758,520	1,645,289	2,209,407	2,301,407	1,013,800	40,631	103,790	335,470	206,590	782,719	119,417		
TIGRAY REGION	1												
Total	1,217,967	368,415	172,707	329,816	175,182	7,742	8,271	35,448	18,537	93,235	8,613		
Male	548,252	206,182	11,412	172,426	89,809	4,036	1,240	14,735	1,235	42,814	4,364		
Female	669,715	162,234	161,295	157,390	85,373	3,707	7,031	20,713	17,302	50,420	4,250		
AFFAR REGION													
Total	283,362	87,044	48,481	92,518	24,065	1,618	1,350	5,482	2,070	18,211	2,522		
Male	140,426	64,102	2,161	46,612	12,275	856	737	2,911	585	8,530	1,657		
Female	142,936	22,943	46,320	45,906	11,790	761	613	2,571	1,485	9,681	865		
AMHARA REGIO													
Total	3,350,976	1,029,373	499,625	885,937	442,455	22,789	29,529	108,417	42,717	246,796	43,338		
Male	1,552,640	633,641	34,230	467,165	230,915	10,564	3,873	45,282	6,528	100,828	19,613		
Female	1,798,336	395,732	465,394	418,772	211,540	12,225	25,656	63,134	36,189	145,968	23,725		
OROMIA REGIO													
Total	4,937,029	1,361,779	765,250	1,495,415	572,182	19,077	38,961	170,883	40,887	392,913	79,683		
Male	2,381,342	951,351	46,558	790,947	293,046	11,619	4,192	70,192	3,226	170,744	39,467		
Female SOMALI REGION	2,555,687 N	410,428	718,692	704,468	279,136	7,458	34,769	100,691	37,660	222,169	40,216		
Total	414,693	88,199	50,493	168,605	68,748	174	2,423	9,616	2,651	20,756	3,027		
Male	203,978	57,721	3,602	86,243	38,516	134	773	4,852	40	10,564	1,533		
Female	210,715	30,479	46,891	82,362	30,232	40	1,650	4,764	2,610	10,192	1,494		

21

Table 3.3 Cont'd

Table 3.3 Cont'd	1										
					ionship to th	e head of ho	usehold				
Region and Sex				Son/ daughter of head	Son/	Son/	Mother/ father of	Sister/ brother of	Domestic workers (include		
		Head of		and	Daughter	daughter of	head/	head/	guard,	Other-	Non-
	Total	Household	Spouse	spouse	of head	spouse	spouse	spouse	driver)	relatives	relatives
BENISHANGUL-GU	MUZ REGION										
Total	196,030	65,608	33,593	55,586	17,332	767	1,289	6,420	1,783	9,026	4,625
Male	96,265	46,132	1,597	28,389	9,273	511	282	2,757	163	4,607	2,555
Female	99,764	19,476	31,996	27,197	8,060	256	1,007	3,663	1,620	4,419	2,070
S.N.N.P REGION											
Total	2,364,569	655,420	334,022	723,254	246,686	11,957	11,473	82,297	34,864	234,414	30,183
Male	1,139,522	460,969	15,385	373,197	132,644	7,291	1,206	35,758	8,450	89,778	14,843
Female	1,225,046	194,451	318,637	350,056	114,042	4,665	10,267	46,539	26,414	144,635	15,339
GAMBELLA REGIOI	N										
Total	124,936	34,667	15,138	29,086	26,024	463	1,145	5,347	582	10,709	1,776
Male	60,086	19,978	1,952	14,380	12,827	305	40	3,121	204	6,019	1,260
Female	64,850	14,688	13,186	14,706	13,197	158	1,105	2,226	378	4,690	516
HARARI REGION											
Total	130,433	39,194	19,393	32,913	20,820	581	1,278	3,364	1,078	10,726	1,086
Male	61,083	23,710	1,269	17,991	11,023	234	274	1,390	-	4,928	263
Female	69,351	15,484	18,125	14,922	9,798	347	1,004	1,974	1,078	5,798	822
ADDIS ABABA CITY	ADMINISTRA										
Total	3,295,206	827,364	428,304	949,764	474,739	15,581	28,016	155,910	80,637	288,345	46,549
Male	1,519,949	476,753	72,970	497,621	244,940	5,099	9,841	72,577	3,451	118,554	18,143
Female	1,775,257	350,611	355,334	452,143	229,798	10,481	18,175	83,332	77,186	169,791	28,405
DIRE DAWA ADMIN	, ,	, -	,	- ,	-,	-,	-, -	· - ,	,	,	-,
Total	282,999	72,607	36,550	75,846	46,096	984	2,784	10,346	4,753	29,368	3,665
Male	136,137	43,844	3,013	42,361	25,262	452	272	4,484	86	14,413	1,951
Female	146,862	28,763	33,537	33,485	20,835	532	2,512	5,862	4,666	14,955	1,714

Table 3.4 Distribution of Urban Population Aged Five Years and Above by Age Group, Sex and School Attendance-COUNTRY TOTAL: 2015

Age Group	Attending Status									
and Sex	All	Currently	Attended in	Never						
	Persons	Attending	the past	Attended						
All Ages		J	,							
Total	14,893,420	5,273,681	7,423,409	2,196,330						
Male	6,963,966	2,551,500	3,842,458	570,008						
Female	7,929,454	2,722,181	3,580,951	1,626,322						
5-9	1,020,101	2,722,101	0,000,001	1,020,022						
Total	1,512,044	1,250,660	31,258	230,125						
Male	740.230	621,559	16,783	101,888						
Female	771,814	629,101	14,476	128,237						
10-14	,	,	,	,						
Total	1,790,429	1,666,288	69,640	54,502						
Male	858,124	813,267	27,604	17,253						
Female	932,305	853,020	42,036	37,249						
15-19	,,,,,,,	,-	,	- ,						
Total	2,041,908	1,424,884	538,575	78,450						
Male	853,760	647,695	188,670	17,395						
Female	1,188,148	777,189	349,905	61,054						
20-24	, ,	•	•	•						
Total	2,011,484	512,049	1,361,941	137,494						
Male	836,505	238,435	570,584	27,487						
Female	1,174,979	273,615	791,358	110,007						
25-29										
Total	2,024,794	227,480	1,560,383	236,931						
Male	978,477	120,380	807,668	50,428						
Female	1,046,317	107,100	752,714	186,503						
30-34										
Total	1,276,614	95,207	1,002,567	178,839						
Male .	663,030	50,739	563,272	49,019						
Female	613,583	44,468	439,295	129,821						
35-39	4 474 440	40.000	045.000	044.000						
Total	1,174,449	43,833	915,623	214,992						
Male	559,061	23,637	485,047	50,377						
Female 40-44	615,388	20,196	430,577	164,615						
Total	713,737	22,520	558,959	132,258						
Male	397,066	14,558	349,836	32,672						
Female	316,672	7,962	209,123	99,586						
45-49	010,072	7,502	200,120	33,300						
Total	561,559	10,480	431,623	119,456						
Male	278,068	8,774	248,932	20,361						
Female	283,491	1,706	182,691	99,095						
50-54	200, 10 1	1,700	102,001	00,000						
Total	422,014	7,355	271,736	142,923						
Male	195,936	4,633	159,538	31,766						
Female	226,078	2,722	112,199	111,157						
55-59	,	•	•	•						
Total	345,943	3,986	221,286	120,670						
Male	154,230	1,874	127,062	25,294						
Female	191,713	2,113	94,224	95,376						
60-64										
Total	310,206	1,536	167,903	140,767						
Male	126,910	1,536	96,835	28,539						
Female	183,297	-	71,069	112,228						
65+	•		,	•						
Total	708,239	7,403	291,914	408,923						
Male	322,571	4,413	200,629	117,530						
Female	385,668	2,990	91,285	291,393						

Table 3.5 Distribution of Urban Population Aged Five Years and Above by School Attendance, Region and Sex: 2015

Above by School Attendance, Region and Sex: 2015 Attended Formal/informal Education												
	A	ttended Formal	/informal Educati	on								
Region and Sex	All Persons	Currently Attending	Attended in	Never Attended								
COUNTRY TOTAL	VII 1-6190118	Auchaling	the past	Auchided								
	44.000.400	E 070 C04	7 400 400	0.400.000								
Total	14,893,420	5,273,681	7,423,409	2,196,330								
Male	6,963,966	2,551,500	3,842,458	570,008								
Female	7,929,454	2,722,181	3,580,951	1,626,322								
TIGRAY REGION	4 070 750			450 400								
Total	1,070,758	363,055	551,511	156,192								
Male	468,608	174,257	261,072	33,278								
Female	602,151	188,798	290,439	122,913								
AFFAR REGION												
Total	247,868	85,319	103,068	59,480								
Male	122,431	42,563	59,046	20,822								
Female	125,437	42,756	44,022	38,659								
AMHARA REGION												
Total	3,001,991	957,793	1,499,884	544,314								
Male	1,377,266	472,381	760,138	144,747								
Female	1,624,725	485,412	739,746	399,567								
OROMIA REGION												
Total	4,381,312	1,554,433	2,169,693	657,185								
Male	2,100,738	764,426	1,156,613	179,699								
Female	2,280,574	790,007	1,013,080	477,487								
SOMALI REGION												
Total	341,272	129,457	87,726	124,088								
Male	165,694	70,837	53,736	41,121								
Female	175,578	58,620	33,990	82,968								
BENISHANGUL-GUI		55,5=5	,	5=,555								
Total	176,002	61,819	82,136	32,046								
Male	86,570	31,766	45,699	9,105								
Female	89,432	30,053	36,437	22,942								
S.N.N.P. REGION	00, 102	30,000	00, 101	22,012								
Total	2,134,717	855,308	1,016,018	263,391								
Male	1,016,085	397,383	550,469	68,234								
Female	1,118,631	457,925	465,549	195,157								
GAMBELLA REGION		407,020	400,040	100,107								
Total	109,667	51,356	43,625	14,686								
Male	52,071	25,337	23,645	3,089								
Female	57,596	26,019	19,980	11,597								
HARARI REGION	57,590	20,019	19,900	11,591								
Total	118,927	44 005	62 620	12 202								
		41,905	63,630	13,392								
Male	55,548	21,140	32,276	2,132								
Female	63,379	20,765	31,354	11,260								
ADDIS ABABA CITY			4 000 000	004.700								
Total	3,053,729	1,078,015	1,680,986	294,729								
Male	1,394,753	498,563	835,887	60,302								
Female	1,658,977	579,451	845,098	234,427								
DIRE DAWA ADMIN			,									
Total	257,177	95,219	125,130	36,827								
Male	124,202	52,846	63,876	7,481								
Female	132,974	42,374	61,254	29,346								

Table 3.6 Distribution of Urban Population of Regions Aged Ten Years and Above by Literacy Status and Sex: 2015

Table 3.0 Distribution of Orbar		. c. itogiono	7.gca .c		teracy Status	oracy Clara	Junu John I	<u> </u>	
Region		All Persons			All Literate		,	All Illiterate	
	Total	Male	Female	Total	Male	Female	Total	Male	Female
COUNTRY TOTAL	13,381,376	6,223,736	7,157,639	11,118,528	5,674,621	5,443,907	2,262,848	549,115	1,713,733
TIGRAY REGION	952,370	411,583	540,787	778,486	378,703	399,783	173,885	32,881	141,004
AFAR REGION	216,983	108,772	108,211	155,842	88,831	67,011	61,141	19,941	41,200
AMHARA REGION	2,722,586	1,240,992	1,481,594	2,121,937	1,093,440	1,028,497	600,649	147,552	453,097
OROMIA REGION	3,898,166	1,863,092	2,035,074	3,247,298	1,696,461	1,550,836	650,868	166,631	484,237
SOMALI REGION BENISHANGUL-GUMUZ	273,737	131,760	141,976	169,838	100,395	69,444	103,899	31,366	72,533
REGION	157,710	77,083	80,627	121,656	67,775	53,881	36,054	9,308	26,746
S.N.N.P REGION	1,887,693	897,420	990,273	1,615,311	824,989	790,322	272,382	72,431	199,951
GAMBELLA REGION	94,945	45,805	49,141	78,820	41,965	36,855	16,125	3,840	12,285
HARARI REGION ADDIS ABABA CITY	106,496	49,819	56,677	92,107	47,697	44,410	14,388	2,122	12,266
ADMINISTRATION	2,844,395	1,289,919	1,554,476	2,552,668	1,234,980	1,317,688	291,727	54,939	236,788
DIRE DAWA ADMINISTRATION	226,294	107,490	118,804	184,564	99,385	85,179	41,729	8,104	33,625

Table 3.7 Distribution of Urban Population Aged Ten Years and Above by Educational Attainment, Age Group and Sex - COUNTRY - TOTAL: 2015

	COUNTINI	TOTAL	0.0									
	Highest Grade Completed Diploma or											
Age Group and Sex	Total	Non Formal	Pre school	Grades 1-8	Secondary not completed	Secondary Completed	Preparatory	Certificate Not Completed	Certificate Completed	Diploma or Degree Not Completed	Diploma Completed	Degree and above Completed
All ages	i otai	1 Omman	0011001		roompiotou	Completed	roparatory	Completed	Completed	Completed	Completed	Completed
Total	11,418,158	332,024	53,243	5,554,955	1,181,522	1,718,812	348,006	59,370	165,382	295,806	959,113	749,925
Male	5,757,742	152,193	22,645	2,642,602	608,233	891,040	177,589	27,800	78,208	144,957	482,613	529,862
Female	5,660,416	179,832	•	2,912,353	573,289	827,772	170,417	31,570	87,174	150,849	476,500	220,063
10-14	, ,	•	·	, ,	•	•	•		·	·	•	·
Total	1,737,163	3,676	29,649	1,678,620	23,733	1,485	-	-	-	-	-	-
Male	841,937	2,070	13,681	814,088	11,515	583	-	-	-	-	-	-
Female	895,226	1,606	15,969	864,531	12,218	902	-	-	-	-	-	-
15-19												
Total	1,963,839	2,962	4,036	1,051,128	394,403	300,173	118,220	16,774	17,155	43,617	15,371	-
Male	836,439	2,218	827	436,901	189,094	123,220	51,117	6,079	7,365	15,270	4,348	-
Female	1,127,399	744	3,209	614,227	205,310	176,953	67,103	10,694	9,790	28,347	11,023	-
20-24												
Total	1,874,671	4,461	1,336	615,445	210,914	405,373	105,068	23,505	52,771	130,081	225,422	100,296
Male .	809,532	1,991	508	251,021	94,896	181,534	52,456	10,933	22,410	57,339	85,405	51,037
Female	1,065,140	2,470	829	364,425	116,017	223,839	52,611	12,571	30,360	72,742	140,017	49,259
25-29	4 700 000	40.400	4 400	F70 440	400 005	240.005	CO 000	44 500	20.004	74.000	270 557	227.400
Total Male	1,788,289 928,475	13,108 7,403	1,492 632	573,443 263,267	160,285 83,690	348,925 186,001	60,823 34,350	11,500 6,624	39,094 15,625	71,960 41,708	270,557 130,484	237,100 158,690
Female	859,814	5,705	860	310,176	76,596	162,924	26,473	4,876	23,469	30,252	140,073	78,411
30-34	059,014	3,703	000	310,170	70,590	102,924	20,473	4,070	23,409	30,232	140,073	70,411
Total	1,097,821	14,404	3,356	404,119	111,549	203.809	36,477	4,915	16,663	17,237	145,298	139,994
Male	614,058	7,902	1,747	204,586	55,814	113,798	21,184	3,277	8,064	9,521	81,268	106,897
Female	483,763	6,501	1,610	199,533	55,735	90,011	15,293	1,639	8,599	7,716	64,030	33,096
35-39	,	-,	,	,	,	,-	-,	,	-,	, -	- ,	,
Total	959,457	22,329	3,153	402,087	107,363	173,410	18,939	529	11,487	10,906	117,085	92,169
Male	508,684	10,321	1,132	193,126	56,015	92,979	12,388	136	7,038	7,119	64,285	64,146
Female	450,773	12,008	2,021	208,961	51,348	80,431	6,551	393	4,449	3,787	52,800	28,023

25

Table 3.7 Cont'd

						Highest (Grade Comple	eted				
Age Group and Sex	Total	Non Formal	Pre school	Grades 1-8	Secondary not completed	Secondary Completed	Preparatory	Certificate Not Completed	Certificate Completed	Diploma or Degree Not Completed	Diploma Completed	Degree and above Completed
40-44												
Total	581,696	33,884	1,660	235,708	61,367	107,345	4,020	340	5,320	7,796	60,501	63,757
Male	364,394	17,426	877	138,186	41,923	69,843	3,580	40	2,283	4,406	34,358	51,471
Female	217,303	16,458	783	97,522	19,444	37,502	440	299	3,037	3,390	26,142	12,287
45-49												
Total	442,103	40,846	827	182,249	45,109	68,565	2,045	459	6,708	6,128	39,545	49,621
Male	257,706	11,435	185	93,199	30,994	47,483	1,311	356	4,158	4,833	23,755	39,997
Female	184,396	29,411	642	89,050	14,115	21,082	734	103	2,550	1,295	15,790	9,624
50-54												
Total	279,091	34,607	785	102,052	29,371	41,958	963	394	5,040	4,341	27,858	31,724
Male	164,170	11,888	523	57,307	17,369	25,508	343	354	3,248	2,376	17,119	28,135
Female	114,921	22,719	262	44,745	12,001	16,450	619	40	1,791	1,965	10,739	3,589
55-59												
Total	225,272	35,288	1,394	98,169	15,189	26,301	425	-	5,757	850	26,108	15,791
Male	128,935	13,391	1,030	52,568	10,164	19,868	40	-	3,892	508	15,809	11,665
Female	96,337	21,897	364	45,601	5,025	6,433	385	-	1,864	343	10,299	4,126
60-64												
Total	169,439	34,009	2,675	80,223	8,567	16,695	125	954	3,940	1,679	13,364	7,208
Male	98,370	16,368	505	44,084	5,978	10,804	67	-	2,676	1,001	10,030	6,858
Female	71,069	17,641	2,170	36,139	2,589	5,891	57	954	1,265	678	3,334	350
65+												
Total	299,316	92,451	2,880	131,712	13,672	24,774	902	-	1,448	1,212	18,003	12,263
Male	205,042	49,778	1,000	94,269	10,781	19,420	752	-	1,448	878	15,750	10,966
Female	94,275	42,672	1,880	37,443	2,891	5,353	151	-	-	334	2,253	1,297

Table 3.8 Distribution of Urban Population of Regions Aged Ten Years and Above by Training Status, Types of Training and Sex: 2015

Table 3.8 Distribut		п г орига	LIOII OI IV	egions A	jeu ren re		Above by Irai	illing State	is, rypes	OI ITAIIII	ing and Se	X. 2013
Training status, Types		<u> </u>				R	egion				A -1 -1: -	D:
of Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
All Persons	Total	rigiay	Allal	Aiiiilala	Oronna	Juliane	Guilluz	SININFIX	Januela	i iai aii	Auaua	Dawa
Total	13,381,376	052 270	216 002	2 722 506	2 202 166	272 727	157 710	1,887,693	04 045	106 406	2,844,395	226 204
Male		•		1,240,992			77,083	897,420	45,805		1,289,919	
Female	7,157,639	540,787	108,211	1,481,594	2,035,074	141,976	80,627	990,273	49,141	56,677	1,554,476	118,804
Trained	0.404.000	000 507	04.000		700 774	04.005	00.044	445.04.4	40.000	00.444	050 047	54.005
Total	3,101,230		31,639	575,777	788,774	31,005	38,041	445,214	19,629	29,441	850,217	54,985
Male	1,892,616		21,697	351,635	498,974	23,865	24,911	278,039	13,244	19,018	491,243	35,466
Female	1,208,614	101,985	9,942	224,141	289,801	7,140	13,130	167,175	6,385	10,424	358,974	19,519
Not Trained	10 000 110	745.004	105011	0.4.40.040	0.400.000	0.40.700	440.000	4 440 470	75.040		4 00 4 470	474.000
Total	10,280,146			2,146,810			119,669	1,442,479	75,316	•	1,994,178	171,308
Male	4,331,121		87,074		1,364,119		52,172	619,381	32,560	30,801	798,676	72,024
Female	5,949,025	438,802	98,269	1,257,453	1,745,273	134,837	67,497	823,098	42,756	46,253	1,195,502	99,285
DETAILED TYPES OF T												
All TRAINED	3,101,230	•	31,639	575,777	788,774	31,005	38,041	445,214	19,629	29,441	850,217	54,985
Male	1,892,616		21,697	351,635	498,974	23,865	24,911	278,039	13,244	19,018	491,243	35,466
Female	1,208,614	101,985	9,942	224,141	289,801	7,140	13,130	167,175	6,385	10,424	358,974	19,519
Social Science												
Total	1,002,568	66,371	9,000	189,970	263,849	12,517	14,052	168,998	8,790	6,918	249,447	12,658
Male	592,876	40,146	6,593	113,524	164,772	9,786	9,562	105,869	6,238	3,844	124,712	7,830
Female	409,693	26,225	2,407	76,445	99,077	2,732	4,490	63,129	2,551	3,074	124,735	4,828
Accounting& Finance												
Total	293,908	22,846	2,839	42,116	83,079	4,098	3,093	35,349	2,410	1,435	92,881	3,763
Male	144,506	12,568	1,388	22,377	43,575	3,225	1,729	18,089	1,627	577	37,759	1,591
Female	149,402	10,278	1,450	19,739	39,504	874	1,364	17,259	783	858	55,122	2,172
Agricultural Economics												
Total	12,603	272	34	1,666	3,871	460	188	2,712	49	142	2,707	502
Male	8,278	-	34	1,071	3,315	460	188	1,809	49	142	1,032	180
Female	4,324	272	-	594	557	-	-	904	-	-	1,676	322
Archeology & Heritage												
Total	418	-	-	98	184	-	-	-	-	49	-	86
Male	135	-	-	-	-	-	-	-	-	49	-	86
Female	283	-	-	98	184	-	-	-	-	-	-	-
Banking &Insurance												
Total	8,016	-	-	94	1,711	-	-	722	_	_	5,489	_
Male	3,951	-	-	-	871	_	-	602	_	-	2,479	_
Female	4,064	-	-	94	840	-	-	121	_	-	3,010	-

Table 3.8 Cont'd

Table 3.8 Cont d	1											
Training status, Types			1	Г	1	K	egion				ا مادا: م	Dia-
of Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Business Administration	n											
Total	29,151	1,448	234	1,783	6,171	350	649	5,608	-	183	12,245	478
Male	18,859	724	234	880	4,390	310	649	4,209	-	140	6,845	478
Female	10,291	725	-	903	1,781	41	-	1,398	-	43	5,400	-
Civics & Ethics												
Total	19,985	2,564	466	7,270	952	336	546	2,728	360	166	4,425	171
Male	11,582	2,202	466	4,293	-	224	209	2,295	278	-	1,445	171
Female	8,403	362	-	2,977	952	112	337	433	82	166	2,981	-
Cooperatives Organiza												
Total	9,862	273	119	874	5,833	268	172	989	192	-	963	180
Male	7,154	273	119	874	3,747	216	172	930	111	-	625	88
Female	2,708	-	-	-	2,086	52	-	59	81	-	338	92
Demography												
Total	7,678	349	253	181	3,101	-	128	1,248	-	-	2,419	-
Male	4,389	-	224	62	1,937	-	-	344	-	-	1,822	-
Female	3,289	349	28	119	1,164	-	128	904	-	-	597	-
Economics												
Total	42,756	2,856	247	10,163	10,197	348	422	4,838	290	151	12,597	647
Male	33,937	2,225	247	7,962	8,531	348	393	3,825	268	151	9,522	465
Female	8,819	630	-	2,201	1,666	-	29	1,013	22	-	3,076	182
Education												
Total	58,713	4,422	727	8,036	19,489	1,324	1,024	9,820	183	563	12,144	982
Male	27,939	2,536	480	3,246	10,512	1,052	640	3,797	94	277	4,803	503
Female	30,774	1,887	248	4,790	8,977	272	384	6,022	89	286	7,341	479
Fine & Theatre Arts												
Total	3,912	226	-	1,091	996	-	154	-	-	-	1,354	92
Male	1,816	226	-	289	627	-	-	-	-	-	674	-
Female	2,096	-	-	801	369	-	154	-	-	-	680	92
Folklore & Literature												
Total	5,118	495	-	1,543	914	-	39	400	-	41	1,613	73
Male	847	495	-	-	199	-	39	-	-	41	-	73
Female	4,271	-	-	1,543	715	-	-	400	-	-	1,613	-
Gender & Development												
Total	1,843	-	-	355	521	-	43	884	40	-	-	-
Male	695	-	-	-	-	-	-	695	-	-	-	-
Female	1,148	-	-	355	521	-	43	189	40	-	-	-

28

Table 3.8 Cont'd

Training status, Types	Region												
of Training status, Types	_						Benishangul-				Addis	Dire	
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa	
Geography													
Total	42,418	3,542	16	12,608	13,300	681	604	7,016	922	231	3,185	314	
Male	30,340	2,685	-	8,878	9,046	509	475	5,972	718	90	1,652	314	
Female	12,078	857	16	3,730	4,254	172	128	1,044	204	141	1,533	-	
Governance & Public Adı													
Total	71,383	5,695	426	10,446	23,092	1,240	977	21,235	763	517	6,322	670	
Male	51,359	3,961	426	7,770	16,745	756	850	14,582	544	207	5,008	511	
Female	20,024	1,734	-	2,675	6,347	484	128	6,654	219	310	1,314	159	
History													
Total	18,645	2,007	-	2,710	5,467	-	129	4,268	175	49	3,840	-	
Male	13,370	2,007	-	1,819	3,260	-	-	3,565	175	-	2,544	-	
Female	5,275	-	-	891	2,207	-	129	703	-	49	1,296	-	
Journalism & Broadcastir													
Total	8,345	452	-	1,389	1,167	-	29	1,375	-	48	3,886	-	
Male	5,443	90	-	1,168	483	-	29	1,096	-	-	2,578	-	
Female	2,902	361	-	221	684	-	-	279	-	48	1,309	-	
Language													
Total	111,858	4,952	993	29,387	25,758	1,041	1,931	24,265	973	1,452	19,833	1,272	
Male	59,457	2,212	753	14,924	13,958	762	1,134	13,876	655	813	9,622	747	
Female	52,401	2,740	240	14,463	11,800	279	798	10,389	318	639	10,212	525	
Law													
Total	70,165	3,538	486	14,365	20,498	764	959	13,390	700	891	14,009	566	
Male	58,694	2,826	454	11,766	18,582	511	638	11,508	572	731	10,540	566	
Female	11,471	712	32	2,600	1,917	253	321	1,882	128	160	3,469	-	
Library & Information Ser													
Total	3,982	256	50	128	989	-	-	1,452	41	125	942	-	
Male	1,619	-	34	-	835	-	-	416	-	81	253	-	
Female	2,363	256	16	128	154	-	-	1,037	41	43	689	-	
Management													
Total	53,346	2,597	1,072	15,127	8,798	708	1,374	7,504	503	380	13,881	1,402	
Male	34,994	1,742	693	9,000	5,737	617	1,116	5,231	463	148	9,118	1,129	
Female	18,351	855	378	6,127	3,061	91	259	2,273	40	232	4,763	273	
Management Information													
Total	2,693	-	-	1,037	821	50	-	784	-	-	-	-	
Male	1,036	-	-	-	780	-	-	257	-	-	-	-	
Female	1,657	-	-	1,037	41	50	-	528	-	-	-	-	
Marketing													
Total	17,225	269	224	976	1,296	-	72	3,650	22	110	10,519	86	
Male	10,169	269	224	791	1,107	-	72	1,735	22	110	5,753	86	
Female	7,055	-	-	185	189	-	-	1,915	-	-	4,766	-	

Philosophy

Table 3.8 Cont'd

l able 3.8 Cont d						R	egion					
Training status, Types	T						Benishangul-				Addis	Dire
of Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Total	1,337	-	224	93	-	-	-	241	_	-	676	102
Male	1,243	-	224	-	-	-	-	241	-	-	676	102
Female	93	-	-	93	-	-	-	-	-	-	-	-
Political Science												
Total	3,172	256	-	-	1,352	40	-	96	-	49	1,379	-
Male	2,437	256	-	-	959	40	-	96	-	49	1,037	-
Female	735	-	-	-	392	-	-	-	-	-	343	-
Procurement & Supply M	1anagement											
Total	6,005	1,591	-	1,005	1,075	84	29	395	150	-	1,585	92
Male	2,411	-	-	547	677	84	-	93	-	-	1,010	-
Female	3,593	1,591	-	459	398	-	29	301	150	-	574	92
Psychology												
Total	8,896	702	-	1,751	1,749	-	158	3,120	-	50	1,367	-
Male	5,352	564	-	1,751	1,352	-	109	1,576	-	-	-	-
Female	3,544	137	-	-	397	-	49	1,544	-	50	1,367	-
Purchasing												
Total	6,422	412	24	1,539	1,185	118	68	395	183	-	2,258	242
Male	1,760	153	24	186	1,152	118	-	-	40	-	-	88
Female	4,661	259	-	1,352	34	-	68	395	144	-	2,258	153
Rural Development												
Total	5,186	-	-	2,068	1,086	165	148	1,286	105	41	287	-
Male	3,745	-	-	1,470	771	165	148	1,045	105	41		-
Female	1,441	-	-	598	315	-	-	241	-	-	287	-
Social Work												
Total	8,425	259	-	848	3,035	-	171	762	159	-	3,190	-
Male	6,157	259	-	788	1,886	-	143	396	97	-	2,589	-
Female	2,267	-	-	60	1,149	-	28	366	62	-	601	-
Sociology & Anthropology												
Total	14,925	500	-	1,474	5,916	185	194	3,852	22	100	2,284	398
Male	9,702	500	-	1,474	3,703	133	165	2,443	22	100	889	272
Female	5,223	-	-	-	2,214	52	28	1,408	-	-	1,395	126
Special Needs education		004		500	700	0.4		4 000	40			
Total	2,789	361	-	596	706	84	-	1,002	40	-	-	-
Male	1,764	-	-	596	644	84	-	399	40	-	-	-
Female	1,026	361	-	-	61	-	-	603	-	-	-	-
Tourism & Hotel Manager			= :									
Total	13,146	779	28	2,517	2,360	41	-	1,421	217	-	5,711	73
Male	4,624	371	28	523	1,459	41	-	542	217	-	1,371	73
Female	8,521	408	-	1,995	900	-	-	879	-	-	4,339	-
Other Social Science Fiel	ds											
Total	38,245	2,454	538	14,635	7,178	133	751	6,194	292	147	5,456	467

Table 3.8 Cont	a											
Training status, Type	76					R	egion					
of Training status, Type	,						Benishangul-				Addis	Dire
or training and sex	` Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Male	23,109	1,003	538	9,018	3,932	133	664	4,206	142	99	3,067	307
Female	15,136	1,451	-	5,617	3,246	-	87	1,988	150	48	2,389	161
Natural Science												
Total	364,756	23,413	3,319	89,618	90,847	2,694	6,319	50,378	3,040	2,494	88,619	4,014
Male	168,962	9,312	1,646	38,899	46,491	1,915	3,627	23,310	1,772	1,413	38,900	1,676
Female	195,793	14,101	1,673	50,719	44,357	779	2,692	27,067	1,268	1,081	49,718	2,338
Biology	,	, -	,	,	,		,	,	,	,	-, -	,
Total	37,452	3,097	230	8,606	8,636	494	770	6,187	839	261	7,965	365
Male	18,376	922	189	3,818	4,926	356	558	2,953	511	48	4,006	88
Female	19,076	2,176	41	4,788	3,710	138	212	3,235	328	213	3,959	277
Biotechnology	.0,0.0	_,	• •	.,. 55	٥, ٠			0,200	0_0		0,000	
Total	2,805	_	_	358	786	_	-	241	88	_	1,333	_
Male	2,199	_	_	358	674	_	_		88	_	1,080	_
Female	606	_	_	-	113	_	_	241	-	_	253	_
Chemistry	000										200	
Total	23,644	2,227	437	3,998	6,201	184	816	2,043	329	89	6,908	411
Male	19,094	1,079	437	3,487	5,747	144	740	1,585	267	89	5,196	324
Female	4,550	1,148		511	454	40	77	458	63	-	1,713	86
Computer Science	4,000	1,140		011	404	40	, ,	400	00		1,7 10	00
Total	137,587	8,134	1,051	33,675	31,951	1,084	1,273	18,018	735	530	39,502	1,631
Male	38,011	2,231	68	8,026	9,652	733	169	3,781	62	138	12,698	453
Female	99,576	5,903	984	25,650	22,300	351	1,104	14,237	673	392	26,804	1,178
Earth Science	00,0.0	0,000		_0,000	,000		.,	,_0.	0.0		_0,00	.,
Total	4,930	706	-	315	1,118	41	-	1,028	22	48	1,652	_
Male	3,226	616	-	223	475	41	-	485	22	48	1,318	_
Female	1,704	90	_	93	644	_	_	543	_	_	334	_
Environmental Science		00		00	0			0.0			00 1	
Total	10,871	_	48	1,715	4,603	101	_	2,542	127	_	1,607	126
Male	6,367	-	32	893	2,700	101	_	1,752	87	_	675	126
Female	4,504	_	16	823	1,903	-	_	790	40	_	932	-
Food Science	1,001				,,,,,,							
Total	5,455	90	_	158	1,233	50	-	280	_	_	3,643	_
Male	1,347	90	_	-	1,200	50	_	187	_	_	1,020	_
Female	4,108	-	_	158	1,233	-	_	93	_	_	2,624	_
Geology	1,100			100	1,200			00			2,021	
Total	1,802	_	-	889	913	_	_	_	_	_	_	_
Male	1,358	_	_	446	913	_	_	_	_	_	_	_
Female	444	_	_	444	-	_	_	_	_	_	_	_
Torridio	777		_	7-7-7							_	
Information Systems												
Total	24,431	658	374	10,343	5,895	_	1,049	2,121	_	261	3,272	458
IUIAI	24,431	000	3/4	10,343	5,095	-	1,049	∠, ۱∠ ۱	-	201	3,212	400

Table 3.8 Cont'd

	ont a					R	Region					
Training status, T							Benishangul-				Addis	Dire
of Training and	Sex Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Male	9,494	90	219	3,808	2,697	-	212	880	-	220	1,367	-
Female	14,937	568	155	6,535	3,199	-	837	1,240	-	41	1,905	458
Mathematics												
Total	50,799	3,894	244	15,827	12,447	120	510	6,561	614	522	9,705	354
Male	32,925	2,506	126	9,440	8,494	120	481	5,517	614	522	5,003	102
Female	17,874	1,388	119	6,388	3,952	-	29	1,044	-	-	4,702	252
Meteorology												
Total	1,782	243	-	-	726	-	49	-	40	41	683	-
Male	856	-	-	-	726	-	49	-	40	41	-	-
Female	926	243	-	-	-	-	-	-	-	-	683	-
Physics												
Total	11,794	800	25	1,856	3,397	40	209	1,803	164	210	3,289	-
Male	9,189	800	25	1,463	3,272	40	209	1,803	-	210	1,365	-
Female	2,606	-	-	393	125	-	-	-	164	-	1,924	-
Sport Science												
Total	27,978	1,972	166	7,773	6,697	424	870	3,929	82	195	5,450	421
Male	16,521	842	55	4,887	4,093	175	703	1,367	82	96	3,800	421
Female	11,457	1,130	111	2,886	2,604	249	167	2,561	-	99	1,650	-
Statistics												
Total	5,416	238	47	481	609	-	377	1,857	-	-	1,720	86
Male	2,682	-	47	420	84	-	285	1,153	-	-	693	-
Female	2,734	238	-	62	525	-	92	704	-	-	1,027	86
Other Natural Sci	ence Fields											
Total	18,010	1,351	696	3,623	5,635	154	395	3,767	-	337	1,889	161
Male	7,318	135	449	1,633	2,039	154	221	1,847	-	-	680	161
Female	10,691	1,216	248	1,990	3,596	-	174	1,920	-	337	1,210	-
Engineering & A	Illied Services											
Total	297,036	25,125	3,416	48,965	68,748	1,168	975	33,816	1,305	1,877	106,520	5,120
Male	195,493	15,919	1,794	31,429	47,214	718	756	22,379	1,062	1,065	69,680	3,477
Female	101,543	9,206	1,622	17,537	21,534	450	219	11,438	244	812	36,840	1,643
Agricultural Engir	neering											
Total	1,019	90	-	477	-	-	-	59	49	-	344	-
Male	1,019	90	-	477	-	-	-	59	49	-	344	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Architecture												
Total	6,288	704	-	92	2,403	50	-	1,183	164	49	1,642	-
Male	3,439	362	-	92	763	50	-	584	164	49	1,375	-

Table 3.8 Cont'e	d											1
Training status, Types	s	T				R	egion		1			
of Training and Sex		Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Female	2,849	342	-	-	1,640	-	-	600	-	-	267	-
Cadastre & mapping S	Service											
Total	1,123	-	-	91	798	-	-	-	-	234	-	-
Male	275	-	-	91	67	-	-	-	-	117	-	-
Female	848	-	-	-	731	-	-	-	-	117	-	-
Chemical Engineering	9											
Total	3,004	756	-	111	319	-	-	222	82	-	1,352	161
Male	2,662	756	-	111	319	-	-	222	82	-	1,010	161
Female	342	-	-	-	-	-	-	-	-	-	342	-
Civil Engineering												
Total	29,981	1,596	112	2,689	5,028	130	130	3,484	22	251	16,007	534
Male	22,864	789	112	1,526	3,556	130	130	3,062	-	251	12,877	432
Female	7,117	807	-	1,163	1,473	-	-	422	22	-	3,130	102
Computer Engineering	I											
Total	20,135	1,649	10	2,477	3,839	52	-	1,891	-	-	10,217	-
Male	8,896	555	10	1,604	1,303	52	-	401	-	-	4,971	-
Female	11,239	1,094	-	873	2,536	-	-	1,490	-	-	5,246	-
Communication Engir	neering											
Total	3,103	133	-	-	790	-	39	-	-	48	1,900	194
Male	2,369	-	-	-	578	-	39	-	-	-	1,559	194
Female	734	133	-	-	212	-	-	-	-	48	342	-
Construction Technol	ogy											
Total	24,130	1,147	239	2,484	9,359	-	57	4,051	49	275	6,280	188
Male	19,271	1,014	239	2,322	7,050	-	-	1,903	-	275	6,280	188
Female	4,858	133	-	162	2,308	-	57	2,149	49	-	-	-
Drafting												
Total	9,122	553	-	2,694	1,310	-	-	241	-	110	4,037	179
Male	4,817	133	-	1,235	768	-	-	-	-	-	2,594	86
Female	4,306	420	-	1,458	541	-	-	241	-	110	1,443	92
Electrical engineering												
Total	50,837	4,930	1,055	10,293	10,261	40	71	6,348	203	148	16,283	1,204
Male	41,694	4,208	468	7,758	7,605	40	42	5,653	122	148	14,640	1,011
Female	9,143	723	587	2,536	2,656	-	29	696	81	-	1,643	194
Esthetics												
Total	1,296	238	_	-	-	-	-	482	-	162	287	126
Male	241	-	-	-	-	-	_	241	-	-	-	-
Female	1,055	238	-	-	-	-	-	241	-	162	287	126

34

Table 3.8 Cont'd

Training status, Types			-			R	egion					-
of Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Food Processing	•								•			
Total	37,544	1,329	-	5,677	4,248	124	-	2,139	40	375	23,188	425
Male	4,983	-	-	573	672	40	-	96	-	-	3,602	-
Female	32,561	1,329	-	5,104	3,576	84	-	2,042	40	375	19,587	425
Industrial Engineering												
Total	6,727	1,374	-	826	962	-	39	1,142	105	-	2,188	92
Male	5,072	1,032	-	118	928	-	39	1,051	53	-	1,851	-
Female	1,656	342	-	709	34	-	-	91	53	-	336	92
Manufacturing Technology												
Total	2,330	439	-	1,084	520	-	-	-	-	-	287	-
Male	971	90	-	594	-	-	-	-	-	-	287	-
Female	1,359	349	-	490	520	-	-	-	-	-	-	_
Mechanical Engineering												
Total	25,278	2,574	458	3,642	7,241	52	93	1,797	419	_	8,728	274
Male	24,184	2,438	458	3,567	7,119	52	57	1,738	419	_	8,062	274
Female	1,095	137	-	, 75	122	_	36	59	_	_	665	-
Metal Engineering	,											
Total	3,763	672	-	175	1,117	_	-	427	_	_	964	408
Male	3,496	405	-	175	1,117	-	-	427	-	-	964	408
Female	267	267	-	-	-	_	-	_	_	_	_	-
Petroleum Engineering												
Total	584	-	-	-	584	-	-	-	-	-	-	-
Male	584	-	-	-	584	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Surveying												
Total	27,063	2,375	704	9,727	4,792	103	231	4,222	-	93	4,095	721
Male	16,219	1,126	449	5,514	3,138	52	192	2,909	-	93	2,485	261
Female	10,844	1,249	255	4,214	1,654	51	39	1,313	-	-	1,609	460
Textile Engineering												
Total	3,397	632	-	749	1,003	-	-	-	-	-	1,013	-
Male	1,319	271	-	593	116	-	-	-	-	-	338	-
Female	2,078	360	-	155	887	-	-	-	-	-	675	-
Urban Land Development S	Service											
Total	2,383	-	_	1,231	_	_	_	478	_	90	583	_
Male	2,053	-	-	1,231	_	-	-	478	_	90	253	-
Female	331	-	_	-,===	-	_	_	-	-	-	331	_
Urban Planning & Manage											001	

35

Table 3.8 Cont'd

Table 3.8 Cont d												1
Training status, Types		1	1			R	egion		T			
of Training and Sex	Ta4-1	Times	۸ 44	A k	0	Com =!!:	Benishangul-	CNINIDD	Combal	Harrari	Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Total	2,116	363	-	-	1,046	-	39	615	53	-	-	-
Male .	1,861	363	-	-	1,046	-	39	360	53	-	-	-
Female	. 255	-	-	-	-	-	-	255	-	-	-	-
Water Resources Engine												
Total	10,621	2,284	779	184	3,424	462	58	2,624	81	41	594	88
Male	7,385	1,219		184	3,032	147	29	2,223	81	41	341	88
Female	3,235	1,066	779	-	392	315	29	401	-	-	253	-
Wood Science Technology												
Total	14,282	1,067	28	2,968	4,530	-	29	733	-	-	4,399	527
Male	12,279	1,067	28	2,968	3,825	-	-	299	-	-	3,717	374
Female	2,003	-	-	-	705	-	29	434	-	-	682	153
Other Engineering Fields												
Total	10,910	218	31	1,293	5,176	154	189	1,679	40	-	2,130	-
Male	7,542	-	31	696	3,628	154	189	675	40	-	2,130	-
Female	3,368	218	-	597	1,548	-	-	1,004	-	-	-	-
Health Science												
Total	240,221	16,585	1,988	43,135	68,680	3,906	6,025	41,038	1,887	3,442	49,784	3,751
Male	104,889	4,585	813	21,283	31,436	2,165	3,223	18,791	1,012	2,016	17,867	1,696
Female	135,332	11,999	1,174	21,851	37,243	1,741	2,803	22,247	875	1,426	31,917	2,055
Anesthesia												
Total	415	223	-	60	-	-	132	-	-	-	-	-
Male	133	133	-	-	-	-	-	-	-	-	-	-
Female	283	90	-	60	-	-	132	-	-	-	-	-
Animal Health Care												
Total	6,379	1,112	138	1,510	1,283	120	108	583	361	41	1,034	91
Male	3,978	-	138	898	1,283	120	29	583	144	-	692	91
Female	2,401	1,112	-	611	-	-	79	-	217	41	342	-
Dentistry												
Total	1,625	-	-	-	123	-	-	95	-	-	1,407	-
Male	946	-	-	-	123	-	-	95	-	-	728	-
Female	678	-	-	-	-	-	-	-	-	-	678	-
Dental Hygiene												
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Food Hygiene												
Total	2,884	-	16	593	128	-	-	96	-	-	2,052	-
Male	156	_	-	-	60	_	-	96	-	-	-	_
Female	2,728	-	16	593	67	-	-	-	-	-	2,052	-
Health Extension	•										•	

Table 3.8 Cont'd

Training status, Types						R	egion					
of Training status, Types –							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Total	16,232	90	47	5,328	3,860	-	753	3,552	126	48	2,274	153
Male	1,555	-	31	445	-	-	92	986	-	-	-	-
Female	14,677	90	16	4,883	3,860	-	661	2,566	126	48	2,274	153
Health Officer												
Total	11,411	-	108	4,143	2,112	172	262	1,931	41	139	2,230	273
Male	8,609	-	-	3,396	1,382	172	262	1,107	41	90	1,888	273
Female	2,802	-	108	748	731	-	-	824	-	49	342	-
Massage & Physiotherap												
Total	1,435	-	-	136	114	-	-	-	-	110	1,076	-
Male	167	-	-	-	57	-	-	-	-	110	-	-
Female	1,269	-	-	136	57	-	-	-	-	-	1,076	-
Medical Doctor												
Total	6,871	1,008	224	127	815	91	-	945	-	48	3,422	190
Male	4,295	617	224	127	732	40	-	945	-	48	1,372	190
Female	2,576	391	-	-	83	51	-	-	-	-	2,051	-
Medical Laboratory												
Total	14,815	497	26	2,678	5,248	50	567	1,785	82	88	3,693	102
Male	8,856	361	10	1,708	3,430	50	408	378	-	88	2,322	102
Female	5,960	136	16	970	1,818	-	159	1,406	82	-	1,372	-
Midwifery												
Total	13,150	2,158	-	2,078	3,298	123	337	1,547	-	581	2,936	92
Male	2,875	-	-	754	1,062	-	79	241	-	303	436	-
Female	10,275	2,158	-	1,324	2,235	123	258	1,307	-	278	2,500	92
Nursing												
Total	102,615	8,221	307	16,591	30,582	2,321	2,126	19,261	710	1,901	18,228	2,369
Male	37,996	1,116	55	8,785	9,768	1,090	1,517	8,884	346	1,078	4,711	648
Female	64,619	7,105	252	7,806	20,814	1,231	609	10,378	364	823	13,517	1,721
Nursing Assistance												
Total	9,113	271	247	538	4,760	276	-	323	-	-	2,572	126
Male	5,705	271	28	538	3,530	40	-	161	-	-	1,010	126
Female	3,408	-	219	-	1,230	235	-	161	-	-	1,562	-
Optometry												
Total	555	-	-	-	314	-	-	241	-	-	-	-
Male	555	-	-	-	314	-	-	241	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Pharmacy												
Total	27,142	1,550	155	4,360	9,022	311	693	3,403	400	357	6,890	_
Male	14,206	766	155	1,798	4,190	311	342	1,960	400	259	4,023	-
Female	12,937	783	-	2,562	4,833	-	351	1,443	-	98	2,867	-
Psychiatry												

37

Table 3.8 Cont'd

Training status, Types						R	egion					
of Training status, Types of Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Total	184	-	-	-	-	-	-	184	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	184	-	-	-	-	-	-	184	-	-	-	-
Public/Environmental Hea												
Total	6,304	266	-	1,466	1,288	-	238	2,823	-	130	-	92
Male	2,580	133	-	952	581	-	29	752	-	41	-	92
Female	3,724	133	-	513	707	-	209	2,071	-	90	-	-
Radiography												
Total	1,034	137	-	118	448	-	-	-	-	-	332	-
Male	702	137	-	118	448	-	-	-	-	-	-	-
Female	332	-	-	-	-	-	-	-	-	-	332	-
Retail Pharmacy Assistan	ice											
Total	752	90	-	_	62	_	-	600	_	_	-	-
Male	690	90	-	-	-	-	-	600	-	-	-	-
Female	62	-	-	-	62	-	-	-	-	-	-	-
Veterinary Medicine												
Total	11,891	618	704	2,642	4,130	442	637	1,491	125	-	1,017	86
Male	8,741	618	156	1,354	3,956	341	465	997	82	_	686	86
Female	3,151	-	548	1,288	173	101	173	494	43	-	331	-
Other Health Fields												
Total	5,412	343	16	767	1,094	-	173	2,177	43	-	621	177
Male	2,143	343	16	411	521	-	-	764	-	-	-	88
Female	3,268	-	-	356	573	-	173	1,413	43	-	621	88
Agricultural Science												
Total	73,075	10,632	1,951	15,709	19,240	1,936	2,304	12,790	1,988	467	5,448	609
Male	48,501	3,957	1,513	11,709	13,772	1,762	1,548	8,800	934	333	3,814	361
Female	24,574	6,674	438	4,000	5,468	174	757	3,990	1,055	134	1,635	249
Agricultural Extension												
Total	16,865	1,488	177	3,650	5,078	381	459	2,661	388	184	2,129	270
Male	12,641	1,128	143	2,804	3,767	381	286	1,923	149	94	1,785	181
Female	4,224	360	34	847	1,311	-	172	738	239	90	344	89
Animal Production & Man					.,							
Total	5,100	2,174	428	278	916	73	239	443	125	_	336	87
Male	2,863	1,182	428	162	149	73	210	301	22	_	336	-
Female	2,237	993	420	116	767	-	29	143	103	_	-	87
Animal Science	2,201	333	-	110	707	_	29	143	103	-	-	01
Total	11,359	1,019	25	1,997	4,895	401	265	1,586	627	197	348	
Male		399	25 25			401 401			284	197		-
	7,437			1,407	3,124	401	68	1,184		197	348	-
Female	3,922	620	-	590	1,770	-	198	401	342	-	-	-
Artificial insemination												

Table 3.8 Cont'd

Training status, Types						R	egion					
of Training status, Types of Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Bamboo Harvesting												
Total	93	-	-	93	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	93	-	-	93	-	-	-	-	-	-	-	-
Beekeeping												
Total	1,031	434	-	493	61	-	-	-	-	43	-	-
Male	554	-	-	493	61	-	-	-	-	-	-	-
Female	477	434	-	-	-	-	-	-	-	43	-	-
Dairy Production												
Total	680	680	-	-	-	-	-	-	-	-	-	-
Male	409	409	-	-	-	-	-	-	-	-	-	-
Female	272	272	-	-	-	-	-	-	-	-	-	-
Field Crops Production												
Total	3,399	-	537	594	1,098	-	121	625		-	342	-
Male	2,744	-	323	594	1,041	-	-	403		-	342	-
Female	655	-	214	-	57	-	121	222	40	-	-	-
Forestry												
Total	3,291	-	24	1,506	770	92	173	143		-	336	-
Male	2,881	-	24	1,506	736	92	173	143		-	-	-
Female	410	-	-	-	34	-	-	-	40	-	336	-
Horticulture												
Total	931	319	-	-	531	-	-	-	81	-	-	-
Male	621	90	-	-	531	-	-	-	-	-	-	-
Female	309	229	-	-	-	-	-	-	81	-	-	-
Leather Basic Production												
Total	1,073	-	-	-	118	-	-	-	-	-	954	-
Male	118	-	-	-	118	-	-	-	-	-	-	-
Female	954	-	-	-	-	-	-	-	-	-	954	-
Natural Resource Manager	ment											
Total	13,392	616	642	4,740	1,568	459	827	3,963	324	-	-	253
Male	9,506	-	483	3,545	1,511	459	667	2,453		-	-	180
Female	3,887	616	159	1,195	57	-	160	1,510		-	-	73
Plant Science	•			•				•				
Total	-	-	-	-	-	-	-	-	-	-	-	-

Table 3.8 Cont'd												
Training status, Types						R	Region					
of Training status, Types of Training and Sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Post Harvest Manageme												
Total	158	-	-	-	-	158	-	-	-	-	-	-
Male	158	-	-	-	-	158	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Poultry Production												
Total	3,410	1,696	-	1,007	523	-	-	184	-	-	-	-
Male	61	-	-	61	-	-	-	-	-	-	-	-
Female	3,349	1,696	-	946	523	-	-	184	-	-	-	-
Range Science												
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Stimulants & Spices Prod	duction											
Total	240	181	-	-	-	-	-	59	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	240	181	-	-	-	-	-	59	-	-	-	-
Sugar Crops Production												
Total	333	-	-	-	-	-	-	-	-	-	333	-
Male	333	-	-	-	-	-	-	-	-	-	333	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Vegetables, Fruits, Flower		n										
Total	1,066	552	-	93	121	-	-	301	-	-	-	-
Male	515	-	-	93	121	-	-	301	-	-	-	-
Female	552	552	-	-	-	-	-	-	-	-	-	-
Other Agricultural Fields												
Total	10,654	1,473	117	1,258	3,562	372	220	2,825	115	42	670	-
Male	7,661	750	86	1,044	2,614	198	143	2,092		42	670	-
Female	2,993	723	31	214	949	174	77	732	93	-	-	-
Construction Industry												
Total	157,661	23,058	1,453	38,631	29,169	621	1,181	14,733		2,013	43,663	3,074
Male	116,352	15,596	1,264	26,476	20,085	398	949	10,823		1,394	37,237	2,066
Female	41,309	7,461	189	12,155	9,084	223	232	3,910	-	619	6,426	1,008
Bar Bending & Concreting	9											
Total	7,843	1,529	450	1,000	993	-	-	986	-	385	2,322	177
Male	6,845	1,376	450	836	993	-	-	986		385	1,641	177
Female	998	153	-	164	-	-	-	-	-	-	681	-
Building Construction				_								
Total	29,701	3,278	16	8,559	4,472	50	573	5,765	22	-	6,790	176

40

Table 3.8 Cont'd Region Training status, Types Addis Benishangul-Dire of Training and Sex Total Tigray Affar Amhara Oromia Somalie Gumuz SNNPR Gambela Harari Ababa Dawa Male 20.746 2.205 4.281 3.003 50 376 4.639 22 6.080 90 Female 8,956 1,073 16 4,278 1,469 197 1,126 711 86 **Building Electrical Installation** Total 12,469 511 3,620 2,421 50 1,557 3,676 415 218 256 2.296 1.527 860 2.782 342 Male 8.111 49 Female 4,358 256 1,324 894 50 697 170 895 73 **Building Metal construction** Total 567 164 241 162 Male 282 241 41 Female 286 164 122 Carpentry & Joinery Total 9.945 1,691 253 2,574 1,048 320 1,536 49 2,389 86 Male 9,436 1,182 253 2,574 1,048 320 1,536 49 2,389 86 Female 509 509 Coble Stone Carving Total 11.850 1.130 3.914 3.205 174 36 91 649 2.320 332 Male 8.631 769 2.805 2.466 462 1.977 153 3,220 739 36 91 343 Female 361 1,110 174 188 179 Coble Stone Road Construction Total 17.959 8,186 3,395 4,842 1,167 195 174 Male 10,095 4,918 2,284 2,514 184 195 -7.863 983 Female 3,267 2,328 174 1.111 Concreting & Reinforcing Works Total 3.933 423 2.047 40 96 1.240 86 Male 3,201 356 1,564 40 1,240 732 67 96 Female 483 86 Construction Equipment Operation Total 152 375 29 88 975 332 Male 601 152 29 332 88 375 375 Female Construction Management Total 11,124 1,199 2,495 2,108 339 41 4,943 Male 9,590 870 1,955 2,074 4,690 Female 1.534 329 539 34 339 41 253 Finishing Construction Works Total 5,206 523 81 36 95 397 744 10 1,627 1,693 1,361 Male 1,819 10 164 81 36 95 73 523 332 324 Female 3.387 744 1.464 Foundry Works 22 22 Total

Table 3.8 Cont'd							logion					
Training status, Types	1	I				, K	egion				۸ مامانه	Dine
of Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Male	22	-	-	-	-	-	-	-	22	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Geo-Technical Drilling												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Masonry Works												
Total	93	-	-	93	-	-	-	-	-	_	-	
Male	93	_	_	93	_	_	-	-	-	_	_	
Female	_	-	_	_	-	-	-	-	-	_	_	
Metal Machining												
Total	4,905	526	10	61	210	_	-	305	-	88	3,452	254
Male	3,736	345	10	61	176	-	-	305	-	88	2,497	254
Female	1,169	181	-	_	34	_	-	_	_	-	954	
Metal Works	1,100											
Total	15,843	1,606	_	3,312	3,200	186	-	400	-	42	6,754	344
Male	13,678	1,606	_	3,220	2,156		-	400		42	5,725	344
Female	2,165	· -	_	92	1,044	_	-	-	-	_	1,029	
Plastering & Paints Works					,						,	
Total	2,646	218	231	-	392	_	-	-	-	_	1,805	
Male	2,105	218	231	-	392	-	-	-	-	-	1,264	
Female	541	-	-	-	-	-	-	-	-	-	541	
Road Construction Maint	tenance											
Total	5,930	135	31	3,034	1,405	-	-	283		-	954	87
Male	4,501	-	31	1,801	1,345	-	-	283	-	-	954	87
Female	1,429	135	-	1,233	60	-	-	-	-	-	-	
Road Surfacing												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Tiling Work												
Tiling Work Total	130				34			96				
Male	130	-	-	_	34	-	-	96 96		-	-	•
Female	130	-	-	_	34	-	-	90	-	_	-	
Tunnel Construction	-	-	-	_	-	-	-	-	-	-	-	•
Total	629	349	96	184	_	-	-	-	-	_	-	
Male	504	349	64	91	_	_	_	_	_	_	_	
iviale	504	343	04	91	-	-	-	-	-	-	-	

Table 3.8 Cont d							•					
Training status, Types		ı		T		K	Region					
of Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Female	125	-	32	92	-	_	-	-	-	-	-	-
Water Well Drilling & C	Construction											
Total	133	-	-	93	-	40	-	-	-	-	-	-
Male	40	-	-	-	-	40	-	-	-	-	-	-
Female	93	-	-	93	-	-	-	-	-	-	-	-
Welding												
Total	8,835	1,111	56	3,083	1,234	-	130	1,196	22	183	1,747	73
Male	7,505	840	56	2,990	710	-	130	1,196	22	84	1,405	73
Female	1,330	272	-	93	524	-	-	-	-	99	342	-
Other Construction Field	ds											
Total	6,922	843	300	848	661	-	58	579	-	-	3,247	386
Male	4,680	662	159	518	83	-	58	-	-	-	2,901	300
Female	2,241	181	141	330	578	-	-	579	-	-	346	86
Transport Operations												
Total	448,215	31,105	4,719	73,669	124,109	4,526	3,229	52,498	1,390	4,961	135,404	12,605
Male	428,046	30,355	4,719	70,725	118,975	4,486	3,108	50,486	1,338	4,920	126,666	12,268
Female	20,169	750	-	2,943	5,133	40	121	2,012	53	41	8,738	337
Aircraft Mechanic												
Total	1,736	-	-	-	380	-	-	-	-	-	1,264	92
Male	1,669	-	-	-	313	-	-	-	-	-	1,264	92
Female	67	-	-	-	67	-	-	-	-	-	-	-
Automotive Driving												
Total	20,487	1,364	245	4,913	8,767	293	220	971	115	238	2,549	813
Male	19,375	1,108	245	4,820	8,244	293	220	730	115	238	2,549	813
Female	1,112	255	-	92	524	-	-	241	-	-	-	-
Automotive Electrical/E		cing										
Total	30,538	2,111	31	7,579	7,749	201	418	3,043	-	89	8,800	518
Male	26,358	1,840	31	6,849	6,939	161	418	2,071	-	48	7,657	346
Female	4,181	272	-	730	810	40	-	972	-	41	1,143	172
Automotive Mechanic												
Total	53,612	6,477	210	8,928	7,708	208	504	6,100	53	129	22,231	1,064
Male	50,359	6,253	210	8,824	6,283	208	412	6,100	-	129	20,875	1,064
Female	3,253	223	-	103	1,425	-	92	-	53	-	1,357	-
Automotive Painting												
Total	1,613	-	-	161	557	-	-	400	-	-	334	161
Male	1,613	-	-	161	557	-	-	400	-	-	334	161
Female	_	-	-	-	-	-	-	-	-	-	-	-
Automotive Servicing N												
Total	2,348	218	-	886	180	-	-	-	-	-	978	86
Male	2,348	218	-	886	180	-	-	-	-	-	978	86
Female	-	-	-	-	-	-	-	-	-	-	-	-

l able 3.8	Cont'd	T											
Training state	us, Types	<u> </u>			ı		R	egion		1			
of Training		Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Automotive T	echnology										•		
Total		3,326	-	-	1,040	1,297	-	-	646	-	-	343	-
Male		3,115	-	-	949	1,177	-	-	646	-	-	343	-
Female		210	-	-	90	120	-	-	-	-	-	-	-
City Bus Drivi	ng												
Total		24,044	2,238	10	5,723	6,073	170	98	1,393	110	313	7,001	917
Male		23,423	2,238	10	5,160	6,015	170	98	1,393	110	313	7,001	917
Female		621	-	-	563	58	-	-	-	-	-	-	-
Taxi Driving													
Total		131,293	7,569	743	16,136	31,570	2,584	877	10,444	378	2,724	51,518	6,750
Male		125,477	7,569	743	15,397	30,529	2,584	877	9,644	378	2,724	48,355	6,677
Female		5,815	-	-	739	1,042	-	-	800	-	-	3,162	73
Other Transp	ort Service												
Total		179,218	11,128	3,480	28,303	59,827	1,070	1,112	29,501	734	1,468	40,387	2,205
Male		174,307	11,128	3,480	27,678	58,738	1,070	1,083	29,501	734	1,468	37,311	2,113
Female		4,911	-	-	625	1,089	-	29	-	-	-	3,076	92
Infrastructure	e Services												
Total		33,290	2,790	1,564	7,123	7,495	114	72	3,241	46	349	9,843	653
Male		23,002	1,369	1,020	3,934	5,199	73	43	1,621	46	307	8,825	565
Female		10,287	1,421	544	3,189	2,296	40	29	1,620	-	42	1,018	88
Biomedical E	quipment S	Servicing											
Total		2,642	-	-	-	520	-	-	96	-	-	2,026	-
Male		2,121	_	_	-	-	_	-	96	-	-	2,026	_
Female		520	_	_	-	520	_	-	-	-	-	-	_
Home/Office E	Elect/Electro	onics Equip. S	Servicing										
Total		7,429	181	_	1,138	1,500	73	-	_	46	91	3,996	403
Male		6,554	90	_	545	1,440	73	_	_	46	49	3,996	314
Female		875	90	_	594	60	_	_	_	_	42	-	88
Hydrometric	Monitorina	0.0											
Total	3	756	633	_	_	123	_	_	_	_	_	_	_
Male		259	137	_	_	123	_	_	_	_	_	_	_
Female		497	497	_	_	120	_	_	_	_	_	_	_
Infrastructure	e operations	_	101										
Total	, орогашот.	1,027	514	23	155	_	_	_	_	_	_	335	_
Male		1,027	514	23	155		_	_		_		335	
Female		1,021	314	23	133	_	_	_	_	_	_	333	_
IT Service Ma	anagement	-	-	-	-	-	-	-	-	-	-	-	-
Total	anagomoni	616	616										
		010	010	-	-	-	-	-	-	-	-	-	-
Male		-	-	-	-	-	-	-	-	-	-	-	-

Table 3.8 Cont'd

Table 3.8 Cont d							egion					
Training status, Types	1	T					Benishangul-				Addis	Dire
of Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Female	616	616	-	-	-	_	-	-	_	-	-	-
IT support Service												
Total	670	-	-	670	-	-	-	-	-	-	-	-
Male	445	-	-	445	-	-	-	-	-	-	-	-
Female	225	-	-	225	-	-	-	-	-	-	-	-
Mechatronics & Instrum	entation servi	cing										
Total	778	-	-	101	34	-	29	190	-	-	424	-
Male	648	-	-	-	34	-	-	190	-	-	424	-
Female	130	-	-	101	-	-	29	-	-	-	-	-
Production Machine Insta	llation											
Total	342	-	-	-	-	-	-	-	-	-	342	-
Male	342	-	-	-	-	-	-	-	-	-	342	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Sanitary Installation												
Total	424	-	-	-	93	-	-	-	-	-	332	-
Male	93	-	-	-	93	-	-	-	-	-	-	-
Female	332	-	-	-	-	-	-	-	-	-	332	-
Telecommunication Wor	ks											
Total	2,704	-	-	1,149	749	-	-	-	-	-	733	73
Male	1,805	-	-	313	686	-	-	-	-	-	733	73
Female	899	-	-	837	62	-	-	-	-	-	-	-
Water Supply & Distributi	on Operations	3										
Total	8,188	436	1,510	1,029	3,230	40	43	1,073	-	110	625	92
Male	5,911	218	966	936	2,187	-	43	734	-	110	625	92
Female	2,278	218	544	93	1,044	40	-	339	-	-	-	-
Web & Multimedia Design	ning											
Total	1,165	273	-	892	-	-	-	-	-	-	-	-
Male	273	273	-	-	-	-	-	-	-	-	-	-
Female	892	-	-	892	-	-	-	-	-	-	-	-
Other Infrastructure Servi	ce Fields											
Total	6,549	137	31	1,987	1,246	-	-	1,882	-	148	1,030	86
Male	3,524	137	31	1,540	637	-	-	601	-	148	344	86
Female	3,024	-	-	447	610	_	-	1,281	_	-	686	-
Production Operations				• • •	0.0			.,_0.				
Total	60,711	8,102	982	9,409	11,166	73	209	5,732	134	445	23,243	1,215
Male	25,969	2,646	722	3,870	3,337	-	209	3,272	134	137	11,488	153
Female	34,742	5,456	260	5,539	7,829	73		2,460	-	308	11,754	1,062
i diliaid	J 1,1 1Z	5, 100	200	5,000	.,020	, 0		۷, ۱۵۵		000	. 1,704	.,002

45

Total Male Female

Total

Male

Female

Bamboo Furniture Making

Sugar & Ethanol Production

1,067

724

343

Table 3.8 Cont'd Region Training status, Types Benishangul-Addis Dire of Training and Sex Total **Tigray** Affar Amhara Oromia Somalie Gumuz SNNPR Gambela Harari Ababa Dawa Animal By-Product Processing Total 566 40 510 16 550 510 40 Male 16 Female 16 Plant By-Product Processing Total 341 341 Male Female 341 341 Hide & Skin Processing Total 408 187 445 2,099 3,139 Male 2,233 255 222 1,756 Female 222 906 153 187 343 **Textile Chemical Processing** Total 253 253 Male 253 253 Female **CNC Machine Operation** Total Male Female Tool & Die Making 2,378 94 Total 2,663 191 94 Male 1,648 191 1,363 Female 1.015 1,015 Dry Food & Spice Processing Total 5,352 255 1,044 1,355 702 49 1,703 244 1,026 Male 1,119 92 Female 4,234 255 951 1,355 702 49 677 244 **Bamboo Derivatives Production**

1,033

690

343

34

Table 3.8 Cont'd

Table 3.8 Cont'd												
Training status, Types		Region										
of Training and Sex		7					Benishangul-				Addis	Dire
_	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Total	26	-	26	-	-	-	-	-	-	-	-	-
Male	26	-	26	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Garment production												
Total	7,431	90	-	1,747	2,614	-	-	162	-	-	2,818	-
Male	2,256	90	-	596	890	-	-	-	-	-	680	-
Female	5,175	-	-	1,151	1,724	-	-	162	-	-	2,138	-
Leather Goods Making												
Total	1,395	-	-	-	121	-	-	-	-	-	1,274	-
Male	1,395	-	-	-	121	-	-	-	-	-	1,274	-
Female	-	-	-	-	-	-	-	-	-	-		-
Metal Machining												
Total	2,425	272	-	488	621	-	-	575	-	41	342	86
Male	1,997	272	_	488	621	_	-	575	_	41	_	-
Female	429	_	_	-	_	_	-	_	_	_	342	86
Millwright	0										0.2	
Total	876	_	_	356	520	_	-	_	_	_	_	_
Male	876	_	_	356	520	_	_	_	_	_	_	_
Female	-	_	_	-	-	_	-	_	_	_	_	_
Root Crops Processing												
Total	_	_	_	_	_	_	_	_	_	_	_	_
Male	_	_	_	_	_	_	_	_	_	_	_	_
Female	_	_	_	_	_	_	_	_	_	_	_	_
Production Technology												
Total	_	_	_	_	_	_	_	_	_	_	_	_
Male	_	_	_	_	_	_	_	_	_	_	_	_
Female	_	_	_	_	_	_	_	_	_	_	_	_
i emale	_	_	_	_	_	_	_	_				_
Furniture Making												
Total	5,080	674	-	513	1,098	-	-	1,124	94	48	1,374	153
Male	3,648	-	-	513	342	-	-	1,124	94	48	1,374	153
Female	1,431	674	-	-	757	-	-	-	-	-	-	-
Dressmaking & Tailoring	•											
Total	26,688	4,845	238	4,200	4,353	73	-	2,313	-	308	9,627	732
Male	7,306	1,175	10	1,500	561	-	-	940	-	48	3,072	-
Female	19,383	3,670	228	2,699	3,792	73	-	1,373	-	259	6,555	732
Cement Production Equip	ment Operati											
Total	-	-	-	-	-	-	-	-	-	-	-	-

Table 3.8 Cont d						R	egion					
Training status, Types							Benishangul-				Addis	Dire
of Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Other Production Opera												
Total	3,409	856	668	780	485	-	209	411	-	-	-	-
Male	1,939	153	653	230	284	-	209	411	-	-	-	-
Female	1,470	703	16	550	201	-	-	-	-	-	-	-
Other Social Services												
Total	423,698	29,328	3,248	59,547	105,472	3,449	3,675	61,990	983	6,475	138,246	11,285
Male	188,526	10,636	1,613	29,786	47,693	2,562	1,888	32,689	644	3,588	52,055	5,374
Female	235,172	18,692	1,634	29,762	57,779	887	1,787	29,301	340	2,888	86,192	5,911
Ambulance Service & Em	nergency Care	9										
Total	1,232	-	31	-	-	-	-	865	-	-	336	-
Male	31	-	31	-	-	-	-	-	-	-	-	-
Female	1,201	-	-	-	-	-	-	865	-	-	336	-
Domestic Care giving												
Total	2,665	-	-	594	1,686	-	-	-	-	88	-	297
Male	521	-	-	-	521	-	-	-	-	-	-	-
Female	2,144	-	-	594	1,165	-	-	-	-	88	-	297
Domestic Service												
Total	3,487	716	-	102	1,241	-	-	603	-	-	825	-
Male	908	_	-	-	621	-	-	-	-	-	287	-
Female	2,579	716	-	102	621	-	-	603	_	-	538	_
Embalming service	,											
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	_	-	-	-	-	-	-	-	-	-	-
Embroidery												
Total	11,085	1,470	28	2,647	2,944	40	-	781	-	298	2,700	176
Male	462	255	28	98	80	-	-	-	-	-		-
Female	10,623	1,215	-	2,549	2,863	40	-	781	-	298	2,700	176
Fashion Designing												
Total	18,467	1,724	-	1,794	4,534	-	77	301	40	171	9,145	681
Male	4,816	961	-	817	585	-	-	-	-	-	2,454	-
Female	13,651	763	-	978	3,950	-	77	301	40	171	6,691	681
Hairdressing Beauty Car	re											
Total	76,051	9,612	286	8,546	19,172	194	-	4,922	-	502	31,648	1,169
Male	9,991	1,727	-	1,068	1,432	-	-	95	-	43	5,480	146
Female	66,061	7,886	286	7,478	17,740	194	-	4,827	-	459	26,168	1,023
Housekeeping & Laundr	y Service											
Total	406	-	-	101	62	-	-	243	-	-	-	-

48

Table 3.8 Cont'd

Table 3.8 Cont d	Region											
Training status, Types	1	I					Benishangul-				Addis	Dire
of Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Male	62	-	-	-	62	-	-	_	-	-	-	-
Female	344	-	-	101	-	-	-	243	-	-	-	-
Mortuary & Cemetery Se												
Total	42	-	-	-	-	-	-	-	-	42	-	-
Male	42	-	-	-	-	-	-	-	-	42	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Music												
Total	5,178	-	-	513	719	-	39	164	-	-	3,589	153
Male	3,766	-	-	-	719	-	39	164	-	-	2,690	153
Female	1,413	-	-	513	-	-	-	-	-	-	899	-
Police												
Total	97,750	6,321	1,325	20,947	17,612	2,311	657	23,709		3,467	17,860	2,986
Male	83,720	4,955	890	17,144	16,345	2,153	657	19,988		3,202	15,441	2,607
Female	14,030	1,366	436	3,803	1,267	158	-	3,721	215	265	2,419	379
Pottery												
Total	988	-	-	-	34	-	-	-	-	-	954	-
Male	34	-	-	-	34	-	-	-	-	-	-	-
Female	954	-	-	-	-	-	-	-	-	-	954	-
Records & Archives Serv	ices											
Total	2,601	1,146	-	363	638	-	-	121	-	-	334	-
Male	604	-	-	270	-	-	-	-	-	-	334	-
Female	1,997	1,146	-	92	638	-	-	121	-	-	-	-
Secretarial Service												
Total	69,315	4,224	721	10,435	17,571	224	1,428	10,351	40	1,508	19,450	3,362
Male	6,371	, <u>-</u>	31	1,652	911	-	143	971	-	211	1,366	1,087
Female	62,944	4,224	690	8,784	16,660	224	1,284	9,381	40	1,297	18,084	2,276
Tour & Travel Service												
Total	730	-	_	-	57	_	-	_	-	-	673	_
Male	730	-	_	-	57	-	-	-	-	-	673	_
Female	_	_	_	_	_	_	_	_	_	_	_	_
Video & Photography												
Total	3,304	796	_	_	663	_	41	396	_	_	1,068	339
Male	1,967	796	_	_	139	_	41	396		_	346	247
Female	1,337	730	_	_	524	_	-	-	_	_	722	92
Waste Handling & Waste		_	_	_	524	_	_	_	_	_	1 22	32
Total	3,527	271			93	169		301	40	83	2,570	
			-	-			-		_		2,370	-
Male	342	- 074	-	-	-	400	-	301	40	-	- 0.570	-
Female	3,185	271	-	-	93	169	-	-	-	83	2,570	-
Weaving												

Table 3.8 Cont'd

Training status, Types						R	egion					
of Training status, Types							Benishangul-				Addis	Dire
or maining and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Total	4,540	133	-	1,316	639	-	-	661	-	41	1,751	-
Male	3,024	133	-	740	639	-	-	59	-	41	1,413	-
Female	1,516	-	-	577	-	-	-	602	-	-	337	-
Other Social Services												
Total	122,329	2,915	856	12,188	37,807	511	1,433	18,571	309	276	45,343	2,121
Male	71,137	1,810	633	7,997	25,549	409	1,007	10,714	264	49	21,571	1,134
Female	51,193	1,106	222	4,191	12,258	102	426	7,856	45	227	23,772	987

Table 3.9 Distribution of Urban Population by Five Years Age Group, Sex and Sex Ratio – COUNTRY TOTAL: 2015

	bex and Sex		11 TOTAL. 2013	
Age Group		Sex		Sex Ratio
3 % P	Total	Male	Female	
All Ages	16,598,199	7,839,679	8,758,520	90
0-4	1,704,779	875,713	829,066	106
5-9	1,512,044	740,230	771,814	96
10-14	1,790,429	858,124	932,305	92
15-19	2,041,908	853,760	1,188,148	72
20-24	2,011,484	836,505	1,174,979	71
25-29	2,024,794	978,477	1,046,317	94
30-34	1,276,614	663,030	613,583	108
35-39	1,174,449	559,061	615,388	91
40-44	713,737	397,066	316,672	125
45-49	561,559	278,068	283,491	98
50-54	422,014	195,936	226,078	87
55-59	345,943	154,230	191,713	80
60-64	310,206	126,910	183,297	69
65+	708,239	322,571	385,668	84

CHAPTER IV

ECONOMIC ACTIVITY STATUS

4.1 Introduction

Statistical data on the economic activity status of a population is essential for planning and formulating policies for better utilization of human resources of a country. The economic activity status indicates the size and distribution of the work force of economically active and not active population during a given reference period at a specified area.

In this regard, a wide range of information was collected on economic activity status of all persons aged five years and above in the 2015 UEUS. For the purpose of gathering information about the economic activities of children, the lower age limit was fixed at five years and hence data were collected from persons aged five years and above. However, for the purpose of measuring the economic activity status of adult population according to the Ethiopian situation, the lower age limit was fixed into ten years.

The 2015 Urban Employment and Unemployment Survey administered detailed labour force questions to measure the economic activity status of persons in urban areas of the country. These includes:-

- whether persons were engaged in economic / productive activities in the last seven days prior to the survey date or reasons for not being engaged;
- whether persons were looking for work or will be available for work in the coming one month or reason for not being available during the reference period; and
- whether persons were engaged in economic / productive activities during most of the last six months or reasons for not being engaged.

Every person aged ten years and above in the sampled household was asked if he/she was engaged in any productive activities during the last seven days, and in the last six months prior to the survey date. Based on the response obtained from respondents, the economic activity status of the population was determined by using the standard definitions of economic and non economic activities.

The 2015 Urban Employment and Unemployment Survey followed the ILO international standard definitions of economic and non-economic activities. The concept of economic activity as adopted by the 13th International Conference on Labour Statistics (ICLS,1982) is defined as the production of goods and/or services that falls within the United Nations System of National Accounts (SNA) production boundary (ILO, 1990). In this survey, economic/ productive activity is defined as work, which involves the production of goods and /or services for sale or exchange. In addition, production of goods and services for own consumption or own uses are also considered as economic activities. These include production of primary products (agriculture, hunting, fishing, forestry and logging, mining and quarrying), for own consumption; processing of primary products by the producers themselves; production of other commodities where part of it is sold on the market; and own account construction and fixed asset formation (expected life use of one year or more). Such economic activities could be performed for an individual, family or private enterprise, government establishment or social organization. The remuneration may be on daily, weekly, monthly, yearly or contract basis. The practical activities of apprentices are also considered as economic activities. On the other hand, unpaid household chores such as preparing food, cleaning the house, taking care of children are not considered as economic activities. Similarly, unpaid community and volunteer services are classified as non-economic activities.

Based on the above ILO definitions, data on economic activity status of persons aged ten years and above was presented in this report. Persons who were engaged in productive work /employed/ or available to be engaged /unemployed/ during the reference period are considered as economically active persons. While those persons who were neither

engaged nor available to be engaged in the production of economic goods and services during the reference period are considered as population not economically active.

4.2 Data Collection Approaches

The 2015 Urban Employment and Unemployment Survey used two approaches in the collection of data about the economic activity status of the population. These are current and usual activity status approaches. The main difference between these two approaches is the length of the reference period.

4.2.1 Current Activity Status Approach

The current activity status approach measures the economic activity status of a person based on a short reference period of seven days before the date of interview. In this approach, all persons aged ten years and above who were engaged in productive work or available to be engaged during the reference week were considered as currently active while those who were neither engaged nor available to be engaged in productive work were taken as currently not active.

4.2.2 Usual Activity Status Approach

The usual activity status approach measures the economic activity status of a person based on a long reference period of the six months prior to the date of interview. In the usual activity status approach, all persons aged ten years and above who were engaged in productive works or available to be engaged in the production of economic goods and/or services during most of the six months were classified as usually active. Persons who were neither employed nor unemployed during most of the six months categorized as usually not active.

The choice of appropriate measurement approach depends on the extent of variability of economic activities of the population during a specified period in a given the study area. In the rural areas where seasonal variation of activities is common, to collect the

economic activity status of the rural population including the slack and peak periods of agricultural activities, a longer reference period (usual activity status approach) is found to be more appropriate. On the other hand, in urban areas, activities are relatively less affected by seasonal variations and hence shorter reference period or current activity status approach is found to be more convenient. But in this survey, both approaches were used to collect data on economic activity status of urban population.

This chapter presents the economic activity status aged ten years and above, size and distribution of economically active and not economically active urban population, activity rate, and reasons for not being economically active of urban population by using only the current status approach at national, regional and major town levels.

4.3 Current Activity Status

As mentioned above, current activity status approach measures the economic activity status of a person in a short reference period, that is, the seven days prior to the date of the interview. In this approach a series of inquiries were administered to determine the economic activity status of the population during the reference week such as engagement in economic activity, attachment to the job, seeking and availability to work, reason for not seeking or available to work, etc. These questions were presented to those persons aged five years and above. For the purpose of reporting, persons aged ten years and above were used to classify the population into three mutually exclusive categories: employed, unemployed, and not in the labour force (Not Economically active). The employed and the unemployed population together constitute the labour force or the currently active population. The third category, those who were not in the labour force are called not currently active population.

As part of the economically active population, employed population consists of persons aged ten years and above who engaged in a productive activity or work at least for one hour during the seven days prior to the date of the interview. In addition, persons who had regular jobs, or business, or holdings to return to but who were temporarily absent from work (not at work or worked less than one hour) for various reasons such as illness or injury, holiday or vacation, strike or lockout, and seasonality of work, annual

leave, temporary closure of establishment were also considered as employed. For a person to be considered as absent from work, he/she must have formal attachment to the job.

The currently unemployed population also comprises those persons without work but looking for work or available and ready to work if any job is found during the reference period of the coming one month. The reference period of the coming one month refers to the survey week plus the next three weeks. If a person who is looking for work but engaged in productive activity during the reference week is recorded as employed but not as unemployed.

4.4 Economically Active and Not active Population and Activity Rate

Based on the above definition of economic activity status, the number of economically active population including employed and unemployed persons, population not active and activity rates of urban population aged ten years and above at national and regional level is presented by sex and age group in Table 4.1 and Table 4.1.1- 4.1.11 using the current status approach, respectively. Besides, the economic activity status and activity rate of selected major towns' population aged ten years and above is also presented in Table 4.2.

The extent of the economic participation of the population is measured using economic activity rates. Economic activity rate or economic participation rate is computed as the percentage of the economically active population over the total of active plus not active population aged ten years and above. The national estimate shows that 63.7 percent of the population aged ten years and above were economically active (Table 4.1). With respect to sex, the activity rate of male is 71.1 percent and the corresponding female activity rate is 57.2 percent. The age distributions of activity rate illustrate that it is low at younger and older age groups. The table also shows a relatively stable for middle ages between 20-59 years, and reaches its peak in age group 35-39 (Table 4.1).

As far as the economic activity status of major towns' population is concerned, 61.8 percent of the populations were economically active while the remaining 38.2 percent were found to be the population not economically active. In terms of sex, the activity rate of male (70.0 per cent) was higher by 15.3 per cent than female (54.9 per cent). Among the major towns, the highest activity rate was recorded in Bahir Dar town (71.9 percent) while the lowest was for Jijiga town (48.1 percent) (Table 4.2).

The size of economically active and not active population has its own implication on the development of the local as well as the national economy of a region. The high activity rate of a population implies the prevalence of huge potential of the human resource or human labour that foster sustainable economic development. On the other hand, the low economic activity rate also implies the lack of adequate and trained human labour or the existence of large number of economically non-productive population which has no contribution for development at the time. Hence, the economic activity rate of the urban population of the Ethiopia can be viewed from these perspectives.

The distribution of economically not active population of urban areas at national and regional level by age group, sex and reason for not being economically active (for those who were neither engaged nor available and/ or seeking work) during the last seven days are presented in Table 4.3 and Table 4.4, respectively. The reasons for not looking for or available to work include; home maker, pregnancy, student, injury/ disability, illness, too young, old age, pensioned, remittance, and other reasons. The result in Table 4.3 shows that the lion share of the population not active were students for the national (63.1 percent). The remaining per cent was the other groups of reasons for the economically not active population. The same holds true for the regions, too (Table 4.4).

Table 4.1 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)- COUNTRY TOTAL: 2015

Nate		Last Ocven L		y Status	Dacily- COUNT	KI IOIAL.
Age Group	All	F.	conomically Act		Economically	Activity
and Sex	Persons				Not Active	Rate
ΛΙΙ Λ σ.σ.ς		Employed	Unemployed	Total Active	NOT ACTIVE	
All Ages	10 001 070	7 000 004	4 40 4 007	0.547.000	4 004 055	00.7
Total	13,381,376	7,083,094	1,434,227	8,517,320	4,864,055	63.7
Male .	6,223,736	3,963,143	462,220	4,425,363	1,798,373	71.1
Female	7,157,639	3,119,950	972,007	4,091,957	3,065,682	57.2
10-14						
Total	1,790,429	194,969	24,390	219,359	1,571,070	12.3
Male	858,124	100,707	9,146	109,853	748,271	12.8
Female	932,305	94,262	15,244	109,506	822,799	11.7
15-19 Tatal	0.044.000	FCF 400	204 540	700.050	4 070 050	27.7
Total	2,041,908	565,103	204,546	769,650	1,272,258	37.7
Male	853,760	224,956	71,926	296,881	556,879	34.8
Female 20-24	1,188,148	340,148	132,621	472,768	715,379	39.8
Total	2,011,484	1,107,918	418,460	1,526,378	485,106	75.9
Male	836,505	546,853	137,166	684,020	152,485	81.8
Female	1,174,979	561,064	281,294	842,358	332,621	71.7
25-29	1,174,979	301,004	201,294	042,330	332,021	71.7
Total	2,024,794	1,455,213	307,539	1,762,753	262,041	87.1
Male	978,477	827,711	104,586	932,297	46,180	95.3
Female	1,046,317	627,502	202,953	830,456	215,861	95.3 79.4
30-34	1,040,517	021,302	202,933	030,430	213,001	73.4
Total	1,276,614	985,745	152,036	1,137,781	138,833	89.1
Male	663,030	599,578	42,939	642,517	20,513	96.9
Female	613,583	386,166	109,097	495,264	118,320	80.7
35-39	010,000	000,100	100,007	100,201	110,020	00.7
Total	1,174,449	935,320	116,225	1,051,546	122,903	89.5
Male	559,061	518,080	26,618	544,699	14,362	97.4
Female	615,388	417,240	89,607	506,847	108,541	82.4
40-44	0.0,000	,	00,00.	000,0		<u></u>
Total	713,737	578,390	55,802	634,192	79,545	88.9
Male	397,066	365,839	17,988	383,827	13,238	96.7
Female	316,672	212,551	37,815	250,365	66,307	79.1
45-49	•	•	,	•	·	
Total	561,559	421,698	54,645	476,342	85,217	84.8
Male	278,068	252,303	14,044	266,347	11,720	95.8
Female	283,491	169,394	40,601	209,995	73,496	74.1
50-54						
Total	422,014	288,013	30,941	318,954	103,060	75.6
Male	195,936	173,544	7,974	181,517	14,419	92.6
Female	226,078	114,470	22,967	137,437	88,641	60.8
55-59						
Total	345,943	210,275	31,904	242,179	103,764	70.0
Male .	154,230	128,640	10,468	139,108	15,122	90.2
Female	191,713	81,635	21,436	103,071	88,642	53.8
60-64	0.4.5.5.5			400 10=	,	
Total	310,206	145,598	14,841	160,439	149,768	51.7
Male .	126,910	90,332	6,664	96,996	29,913	76.4
Female	183,297	55,266	8,177	63,442	119,855	34.6
65+ Tatal	700.000	404.054	00.007	047.740	400 404	00.7
Total Male	708,239	194,851	22,897	217,748	490,491 175,271	30.7 45.7
	322,571	134,599	12,702	147,300	175,271	
Female	385,668	60,253	10,195	70,448	315,220	18.3

Table 4.1.1 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-TIGRAY REGION: 2015

Age Group	All	Activity Status		Activity
and Sex	Persons	Economically Active	Economically	Rate

1	1 [Total	Not Active	Ĺ
		Employed	Unemployed	Active	NOT ACTIVE	
All Ages	1	' '		'		
Total	952,370	477,813	109,321	587,134	365,237	61.6
Male	411,583	253,254	27,218	280,472	131,111	68.1
Female	540,787	224,559	82,103	306,662	234,126	56.7
10-14						
Total	116,810	6,281	1,253	7,535	109,275	6.5
Male	56,402	2,213	591	2,804	53,598	5.0
Female	60,408	4,068	662	4,730	55,677	7.8
15-19						
Total	154,538	42,231	18,958	61,189	93,350	39.6
Male	67,758	16,565	8,108	24,673	43,085	36.4
Female	86,780	25,665	10,850	36,515	50,265	42.1
20-24						
Total	146,474	74,609	37,978	112,587	33,886	76.9
Male	44,307	27,680	7,766	35,446	8,861	80.0
Female	102,167	46,929	30,212	77,142	25,026	75.5
25-29						
Total	143,397	108,332	18,047	126,379	17,018	88.1
Male	63,153	56,044	3,801	59,845	3,308	94.8
Female	80,244	52,288	14,246	66,534	13,710	82.9
30-34						
Total	101,199	74,997	15,205	90,203	10,996	89.1
Male	52,212	47,385	4,045	51,430	781	98.5
Female	48,987	27,612	11,160	38,772	10,215	79.1
35-39						
Total	71,373	54,086	5,930	60,016	11,358	84.1
Male .	30,622	28,626	90	28,716	1,906	93.8
Female	40,751	25,460	5,839	31,299	9,452	76.8
40-44	00.004	00.407	4.400	00.000	0.055	0.4.7
Total	39,684	29,467	4,162	33,629	6,055	84.7
Male	22,539	20,162	1,544	21,705	834	96.3
Female	17,144	9,305	2,619	11,924	5,220	69.6
45-49	25 527	26.062	2 722	20.705	E 7E0	83.8
Total Male	35,537 16,500	26,063 16,193	3,722 153	29,785 16,346	5,752 153	83.8 99.1
Female	19,038	9,869	3,569	13,439	5,599	70.6
50-54	19,030	9,009	3,309	13,439	3,399	70.0
Total	31,595	21,511	1,260	22,771	8,824	72.1
Male	17,425	15,541	135	15,676	1,749	90.0
Female	14,170	5,970	1,125	7,095	7,075	50.1
55-59	,	3,3.3	.,0	.,000	.,	
Total	21,438	13,031	959	13,990	7,448	65.3
Male	7,149	5,982	229	6,211	938	86.9
Female	14,289	7,049	730	7,779	6,510	54.4
60-64	,	, -	_	,	,	
Total	22,037	9,713	363	10,077	11,961	45.7
Male	8,081	6,637	-	6,637	1,443	82.1
Female	13,957	3,076	363	3,439	10,517	24.6
65+	-,	-,-		-,	-,	-
Total	68,288	17,492	1,482	18,974	49,314	27.8
Male	25,436	10,226	756	10,982	14,454	43.2
Female	42,852	7,267	726	7,993	34,859	18.7

Female 42,852 7,267 726 7,993 34,859 18.7

Table 4.1.2 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-AFFAR REGION: 2015

Age Group	All	Eco	onomically Activ	ve		Activity
and Sex	Persons			Total	Economically	Rate
		Employed	Unemployed	Active	Not Active	
All Ages						
Total	216,983	131,439	19,977	151,415	65,567	69.8
Male	108,772	75,452	6,894	82,346	26,426	75.7

Female 10-14	108,211	55,987	13,083	69,069	39,142	63.8
Total	29,479	4,059	224	4,283	25,196	14.5
Male	15,835	1,826	224	2,050	13,785	12.9
Female	13,643	2,233		2,233	11,411	16.4
15-19	,	_,		_,	,	
Total	28,316	9,262	1,970	11,232	17,084	39.7
Male	11,399	3,692	1,083	4,775	6,624	41.9
Female	16,918	5,570	887	6,457	10,461	38.2
20-24						
Total	30,313	18,123	6,704	24,827	5,486	81.9
Male	16,846	11,287	3,230	14,517	2,329	86.2
Female	13,466	6,835	3,475	10,310	3,156	76.6
25-29						
Total	36,477	27,089	4,748	31,837	4,641	87.3
Male	15,061	13,825	754	14,580	481	96.8
Female	21,416	13,263	3,994	17,257	4,159	80.6
30-34 Total	21,738	18,064	1,799	19,863	1,875	91.4
Male	12,072	11,713	1,799	19,003	350	91. 4 97.1
Female	9,666	6,351	1,790	8,141	1,525	84.2
35-39	9,000	0,331	1,790	0,141	1,323	04.2
Total	22,868	18,805	1,402	20,207	2,661	88.4
Male	12,303	11,486	544	12,029	274	97.8
Female	10,565	7,319	858	8,177	2,387	77.4
40-44	10,000	7,010	000	0,177	2,007	11.4
Total	12,518	10,597	1,078	11,674	844	93.3
Male	6,660	6,291	370	6,660	-	100.0
Female	5,858	4,306	708	5,014	844	85.6
45-49	2,222	1,000		2,2 : :		
Total	10,638	9,295	725	10,020	619	94.2
Male	5,117	4,708	377	5,086	31	99.4
Female	5,522	4,587	347	4,934	588	89.4
50-54	-,	1,001		1,00		•••
Total	8,602	6,314	256	6,571	2,031	76.4
Male	4,382	4,338	-	4,338	44	99.0
Female	4,219	1,976	256	2,233	1,986	52.9
55-59	1,210	1,070	200	2,200	1,000	02.0
Total	6,466	3,769	806	4,575	1,891	70.8
Male	2,348	1,714	39	1,753	595	74.6
Female	4,118	2,055	768	2,822	1,295	68.5
60-64	4,110	2,055	700	2,022	1,295	00.5
	F 000	0.744	440	0.004	4.044	75.5
Total	5,062	3,711	110	3,821	1,241	75.5
Male	3,307	2,721	110	2,831	476	85.6
Female	1,755	990	-	990	765	56.4
65+		0.0		0.500	0.555	
Total	4,506	2,352	154	2,506	2,000	55.6
Male	3,439	1,850	154	2,004	1,435	58.3
Female	1,066	502	-	502	565	47.0

Female 1,066 502 - 502 565 47.0

Table 4.1.3 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-AMHARA REGION: 2015

		•	Activity	Status		
Age Group	All	Eco	onomically Activ	/e		Activity
and Sex	Persons			Total	Economically	Rate
		Employed	Unemployed	Active	Not Active	
All Ages						
Total	2,722,586	1,449,091	339,154	1,788,246	934,341	65.7
Male	1,240,992	795,183	100,100	895,283	345,710	72.1
Female	1,481,594	653,908	239,055	892,963	588,631	60.3
10-14						
Total	352,195	44,588	1,674	46,262	305,934	13.1
Male	176,748	22,776	431	23,207	153,541	13.1
Female	175,448	21,812	1,243	23,055	152,393	13.1

iotai	- 10,371	100,773	- 0,032	132,307	200,707	37.1
Male	164,308	42,682	14,065	56,747	107,561	34.5
Female	246,663	63,793	32,027	95,820	150,843	38.8
20-24	,	,	,	,	•	
Total	425,178	232,951	109,704	342,654	82,524	80.6
Male	172,718	109,983	31,790	141,773	30,944	82.1
Female	252,460	122,967	77,914	200,881	51,579	79.6
25-29	232,400	122,307	77,514	200,001	31,379	73.0
Total	412,707	299,011	71,007	370,018	42,689	89.7
Male	203,310	176,619	22,335	198,954	4,356	97.9
Female	209,397	122,393	48,672	171,064	38,333	81.7
30-34						
Total	252,372	190,781	35,281	226,062	26,310	89.6
Male	127,941	116,417	9,702	126,119	1,821	98.6
Female	124,431	74,364	25,579	99,943	24,489	80.3
35-39						
Total	225,093	182,038	25,317	207,355	17,738	92.1
Male	106,123	97,487	6,353	103,841	2,283	97.8
Female	118,969	84,550	18,963	103,514	15,456	87.0
40-44	-,	, , , , , ,	-,	, -	-,	
Total	129,669	108,134	12,778	120,913	8,756	93.2
Male	74,582	69,255	3,272	72,528	2,055	97.2
Female	55,087	38,879	9,506	48,385	6,702	87.8
45-49	,	,-	-,	-,	-, -	
Total	109,362	87,212	10,106	97,318	12,044	89.0
Male	50,160	46,297	2,193	48,489	1,670	96.7
Female	59,202	40,915	7,913	48,828	10,374	82.5
50-54	,	,	,	,	•	
Total	92,623	62,033	12,287	74,320	18,303	80.2
Male	34,403	30,169	2,671	32,840	1,563	95.5
Female	58,220	31,864	9,616	41,480	16,739	71.2
55-59	,	, , , ,	-,-	,	-,	
Total	79,419	50,755	6,735	57,490	21,929	72.4
Male	35,936	31,112	1,628	32,740	3,196	91.1
Female	43,483	19,643	5,108	24,750	18,733	56.9
60-64	10,100	,	2,:00	_ :,: = :	,	
Total	60,732	34,241	3,993	38,235	22,498	63.0
Male	25,448	20,928	1,675	22,604	2,844	88.8
Female	35,285	13,313	2,318	15,631	19,653	44.3
65+	00,200	10,010	2,0.0	10,001	10,000	
Total	172,264	50,872	4,180	55,052	117,213	32.0
Male	69,316	31,457	3,984	35,441	33,875	51.1
Female	102,948	19,415	196	19.611	83,338	19.0
гентане	102,940	19,413	190	19,011	03,330	19.0

15-19 Total

410,971

106,475

46,092

152,567

258,404

37.1

Table 4.1.4 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-OROMIA REGION: 2015

			Activity Status				
Age Group	All	Eco	Economically Active			Activity	
and Sex	Persons			Total	Economically	Rate	
		Employed	Unemployed	Active	Not Active		
All Ages							
Total	3,898,166	2,102,417	386,972	2,489,389	1,408,777	63.9	
Male	1,863,092	1,213,723	128,166	1,341,888	521,204	72.0	
Female	2,035,074	888,694	258,806	1,147,501	887,573	56.4	
10-14							
Total	563,599	67,735	7,817	75,552	488,047	13.4	
Male	272,782	41,900	2,873	44,773	228,009	16.4	
Female	290,817	25,835	4,944	30,778	260,038	10.6	
15-19							
Total	632,697	185,609	64,916	250,525	382,172	39.6	
Male	275,662	82,315	23,927	106,242	169,420	38.5	
Female	357,035	103,293	40,990	144,283	212,752	40.4	
20-24							

TOtal	002,204	340,212	114,004	400,293	141,900	70.4
Male	255,222	183,385	36,756	220,141	35,081	86.3
Female	347,041	162,826	77,328	240,154	106,887	69.2
25-29						
Total	608,648	434,406	87,835	522,240	86,408	85.8
Male	302,293	257,771	29,632	287,404	14,889	95.1
Female	306,355	176,634	58,202	234,837	71,518	76.7
30-34						
Total	349,526	280,755	34,057	314,812	34,714	90.1
Male	191,280	174,014	9,880	183,894	7,386	96.1
Female	158,246	106,741	24,177	130,918	27,328	82.7
35-39	,	,	,	,-	,	
Total	334,301	272,233	30,808	303,041	31,260	90.6
Male	159,482	148,999	8,129	157,128	2,354	98.5
Female	174,819	123,234	22,679	145,913	28,906	83.5
40-44						
Total	210,778	171,711	12,518	184,229	26,549	87.4
Male	121,589	111,982	5,956	117,938	3,651	97.0
Female	89,189	59,729	6,562	66,291	22,898	74.3
45-49						
Total	163,564	120,390	14,673	135,063	28,501	82.6
Male	79,937	72,919	3,167	76,086	3,852	95.2
Female	83,627	47,471	11,507	58,977	24,650	70.5
50-54						
Total	95,344	72,481	4,851	77,332	18,012	81.1
Male	50,811	45,658	1,690	47,348	3,463	93.2
Female	44,533	26,823	3,160	29,984	14,549	67.3
55-59						
Total	87,204	57,461	5,137	62,598	24,605	71.8
Male	39,523	34,810	1,700	36,510	3,013	92.4
Female	47,681	22,651	3,437	26,088	21,593	54.7
60-64	17,001	22,00	0, 101	20,000	21,000	0
Total	87,711	44,309	4,162	48,470	39,241	55.3
Male	37,757	27,178	1,749	28,927	8,830	76.6
Female	49,954	17,131	2,413	19,544	30,410	39.1
65+	10,001	17,101	2,110	10,011	00,110	00.1
Total	162,531	49,116	6,116	55,232	107,299	34.0
Male	76,752	32,790	2,707	35,497	41,255	46.2
Female	85,778	16,325	3,409	19,734	66,044	23.0
remale	00,770	10,323	3,409	19,734	00,044	23.0

Total

602,264

346,212

114,084

460,295

141,968

76.4

Table 4.1.5 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-SOMALI REGION: 2015

	Activity Status					
Age Group	All	Eco	onomically Activ	ve		Activity
and Sex	Persons			Total	Economically	Rate
		Employed	Unemployed	Active	Not Active	
All Ages						
Total	273,737	119,082	20,846	139,928	133,809	51.1
Male	131,760	67,114	8,066	75,181	56,580	57.1
Female	141,976	51,967	12,780	64,747	77,229	45.6
10-14						
Total	52,965	6,236	1,043	7,278	45,687	13.7
Male	27,439	2,695	227	2,923	24,517	10.7
Female	25,526	3,540	816	4,356	21,170	17.1
15-19						
Total	47,750	10,505	1,838	12,343	35,407	25.8
Male	22,099	3,950	650	4,600	17,499	20.8
Female	25,651	6,555	1,188	7,743	17,908	30.2
20-24						
Total	35,088	13,337	5,952	19,289	15,799	55.0
Male	15,543	6,860	2,415	9,274	6,268	59.7
Female	19,546	6,477	3,538	10,015	9,531	51.2
25-29						
Total	34,163	21,263	3,479	24,742	9,421	72.4

Male	16,071	13,257	836	14,094	1,977	87.7	
Female	18,093	8,006	2,643	10,649	7,444	58.9	
30-34							
Total	22,497	15,175	2,731	17,905	4,592	79.6	
Male	10,907	9,835	750	10,585	322	97.0	
Female	11,590	5,339	1,981	7,320	4,270	63.2	
35-39							
Total	21,200	15,790	1,830	17,619	3,581	83.1	
Male	9,202	8,408	664	9,072	130	98.6	
Female	11,998	7,382	1,165	8,547	3,451	71.2	
40-44							
Total	18,841	14,072	1,451	15,523	3,318	82.4	
Male	8,729	7,918	492	8,410	319	96.3	
Female	10,112	6,154	959	7,113	2,998	70.3	
45-49							
Total	12,348	9,869	793	10,662	1,686	86.3	
Male	6,989	5,840	742	6,582	407	94.2	
Female	5,359	4,029	51	4,080	1,279	76.1	
50-54							
Total	8,798	5,320	968	6,288	2,510	71.5	
Male	4,762	3,568	631	4,200	562	88.2	
Female	4,036	1,752	336	2,088	1,947	51.7	
55-59							
Total	3,218	1,331	226	1,556	1,661	48.4	
Male .	1,885	1,024	174	1,198	687	63.6	
Female	1,333	307	52	359	975	26.9	
60-64	0.405	0.004	0.4.0	0.004	0.005	=0.4	
Total	6,185	3,091	210	3,301	2,885	53.4	
Male	3,231	1,855	210	2,065	1,166	63.9	
Female	2,954	1,236	-	1,236	1,719	41.8	
65+ Total	10.605	2.004	226	2 420	7.064	22.0	
Total	10,685	3,094	326	3,420	7,264	32.0	
Male	4,905	1,903	276	2,179	2,726	44.4	
Female	5,780	1,191	51	1,242	4,538	21.5	-

Table 4.1.6 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-**BENISHANGUL-GUMUZ REGION: 2015**

Age Group	All	Eco	onomically Activ	ve		Activity
and Sex	Persons			Total	Economically	Rate
		Employed	Unemployed	Active	Not Active	
All Ages						
Total	157,710	107,053	9,731	116,784	40,926	74.0
Male	77,083	58,730	2,830	61,559	15,524	79.9
Female	80,627	48,323	6,901	55,224	25,403	68.5
10-14						
Total	22,907	7,388	43	7,430	15,476	32.4
Male	9,561	2,760	43	2,802	6,758	29.3
Female	13,346	4,628	-	4,628	8,718	34.7
15-19						
Total	23,054	9,970	1,012	10,982	12,072	47.6
Male	11,884	4,889	348	5,236	6,647	44.1
Female	11,170	5,081	664	5,745	5,425	51.4
20-24						
Total	26,866	18,860	4,000	22,861	4,005	85.1
Male	10,504	8,763	1,208	9,971	533	94.9
Female	16,361	10,098	2,792	12,890	3,472	78.8
25-29						
Total	27,224	23,990	1,450	25,440	1,783	93.4
Male	15,121	14,366	394	14,760	361	97.6
Female	12,103	9,624	1,056	10,680	1,423	88.2
30-34						
Total	14,432	13,065	876	13,941	491	96.6

Male	7,513	7,378	135	7,513	-	100.0
Female	6,919	5,687	741	6,428	491	92.9
35-39						
Total	13,439	11,685	532	12,218	1,222	90.9
Male	7,116	6,825	-	6,825	291	95.9
Female	6,323	4,861	532	5,393	930	85.3
40-44						
Total	7,539	6,630	605	7,235	303	96.0
Male	3,468	3,392	77	3,468	-	100.0
Female	4,070	3,238	529	3,767	303	92.5
45-49						
Total	7,049	6,087	253	6,340	709	89.9
Male	4,169	4,050	119	4,169	-	100.0
Female	2,880	2,037	134	2,171	709	75.4
50-54						
Total	4,966	4,104	215	4,319	647	87.0
Male	2,842	2,671	172	2,842	-	100.0
Female	2,124	1,434	43	1,477	647	69.5
55-59						
Total	3,918	2,293	349	2,643	1,275	67.4
Male	1,936	1,632	172	1,804	132	93.2
Female	1,982	661	177	839	1,143	42.3
60-64						
Total	1,778	977	163	1,140	637	64.1
Male	759	597	163	759	-	100.0
Female	1,018	381	-	381	637	37.4
65+						
Total	4,539	2,001	233	2,234	2,305	49.2
Male	2,209	1,408	-	1,408	801	63.7
Female	2,330	593	233	826	1,504	35.4

Table 4.1.7 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-S.N.N.P REGION: 2015

Age Group	All	Eco	onomically Activ	ve		Activity
and Sex	Persons			Total	Economically	Rate
		Employed	Unemployed	Active	Not Active	
All Ages						
Total	1,887,693	1,096,887	142,947	1,239,834	647,859	65.7
Male	897,420	603,653	46,009	649,662	247,758	72.4
Female	990,273	493,233	96,939	590,172	400,100	59.6
10-14						
Total	301,593	39,806	5,106	44,912	256,681	14.9
Male	134,093	18,628	1,174	19,802	114,291	14.8
Female	167,500	21,178	3,932	25,110	142,390	15.0
15-19						
Total	320,875	108,119	27,595	135,714	185,161	42.3
Male	132,561	45,420	10,016	55,436	77,125	41.8
Female	188,314	62,699	17,579	80,278	108,036	42.6
20-24						
Total	277,103	180,025	36,141	216,166	60,936	78.0
Male	125,632	92,276	12,436	104,712	20,920	83.3
Female	151,471	87,749	23,705	111,454	40,017	73.6
25-29						
Total	279,538	223,124	32,126	255,250	24,288	91.3
Male	139,390	123,468	11,276	134,744	4,646	96.7
Female	140,148	99,656	20,850	120,506	19,642	86.0
30-34						
Total	149,010	127,137	12,676	139,813	9,197	93.8
Male	79,612	75,130	3,612	78,742	871	98.9
Female	69,398	52,007	9,064	61,071	8,327	88.0
35-39						
Total	163,312	141,276	8,956	150,232	13,080	92.0
Male	76,667	73,033	1,310	74,343	2,324	97.0
Female	86,645	68,243	7,646	75,889	10,756	87.6

40-44						
Total	109,822	95,357	4,809	100,167	9,655	91.2
Male	62,386	59,331	1,229	60,560	1,826	97.1
Female	47,436	36,026	3,581	39,607	7,829	83.5
45-49						
Total	75,533	64,394	4,554	68,948	6,585	91.3
Male	42,846	40,865	661	41,526	1,320	96.9
Female	32,687	23,529	3,894	27,423	5,264	83.9
50-54						
Total	56,464	42,277	4,162	46,438	10,025	82.2
Male	29,101	26,341	1,258	27,599	1,503	94.8
Female	27,362	15,936	2,904	18,840	8,523	68.9
55-59						
Total	39,865	28,495	2,497	30,991	8,873	77.7
Male	22,323	19,472	1,460	20,932	1,391	93.8
Female	17,542	9,023	1,037	10,059	7,482	57.3
60-64						
Total	34,855	18,241	2,052	20,293	14,562	58.2
Male	10,038	7,137	891	8,027	2,011	80.0
Female	24,817	11,105	1,161	12,266	12,551	49.4
65+	,	·	·	·	,	
Total	79,724	28,636	2,272	30,908	48,815	38.8
Male	42,771	22,554	687	23,240	19,531	54.3
Female	36,953	6,083	1,586	7,668	29,284	20.8

Table 4.1.8 Urban population Aged Ten Years and above by Age Group, Sex, Activity
Status and Rate during the Last Seven Days (Current Status Approach)GAMBELLA REGION: 2015

Ago Croup	All		Activity	Status		A ativity
Age Group and Sex		Ec	onomically Activ	/e	Economically	Activity Rate
and Sex	Persons	Employed	Unemployed	Total Active	Not Active	Rate
All Ages						
Total	94,945	61,809	5,101	66,910	28,035	70.5
Male	45,805	31,771	1,511	33,282	12,523	72.7
Female	49,141	30,038	3,590	33,628	15,513	68.4
10-14						
Total	16,286	4,843	121	4,965	11,322	30.5
Male	7,967	2,040	40	2,081	5,886	26.1
Female	8,319	2,803	81	2,884	5,436	34.7
15-19	45.000		710	7 400	0.505	40.0
Total	15,992	6,775	713	7,488	8,505	46.8
Male Female	7,729	3,251	162	3,413	4,316	44.2
20-24	8,264	3,524	551	4,075	4,189	49.3
Zu-Z4 Total	40.000	10 110	4.004	44.027	0.040	82.7
Male	16,983	12,116 5,364	1,921 536	14,037 5,900	2,946	82.7 81.2
Female	7,264 9,719	6,753	1,384	8,137	1,364 1,582	83.7
25-29	9,719	0,755	1,304	0,137	1,362	03.7
Total	15,908	13,679	1,123	14,802	1,105	93.1
Male -	7,715	7,128	359	7,487	227	97.1
Female	8,193	6,551	764	7,315	878	89.3
30-34						
Total	10,160	8,929	455	9,384	776	92.4
Male	4,981	4,940	-	4,940	41	99.2
Female	5,179	3,989	455	4,444	735	85.8
35-39	•	•		,		
Total	6,677	6,056	134	6,190	487	92.7
Male	3,941	3,839	41	3,880	62	98.4
Female	2,735	2,217	93	2,310	425	84.5
40-44	2,733	2,217	95	2,310	423	04.5
Total	3,861	3,489	174	3,663	198	94.9
Male	2,091	1,976	93	2,069	22	99.0
Female	1,770	1,513	81	1,594	176	90.0
45-49	.,	.,3.0	•	.,		
Total	3,342	2,652	364	3,016	326	90.2
1 0101	0,072	2,302	304	3,310	320	00.2

Male	1,344	1,119	204	1,323	22	98.4	
Female	1,997	1,534	159	1,693	304	84.8	
50-54							
Total	2,213	1,995	-	1,995	218	90.2	
Male	1,201	1,180	-	1,180	22	98.2	
Female	1,012	816	-	816	196	80.6	
55-59	,						
Total	1,206	628	75	703	503	58.3	
Male	744	582	75	657	88	88.2	
Female	462	46	-	46	416	10.0	
60-64							
Total	870	341	-	341	529	39.2	
Male	375	177	-	177	198	47.1	
Female	495	165	-	165	330	33.2	
65+							
Total	1,448	305	22	326	1,122	22.5	
Male	453	176	-	176	277	38.9	
Female	995	129	22	150	845	15.1	

Table 4.1.9 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-HARARI REGION: 2015

and Sex Persons Total Economically Ra	vity ite
	ite
Francisco de la la constancia de la Autoria	
Employed Unemployed Active Not Active	
All Ages	
	63.2
Male 49,819 29,900 2,716 32,616 17,203	65.5
Female 56,677 29,210 5,493 34,703 21,974	61.2
10-14	
Total 14,752 567 97 663 14,089	4.5
Male 8,160 181 49 229 7,930	2.8
Female 6,592 386 48 434 6,158	6.6
15-19	
	21.3
Male 6,411 729 317 1,046 5,365	16.3
	25.6
20-24	
Total 11,449 6,836 1,611 8,446 3,003	73.8
Male 4,494 2,840 528 3,368 1,126	74.9
Female 6,955 3,996 1,083 5,078 1,877	73.0
25-29	
Total 13,804 10,935 1,863 12,798 1,006	92.7
	97.4
Female 6,917 4,603 1,487 6,090 827	0.88
30-34	
Total 9,960 8,560 697 9,256 704	92.9
Male 4,839 4,462 229 4,691 149	96.9
Female 5,120 4,098 468 4,565 555	89.2
35-39	
	89.8
· · · · · · · · · · · · · · · · · · ·	93.9
	87.5
40-44	
Total 7,203 6,408 586 6,995 208	97.1
Male 3,901 3,599 262 3,861 40	99.0
Female 3,302 2,810 324 3,134 168	94.9
45-49	
Total 6,005 5,173 447 5,620 385	93.6
Male 3,201 3,069 132 3,201 -	0.00
	86.3
50-54	

Total	5,044	4,057	237	4,294	750	85.1	
Male	2,573	2,227	99	2,326	247	90.4	
Female	2,470	1,830	138	1,968	502	79.7	
55-59							
Total	3,020	1,798	365	2,163	858	71.6	
Male	1,297	1,071	186	1,257	41	96.9	
Female	1,723	727	179	906	817	52.6	
60-64							
Total	3,292	1,848	41	1,888	1,404	57.4	
Male	1,531	1,138	-	1,138	394	74.3	
Female	1,761	710	41	751	1,010	42.6	
65+							
Total	6,518	1,581	215	1,795	4,722	27.5	
Male	2,249	691	86	777	1,472	34.5	
Female	4,268	889	129	1,019	3,250	23.9	

Table 4.1.10 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-ADDIS ABABA CITY ADMINISTRATION: 2015

		Activity Status					
Age Group	All	Eco	onomically Activ	ve		Activity	
and Sex	Persons			Total	Economically	Rate	
		Employed	Unemployed	Active	Not Active		
All Ages		1 3				•	
Total	2,844,395	1,362,069	366,227	1,728,296	1,116,099	60.8	
Male	1,289,919	772,367	130,123	902,490	387,430	70.0	
Female	1,554,476	589,703	236,103	825,806	728,670	53.1	
10-14	.,,	,		,	1 = 0,010		
Total	288,827	11,109	6,677	17,786	271,040	6.2	
Male	132,572	4,394	3,339	7,733	124,839	5.8	
Female	156,254	6,715	3,338	10,053	146,201	6.4	
15-19	100,201	0,7 10	0,000	10,000	110,201	0.1	
Total	358,193	73,439	37,991	111,430	246,763	31.1	
Male	136,563	17,715	12,287	30,002	106,562	22.0	
Female	221,629	55,724	25,704	81,428	140,201	36.7	
20-24	221,020	00,121	20,701	01,120	1.10,201	00	
Total	410,305	189,657	94,289	283,945	126,360	69.2	
Male	170,339	89,901	38,828	128,729	41,610	75.6	
Female	239,966	99,756	55,460	155,216	84,750	64.7	
25-29	_00,000	00,.00	33, .33	.00,0	0.,.00	•	
Total	423,669	272,641	81,166	353,807	69,862	83.5	
Male	195,870	148,003	33,267	181,270	14,600	92.5	
Female	227,799	124,638	47,900	172,537	55,261	75.7	
30-34	,	,	,	,	,	-	
Total	327,199	234,644	45,908	280,552	46,648	85.7	
Male	162,345	140,195	13,786	153,982	8,363	94.8	
Female	164,854	94,449	32,122	126,570	38,284	76.8	
35-39	,	,	,	•	,		
Total	284,287	209,490	36,758	246,248	38,039	86.6	
Male	140,316	127,903	8,211	136,113	4,203	97.0	
Female	143,971	81,587	28,547	110,135	33,837	76.5	
40-44	,	,	,	•	,		
Total	160,599	122,301	16,532	138,833	21,766	86.4	
Male	85,122	76,663	4,269	80,932	4,190	95.1	
Female	75,476	45,637	12,263	57,901	17,576	76.7	
45-49	70,170	10,007	12,200	07,001	17,070	70.7	
Total	122,300	77,769	17,061	94,830	27,470	77.5	
Male	60,128	50,325	5,629	55,954	4,174	93.1	
Female	62,172	27,444	11,432	38,876	23,296	62.5	
50-54	02,172	21,177	11,702	00,070	20,200	02.0	
Total	108,325	62,695	5,526	68,221	40,104	63.0	
Male	45,443	39,384	974	40,358	5,085	88.8	
Female	62,882	23,311	4,552	27,863	35,019	44.3	
55-59	02,002	20,011	7,552	21,000	33,013	77.0	
Total	92,115	45,449	13,729	59,178	32,937	64.2	
iotai	52,115	70,770	10,120	55,176	02,001	07.2	

Male Female	37,064 55,051	27,961 17,487	4,338 9,391	32,299 26,878	4,765 28,173	87.1 48.8	
60-64							
Total	82,066	26,286	3,394	29,680	52,386	36.2	
Male	33,743	20,138	1,691	21,829	11,914	64.7	
Female	48,323	6,147	1,704	7,851	40,472	16.2	
65+							
Total	186,510	36,591	7,195	43,785	142,724	23.5	
Male	90,413	29,784	3,504	33,288	57,124	36.8	
Female	96,097	6,806	3,691	10,497	85,600	10.9	

Table 4.1.11 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-DIRE DAWAADMINISTRATION:2015

		DIMINIOTRA	Activity	Status	T	
Age Group	AII	Eco	onomically Activ			Activity
and Sex	Persons			Total	Economically	Rate
		Employed	Unemployed	Active	Not Active	
All Ages	· '	• •			<u>'</u>	
Total	226,294	116,325	25,742	142,066	84,227	62.8
Male	107,490	61,997	8,587	70,585	36,905	65.7
Female	118,804	54,327	17,154	71,482	47,322	60.2
10-14						
Total	31,017	2,358	334	2,693	28,324	8.7
Male	16,564	1,295	153	1,448	15,117	8.7
Female	14,452	1,064	181	1,245	13,208	8.6
15-19						
Total	35,714	10,537	2,702	13,240	22,474	37.1
Male	17,386	3,747	965	4,712	12,674	27.1
Female	18,328	6,790	1,738	8,528	9,800	46.5
20-24						
Total	29,463	15,192	6,077	21,270	8,193	72.2
Male .	13,637	8,515	1,673	10,188	3,448	74.7
Female	15,826	6,677	4,404	11,081	4,745	70.0
25-29						
Total	29,258	20,743	4,695	25,438	3,820	86.9
Male	13,607	10,896	1,555	12,451	1,155	91.5
Female	15,652	9,847	3,140	12,987	2,665	83.0
30-34 Total	10 501	12 620	2.252	15 000	2 521	06.2
Total Male	18,521 9,328	13,638 8,109	2,352 790	15,990 8,899	2,531 429	86.3 95.4
Female	9,193	5,529	1,562	7,091	2,102	77.1
35-39	3,133	3,323	1,502	7,031	2,102	77.1
Total	20,257	14,696	3,266	17,962	2,295	88.7
Male	9,013	7,914	822	8,736	277	96.9
Female	11,244	6,781	2,444	9,225	2,018	82.0
40-44	,	0,701	_,	0,220	2,010	02.0
Total	13,224	10,223	1,108	11,331	1,893	85.7
Male	5,997	5,272	424	5,696	301	95.0
Female	7,227	4,952	684	5,636	1,592	78.0
45-49						
Total	15,880	12,794	1,947	14,740	1,139	92.8
Male	7,677	6,918	668	7,586	91	98.8
Female	8,203	5,876	1,279	7,155	1,048	87.2
50-54						
Total	8,041	5,225	1,179	6,404	1,637	79.6
Male	2,990	2,466	344	2,811	179	94.0
Female	5,051	2,759	835	3,594	1,458	71.1
55-59						
Total	8,074	5,266	1,026	6,292	1,782	77.9
Male	4,024	3,280	468	3,748	277	93.1
Female	4,050	1,986	558	2,544	1,505	62.8
60-64						
Total	5,618	2,839	353	3,193	2,425	56.8

Male Female	2,639 2,979	1,826 1,013	177 177	2,002 1,190	636 1,789	75.9 39.9	
65+							
Total	11,227	2,813	702	3,515	7,713	31.3	
Male	4,628	1,759	549	2,308	2,320	49.9	
Female	6,599	1,053	153	1,207	5,393	18.3	

Table 4.2 Populations of Major Towns Aged Ten Years and Above by Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach): 2015

rate daring	the Last Seve					
Major Town	All		•	y Status		Activity
and Sex	Persons	Е	conomically Acti		Economically	Rate
		Employed	Unemployed	Total Active	Not Active	
MAJOR TOWNS TOTAL		Zilipioyou	Champioyed	7101170		
Total	5,120,995	2,555,307	610,986	3,166,293	1,954,702	61.8
Male	2,342,960	1,435,723	204,395	1,640,118	702,843	70.0
Female	2,778,035	1,119,585	406,591	1,526,176	1,251,859	54.9
MEKELE TOWN	_,,	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	100,001	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	,,	
Total	247,325	131,641	24,574	156,214	91,111	63.2
Male	111,726	70,849	7,043	77,892	33,835	69.7
Female	135,599	60,792	17,531	78,323	57,277	57.8
ASAYITA TOWN	•	,	·	·	•	
Total	20,428	12,819	1,299	14,118	6,310	69.1
Male	10,263	7,356	412	7,769	2,494	75.7
Female	10,166	5,463	887	6,350	3,816	62.5
GONDER TOWN						
Total	255,459	148,668	26,351	175,018	80,441	68.5
Male	111,299	77,623	7,853	85,476	25,822	76.8
Female	144,161	71,045	18,498	89,542	54,618	62.1
DESSIE TOWN						
Total	154,384	74,940	23,326	98,266	56,119	63.7
Male	70,111	44,569	6,878	51,447	18,664	73.4
Female	84,273	30,371	16,448	46,819	37,454	55.6
BAHIR DAR TOWN						
Total	203,038	121,879	24,142	146,021	57,016	71.9
Male	90,208	62,613	8,315	70,928	19,280	78.6
Female	112,829	59,266	15,827	75,093	37,736	66.6
BISHOFTU TOWN						
Total	125,511	60,204	18,130	78,334	47,176	62.4
Male	56,832	34,597	4,722	39,319	17,514	69.2
Female	68,678	25,607	13,409	39,016	29,663	56.8
ADAMA TOWN						
Total	263,617	122,049	37,097	159,146	104,472	60.4
Male	124,000	72,253	11,552	83,804	40,196	67.6
Female	139,617	49,796	25,545	75,341	64,276	54.0
JIMMA TOWN						
Total	137,760	75,204	12,922	88,126	49,634	64.0
Male	64,432	42,363	4,041	46,404	18,028	72.0
Female	73,328	32,841	8,881	41,722	31,606	56.9
SHASHEMENE TOWN						
Total	112,476	55,806	9,175	64,981	47,495	57.8
Male	51,866	31,944	2,479	34,423	17,442	66.4
Female	60,611	23,862	6,696	30,558	30,053	50.4
JIJIGA TOWN						
Total	101,901	40,815	8,218	49,033	52,868	48.1
Male	49,905	25,947	3,568	29,515	20,390	59.1
Female	51,996	14,868	4,650	19,518	32,478	37.5
ASOSA TOWN						
Total	35,746	21,112	3,087	24,199	11,547	67.7
Male	17,999	12,786	1,012	13,798	4,201	76.7
Female	17,747	8,326	2,075	10,401	7,346	58.6

Table 4.4 Economically Not Active population of Urban Areas Aged Ten Years and Above by Region, Sex and Reasons for Not Being Active during the Last Seven Days (Current Status Approach): 2015

	Reason for Not Being Active										
Age Group and Sex	Economically	Home			Injury/		Too	Old			
and Sex	Not Active	Maker	Pregnancy	Student	Disabled	Illness	Young	Age	Pensioned	Remittance	Others
COUNTRY TOTAL											
Total	4,864,055	586,999	192,850	3,070,667	26,030	295,567	49,655	416,641	110,753	47,705	67,187
Male	1,798,373	16,374	-	1,438,159	14,266	105,832	24,123	105,198	62,605	14,012	17,805
Female	3,065,682	570,625	192,850	1,632,508	11,764	189,736	25,532	311,444	48,148	33,693	49,383
TIGRAY REGION											
Total	365,237	50,142	10,794	217,414	3,295	23,474	3,449	47,315	3,831	1,976	3,546
Male	131,111	362	-	103,295	1,614	8,184	2,008	11,436	2,334	706	1,172
Female	234,126	49,780	10,794	114,118	1,681	15,290	1,441	35,879	1,497	1,270	2,374
AFFAR REGION											
Total	65,567	8,214	4,176	43,153	329	3,802	1,403	2,703	450	525	813
Male	26,426	355	-	22,221	-	1,314	692	1,248	450	34	111
Female	39,142	7,859	4,176	20,933	329	2,488	711	1,455	-	491	702
AMHARA REGION											
Total	934,341	87,807	43,527	601,828	3,309	69,852	5,123	98,958	11,553	2,806	9,578
Male	345,710	2,248	-	284,567	1,733	21,100	2,909	21,604	7,690	207	3,651
Female	588,631	85,558	43,527	317,261	1,576	48,751	2,214	77,354	3,863	2,600	5,927
OROMIA REGION											
Total	1,408,777	182,783	61,082	916,441	9,215	84,399	16,377	92,692	23,391	9,529	
Male	521,204	5,635	-	429,192	4,190	31,695	7,855	25,167	11,095	2,878	3,498
Female	887,573	177,148	61,082	487,248	5,025	52,704	8,522	67,525	12,296	6,651	9,370
SOMALI REGION											
Total	133,809	25,030	3,641	77,274	1,088	5,386	4,634	7,743	373	8,174	467
Male	56,580	457	-	44,737	675	3,552	2,001	2,524	161	2,302	171
Female	77,229	24,573	3,641	32,537	413	1,834	2,633	5,219	212	5,872	296
BENISHANGUL-GUN											
Total	40,926	2,217	2,289	29,065	192	3,425	516	2,442	-	315	466
Male	15,524	185	-	13,979	-	730	117	329	-	-	183
Female	25,403	2,032	2,289	15,085	192	2,695	399	2,113	-	315	283
S.N.N.P REGION											
Total	647,859	62,410	18,722	466,945	3,251	27,695	8,322	38,768	9,789	8,583	3,372
Male	247,758	2,388	-	206,157	2,428	12,152	4,517	11,179	5,185	2,298	1,454
Female	400,100	60,022	18,722	260,788	823	15,543	3,805	27,589	4,604	6,285	1,918

73

Table 4.4 Cont'd

Age Group					Reason for	Not Being Act	tive				
and Sex	Economically	Home			Injury/		Too	Old			
and Sex	Not Active	Maker	Pregnancy	Student	Disabled	Illness	Young	Age	Pensioned	Remittance	Others
GAMBELLA REGION											
Total	28,035	2,373	439	20,942	246	1,012	615	1,708	183	290	227
Male	12,523	116	-	11,129	193	288	217	332	22	62	164
Female	15,513	2,258	439	9,814	53	723	398	1,376	162	228	63
HARARI REGION											
Total	39,177	3,409	1,140	25,251	398	2,088	1,396	3,875	790	692	138
Male	17,203	-	-	13,601	398	1,091	756	916	294	147	-
Female	21,974	3,409	1,140	11,649	-	998	641	2,959	496	544	138
ADDIS ABABA CITY	ADMINISTRATIO	N									
Total	1,116,099	154,155	44,492	620,704	4,367	70,098	3,827	112,813	59,137	11,904	34,602
Male	387,430	4,628	-	280,111	2,948	23,952	1,010	28,630	34,702	4,564	6,883
Female	728,670	149,527	44,492	340,593	1,419	46,146	2,817	84,182	24,435	7,340	27,719
DIRE DAWA ADMINI	STRATION										
Total	84,227	8,458	2,548	51,650	340	4,336	3,993	7,626	1,255	2,911	1,111
Male	36,905	-	-	29,169	86	1,773	2,041	1,832	673	813	517
Female	47,322	8,458	2,548	22,481	253	2,563	1,952	5,794	582	2,097	594

Table 4.2 Cont'd

Major Town	All	Е	conomically Acti	ve	Economically	Activity
and Sex	Persons	Employed	Unemployed	Total Active	Not Active	Rate
HAWASSA TOWN						
Total	236,248	124,702	18,436	143,138	93,109	60.6
Male	113,068	73,264	3,887	77,151	35,917	68.2
Female	123,180	51,439	14,549	65,987	57,193	53.6
GAMBELLA TOWN						
Total	49,916	27,965	4,052	32,018	17,899	64.1
Male	24,022	15,295	1,207	16,502	7,520	68.7
Female	25,894	12,670	2,846	15,515	10,379	59.9
HARAR TOWN						
Total	106,496	59,109	8,209	67,319	39,177	63.2
Male	49,819	29,900	2,716	32,616	17,203	65.5
Female	56,677	29,210	5,493	34,703	21,974	61.2
ADDIS ABABA CITY						
Total	2,844,395	1,362,069	366,227	1,728,296	1,116,099	60.8
Male	1,289,919	772,367	130,123	902,490	387,430	70.0
Female	1,554,476	589,703	236,103	825,806	728,670	53.1
DIRE DAWA TOWN						
Total	226,294	116,325	25,742	142,066	84,227	62.8
Male	107,490	61,997	8,587	70,585	36,905	65.7
Female	118,804	54,327	17,154	71,482	47,322	60.2

Table 4.3 Economically Not Active Population of Urban Areas Aged Ten Years and Above by Age Group, Sex and Reasons for Not Being Active During the Last Seven Days (Current Status Approach) - COUNTRY TOTAL: 2015

Ago Croup				•	asons for Not			IOIAL. Z			
Age Group and Sex	Economically	Home			Injury/	-	Too	Old			
and Sex	Not Active	Maker	Pregnancy	Student	Disabled	Illness	Young	Age	Pensioned	Remittance	Others
All Ages								-			
Total	4,864,055	586,999	192,850	3,070,667	26,030	295,567	49,655	416,641	110,753	47,705	67,187
Male	1,798,373	16,374	-	1,438,159	14,266	105,832	24,123	105,198	62,605	14,012	17,805
Female	3,065,682	570,625	192,850	1,632,508	11,764	189,736	25,532	311,444	48,148	33,693	49,383
10-14											
Total	1,571,070	25,597	-	1,486,178	2,156	4,964	49,655	-	-	576	1,944
Male	748,271	2,640	-	715,346	2,156	2,944	24,123	-	-	576	485
Female	822,799	22,957	-	770,831	-	2,020	25,532	-	-	-	1,459
15-19											
Total	1,272,258	50,837	13,080	1,188,342	3,322	11,015	-	-	-	549	5,114
Male	556,879	4,155	-	543,980	1,782	4,581	-	-	-	214	2,168
Female	715,379	46,682	13,080	644,362	1,540	6,434	-	-	-	335	2,946
20-24											
Total	485,106	79,740	64,094	307,154	2,433	17,556	-	-	-	1,020	13,110
Male	152,485	1,687	-	138,125	889	8,581	-	-	-	403	2,800
Female	332,621	78,053	64,094	169,028	1,544	8,975	-	-	-	617	10,310
25-29											
Total	262,041	99,633	67,171	57,680	1,261	20,605	-	-	-	4,629	11,063
Male	46,180	2,033	-	28,873	1,089	9,701	-	-	-	1,918	2,566
Female	215,861	97,600	67,171	28,806	172	10,903	-	-	-	2,712	8,497
30-34											
Total	138,833	60,777	30,271	16,332	1,986	18,593	-	-	-	4,460	6,412
Male	20,513	2,083	-	6,669	1,775	7,960	-	-	-	331	1,695
Female	118,320	58,694	30,271	9,664	211	10,633	-	-	-	4,130	4,716
35-39											
Total	122,903	72,498	13,210	3,145	1,935	19,697	-	-	-	4,382	8,037
Male	14,362	894	-	1,193	390	7,350	-	-	-	1,505	3,030
Female	108,541	71,604	13,210	1,952	1,544	12,347	-	-	-		5,007
40-44											
Total	79,545	43,869	2,104	1,635	2,243	19,347	-	-	-	4,990	5,357
Male	13,238	274	· -	970	1,451	7,274	-	-	-	1,446	1,824
Female	66,307	43,596	2,104	666	792	12,073	-	-	-	3,544	3,533

 \geq

Table 4.2 Cont'd

Ago Croup				Re	asons for Not	Being Acti	ve				
Age Group and Sex	Economically Not Active	Home Maker	Pregnancy	Student	Injury/ Disabled	Illness	Too Young	Old Age	Pensioned	Remittance	Others
45-49			<u> </u>	1		'					
Total	85,217	43,430	1,024	1,066	2,527	26,436	-	-	3,268	2,249	5,216
Male	11,720	1,111	-	101	837	7,489	-	-	-	712	1,469
Female 50-54	73,496	42,319	1,024	965	1,690	18,947	-	-	3,268	1,537	3,747
Total	103,060	40,540	529	900	2,039	28,868	-	15,005	5,707	3,908	5,565
Male	14,419	1,212	-	-	1,000	7,776	-	1,419	1,845	236	932
Female 55-59	88,641	39,328	529	900	1,039	21,092	-	13,586	3,862	3,673	4,633
Total	103,764	28,216	882	618	1,308	30,775	-	25,747	10,252	3,619	2,347
Male	15,122	-	-	-	395	7,510	-	1,950	4,515	531	222
Female 60-64	88,642	28,216	882	618	913	23,265	-	23,798	5,737	3,088	2,125
Total	149,768	21,177	90	1,049	3,586	27,319	-	60,908	27,747	5,386	2,504
Male	29,913	253	-	133	1,866	7,562	-	6,211	12,786	727	375
Female	119,855	20,924	90	916	1,720	19,756	-	54,698	14,961	4,660	2,129
65+											
Total	490,491	20,686	395	6,569	1,235	70,393	-	314,980	63,779	11,935	518
Male	175,271	34	-	2,768	636	27,104	-	95,618	43,459	5,413	239
Female	315,220	20,652	395	3,801	600	43,289	-	219,362	20,319	6,522	280

CHAPTER IV

ECONOMIC ACTIVITY STATUS

4.1 Introduction

Statistical data on the economic activity status of a population is essential for planning and formulating policies for better utilization of human resources of a country. The economic activity status indicates the size and distribution of the work force of economically active and not active population during a given reference period at a specified area.

In this regard, a wide range of information was collected on economic activity status of all persons aged five years and above in the 2015 UEUS. For the purpose of gathering information about the economic activities of children, the lower age limit was fixed at five years and hence data were collected from persons aged five years and above. However, for the purpose of measuring the economic activity status of adult population according to the Ethiopian situation, the lower age limit was fixed into ten years.

The 2015 Urban Employment and Unemployment Survey administered detailed labour force questions to measure the economic activity status of persons in urban areas of the country. These includes:-

- whether persons were engaged in economic / productive activities in the last seven days prior to the survey date or reasons for not being engaged;
- whether persons were looking for work or will be available for work in the coming one month or reason for not being available during the reference period; and
- whether persons were engaged in economic / productive activities during most of the last six months or reasons for not being engaged.

Every person aged ten years and above in the sampled household was asked if he/she was engaged in any productive activities during the last seven days, and in the last six months prior to the survey date. Based on the response obtained from respondents, the economic activity status of the population was determined by using the standard definitions of economic and non economic activities.

The 2015 Urban Employment and Unemployment Survey followed the ILO international standard definitions of economic and non-economic activities. The concept of economic activity as adopted by the 13th International Conference on Labour Statistics (ICLS,1982) is defined as the production of goods and/or services that falls within the United Nations System of National Accounts (SNA) production boundary (ILO, 1990). In this survey, economic/ productive activity is defined as work, which involves the production of goods and /or services for sale or exchange. In addition, production of goods and services for own consumption or own uses are also considered as economic activities. These include production of primary products (agriculture, hunting, fishing, forestry and logging, mining and quarrying), for own consumption; processing of primary products by the producers themselves; production of other commodities where part of it is sold on the market; and own account construction and fixed asset formation (expected life use of one year or more). Such economic activities could be performed for an individual, family or private enterprise, government establishment or social organization. The remuneration may be on daily, weekly, monthly, yearly or contract basis. The practical activities of apprentices are also considered as economic activities. On the other hand, unpaid household chores such as preparing food, cleaning the house, taking care of children are not considered as economic activities. Similarly, unpaid community and volunteer services are classified as non-economic activities.

Based on the above ILO definitions, data on economic activity status of persons aged ten years and above was presented in this report. Persons who were engaged in productive work /employed/ or available to be engaged /unemployed/ during the reference period are considered as economically active persons. While those persons who were neither

engaged nor available to be engaged in the production of economic goods and services during the reference period are considered as population not economically active.

4.2 Data Collection Approaches

The 2015 Urban Employment and Unemployment Survey used two approaches in the collection of data about the economic activity status of the population. These are current and usual activity status approaches. The main difference between these two approaches is the length of the reference period.

4.2.1 Current Activity Status Approach

The current activity status approach measures the economic activity status of a person based on a short reference period of seven days before the date of interview. In this approach, all persons aged ten years and above who were engaged in productive work or available to be engaged during the reference week were considered as currently active while those who were neither engaged nor available to be engaged in productive work were taken as currently not active.

4.2.2 Usual Activity Status Approach

The usual activity status approach measures the economic activity status of a person based on a long reference period of the six months prior to the date of interview. In the usual activity status approach, all persons aged ten years and above who were engaged in productive works or available to be engaged in the production of economic goods and/or services during most of the six months were classified as usually active. Persons who were neither employed nor unemployed during most of the six months categorized as usually not active.

The choice of appropriate measurement approach depends on the extent of variability of economic activities of the population during a specified period in a given the study area. In the rural areas where seasonal variation of activities is common, to collect the

economic activity status of the rural population including the slack and peak periods of agricultural activities, a longer reference period (usual activity status approach) is found to be more appropriate. On the other hand, in urban areas, activities are relatively less affected by seasonal variations and hence shorter reference period or current activity status approach is found to be more convenient. But in this survey, both approaches were used to collect data on economic activity status of urban population.

This chapter presents the economic activity status aged ten years and above, size and distribution of economically active and not economically active urban population, activity rate, and reasons for not being economically active of urban population by using only the current status approach at national, regional and major town levels.

4.3 Current Activity Status

As mentioned above, current activity status approach measures the economic activity status of a person in a short reference period, that is, the seven days prior to the date of the interview. In this approach a series of inquiries were administered to determine the economic activity status of the population during the reference week such as engagement in economic activity, attachment to the job, seeking and availability to work, reason for not seeking or available to work, etc. These questions were presented to those persons aged five years and above. For the purpose of reporting, persons aged ten years and above were used to classify the population into three mutually exclusive categories: employed, unemployed, and not in the labour force (Not Economically active). The employed and the unemployed population together constitute the labour force or the currently active population. The third category, those who were not in the labour force are called not currently active population.

As part of the economically active population, employed population consists of persons aged ten years and above who engaged in a productive activity or work at least for one hour during the seven days prior to the date of the interview. In addition, persons who had regular jobs, or business, or holdings to return to but who were temporarily absent from work (not at work or worked less than one hour) for various reasons such as illness or injury, holiday or vacation, strike or lockout, and seasonality of work, annual

leave, temporary closure of establishment were also considered as employed. For a person to be considered as absent from work, he/she must have formal attachment to the job.

The currently unemployed population also comprises those persons without work but looking for work or available and ready to work if any job is found during the reference period of the coming one month. The reference period of the coming one month refers to the survey week plus the next three weeks. If a person who is looking for work but engaged in productive activity during the reference week is recorded as employed but not as unemployed.

4.4 Economically Active and Not active Population and Activity Rate

Based on the above definition of economic activity status, the number of economically active population including employed and unemployed persons, population not active and activity rates of urban population aged ten years and above at national and regional level is presented by sex and age group in Table 4.1 and Table 4.1.1- 4.1.11 using the current status approach, respectively. Besides, the economic activity status and activity rate of selected major towns' population aged ten years and above is also presented in Table 4.2.

The extent of the economic participation of the population is measured using economic activity rates. Economic activity rate or economic participation rate is computed as the percentage of the economically active population over the total of active plus not active population aged ten years and above. The national estimate shows that 63.7 percent of the population aged ten years and above were economically active (Table 4.1). With respect to sex, the activity rate of male is 71.1 percent and the corresponding female activity rate is 57.2 percent. The age distributions of activity rate illustrate that it is low at younger and older age groups. The table also shows a relatively stable for middle ages between 20-59 years, and reaches its peak in age group 35-39 (Table 4.1).

As far as the economic activity status of major towns' population is concerned, 61.8 percent of the populations were economically active while the remaining 38.2 percent were found to be the population not economically active. In terms of sex, the activity rate of male (70.0 per cent) was higher by 15.3 per cent than female (54.9 per cent). Among the major towns, the highest activity rate was recorded in Bahir Dar town (71.9 percent) while the lowest was for Jijiga town (48.1 percent) (Table 4.2).

The size of economically active and not active population has its own implication on the development of the local as well as the national economy of a region. The high activity rate of a population implies the prevalence of huge potential of the human resource or human labour that foster sustainable economic development. On the other hand, the low economic activity rate also implies the lack of adequate and trained human labour or the existence of large number of economically non-productive population which has no contribution for development at the time. Hence, the economic activity rate of the urban population of the Ethiopia can be viewed from these perspectives.

The distribution of economically not active population of urban areas at national and regional level by age group, sex and reason for not being economically active (for those who were neither engaged nor available and/ or seeking work) during the last seven days are presented in Table 4.3 and Table 4.4, respectively. The reasons for not looking for or available to work include; home maker, pregnancy, student, injury/ disability, illness, too young, old age, pensioned, remittance, and other reasons. The result in Table 4.3 shows that the lion share of the population not active were students for the national (63.1 percent). The remaining per cent was the other groups of reasons for the economically not active population. The same holds true for the regions, too (Table 4.4).

Table 4.1 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)- COUNTRY TOTAL: 2015

Kale	during the l	Lasi Sevell D	Activity Status							
Age Group	All	L-			F	Activity				
and Sex	Persons		onomically Acti		Economically	Rate				
		Employed	Unemployed	Total Active	Not Active					
All Ages										
Total	13,381,376	7,083,094	1,434,227	8,517,320	4,864,055	63.7				
Male	6,223,736	3,963,143	462,220	4,425,363	1,798,373	71.1				
Female	7,157,639	3,119,950	972,007	4,091,957	3,065,682	57.2				
10-14										
Total	1,790,429	194,969	24,390	219,359	1,571,070	12.3				
Male	858,124	100,707	9,146	109,853	748,271	12.8				
Female	932,305	94,262	15,244	109,506	822,799	11.7				
15-19										
Total	2,041,908	565,103	204,546	769,650	1,272,258	37.7				
Male	853,760	224,956	71,926	296,881	556,879	34.8				
Female	1,188,148	340,148	132,621	472,768	715,379	39.8				
20-24	0.044.404	4 407 040	440,400	4 500 070	405 400	75.0				
Total	2,011,484	1,107,918	418,460	1,526,378	485,106	75.9				
Male	836,505	546,853	137,166	684,020	152,485	81.8				
Female	1,174,979	561,064	281,294	842,358	332,621	71.7				
25-29	0.004.704	4 455 040	007.500	4 700 750	000 044	07.4				
Total	2,024,794	1,455,213	307,539	1,762,753	262,041	87.1				
Male	978,477	827,711	104,586	932,297	46,180	95.3				
Female	1,046,317	627,502	202,953	830,456	215,861	79.4				
30-34	4 070 044	005.745	450.000	4 407 704	400.000	20.4				
Total	1,276,614	985,745	152,036	1,137,781	138,833	89.1				
Male	663,030	599,578	42,939	642,517	20,513	96.9				
Female	613,583	386,166	109,097	495,264	118,320	80.7				
35-39	1 174 110	025 220	116 225	1 051 546	122.002	90 F				
Total	1,174,449	935,320	116,225	1,051,546	122,903	89.5				
Male	559,061	518,080	26,618	544,699	14,362	97.4				
Female 40-44	615,388	417,240	89,607	506,847	108,541	82.4				
	710 707	E70 200	EE 000	624 402	70 F 4F	99.0				
Total Male	713,737 397,066	578,390 365,839	55,802 17,988	634,192 383,827	79,545 13,238	88.9 96.7				
Female	316,672	212,551	37,815	250,365	66,307	79.1				
45-49	310,072	212,001	37,013	250,505	00,307	73.1				
Total	561,559	421,698	54,645	476,342	85,217	84.8				
Male	278,068	252,303	14,044	266,347	11,720	95.8				
Female	283,491	169,394	40,601	209,995	73,496	74.1				
50-54	200,401	100,004	40,001	200,000	70,400	7 - 7. 1				
Total	422,014	288,013	30,941	318,954	103,060	75.6				
Male	195,936	173,544	7,974	181,517	14,419	92.6				
Female	226,078	114,470	22,967	137,437	88,641	60.8				
55-59	220,070	, 0	22,001	101,101	00,011	00.0				
Total	345,943	210,275	31,904	242,179	103,764	70.0				
Male	154,230	128,640	10,468	139,108	15,122	90.2				
Female	191,713	81,635	21,436	103,071	88,642	53.8				
60-64	•	•	•	•	•					
Total	310,206	145,598	14,841	160,439	149,768	51.7				
Male	126,910	90,332	6,664	96,996	29,913	76.4				
Female	183,297	55,266	8,177	63,442	119,855	34.6				
65+	•	,	•	•	•					
Total	708,239	194,851	22,897	217,748	490,491	30.7				
Male	322,571	134,599	12,702	147,300	175,271	45.7				
Female	385,668	60,253	10,195	70,448	315,220	18.3				

Table 4.1.1 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-TIGRAY REGION: 2015

Age Group	All	Activity Status	Activity	
and Sex	Persons	Economically Active	Economically	Rate

1] [Total	Not Active	
		Employed	Unemployed	Active		
All Ages						
Total	952,370	477,813	109,321	587,134	365,237	61.6
Male	411,583	253,254	27,218	280,472	131,111	68.1
Female	540,787	224,559	82,103	306,662	234,126	56.7
10-14						
Total	116,810	6,281	1,253	7,535	109,275	6.5
Male	56,402	2,213	591	2,804	53,598	5.0
Female	60,408	4,068	662	4,730	55,677	7.8
15-19						
Total	154,538	42,231	18,958	61,189	93,350	39.6
Male	67,758	16,565	8,108	24,673	43,085	36.4
Female	86,780	25,665	10,850	36,515	50,265	42.1
20-24						
Total	146,474	74,609	37,978	112,587	33,886	76.9
Male	44,307	27,680	7,766	35,446	8,861	80.0
Female	102,167	46,929	30,212	77,142	25,026	75.5
25-29						
Total	143,397	108,332	18,047	126,379	17,018	88.1
Male	63,153	56,044	3,801	59,845	3,308	94.8
Female	80,244	52,288	14,246	66,534	13,710	82.9
30-34						
Total	101,199	74,997	15,205	90,203	10,996	89.1
Male	52,212	47,385	4,045	51,430	781	98.5
Female	48,987	27,612	11,160	38,772	10,215	79.1
35-39						
Total	71,373	54,086	5,930	60,016	11,358	84.1
Male	30,622	28,626	90	28,716	1,906	93.8
Female	40,751	25,460	5,839	31,299	9,452	76.8
40-44 Total	39,684	29,467	4,162	33,629	6,055	84.7
Male	22,539	29,467	1,544	21,705	834	96.3
Female	17,144	9,305	2,619	11,924	5,220	69.6
45-49	17,144	9,305	2,019	11,924	5,220	09.0
Total	35,537	26,063	3,722	29,785	5,752	83.8
Male	16,500	16,193	153	16,346	153	99.1
Female	19,038	9,869	3,569	13,439	5,599	70.6
50-54	.0,000	0,000	0,000	10, 100	0,000	. 0.0
Total	31,595	21,511	1,260	22,771	8,824	72.1
Male	17,425	15,541	135	15,676	1,749	90.0
Female	14,170	5,970	1,125	7,095	7,075	50.1
55-59						
Total	21,438	13,031	959	13,990	7,448	65.3
Male	7,149	5,982	229	6,211	938	86.9
Female	14,289	7,049	730	7,779	6,510	54.4
60-64						
Total	22,037	9,713	363	10,077	11,961	45.7
Male	8,081	6,637	-	6,637	1,443	82.1
Female	13,957	3,076	363	3,439	10,517	24.6
65+						
Total	68,288	17,492	1,482	18,974	49,314	27.8
Male	25,436	10,226	756	10,982	14,454	43.2
Female	42,852	7,267	726	7,993	34,859	18.7

Female 42,852 7,267 726 7,993 34,859 18.7

Table 4.1.2 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-AFFAR REGION: 2015

Age Group	All	Economically Active				Activity
and Sex	Persons			Total	Economically	Rate
		Employed	Unemployed	Active	Not Active	
All Ages						
Total	216,983	131,439	19,977	151,415	65,567	69.8
Male	108,772	75,452	6,894	82,346	26,426	75.7

Female 10-14	108,211	55,987	13,083	69,069	39,142	63.8
Total	29,479	4,059	224	4,283	25,196	14.5
Male	15,835	1,826	224	2,050	13,785	12.9
Female	13,643	2,233		2,233	11,411	16.4
15-19	,	_,		_,	,	
Total	28,316	9,262	1,970	11,232	17,084	39.7
Male	11,399	3,692	1,083	4,775	6,624	41.9
Female	16,918	5,570	887	6,457	10,461	38.2
20-24						
Total	30,313	18,123	6,704	24,827	5,486	81.9
Male	16,846	11,287	3,230	14,517	2,329	86.2
Female	13,466	6,835	3,475	10,310	3,156	76.6
25-29						
Total	36,477	27,089	4,748	31,837	4,641	87.3
Male	15,061	13,825	754	14,580	481	96.8
Female	21,416	13,263	3,994	17,257	4,159	80.6
30-34 Total	21,738	18,064	1,799	19,863	1,875	91.4
Male	12,072	11,713	1,799	19,003	350	91. 4 97.1
Female	9,666	6,351	1,790	8,141	1,525	84.2
35-39	9,000	0,331	1,790	0,141	1,323	04.2
Total	22,868	18,805	1,402	20,207	2,661	88.4
Male	12,303	11,486	544	12,029	274	97.8
Female	10,565	7,319	858	8,177	2,387	77.4
40-44	10,000	7,010	000	0,177	2,007	11.4
Total	12,518	10,597	1,078	11,674	844	93.3
Male	6,660	6,291	370	6,660	-	100.0
Female	5,858	4,306	708	5,014	844	85.6
45-49	2,222	1,000		2,2 : :		
Total	10,638	9,295	725	10,020	619	94.2
Male	5,117	4,708	377	5,086	31	99.4
Female	5,522	4,587	347	4,934	588	89.4
50-54	-,	1,001		1,00		•••
Total	8,602	6,314	256	6,571	2,031	76.4
Male	4,382	4,338	-	4,338	44	99.0
Female	4,219	1,976	256	2,233	1,986	52.9
55-59	1,210	1,070	200	2,200	1,000	02.0
Total	6,466	3,769	806	4,575	1,891	70.8
Male	2,348	1,714	39	1,753	595	74.6
Female	4,118	2,055	768	2,822	1,295	68.5
60-64	4,110	2,055	700	2,022	1,295	00.5
	F 000	0.744	440	0.004	4.044	75.5
Total	5,062	3,711	110	3,821	1,241	75.5
Male	3,307	2,721	110	2,831	476	85.6
Female	1,755	990	-	990	765	56.4
65+		0.0		0.500	0.555	
Total	4,506	2,352	154	2,506	2,000	55.6
Male	3,439	1,850	154	2,004	1,435	58.3
Female	1,066	502	-	502	565	47.0

Female 1,066 502 - 502 565 47.0

Table 4.1.3 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-AMHARA REGION: 2015

		Activity Status				
Age Group	All	Eco	onomically Activ		Activity	
and Sex	Persons			Total	Economically	Rate
		Employed	Unemployed	Active	Not Active	
All Ages						
Total	2,722,586	1,449,091	339,154	1,788,246	934,341	65.7
Male	1,240,992	795,183	100,100	895,283	345,710	72.1
Female	1,481,594	653,908	239,055	892,963	588,631	60.3
10-14						
Total	352,195	44,588	1,674	46,262	305,934	13.1
Male	176,748	22,776	431	23,207	153,541	13.1
Female	175,448	21,812	1,243	23,055	152,393	13.1

iotai	- 10,371	100,773	- 0,032	132,307	200,707	37.1
Male	164,308	42,682	14,065	56,747	107,561	34.5
Female	246,663	63,793	32,027	95,820	150,843	38.8
20-24	,	,	,	,	•	
Total	425,178	232,951	109,704	342,654	82,524	80.6
Male	172,718	109,983	31,790	141,773	30,944	82.1
Female	252,460	122,967	77,914	200,881	51,579	79.6
25-29	232,400	122,307	77,514	200,001	31,379	73.0
Total	412,707	299,011	71,007	370,018	42,689	89.7
Male	203,310	176,619	22,335	198,954	4,356	97.9
Female	209,397	122,393	48,672	171,064	38,333	81.7
30-34						
Total	252,372	190,781	35,281	226,062	26,310	89.6
Male	127,941	116,417	9,702	126,119	1,821	98.6
Female	124,431	74,364	25,579	99,943	24,489	80.3
35-39						
Total	225,093	182,038	25,317	207,355	17,738	92.1
Male	106,123	97,487	6,353	103,841	2,283	97.8
Female	118,969	84,550	18,963	103,514	15,456	87.0
40-44	-,	, , , , , ,	-,	, -	-,	
Total	129,669	108,134	12,778	120,913	8,756	93.2
Male	74,582	69,255	3,272	72,528	2,055	97.2
Female	55,087	38,879	9,506	48,385	6,702	87.8
45-49	,	,-	-,	-,	-, -	
Total	109,362	87,212	10,106	97,318	12,044	89.0
Male	50,160	46,297	2,193	48,489	1,670	96.7
Female	59,202	40,915	7,913	48,828	10,374	82.5
50-54	,	,	,	,	•	
Total	92,623	62,033	12,287	74,320	18,303	80.2
Male	34,403	30,169	2,671	32,840	1,563	95.5
Female	58,220	31,864	9,616	41,480	16,739	71.2
55-59	,	, , , ,	-,-	,	-,	
Total	79,419	50,755	6,735	57,490	21,929	72.4
Male	35,936	31,112	1,628	32,740	3,196	91.1
Female	43,483	19,643	5,108	24,750	18,733	56.9
60-64	10,100	,	2,:00	_ :,: = :	,	
Total	60,732	34,241	3,993	38,235	22,498	63.0
Male	25,448	20,928	1,675	22,604	2,844	88.8
Female	35,285	13,313	2,318	15,631	19,653	44.3
65+	00,200	10,010	2,0.0	10,001	10,000	
Total	172,264	50,872	4,180	55,052	117,213	32.0
Male	69,316	31,457	3,984	35,441	33,875	51.1
Female	102,948	19,415	196	19.611	83,338	19.0
гентаве	102,940	19,413	190	19,011	03,330	19.0

15-19 Total

410,971

106,475

46,092

152,567

258,404

37.1

Table 4.1.4 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-OROMIA REGION: 2015

			Activity Status				
Age Group	All	Eco	onomically Activ		Activity		
and Sex	Persons			Total	Economically	Rate	
		Employed	Unemployed	Active	Not Active		
All Ages							
Total	3,898,166	2,102,417	386,972	2,489,389	1,408,777	63.9	
Male	1,863,092	1,213,723	128,166	1,341,888	521,204	72.0	
Female	2,035,074	888,694	258,806	1,147,501	887,573	56.4	
10-14							
Total	563,599	67,735	7,817	75,552	488,047	13.4	
Male	272,782	41,900	2,873	44,773	228,009	16.4	
Female	290,817	25,835	4,944	30,778	260,038	10.6	
15-19							
Total	632,697	185,609	64,916	250,525	382,172	39.6	
Male	275,662	82,315	23,927	106,242	169,420	38.5	
Female	357,035	103,293	40,990	144,283	212,752	40.4	
20-24							

TOtal	002,204	340,212	114,004	400,293	141,900	70.4
Male	255,222	183,385	36,756	220,141	35,081	86.3
Female	347,041	162,826	77,328	240,154	106,887	69.2
25-29						
Total	608,648	434,406	87,835	522,240	86,408	85.8
Male	302,293	257,771	29,632	287,404	14,889	95.1
Female	306,355	176,634	58,202	234,837	71,518	76.7
30-34						
Total	349,526	280,755	34,057	314,812	34,714	90.1
Male	191,280	174,014	9,880	183,894	7,386	96.1
Female	158,246	106,741	24,177	130,918	27,328	82.7
35-39	,	,	,	,-	,	
Total	334,301	272,233	30,808	303,041	31,260	90.6
Male	159,482	148,999	8,129	157,128	2,354	98.5
Female	174,819	123,234	22,679	145,913	28,906	83.5
40-44						
Total	210,778	171,711	12,518	184,229	26,549	87.4
Male	121,589	111,982	5,956	117,938	3,651	97.0
Female	89,189	59,729	6,562	66,291	22,898	74.3
45-49						
Total	163,564	120,390	14,673	135,063	28,501	82.6
Male	79,937	72,919	3,167	76,086	3,852	95.2
Female	83,627	47,471	11,507	58,977	24,650	70.5
50-54						
Total	95,344	72,481	4,851	77,332	18,012	81.1
Male	50,811	45,658	1,690	47,348	3,463	93.2
Female	44,533	26,823	3,160	29,984	14,549	67.3
55-59						
Total	87,204	57,461	5,137	62,598	24,605	71.8
Male	39,523	34,810	1,700	36,510	3,013	92.4
Female	47,681	22,651	3,437	26,088	21,593	54.7
60-64	17,001	22,00	0, 101	20,000	21,000	0
Total	87,711	44,309	4,162	48,470	39,241	55.3
Male	37,757	27,178	1,749	28,927	8,830	76.6
Female	49,954	17,131	2,413	19,544	30,410	39.1
65+	10,001	17,101	2,110	10,011	00,110	00.1
Total	162,531	49,116	6,116	55,232	107,299	34.0
Male	76,752	32,790	2,707	35,497	41,255	46.2
Female	85,778	16,325	3,409	19,734	66,044	23.0
remale	00,770	10,323	3,409	19,734	00,044	23.0

Total

602,264

346,212

114,084

460,295

141,968

76.4

Table 4.1.5 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-SOMALI REGION: 2015

		Activity Status				
Age Group	All	Eco	onomically Activ	ve		Activity
and Sex	Persons			Total	Economically	Rate
		Employed	Unemployed	Active	Not Active	
All Ages						
Total	273,737	119,082	20,846	139,928	133,809	51.1
Male	131,760	67,114	8,066	75,181	56,580	57.1
Female	141,976	51,967	12,780	64,747	77,229	45.6
10-14						
Total	52,965	6,236	1,043	7,278	45,687	13.7
Male	27,439	2,695	227	2,923	24,517	10.7
Female	25,526	3,540	816	4,356	21,170	17.1
15-19						
Total	47,750	10,505	1,838	12,343	35,407	25.8
Male	22,099	3,950	650	4,600	17,499	20.8
Female	25,651	6,555	1,188	7,743	17,908	30.2
20-24						
Total	35,088	13,337	5,952	19,289	15,799	55.0
Male	15,543	6,860	2,415	9,274	6,268	59.7
Female	19,546	6,477	3,538	10,015	9,531	51.2
25-29						
Total	34,163	21,263	3,479	24,742	9,421	72.4

Male	16,071	13,257	836	14,094	1,977	87.7
Female	18,093	8,006	2,643	10,649	7,444	58.9
30-34	,	,	•	,	,	
Total	22,497	15,175	2,731	17,905	4,592	79.6
Male	10,907	9,835	750	10,585	322	97.0
Female	11,590	5,339	1,981	7,320	4,270	63.2
35-39	,	2,222	1,221	1,0=0	-,	
Total	21,200	15,790	1,830	17,619	3,581	83.1
Male	9,202	8,408	664	9,072	130	98.6
Female	11,998	7,382	1,165	8,547	3,451	71.2
40-44	,	,	•	,	,	
Total	18,841	14,072	1,451	15,523	3,318	82.4
Male	8,729	7,918	492	8,410	319	96.3
Female	10,112	6,154	959	7,113	2,998	70.3
45-49						
Total	12,348	9,869	793	10,662	1,686	86.3
Male	6,989	5,840	742	6,582	407	94.2
Female	5,359	4,029	51	4,080	1,279	76.1
50-54						
Total	8,798	5,320	968	6,288	2,510	71.5
Male	4,762	3,568	631	4,200	562	88.2
Female	4,036	1,752	336	2,088	1,947	51.7
55-59						
Total	3,218	1,331	226	1,556	1,661	48.4
Male	1,885	1,024	174	1,198	687	63.6
Female	1,333	307	52	359	975	26.9
60-64						
Total	6,185	3,091	210	3,301	2,885	53.4
Male	3,231	1,855	210	2,065	1,166	63.9
Female	2,954	1,236	-	1,236	1,719	41.8
65+						
Total	10,685	3,094	326	3,420	7,264	32.0
Male	4,905	1,903	276	2,179	2,726	44.4
Female	5,780	1,191	51	1,242	4,538	21.5
_						•

Table 4.1.6 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-**BENISHANGUL-GUMUZ REGION: 2015**

Age Group	All	Economically Active				Activity
and Sex	Persons			Total	Economically	Rate
		Employed	Unemployed	Active	Not Active	
All Ages						
Total	157,710	107,053	9,731	116,784	40,926	74.0
Male	77,083	58,730	2,830	61,559	15,524	79.9
Female	80,627	48,323	6,901	55,224	25,403	68.5
10-14						
Total	22,907	7,388	43	7,430	15,476	32.4
Male	9,561	2,760	43	2,802	6,758	29.3
Female	13,346	4,628	-	4,628	8,718	34.7
15-19						
Total	23,054	9,970	1,012	10,982	12,072	47.6
Male	11,884	4,889	348	5,236	6,647	44.1
Female	11,170	5,081	664	5,745	5,425	51.4
20-24						
Total	26,866	18,860	4,000	22,861	4,005	85.1
Male	10,504	8,763	1,208	9,971	533	94.9
Female	16,361	10,098	2,792	12,890	3,472	78.8
25-29						
Total	27,224	23,990	1,450	25,440	1,783	93.4
Male	15,121	14,366	394	14,760	361	97.6
Female	12,103	9,624	1,056	10,680	1,423	88.2
30-34						
Total	14,432	13,065	876	13,941	491	96.6

Male	7,513	7,378	135	7,513	-	100.0
Female	6,919	5,687	741	6,428	491	92.9
35-39						
Total	13,439	11,685	532	12,218	1,222	90.9
Male	7,116	6,825	-	6,825	291	95.9
Female	6,323	4,861	532	5,393	930	85.3
40-44						
Total	7,539	6,630	605	7,235	303	96.0
Male	3,468	3,392	77	3,468	-	100.0
Female	4,070	3,238	529	3,767	303	92.5
45-49						
Total	7,049	6,087	253	6,340	709	89.9
Male	4,169	4,050	119	4,169	-	100.0
Female	2,880	2,037	134	2,171	709	75.4
50-54						
Total	4,966	4,104	215	4,319	647	87.0
Male	2,842	2,671	172	2,842	-	100.0
Female	2,124	1,434	43	1,477	647	69.5
55-59						
Total	3,918	2,293	349	2,643	1,275	67.4
Male	1,936	1,632	172	1,804	132	93.2
Female	1,982	661	177	839	1,143	42.3
60-64						
Total	1,778	977	163	1,140	637	64.1
Male	759	597	163	759	-	100.0
Female	1,018	381	-	381	637	37.4
65+						
Total	4,539	2,001	233	2,234	2,305	49.2
Male	2,209	1,408	-	1,408	801	63.7
Female	2,330	593	233	826	1,504	35.4

Table 4.1.7 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-S.N.N.P REGION: 2015

	Activity Status							
Age Group	All	Eco	onomically Activ	ve		Activity		
and Sex	Persons			Total	Economically	Rate		
		Employed	Unemployed	Active	Not Active			
All Ages								
Total	1,887,693	1,096,887	142,947	1,239,834	647,859	65.7		
Male	897,420	603,653	46,009	649,662	247,758	72.4		
Female	990,273	493,233	96,939	590,172	400,100	59.6		
10-14								
Total	301,593	39,806	5,106	44,912	256,681	14.9		
Male	134,093	18,628	1,174	19,802	114,291	14.8		
Female	167,500	21,178	3,932	25,110	142,390	15.0		
15-19								
Total	320,875	108,119	27,595	135,714	185,161	42.3		
Male	132,561	45,420	10,016	55,436	77,125	41.8		
Female	188,314	62,699	17,579	80,278	108,036	42.6		
20-24								
Total	277,103	180,025	36,141	216,166	60,936	78.0		
Male	125,632	92,276	12,436	104,712	20,920	83.3		
Female	151,471	87,749	23,705	111,454	40,017	73.6		
25-29								
Total	279,538	223,124	32,126	255,250	24,288	91.3		
Male	139,390	123,468	11,276	134,744	4,646	96.7		
Female	140,148	99,656	20,850	120,506	19,642	86.0		
30-34								
Total	149,010	127,137	12,676	139,813	9,197	93.8		
Male	79,612	75,130	3,612	78,742	871	98.9		
Female	69,398	52,007	9,064	61,071	8,327	88.0		
35-39								
Total	163,312	141,276	8,956	150,232	13,080	92.0		
Male	76,667	73,033	1,310	74,343	2,324	97.0		
Female	86,645	68,243	7,646	75,889	10,756	87.6		

40-44						
Total	109,822	95,357	4,809	100,167	9,655	91.2
Male	62,386	59,331	1,229	60,560	1,826	97.1
Female	47,436	36,026	3,581	39,607	7,829	83.5
45-49						
Total	75,533	64,394	4,554	68,948	6,585	91.3
Male	42,846	40,865	661	41,526	1,320	96.9
Female	32,687	23,529	3,894	27,423	5,264	83.9
50-54						
Total	56,464	42,277	4,162	46,438	10,025	82.2
Male	29,101	26,341	1,258	27,599	1,503	94.8
Female	27,362	15,936	2,904	18,840	8,523	68.9
55-59						
Total	39,865	28,495	2,497	30,991	8,873	77.7
Male	22,323	19,472	1,460	20,932	1,391	93.8
Female	17,542	9,023	1,037	10,059	7,482	57.3
60-64						
Total	34,855	18,241	2,052	20,293	14,562	58.2
Male	10,038	7,137	891	8,027	2,011	80.0
Female	24,817	11,105	1,161	12,266	12,551	49.4
65+	,	·	·	·	,	
Total	79,724	28,636	2,272	30,908	48,815	38.8
Male	42,771	22,554	687	23,240	19,531	54.3
Female	36,953	6,083	1,586	7,668	29,284	20.8

Table 4.1.8 Urban population Aged Ten Years and above by Age Group, Sex, Activity
Status and Rate during the Last Seven Days (Current Status Approach)GAMBELLA REGION: 2015

Ago Croup	All		Activity	Status		A ativity
Age Group and Sex		Ec	onomically Activ	/e	Economically	Activity Rate
and Sex	Persons	Employed	Unemployed	Total Active	Not Active	Rate
All Ages						
Total	94,945	61,809	5,101	66,910	28,035	70.5
Male	45,805	31,771	1,511	33,282	12,523	72.7
Female	49,141	30,038	3,590	33,628	15,513	68.4
10-14						
Total	16,286	4,843	121	4,965	11,322	30.5
Male	7,967	2,040	40	2,081	5,886	26.1
Female	8,319	2,803	81	2,884	5,436	34.7
15-19	45.000		710	7 400	0.505	40.0
Total	15,992	6,775	713	7,488	8,505	46.8
Male Female	7,729	3,251	162	3,413	4,316	44.2
20-24	8,264	3,524	551	4,075	4,189	49.3
Zu-Z4 Total	40.000	10 110	4.004	44.027	2.040	82.7
Male	16,983	12,116 5,364	1,921 536	14,037 5,900	2,946	82.7 81.2
Female	7,264 9,719	6,753	1,384	8,137	1,364 1,582	83.7
25-29	9,719	0,755	1,304	0,137	1,362	03.7
Total	15,908	13,679	1,123	14,802	1,105	93.1
Male -	7,715	7,128	359	7,487	227	97.1
Female	8,193	6,551	764	7,315	878	89.3
30-34						
Total	10,160	8,929	455	9,384	776	92.4
Male	4,981	4,940	-	4,940	41	99.2
Female	5,179	3,989	455	4,444	735	85.8
35-39	•	•		,		
Total	6,677	6,056	134	6,190	487	92.7
Male	3,941	3,839	41	3,880	62	98.4
Female	2,735	2,217	93	2,310	425	84.5
40-44	2,733	2,217	95	2,310	423	04.5
Total	3,861	3,489	174	3,663	198	94.9
Male	2,091	1,976	93	2,069	22	99.0
Female	1,770	1,513	81	1,594	176	90.0
45-49	.,	.,3.0	•	.,		
Total	3,342	2,652	364	3,016	326	90.2
1 0101	0,072	2,302	304	3,310	320	00.2

Male	1,344	1,119	204	1,323	22	98.4	
Female	1,997	1,534	159	1,693	304	84.8	
50-54							
Total	2,213	1,995	-	1,995	218	90.2	
Male	1,201	1,180	-	1,180	22	98.2	
Female	1,012	816	-	816	196	80.6	
55-59	,						
Total	1,206	628	75	703	503	58.3	
Male	744	582	75	657	88	88.2	
Female	462	46	-	46	416	10.0	
60-64							
Total	870	341	-	341	529	39.2	
Male	375	177	-	177	198	47.1	
Female	495	165	-	165	330	33.2	
65+							
Total	1,448	305	22	326	1,122	22.5	
Male	453	176	-	176	277	38.9	
Female	995	129	22	150	845	15.1	

Table 4.1.9 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-HARARI REGION: 2015

		Activity Status					
Age Group	All	Eco	onomically Activ	/e		Activity	
and Sex	Persons			Total	Economically	Rate	
		Employed	Unemployed	Active	Not Active		
All Ages							
Total	106,496	59,109	8,209	67,319	39,177	63.2	
Male	49,819	29,900	2,716	32,616	17,203	65.5	
Female	56,677	29,210	5,493	34,703	21,974	61.2	
10-14							
Total	14,752	567	97	663	14,089	4.5	
Male	8,160	181	49	229	7,930	2.8	
Female	6,592	386	48	434	6,158	6.6	
15-19							
Total	13,807	2,181	759	2,941	10,867	21.3	
Male	6,411	729	317	1,046	5,365	16.3	
Female	7,396	1,452	442	1,894	5,502	25.6	
20-24							
Total	11,449	6,836	1,611	8,446	3,003	73.8	
Male	4,494	2,840	528	3,368	1,126	74.9	
Female	6,955	3,996	1,083	5,078	1,877	73.0	
25-29							
Total	13,804	10,935	1,863	12,798	1,006	92.7	
Male	6,887	6,332	376	6,708	179	97.4	
Female	6,917	4,603	1,487	6,090	827	88.0	
30-34							
Total	9,960	8,560	697	9,256	704	92.9	
Male	4,839	4,462	229	4,691	149	96.9	
Female	5,120	4,098	468	4,565	555	89.2	
35-39							
Total	11,642	9,167	1,292	10,459	1,183	89.8	
Male	4,275	3,561	454	4,015	260	93.9	
Female	7,367	5,605	839	6,444	923	87.5	
40-44							
Total	7,203	6,408	586	6,995	208	97.1	
Male	3,901	3,599	262	3,861	40	99.0	
Female	3,302	2,810	324	3,134	168	94.9	
45-49							
Total	6,005	5,173	447	5,620	385	93.6	
Male	3,201	3,069	132	3,201	-	100.0	
Female	2,805	2,104	315	2,420	385	86.3	
50-54							

Total	5,044	4,057	237	4,294	750	85.1	
Male	2,573	2,227	99	2,326	247	90.4	
Female	2,470	1,830	138	1,968	502	79.7	
55-59							
Total	3,020	1,798	365	2,163	858	71.6	
Male	1,297	1,071	186	1,257	41	96.9	
Female	1,723	727	179	906	817	52.6	
60-64							
Total	3,292	1,848	41	1,888	1,404	57.4	
Male	1,531	1,138	-	1,138	394	74.3	
Female	1,761	710	41	751	1,010	42.6	
65+							
Total	6,518	1,581	215	1,795	4,722	27.5	
Male	2,249	691	86	777	1,472	34.5	
Female	4,268	889	129	1,019	3,250	23.9	

Table 4.1.10 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-ADDIS ABABA CITY ADMINISTRATION: 2015

		Activity Status					
Age Group	All	Eco	onomically Activ	ve		Activity	
and Sex	Persons			Total	Economically	Rate	
		Employed	Unemployed	Active	Not Active		
All Ages		1 3				•	
Total	2,844,395	1,362,069	366,227	1,728,296	1,116,099	60.8	
Male	1,289,919	772,367	130,123	902,490	387,430	70.0	
Female	1,554,476	589,703	236,103	825,806	728,670	53.1	
10-14	.,,	,		,	1 = 0,010		
Total	288,827	11,109	6,677 17,786		271,040	6.2	
Male	132,572	4,394	3,339	7,733	124,839	5.8	
Female	156,254	6,715	3,338	10,053	146,201	6.4	
15-19	100,201	0,7 10	0,000	10,000	110,201	0.1	
Total	358,193	73,439	37,991	111,430	246,763	31.1	
Male	136,563	17,715	12,287	30,002	106,562	22.0	
Female	221,629	55,724	25,704	81,428	140,201	36.7	
20-24	221,020	00,121	20,701	01,120	1.10,201	00	
Total	410,305	189,657	94,289	283,945	126,360	69.2	
Male	170,339	89,901	38,828	128,729	41,610	75.6	
Female	239,966	99,756	55,460	155,216	84,750	64.7	
25-29	_00,000	00,.00	33, .33	.00,0	0.,.00	•	
Total	423,669	272,641	81,166	353,807	69,862	83.5	
Male	195,870	148,003	33,267	181,270	14,600	92.5	
Female	227,799	124,638	47,900	172,537	55,261	75.7	
30-34	,	,	,	,	,	_	
Total	327,199	234,644	45,908	280,552	46,648	85.7	
Male	162,345	140,195	13,786	153,982	8,363	94.8	
Female	164,854	94,449	32,122	126,570	38,284	76.8	
35-39	,	,	,	•	,		
Total	284,287	209,490	36,758	246,248	38,039	86.6	
Male	140,316	127,903	8,211	136,113	4,203	97.0	
Female	143,971	81,587	28,547	110,135	33,837	76.5	
40-44	,	,	,	•	,		
Total	160,599	122,301	16,532	138,833	21,766	86.4	
Male	85,122	76,663	4,269	80,932	4,190	95.1	
Female	75,476	45,637	12,263	57,901	17,576	76.7	
45-49	70,170	10,007	12,200	07,001	17,070	70.7	
Total	122,300	77,769	17,061	94,830	27,470	77.5	
Male	60,128	50,325	5,629	55,954	4,174	93.1	
Female	62,172	27,444	11,432	38,876	23,296	62.5	
50-54	02,172	21,177	11,702	00,070	20,200	02.0	
Total	108,325	62,695	5,526	68,221	40,104	63.0	
Male	45,443	39,384	974	40,358	5,085	88.8	
Female	62,882	23,311	4,552	27,863	35,019	44.3	
55-59	02,002	20,011	7,552	21,000	33,013	77.0	
Total	92,115	45,449	13,729	59,178	32,937	64.2	
iotai	52,115	70,770	10,120	55,176	02,001	07.2	

Male	37,064	27,961	4,338	32,299	4,765	87.1	
Female	55,051	17,487	9,391	26,878	28,173	48.8	
60-64							
Total	82,066	26,286	3,394	29,680	52,386	36.2	
Male	33,743	20,138	1,691	21,829	11,914	64.7	
Female	48,323	6,147	1,704	7,851	40,472	16.2	
65+							
Total	186,510	36,591	7,195	43,785	142,724	23.5	
Male	90,413	29,784	3,504	33,288	57,124	36.8	
Female	96,097	6,806	3,691	10,497	85,600	10.9	

Table 4.1.11 Urban population Aged Ten Years and above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach)-DIRE DAWAADMINISTRATION:2015

		Activity Status				
Age Group	All	Eco	onomically Activ			Activity
and Sex	Persons			Total	Economically	Rate
		Employed	Unemployed	Active	Not Active	
All Ages			· · · · · · · · · · · · · · · · · · ·			
Total	226,294	116,325	25,742	142,066	84,227	62.8
Male	107,490	61,997	8,587	70,585	36,905	65.7
Female	118,804	54,327	17,154	71,482	47,322	60.2
10-14						
Total	31,017	2,358	334	2,693	28,324	8.7
Male	16,564	1,295	153	1,448	15,117	8.7
Female	14,452	1,064	181	1,245	13,208	8.6
15-19						
Total	35,714	10,537	2,702	13,240	22,474	37.1
Male	17,386	3,747	965	4,712	12,674	27.1
Female	18,328	6,790	1,738	8,528	9,800	46.5
20-24						
Total	29,463	15,192	6,077	21,270	8,193	72.2
Male	13,637	8,515	1,673	10,188	3,448	74.7
Female	15,826	6,677	4,404	11,081	4,745	70.0
25-29						
Total	29,258	20,743	4,695	25,438	3,820	86.9
Male .	13,607	10,896	1,555	12,451	1,155	91.5
Female	15,652	9,847	3,140	12,987	2,665	83.0
30-34	40.504	40.000	0.050	45.000	0.504	00.0
Total	18,521	13,638	2,352	15,990	2,531	86.3
Male Female	9,328	8,109 5,530	790 1,562	8,899 7,091	429	95.4 77.1
35-39	9,193	5,529	1,302	7,091	2,102	77.1
Total	20,257	14,696	3,266	17,962	2,295	88.7
Male	9,013	7,914	822	8,736	2,293	96.9
Female	11,244	6,781	2,444	9,225	2,018	82.0
40-44	11,277	0,701	۷,٦٦٦	3,223	2,010	02.0
Total	13,224	10,223	1,108	11,331	1,893	85.7
Male	5,997	5,272	424	5,696	301	95.0
Female	7,227	4,952	684	5,636	1,592	78.0
45-49	,	,		-,	,	
Total	15,880	12,794	1,947	14,740	1,139	92.8
Male	7,677	6,918	668	7,586	91	98.8
Female	8,203	5,876	1,279	7,155	1,048	87.2
50-54	-,	-,	- ,	.,.50	.,	- · · -
Total	8,041	5,225	1,179	6,404	1,637	79.6
Male	2,990	2,466	344	2,811	179	94.0
Female	5,051	2,759	835	3,594	1,458	71.1
55-59						
Total	8,074	5,266	1,026	6,292	1,782	77.9
Male	4,024	3,280	468	3,748	277	93.1
Female	4,050	1,986	558	2,544	1,505	62.8
60-64	, -	,		•	,	
Total	5,618	2,839	353	3,193	2,425	56.8
	, -	,		,	, -	_

Male Female	2,639 2,979	1,826 1,013	177 177	2,002 1,190	636 1,789	75.9 39.9	
65+							
Total	11,227	2,813	702	3,515	7,713	31.3	
Male	4,628	1,759	549	2,308	2,320	49.9	
Female	6,599	1,053	153	1,207	5,393	18.3	

Table 4.2 Populations of Major Towns Aged Ten Years and Above by Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach): 2015

rate daring	the Last Seve					
Major Town	All		•	y Status		Activity
and Sex	Persons	Е	conomically Acti		Economically	Rate
		Employed	Unemployed	Total Active	Not Active	
MAJOR TOWNS TOTAL		Zilipioyou	Champioyed	7101170		
Total	5,120,995	2,555,307	610,986	3,166,293	1,954,702	61.8
Male	2,342,960	1,435,723	204,395	1,640,118	702,843	70.0
Female	2,778,035	1,119,585	406,591	1,526,176	1,251,859	54.9
MEKELE TOWN	_,,	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	100,001	1,0=0,110	,,	
Total	247,325	131,641	24,574	156,214	91,111	63.2
Male	111,726	70,849	7,043	77,892	33,835	69.7
Female	135,599	60,792	17,531	78,323	57,277	57.8
ASAYITA TOWN	•	,	·	·	•	
Total	20,428	12,819	1,299	14,118	6,310	69.1
Male	10,263	7,356	412	7,769	2,494	75.7
Female	10,166	5,463	887	6,350	3,816	62.5
GONDER TOWN						
Total	255,459	148,668	26,351	175,018	80,441	68.5
Male	111,299	77,623	7,853	85,476	25,822	76.8
Female	144,161	71,045	18,498	89,542	54,618	62.1
DESSIE TOWN						
Total	154,384	74,940	23,326	98,266	56,119	63.7
Male	70,111	44,569	6,878	51,447	18,664	73.4
Female	84,273	30,371	16,448	46,819	37,454	55.6
BAHIR DAR TOWN						
Total	203,038	121,879	24,142	146,021	57,016	71.9
Male	90,208	62,613	8,315	70,928	19,280	78.6
Female	112,829	59,266	15,827	75,093	37,736	66.6
BISHOFTU TOWN						
Total	125,511	60,204	18,130	78,334	47,176	62.4
Male	56,832	34,597	4,722	39,319	17,514	69.2
Female	68,678	25,607	13,409	39,016	29,663	56.8
ADAMA TOWN						
Total	263,617	122,049	37,097	159,146	104,472	60.4
Male	124,000	72,253	11,552	83,804	40,196	67.6
Female	139,617	49,796	25,545	75,341	64,276	54.0
JIMMA TOWN						
Total	137,760	75,204	12,922	88,126	49,634	64.0
Male	64,432	42,363	4,041	46,404	18,028	72.0
Female	73,328	32,841	8,881	41,722	31,606	56.9
SHASHEMENE TOWN						
Total	112,476	55,806	9,175	64,981	47,495	57.8
Male	51,866	31,944	2,479	34,423	17,442	66.4
Female	60,611	23,862	6,696	30,558	30,053	50.4
JIJIGA TOWN						
Total	101,901	40,815	8,218	49,033	52,868	48.1
Male	49,905	25,947	3,568	29,515	20,390	59.1
Female	51,996	14,868	4,650	19,518	32,478	37.5
ASOSA TOWN						
Total	35,746	21,112	3,087	24,199	11,547	67.7
Male	17,999	12,786	1,012	13,798	4,201	76.7
Female	17,747	8,326	2,075	10,401	7,346	58.6

Table 4.4 Economically Not Active population of Urban Areas Aged Ten Years and Above by Region, Sex and Reasons for Not Being Active during the Last Seven Days (Current Status Approach): 2015

	Ct Being Active			, , , , ,		Not Being Ac	_				
Age Group and Sex	Economically	Home			Injury/		Too	Old			
and Sex	Not Active	Maker	Pregnancy	Student	Disabled	Illness	Young	Age	Pensioned	Remittance	Others
COUNTRY TOTAL											
Total	4,864,055	586,999	192,850	3,070,667	26,030	295,567	49,655	416,641	110,753	47,705	67,187
Male	1,798,373	16,374	-	1,438,159	14,266	105,832	24,123	105,198	62,605	14,012	17,805
Female	3,065,682	570,625	192,850	1,632,508	11,764	189,736	25,532	311,444	48,148	33,693	49,383
TIGRAY REGION											
Total	365,237	50,142	10,794	217,414	3,295	23,474	3,449	47,315	3,831	1,976	3,546
Male	131,111	362	-	103,295	1,614	8,184	2,008	11,436	2,334	706	1,172
Female	234,126	49,780	10,794	114,118	1,681	15,290	1,441	35,879	1,497	1,270	2,374
AFFAR REGION											
Total	65,567	8,214	4,176	43,153	329	3,802	1,403	2,703	450	525	813
Male	26,426	355	-	22,221	-	1,314	692	1,248	450	34	111
Female	39,142	7,859	4,176	20,933	329	2,488	711	1,455	-	491	702
AMHARA REGION											
Total	934,341	87,807	43,527	601,828	3,309	69,852	5,123	98,958	11,553	2,806	9,578
Male	345,710	2,248	-	284,567	1,733	21,100	2,909	21,604	7,690	207	3,651
Female	588,631	85,558	43,527	317,261	1,576	48,751	2,214	77,354	3,863	2,600	5,927
OROMIA REGION											
Total	1,408,777	182,783	61,082	916,441	9,215	84,399	16,377	92,692	23,391	9,529	
Male	521,204	5,635	-	429,192	4,190	31,695	7,855	25,167	11,095	2,878	3,498
Female	887,573	177,148	61,082	487,248	5,025	52,704	8,522	67,525	12,296	6,651	9,370
SOMALI REGION											
Total	133,809	25,030	3,641	77,274	1,088	5,386	4,634	7,743	373	8,174	467
Male	56,580	457	-	44,737	675	3,552	2,001	2,524	161	2,302	171
Female	77,229	24,573	3,641	32,537	413	1,834	2,633	5,219	212	5,872	296
BENISHANGUL-GUN											
Total	40,926	2,217	2,289	29,065	192	3,425	516	2,442	-	315	466
Male	15,524	185	-	13,979	-	730	117	329	-	-	183
Female	25,403	2,032	2,289	15,085	192	2,695	399	2,113	-	315	283
S.N.N.P REGION											
Total	647,859	62,410	18,722	466,945	3,251	27,695	8,322	38,768	9,789	8,583	3,372
Male	247,758	2,388	-	206,157	2,428	12,152	4,517	11,179	5,185	2,298	1,454
Female	400,100	60,022	18,722	260,788	823	15,543	3,805	27,589	4,604	6,285	1,918

73

Table 4.4 Cont'd

Age Group	Reason for Not Being Active											
and Sex	Economically	Home			Injury/		Too	Old				
and Sex	Not Active	Maker	Pregnancy	Student	Disabled	Illness	Young	Age	Pensioned	Remittance	Others	
GAMBELLA REGION												
Total	28,035	2,373	439	20,942	246	1,012	615	1,708	183	290	227	
Male	12,523	116	-	11,129	193	288	217	332	22	62	164	
Female	15,513	2,258	439	9,814	53	723	398	1,376	162	228	63	
HARARI REGION												
Total	39,177	3,409	1,140	25,251	398	2,088	1,396	3,875	790	692	138	
Male	17,203	-	-	13,601	398	1,091	756	916	294	147	-	
Female	21,974	3,409	1,140	11,649	-	998	641	2,959	496	544	138	
ADDIS ABABA CITY ADMINISTRATION												
Total	1,116,099	154,155	44,492	620,704	4,367	70,098	3,827	112,813	59,137	11,904	34,602	
Male	387,430	4,628	-	280,111	2,948	23,952	1,010	28,630	34,702	4,564	6,883	
Female	728,670	149,527	44,492	340,593	1,419	46,146	2,817	84,182	24,435	7,340	27,719	
DIRE DAWA ADMINISTRATION												
Total	84,227	8,458	2,548	51,650	340	4,336	3,993	7,626	1,255	2,911	1,111	
Male	36,905	-	-	29,169	86	1,773	2,041	1,832	673	813	517	
Female	47,322	8,458	2,548	22,481	253	2,563	1,952	5,794	582	2,097	594	

Table 4.2 Cont'd

Major Town	All	Е	conomically Acti	Economically	Activity		
and Sex	Persons	Employed	Unemployed	Total Active	Not Active	Rate	
HAWASSA TOWN							
Total	236,248	124,702	18,436	143,138	93,109	60.6	
Male	113,068	73,264	3,887	77,151	35,917	68.2	
Female 123,1		51,439	14,549	65,987	57,193	53.6	
GAMBELLA TOWN							
Total	49,916	27,965	4,052	32,018	17,899	64.1	
Male	lale 24,022		1,207	16,502	7,520	68.7	
Female 25,89		12,670	2,846	15,515	10,379	59.9	
HARAR TOWN							
Total	106,496	59,109	8,209	67,319	39,177	63.2	
Male	49,819	29,900	2,716	32,616	17,203	65.5	
Female	56,677	29,210	5,493	34,703	21,974	61.2	
ADDIS ABABA CITY							
Total	2,844,395	1,362,069	366,227	1,728,296	1,116,099	60.8	
Male	1,289,919	772,367	130,123	902,490	387,430	70.0	
Female 1,554,476		589,703	236,103	825,806	728,670	53.1	
DIRE DAWA TOWN							
Total	226,294	116,325	25,742	142,066	84,227	62.8	
Male	107,490	61,997	8,587	70,585	36,905	65.7	
Female	118,804	54,327	17,154	71,482	47,322	60.2	

Table 4.3 Economically Not Active Population of Urban Areas Aged Ten Years and Above by Age Group, Sex and Reasons for Not Being Active During the Last Seven Days (Current Status Approach) - COUNTRY TOTAL: 2015

Age Group	Reasons for Not Being Active										
and Sex	Economically	Home			Injury/	-	Too	Old			
	Not Active	Maker	Pregnancy	Student	Disabled	Illness	Young	Age	Pensioned	Remittance	Others
All Ages								-			
Total	4,864,055	586,999	192,850	3,070,667	26,030	295,567	49,655	416,641	110,753	47,705	67,187
Male	1,798,373	16,374	-	1,438,159	14,266	105,832	24,123	105,198	62,605	14,012	17,805
Female	3,065,682	570,625	192,850	1,632,508	11,764	189,736	25,532	311,444	48,148	33,693	49,383
10-14											
Total	1,571,070	25,597	-	1,486,178	2,156	4,964	49,655	-	-	576	1,944
Male	748,271	2,640	-	715,346	2,156	2,944	24,123	-	-	576	485
Female	822,799	22,957	-	770,831	-	2,020	25,532	-	-	-	1,459
15-19											
Total	1,272,258	50,837	13,080	1,188,342	3,322	11,015	-	-	-	549	5,114
Male	556,879	4,155	-	543,980	1,782	4,581	-	-	-	214	2,168
Female	715,379	46,682	13,080	644,362	1,540	6,434	-	-	-	335	2,946
20-24											
Total	485,106	79,740	64,094	307,154	2,433	17,556	-	-	-	1,020	13,110
Male	152,485	1,687	-	138,125	889	8,581	-	-	-	403	2,800
Female	332,621	78,053	64,094	169,028	1,544	8,975	-	-	-	617	10,310
25-29											
Total	262,041	99,633	67,171	57,680	1,261	20,605	-	-	-	4,629	11,063
Male	46,180	2,033	-	28,873	1,089	9,701	-	-	-	1,918	2,566
Female	215,861	97,600	67,171	28,806	172	10,903	-	-	-	2,712	8,497
30-34											
Total	138,833	60,777	30,271	16,332	1,986	18,593	-	-	-	4,460	6,412
Male	20,513	2,083	-	6,669	1,775	7,960	-	-	-	331	1,695
Female	118,320	58,694	30,271	9,664	211	10,633	-	-	-	4,130	4,716
35-39											
Total	122,903	72,498	13,210	3,145	1,935	19,697	-	-	-	4,382	8,037
Male	14,362	894	-	1,193	390	7,350	-	-	-	1,505	3,030
Female	108,541	71,604	13,210	1,952	1,544	12,347	-	-	-		5,007
40-44											
Total	79,545	43,869	2,104	1,635	2,243	19,347	-	-	-	4,990	5,357
Male	13,238	274	-	970	1,451	7,274	-	-	-	1,446	1,824
Female	66,307	43,596	2,104	666	792	12,073	-	-	-	3,544	3,533

 \geq

Table 4.2 Cont'd

Age Group and Sex		Reasons for Not Being Active											
	Economically Not Active	Home Maker	Pregnancy	Student	Injury/ Disabled	Illness	Too Young	Old Age	Pensioned	Remittance	Others		
45-49	1		<u> </u>	'	'	Ц							
Total	85,217	43,430	1,024	1,066	2,527	26,436	-	-	3,268	2,249	5,216		
Male	11,720	1,111	-	101	837	7,489	-	-	-	712	1,469		
Female 50-54	73,496	42,319	1,024	965	1,690	18,947	-	-	3,268	1,537	3,747		
Total	103,060	40,540	529	900	2,039	28,868	-	15,005	5,707	3,908	5,565		
Male	14,419	1,212	-	-	1,000	7,776	-	1,419	1,845	236	932		
Female 55-59	88,641	39,328	529	900	1,039	21,092	-	13,586	3,862	3,673	4,633		
Total	103,764	28,216	882	618	1,308	30,775	-	25,747	10,252	3,619	2,347		
Male	15,122	-	-	-	395	7,510	-	1,950	4,515	531	222		
Female 60-64	88,642	28,216	882	618	913	23,265	-	23,798	5,737	3,088	2,125		
Total	149,768	21,177	90	1,049	3,586	27,319	-	60,908	27,747	5,386	2,504		
Male	29,913	253	-	133	1,866	7,562	-	6,211	12,786	727	375		
Female	119,855	20,924	90	916	1,720	19,756	-	54,698	14,961	4,660	2,129		
65+													
Total	490,491	20,686	395	6,569	1,235	70,393	-	314,980	63,779	11,935	518		
Male	175,271	34	-	2,768	636	27,104	-	95,618	43,459	5,413	239		
Female	315,220	20,652	395	3,801	600	43,289	-	219,362	20,319	6,522	280		

CHAPTER V

CHARACTERISTICS OF THE CURRENTLY URBAN EMPLOYED POPULATION

5. 1. Introduction

The previous chapter has defined the economically active population to be the sum of the employed and the unemployed population. This chapter presents statistical data on the size, distribution and characteristics of the currently employed population based on the definition given in chapter IV. The survey has collected the following information regarding the characteristics of the currently employed population:

- a) Main type of occupation;
- b) Main product or service of the establishment or industry;
- c) Employment status of the main activity and terms of employment;
- d) Reasons for absence from work for those employed persons who worked less than one hour or does not work at all but have formal job attachment during the reference period;
- e) Amount of earnings from paid employees;
- f) Hours worked per week and
- g) Whether employed persons were available or seeking to work additional hours.
- h) Sector of the economy (Formal or informal employment sector);

5.2. Employment to Population Ratio

According to the 18 Key Indicators of Labour Market (KILM) used by the ILO, employment to population ratio is calculated as a percentage of total employed persons to that of the working age population (ILO, 1999). The distribution of employment to population ratio of urban areas of each region by age group and sex is shown in Table 5.1. The data for employment to population ratio of each selected major towns by age group and sex is presented in Table 5.2.

5.3. Occupation and Industry

In this survey, the currently employed persons were asked about the type of main activity or occupation and major product or service of the establishment or industry in which they were engaged during the survey reference period. The questions used to identify the type of occupation and industry, were left open ended so that they will be filled in with the fullest description and its code in the field. For those employed persons engaged in multiple activities, the activity that took the largest share of the respondents' time was taken as the main type of activity.

The responses of the descriptions and codes for the type of occupation and industry filled in by enumerators were further checked and edited at the head quarters using the National Occupational and Industrial Classifications (NOIC) code-book. The NOIC codes were adopted from the International Standard Classification of Occupation (ISCO-08) and International Standard Industrial Classification (ISIC, Revision 4) by taking into account the prevailing national socio-economic conditions.

The NOIC uses a 4-digit coding systems corresponding to 4 level classifications in successively finer detail. In the cases of occupation and industry, the levels are ordered as major divisions, sub major divisions, minor divisions and unit divisions. The occupational classification was categorized into 9 major groups, 42 sub major groups, 132 minor groups and 467 unit groups. The industrial classification contains 21, 88, 238 and 425 major, sub major, minor and unit divisions respectively.

The distribution of employed population of urban areas by major occupational groups, age group and sex for the country total and each region is presented in Table 5.3 and 5.4, respectively. Table 5.5 presents the distribution of employed population by major occupation, sex for each selected major towns. The distribution of employed population by minor occupational groups for each region classified by sex appears in Table 5.6. One should note that in Table 5.6 and 5.10 major and sub major occupation and industrial classification are written in capital letter, while minor occupation and industrial classification are written in small letter, respectively. Analogous information, which deals with the industrial divisions are also shown in Table 5.7 to 5.10.

5.4. Status in Employment

Employment status of a person indicates the level of involvement and degree of decision-making in his main activity. In this survey employed persons were asked about the employment status of their main activity which was classified into: employer, paid employees, self-employed, unpaid family worker, apprentice, and member's of producers, cooperative and others categories. Paid employees were further disaggregated into employees-government, employees government parastatal, employees NGO's, employees private organization, employees domestic and other paid employees.

Table 5.11 and Table 5.12 present the distribution of status in employment by age group and sex for urban areas of country and each region. The distribution of status in employment for major towns classified by sex is presented in Tables 5.13. Table 5.14 contains data on major occupational groups cross classified by major industrial divisions and sex for urban areas of the country total. The status in employment by major industrial divisions and sex for a country total is shown in Table 5.15. Also, the figures for status in employment classified by major occupation and sex for country total are presented in Table 5.16.

5.5. Number of Hours worked

The absence of standard working hours and irregular nature of working days in the informal sectors or in the self employment, data collection as well as getting accurate data on hours of work were found to be difficult. In addition, the reliability of the data collection is likely to be affected due to memory lapse and lack of knowledge or information about the concepts of time on the part of the respondent. Thus, considering the inherent data collection problem on hours of work in developing countries, the figures presented in these tables should be regarded as indicative and not the true levels of intensity of work.

The 2015 Urban Employment and Unemployment Survey included questions on the number of hours worked for two purposes. First, the response on the number of hours worked during the seven days prior to the date of the interview is used to classify the respondents as employed or not on the basis of the given criteria of the minimum required hours of work. As described in Chapter IV, persons who worked at least one hour or more, and those who were not worked or worked less than 1 hour but had a job to return to were considered as employed. The rest of persons were then subjected to other filtering questions for unemployment and not active. The second objective of including questions on number of hours worked was to gather data on intensity of work among the employed population.

In this survey, all persons who were engaged in productive activities during the seven days prior to the date of interview were asked to provide the number of hours they worked in all of their productive activities. The response was recorded for each of the days in a separate notebook so as to get the total actual hours of work during the reference week. Then the hours so obtained were added together for the week and finally filled in the main questionnaire.

According to the resolution concerning statistics on hours actually worked that adopted by the tenth ICLS in 1962 (ILO, 1976), in this survey, the hours actually worked included:

- a) Hours actually worked during normal periods of work and time spent for waiting for the market;
- b) Time worked in addition to hours worked during normal periods of work, and generally paid at higher rates than normal rates (overtime);
- c) Time spent at place of work on activities such as the preparation of the work place; repairing maintenance, preparing and cleaning of tools and others;
- d) Time spent at the place of work waiting or standing by for such reasons as lack of supply of work, break down of machinery, or accidents, or time spent at the place of wok during which no work is done but for which payment is made under a guaranteed employment contract and
- e) Time corresponding to short rest periods at the work place including tea and coffee breaks;

According to the resolution in recording the number of hours worked, care was taken to exclude hours paid for but not worked, such as paid annual leave, paid public holidays or paid sick leave. Also excluded are meal breaks, time spent on travel from home to work and vice versa /for those who have specified place of work/, and hours spent on household activities that were not considered as productive. For employed persons who were not at work during the seven days prior to the date of interview, the number of hours of work is recorded as zero. On the other hand, any time that employed persons have spent in productive activity in places other than work site is considered as working hour.

Table 5.17 presents the distribution of the number of hours worked during the last seven days by age group and sex at country urban level. Similarly, the distribution for each region classified by sex appears in Table 5.18. The distribution of number of hours worked by major industrial divisions, major occupation, and status in employment for country total disaggregated by sex are shown in Table 5.19, 5.20 and 5.21, in that order.

5.6. Earning from Paid Employment

Paid employment jobs are those jobs where the employees hold explicit (written or oral) or implicit employment contracts, which give them a basic remuneration. Some or all of the tools, capitals equipment, information systems and/or premises used by the employees may be owned by others, and the employees may work under direct supervision or according to the strict guidelines set by the owner(s) or persons in the owners' employment. Persons in "paid employment" jobs are typically remunerated by wages and salaries, but may be paid by commission from sales, by piece rates, bonuses or in kind payments such as food, housing or clothing. In this survey, earnings for paid employees refer to gross remuneration and include bonus, overtime, allowances and other benefits that are obtained only from the main job.

Table 5.22 shows the distribution of paid employees disaggregated by types of payment and sex for urban areas of each region. The distribution of paid employees in urban areas by sex and period of payment and sex at country level is presented in Table 5.23. The distribution of paid employees in urban areas of the country classified by amount and period of payment and sex is presented in table 5.24. Paid employees of urban areas of the country by major industrial divisions, sex and amount of payment per month has also shown in Table 5.25.

5.7. Absence from Work

Employed persons that were considered to be absent from work were those who were not engaged in any productive activity or who work less than one hour during the seven days prior to the date of the interview but have formal job attachment. These persons were asked their main reason for being absent from work. Reasons for absence from work was classified as illness, holiday /vacation, Training/education, off-season, reduction in economic activity and "others" categories. The data presented in Table 5.26 indicates the distribution of employed population for their reasons for being absent from work by region and sex.

5.8. Availability to Work Additional Hours

This survey collected data on time related under employment in an attempt to capture the insufficiency of the hours of work of employed persons in relation to an alternative employment situation in which persons are willing and available to be engaged in. The response of employed persons depending on their availability and readiness or not to work for additional hours was coded as "yes" or "no". The response for additional hours available could be for another job (jobs) in addition to the current job; replacing any of current jobs by another job with increased hours of work or longer time on the current job.

The distribution of employed population whether they were available or ready to work additional hours for urban areas of the country by age group and sex is shown in Table 5.27. Table 5.28 shows distribution of employed persons who sought or available to work additional hours together with the actual number of hours worked by age group and sex for the total urban areas of the country. Table 5.29 presents the distribution of urban employed population of each region who were sought or available to work additional hours classified by type of work sought or available for and sex.

5.9. Formal and Informal Sectors of the Economy

For statistical purposes, the informal sector is regarded as a group of production units, which according to the definitions and classification provided in the United Nation System of National Accounts (SNA Rev.4), form part of the household sector as household enterprises or, equivalently, unincorporated enterprises owned by households. The informal sector is defined irrespective of the kind of work place where the productive activities are carried out, the extent of fixed capital used, the duration of the enterprise and its operation as main or secondary activity of the owner.

The figures here were not referring to the whole employed population. In this survey, according to the 15th ICLS recommendations, those employed persons who were engaged in subsistence farming and work in private households were exempted from the analysis of the formal and informal sectors of the economic activity.

This survey includes questions to identify the employment sector of the economy and some of the characteristics in which the employed persons are engaged in their main activity. However, employed persons whose employment status were classified as employee-government, employee-government Development org., employee-non-governmental organization, or members of producers' cooperative were not necessary to ask such questions related to the activities of their sectors because they were considered as formal and later classified as being working in the formal sector.

On the other hand, other employed persons whose employment status were employer, employee of private organization, self employed, unpaid family worker and apprentice were asked the following questions whether the business /enterprise they were engaged in,

- a) Does the enterprise have book of account; or
- b) Does the enterprise have a license; or
- c) Does the product/Service of the organization you engaged in is mainly for market?

Based on the response to these three questions, classification on sector of economy was made as: formal, informal or "not identified". Employed persons who satisfy at least one of the above conditions ('a' or 'b' answered 'yes') were considered as working in the formal sector. On the contrary, those who respond "no" for 'a' and 'b' and 'yes' for 'c', the activity was taken as informal. For those who don't know the situation about their main activity/business/ enterprise with respect to the above questions was labeled as "not identified".

Table 5.30 presents the distribution of employed population of urban areas of the country in relation to the sector of the economy classified by age group and sex. The distribution of employed population by sector of economy, region and sex is presented in 5.31. Similar information for selected major town is shown in Table 5.32. Table 5.33-5.36 presents the cross classification of the informal sector by sex and region by some of the selected variables, such as major occupation and branches of economic sector, educational attainment and working place. The last Table 5.37 has shown the distribution of total employed population by Mean number of hours worked with major branches of economic sector in each region and sex.

Table 5.1 Urban Population of Regions Aged Ten Years and Above by Age Group, Sex and Employment to Population Ratio: 2015

	Total Population Total Employed Population Employment to Population Total Employed Population Total Employed Population								
Age Group		tal Population						Ratio	
Стоир	Total	Male	Female	Total	Male	Female	Total	Male	Female
			C	OUNTRY T	OTAL				
Total	13,381,376		7,157,639	7,083,094	3,963,143	3,119,950	52.9	63.7	43.6
10-14	1,790,429	858,124	932,305	194,969	100,707	94,262	10.9	11.7	
15-19	2,041,908	853,760	1,188,148	565,103	224,956	340,148	27.7	26.3	28.6
20-24	2,011,484	836,505	1,174,979	1,107,918	546,853	561,064	55.1	65.4	47.8
25-29	2,024,794	978,477	1,046,317	1,455,213	827,711	627,502	71.9	84.6	60.0
30-34	1,276,614	663,030	613,583	985,745	599,578	386,166	77.2	90.4	62.9
35-39	1,174,449	559,061	615,388	935,320	518,080	417,240	79.6	92.7	67.8
40-44	713,737	397,066	316,672	578,390	365,839	212,551	81.0	92.1	67.1
45-49	561,559	278,068	283,491	421,698	252,303	169,394	75.1	90.7	59.8
50-54	422,014	195,936	226,078	288,013	173,544	114,470	68.2	88.6	50.6
55-59	345,943	154,230	191,713	210,275	128,640	81,635	60.8	83.4	42.6
60-64	310,206	126,910	183,297	145,598	90,332	55,266	46.9	71.2	
65+	708,239	322,571	385,668	194,851	134,599	60,253	27.5	41.7	15.6
			7	TIGRAY RE	GION				
Total	952,370	411,583	540,787	477,813	253,254	224,559	50.2	61.5	41.5
10-14	116,810	56,402	60,408	6,281	2,213	4,068	5.4	3.9	
15-19	154,538	67,758	86,780	42,231	16,565	25,665	27.3	24.4	29.6
20-24	146,474	44,307	102,167	74,609	27,680	46,929	50.9	62.5	45.9
25-29	143,397	63,153	80,244	108,332	56,044	52,288	75.5	88.7	65.2
30-34	101,199	52,212	48,987	74,997	47,385	27,612	74.1	90.8	56.4
35-39	71,373	30,622	40,751	54,086	28,626	25,460	75.8	93.5	62.5
40-44	39,684	22,539	17,144	29,467	20,162	9,305	74.3	89.5	54.3
45-49	35,537	16,500	19,038	26,063	16,193	9,869	73.3	98.1	51.8
50-54	31,595	17,425	14,170	21,511	15,541	5,970	68.1	89.2	42.1
55-59	21,438	7,149	14,289	13,031	5,982	7,049	60.8	83.7	49.3
60-64	22,037	8,081	13,957	9,713	6,637	3,076	44.1	82.1	22.0
65+	68,288	25,436	42,852	17,492	10,226	7,267	25.6	40.2	17.0
				AFFAR REG	SION				
Total	216,983	108,772	108,211	131,439	75,452	55,987	60.6	69.4	51.7
10-14	29,479	15,835	13,643	4,059	1,826	2,233	13.8	11.5	16.4
15-19	28,316	11,399	16,918	9,262	3,692	5,570	32.7	32.4	32.9
20-24	30,313	16,846	13,466	18,123	11,287	6,835	59.8	67.0	50.8
25-29	36,477	15,061	21,416	27,089	13,825	13,263	74.3	91.8	61.9
30-34	21,738	12,072	9,666	18,064	11,713	6,351	83.1	97.0	65.7
35-39	22,868	12,303	10,565	18,805	11,486	7,319	82.2	93.4	69.3
40-44	12,518	6,660	5,858	10,597	6,291	4,306	84.7	94.4	73.5
45-49	10,638	5,117	5,522	9,295	4,708	4,587	87.4	92.0	83.1
50-54	8,602	4,382	4,219	6,314	4,338	1,976	73.4	99.0	46.8
55-59	6,466	2,348	4,118	3,769	1,714	2,055	58.3	73.0	49.9
60-64	5,062	3,307	1,755	3,711	2,721	990	73.3	82.3	56.4
65+	4,506	3,439	1,066	2,352	1,850	502	52.2	53.8	47.0

Table 5.1 Cont'd

Table 5.1	Cont'd									
Age Group		tal Populatio	n	Total En	nployed Pop	ulation		ent to Por Ratio		
Group	Total	Male	Female	Total	Male	Female	Total	Male	Female	
			Al	MHARA REG	SION					
Total	2,722,586	1,240,992	1,481,594	1,449,091	795,183	653,908	53.2	64.1	44.1	
10-14	352,195	176,748	175,448	44,588	22,776	21,812	12.7	12.9	12.4	
15-19	410,971	164,308	246,663	106,475	42,682	63,793	25.9	26.0	25.9	
20-24	425,178	172,718	252,460	232,951	109,983	122,967	54.8	63.7	48.7	
25-29	412,707	203,310	209,397	299,011	176,619	122,393	72.5	86.9	58.4	
30-34	252,372	127,941	124,431	190,781	116,417	74,364	75.6	91.0	59.8	
35-39	225,093	106,123	118,969	182,038	97,487	84,550	80.9	91.9	71.1	
40-44	129,669	74,582	55,087	108,134	69,255	38,879	83.4	92.9	70.6	
45-49	109,362	50,160	59,202	87,212	46,297	40,915	79.7	92.3	69.1	
50-54	92,623	34,403	58,220	62,033	30,169	31,864	67.0	87.7	54.7	
55-59	79,419	35,936	43,483	50,755	31,112	19,643	63.9	86.6	45.2	
60-64	60,732	25,448	35,285	34,241	20,928	13,313	56.4	82.2	37.7	
65+	172,264	69,316	102,948	50,872	31,457	19,415	29.5	45.4	18.9	
			0	ROMIA REG	ION					
Total	3,898,166	1,863,092	2,035,074	2,102,417	1,213,723	888,694	53.9	65.1	43.7	
10-14	563,599	272,782	290,817	67,735	41,900	25,835	12.0	15.4	8.9	
15-19	632,697	275,662	357,035	185,609	82,315	103,293	29.3	29.9	28.9	
20-24	602,264	255,222	347,041	346,212	183,385	162,826	57.5	71.9	46.9	
25-29	608,648	302,293	306,355	434,406	257,771	176,634	71.4	85.3	57.7	
30-34	349,526	191,280	158,246	280,755	174,014	106,741	80.3	91.0	67.5	
35-39	334,301	159,482	174,819	272,233	148,999	123,234	81.4	93.4	70.5	
40-44	210,778	121,589	89,189	171,711	111,982	59,729	81.5	92.1	67.0	
45-49	163,564	79,937	83,627	120,390	72,919	47,471	73.6	91.2	56.8	
50-54	95,344	50,811	44,533	72,481	45,658	26,823	76.0	89.9	60.2	
55-59	87,204	39,523	47,681	57,461	34,810	22,651	65.9	88.1	47.5	
60-64	87,711	37,757	49,954	44,309	27,178	17,131	50.5	72.0	34.3	
65+	162,531	76,752	85,778	49,116	32,790	16,325	30.2	42.7	19.0	
			S	OMALI REG	ION					
Total	273,737	131,760	141,976	119,082	67,114	51,967	43.5	50.9	36.6	
10-14	52,965	27,439	25,526	6,236	2,695	3,540	11.8	9.8	13.9	
15-19	47,750	22,099	25,651	10,505	3,950	6,555	22.0	17.9	25.6	
20-24	35,088	15,543	19,546	13,337	6,860	6,477	38.0	44.1	33.1	
25-29	34,163	16,071	18,093	21,263	13,257	8,006	62.2	82.5	44.2	
30-34	22,497	10,907	11,590	15,175	9,835	5,339	67.5	90.2	46.1	
35-39	21,200	9,202	11,998	15,790	8,408	7,382	74.5	91.4	61.5	
40-44	18,841	8,729	10,112	14,072	7,918	6,154	74.7	90.7	60.9	
45-49	12,348	6,989	5,359	9,869	5,840	4,029	79.9	83.6	75.2	
50-54	8,798	4,762	4,036	5,320	3,568	1,752	60.5	74.9	43.4	
55-59	3,218	1,885	1,333	1,331	1,024	307	41.3	54.3	23.0	
60-64	6,185	3,231	2,954	3,091	1,855	1,236	50.0	57.4	41.8	
65+	10,685	4,905	5,780	3,094	1,903	1,191	29.0	38.8	20.6	

Table 5.1 Cont'd

Table 5.1	Cont'd						Employment to Population			
Age Group		tal Populatio	n		ployed Pop			Ratio	oulation	
Group	Total	Male	Female	Total	Male	Female	Total	Male	Female	
			BENISHA	NGUL-GUML	JZ REGION					
Total	157,710	77,083	80,627	107,053	58,730	48,323	67.9	76.2	59.9	
10-14	22,907	9,561	13,346	7,388	2,760	4,628	32.3	28.9	34.7	
15-19	23,054	11,884	11,170	9,970	4,889	5,081	43.2	41.1	45.5	
20-24	26,866	10,504	16,361	18,860	8,763	10,098	70.2	83.4	61.7	
25-29	27,224	15,121	12,103	23,990	14,366	9,624	88.1	95.0	79.5	
30-34	14,432	7,513	6,919	13,065	7,378	5,687	90.5	98.2	82.2	
35-39	13,439	7,116	6,323	11,685	6,825	4,861	86.9	95.9	76.9	
40-44	7,539	3,468	4,070	6,630	3,392	3,238	87.9	97.8	79.6	
45-49	7,049	4,169	2,880	6,087	4,050	2,037	86.4	97.1	70.7	
50-54	4,966	2,842	2,124	4,104	2,671	1,434	82.6	94.0	67.5	
55-59	3,918	1,936	1,982	2,293	1,632	661	58.5	84.3	33.4	
60-64	1,778	759	1,018	977	597	381	55.0	78.6	37.4	
65+	4,539	2,209	2,330	2,001	1,408	593	44.1	63.7	25.5	
			S	.N.N.P. REGI	ON					
Total	1,887,693	897,420	990,273	1,096,887	603,653	493,233	58.1	67.3	49.8	
10-14	301,593	134,093	167,500	39,806	18,628	21,178	13.2	13.9	12.6	
15-19	320,875	132,561	188,314	108,119	45,420	62,699	33.7	34.3	33.3	
20-24	277,103	125,632	151,471	180,025	92,276	87,749	65.0	73.4	57.9	
25-29	279,538	139,390	140,148	223,124	123,468	99,656	79.8	88.6	71.1	
30-34	149,010	79,612	69,398	127,137	75,130	52,007	85.3	94.4	74.9	
35-39	163,312	76,667	86,645	141,276	73,033	68,243	86.5	95.3	78.8	
40-44	109,822	62,386	47,436	95,357	59,331	36,026	86.8	95.1	75.9	
45-49	75,533	42,846	32,687	64,394	40,865	23,529	85.3	95.4	72.0	
50-54	56,464	29,101	27,362	42,277	26,341	15,936	74.9	90.5	58.2	
55-59	39,865	22,323	17,542	28,495	19,472	9,023	71.5	87.2	51.4	
60-64	34,855	10,038	24,817	18,241	7,137	11,105	52.3	71.1	44.7	
65+	79,724	42,771	36,953	28,636	22,554	6,083	35.9	52.7	16.5	
			GA	MBELLA REC	SION					
Total	94,945	45,805	49,141	61,809	31,771	30,038	65.1	69.4	61.1	
10-14	16,286	7,967	8,319	4,843	2,040	2,803	29.7	25.6	33.7	
15-19	15,992	7,729	8,264	6,775	3,251	3,524	42.4	42.1	42.6	
20-24	16,983	7,264	9,719	12,116	5,364	6,753	71.3	73.8	69.5	
25-29	15,908	7,715	8,193	13,679	7,128	6,551	86.0	92.4	80.0	
30-34	10,160	4,981	5,179	8,929	4,940	3,989	87.9	99.2	77.0	
35-39	6,677	3,941	2,735	6,056	3,839	2,217	90.7	97.4	81.0	
40-44	3,861	2,091	1,770	3,489	1,976	1,513	90.4	94.5	85.5	
45-49	3,342	1,344	1,997	2,652	1,119	1,534	79.4	83.2	76.8	
50-54	2,213	1,201	1,012	1,995	1,180	816	90.2	98.2	80.6	
55-59	1,206	744	462	628	582	46	52.1	78.1	10.0	
60-64	870	375	495	341	177	165	39.2	47.1	33.2	
65+	1,448	453	995	305	176	129	21.0	38.9	12.9	

Table 5.1 Cont'd

Table 5.1	Cont'd			I.E. J. (C. D. LC.					
Age Group		tal Populatio	n		ployed Pop	ulation		ent to Por Ratio	oulation
Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
			Н	ARARI REGI	NC				
Total	106,496	49,819	56,677	59,109	29,900	29,210	55.5	60.0	51.5
10-14	14,752	8,160	6,592	567	181	386	3.8	2.2	5.9
15-19	13,807	6,411	7,396	2,181	729	1,452	15.8	11.4	19.6
20-24	11,449	4,494	6,955	6,836	2,840	3,996	59.7	63.2	57.4
25-29	13,804	6,887	6,917	10,935	6,332	4,603	79.2	91.9	66.5
30-34	9,960	4,839	5,120	8,560	4,462	4,098	85.9	92.2	80.0
35-39	11,642	4,275	7,367	9,167	3,561	5,605	78.7	83.3	76.1
40-44	7,203	3,901	3,302	6,408	3,599	2,810	89.0	92.2	85.1
45-49	6,005	3,201	2,805	5,173	3,069	2,104	86.1	95.9	75.0
50-54	5,044	2,573	2,470	4,057	2,227	1,830	80.4	86.6	74.1
55-59	3,020	1,297	1,723	1,798	1,071	727	59.5	82.5	42.2
60-64	3,292	1,531	1,761	1,848	1,138	710	56.1	74.3	40.3
65+	6,518	2,249	4,268	1,581	691	889	24.3	30.7	20.8
		A	ADDIS ABAE	BA CITY ADM	INISTRATIO	NC			
Total	2,844,395	1,289,919	1,554,476	1,362,069	772,367	589,703	47.9	59.9	37.9
10-14	288,827	132,572	156,254	11,109	4,394	6,715	3.8	3.3	4.3
15-19	358,193	136,563	221,629	73,439	17,715	55,724	20.5	13.0	25.1
20-24	410,305	170,339	239,966	189,657	89,901	99,756	46.2	52.8	41.6
25-29	423,669	195,870	227,799	272,641	148,003	124,638	64.4	75.6	54.7
30-34	327,199	162,345	164,854	234,644	140,195	94,449	71.7	86.4	57.3
35-39	284,287	140,316	143,971	209,490	127,903	81,587	73.7	91.2	56.7
40-44	160,599	85,122	75,476	122,301	76,663	45,637	76.2	90.1	60.5
45-49	122,300	60,128	62,172	77,769	50,325	27,444	63.6	83.7	44.1
50-54	108,325	45,443	62,882	62,695	39,384	23,311	57.9	86.7	37.1
55-59	92,115	37,064	55,051	45,449	27,961	17,487	49.3	75.4	31.8
60-64	82,066	33,743	48,323	26,286	20,138	6,147	32.0	59.7	12.7
65+	186,510	90,413	96,097	36,591	29,784	6,806	19.6	32.9	7.1
			DIRE DA	WA ADMINIS	STRATION				
Total	226,294	107,490	118,804	116,325	61,997	54,327	51.4	57.7	45.7
10-14	31,017	16,564	14,452	2,358	1,295	1,064	7.6	7.8	7.4
15-19	35,714	17,386	18,328	10,537	3,747	6,790	29.5	21.6	37.0
20-24	29,463	13,637	15,826	15,192	8,515	6,677	51.6	62.4	42.2
25-29	29,258	13,607	15,652	20,743	10,896	9,847	70.9	80.1	62.9
30-34	18,521	9,328	9,193	13,638	8,109	5,529	73.6	86.9	60.1
35-39	20,257	9,013	11,244	14,696	7,914	6,781	72.5	87.8	60.3
40-44	13,224	5,997	7,227	10,223	5,272	4,952	77.3	87.9	68.5
45-49	15,880	7,677	8,203	12,794	6,918	5,876	80.6	90.1	71.6
50-54	8,041	2,990	5,051	5,225	2,466	2,759	65.0	82.5	54.6
55-59	8,074	4,024	4,050	5,266	3,280	1,986	65.2	81.5	49.0
60-64	5,618	2,639	2,979	2,839	1,826	1,013	50.5	69.2	34.0
65+	11,227	4,628	6,599	2,813	1,759	1,053	25.1	38.0	16.0
	11,55	.,020	3,000	_,0.0	.,,,,	.,000		30.0	

34 Table 5.2 Population of Major Towns Aged Ten Years and Above by Age Group, Sex and Employment to Population Ratio: 2015

Age Group	To	otal Population			mployed Pop	ulation	Employm	ent to P	opulation
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
			MAJO	R TOWNS TO	OTAL				
A.I. A	F 077 047	0.044.055				4 440 004	50.4	04.7	40.0
All Ages	5,077,247	2,341,255	2,735,993	2,560,286	1,443,652	1,116,634	50.4	61.7	
10-14	588,702	282,846	305,856	54,284	26,644	27,640	9.2	9.4	9.0
15-19	701,143	282,624	418,519	178,024	59,099	118,925	25.4	20.9	28.4
20-24	747,464	309,288	438,176	373,461	179,472	193,989	50.0	58.0	44.3
25-29	771,475	360,738	410,737	518,617	288,787	229,830	67.2	80.1	56.0
30-34	543,817	275,902	267,915	400,873	241,941	158,931	73.7	87.7	59.3
35-39	474,942	234,107	240,835	357,279	213,752	143,527	75.2	91.3	59.6
40-44	277,776	148,805	128,971	215,534	134,251	81,283	77.6	90.2	63.0
45-49	219,362	110,345	109,017	153,341	96,439	56,903	69.9	87.4	52.2
50-54	181,007	78,565	102,442	111,163	67,515	43,648	61.4	85.9	42.6
55-59	150,422	65,114	85,309	82,288	51,393	30,895	54.7	78.9	36.2
60-64	130,003	55,881	74,122	51,024	36,567	14,456	39.2	65.4	19.5
65+	291,134	137,041	154,093	64,399	47,792	16,608	22.1	34.9	10.8
			ME	EKELE TOW	N				
All Ages	247,325	111,726	135,599	131,641	70,849	60,792	53.2	63.4	44.8
10-14	30,885	15,838	15,047	2,834	1,430	1,403	9.2	9.0	9.3
15-19	33,423	14,322	19,101	11,652	3,880	7,772	34.9	27.1	40.7
20-24	39,143	13,631	25,512	20,309	7,976	12,334	51.9	58.5	48.3
25-29	43,700	17,892	25,808	29,474	14,503	14,971	67.4	81.1	58.0
30-34	29,508	15,112	14,395	22,540	14,036	8,504	76.4	92.9	59.1
35-39	19,728	10,235	9,493	15,682	10,144	5,538	79.5	99.1	58.3
40-44	11,710	6,484	5,226	9,110	5,654	3,456	77.8	87.2	66.1
45-49	8,058	3,986	4,072	5,868	3,986	1,881	72.8	100.0	46.2
50-54	7,411	3,686	3,725	4,505	2,939	1,566	60.8	79.7	42.1
55-59	5,778	2,683	3,096	4,005	2,459	1,545	69.3	91.7	49.9
60-64	5,466	2,416	3,050	2,764	1,969	795	50.6	81.5	26.1
65+	12,516	5,442	7,074	2,898	1,871	1,026	23.2	34.4	14.5
			AS	SAYITA TOW	N				
All Ages	20,428	10,263	10,166	12,819	7,356	5,463	62.8	71.7	53.7
			941		7,356 34				
10-14	2,062	1,121		119		85 470	5.8	3.0	9.0
15-19	2,504	1,046	1,458	718	240	478	28.7	23.0	32.8
20-24	2,996	1,294	1,702	1,955	1,078	877	65.3	83.4	
25-29	3,838	1,809	2,029	2,970	1,658	1,313	77.4	91.6	
30-34	1,896	995	901	1,619	963	656	85.4	96.8	
35-39	1,953	1,201	753	1,695	1,144	551	86.8	95.3	
40-44	1,327	699	628	1,160	659	501	87.4	94.3	
45-49	1,082	642	440	937	563	374	86.6	87.7	
50-54	800	314	486	494	270	224	61.7	85.9	46.1
55-59	690	296	394	401	203	198	58.2	68.7	
60-64	477	249	227	339	212	127	71.1	84.9	55.9
65+	804	598	207	412	332	80	51.2	55.5	38.8
				87					

Table 5.2 Cont'd

Table 5.2 Co	ont'd						I =		andati	
Age Group	To	tal Population	on	Total E	mployed Po	pulation	Employment to Population Ratio			
	Total	Male	Female	Total	Male	Female	Total	Male	Female	
			GC	NDER TOW	'N					
All Ages	245,910	112,789	133,121	162,404	85,621	76,782	66.0	75.9	57.7	
10-14	41,372	19,979	21,392	22,957	11,963	10,994	55.5	59.9	51.4	
15-19	38,583	16,080	22,503	20,363	9,394	10,969	52.8	58.4	48.7	
20-24	33,500	15,562	17,938	22,609	11,760	10,849	67.5	75.6	60.5	
25-29	32,351	15,018	17,332	24,427	12,980	11,447	75.5	86.4	66.0	
30-34	21,265	10,528	10,737	17,702	10,117	7,585	83.2	96.1	70.6	
35-39	19,163	6,737	12,426	14,926	6,125	8,801	77.9	90.9	70.8	
40-44	12,923	8,015	4,907	11,104	7,244	3,860	85.9	90.4	78.7	
45-49	11,038	6,041	4,997	8,952	5,482	3,471	81.1	90.7	69.4	
50-54	7,187	2,267	4,921	4,606	1,986	2,620	64.1	87.6	53.2	
55-59	7,947	3,206	4,742	5,418	2,895	2,522	68.2	90.3	53.2	
60-64	6,695	3,509	3,186	3,780	2,380	1,400	56.5	67.8	44.0	
65+	13,886	5,848	8,039	5,559	3,296	2,264	40.0	56.4	28.2	
			DI	ESSIE TOW	N					
All Ages	144,091	68,328	75,764	71,530	43,161	28,369	49.6	63.2	37.4	
10-14	14,552	6,811	7,741	647	124	523	4.4	1.8	6.8	
15-19	20,598	9,229	11,370	4,846	1,933	2,914	23.5	20.9	25.6	
20-24	20,389	9,339	11,050	10,422	6,026	4,396	51.1	64.5	39.8	
25-29	21,661	10,384	11,277	14,155	9,210	4,945	65.3	88.7	43.9	
30-34	13,472	7,106	6,367	9,980	6,561	3,420	74.1	92.3	53.7	
35-39	12,812	6,316	6,495	9,047	5,360	3,687	70.6	84.9	56.8	
40-44	8,780	3,940	4,840	5,703	3,072	2,631	65.0	78.0	54.4	
45-49	6,223	3,536	2,688	4,827	3,256	1,572	77.6	92.1	58.5	
50-54	6,240	2,746	3,494	4,294	2,456	1,837	68.8	89.4	52.6	
55-59	4,256	2,000	2,257	2,267	1,426	841	53.3	71.3	37.3	
60-64	4,288	1,949	2,339	2,502	1,703	799	58.3	87.4	34.1	
65+	10,820	4,974	5,846	2,839	2,036	803	26.2	40.9	13.7	
			BAH	IIR DAR TO\	WN					
All Ages	196,160	91,444	104,717	124,310	66,389	57,921	63.4	72.6	55.3	
10-14	23,640	13,008	10,633	7,334	5,210	2,124	31.0	40.1	20.0	
15-19	31,117	13,388	17,729	14,241	6,295	7,946	45.8	47.0	44.8	
20-24	33,127	10,648	22,478	21,480	7,647	13,834	64.8	71.8	61.5	
25-29	33,649	16,752	16,897	25,674	14,295	11,379	76.3	85.3	67.3	
30-34	20,417	10,115	10,302	16,842	9,074	7,767	82.5	89.7	75.4	
35-39	16,165	8,874	7,291	13,826	8,208	5,618	85.5	92.5	77.1	
40-44	8,211	4,407	3,803	6,975	4,247	2,728	85.0	96.4	71.7	
45-49	7,650	4,794	2,855	6,843	4,727	2,116	89.5	98.6	74.1	
50-54	4,718	1,363	3,355	3,188	1,363	1,825	67.6	100.0	54.4	
55-59	4,955	2,053	2,902	3,686	1,960	1,726	74.4	95.5	59.5	
60-64	3,464	1,434	2,030	1,597	1,250	347	46.1	87.2	17.1	
65+	9,048	4,607	4,441	2,622	2,113	509	29.0	45.9	11.5	
	,	,	,	88	,	-				

Table 5.2 Cont'd

Table 5.2 Co	ont'd								
Age Group		tal Population	on		mployed Por	ı	Employm	ent to Po Ratio	-
	Total	Male	Female	Total	Male	Female	Total	Male	Female
			BIS	HOFTU TOV	/N				
All Ages	119,970	55,080	64,890	57,994	33,733	24,261	48.3	61.2	37.4
10-14	13,893	5,900	7,993	171	114	57	1.2	1.9	0.7
15-19	16,657	6,401	10,256	2,182	570	1,612	13.1	8.9	15.7
20-24	18,081	7,714	10,367	10,114	4,968	5,147	55.9	64.4	49.6
25-29	19,639	9,609	10,030	12,976	8,000	4,976	66.1	83.3	49.6
30-34	13,621	6,722	6,899	10,085	5,792	4,293	74.0	86.2	62.2
35-39	10,691	5,791	4,900	7,826	5,032	2,794	73.2	86.9	57.0
40-44	5,523	2,867	2,657	4,428	2,667	1,761	80.2	93.0	66.3
45-49	5,074	2,495	2,579	3,810	2,266	1,544	75.1	90.8	59.9
50-54	3,945	1,928	2,017	2,204	1,471	732	55.9	76.3	36.3
55-59	3,613	1,684	1,929	1,778	1,148	631	49.2	68.2	32.7
60-64	2,837	1,145	1,692	1,075	919	155	37.9	80.3	9.2
65+	6,395	2,824	3,571	1,344	786	558	21.0	27.8	15.6
			AI	NOT AMAC	N				
All Ages	255,988	122,131	133,856	117,828	69,953	47,874	46.0	57.3	35.8
10-14	28,552	15,624	12,927	980	436	544	3.4	2.8	4.2
15-19	39,362	16,673	22,689	7,934	3,127	4,808	20.2	18.8	21.2
20-24	40,120	16,551	23,569	19,459	9,598	9,861	48.5	58.0	41.8
25-29	42,962	20,432	22,530	27,068	16,018	11,050	63.0	78.4	49.0
30-34	24,186	12,923	11,263	15,247	9,967	5,280	63.0	77.1	46.9
35-39	22,832	12,008	10,825	15,909	10,521	5,388	69.7	87.6	49.8
40-44	12,848	6,926	5,922	9,425	6,141	3,284	73.4	88.7	55.4
45-49	10,156	4,443	5,713	6,336	3,739	2,597	62.4	84.2	45.5
50-54	9,109	4,667	4,442	5,610	3,475	2,135	61.6	74.5	48.1
55-59	7,509	3,499	4,010	4,177	3,041	1,136	55.6	86.9	28.3
60-64	6,419	3,082	3,337	3,218	2,261	958	50.1	73.4	28.7
65+	11,933	5,304	6,629	2,465	1,629	836	20.7	30.7	12.6
			JI	MMA TOWN	I				
All Ages	138,824	65,928	72,896	75,289	42,862	32,428	54.2	65.0	44.5
10-14	20,241	9,819	10,422	2,130	436	1,693	10.5	4.4	16.2
15-19	22,138	8,455	13,683	7,784	2,844	4,940	35.2	33.6	36.1
20-24	22,224	9,760	12,464	12,581	7,078	5,503	56.6	72.5	44.2
25-29	23,181	11,956	11,226	16,772	10,619	6,153	72.3	88.8	54.8
30-34	12,183	7,261	4,922	9,627	6,386	3,241	79.0	87.9	65.8
35-39	10,504	4,153	6,351	7,891	3,756	4,135	75.1	90.4	65.1
40-44	6,672	3,451	3,221	5,772	3,231	2,541	86.5	93.6	78.9
45-49	5,747	3,272	2,474	4,555	3,003	1,552	79.3	91.8	62.7
50-54	4,225	2,303	1,923	3,227	2,148	1,078	76.4	93.3	56.1
55-59	3,806	1,990	1,816	2,422	1,781	641	63.6	89.5	35.3
60-64	2,293	830	1,463	1,090	483	607	47.5	58.2	41.5
65+	5,609	2,678	2,931	1,440	1,095	344	25.7	40.9	11.7

Table 5.2 Cont'd

Table 5.2 Co	ont'd								
Age Group		tal Population	on		mployed Por	Employment to Population Ratio			
	Total	Male	Female	Total	Male	Female	Total	Male	Female
			SHAS	HEMENE TO	NWC				
All Ages	107,554	51,342	56,212	54,374	32,171	22,203	50.6	62.7	39.5
10-14	14,997	7,709	7,288	662	320	341	4.4	4.2	4.7
15-19	16,852	7,124	9,728	3,942	1,725	2,216	23.4	24.2	22.8
20-24	16,114	5,954	10,160	7,354	3,708	3,646	45.6	62.3	35.9
25-29	19,119	9,029	10,089	13,747	7,986	5,762	71.9	88.4	57.1
30-34	11,285	6,202	5,083	8,748	5,572	3,176	77.5	89.8	62.5
35-39	10,518	6,157	4,361	8,550	5,893	2,656	81.3	95.7	60.9
40-44	5,233	3,151	2,083	4,509	3,042	1,466	86.2	96.6	70.4
45-49	3,403	1,562	1,841	2,498	1,337	1,161	73.4	85.6	63.1
50-54	2,509	908	1,602	1,378	706	672	54.9	77.8	42.0
55-59	2,356	1,300	1,056	1,337	986	351	56.8	75.8	33.3
60-64	1,327	653	673	778	550	228	58.7	84.2	33.9
65+	3,842	1,595	2,247	872	346	526	22.7	21.7	23.4
			JI	JIGA TOWN	l				
All Ages	101,901	49,905	51,996	40,815	25,947	14,868	40.1	52.0	28.6
10-14	17,723	9,154	8,569	424	131	293	2.4	1.4	3.4
15-19	15,765	6,976	8,790	3,346	1,539	1,808	21.2	22.1	20.6
20-24	15,379	7,346	8,034	5,369	3,114	2,256	34.9	42.4	28.1
25-29	13,626	6,403	7,223	8,041	5,392	2,650	59.0	84.2	36.7
30-34	9,084	4,652	4,432	5,672	3,979	1,693	62.4	85.5	38.2
35-39	8,021	3,436	4,585	5,636	3,079	2,557	70.3	89.6	55.8
40-44	6,992	3,732	3,260	4,851	3,215	1,636	69.4	86.2	50.2
45-49	4,119	2,725	1,394	2,949	2,276	674	71.6	83.5	48.3
50-54	4,230	2,346	1,884	2,324	1,714	610	54.9	73.1	32.4
55-59	1,397	737	660	506	314	193	36.2	42.5	29.2
60-64	1,863	776	1,087	936	654	282	50.3	84.3	25.9
65+	3,700	1,623	2,077	760	541	219	20.5	33.3	10.5
			AS	SOSA TOWN	N				
All Ages	35,746	17,999	17,747	21,112	12,786	8,326	59.1	71.0	46.9
10-14	4,882	2,354	2,529	191	100	91	3.9	4.3	3.6
15-19	4,743	2,206	2,537	1,305	741	563	27.5	33.6	22.2
20-24	6,874	2,457	4,417	4,124	1,819	2,304	60.0	74.1	52.2
25-29	7,224	3,989	3,235	5,852	3,643	2,209	81.0	91.3	68.3
30-34	3,433	2,163	1,270	2,775	2,028	747	80.8	93.8	58.8
35-39	2,995	1,875	1,120	2,582	1,804	778	86.2	96.2	69.5
40-44	1,741	839	902	1,563	839	724	89.7	100.0	80.2
45-49	1,358	816	542	1,122	774	348	82.6	94.8	64.2
50-54	972	590	382	747	547	199	76.8	92.9	52.1
55-59	509	260	248	380	217	162	74.6	83.6	65.3
60-64	250	172	78	201	172	29	80.4	100.0	36.9
65+	766	279	487	271	100	171	35.3	35.7	35.1

Table 5.2 Cont'd

Table 5.2 C	ont'd						T			
Age Group		otal Populatio	1		mployed Pop	1	Employment to Population Ratio			
	Total	Male	Female	Total	Male	Female	Total	Male	Female	
			HAV	WASSA TOW	/N					
All Ages	236,248	113,068	123,180	124,702	73,264	51,439	52.8	64.8	41.8	
10-14	33,672	14,566	19,107	1,401	312	1,089	4.2	2.1	5.7	
15-19	43,052	16,571	26,480	11,146	3,515	7,631	25.9	21.2	28.8	
20-24	39,012	16,690	22,322	20,475	11,016	9,459	52.5	66.0	42.4	
25-29	35,937	17,365	18,573	26,795	15,911	10,884	74.6	91.6	58.6	
30-34	22,352	13,026	9,326	18,594	12,157	6,436	83.2	93.3	69.0	
35-39	19,098	11,106	7,992	16,541	10,793	5,748	86.6	97.2	71.9	
40-44	12,676	8,012	4,663	10,206	7,505	2,701	80.5	93.7	57.9	
45-49	9,407	4,115	5,292	7,531	3,867	3,664	80.1	94.0	69.2	
50-54	7,228	3,879	3,350	5,702	3,820	1,883	78.9	98.5	56.2	
55-59	3,717	2,504	1,212	2,958	2,209	749	79.6	88.2	61.7	
60-64	3,178	1,561	1,617	1,525	831	694	48.0	53.2	42.9	
65+	6,918	3,672	3,246	1,828	1,328	500	26.4	36.2	15.4	
			GAN	MBELLA TOV	VN					
All Ages	49,916	24,022	25,894	27,965	15,295	12,670	56.0	63.7	48.9	
10-14	7,635	3,668	3,968	400	164	236	5.2	4.5	6.0	
15-19	8,633	3,793	4,840	2,406	1,105	1,301	27.9	29.1	26.9	
20-24	9,289	3,873	5,416	5,524	2,429	3,095	59.5	62.7	57.1	
25-29	7,857	3,736	4,121	6,346	3,343	3,003	80.8	89.5	72.9	
30-34	5,434	2,584	2,851	4,601	2,543	2,059	84.7	98.4	72.2	
35-39	4,276	2,615	1,661	3,815	2,513	1,301	89.2	96.1	78.4	
40-44	2,114	1,262	853	1,796	1,200	596	84.9	95.1	69.9	
45-49	1,861	912	949	1,376	850	525	73.9	93.2	55.4	
50-54	1,022	563	459	907	541	366	88.8	96.2	79.7	
55-59	681	517	164	442	442	-	64.8	85.5	-	
60-64	472	192	280	246	82	165	52.2	42.5	58.9	
65+	641	308	333	106	84	22	16.6	27.3	6.7	
			H	ARAR TOWN	١					
All Ages	106,496	49,819	56,677	59,109	29,900	29,210	55.5	60.0	51.5	
10-14	14,752	8,160	6,592	567	181	386	3.8	2.2	5.9	
15-19	13,807	6,411	7,396	2,181	729	1,452	15.8	11.4	19.6	
20-24	11,449	4,494	6,955	6,836	2,840	3,996	59.7	63.2	57.4	
25-29	13,804	6,887	6,917	10,935	6,332	4,603	79.2	91.9	66.5	
30-34	9,960	4,839	5,120	8,560	4,462	4,098	85.9	92.2	80.0	
35-39	11,642	4,275	7,367	9,167	3,561	5,605	78.7	83.3	76.1	
40-44	7,203	3,901	3,302	6,408	3,599	2,810	89.0	92.2	85.1	
45-49	6,005	3,201	2,805	5,173	3,069	2,104	86.1	95.9	75.0	
50-54	5,044	2,573	2,470	4,057	2,227	1,830	80.4	86.6	74.1	
55-59	3,020	1,297	1,723	1,798	1,071	727	59.5	82.5	42.2	
60-64	3,292	1,531	1,761	1,848	1,138	710	56.1	74.3	40.3	
65+	6,518	2,249	4,268	1,581	691	889	24.3	30.7	20.8	

Table 5.2 Cont'd

Age Group	To	otal Population	on	Total Er	oulation	Employment to Population Ratio			
	Total	Male	Female	Total	Male	Female	Total	Male	Female
			ADD	IS ABABA C	TY				
All Ages	2,844,395	1,289,919	1,554,476	1,362,069	772,367	589,703	47.9	59.9	37.9
10-14	288,827	132,572	156,254	11,109	4,394	6,715	3.8	3.3	4.3
15-19	358,193	136,563	221,629	73,439	17,715	55,724	20.5	13.0	25.1
20-24	410,305	170,339	239,966	189,657	89,901	99,756	46.2	52.8	41.6
25-29	423,669	195,870	227,799	272,641	148,003	124,638	64.4	75.6	54.7
30-34	327,199	162,345	164,854	234,644	140,195	94,449	71.7	86.4	57.3
35-39	284,287	140,316	143,971	209,490	127,903	81,587	73.7	91.2	56.7
40-44	160,599	85,122	75,476	122,301	76,663	45,637	76.2	90.1	60.5
45-49	122,300	60,128	62,172	77,769	50,325	27,444	63.6	83.7	44.1
50-54	108,325	45,443	62,882	62,695	39,384	23,311	57.9	86.7	37.1
55-59	92,115	37,064	55,051	45,449	27,961	17,487	49.3	75.4	31.8
60-64	82,066	33,743	48,323	26,286	20,138	6,147	32.0	59.7	12.7
65+	186,510	90,413	96,097	36,591	29,784	6,806	19.6	32.9	7.1
			DIRE	E DAWA TOV	VN				
All Ages	226,294	107,490	118,804	116,325	61,997	54,327	51.4	57.7	45.7
10-14	31,017	16,564	14,452	2,358	1,295	1,064	7.6	7.8	7.4
15-19	35,714	17,386	18,328	10,537	3,747	6,790	29.5	21.6	37.0
20-24	29,463	13,637	15,826	15,192	8,515	6,677	51.6	62.4	42.2
25-29	29,258	13,607	15,652	20,743	10,896	9,847	70.9	80.1	62.9
30-34	18,521	9,328	9,193	13,638	8,109	5,529	73.6	86.9	60.1
35-39	20,257	9,013	11,244	14,696	7,914	6,781	72.5	87.8	60.3
40-44	13,224	5,997	7,227	10,223	5,272	4,952	77.3	87.9	68.5
45-49	15,880	7,677	8,203	12,794	6,918	5,876	80.6	90.1	71.6
50-54	8,041	2,990	5,051	5,225	2,466	2,759	65.0	82.5	54.6
55-59	8,074	4,024	4,050	5,266	3,280	1,986	65.2	81.5	49.0
60-64	5,618	2,639	2,979	2,839	1,826	1,013	50.5	69.2	34.0
65+	11,227	4,628	6,599	2,813	1,759	1,053	25.1	38.0	16.0

Table 5.3 Employed Population of Urban Areas Aged Ten Years and Above by Major Occupational Groups, Age Group, and Sex - COUNTRY TOTAL: 2015

					Major Oc	cupational (Groups				
					-		Skilled		Plant and		
Age Group				Technicians			agricultural,	Craft and	machine		
and Sex	Total			and	Clerical	Service	forestry	related	operators,		
	Employed			associate	support	and sales	and fishery	trades	and	Elementary	
	Person	Managers	Professionals	professionals	workers	workers	workers	workers	assemblers	occupations	Others
All Ages											
Total	7,083,094	155,628	560,307	•		2,162,111	387,801	1,089,180	458,982	1,434,946	9,032
Male	3,963,143	119,462	370,102	345,470	90,129	912,282	270,676	696,336	418,261	733,172	7,254
Female	3,119,950	36,166	190,206	221,219	168,287	1,249,829	117,126	392,845	40,721	701,775	1,777
10-14											
Total	194,969	-	-	-	-	51,322	24,934	14,903	394	103,416	-
Male	100,707	-	-	-	-	19,520	19,517	8,712	394	52,564	-
Female	94,262	-	-	-	-	31,802	5,417	6,190	-	50,852	-
15-19											
Total	565,103	1,568	-	9,268	7,733	221,013	27,135	80,721	14,139	203,527	-
Male	224,956	889	-	2,024	2,222	61,970	18,707	45,929	10,207	83,007	-
Female	340,148	679	-	7,244	5,512	159,043	8,427	34,791	3,933	120,520	-
20-24											
Total	1,107,918	12,002	66,532	85,662	47,973	361,432	28,822	179,514	76,481	249,138	361
Male	546,853	7,069	32,104	40,507	9,878	135,119	19,884	112,924	63,892	125,163	313
Female	561,064	4,933	34,427	45,155	38,095	226,313	8,938	66,591	12,589	123,975	48
25-29											
Total	1,455,213	25,777	165,060	154,858	76,700	395,727	46,407	224,843	112,487	251,069	2,284
Male	827,711	18,457	99,820	88,452	26,866	161,451	33,696	153,300	104,808	138,961	1,900
Female	627,502	7,320	65,240	66,406	49,834	234,277	12,711	71,543	7,679	112,108	384
30-34											
Total	985,745	33,891	105,022	87,616	33,390	277,330	41,876	156,967	89,541	157,825	2,287
Male	599,578	27,412	73,989	53,735	10,002	132,973	27,438	110,795	82,461	79,696	1,078
Female	386,166	6,480	31,033	33,881	23,389	144,356	14,438	46,171	7,080	78,129	1,209
35-39											
Total	935,320	22,449	82,945	83,157	39,484	296,752	44,179	145,396	64,095	155,529	1,334
Male	518,080	16,242	56,205	53,829	12,681	126,143	25,530	95,430	60,183	70,503	1,334
Female	417,240	6,207	26,740	29,328	26,803	170,609	18,649	49,966	3,912	85,026	-
40-44											
Total	578,390	21,378	49,949	45,947	18,865	183,746	34,692	93,793	40,015	89,606	401
Male	365,839	16,656	38,114	30,146	7,746	92,602	25,779	64,715	38,220	51,461	401
Female	212,551	4,722	11,835	15,801	11,119	91,143	8,913	29,078	1,795	38,145	-

Table 5.3 Cont'd

					Major Oc	cupational (Groups				
							Skilled		Plant and		
Age Group				Technicians			agricultural,	Craft and	machine		
and Sex	Total			and	Clerical	Service	forestry	related	operators,		
	Employed			associate	support	and sales	and fishery	trades	and	Elementary	
	Person	Managers	Professionals	professionals	workers	workers	workers	workers	assemblers	occupations	Others
45-49											
Total	421,698	14,120	38,576	35,733	11,648	132,626	33,518	61,913	24,193	67,643	1,726
Male	252,303	10,710	26,032	27,448	6,284	61,691	23,205	34,983	22,450	37,775	1,726
Female	169,394	3,410	12,545	8,285	5,364	70,935	10,313	26,931	1,743	29,868	-
50-54											
Total	288,013	11,098	24,937	29,761	9,312	78,119	24,146	44,186	14,799	51,517	136
Male	173,544	9,724	21,196	22,172	5,439	36,697	12,886	21,622	13,747	30,059	-
Female	114,470	1,374	3,741	7,589	3,873	41,422	11,260	22,564	1,052	21,458	136
55-59											
Total	210,275	5,187	14,593	19,225	7,705	61,134	22,404	33,048	10,344	36,133	503
Male	128,640	4,722	10,565	14,687	5,581	28,159	15,866	17,154	10,073	21,331	503
Female	81,635	465	4,029	4,539	2,124	32,975	6,538	15,894	271	14,801	-
60-64	,		,	•	,	,	•	ŕ		,	
Total	145,598	3,316	6,730	6,981	2,450	42,805	21,718	24,681	5,975	30,941	-
Male	90,332	2,793	6,730	5,431	1,731	21,274	15,566	13,407	5,644	17,757	-
Female	55,266		-	1,549	719	21,531	6,152	11,274		13,184	-
65+	,			•		,	•	ŕ		ŕ	
Total	194,851	4,841	5,963	8,482	3,154	60,105	37,971	29,215	6,519	38,602	-
Male	134,599		5,346	7,040	1,699	34,683	32,603	17,364		24,894	
Female	60,253	53	617	1,442	1,455	25,422	5,369	11,851	336	13,708	-

Table 5.4 Employed Population of Urban Areas Aged Ten Years and Above by Major Occupational Groups, Region and Sex: 2015

Table 5.4	Employed	Popu	ilation of	Urban Areas	Aged Ten Ye	ears and	Above by	Major Occ	upational	Groups, Re	egion and S	<u>ex: 201</u> 5
						Major Oc	cupational (
								Skilled		Plant and		
Region					Technicians			agricultural,	Craft and	machine		
and Sex					and .	Clerical	Service	forestry	related	operators,		
	Emplo	-	N4	Dantanainanala	associate	support	and sales	and fishery	trades	and	Elementary	041
	pers	ons	ivianagers	Professionals	professionals	workers	workers	workers	workers	assemblers	occupations	Otners
COUNTRY T	OTAL											
Total	7,083	,094	155,628	560,307	566,689	258,416	2,162,111	387,801	1,089,180	458,982	1,434,946	9,032
Male	3,963	,143	119,462	370,102	345,470	90,129	912,282	270,676	696,336	418,261	733,172	7,254
Female	3,119	,950	36,166	190,206	221,219	168,287	1,249,829	117,126	392,845	40,721	701,775	1,777
TIGRAY REG	SION											
Total	477	,813,	10,447	31,666	36,835	20,705	141,534	21,439	82,120	41,456	91,387	223
Male	253	,254	8,451	20,400	19,812	3,901	47,346	14,809	51,394	37,408	49,511	223
Female	224	,559	1,996	11,266	17,023	16,804	94,187	6,630	30,726	4,049	41,877	-
AFFAR REG	ION											
Total	131	,439	2,356	3,323	8,681	3,746	35,201	17,018	12,248	6,651	42,198	16
Male	75	,452	1,705	2,343	6,464	1,278	12,963	13,219	7,666	6,620	23,177	16
Female	55	,987	650	981	2,217	2,468	22,238	3,799	4,582	31	19,020	-
AMHARA RE	GION											
Total	1,449	,091	26,762	104,276	108,766	38,951	409,098	118,530	209,106	66,006	367,005	594
Male		,183	21,657	72,716	67,366	10,620	155,757	85,353	134,065	61,977	185,079	594
Female	653	,908	5,105	31,560	41,400	28,331	253,341	33,177	75,041	4,029	181,925	-
OROMIA RE												
Total	2,102	,417	46,792	146,128	146,441	58,151	697,173	99,870	357,711	133,386	413,083	3,682
Male	1,213		35,695	96,211	92,697	25,874	304,109	71,836	223,891	119,114	241,186	3,111
Female	888	,694	11,096	49,916	53,745	32,277	393,065	28,034	133,821	14,272	171,897	572
SOMALI REC	GION		·	•	·	,	,	·	,	,	·	
Total	119	,082	1,779	8,534	8,711	2,877	44,279	9,860	11,017	4,210	27,712	103
Male	67	,114	1,513	7,164	6,617	1,498	19,558	5,864	6,805	4,045	13,947	103
Female		,967	266	1,370	2,095	1,379	24,721	3,996	4,212	165	13,765	_
	UL -GUMUZ			,	•	,	,	,	,		•	
Total	107	,053	2,092	11,946	10,593	4,292	25,035	17,787	12,229	3,343	19,606	130
Male		,730	1,484	8,672	6,561	1,293	8,670	10,071	8,386	3,269	10,193	130
Female		,323	607	3,274	4,031	2,999	16,365	7,716	3,843	, 75	9,413	_
S.N.N.P. REC		,		-,	,	,	-,	, -	-,-		-,	
Total	1,096	,887	19,155	99,730	103,078	36,309	344,946	74,916	152,338	44,435	220,714	1,264
Male	•	,653	16,831	70,273	54,735	13,417	155,897	51,351	93,242	43,468	103,175	1,264
Female		,233	2,324	29,458	48,343	22,892	189,048	23,565	59,096	967	117,539	,
	100	,	_,=_	=5,100	.5,510	,		_0,000	55,500	501	,500	

Table 5.4 Cont'd

Table 3.4 Cont	<u></u>				Maior Oc	cupational (Groups				
							Skilled		Plant and		
Region				Technicians			agricultural,	Craft and	machine		
and Sex	Total			and	Clerical	Service	forestry	related	operators,		
	Employed			associate	support	and sales	and fishery	trades	and	Elementary	
	persons	Managers	Professionals	professionals	workers	workers	workers	workers	assemblers	occupations	Others
GAMBELLA REGIO	N										
Total	61,809	1,868	5,978	4,478	2,639	14,776	5,473	5,131	1,820	19,645	-
Male	31,771	1,203	4,431	2,629	1,488	7,009	3,814	2,847	1,678	6,671	-
Female	30,038	666	1,546	1,849	1,151	7,767	1,659	2,284	141	12,975	-
HARARI REGION											
Total	59,109	586	3,586	6,216	2,297	22,145	405	9,214	3,943	10,636	81
Male	29,900	417	2,417	4,148	396	8,631	215	5,422	3,862	4,312	81
Female	29,210	169	1,169	2,069	1,901	13,514	189	3,792	81	6,324	-
ADDIS ABABA CITY	ADMINIST	TRATION									
Total	1,362,069	41,656	139,267	124,358	84,806	384,671	18,616	221,334	143,792	200,723	2,847
Male	772,367	28,962	81,498	78,546	28,771	176,066	12,045	152,494	127,937	84,405	1,642
Female	589,703	12,694	57,769	45,811	56,035	208,604	6,571	68,839	15,856	116,318	1,206
DIRE DAWA ADMIN	IISTRATIO	N									
Total	116,325	2,135	5,873	8,533	3,644	43,254	3,888	16,732	9,940	22,237	90
Male	61,997	1,542	3,977	5,897	1,593	16,276	2,099	10,125	8,884	11,515	90
Female	54,327	592	1,897	2,636	2,051	26,978	1,789	6,608	1,056	10,722	-

Table 5.5 Employed Population of Major Towns Aged Ten Years and above by Major Occupational Groups and Sex: 2015

Table 5.5 Emp	ioyea Popu	iation of M	ajor Towns A	iged Ten Yea				ai Groups	and Sex: 20	15	
					Major O	ccupational Gr					
							Skilled		Plant and		
NA . T				Technicians			agricultural,	Craft and	machine		
Major Towns	Total			and	Clerical	Service	forestry and	related	operators,	-	
and Sex	Employed		D . (associate	support	and sales	fishery	trades	and	Elementary	0.1
00111170117011	Persons	Managers	Professionals	professionals	workers	workers	workers	workers	assemblers	occupations	Otners
COUNTRY TOTAL											
Total	2,560,286	66,833	222,101	212,650	129,367	734,939	71,215	401,403	236,478	481,225	4,076
Male .	1,443,652	48,414	137,876	134,333	42,751	316,128	48,526	279,329	212,940	220,532	2,822
Female	1,116,634	18,419	84,225	78,317	86,616	418,810	22,688	122,074	23,537	260,694	1,254
MEKELE TOWN											
Total	131,641	4,272	11,608	10,276	5,676	36,342	2,861	24,159	15,444	20,778	223
Male	70,849	2,911	6,560	6,510	1,032	10,921	1,609	16,772	15,217	9,093	223
Female	60,792	1,361	5,048	3,766	4,644	25,421	1,253	7,388	227	11,685	-
ASAYITA TOWN											
Total	12,819	63	685	1,189	456	3,471	667	1,516	649	4,108	16
Male	7,356	63	479	740	186	1,513	635	1,240	618	1,868	16
Female	5,463	-	206	449	270	1,959	32	276	31	2,240	-
GONDER TOWN											
Total	162,404	1,584	5,677	5,136	3,441	27,287	23,596	15,496	4,186	76,001	-
Male	85,621	1,155	3,707	2,777	1,173	9,040	16,453	11,109	4,186	36,022	-
Female	76,782	430	1,970	2,360	2,268	18,248	7,143	4,386	-	39,979	-
DESSIE TOWN											
Total	71,530	1,820	5,174	6,505	2,929	25,459	1,772	10,896	4,481	12,494	-
Male	43,161	1,581	4,367	4,246	917	11,786	1,216	7,465	4,358	7,226	-
Female	28,369	239	807	2,259	2,012	13,673	556	3,431	123	5,269	-
BAHIR DAR TOWN											
Total	124,310	2,249	7,441	7,863	4,471	25,804	7,808	15,928	7,476	45,270	-
Male	66,389	1,882	4,584	4,507	1,321	7,324	6,683	10,665	7,200	22,222	-
Female	57,921	367	2,857	3,355	3,150	18,480	1,125	5,263	276	23,048	-
BISHOFTU TOWN											
Total	57,994	1,096	4,375	4,138	1,823	15,680	2,527	12,619	6,463	9,216	58
Male	33,733	982	2,539	2,562	399	7,066	1,213	9,305	5,421	4,189	58
Female	24,261	114	1,836	1,576	1,424	8,615	1,313	3,314	1,042	5,027	-
ADAMA TOWN											
Total	117,828	1,939	6,772	9,556	4,836	35,337	3,736	19,267	15,517	20,501	366
Male	69,953	1,328	4,870	5,797	2,115	15,787	2,790	14,475	12,198	10,228	366
Female	47,874	611	1,903	3,759	2,722	19,549	946	4,793	3,319	10,273	-
JIMMA TOWN											
Total	75,289	1,829	5,988	5,819	3,585	24,395	778	12,918	4,301	15,677	-
Male	42,862	1,645	4,185	3,345	1,089	11,242	657	9,010	4,142	7,547	-
Female	32,428	184	1,803	2,474	2,496	13,153	121	3,908	159	8,130	-

Table 5.5 Cont'd

Table 3.3 Cont	<u>, </u>				Maior O	ccupational Gr	ouds				
				Technicians	ajo: o		Skilled agricultural,	Craft and	Plant and machine		
Major Towns	Total			and	Clerical	Service	forestry and	related	operators,		
and Sex	Employed			associate	support	and sales	fishery	trades	and	Elementary	
	Persons	Managers	Professionals	professionals	workers	workers	workers	workers	assemblers	occupations	Others
SHASHEMENE TO											
Total	54,374	612	2,674	3,758	2,117	22,243	1,531	8,211	4,989	8,110	130
Male	32,171	424	2,074	2,470	958	8,814	1,116	6,463	4,624	5,147	82
Female	22,203	188	600	1,288	1,159	13,428	414	1,748	366	2,962	48
JIJIGA TOWN											
Total	40,815	1,057	3,377	4,019	1,161	15,768	752	4,793	3,062	6,724	103
Male	25,947	791	2,628	3,027	561	7,459	471	3,625	2,897	4,386	103
Female	14,868	266	749	992	599	8,309	281	1,168	165	2,338	-
ASOSA TOWN											
Total	21,112	443	2,497	2,785	1,331	5,516	126	3,530	1,228	3,656	-
Male	12,786	414	2,052	1,678	359	2,303	126	2,714	1,154	1,987	-
Female	8,326	28	445	1,108	973	3,213	-	816	75	1,669	-
HAWASSA TOWN	101700	4.404	40.000	0.040	5 0 4 0		4 007	00.400	0.004	10 700	400
Total	124,702	4,164	13,888	8,648	5,242	39,983	1,837	22,166	9,884	18,728	162
Male	73,264	3,268	9,347	5,677	1,220	18,071	949	17,157	9,218	8,196	162
Female	51,439	896	4,541	2,971	4,022	21,912	888	5,009	666	10,533	-
GAMBELLA TOWN		4.000	0.040	0.050	4.550	7.505	045	0.004	4 404	0.007	
Total	27,965	1,330	3,218	3,852	1,553	7,585	315	2,624	1,121	6,367	-
Male	15,295	1,050	2,593	2,408	661	3,829	249	1,289	1,025	2,190	-
Female	12,670	280	624	1,444	892	3,756	66	1,335	97	4,177	-
HARAR TOWN	FO 400	F0C	0.500	0.040	0.007	00.445	405	0.044	2.042	10.000	04
Total	59,109	586	3,586	6,216	2,297	22,145	405	9,214	3,943	10,636	81
Male	29,900	417	2,417	4,148	396	8,631	215	5,422	3,862	4,312	81
Female ADDIS ABABA CIT	29,210	169	1,169	2,069	1,901	13,514	189	3,792	81	6,324	-
Total	1,362,069	44 CEC	120.267	124,358	04.006	204 674	18,616	224 224	143,792	200,723	2,847
Male		41,656 28,962	139,267	•	84,806	384,671 176,066	12,045	221,334	,	84,405	
iviale Female	772,367 589,703	•	81,498 57,760	78,546 45,811	28,771	208,604	•	152,494	127,937	•	1,642
DIRE DAWA TOWN		12,694	57,769	40,011	56,035	200,004	6,571	68,839	15,856	116,318	1,206
Total	116,325	2,135	5,873	8,533	3,644	43,254	3,888	16,732	9,940	22,237	90
Male	61,997	2,135 1,542	3,977	6,533 5,897	3,6 44 1,593	43,25 4 16,276	3,000 2,099	10,732	9,940 8,884	22,23 <i>1</i> 11,515	90
	•	1,542 592	,	•		•		•	•		
Female	54,327	၁ 92	1,897	2,636	2,051	26,978	1,789	6,608	1,056	10,722	-

Table 5.6 Employed Population of Urban Areas Aged Ten Years and Above by Minor Occupational Groups, Sex and Region: 2015

Table 5.6 Emplo	yea Popula	ation of U	rban Area	as Aged Te	en Years ar	na Above	by Minor Oc	cupational	Groups, S	ex and	Region: 20	15
Minor						Re	egion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
TOTALEMPLOYE	D POPUL	ATION										
Total	7,083,094	477,813	131,439	1,449,091	2,102,417	119,082	107,053	1,096,887	61,809	59,109	1,362,069	116,325
Male	3,963,143	253,254	75,452	795,183	1,213,723	67,114	58,730	603,653	31,771	29,900	772,367	61,997
Female	3,119,950	224,559	55,987	653,908	888,694	51,967	48,323	493,233	30,038	29,210	589,703	54,327
MAJOR GROUP	1: MANAC	SERS										
Total	153,505	10,447	2,356	26,762	45,151	1,779	2,092	19,155	1,868	586	41,174	2,135
Male	117,550	8,451	1,705	21,657	34,129	1,513	1,484	16,831	1,203	417	28,616	1,542
Female	35,956	1,996	650	5,105	11,022	266	607	2,324	666	169	12,558	592
CHIEF EXECU												
Total	19,668	1,313	357	4,539	4,432	494	441	4,421	209	98	3,291	73
Male	17,378	609	357	4,539	3,121	454	441	4,260	209	98	3,291	-
Female	2,289	704	-	-	1,311	40	-	161	-	-	-	73
Legislators and												
Total	11,299	1,046	357	1,783	3,168	454	185	3,116	116	49	953	73
Male	9,010	342	357	1,783	1,858	414	185	2,955	116	49	953	-
Female	2,289	704	-	-	1,311	40	-	161	-	-	-	73
Managing dire			ıtives									
Total	8,368	267	-	2,756	1,264	40	256	1,305	93	49	2,338	-
Male	8,368	267	-	2,756	1,264	40	256	1,305	93	49	2,338	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
ADMINISTRAT	IVE AND C	OMMERC	IAL MAN	AGERS								
Total	42,054	2,396	-	4,667	11,528	482	471	5,146	769	81	15,447	1,065
Male	32,870	1,988	-	4,173	8,431	482	343	4,175	626	41	11,901	711
Female	9,184	408	-	495	3,096	-	128	971	143	41	3,546	355
Business serv			on manag									
Total	33,809	2,306	-	4,157	10,411	442	342	3,984	647	41	10,507	974
Male	28,302	1,898	-	3,780	7,695	442	214	3,494	504	-	9,567	711
Female	5,507	408	-	377	2,716	-	128	490	143	41	941	263
Sales, marketi	ing and dev	elopment	managers									
Total	8,245	90	-	510	1,117	40	129	1,163	122	41	4,940	92
Male	4,568	90	-	393	737	40	129	681	122	41	2,334	-
Female	3,677	-	-	118	380	-	-	481	-	-	2,606	92
PRODUCTION	AND SPEC	CIALIZED	SERVICE	S MANAG	ERS							
Total	44,510	3,455	436	9,210	13,557	411	410	6,537	714		8,795	749
Male	35,246	3,455	218	6,966	11,230	411	410	5,559	192	147	6,002	657
Female	9,264	-	219	2,244	2,327	-	-	978	523	88	2,794	92

99

Table 5.6 Cont'd

Minor						Re	gion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie		SNNPR	Gambela	Harari	Ababa	Dawa
Production mai						Comanc	Odmaz	ONNIN	Cambeia	Haran	Noaba	Dawa
Total	12,261	9,100,1010, 1,472	- 10103tiy t	4,378	3,839	_	_	1,363	439	49	628	93
Male	10,762	1,472		3,265	3,839	_	_	1,363		49	628	93
Female	1,500	1,412	_	1,114	5,059	_	_	1,303	386	-	020	93
Manufacturing,	,	- netruction	- and dict	,	-	_	_	_	300	_	_	_
Total	11,186	1,354	, and dist	634	2,638	123	134	999	74		5,013	
Male	8,763	1,354	218	572	1,856	123	134	397	53	-	4,059	-
Female	2,423	1,334	210	62	782	123	134	603	22	-	954	-
		- nationa ta	- Vanalaavu			-	-	603	22	-	954	-
Information and		cations tec	chhology	service mar								
Total	521	-	-	-	521	-	-	-	-	-	-	-
Male	521	-	-	-	521	-	-	-	-	-	-	-
Female	- :	-	-	-	-	-	-	-	-	-	-	-
Professional se			040	4.407	0.550	000	077	4 4 7 4	004	400	0.454	0.57
Total	20,541	629	219	4,197	6,559	288	277	4,174	201	186	3,154	657
Male .	15,200	629	-	3,129	5,014	288	277	3,799	86	98	1,315	565
Female	5,342	<u>-</u>	219	1,069	1,545	-	-	375	115	88	1,839	92
HOSPITALITY,												
Total	47,274	3,282	1,562	8,346	15,634	391	769	3,051	177	172	13,641	247
Male	32,055	2,398	1,131	5,980	11,346	166	290	2,838	177	132	7,424	175
Female	15,219	884	431	2,366	4,288	225	479	214	-	41	6,217	73
Hotel and resta	aurant mana											
Total	19,553	493	218	3,025	8,078	156	555	1,720	62	-	5,174	73
Male	11,845	493	-	1,823	5,588	52	219	1,506	62	-	2,030	73
Female	7,708	-	218	1,202	2,490	104	336	214	-	-	3,145	=
Retail and who	lesale trade	e manage	rs									
Total	11,285	1,134	-	1,719	2,396	81	29	-	53	81	5,690	102
Male	7,226	612	-	1,274	1,827	-	29	-	53	41	3,290	102
Female	4,059	523	-	445	570	81	-	-	-	41	2,400	-
Other services	managers											
Total	16,435	1,655	1,345	3,602	5,160	154	185	1,332	62	91	2,777	73
Male	12,984	1,294	1,131	2,883	3,932	114	41	1,332	62	91	2,104	_
Female	3,451	361	214	719	1,228	40	143	-	-	-	672	73

Table 5.6 Cont'd

Minor						Re	gion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
MAJOR GROUP 2	: PROFESS		'	•		•			•	•		
Total	558,734	31,666	3,323	104,276	146,071	8,534	11,946	99,730	5,835	3,586	137,893	5,873
Male	369,274	20,400	2,343	72,716	96,211	7,164	8,672	70,273	4,289	2,417	80,813	3,977
Female	189,460	11,266	981	31,560	49,859	1,370	3,274	29,458	1,546	1,169	57,080	1,897
SCIENCE AND	ENGINEER	RING PRO)FESSION	NALS								
Total	60,931	1,519	405	8,698	13,312	1,389	1,081	7,709	680	257	25,127	754
Male	50,126	1,385	266	6,874	11,743	1,175	777	6,147	571	140	20,384	665
Female	10,805	133	140	1,824	1,569	214	304	1,562	109	117	4,744	89
Physical and e	earth science	e professi	onals									
Total	4,211	-	224	596	1,202	40	49	-	162	-	1,697	240
Male	3,194	-	224	-	1,168	=	49	-	162	-	1,351	240
Female	1,017	-	-	596	34	40	-	-	-	-	346	-
Mathematiciar	ns, actuaries	and statis	sticians									
Total	3,010	-	31	-	1,048	82	49	241	22	-	1,536	-
Male	2,083	-	16	-	1,048	82	49	241	22	-	625	-
Female	928	-	16	-	-	-	-	-	-	-	912	-
Life science pi	rofessionals											
Total	16,958	489	134	4,861	3,987	912	706	2,481	140	49	3,110	89
Male	13,507	489	26	4,172	2,766	912	402	1,879	53	49	2,760	-
Female	3,451	-	108	690	1,221	-	304	602	88	-	350	89
Engineering p	rofessionals	(excludin	g electro	technology)								
Total	24,466	580	16	2,550	5,104	304	237	2,836	151	91	12,171	425
Male	21,879	580	-	2,012	5,104	130	237	2,836	129	91	10,333	425
Female	2,587	-	16	538	-	174	-	-	22	-	1,838	-
Electro techno	logy engine	ers										
Total	5,950	226	-	186	1,391	-	-	1,150	-	-	2,998	-
Male	3,721	226	-	186	1,077	-	-	189	-	-	2,043	-
Female	2,229	-	-	-	314	-	-	961	-	-	954	-
Architects, pla	nners, surve	eyors and	designers	3								
Total	6,336	224	-	503	580	51	39	1,001	205	117	3,615	-
Male	5,741	90	-	503	580	51	39	1,001	205	-	3,272	-
Female	594	133	-	-	-	-	-	-	-	117	344	-

Table 5.6 Cont'd

Table 5.6 Cont o	ı .											
Minor						Re	gion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
HEALTH PROFES		<u> </u>					'			l. U		
Total	65,075	4,475	742	10,174	21,091	820	1,268	12,006	260	472	12,911	855
Male	31,699	2,448	234	5,993	10,026	780	693	6,584	239	350	3,692	661
Female	33,375	2,027	508	4,181	11,065	40	575	5,422	22	122	9,219	194
Medical docto		,		•	,			•			,	
Total	6,488	616	16	893	1,603	212	29	1,889	-	-	1,026	204
Male	4,483	361	-	832	1,079	212	-	1,794	-	-	· -	204
Female	2,005	255	16	60	523	-	29	95	-	-	1,026	-
Nursing and m	nidwifery pro	ofessional	S								,	
Total	26,529	2,194	126	4,289	6,870	269	132	3,608	105	220	8,521	194
Male	7,749	708	-	2,550	1,285	269	-	1,240	83	220	1,301	92
Female	18,779	1,486	126	1,740	5,585	-	132	2,368	22	-	7,220	102
Traditional and	d compleme	ntary med	dicine prof	essionals	•			•				
Total	616	· -	34	-	523	-	-	59	-	-	-	-
Male	523	-	-	-	523	-	-	-	-	-	-	-
Female	93	-	34	-	-	-	-	59	-	-	-	-
Paramedical p	ractitioners											
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Veterinarians												
Total	5,623	-	334	-	3,685	40	77	1,445	40	-	-	-
Male	4,644	-	110	-	3,628	40	-	825	40	-	-	-
Female	978	-	224	-	57	-	77	620	-	-	-	-
Other health p	rofessionals	3										
Total	25,819	1,665	232	4,992	8,410	298	1,030	5,004	115	252	3,364	457
Male	14,299	1,378	124	2,612	3,510	258	693	2,725	115	130	2,390	365
Female	11,519	287	108	2,380	4,900	40	337	2,279	-	122	974	92
TEACHING PR	OFESSION	ALS										
Total	197,039	14,132	1,159	42,659	49,206	3,009	4,636	41,021	3,259	1,655	34,190	2,113
Male	127,976	8,366	1,104	29,086	30,957	2,238	3,439	27,564	2,134	1,152	20,725	1,211
Female	69,063	5,766	55	13,572	18,249	771	1,197	13,457	1,125	503	13,465	902
University and	higher edu	cation tea	chers									
Total	30,609	2,945	203	4,790	7,629	212	795	7,066	124	346	6,314	185
Male	25,936	1,821	180	4,125	7,392	172	795	5,909	124	297	4,936	185
Female	4,672	1,123	24	665	237	40	-	1,156	-	49	1,378	-

Table 5.6 Cont'd

Minor						Re	gion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Vocational edu	cation teac											
Total	10,106	1,329	152	2,271	2,089	194	57	3,241	22	41	625	86
Male	6,935	964	152	1,377	1,935	154	29	2,263	22	41	-	-
Female	3,170	365	-	894	154	40	28	978	-	-	625	86
Secondary edu	ucation tead	hers										
Total	77,424	6,255	454	17,335	16,730	1,320	1,995	15,291	2,643	685	14,068	648
Male	55,774	4,475	454	14,567	11,007	823	1,548	11,549	1,854	555	8,471	469
Female	21,650	1,779	-	2,768	5,723	497	447	3,742	789	130	5,598	179
Primary school	I and early o		teachers									
Total	63,552	3,207	66	13,784	19,835	1,022	1,459	11,807	471	584	10,422	897
Male	28,703	799	34	6,737	7,761	828	738	5,502	134	259	5,651	260
Female	34,849	2,408	32	7,047	12,074	193	722	6,306	336	325	4,770	637
Other teaching	profession	als										
Total	15,348	396	284	4,479	2,923	261	330	3,616	-	-	2,761	297
Male	10,626	306	284	2,279	2,861	261	330	2,341	-	-	1,666	297
Female	4,721	90	-	2,200	62	-	-	1,275	-	-	1,095	-
BUSINESS AND	D ADMINIS	TRATION	PROFES	SIONALS								
Total	146,618	8,257	694	29,406	37,349	1,747	3,095	21,744	1,134	596	41,382	1,215
Male	91,780	5,487	426	20,244	24,033	1,615	2,507	16,241	896	357	19,379	595
Female	54,839	2,770	268	9,162	13,316	132	588	5,503	238	239	22,003	620
Finance profes	ssionals											
Total	80,333	5,003	384	12,261	20,487	1,054	1,221	9,305	767	387	28,690	774
Male	45,452	3,350	160	7,002	11,940	963	874	6,870	611	148	13,042	493
Female	34,881	1,654	224	5,259	8,546	91	348	2,435	156	239	15,648	281
Administration	professiona	als										
Total	53,226	2,668	310	14,204	14,504	692	1,494	9,796	245	50	8,907	355
Male	38,579	1,642	266	11,607	10,731	652	1,254	7,646	245	50	4,385	102
Female	14,647	1,026	44	2,598	3,773	41	240	2,150	-	-	4,522	253
Sales, marketii	ng and publ	ic relation	s profess	ionals								
Total	13,059	585	-	2,941	2,358	-	379	2,643	122	159	3,785	86
Male	7,749	495	-	1,636	1,362	-	379	1,725	41	159	1,952	-
Female	5,311	90	-	1,305	996	-	-	918	82	-	1,833	86
INFORMATION	AND COM		TIONS TE	CHNOLOG	Y PROFES	SSIONALS	3					
Total	9,963	363	-	1,211	2,468	80	28	966	62	-	4,784	-
Male	6,922	363	-	1,211	1,971	80	28	304	62	-	2,902	-
Female	3,040				497	_		662			1,882	

Table 5.6 Cont'd

Minor						Re	gion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Software and a		develope	rs and an									
Total	3,530	-	-	237	122	-	-	603	-	-	2,568	-
Male	1,719	-	-	237	122	-	-	-	-	-	1,360	-
Female	1,811	-	-	-	-	-	-	603	-	-	1,208	-
Database and	network pro	ofessionals	S									
Total	6,433	363	-	974	2,345	80	28	363	62	-	2,216	-
Male	5,203	363	-	974	1,848	80	28	304	62	-	1,543	-
Female	1,229	-	-	-	497	-	-	59	-	-	673	-
LEGAL, SOCIA	L AND CUL	TURAL P	ROFESS	IONALS								
Total	76,214	2,921	323	11,327	21,977	1,489	1,810	15,888	439	606	18,831	602
Male	58,008	2,351	314	8,609	16,814	1,277	1,228	13,037	386	418	13,065	509
Female	18,206	570	10	2,718	5,163	212	583	2,852	53	188	5,767	92
Legal profession	onals											
Total	34,396	1,163	235	6,183	10,673	1,327	736	7,175	182	140	6,389	194
Male	29,277	1,163	235	4,874	9,494	1,114	323	6,173	182	90	5,434	194
Female	5,119	-	-	1,309	1,179	212	413	1,001	-	50	955	-
Librarians, arc	hivists and	curators										
Total	11,519	647	57	1,987	4,675	-	349	1,772		231	1,637	90
Male	5,791	392	47	1,306	1,589	-	256	934	74	140	963	90
Female	5,728	256	10	681	3,087	-	92	838	-	91	674	-
Social and relig		ssionals										
Total	21,890	-	32	2,259	5,360	82	726	4,214	101	99	8,863	153
Male	16,207	-	32	1,666	4,523	82	648	3,297	49	99	5,657	153
Female	5,683	-	-	594	836	-	77	917	53	-	3,206	-
Authors, journa												
Total	5,078	839	-	644	123	-	-	2,204		136	927	165
Male	3,808	525	-	577	62	-	-	2,109	40	89	333	73
Female	1,270	314	-	67	61	-	-	95	-	48	593	92
Creative and p	erforming a	ırtists										
Total	3,331	271	-	253	1,147	80	-	524	41	-	1,015	-
Male	2,926	271	-	186	1,147	80	-	524	41	-	678	-
Female	405	-	-	67	-	-	-	-	-	-	338	-
MAJOR GROUP	3: TECHN					IALS						
Total	564,911	36,473	8,681	108,515	145,797	8,711	10,593	102,957	4,426	6,216	124,011	8,533
Male	344,949	19,812	6,464	67,366	92,176	6,617	6,561	54,735	2,629	4,148	78,546	5,897
Female	219,962	16,661	2,217	41,149	53,622	2,095	4,031	48,222	1,796	2,069	45,464	2,636

Table 5.6 Cont'd

Minor	-					Re	gion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
SCIENCE AND						I				l l		
Total	66,794	7,139	705	9,116	13,154	361	1,113	10,775	686	269	21,956	1,520
Male	53,570	5,528	595	8,215	9,740	310		8,397	382	227	18,368	1,136
Female	13,224	1,612	110	901	3,414	51	441	2,378	304	42	3,588	384
Physical and e	engineering	science te	echnicians	6	•			,			,	
Total	32,790	3,035	119	2,378	7,293	300	459	2,999	40	48	15,588	532
Male	27,958	2,945	10	2,125	5,635	249	231	2,601	40	48	13,629	445
Female	4,832	90	110	253	1,658	51	228	397	-	-	1,958	86
Mining, manuf	acturing and	d construc	tion supe	rvisors								
Total	7,383	2,079	-	1,273	469	-	-	667	-	138	2,280	478
Male	5,481	1,942	-	869	61	-	-	667	-	138	1,326	478
Female	1,902	137	-	404	408	-	-	-	-	-	954	-
Process contro	ol technician	าร										
Total	8,206	271	231	2,452	1,885	-	-	1,178	-	83	1,722	384
Male	5,730	-	231	2,391	992	-	-	483	-	41	1,380	213
Female	2,477	271	-	61	894	-	-	696	-	42	341	171
Life science te	chnicians a	nd related	d associate	e professior	nals (exclud	ling medic	al)					
Total	16,774	1,349	355	2,920	3,506	61	654	5,931	645	-	1,352	-
Male	13,114	368	355	2,830	3,052	61	441	4,646		-	1,018	-
Female	3,660	980	-	91	454	-	213	1,285	304	-	334	-
Ship and aircra	aft controlle	rs and tec	hnicians									
Total	1,640	406	-	93	-	-	-	-	-	-	1,015	126
Male	1,287	273	-	-	-	-	-	-	-	-	1,015	-
Female	353	133	-	93	-	-	-	-	-	-	-	126
HEALTH ASSC	CIATE PRO	DFESSIO	NALS									
Total	82,846	5,285	709	13,789	21,872	2,124	3,707	13,519	868	1,813	18,072	1,089
Male	37,080	1,291	583	8,051	7,921	1,189	2,251	6,323	511	1,107	7,508	344
Female	45,766	3,994	127	5,737	13,950	935	1,456	7,196	356	706	10,564	744
Medical and p	harmaceutio	cal technic	cians									
Total	16,021	983	152	3,848	3,558	144	377	2,533		171	4,001	92
Male	7,352	361	41	1,417	1,148	144	154	1,433		171	2,308	92
Female	8,669	621	111	2,431	2,410	-	223	1,100	82	-	1,693	-
Nursing and m	nidwifery ass	sociate pro	ofessional	ls								
Total	36,317	3,096	165	5,752	9,012	1,355	1,628	6,062		1,451	6,904	710
Male	15,401	346	149	4,528	2,999	503	847	2,740		745	2,190	253
Female	20,916	2,750	16	1,224	6,013	853	781	3,322	81	706	4,714	457

106

Table 5.6 Cont'd

Occupational Groups and Sex Total Tigray Affar Amhara Oromia Somalie Benishangul -Gumuz SNNPR Gambela Harari Addis Ababa Traditional and complementary medicine associate professionals Total 1,707 191 - 863 521 - 132 - - - - Male 1,647 191 - 802 521 - 132 - - - - Female 60 - - 60 - </th <th>-</th>	-
Traditional and complementary medicine associate professionals Total 1,707 191 - 863 521 - 132 - - - Male 1,647 191 - 802 521 - 132 - - - Female 60 - - 60 -	
Total 1,707 191 - 863 521 - 132 - - - Male 1,647 191 - 802 521 - 132 - - - Female 60 - - 60 - - - - - - Veterinary technicians and assistants Total 5,286 393 140 92 371 330 760 1,202 278 - 1,720 Male 3,990 393 140 92 315 330 516 600 226 - 1,370	-
Male 1,647 191 - 802 521 - 132 - - - Female 60 - - 60 -<	-
Female 60 - </td <td>-</td>	-
Veterinary technicians and assistants Total 5,286 393 140 92 371 330 760 1,202 278 - 1,720 Male 3,990 393 140 92 315 330 516 600 226 - 1,370	-
Total 5,286 393 140 92 371 330 760 1,202 278 - 1,720 Male 3,990 393 140 92 315 330 516 600 226 - 1,376	-
Total 5,286 393 140 92 371 330 760 1,202 278 - 1,720 Male 3,990 393 140 92 315 330 516 600 226 - 1,376	-
Female 1,297 56 - 244 602 53 - 343	
Other health associate professionals	
Total 23,515 622 253 3,233 8,410 295 811 3,722 244 191 5,44	287
Male 8,691 - 253 1,212 2,938 212 602 1,549 103 191 1,633	-
Female 14,824 622 - 2,022 5,472 82 209 2,173 141 - 3,819	287
BUSINESS AND ADMINISTRATION ASSOCIATE PROFESSIONALS	
Total 180,256 10,184 3,424 35,821 47,960 3,509 2,613 30,594 1,173 1,950 39,76	3,262
Male 116,273 5,289 2,622 24,150 32,767 2,864 1,655 16,278 620 1,568 26,12	2,335
Female 63,983 4,896 802 11,671 15,194 645 959 14,316 553 383 13,64	926
Financial and mathematical associate professionals	
Total 52,960 3,773 925 8,301 16,017 577 692 11,887 338 191 8,933	1,326
Male 28,914 918 807 5,208 9,583 526 389 6,288 142 191 4,14	717
Female 24,045 2,855 118 3,093 6,434 51 303 5,598 196 - 4,78	610
Sales and purchasing agents and brokers	
Total 37,103 2,630 549 3,597 7,681 1,658 499 5,474 230 523 13,560	701
Male 28,544 2,108 331 2,254 6,922 1,216 378 3,234 115 413 10,96	613
Female 8,558 522 219 1,343 759 442 121 2,239 115 110 2,600	89
Business services agents	
Total 7,990 495 - 2,028 1,878 - 29 2,446 1,029	86
Male 4,576 1,294 1,844 1,009 343	86
Female 3,414 495 - 734 34 - 29 1,436 686	-
Administrative and specialized secretaries	
Total 24,361 1,819 676 3,363 8,161 387 120 3,025 139 91 6,18	394
Male 14,148 1,552 539 1,551 3,535 337 120 2,625 53 - 3,54	292
Female 10,213 267 138 1,812 4,626 51 - 400 87 91 2,640	102

Table 5.6 Cont'd

Minor	-					Re	gion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Regulatory governi	ment associ		ssionals			•	1					
Total	57,843	1,467	1,274	18,532	14,224	886	1,273	7,763	465	1,145	10,060	754
Male	40,091	711	945	13,844	10,882	785	767	3,121	310	963	7,135	628
Female	17,752	756	328	4,688	3,342	101	506	4,642	156	182	2,925	126
LEGAL, SOCIAL, CULTURAL AND RELATED ASSOCIATE PROFESSIONALS												
Total	73,073	3,824	326	11,646	16,223	446	1,714	16,212	987	344	20,079	1,272
Male	48,880	2,346	279	7,380	12,893		1,226	7,590	578	234	14,954	1,005
Female	24,193	1,478	47	4,266	3,330	51	488	8,622	409	110	5,125	267
Legal, social a												
Total	38,578	1,862	251	7,783	8,207	314	1,074	8,181	493	297	9,672	445
Male	30,317	1,153	220	5,880	6,833	314	815	5,538	363	186	8,662	353
Female	8,261	709	32	1,903	1,373	-	259	2,643	130	110	1,010	92
Sports and fitn												
Total	12,764	412	-	1,777	4,094	-	43	1,648	174	-	4,457	159
Male	9,178	259	-	1,016	4,011	-	43	651	174	-	2,951	73
Female	3,586	153	-	760	83	-	-	997	-	-	1,506	86
Artistic, cultura												
Total	21,731	1,550	75	2,086	3,922	132	597	6,382	320	48	5,951	668
Male	9,385	935	59	484	2,049	81	368	1,401	40	48	3,341	579
Female	12,347	616	16	1,603	1,874	51	229	4,981	279	-	2,610	88
Information and												
Total	16,731	905	227	3,202	2,473	52	99	3,657	122	344	5,577	73
Male .	9,598	905	10	2,335	1,382	52	29	1,743	82	196	2,791	73
Female	7,134	<u>-</u>	218	866	1,091	-	70	1,914	41	147	2,786	-
Information an												
Total	12,314	905	227	2,843	1,589	52	99	3,057	41	296	3,132	73
Male	7,108	905	10	1,977	1,008		29	1,143	-	196	1,716	73
Female	5,206		218	866	582	-	70	1,914	41	99	1,416	-
Telecommunic		broadcast	ing techni									
Total	4,417	-	-	359	884	-	-	600	82	48	2,445	-
Male	2,489	-	-	359	375	-	-	600	82	-	1,075	-
Female	1,928	-	-	-	509	-	-	-	-	48	1,370	-
TEACHING AS												
Total	126,161	8,057	2,841	28,939	39,322	1,673	1,346	25,910	590	1,496	14,669	1,319
Male .	63,440	3,465	1,927	12,282	22,680	1,300	728	12,536	457	816	6,245	1,004
Female	62,721	4,591	914	16,657	16,642	372	617	13,374	134	680	8,424	315

Table 5.6 Cont'd

Minor	-					Re	gion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Teaching asso	ciate profes	sionals	'	•		•			•		1	
Total	126,161	8,057	2,841	28,939	39,322	1,673	1,346	25,910	590	1,496	14,669	1,319
Male	63,440	3,465	1,927	12,282	22,680	1,300	728	12,536	457	816	6,245	1,004
Female	62,721	4,591	914	16,657	16,642	372	617	13,374	134	680	8,424	315
SPECIAL EDUC	CATION TEA	ACHING A	SSOCIA	TE PROFES	SSIONALS							
Total	19,050	1,078	449	6,003	4,793	548	-	2,290	-	-	3,890	-
Male	16,109	988	449	4,952	4,793	508	-	1,868	-	-	2,553	-
Female	2,941	90	-	1,050	-	40	-	422	-	-	1,337	-
Special educa	tion teaching	g associat	te profess	ionals								
Total	19,050	1,078	449	6,003	4,793	548	-	2,290	-	-	3,890	-
Male	16,109	988	449	4,952	4,793	508	-	1,868	-	-	2,553	-
Female	2,941	90	-	1,050	-	40	-	422	-	-	1,337	-
MAJOR GROUP	4 : CLERIC	CAL SUPI	PORT WO	ORKERS								
Total	258,416	20,705	3,746	38,951	58,151	2,877	4,292	36,309	2,639	2,297	84,806	3,644
Male	90,129	3,901	1,278	10,620	25,874	1,498	1,293	13,417	1,488	396	28,771	1,593
Female	168,287	16,804	2,468	28,331	32,277	1,379	2,999	22,892	1,151	1,901	56,035	2,051
GENERAL AND) KEYBOAR	D CLERK										
Total	88,417	6,321	1,463	14,146	21,005	1,679	2,011	13,399		974	25,609	1,242
Male	12,058	218	241	1,151	2,787	657	441	1,633		110	4,343	348
Female	76,359	6,103	1,222	12,996	18,218	1,022	1,570	11,767	438	864	21,266	894
General office	clerks											
Total	3,765	218	369	255	706	803	36	366	22	40	949	-
Male	1,711	218	23	154	122	544	36	-	-	-	615	-
Female	2,054	-	347	101	584	260	-	366	22	40	334	-
Secretaries (g	eneral)											
Total	36,733	2,167	360	7,093	9,352	616	1,022	3,039	175	614	11,856	440
Male	2,880	-	-	195	1,246	-	78	-	-	-	1,013	348
Female	33,853	2,167	360	6,898	8,106	616	943	3,039	175	614	10,843	92
Keyboard ope	rators											
Total	47,919	3,936	735	6,799	10,947	260	954	9,994	369	320	12,804	803
Male	7,467	-	219	803	1,419	114	327	1,633	128	110	2,715	-
Female	40,452	3,936	516	5,996	9,528	147	626	8,362	241	210	10,089	803
CUSTOMER SI												
Total	90,523	5,385	741	14,100	21,165	647	1,240	13,429		532	31,471	888
Male	44,720	912	306	5,232	14,203	504	728	7,020		140	14,690	277
Female	45,803	4,473	435	8,869	6,962	143	512	6,409	215	393	16,781	611

Table 5.6 Cont'd

Minor	•					D-	gion					
Minor				1		Ke	gion			ı	A .1.1' - 1	D'
Occupational		- .	A				Benishangul	011125			Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Tellers, money				_								
Total	60,353	4,072	468	9,972	11,478	557	325	7,558	650	492	24,270	510
Male	30,092	640	251	4,255	8,405	414	116	3,777	479	99	11,471	185
Female	30,260	3,433	218	5,717	3,073	143	209	3,781	171	393	12,799	326
Client informat												
Total	30,170	1,313	273	4,128	9,687	90	915	5,870	274	41	7,201	378
Male	14,628	273	55	977	5,798	90	612	3,242	230	41	3,219	92
Female	15,542	1,041	218	3,152	3,889	-	303	2,628	44	-	3,982	286
NUMERICAL A												
Total	37,143	5,019	812	4,459	7,955	337	331	4,557	694	141	11,899	939
Male	20,641	1,960	484	2,358	5,928	337	36	3,215	459	49	5,207	607
Female	16,503	3,059	327	2,101	2,027	-	295	1,342	234	92	6,692	332
Numerical cler												
Total	21,073	1,622	546	1,262	3,411	205	215	3,404	600	92	9,279	436
Male	11,845	586	219	1,048	2,546	205	36	2,572	415	-	3,962	258
Female	9,227	1,036	327	214	865	-	180	831	186	92	5,317	179
Material-record	ding and tra	nsport cle	rks									
Total	16,071	3,397	266	3,197	4,544	132	116	1,153	94	49	2,620	503
Male	8,796	1,374	266	1,311	3,382	132	-	642	45	49	1,245	349
Female	7,275	2,023	-	1,887	1,162	-	116	511	49	-	1,375	153
OTHER CLERIC	CAL SUPPO	ORT WOR	KERS									
Total	42,333	3,979	730	6,244	8,026	214	710	4,924	455	649	15,827	575
Male	12,710	810	247	1,879	2,956	-	87	1,550	191	97	4,531	362
Female	29,623	3,169	484	4,366	5,070	214	623	3,374	263	552	11,297	213
Other clerical	support wor	kers										
Total	42,333	3,979	730	6,244	8,026	214	710	4,924	455	649	15,827	575
Male	12,710	810	247	1,879	2,956	-	87	1,550	191	97	4,531	362
Female	29,623	3,169	484	4,366	5,070	214	623	3,374	263	552	11,297	213
MAJOR GROUP 5			LES WOF		•			•				
Total	2,162,111	141,534	35,201	409,098	697,173	44,279	25,035	344,946	14,776	22,145	384,671	43,254
Male	912,282	47,346	12,963	155,757	304,109	19,558	8,670	155,897	7,009	8,631	176,066	16,276
Female	1,249,829	94,187	22,238	253,341	393,065	24,721	16,365	189,048		13,514	208,604	26,978
PERSONAL SE			•	•	,	•	•	,	•	,	•	,
Total	507,355	44,139	6,654	81,064	147,500	6,998	6,402	78,030	4,345	3,301	119,109	9,813
Male	145,730	6,501	2,652	22,087	55,004	1,790	1,936	27,454	1,779	598	24,781	1,148
Female	361,625	37,638	4,003	58,977	92,496	5,208	4,465	50,576	2,566	2,703	94,328	8,665

Table 5.6 Cont'd

Minor		-				Re	gion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Travel attenda			juides									
Total	11,496	153	-	1,965	4,057	287	238	2,675	196	90	1,473	363
Male	8,702	-	-	1,388	3,334	114	238	1,599	196	90	1,473	270
Female	2,793	153	-	577	723	173	-	1,076	-	-	-	92
Cooks												
Total	79,207	4,406	1,442	8,969	26,082	1,919	1,120	15,347	260	505	18,822	335
Male	15,147	90	149	431	6,989	596	-	3,441	46	-	3,404	-
Female	64,061	4,316	1,293	8,537	19,093	1,323	1,120	11,906	214	505	15,418	335
Waiters and ba	artenders											
Total	130,296	9,107	1,992	29,572	32,502	507	3,448	22,871	3,237	197	25,068	1,793
Male	47,955	1,222	1,018	6,195	17,653	40	1,287	11,213	1,448	49	7,308	522
Female	82,340	7,885	974	23,377	14,850	466	2,161	11,659	1,789	148	17,759	1,271
Hairdressers,	beauticians	and relate	ed workers	S								
Total	94,714	9,751	981	13,873	31,469	743	278	13,653	183	563	22,514	704
Male	39,209	2,313	398	7,608	14,486	611	249	6,052	46	191	7,094	161
Female	55,504	7,438	584	6,265	16,983	132	29	7,600	137	372	15,420	543
Building and h	ousekeepin	g supervis	sors									
Total	145,229	15,046	1,110	18,431	34,982	3,245	1,119	17,514	400	1,718	45,677	5,986
Male	6,694	922	220	895	981	131	-	1,619	22	90	1,816	-
Female	138,535	14,124	890	17,536	34,001	3,114	1,119	15,896	379	1,628	43,861	5,986
Other persona	al services w	orkers										
Total	46,414	5,676	1,129	8,254	18,408	298	199	5,968	68	228	5,555	631
Male	28,022	1,954	868	5,569	11,561	298	163	3,529	22	179	3,686	194
Female	18,392	3,722	261	2,685	6,847	-	36	2,439	46	49	1,869	437
SALES WORKI	ERS											
Total	1,464,338	84,504	25,958	297,827	504,743	31,777	16,364	236,987	8,430	15,385	214,394	27,968
Male	613,892	29,287	8,296	112,141	213,070	12,971	4,838	103,146	3,515	5,086	111,143	10,399
Female	850,446	55,217	17,662	185,686	291,673	18,807	11,526	133,841	4,915	10,299	103,251	17,569
Street and ma	rket salespe	ersons										
Total	511,051	18,266	11,731	99,058	209,083	10,546	2,266	83,081	767	4,999	62,184	9,069
Male	202,399	4,990	3,286	43,674	79,717	2,835	462	36,346	517	789	28,569	1,215
Female	308,652	13,276	8,446	55,384	129,365	7,711	1,804	46,735	251	4,210	33,615	7,854
Shop salesper	rsons											
Total	525,167	36,274	6,287	85,761	159,860	14,463	5,780	77,588	3,353	6,355	116,816	12,631
Male	285,096	15,627	2,164	43,650	90,218	7,128	3,026	45,648	1,971	3,194	66,123	6,346
Female	240,072	20,647	4,124	42,110	69,641	7,335	2,754	31,939	1,382	3,162	50,692	6,285

Table 5.6 Cont'd

Minor						Re	gion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Cashiers and ti	cket clerks											
Total	21,165	1,869	34	4,539	6,742	-	206	3,717		570	2,588	717
Male	8,564	547	34	1,854	2,696	-	206	1,798		224	955	88
Female	12,601	1,322	-	2,684	4,046	-	-	1,919	22	346	1,633	629
Other sales wo	rkers											
Total	231,609	16,830	4,958	54,504	78,284	6,592	1,863	31,640		3,333	26,129	5,357
Male	106,715	7,898	2,385	22,185	35,957	2,832	805	15,374	798	880	14,851	2,750
Female	124,894	8,932	2,573	32,319	42,327	3,760	1,057	16,265	1,322	2,453	11,278	2,607
Katikala retaile												
Total	75,392	3,650	214	13,755	24,933	-	3,523	25,601	1,727	41	1,948	-
Male	4,735	226	-	778	1,014	-	209	1,796		-	645	-
Female	70,656	3,424	214	12,976	23,920	-	3,314	23,805	1,659	41	1,303	-
"Teji " retailers												
Total	12,810	-	428	3,085	6,718	-	-	2,580	-	-	-	-
Male	3,073	-	428	-	1,084	-	-	1,561	-	-	-	-
Female	9,737	-	-	3,085	5,633	-	-	1,019	-	-	-	-
"Tela" retailers												
Total	87,145	7,615	2,306	37,127	19,124	176	2,726	12,781	280	87	4,729	194
Male	3,310	-	-	-	2,383	176	129	622		-	-	-
Female	83,834	7,615	2,306	37,127	16,740	-	2,597	12,159	280	87	4,729	194
PERSONAL CA	RE WORK	ERS										
Total	24,452	447	131	5,548	7,189	-	213	1,883	-	203	8,498	339
Male	5,246	90	-	1,184	1,319	-	-	945	-	122	1,586	-
Female	19,206	357	131	4,364	5,870	-	213	938	-	81	6,912	339
Child care wor												
Total	14,296	447	19	4,955	2,901	-	213	1,100	-	203	4,372	87
Male	2,247	90	-	1,184	105	-	-	403	-	122	343	-
Female	12,049	357	19	3,771	2,795	-	213	697	-	81	4,029	87
Personal care	workers in I	health ser	vices									
Total	10,156	-	112	594	4,289	-	-	783	-	-	4,127	252
Male	2,999	-	-	-	1,214	-	-	542	-	-	1,243	-
Female	7,157	-	112	594	3,075	-	-	241	-	-	2,883	252
PROTECTIVE S												
Total	165,965	12,443	2,458	24,659	37,741	5,504	2,055	28,046	,	3,256	42,669	5,134
Male	147,414	11,468	2,016	20,345	34,716	4,798	1,895	24,353	1,714	2,824	38,556	4,729
Female	18,551	975	442	4,314	3,025	706	160	3,693	286	432	4,113	405

Table 5.6 Cont'd

Table 5.6 Cont C	ı .											
Minor						Re	gion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Protective serv	ices worker	rs				•				•		
Total	165,965	12,443	2,458	24,659	37,741	5,504	2,055	28,046	2,000	3,256	42,669	5,134
Male	147,414	11,468	2,016	20,345	34,716	4,798	1,895	24,353	1,714	2,824	38,556	4,729
Female	18,551	975	442	4,314	3,025	706	160	3,693	286	432	4,113	405
MAJOR GROUP 6: SKILLED AGRICULTURAL, FORESTRY AND FISHERY WORKERS												
Total	387,801	21,439	17,018	118,530	99,870	9,860	17,787	74,916	5,473	405	18,616	3,888
Male	270,676	14,809	13,219	85,353	71,836	5,864	10,071	51,351	3,814	215	12,045	2,099
Female	117,126	6,630	3,799	33,177	28,034	3,996	7,716	23,565	1,659	189	6,571	1,789
MARKET-ORIE												
Total	117,603	7,023	9,479	14,007	49,267	1,685	3,360	17,267	1,396	215	13,040	864
Male	79,664	3,630	7,410	7,861	35,052	1,339	2,567	11,814	933	166	8,115	777
Female	37,938	3,393	2,068	6,146	14,215	346	793	5,453	463	49	4,925	86
Market garden												
Total	70,879	3,756	8,871	9,653	30,672	983	1,966	7,222	627	117	6,674	339
Male	49,349	1,859	6,931	5,122	21,004	983	1,746	6,182	354	117	4,711	339
Female	21,530	1,897	1,940	4,531	9,667	-	220	1,039	273	-	1,962	-
Animal produc	ers											
Total	33,282	3,086	608	3,762	11,979	303	28	6,873	769	-	5,350	525
Male	18,754	1,590	479	2,147	7,954	90	28	3,062	579	-	2,388	438
Female	14,528	1,496	128	1,615	4,026	212	-	3,811	191	-	2,963	86
Mixed crop an	d animal pro	oducers										
Total	13,442	181	-	592	6,616	400	1,366	3,173	-	98	1,016	-
Male	11,562	181	-	592	6,094	267	793	2,570	-	49	1,016	-
Female	1,880	-	-	-	522	133	573	603	-	49	-	-
MARKET-ORIE	NTED SKIL	LED FOR	ESTRY, I	FISHERY A	ITNUH DN.	NG WOR	KERS					
Total	12,515	259	2,897	2,323	1,314	363	314	221	608	-	4,217	-
Male	9,080	259	2,690	1,731	180	322	314	125	209	-	3,250	-
Female	3,435	-	207	592	1,134	40	-	95	399	-	967	-
Forestry and r	elated worke	ers										
Total	9,867	259	2,787	1,729	638	363	314	95	419	-	3,263	-
Male	7,114	259	2,579	1,138	60	322	314	-	146	-	2,296	-
Female	2,753	-	207	592	577	40	-	95	273	-	967	-
Fishery worke	rs, hunters a	and trappe	ers									
Total	2,648	-	111	594	676	-	-	125	188	-	954	-
Male	1,966	-	111	594	120	-	-	125	63	-	954	-
Female	682	-	-	-	556	-	-	-	126	-	-	-

Table 5.6 Cont'd

Minor	•					Re	gion					
Occupational						1	Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
SUBSISTENCE							Garraz	01111111	Jambola	Haran	7 lbaba	Dana
Total	257,684	14,158	4,642	102,199	49,289	7,812	14,113	57,429	3,470	189	1,359	3,024
Male	181,931	10,921	3,119	75,760	36,603	4,203	7,191	39,412	2,672	49	679	1,321
Female	75,753	3,237	1,523	26,439	12,686	3,609	6,923	18,017	797	140	679	1,703
Subsistence ci			,	,	•	,	•	,				,
Total	71,484	9,344	862	19,030	19,543	1,915	129	17,858	2,503	50	-	250
Male	60,035	8,254	846	17,099	16,800	1,301	129	13,549	1,970	-	-	86
Female	11,449	1,090	16	1,931	2,744	614	-	4,309	533	50	-	164
Subsistence liv	vestock farn	ners										
Total	98,705	2,848	3,563	28,038	19,737	5,583	10,073	25,054	228	49	1,025	2,509
Male	53,248	701	2,056	13,452	12,768	2,748	4,176	15,560	139	-	679	970
Female	45,457	2,147	1,507	14,586	6,968	2,835	5,897	9,494	89	49	346	1,539
Subsistence m	nixed crop a		ck farmers	5								
Total	86,204	1,966	-	55,131	10,009	315	3,783	14,032		91	-	265
Male	67,691	1,966	-	45,208	7,035	154	2,757	9,818		49	-	265
Female	18,513	-	-	9,923	2,974	160	1,026	4,214	175	41	-	-
Subsistence fis	shers, hunte	ers, trappe		therers								
Total	1,291	-	218	-	-	-	128	486		-	333	-
Male	957	-	218	-	-	-	128	486	126	-	-	-
Female	333	-	-	-	-	-	-	-	-	-	333	-
MAJOR GROUP	7: CRAFT			RADES WO	RKERS							
Total	1,089,180	82,120	12,248	209,106	357,711	11,017	12,229	152,338	5,131	9,214	221,334	16,732
Male	696,336	51,394	7,666	134,065	223,891	6,805	8,386	93,242	2,847	5,422	152,494	10,125
Female	392,845	30,726	4,582	75,041	133,821	4,212	3,843	59,096	2,284	3,792	68,839	6,608
BUILDING AND												
Total	324,101	21,356	4,539	62,392	111,823	4,360	4,701	48,896		2,804	56,358	5,646
Male	290,753	18,991	4,489	54,670	100,947	3,994	4,327	45,740		2,764	48,790	5,116
Female	33,347	2,365	50	7,722	10,876	366	374	3,157	298	41	7,568	530
Building frame			orkers									
Total	283,032	19,003	4,134	54,135	97,716	3,813	4,502	45,644		2,471	45,682	4,836
Male	252,340	16,638	4,103	46,413	87,708	3,762	4,128	42,488	886	2,471	39,437	4,306
Female	30,693	2,365	31	7,722	10,009	51	374	3,157	209	-	6,245	530
Building finishe	ers and rela	ted trades	workers									
Total	20,365	1,295	347	5,069	6,665	473	92	1,429		202	4,490	214
Male	18,815	1,295	347	5,069	6,152	158	92	1,429		202	3,856	214
Female	1,550	-	-	-	513	315	-	-	89	-	633	-

Table 5.6 Cont'd

Minor						Re	gion					
Occupational						110	Benishangul			Γ	Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Painters, build						Sumane	-Guilluz	SININFIX	Garribeia	Halall	Ababa	Dawa
Total	20,703	1,059	59	3,188	7,441	73	107	1,822	40	131	6,187	596
Male	19,599	1,059	40	3,188	7,441	73		1,822		91	5,496	596
Female	1,104	1,059	19	3,100	354	73		1,022	40	41	690	590
METAL, MACH		- NDELATE		- 		-	-	-	-	41	090	-
Total	120,403	12,708	1,138	23,194	30,065	895	1,558	7,698	685	791	38,946	2,725
Male	112,591	11,853	1,138	20,426	28,229	895	1,427	7,509		791 791	37,251	2,725
Female	7,812	855	1,130	20,426	1,836	090	1,427	189	000	791	1,695	339
Sheet and stru						tod worke		109	-	-	1,095	339
Total	55,100	6,258	272	13,075	13,442	233	656	2,965	557	304	16,391	947
Male	52,384	6,256	272	13,075	12,271	233				304	15,029	855
Female	2,715	90	212	13,075		233	000	2,965		304	1,362	92
	,				1,171	-	-	-	-	-	1,362	92
Blacksmiths, to					4 040	200	207	101		4.45	4.500	000
Total	17,481	2,252	344	4,115	4,618	200	207	191	-	145	4,599	809
Male	12,969	1,577	344	1,451	4,156	200		191	-	145	4,267	562
Female	4,512	674	-	2,664	463	-	130	-	-	-	333	247
Machinery me				0.004	40.005	400	005	4.540	400	0.40	47.055	000
Total	47,822	4,198	522	6,004	12,005	462	695	4,543		342	17,955	969
Male	47,237	4,108	522	5,900	11,803	462		4,354		342	17,955	969
Female	585	90	-	103	202	-	-	189	-	-	-	-
HANDICRAFT A				45.000	04.404	000	4 00 4	47.000	4 070	4 400	00.000	007
Total	179,790	12,827	680	45,200	61,164	699	1,604	17,902		1,198	36,809	627
Male	97,651	4,221	250	19,948	37,240	296	864	12,295		568	20,773	501
Female	82,140	8,606	431	25,252	23,925	403	739	5,608	385	629	16,036	126
Handicraft wor												
Total	171,453	12,327	680	44,439	61,131	699	1,604	17,535		1,198	30,134	627
Male	94,358	4,085	250	19,280	37,206	296	864	11,927		568	18,687	501
Female	77,095	8,242	431	25,159	23,925	403	739	5,608	385	629	11,448	126
Printing trades												
Total	8,337	500	-	761	34	-	-	368	-	-	6,675	-
Male	3,293	136	-	668	34	-	-	368	-	-	2,087	-
Female	5,045	363	-	93	-	-	-	-	-	-	4,589	-
ELECTRICAL A												
Total	48,300	2,704	690	8,115	13,053	427	29	4,964		382	16,757	990
Male	43,668	2,704	482	7,393	12,411	427	29	4,561	137	382	14,239	904
Female	4,632	-	207	722	642	-	-	403	53	-	2,518	87

Table 5.6 Cont'd

Table 5.6 Cont C	ı											
Minor						Re	gion		,			
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Electrical equi												
Total	36,671	1,980	690	5,213	9,565	375	29	3,480	149	195	14,253	743
Male	34,547	1,980	482	4,491	9,446	375	29	3,480	97	195	13,231	743
Female	2,124	-	207	722	119	-	-	-	53	-	1,022	-
Electronics an			installers									
Total	11,629	725	-	2,902	3,488	52	-	1,484	40	187	2,504	248
Male	9,121	725	-	2,902	2,965	52	-	1,081	40	187	1,009	161
Female	2,508	-	-	-	523	-	-	403	-	-	1,495	87
FOOD PROCES	,		,						WORKER	S		
Total	416,587	32,524	5,201	70,204	141,607	4,636	4,338	72,877	1,953	4,040	72,463	6,744
Male	151,673	13,625	1,307	31,627	45,064	1,194	1,738	23,138	404	918	31,441	1,218
Female	264,914	18,900	3,894	38,576	96,542	3,443	2,600	49,740	1,549	3,123	41,022	5,526
Food processi												
Total	259,163	14,867	4,681	41,814	95,163	4,076	2,999	55,405	1,600	3,420	29,717	5,422
Male	54,088	5,030	994	10,331	16,794	713	817	10,554	154	485	7,702	515
Female	205,076	9,837	3,687	31,483	78,369	3,363	2,182	44,851	1,446	2,936	22,015	4,907
Wood treaters												
Total	25,532	994	76	3,095	7,953	-	693	3,794	121	-	8,646	161
Male	23,925	994	76	3,095	7,444	-	275	3,794	121	-	7,966	161
Female	1,607	-	-	-	509	-	418	-	-	-	681	-
Garment and i		es workers	3									
Total	125,939	16,663	444	23,645	36,803	560	646	13,437	233	620	31,905	984
Male	70,397	7,600	237	16,997	20,185	480	646	8,790	130	433	14,445	454
Female	55,542	9,063	207	6,647	16,618	80	-	4,648	103	187	17,460	530
Other craft and	d related wo	rkers										
Total	5,952	-	-	1,651	1,688	-	-	241	-	-	2,195	177
Male	3,262	-	-	1,204	641	-	-	-	-	-	1,328	89
Female	2,690	-	-	446	1,047	-	-	241	-	-	867	89
MAJOR GROUP												
Total	458,982	41,456	6,651	66,006	133,386	4,210	3,343	44,435	1,820	3,943	143,792	9,940
Male	418,261	37,408	6,620	61,977	119,114	4,045	3,269	43,468	1,678	3,862	127,937	8,884
Female	40,721	4,049	31	4,029	14,272	165	75	967	141	81	15,856	1,056
STATIONARY I												
Total	93,677	10,813	1,288	9,220	32,071	336	929	4,200	159	286	32,878	1,498
Male	63,316	7,110	1,288	7,312	20,517	244	893	3,491	137	205	21,604	515
Female	30,361	3,703	-	1,908	11,554	91	36	709	22	81	11,274	983

Table 5.6 Cont'd

Minor						Re	gion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Mining and mir	neral proces	ssing plan	t operator									
Total	8,511	702	426	2,128	2,705	120	514	1,189	66	-	574	86
Male	7,766	702	426	1,682	2,503	120	514	1,093	66	-	574	86
Female	745	-	-	446	202	-	-	96	-	-	-	-
Metal processi	ng and finis	shing plant	t operator	S								
Total	3,303	1,066	142	177	776	40	130	-	-	-	973	-
Male	2,849	1,066	142	177	657	40	130	-	-	-	637	-
Female	454	-	-	-	119	-	-	-	-	-	336	-
Chemical and		ic product	s plant ar	d machine	operators							
Total	3,201	153	-	61	1,322	-	-	191	-	48	1,353	73
Male	1,671	-	-	61	961	-	-	191	-	48	336	73
Female	1,530	153	-	-	361	-	-	-	-	-	1,017	-
Rubber, plastic	and paper	products	machine	operators								
Total	3,881	-	-	-	1,967	84	-	496	-	41	1,294	-
Male	2,349	-	-	-	750	84	-	496	-	-	1,019	-
Female	1,532	-	-	-	1,216	-	-	-	-	41	275	-
Textile, fur and	l leather pro	oducts ma	chine ope	rators								
Total	41,135	6,209	-	3,744	9,749	-	29	1,041	-	-	19,541	822
Male	27,307	3,916	-	2,850	6,642	-	29	576	-	-	13,117	179
Female	13,828	2,293	-	894	3,108	-	-	466	-	-	6,424	643
Food and relate		s machine	operators	3								
Total	19,363	1,191	440	2,519	10,965	91	220	371	93	81	3,392	-
Male	11,316	223	440	1,952	6,292	=	220	371	71	41	1,705	-
Female	8,047	967	-	567	4,673	91	-	-	22	41	1,687	-
Wood processi	ing and pap		plant ope	erators								
Total	925	90	-	-	678	=	36	121	-	-	-	-
Male	365	90	-	-	154	-	-	121	-	-	-	-
Female	560	-	-	-	524	-	36	-	-	-	-	-
Other stationar	ry plant and	l machine	operators	;								
Total	13,357	1,403	280	591	3,909	=	-	790	-	117	5,751	517
Male	9,692	1,113	280	591	2,557	-	-	643	-	117	4,214	177
Female	3,666	290	-	-	1,352	=	-	148	-	-	1,537	340
ASSEMBLERS												
Total	38,765	2,957	798	6,195	11,229	205	116	4,104	434	92	12,047	589
Male	33,902	2,867	766	5,008	10,088	131	77	4,008	314	92	9,963	589
Female	4,863	90	31	1,187	1,142	73	39	96	120	-	2,084	-

Table 5.6 Cont'd

Table 5.6 Contro	ן 											
Minor						Reg	gion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Assemblers		<u> </u>		•		l l	'			. U.		-
Total	38,765	2,957	798	6,195	11,229	205	116	4,104	434	92	12,047	589
Male	33,902	2,867	766	5,008	10,088	131	77	4,008	314	92	9,963	589
Female	4,863	90	31	1,187	1,142	73	39	96	120	-	2,084	-
DRIVERS AND	MOBILE P		ERATORS	S								
Total	326,540	27,686	4,566	50,590	90,086	3,669	2,299	36,131	1,227	3,565	98,868	7,853
Male	321,043	27,431	4,566	49,657	88,509	3,669	2,299	35,969	1,227	3,565	96,370	7,780
Female	5,497	255	-	934	1,576	-	-	162	-	-	2,498	73
Locomotive er		and relat	ed worke									
Total	905	-	-	103	34	92	29	-	-	-	333	314
Male	905	-	-	103	34	92	29	-	-	-	333	314
Female	-	-	-	-	-	-	-	-	-	-	-	-
Car, van and r												
Total	221,780	18,687	4,289	38,433	56,844	2,690	1,319	25,878	1,092	2,845	63,640	6,062
Male	218,241	18,432	4,289	37,946	56,043	2,690	1,319	25,878	1,092	2,845	61,718	5,989
Female	3,539	255	-	488	802	-	-	-	-	-	1,922	73
Heavy truck a												
Total	96,114	8,743	120	11,240	32,156	847	821	9,264	96	671	30,849	1,306
Male	94,405	8,743	120	10,795	31,469	847	821	9,264	96	671	30,273	1,306
Female	1,709	-	-	446	687	-	-	-	-	-	576	-
Mobile plant o												
Total	7,684	256	157	813	994	40	129	989	40	49	4,045	171
Male	7,435	256	157	813	907	40	129	827	40	49	4,045	171
Female	249		-	-	88	-	-	162	-	-	-	-
Ships' deck cr		ated worke	ers									
Total	57	-	-	-	57	-	-	-	-	-	-	-
Male .	57	-	-	-	57	-	-	-	-	-	-	-
Female	-	- NIT 4 D.V. O	-	-	-	-	-	-	-	-	-	-
MAJOR GROUP					445 405	07.740	40.000	000 000	40.044	40.000	000 000	00.007
Total	1,440,420	91,750	42,198	367,255	415,425	27,712	19,606	220,836		10,636	202,926	22,237
Male	736,433	49,511	23,177	185,079	243,273	13,947	10,193	103,175	6,813	4,312	85,436	11,515
Female	703,988	42,239	19,020	182,176	172,152	13,765	9,413	117,660	13,028	6,324	117,490	10,722
CLEANERS AN			2.040	20.004	E0 400	0.404	2.050	20.042	2.700	2.045	C4 0E0	2 000
Total	201,633	11,695	3,942	28,891	52,102	2,434	2,059	28,913	3,789	2,945	61,059	3,806
Male	37,267	44.005	2.040	3,690	11,404	322	337	8,434	1,012	98	10,786	1,184
Female	164,366	11,695	3,942	25,202	40,698	2,112	1,722	20,479	2,776	2,847	50,273	2,621

Table 5.6 Cont'd

Minor						Re	gion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Domestic, hote	el and office	cleaners	and helpe	ers		-						
Total	127,205	9,369	3,869	18,106	30,555	1,696	1,791	17,647	1,071	2,296	39,241	1,566
Male	7,451	-	-	666	1,074	184	288	2,123	-	49	2,978	88
Female	119,754	9,369	3,869	17,440	29,480	1,512	1,502	15,523	1,071	2,247	36,262	1,478
Vehicle, windo	w, laundry	and other		aning worke	rs							
Total	74,428	2,326	73	10,786	21,547		269	11,267	2,718	649	21,818	2,240
Male	29,816	-	-	3,024	10,329	138	49	6,311	1,012	49	7,808	1,096
Female	44,612	2,326	73	7,761	11,217	599	219	4,956	1,706	600	14,010	1,144
AGRICULTUR/	AL, FORES	TRY AND	FISHERY	/ LABOURE	RS							
Total	129,724	3,481	16,847	62,009	34,143	1,650	1,812	6,462	701	-	2,379	240
Male	88,772	2,576	11,472	41,524	26,071	581	746	4,361	175	-	1,026	240
Female	40,951	905	5,375	20,485	8,072	1,069	1,066	2,101	525	-	1,353	-
Agricultural, fo		ishery lab	ourers									
Total	129,724	3,481	16,847	62,009	34,143	1,650	1,812	6,462	701	-	2,379	240
Male	88,772	2,576	11,472	41,524	26,071	581	746	4,361	175	-	1,026	240
Female	40,951	905	5,375	20,485	8,072	1,069	1,066	2,101	525	-	1,353	-
LABOURERS I	N MINING,	CONSTRI	UCTION,	MANUFAC	TURING A	ND TRAN	SPORT					
Total	313,217	24,234	6,206	71,929	113,096	3,532	2,895	34,144	2,012	3,009	45,327	6,832
Male	234,670	18,433	4,351	50,489	86,301	3,101	2,280	27,314	1,787	2,574	31,991	6,050
Female	78,547	5,802	1,855	21,440	26,795	431	615	6,830	226	435	13,336	782
Mining and co	nstruction la	abourers										
Total	128,525	12,586	1,844	35,695	44,059	1,424	1,218	10,320	862	1,141	17,556	1,819
Male	89,332	8,392	568	22,014	35,881	1,291	944	5,094	676	749	11,990	1,730
Female	39,193	4,194	1,276	13,681	8,178	133	274	5,226	185	392	5,566	89
Manufacturing	labourers											
Total	88,687	2,880	2,260	17,690	41,999	730	581	6,179	177	362	13,496	2,334
Male	53,754	2,247	1,697	11,241	24,439	432	240	4,576	177	318	6,747	1,641
Female	34,933	633	563	6,449	17,560	298	342	1,604	-	43	6,749	693
Transport and	storage lab											
Total	96,005	8,768	2,101	18,544	27,038	1,378	1,096	17,644	974	1,506	14,275	2,679
Male	91,584	7,793	2,085	17,234	25,981	1,378	1,096	17,644	934	1,506	13,254	2,679
Female	4,421	975	16	1,310	1,058	-	-	-	40	-	1,022	-
FOOD PREPAR	RATION AS	SISTANT	S									
Total	56,301	6,778	382	9,526	18,228	264	402	7,900	328	252	11,805	437
Male	6,689	408	-	537	2,934	51	-	1,215	164	-	1,381	-
Female	49,612	6,370	382	8,989	15,294	213	402	6,685	164	252	10,424	437

_

Table 5.6 Cont'd

Minor						Re	gion					
Occupational							Benishangul				Addis	Dire
Groups and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Food preparat	ion assistan		'				•		•		•	
Total	56,301	6,778	382	9,526	18,228	264	402	7,900	328	252	11,805	437
Male	6,689	408	-	537	2,934	51	-	1,215	164	-	1,381	-
Female	49,612	6,370	382	8,989	15,294	213	402	6,685	164	252	10,424	437
STREET AND I	RELATED S	SALES AN	D SERVI	CE WORKE	RS							
Total	61,548	3,776	2,393	12,075	15,882	560	320	11,370	488	740	12,661	1,284
Male	32,472	1,218	389	4,881	11,139	329	262	5,300	195	415	7,909	436
Female	29,075	2,558	2,004	7,194	4,743	231	57	6,070	293	325	4,751	848
Street and rela	ated service	workers	•		•			•				
Total	35,133	3,148	1,826	8,611	8,916	100	-	4,183	205	125	7,325	695
Male	18,454	1,026	175	3,688	6,523	50	-	2,558	124	84	3,882	344
Female	16,680	2,122	1,650	4,923	2,394	50	-	1,625	81	40	3,443	351
Street vendors	s (excluding	food)	•		•			•				
Total	26,415	627	567	3,464	6,965	460	320	7,187	284	615	5,335	589
Male	14,019	191	214	1,193	4,616	278	262	2,742	71	331	4,027	92
Female	12,396	436	353	2,271	2,349	181	57	4,445	213	285	1,308	497
REFUSE WOR		OTHER E	ELEMENT					•			,	
Total	677,998	41,785	12,428	182,825	181,974	19,273	12,118	132,048	12,523	3,690	69,695	9,637
Male	336,561	26,877	6,966	83,959	105,424	9,563	6,568	56,552	3,480	1,225	32,343	3,605
Female	341,436	14,909	5,463	98,866	76,551	9,710	5,550	75,495	9,042	2,465	37,353	6,033
Refuse worker	rs		•		•		·	•				
Total	11,368	-	-	4,036	1,126	82	-	2,105	313	49	3,328	329
Male	7,252	-	-	3,079	1,126	-	-	301	89	-	2,657	-
Female	4,116	-	-	956	-	82	-	1,804	224	49	671	329
Other element	ary workers											
Total	666,630	41,785	12,428	178,789	180,849	19,191	12,118	129,943	12,210	3,641	66,368	9,308
Male	329,310	26,877	6,966	80,880	104,298	9,563	6,568	56,251	3,391	1,225	29,686	3,605
Female	337,321	14,909	5,463	97,909	76,551	9,627	5,550	73,692	8,818	2,416	36,682	5,704
OTHERS												
Total	9,032	223	16	594	3,682	103	130	1,264	-	81	2,847	90
Male	7,254	223	16	594	3,111	103	130	1,264	-	81	1,642	90
Female	1,777	-	-	-	572	-	-	-	-	-	1,206	-

Table 5.7 Employed Population of Urban Areas Aged Ten Years and Above by Major Industrial Divisions, Age Group and Sex - COUNTRY TOTAL: 2015

Major Industrial						Age	e Group						
Divisions and													
Sex	All Ages	10-14	15-19	20-24	25-29	30-34	35-39	40-44	45-49	50-54	55-59	60-64	65+
Total Employed F	•												
Total	7,083,094	194,969	565,103	1,107,918	1,455,213	985,745		578,390	421,698	288,013		145,598	194,851
Male	3,963,143	100,707	224,956	546,853	827,711	599,578	518,080	365,839	252,303	173,544	128,640	90,332	134,599
Female	3,119,950	94,262	340,148	561,064	627,502	386,166	417,240	212,551	169,394	114,470	81,635	55,266	60,253
Agriculture, forest													
Total	567,067	56,161	52,065	47,889	71,536	59,087	62,985	46,494	43,489	33,042	29,383	21,417	43,519
Male	382,749	42,609	33,585	31,213	50,002	38,204	34,214	31,910	28,644	20,258	19,856	15,431	36,824
Female	184,319	13,552	18,479	16,676	21,534	20,883	28,772	14,585	14,845	12,784	9,527	5,987	6,695
Mining and quarry	ying												
Total	35,165	727	3,161	4,123	9,168	7,258	4,079	2,494	1,987	692	443	261	772
Male	28,221	367	1,453	2,842	6,998	6,850	3,180	2,494	1,987	692	383	261	715
Female	6,944	360	1,708	1,281	2,171	408	899	-	-	-	61	-	57
Manufacturing													
Total	929,847	17,158	60,579	141,962	184,748	126,202	117,323	77,050	62,223	47,214	32,884	27,719	34,784
Male	489,346	8,862	33,742	77,648	106,716	66,571	55,302	43,719	26,884	19,835	17,839	12,891	19,337
Female	440,501	8,296	26,837	64,314	78,033	59,631	62,021	33,331	35,339	27,380	15,045	14,828	15,447
Electricity, gas, st	team and air d	conditioning	g supply										
Total	21,693	-	1,142	1,923	6,284	1,962	2,696	676	4,083	949	899	182	896
Male	17,435	-	287	779	5,721	1,869	1,968	334	3,641	949	808	182	896
Female	4,258	-	855	1,143	563	93	728	343	442	-	91	-	-
Water supply; sev	werage, waste	e managen	nent and re	mediation a	ctivities								
Total	55,487	3,962	2,221	5,824	13,425	5,569	7,300	4,724	4,500	2,396	1,841	1,448	2,277
Male	34,784	1,465	1,174	4,381	9,987	3,551	4,330	3,299	2,904	1,617	1,479	542	57
Female	20,703	2,497	1,047	1,443	3,439	2,018	2,971	1,426	1,596	779	362	906	2,220
Construction													
Total	507,021	1,954	37,254	90,520	113,143	81,699	69,440	44,256	22,955	17,174	9,930	9,368	9,327
Male	423,634	1,552	21,095	63,186	96,892	74,509	64,352	40,527	20,080	14,862	9,331	8,670	8,579
Female	83,387	402	16,159	27,334	16,252	7,190	5,089	3,729	2,875	2,312	599	698	747
Wholesale and re	tail trade; rep	air of moto	r vehicles	and motorcy	cles	,	,	,	,	,			
Total	1,465,772	26,984	100,111	206,611	278,748	203,274	223,232	145,575	101,531	56,334	45,888	36,281	41,201
Male	706,136	14,503	48,572	99,828	134,865	102,727	101,957	70,652	46,964	26,273	19,726	18,192	21,878
Female	759,636	12,482	51,539	106,784	143.884	100,546	121,275	74,923	54,568	30,061	26,162	18,089	19,323
	. 55,550	, .5_	0.,000		,		, 0	,==0	0 .,000	55,551	_0,.5_	. 5,550	. 5,526

Table 5.7 Cont'd

Major Industrial						Age	e Group						
Divisions and													
Sex	All Ages	10-14	15-19	20-24	25-29	30-34	35-39	40-44	45-49	50-54	55-59	60-64	65+
Transportation an	•												
Total	389,343	2,312	19,478	78,071	97,007	65,681	56,847	27,844	15,923	9,199	6,993	6,457	3,531
Male	368,724	2,120	18,944	72,366	91,372	62,505	54,361	27,721	15,343	7,897	6,381	6,181	3,531
Female	20,619	191	534	5,705	5,635	3,176	2,486	123	579	1,302	612	276	-
Accommodation a													
Total	406,137	8,388	51,922	86,432	76,421	48,968	52,214	23,595	19,835	11,903	9,950	6,702	9,808
Male	115,886	2,687	10,499	24,982	17,768	16,095	13,878	7,561	7,110	3,805	1,939	3,437	6,126
Female	290,251	5,701	41,423	61,450	58,653	32,873	38,336	16,034	12,725	8,098	8,011	3,266	3,681
Information and co		1											
Total	73,483	-	2,226	13,104	19,109	14,477	8,290	4,248	2,835	3,648	2,555	1,783	1,209
Male	40,126	-	1,169	6,765	8,422	9,239	3,557	1,752	2,149	3,122	1,689	1,053	1,209
Female	33,357	-	1,057	6,339	10,687	5,239	4,732	2,496	685	526	866	730	-
Financial and insu	ırance activiti	es											
Total	171,923	-	1,388	30,636	43,965	31,592	22,104	17,792	10,522	5,858	3,269	2,001	2,798
Male	108,035	-	947	11,592	28,051	18,096	16,689	14,210	7,643	4,596	1,528	2,001	2,681
Female	63,887	-	441	19,043	15,914	13,496	5,415	3,582	2,879	1,261	1,741	-	116
Real estate activit	ies												
Total	5,339	-	241	-	1,570	539	1,271	954	-	-	764	-	-
Male	2,909	-	241	-	615	342	336	954	-	-	421	-	-
Female	2,430	-	-	-	954	197	935	-	-	-	343	-	-
Professional, scie	ntific and tecl	hnical activ	ities										
Total	126,086	-	1,482	14,966	30,524	24,872	19,065	12,764	10,285	7,315	1,674	1,526	1,612
Male	85,419	-	1,138	6,600	18,487	19,216	11,973	8,877	8,809	6,259	1,285	1,180	1,596
Female	40,667	-	344	8,366	12,037	5,656	7,092	3,887	1,476	1,056	389	347	16
Administrative and	d support serv	vice activiti	es										
Total	146,604	380	3,550	20,105	26,056	25,329	27,645	13,742	11,570	8,535	5,938	1,826	1,930
Male	87,366	-	1,104	7,973	13,957	14,466	14,718	9,352	9,761	7,513	5,477	1,115	1,930
Female	59,238	380	2,446	12,132	12,099	10,863	12,926	4,389	1,809	1,022	461	710	-
Public administrat	ion and defer	nse; compu	Isory socia	l security									
Total	349,400	-	2,663	41,560	84,573	65,104	53,832	39,798	28,663	16,403	9,129	3,286	4,389
Male	235,776	-	1,148	18,382	51,029	48,055	36,481	30,256	22,109	13,387	7,552	3,052	4,327
Female	113,623	-	1,516	23,178	33,544	17,049	17,351	9,542	6,554	3,016	1,577	235	62

Table 5.7 Cont'd

Major Industrial						Age	Group						
Divisions and Sex	All Ages	10-14	15-19	20-24	25-29	30-34	35-39	40-44	45-49	50-54	55-59	60-64	65+
Education	7 til 7 tg00	10 11	10 10	20 2 1	20 20	00 01	00 00	10 11	10 10	00 0 1	00 00	00 01	001
Total	469,184	153	5,442	70,834	124,091	79,120	70,743	37,106	28,423	22,245	18,404	8,624	3,999
Male	254,406	_	458	27,237	65,117	46,793	35,831	22,396	16,695	15,648	12,383	7,943	3,905
Female	214,778	153	4,984	43,597	58,973	32,327	34,912	14,710	11,728	6,597	6,021	682	93
Human health and	d social work	activities											
Total	189,014	96	3,576	26,914	64,764	24,412	27,781	12,832	10,441	7,732	6,666	552	3,247
Male	78,247	-	838	4,973	24,951	8,694	11,222	6,896	6,572	5,000	5,752	375	2,976
Female	110,766	96	2,738	21,941	39,813	15,718	16,559	5,936	3,870	2,732	914	178	272
Arts, entertainme	nt and recreat	ion											
Total	49,289	-	6,535	9,839	11,283	7,275	5,072	2,745	2,097	2,868	933	91	553
Male	28,878	-	2,857	6,639	8,005	3,803	2,288	1,827	952	1,745	450	91	221
Female	20,411	-	3,678	3,200	3,278	3,472	2,784	918	1,144	1,123	483	-	332
Other service acti	vities												
Total	440,377	7,815	38,694	93,744	88,051	55,185	51,445	34,190	21,127	16,603	13,318	6,865	13,340
Male	288,385	6,797	21,568	51,723	56,517	38,331	34,289	25,117	16,550	12,559	9,992	4,469	10,473
Female	151,992	1,017	17,126	42,021	31,534	16,854	17,156	9,074	4,577	4,044	3,326	2,396	2,867
Activities of house	eholds as emp	oloyers; un	differentiate	ed goods- an	d services-p	roducing a	ctivities of	household	ls for own	use			
Total	606,525	68,880	170,929	113,438	94,867	43,493	42,229	19,594	12,510	12,037	7,021	8,035	13,492
Male	139,157	19,745	24,075	22,962	25,649	10,977	12,403	7,977	2,737	2,761	2,265	2,437	5,169
Female	467,368	49,135	146,854	90,476	69,218	32,516	29,825	11,617	9,774	9,276	4,756	5,598	8,323
Activities of extrat	erritorial orga	nizations a	nd bodies I	Description									
Total	78,337	-	445	9,423	15,881	14,648	9,727	9,916	6,700	5,865	2,391	1,172	2,169
Male	47,522	-	63	4,784	6,592	8,686	4,751	8,010	4,770	4,764	2,104	830	2,169
Female	30,815	-	383	4,639	9,289	5,962	4,976	1,906	1,930	1,102	287	342	

Table 5.8 Employed Population of Urban Areas Aged Ten Years and Above by Major Industrial Divisions, Sex and Region-COUNTRY TOTAL: 2015

Major Industrial						R	Region					
Divisions and							Benishangul				Addis	Dire
Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Total Employed P	opulation											
Total	7,083,094	477,813	131,439	1,449,091	2,102,417	119,082	107,053	1,096,887	61,809	59,109	1,362,069	116,325
Male	3,963,143	253,254	75,452	795,183	1,213,723	67,114	58,730	603,653	31,771	29,900	772,367	61,997
Female	3,119,950	224,559	55,987	653,908	888,694	51,967	48,323	493,233	30,038	29,210	589,703	54,327
Agriculture, foresti	ry and fishin	ıg										
Total	567,067	31,685	34,223	185,318	159,457	14,676	21,054	94,138	6,600	536	15,768	3,613
Male	382,749	21,450	25,357	129,045	107,658	8,462	11,711	62,298	4,570	307	10,138	1,753
Female	184,319	10,236	8,866	56,273	51,799	6,213	9,343	31,840	2,030	230	5,631	1,860
Mining and quarry	ring											
Total	35,165	3,505	111	9,019	15,908	234	552	1,459	182	-	4,197	-
Male	28,221	3,143	111	6,543	12,977	194	480	1,099	142	-	3,534	-
Female	6,944	362	-	2,476	2,932	40	72	360	40	-	663	-
Manufacturing												
Total	929,847	72,036	19,836	180,459	300,099	4,450	10,875	119,318	3,911	4,650	200,629	13,585
Male	489,346	41,861	10,157	89,051	157,750	2,376	3,758	47,621	1,871	2,159	125,317	7,426
Female	440,501	30,175	9,679	91,408	142,349	2,074	7,117	71,697	2,040	2,491	75,313	6,159
Electricity, gas, ste	eam and air	conditioni	ng supply									
Total	21,693	2,916	282	5,154	5,889	152	68	1,982	114	91	4,103	942
Male	17,435	2,300	282	4,685	4,781	101	68	1,117	22	49	3,088	942
Female	4,258	616	-	469	1,108	50	-	865	93	42	1,015	-
Water supply; sew	verage, was	te manage	ement and i	remediation a	activities							
Total	55,487	2,099	3,988	14,085	14,387	3,072	143	9,941	394	453	6,018	907
Male	34,784	1,646	1,959	10,438	8,900	1,899	107	5,281	180	354	3,838	182
Female	20,703	453	2,029	3,647	5,487	1,173	36	4,660	213	99	2,181	725
Construction												
Total	507,021	47,379	8,642	107,696	148,626	5,881	7,385	64,098	2,401	4,027	101,468	9,419
Male	423,634	39,474	7,104	85,168	128,706	5,634	6,491	56,496	2,041	3,589	80,907	8,024
Female	83,387	7,905	1,538	22,528	19,920	247	894	7,602	359	438	20,561	1,395
	,											
	,		of motor ve	ehicles and m	notorcycles							
	,	de; repair 90,787	of motor ve 26,116	ehicles and m 269,813	480,106	35,862	12,011	231,455	8,292	17,121	262,031	32,179
Wholesale a	and retail tra	de; repair			-	35,862 13,579 22,283	12,011 5,195 6,816	231,455 113,143 118,312	8,292 4,280 4,012	17,121 6,207	262,031 148,267	32,179 13,396

Table 5.8 Cont'd

Major Industrial	<u>u</u>					F	Region					
Divisions and							Benishangul				Addis	Dire
Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Transportation and	d storage											
Total	389,343	25,212	3,600	63,314	118,525	4,269	3,099	50,421	2,171	3,671	105,937	9,124
Male	368,724	23,333	3,373	60,105	114,599	4,229	3,022	48,595	2,067	3,312	97,414	8,674
Female	20,619	1,879	227	3,209	3,926	40	77	1,826	103	358	8,523	450
Accommodation a	and food ser	vice activ	ities									
Total	406,137	28,127	9,421	73,137	136,982	1,886	8,088	68,290	4,382	2,133	68,049	5,642
Male	115,886	4,186	1,534	13,389	43,137	761	2,516	26,670	1,423	460	20,943	865
Female	290,251	23,941	7,887	59,748	93,845	1,124	5,572	41,620	2,959	1,673	47,106	4,777
Information and co	ommunicati	on										
Total	73,483	3,061	836	9,174	11,578	422	649	12,865	289	518	33,221	871
Male	40,126	2,174	184	6,687	6,892	250	203	7,850	184	326	15,033	345
Female	33,357	887	653	2,487	4,686	172	446	5,015	105	192	18,188	526
Financial and insu	ırance activ	ities										
Total	171,923	9,289	921	30,152	42,446	3,008	1,713	25,059	1,448	659	55,125	2,102
Male	108,035	3,680	646	19,252	30,941	2,269	1,362	16,071	923	339	31,129	1,421
Female	63,887	5,609	274	10,900	11,505	739	351	8,988	525	320	23,996	681
Real estate activit	ies											
Total	5,339	226	-	654	-	-	-	241	-	-	4,218	-
Male	2,909	90	-	-	-	-	-	241	-	-	2,578	-
Female	2,430	135	-	654	-	-	-	-	-	-	1,640	-
Professional, sc	cientific and	technical	activities									
Total	126,086	5,109	1,511	13,047	32,310	2,722	2,917	20,538	1,891	542	43,792	1,706
Male	85,419	3,815	561	9,772	23,614	2,368	1,900	14,675	1,008	259	25,991	1,459
Female	40,667	1,295	950	3,275	8,697	355	1,017	5,863	884	283	17,801	248
Administrative and	d support se	ervice acti	vities									
Total	146,604	7,264	734	17,841	39,897	2,350	859	30,027	1,053	648	44,865	1,065
Male	87,366	5,399	369	10,181	25,545	1,589	545	15,971	620	139	26,313	696
Female	59,238	1,865	365	7,660	14,352	761	314	14,057	434	509	18,552	369
Public administrat	ion and def	ense; com	npulsory s	ocial securi	ty							
Total	349,400	27,395	6,243	82,625	81,495	6,371	10,615	63,604	3,733	7,190	54,729	5,399
Male	235,776	18,775	4,298	60,109	55,346	4,915	7,119	41,426	2,223	4,963	32,925	3,677
Female	113,623	8,620	1,945	22,516	26,149	1,456	3,497	22,178	1,510	2,227	21,804	1,722

Table 5.8 Cont'd

Major Industrial	_					F	Region					
Divisions and							Benishangul				Addis	Dire
Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Education												
Total	469,184	37,627	5,085	111,552	120,381	6,296	7,812	98,077	4,681	4,977	66,485	6,211
Male	254,406	17,996	4,043	61,365	66,927	4,518	4,879	54,063	2,989	2,364	32,080	3,182
Female	214,778	19,631	1,042	50,187	53,454	1,778	2,932	44,015	1,692	2,613	34,405	3,029
Human health an	d social wor	k activities	3									
Total	189,014	15,351	3,147	31,631	54,186	2,762	5,353	28,964	1,537	3,093	39,654	3,335
Male	78,247	6,514	1,040	15,923	21,624	1,592	2,483	13,130	673	1,318	12,593	1,355
Female	110,766	8,837	2,107	15,708	32,561	1,170	2,870	15,834	864	1,774	27,060	1,980
Arts, entertainme	nt and recre	ation										
Total	49,289	3,960	673	12,003	9,490	173	405	6,238	189	287	15,350	521
Male	28,878	2,655	460	4,685	7,678	173	253	3,357	144	81	8,957	435
Female	20,411	1,305	214	7,318	1,811	-	152	2,881	45	206	6,393	86
Other service act	ivities											
Total	440,377	23,070	3,010	93,769	147,951	7,028	3,052	69,200	4,878	3,357	77,147	7,916
Male	288,385	9,291	2,086	60,685	104,997	6,040	2,225	45,740	2,330	2,652	46,549	5,792
Female	151,992	13,780	924	33,084	42,954		827	23,460	2,548	705	30,598	2,124
Activities of hous	eholds as er	nployers;	undifferer	itiated good	ls- and serv	/ices-produ	ucing activities of	households	for own us	e		
Total	606,525	34,959	1,938	130,076	163,314	15,118	9,662	89,617	11,671	4,642	134,455	11,072
Male	139,157	7,274	343	20,687	45,301	4,224	3,879	20,964	2,567	679	31,296	1,943
Female	467,368	27,685	1,595	109,389	118,014	10,894	5,783	68,653	9,104	3,963	103,158	9,129
Activities of extra	territorial org	ganizations	s and bod	ies Descrip	tion							
Total	78,337	6,757	1,122	8,574	19,391	2,351	742	11,353	1,990	514	24,828	716
Male	47,522	4,891	543	5,221	10,780	1,943	534	7,846	1,512	343	13,477	432
Female	30,815	1,866	579	3,353	8,611	408	209	3,507	478	171	11,351	284

Table 5.9 Employed Population of Major Towns Aged Ten Years and Above by Major Industrial Divisions and Sex: 2015 Part A

Table 5.9 Er	mployed	Population	on of Ma	jor Towns /	Aged Ten	Years and Abo		or Industria	al Divisions	and Sex: 20	15 Part A	
						Major Industria	l Divisions					
								Wholesale				
					Electricity,	Water supply;		and retail				
					gas, steam	sewerage, waste		trade; repair		Accommodation		Financial
Major Towns		Agriculture,	Mining		and air	management and		of motor		and food	Information	and
and Sex		forestry	and		conditioning	remediation		vehicles and	Transportation	service	and	insurance
	Total	and fishing	quarrying	Manufacturing	supply	activities	Construction	motorcycles	and storage	activities	communication	activities
COUNTRY TOTAL												
Total	2,560,286	126,203	11,274	332,689	9,308	14,610	202,382	511,861	179,247	136,960	46,100	77,477
Male	1,443,652	83,438	8,998	201,868	7,577	9,235	164,762	264,792	167,503	39,082	22,527	44,131
Female	1,116,634	42,765	2,276	130,821	1,731	5,375	37,620	247,069	11,744	97,878	23,573	33,347
MEKELE TOWN	, ,		,	,	,		•		,	•	,	,
Total	131,641	4,061	1,017	20,108	1,157	1,005	12,149	22,085	10,634	5,119	1,995	2,828
Male	70,849	2,280	884	11,309	1,157	915	10,038	9,827	10,047	994	1,457	1,385
Female	60,792	1,781	133	8,799		90	2,111	12,258	587	4,124	538	1,443
ASAYITA TOWN	**/	.,		27			_,	,		.,		.,
Total	12,819	922	_	1,897	63	95	888	3,942	552	399	302	167
Male	7,356	868	-	1,437	63	63	815	1,493	543	-	184	110
Female	5,463	54	_	461	-	32	73	2,449	10	399	118	57
GONDER TOWN	0,100	0.1		101		02	, 0	2,117	10	0,,	110	0,
Total	162,404	48,472	2,986	9,662	203	300	13,280	23,049	6,419	4,903	281	2,382
Male	85,621	34,894	2,611	5,101	101	150	10,170	8,400	6,419	1,421	281	1,356
Female	76,782	13,578	375	4,561	101	150	3,110	14,649	-	3,482	-	1,026
DESSIE TOWN	70,702	.0,0.0	0.0	.,00.		.00	3,	,		0,102		.,020
Total	71,530	1,735	245	7,840	583	781	4,049	22,251	4,150	3,025	1,866	1,341
Male	43,161	1,178	184	5,597	583	623	3,536	11,033	4,089	815	1,410	672
Female	28,369	556	61	2,243	-	158	512	11,218	61	2,210	457	669
BAHIR DAR TOWN		330	01	2,243	-	130	312	11,210	01	2,210	437	007
Total	124,310	31,993	515	11,261	809	1,526	12,292	14,569	5,537	11,361	999	2,212
Male	66,389	19,472	332	5,632	442	928	8,189	7,111	5,537	2,169	518	1,413
Female	57,921	12,522	183	5,628	367	597	4,103	7,458	-	9,192	480	799
BISHOFTU TOWN		F F 40	404	40.000		201	. 074	40.040	4 440	2 222	000	4 404
Total	57,994	5,540	491	12,022	-	286	6,271	10,043	4,418	3,008	229	1,181
Male	33,733	2,735	255	6,869	-	171	5,784	3,880	4,205	1,014	114	613
Female	24,261	2,805	236	5,153	-	114	487	6,164	213	1,994	115	568
ADAMA TOWN												
Total	117,828	5,025	717	18,967	245	369	9,484	28,557	8,914	8,886	1,365	1,223
Male	69,953	4,124	597	10,244	245	123	7,868	15,856	8,060	2,025	610	619
Female	47,874	901	120	8,723	-	246	1,616	12,701	854	6,860	755	604
JIMMA TOWN												
Total	75,289	1,806	376	6,321	399	245	5,761	15,993	3,464	4,799	726	1,808
Male	42,862	1,294	97	4,357	337	245	5,193	9,379	3,341	1,327	367	1,145
Female	32,428	512	279	1,964	62	-	568	6,614	123	3,472	359	663
Table 5.9 Co	nt'd										Part A	
	Major Indu	strial Division	S									
		_										

Major Towns and Sex		Agriculture,	Mining		Electricity, gas, steam and air	Water supply; sewerage, waste management and		Wholesale and retail trade; repair of motor		Accommodation and food	Information	Financial and
dia Sex		forestry	and		conditioning	remediation		vehicles and	Transportation	service	and	insurance
	Total		quarrying	Manufacturing		activities	Construction		and storage	activities	communication	activities
SHASHEMENE TO	WN											
Total	54,374	2,030	343	6,476	105	67	3,330	14,504	5,312	3,221	702	1,096
Male	32,171	1,285	269	4,464	105	67	2,986	6,665	4,984	1,342	453	449
Female	22,203	745	74	2,012	-	-	344	7,839	328	1,879	249	647
JIJIGA TOWN												
Total	40,815	874	40	1,805	91	722	3,168	11,374	2,650	1,041	338	1,035
Male	25,947	633	-	1,275	40	366	3,035	5,190	2,610	388	166	687
Female	14,868	241	40	530	50	356	133	6,184	40	653	172	348
ASOSA TOWN												
Total	21,112	309	172	1,095	68	143	2,672	4,103	1,261	1,135	78	197
Male	12,786	280	100	753	68	107	2,263	1,660	1,261	313	49	197
Female	8,326	29	72	342	-	36	409	2,443	-	822	29	-
HAWASSA TOWN												
Total	124,702	2,211	95	15,407	334	1,300	12,198	26,533	5,825	12,292	2,320	2,778
Male	73,264	1,192	95	9,471	334	923	10,798	14,623	5,731	4,673	1,030	1,725
Female	51,439	1,018	-	5,936	-	376	1,399	11,910	94	7,619	1,290	1,054
GAMBELLA TOWN		4 007		2/5			4 007					
Total	27,965	1,307	80	965	114	394	1,927	3,526	1,380	1,947	289	1,342
Male	15,295	1,006	40	457	22	180	1,568	1,805	1,276	333	184	870
Female	12,670	301	40	507	93	213	359	1,721	103	1,614	105	472
HARAR TOWN	E0 100	F0/		4.450	01	450	4.007	17.101	2 /71	0.400	F40	/50
Total	59,109	536	-	4,650	91	453	4,027	17,121	3,671	2,133	518	659
Male	29,900	307	-	2,159	49 42	354 99	3,589	6,207	3,312	460	326	339
Female ADDIS ABABA CIT	29,210 V ADMINIST	230	-	2,491	42	99	438	10,914	358	1,673	192	320
Total	1,362,069	15,768	4,197	200,629	4,103	6,018	101,468	262,031	105,937	68,049	33,221	55,125
Male	772,367	10,138	3,534	125,317	3,088	3,838	80,907	148,267	97,414	20,943	15,033	31,129
Female	589,703	5,631	663	75,317	1,015	2,181	20,561	113,764	8,523	47,106	18,188	23,996
DIRE DAWA TOWN		3,031	003	75,515	1,013	2,101	20,301	113,704	0,323	47,100	10,100	23,770
Total	116,325	3,613		13,585	942	907	9,419	32,179	9,124	5,642	871	2,102
Male	61,997	1,753	_	7,426	942	182	8,024	13,396	8,674	865	345	1,421
Female	54,327	1,733	_	6,159	742	725	1,395	18,784	450	4,777	526	681
1 CITIAIC	J4,JZ1	1,000		0,137	-	723	1,373	10,704	430	4,///	520	001

Table 5.9 Employed Population of Major Towns Aged Ten Years and Above by Major Industrial Divisions and Sex: 2015 Part B

Table 5.9 Emple	year opur	anon or maje	n 10miis Agel	a ron rouls (strial Divisions		11310113 6	iliu Sex. 2015 Pai	
				Public	iviajoi mut	Journal Divisions			Activities of households as employers;	Activities of
Major Towns and Sex	Real estate activities	Professional, scientific and technical activities	Administrative and support service activities	administration and defense; compulsory social security	Education	Human health and social work activities	Arts, entertainment and recreation	Other service activities	undifferentiated goods- and services-producing activities of households for own use	extraterritorial organizations and bodies Description
MAJOR TOWN TOTAL										_
Total Male Female MEKELE TOWN	4,505 2,668 1,837	58,863 37,437 21,426	60,588 35,824 24,764	118,520 75,693 42,827	145,402 72,551 72,851	75,484 27,918 47,566	24,203 14,305 9,898	141,559 88,555 53,005	245,391 52,130 193,261	37,659 22,659 15,000
Total Male Female ASAYITA TOWN	226 90 135	2,059 1,477 582	1,670 765 905	7,968 4,583 3,385	8,165 3,832 4,334	4,953 1,873 3,080	2,718 1,719 998	6,670 1,949 4,721	13,259 2,866 10,393	1,793 1,401 392
Total Male Female GONDER TOWN	- - -	300 124 176	179 39 140	771 586 185	771 395 376	305 163 143	28 28 -	372 214 157	735 125 609	130 107 24
Total Male Female DESSIE TOWN	- - -	591 487 103	739 358 381	3,144 2,688 456	5,969 2,395 3,575	1,255 310 945	383 280 103	6,872 4,517 2,355	30,672 3,148 27,523	844 535 309
Total Male Female BAHIR DAR TOWN	62 - 62	612 612 -	485 425 60	3,635 2,202 1,433	5,813 3,828 1,985	2,616 1,579 1,037	343 160 183	4,845 3,703 1,142	4,962 763 4,199	292 171 122
Total Male Female BISHOFTU TOWN	- -	580 461 119	2,139 1,424 714	6,114 4,036 2,079	7,615 3,608 4,007	2,627 899 1,727	1,268 618 651	4,700 2,680 2,020	4,930 93 4,836	1,263 825 437
Total Male Female ADAMA TOWN	- - -	671 292 380	762 452 310	2,272 1,732 540	3,102 1,791 1,312	1,408 661 747	472 373 98	2,411 1,164 1,247	2,924 1,401 1,523	483 228 255
Total Male Female	- - -	2,611 2,278 333	2,294 1,793 501	2,888 2,643 245	5,840 2,876 2,964	2,944 981 1,963	606 364 243	6,113 4,147 1,966	9,696 3,767 5,930	1,083 732 350

Table 5.9 Cont'd Part B

Major Industrial Divisions

Major Towns and Sex	Real estate activities	Professional, scientific and technical activities	Administrative and support service activities	Public administration and defense; compulsory social security	Education	Human health and social work activities	Arts, entertainment and recreation	Other service activities	Activities of households as employers; undifferentiated goods- and services-producing activities of households for own use	Activities of extraterritorial organizations and bodies Description
JIMMA TOWN	activities	activities	Service activities	Social Security	Luucation	activities	and recreation	activities	TOI OWITUSE	Description
Total	_	610	386	2,744	10,477	3,940	373	5,088	9,394	579
Male	_	457	201	1,905	4,547	1,963	290	4,174	2,056	185
Female	_	153	185	839	5,930	1,976	83	914	7,338	394
SHASHEMENE TOWN		100	100	007	0,700	1,770	00	711	7,000	371
Total	-	1,119	1,777	693	1,738	1,412	257	5,943	3,821	427
Male	-	870	1,050	458	1,004	532	176	3,605	980	427
Female	_	249	727	235	734	880	80	2,338	2,841	-
JIJIGA TOWN		2.,		200		000		2,000	2/0	
Total	-	873	577	3,586	3,274	1,197	173	3,005	3,838	1,153
Male	-	690	476	2,676	2,147	, 775	173	2,431	1,279	910
Female	-	183	100	910	1,127	422	-	574	2,559	243
ASOSA TOWN										
Total	-	637	39	4,846	1,648	601	159	890	744	316
Male	-	464	-	2,960	967	274	123	704	58	184
Female	-	172	39	1,886	681	327	36	186	686	132
HAWASSA TOWN										
Total	-	1,321	1,964	9,387	11,338	4,688	1,075	4,965	6,611	2,060
Male	-	977	1,073	5,743	6,342	2,009	383	3,640	779	1,721
Female	-	344	891	3,644	4,996	2,680	692	1,324	5,832	339
GAMBELLA TOWN										
Total	-	840	1,000	3,153	1,976	1,457	189	1,265	3,637	1,179
Male	-	540	620	1,917	1,193	632	144	632	896	981
Female	-	300	381	1,236	783	824	45	633	2,741	198
HARAR TOWN										
Total	-	542	648	7,190	4,977	3,093	287	3,357	4,642	514
Male	-	259	139	4,963	2,364	1,318	81	2,652	679	343
Female	-	283	509	2,227	2,613	1,774	206	705	3,963	171
ADDIS ABABA CITY AD										
Total	4,218	43,792	44,865	54,729	66,485	39,654	15,350	77,147	134,455	24,828
Male	2,578	25,991	26,313	32,925	32,080	12,593	8,957	46,549	31,296	13,477
Female	1,640	17,801	18,552	21,804	34,405	27,060	6,393	30,598	103,158	11,351
DIRE DAWA TOWN		1 70/	10/5	F 200	. 011	2.225	F04	7.01/	11.070	747
Total	-	1,706	1,065	5,399	6,211	3,335	521	7,916	11,072	716
Male	-	1,459	696	3,677	3,182	1,355	435	5,792	1,943	432
Female	-	248	369	1,722	3,029	1,980	86	2,124	9,129	284

Table 5.10 Employed Population of Urban Areas Aged Ten Years and Above by Minor Industrial Divisions, Sex and Region – COUNTRY TOTAL: 2015

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Total Employed Persons												
Total	7,083,094	477,813	131,439	1,449,091	2,102,417	119,082	107,053	1,096,887	61,809	59,109	1,362,069	116,325
Male	3,963,143	253,254	75,452	795,183	1,213,723	67,114	58,730	603,653	31,771	29,900	772,367	61,997
Female	3,119,950	224,559	55,987	653,908	888,694	51,967	48,323	493,233	30,038	29,210	589,703	54,327
SECTION A: AGRICULTUR	E, FOREST	RY AND F	ISHING									
Total	567,067	31,685	34,223	185,318	159,457	14,676	21,054	94,138	6,600	536	15,768	3,613
Male	382,749	21,450	25,357	129,045	107,658	8,462	11,711	62,298	4,570	307	10,138	1,753
Female	184,319	10,236	8,866	56,273	51,799	6,213	9,343	31,840	2,030	230	5,631	1,860
DIVISION 01 CROP ANI	D ANIMAL	PRODU(CTION, F	HUNTING A	AND RELA	TED SE	RVICE ACTI\	/ITIES				
Total	534,794	30,854	29,021	179,956	153,482	13,279	20,492	85,234	5,989	445	12,589	3,452
Male	360,824	20,618	20,385	124,790	103,802	8,068	11,359	57,377	4,147	265	8,261	1,753
Female	173,970	10,236	8,636	55,165	49,680	5,212	9,134	27,858	1,842	180	4,329	1,699
Growing of non-perennial	crops											
Total	159,601	15,466	18,736	34,011	52,913	1,322	1,602	29,166	2,596	50	3,385	355
Male	125,128	13,108	14,293	27,571	41,568	1,148	1,100	21,714	1,574	-	2,787	265
Female	34,473	2,357	4,444	6,440	11,344	174	502	7,452	1,022	50	598	91
Growing of perennial crops	S											
Total	43,582	2,565	10	4,193	20,431	4,215	1,640	7,414	744	-	2,297	73
Male	25,534	816	10	2,181	10,941	2,357	1,144	6,410	698	-	978	-
Female	18,048	1,749	-	2,012	9,490	1,858	496	1,004	46	-	1,320	73
Plant propagation												
Total	15,725	-	16	1,433	11,043	-	116	2,649	121	-	346	-
Male	8,075	-	16	990	5,067	-	116	1,805	81	-	-	-
Female	7,650	-	-	443	5,976	-	-	844	40	-	346	-
Animal production												
Total	155,016	9,788	4,116	48,872	37,411	5,512	10,251	31,351	1,598	49	3,472	2,597
Male	88,641	3,659	2,273	30,475	24,683	2,688	4,490	16,396	1,165	-	1,751	1,062
Female	66,375	6,129	1,843	18,397	12,728	2,824	5,761	14,955	433	49	1,721	1,535
Mixed farming												
Total	123,228	2,548	-	87,615	19,877	218	6,039	5,969	327	208	-	426
Male	85,666	2,548	-	61,182	13,990	218	3,807	3,164	163	166	-	426
Female	37,562	-	-	26,433	5,887	-	2,231	2,806	164	41	-	-

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Support activities to agricu	lture and po	st-harvest	crop activ	vities								
Total	37,643	487	6,143	3,831	11,808	2,013	844	8,685	603	139	3,089	-
Male	27,780	487	3,794	2,390	7,553	1,657	701	7,888	467	99	2,745	-
Female	9,863	-	2,349	1,441	4,254	356	143	797	137	41	344	-
Hunting, trapping and relat	ed service a	activities										
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	_
Female	-	-	-	-	-	-	-	-	-	-	-	_
DIVISION 02 FORESTRY A	ND LOGGI	NG										
Total	29,157	215	4,984	4,799	5,378	1,396	562	8,324	398	91	2,848	161
Male	19,207	215	4,754	3,759	3,260	394	353	4,342	211	42	1,877	-
Female	9,950	-	230	1,040	2,118	1,002	209	3,982	188	49	971	161
Silviculture and other fores	try activities	;										
Total	3,413	-	654	891	618	50	-	301	-	-	899	_
Male	3,138	-	654	891	618	50	-	301	-	-	625	-
Female	275	-	-	-	-	-	-	-	-	-	275	-
Logging												
Total	13,927	-	4,167	590	1,042	1,214	-	6,353	-	49	350	161
Male	8,199	-	3,937	590	485	213	-	2,974	-	-	-	-
Female	5,729	-	230	-	557	1,002	-	3,380	-	49	350	161
Gathering of non-wood for	est products	;										
Total	4,176	-	28	684	2,278	-	-	724	188	-	275	-
Male	1,728	-	28	-	1,303	-	-	122	-	-	275	-
Female	2,449	-	-	684	975	-	-	603	188	-	-	-
Support services to forestr												
Total	7,641	215	136	2,634	1,440	132	562	945	211	42	1,324	-
Male	6,143	215	136	2,278	854	132	353	945	211	42	978	_
Female	1,498	-	-	356	587	-	209	-	-	-	347	-
DIVISION 03 FISHING AND		TURE										
Total	3,116	616	218	563	597	_	-	579	213	_	331	_
Male	2,718	616	218	496	597	_	-	579	213	_	-	_
Female	398	-	-	67	-	_	-	-	-	_	331	_

Table 5.10 Cont'd

Total	Minor Industrial Divisions						Re	egion					
Fishing								Benishangul-					
Total 1,695 - 218 496 189 579 213 Male 1,695 - 218 496 189 579 213 Male 1,695 - 218 496 189 579 213 579 213		Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Male Female 1,695 - 218 496 189													
Female -			-				-	-			-	-	-
Aquaculture		1,695	-	218	496	189	-	-	579	213	-	-	-
Total 1,421 616 - 67 408 - - - - 331 Male 1,023 616 - - 408 -<		-	-	-	-	-	-	-	-	-	-	-	-
Male 1,023 616 - - 408 - <t< td=""><td>Aquaculture</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>	Aquaculture												
Female 398 - - 67 - - - - - - - 331 SECTION B: MINING AND QUARRYING		1,421		-	67		-	-	-	-	-	331	-
Total 35,165 3,505 111 9,019 15,908 234 552 1,459 182 - 4,197	Male	1,023	616	-	-	408	-	-	-	-	-	-	-
Total 35,165 3,505 111 9,019 15,908 234 552 1,459 182 - 4,197 Male 28,221 3,143 111 6,543 12,977 194 480 1,099 142 - 3,534 Female 6,944 362 - 2,476 2,932 40 72 360 40 - 663 DIVISION 05 MINING OF COAL AND LIGNITE Total 2,834 - 6 614 1,829 - 5 53 - 53 - 337 Male 2,219 - 614 1,829 - 5 53 - 53 - 337 Female 614 - 614 1,306 - 5 53 - 53 - 337 Male 1,973 - 614 1,306 - 5 53 - 53 - 53 - 614 Mining of hard coal Total 1,973 - 614 1,306 - 5 53 - 53 - 53 - 614 Mining of lignite Total 860 - 5 5614 - 523 - 523 - 5 5 5 - 360 - 5 - 337 Male 860 - 5 523 - 523 - 5 - 360 - 5 - 520 DIVISION 06 EXTRACTION OF CRUDE PETROLEUM AND NATURAL GAS				-	67	-	-	-	-	-	-	331	-
Male 28,221 3,143 111 6,543 12,977 194 480 1,099 142 - 3,534 Female 6,944 362 - 2,476 2,932 40 72 360 40 - 663 DIVISION 05 MINNING OF COAL AND LIGNITE Total 2,834 - - 614 1,829 - - - 53 - 337 Male 2,219 - - 614 - - - 53 - 337 Female 614 - - 614 -	SECTION B: MINING AND Q		ì										
Female 6,944 362 - 2,476 2,932 40 72 360 40 - 663 DIVISION 05 MINING OF COAL AND LIGNITE Total 2,834 - - 614 1,829 - - - 53 - 337 Male 2,219 - - 1,829 - - - 53 - 337 Female 614 - - 614 -	Total	35,165	3,505		9,019	15,908	234		1,459		-	4,197	-
DIVISION 05 MINING OF COAL AND LIGNITE Total 2,834 614 1,829 53 - 337 Male 2,219 1,829 53 - 337 Female 614 614 53 - 53 - 337 Mining of hard coal Total 1,973 614 1,306 5 53 - 53 - 5 Male 1,359 614 1,306 5 53 - 53 - 5 Female 614 - 614 614 5 53 - 53 - 5 Female 614 614 5 53 - 53 - 5 Mining of lignite Total 860 5 523 3 523 3 37 Male 860 5 523 3 37 Male 860 5 523 3 37 Male 860 3 523 3 37 Male 860 3 523 3 37 Male 860 3 523 3 360 337 Male 860 3 523 3 360	Male	28,221	3,143	111	6,543	12,977	194	480	1,099	142	-	3,534	-
Total 2,834 614 1,829 53 - 337 Male 2,219 1,829 53 - 337 Female 614 614 614 53 53 614 Mining of hard coal Total 1,973 614 1,306 53 53 Male 1,359 614 1,306 53 53 Female 614 614 53 53 Mining of lignite Total 860 614 523 337 Male 860 523 337 Male 860 523 337 DIVISION 06 EXTRACTION OF CRUDE PETROLEUM AND NATURAL GAS Total 714 355 360				-	2,476	2,932	40	72	360	40	-	663	-
Male 2,219 - - 1,829 - - - 53 - 337 Female 614 - - 614 -	DIVISION 05 MINING OF CO	OAL AND LI	GNITE										
Female 614 - - 614 -	Total	2,834	-	-	614	1,829	-	-	-	53	-	337	-
Mining of hard coal Total 1,973 - - 614 1,306 - - - 53 - - Male 1,359 - - 1,306 - - - 53 - - Female 614 - - 614 - <td>Male</td> <td>2,219</td> <td>-</td> <td>-</td> <td>-</td> <td>1,829</td> <td>-</td> <td>-</td> <td>-</td> <td>53</td> <td>-</td> <td>337</td> <td>-</td>	Male	2,219	-	-	-	1,829	-	-	-	53	-	337	-
Total 1,973 - - 614 1,306 - - - 53 - - Male 1,359 - - 1,306 - - - 53 - - Female 614 - - 614 - </td <td></td> <td>614</td> <td>-</td> <td>-</td> <td>614</td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td>-</td> <td>-</td>		614	-	-	614	-	-	-	-	-	-	-	-
Male 1,359 - - 1,306 - - - 53 - - Female 614 - - 614 - 337 - 337 - - - - - - - - - - - - <td< td=""><td>Mining of hard coal</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></td<>	Mining of hard coal												
Female 614 - - 614 -			-	-	614		-	-	-	53	-	-	-
Mining of lignite Total 860 - - 523 - - - - 337 Male 860 - - 523 - - - - - 337 Female - <	Male		-	-	-	1,306	-	-	-	53	-	-	-
Total 860 - - 523 - - - - 337 Male 860 - - 523 - - - - 337 Female - <t< td=""><td>Female</td><td>614</td><td>-</td><td>-</td><td>614</td><td>-</td><td>-</td><td>-</td><td>-</td><td>-</td><td>-</td><td>-</td><td>-</td></t<>	Female	614	-	-	614	-	-	-	-	-	-	-	-
Male 860 - - 523 - - - - - 337 Female -	Mining of lignite												
Female	Total	860	-	-	-	523	-	-	-	-	-	337	-
DIVISION 06 EXTRACTION OF CRUDE PETROLEUM AND NATURAL GAS Total 714 355 360	Male	860	-	-	-	523	-	-	-	-	-	337	-
Total 714 355 360	Female	-	-	-	-	-	-	-	-	-	-	-	-
	DIVISION 06 EXTRACTION	OF CRUDE	PETRO	LEUM AN	D NATURA	L GAS							
Male 34 34	Total	714	-	-	-	355	-	-	360	-	-	-	-
IVIAIG 54 34	Male	34	-	-	_	34	-	-	-	-	-	-	-
Female 681 321 360	Female	681	-	-	-	321	-	-	360	-	-	-	-
Extraction of crude petroleum	Extraction of crude petrole	um											
Total 34 34	Total	34	-	-	-	34	-	-	-	-	-	-	-
Male 34 34	Male	34	-	-	-	34	-	-	-	-	-	-	-
Female	Female	-	-	-	-	-	-	-	-	-	-	-	-
Extraction of natural gas	Extraction of natural gas												
Total 681 321 360	Total	681	-	-	-	321	-	-	360	-	-	-	-
Male	Male	-	-	-	-	-	-	-	-	-	-	-	-
Female 681 321 360	Female	681	-	-	-	321	-	-	360	-	-	-	-

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
DIVISION 07 MINING OF ME												
Total	2,318	-	-	165	1,049	-	71	-	-	-	1,033	-
Male	2,099	-	-	165	872	-	29	-	-	-	1,033	-
Female	220	-	-	-	177	-	43	-	-	-	-	-
Mining of iron ores												
Total	911	-	-	104	415	-	43	-	-	-	350	-
Male	749	-	-	104	295	-	-	-	-	-	350	-
Female	162	-	-	-	120	-	43	-	-	-	-	-
Mining of non-ferrous meta												
Total	1,407	-	-	61	634	-	29	-	-	-	683	-
Male	1,350	-	-	61	577	-	29	-	-	-	683	-
Female	57	-	-	-	57	-	-	-	-	-	-	-
DIVISION 08 OTHER MININ				0.000	40.007	400	400	4 000	400		4.040	
Total	27,279	3,276	111	8,239	12,007	120	480	1,099		-	1,818	-
Male	22,588	2,914	111	6,378	9,981	80	451	1,099		-	1,486	-
Female	4,691	362	-	1,861	2,026	40	29	-	40	-	332	-
Quarrying of stone, sand a	•	000		0.070	0.700	00	400		400		004	
Total	14,998	663	-	6,272	6,782	80	408	-	129	-	664	-
Male	12,312	663	-	5,555	5,186	80	408	-	89	-	332	-
Female	2,685	-	-	717	1,596	-	-	-	40	-	332	-
Mining and quarrying not e												
Total	12,281	2,613	111	1,967	5,225	40	72	1,099	-	-	1,154	-
Male	10,275	2,251	111	823	4,795	-	43	1,099	-	-	1,154	-
Female	2,006	362	-	1,144	430	40	29	-	-	-	-	-
DIVISION 09 MINING SUPF			VIIIES									
Total	2,020	229	-	-	669	114	-	-	-	-	1,008	-
Male	1,282	229	-	-	261	114	-	-	-	-	678	-
Female	738	-	-	-	408	-	-	-	-	-	331	-
Support activities for petrol		tural gas e	extraction									
Total	114	-	-	-	-	114	-	-	-	-	-	-
Male	114	-	-	-	-	114	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Support activities for other	-	quarrying										
Total	1,906	229	-	-	669	-	-	-	-	-	1,008	-
Male	1,168	229	-	-	261	-	-	-	-	-	678	-
Female	738	-	-	-	408	-	-	-	-	-	331	-

Table 5.10 Cont'd

Minor Industrial Divisions						Re	gion					
and sex							Benishangul-				Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
SECTION C: MANUFACTUR	RING											
Total	929,847	72,036	19,836	180,459	300,099	4,450	10,875	119,318	3,911	4,650	200,629	13,585
Male	489,346	41,861	10,157	89,051	157,750	2,376	3,758	47,621	1,871	2,159	125,317	7,426
Female	440,501	30,175	9,679	91,408	142,349	2,074	7,117	71,697	2,040	2,491	75,313	6,159
DIVISION 10 MANUFACTU	RE OF FOO	D PRODU	JCTS									
Total	231,262	11,534	14,303	28,349	93,257	2,033	1,239	40,146	1,347	1,957	32,206	4,891
Male	90,628	4,782	8,433	13,873	38,857	568	799	9,834	548	513	10,972	1,450
Female	140,634	6,752	5,870	14,477	54,400	1,466	440	30,312	799	1,444	21,234	3,441
Processing and preserving	of meat											
Total	8,019	918	-	626	1,350	279	86	997	49	-	3,714	-
Male	7,186	765	-	535	1,350	121	86	901	49	-	3,378	-
Female	833	153	-	91	-	158	-	96	-	-	336	-
Processing and preserving	of fish, crus	taceans a	nd mollu	scs								
Total	223	133	-	90	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	223	133	-	90	-	-	-	-	-	-	-	-
Processing and preserving	of fruit and	vegetable	S									
Total	6,710	90	16	2,802	2,047	-	-	96	-	-	1,658	-
Male	2,052	90	16	1,187	759	-	-	-	-	-	-	-
Female	4,658	-	-	1,615	1,288	-	-	96	-	-	1,658	-
Manufacture of vegetable a	and animal c	ils and fa	its									
Total	3,212	-	-	1,278	1,100	80	-	-	49	261	342	102
Male	2,880	-	-	1,278	1,044	80	-	-	49	87	342	-
Female	332	-	-	-	56	-	-	-	-	174	-	102
Manufacture of dairy produ												
Total	5,648	690	-	591	2,101	212	-	907	-	-	625	522
Male	1,632	404	-	-	179	-	-	416	-	-	287	346
Female	4,016	286	-	591	1,921	212	-	491	-	-	337	176
Manufacture of grain mill p	roducts, stai	ches and	starch pr									
Total	19,557	1,277	109	4,596	7,525	84	378	3,572	-	41	1,596	380
Male	11,787	751	-	2,670	5,306	-	349	2,233	-	-	287	190
Female	7,770	526	109	1,926	2,218	84	28	1,339	-	41	1,308	190
Manufacture of other food	products											
Total	185,960	8,425	13,348	18,365	78,830	1,378	775	33,774	1,250	1,656	24,272	3,887
Male	64,237	2,771	8,210	8,203	29,971	366	363	5,884	451	426	6,678	914
Female	121,723	5,653	5,139	10,162	48,859	1,012	412	27,890	799	1,230	17,594	2,973

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Manufacture of prepared a	nimal feeds											
Total	1,934	-	830	-	304	-	-	800	-	-	-	-
Male	854	-	207	-	246	-	-	400	-	-	-	-
Female	1,080	-	622	-	58	-	-	400	-	-	-	-
DIVISION 11 MANUFACTU	RE OF BEV	ERAGES										
Total	144,645	4,058	2,717	45,760	38,351	92	4,532	36,926	1,087	573	8,900	1,648
Male	23,290	1,205	428	6,117	6,045	-	206	4,249	-	279	3,817	944
Female	121,355	2,853	2,290	39,643	32,307	92	4,326	32,677	1,087	294	5,083	704
Manufacture of beverages												
Total	144,645	4,058	2,717	45,760	38,351	92	4,532	36,926	1,087	573	8,900	1,648
Male	23,290	1,205	428	6,117	6,045	-	206	4,249	-	279	3,817	944
Female	121,355	2,853	2,290	39,643	32,307	92	4,326	32,677	1,087	294	5,083	704
DIVISION 12 MANUFACTU		ACCO PR										
Total	1,879	-	331	684	96	40	-	96	-	-	632	-
Male	819	-	-	91	96	-	-	-	-	-	632	-
Female	1,060	-	331	593	-	40	-	96	-	-	-	-
Manufacture of tobacco pro												
Total	1,879	-	331	684	96	40	-	96	-	-	632	-
Male	819	-	-	91	96	-	-	-	-	-	632	-
Female	1,060	-	331	593	-	40	-	96	-	-	-	-
DIVISION 13 MANUFACTU												
Total	187,723	24,887	840	42,991	67,889	236	134	8,517	148	655	39,565	1,860
Male	93,579	10,402	-	21,461	35,927	50	57	3,804	85	172	21,177	445
Female	94,143	14,485	840	21,531	31,963	186	77	4,712	63	484	18,388	1,415
Spinning, weaving and finis	•											
Total	121,306	15,787	415	29,271	41,515	-	106	3,855	-	131	28,705	1,521
Male	62,488	6,275	-	11,568	25,443	-	29	2,326	-	81	16,411	356
Female	58,817	9,512	415	17,704	16,072	-	77	1,528	-	50	12,294	1,165
Manufacture of other textile												
Total	66,417	9,101	425	13,720	26,374	236	29	4,662	148	524	10,861	338
Male	31,091	4,127	-	9,893	10,484	50	29	1,478	85	91	4,767	88
Female	35,326	4,974	425	3,827	15,890	186	-	3,184	63	433	6,094	250
Division 14 Manufacture of v	wearing appa	arel										
Total	40,836	3,915	10	10,294	9,309	316	380	6,829	130	49	9,193	412
Male	29,627	2,748	10	9,122	7,409	232	380	5,536	130	49	3,771	240
Female	11,210	1,167	-	1,171	1,899	84	-	1,294	-	-	5,422	173

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Manufacture of wearing app	parel, excep	t fur appa	rel									
Total	31,887	3,146	10	10,088	7,319	316	380	2,138	130	49	7,899	412
Male	23,244	2,207	10	9,122	5,792	232	380	1,644	130	49	3,438	240
Female	8,643	938	-	966	1,527	84	-	494	-	-	4,461	173
Manufacture of articles of fu	ur											
Total	1,152	-	-	-	523	-	-	-	-	-	629	-
Male	523	-	-	-	523	-	-	-	-	-	-	-
Female	629	-	-	-	-	-	-	-	-	-	629	-
Manufacture of knitted and	crocheted a	apparels D	Description	า								
Total	7,797	770		205	1,466	-	-	4,691	-	-	666	-
Male	5,859	541	-	-	1,094	-	-	3,892	-	-	333	-
Female	1,938	229	-	205	372	-	-	799	-	-	333	-
DIVISION 15 MANUFACTUR			D RELAT	ED PRODU	ICTS							
Total	26,114	1,769	-	1,242	3,999	158	-	566	-	87	18,293	-
Male	12,674	903	-	-	1,842	158	-	343	-	87	9,341	-
Female	13,440	867	-	1,242	2,157	-	-	222	-	-	8,952	-
Tanning and dressing of lea			luggage,		saddler and	d harness;	dressing and c					
Total	5,542	1,632	-	1,242	1,687	-	-	445	-	-	536	-
Male	2,076	765	-	-	899	-	-	222	-	-	189	-
Female	3,466	867	-	1,242	788	-	-	222	-	-	347	-
Manufacture of footwear												
Total	20,571	137	-	-	2,312		-	121	-	87	17,757	-
Male	10,598	137	-	-	943	158	-	121	-	87	9,152	-
Female	9,974	-	-	. .	1,369		-	-	-	-	8,605	-
DIVISION 16 MANUFACTUI FURNITURE;MANUFACTUI							XCEPT					
Total	52,153	2,277	98	7,117	18,051	-	2,659	9,395	121	335	11,851	249
Male	45,356	2,277	98	5,873	15,717	-	1,142	9,274	121	335	10,270	249
Female	6,797	-	-	1,244	2,334	-	1,517	121	-	-	1,581	-
Sawmilling and planning of	wood											
Total	18,176	695	23	2,093	6,193	-	682	2,924	121	49	5,237	161
Male	17,512	695	23	2,093	6,193	-	682	2,924	121	49	4,572	161
Female	665	-	-	· -	· -	-	-		-	-	665	-
Manufacture of products of		straw an	d plaiting	materials								
Total	33,977	1,582	76	5,025	11,858	_	1,977	6,471	_	286	6,615	88
Male	27,845	1,582	76	3,781	9,524	_	460	6,350	_	286	5,698	88
Female	6,132	,	-	1,244	2,334	-	1,517	121	_		917	-

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
DIVISION 17 MANUFACTUI	RE OF PAP	ER AND	PAPER PI	RODUCTS								
Total	2,473	-	-	185	995	-	-	-	-	-	1,292	-
Male	1,158	-	-	-	471	-	-	-	-	-	687	-
Female	1,314	-	-	185	524	-	-	-	-	-	605	-
Manufacture of paper and p	paper produ	cts										
Total	2,473	-	-	185	995	-	-	-	-	-	1,292	-
Male	1,158	-	-	-	471	-	-	-	-	-	687	-
Female	1,314	-	-	185	524	-	-	-	-	-	605	-
DIVISION 18 PRINTING AN	ID REPROD	UCTION	OF RECC	RDED MEI	DIA							
Total	5,646	454	-	749	435	-	92	524	-	81	3,310	-
Male	2,240	227	-	60	-	-	-	524	-	-	1,428	-
Female	3,407	227	-	689	435	-	92	-	-	81	1,882	-
Printing and service activities		printing										
Total	4,736	454	-	153	122	-	92	524	-	81	3,310	-
Male	2,240	227	-	60	-	-	-	524	-	-	1,428	-
Female	2,497	227	-	93	122	-	92	-	-	81	1,882	-
Reproduction of recorded n	nedia											
Total	910	-	-	596	314	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	910	-	-	596	314		-	-	-	-	-	-
DIVISION 19 MANUFACTUI	RE OF COK	E AND R	EFINED F	PETROLEU								
Total	84	-	-	-	84	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	84	-	-	-	84	-	-	-	-	-	-	-
Manufacture of coke oven	products											
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of refined petr	roleum prod	ucts										
Total	84	-	-	-	84	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	84	-	-	-	84	-	-	-	-	-	-	-
DIVISION 20 MANUFACTUI	RE OF CHE	MICALS	AND CHE	MICAL PRO	DDUCTS							
Total	14,562	153	16	1,209	9,844	-	-	412	-	-	2,928	-
Male	10,261	153	-	536	7,140	-	-	412	-	-	2,021	-

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Manufacture of basic chem		ers and ni	itrogen co	mpounds, p		synthetic	rubber in prima	ary forms		•	'	
Total	4,212	153	16	-	3,673	-	-	95	-	-	275	-
Male	2,134	153	-	-	1,611	-	-	95	-	-	275	-
Female	2,078	-	16	-	2,062	-	-	-	-	-	-	-
Manufacture of other chem	•	3										
Total	9,255	-	-	638	5,647	-	-	317	-	-	2,653	-
Male	7,603	-	-	536	5,005	-	-	317	-	-	1,746	-
Female	1,651	-	-	102	642	-	-	-	-	-	907	-
Manufacture of man-made												
Total	1,095	-	-	572	524	-	-	-	-	-	-	-
Male	524	-	-	-	524	-	-	-	-	-	-	-
Female	572	-	-	572	-	-	-	-	-	-	-	-
DIVISION 21 MANUFACTUI		RMACEU	TICALS,	MEDICINAI		L AND BO	OTANICAL PRO	DDUCTS				
Total	5,082	765	-	355	851	-	-	241	-	-	2,870	-
Male	2,348	306	-	355	114	-	-	-	-	-	1,573	-
Female	2,734	459	-	-	737	-	-	241	-	-	1,297	-
Manufacture of pharmaceu	ticals, medic	cinal chem	nical and I	ootanical pr	oducts							
Total	5,082	765	-	355	851	-	-	241	-	-	2,870	-
Male	2,348	306	-	355	114	-	-	-	-	-	1,573	-
Female	2,734	459	-	-	737	-	-	241	-	-	1,297	-
DIVISION 22 MANUFACTUI	RE OF RUB	BER AND	PLASTIC	CS PRODU	CTS							
Total	11,208	358	-	781	3,500	84	-	865	-	-	5,355	265
Male	9,089	223	-	688	2,563	84	-	865	-	-	4,401	265
Female	2,119	135	-	92	937	-	-	-	-	-	954	-
Manufacture of rubber proc												
Total	3,916	90	-	781	811	84	-	865	-	-	1,196	89
Male	3,824	90	-	688	811	84	-	865	-	-	1,196	89
Female	92	-	-	92	-	-	-	-	-	-	-	-
Manufacture of plastics pro	ducts											
Total	7,292	268	-	-	2,689	_	-	-	-	-	4,159	176
Male	5,265	133	-	-	1,752	_	-	-	-	-	3,205	176
Female	2,026	135	-	-	937	-	-	-	-	_	954	-
DIVISION 23 MANUFACTUI			METALLI	C MINERA		TS						
Total	37,709	4,511	654	7,477	12,008	-	42	3,483	92	166	7,344	1,933
Male	29,382	3,491	654	5,318	9,351	_	42	2,809	46	124	5,701	1,847
Female	8,327	1,020	-	2,159	2,657	-	-	674	46	42	1,644	86

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
Minor Industrial Divisions							Benishangul-			I	Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Manufacture of glass and g	glass produc	ts				•			•	•		
Total	1,839	272	-	-	95	-	-	96	92	-	1,283	-
Male	1,457	272	-	-	95	-	-	96	46	-	948	-
Female	382	-	-	-	-	-	-	-	46	-	336	-
Manufacture of non-metalli	c mineral pr	oducts no	t elsewhe	re classified								
Total	35,871	4,239	654	7,477	11,913	-	42	3,386	-	166	6,061	1,933
Male	27,925	3,219	654	5,318	9,256	-	42	2,713	-	124	4,753	1,847
Female	7,945	1,020	-	2,159	2,657	_	-	674	-	42	1,308	86
DIVISION 24 MANUFACTU	RE OF BAS	IC METAL	_S	,	,						,	
Total	28,682	4,478	230	4,515	7,749	120	308	1,244	492	-	8,863	682
Male	24,822	3,537	16	4,067	6,600	120	130	1,244	492	-	8,185	430
Female	3,861	942	214	448	1,149	_	177	· -	-	-	678	253
Manufacture of basic iron a	and steel				,							
Total	10,760	627	-	524	3,569	-	-	338	-	-	5,525	177
Male	9,259	627	-	75	2,858	-	-	338	-	-	5,183	177
Female	1,501	-	-	448	711	-	-	-	-	-	342	-
Manufacture of basic preci-	ous and oth	er non-fer	rous meta	ls								
Total	1,330	535	-	-	473	-	-	-	-	-	-	322
Male	779	267	-	-	350	-	-	-	-	-	-	161
Female	551	267	-	-	123	-	-	-	-	-	-	161
Casting of metals												
Total	16,592	3,317	230	3,991	3,707	120	308	907	492	-	3,338	183
Male	14,784	2,642	16	3,991	3,392	120	130	907	492	-	3,002	92
Female	1,808	674	214	-	315	-	177	-	-	-	336	92
DIVISION 25 MANUFACTU	RE OF FAB	RICATED	METAL F	PRODUCTS	S, EXCEPT	MACHINE	RY AND EQU	IPMENT				
Total	25,468	3,663	328	2,926	5,550	272	383	1,556	40	233	10,121	395
Male	22,589	3,572	328	2,926	4,281	272	383	1,556	40	233	8,602	395
Female	2,878	90	-	-	1,269	-	-	-	-	-	1,519	-
Manufacture of structural m	netal produc	ts, tanks,	reservoirs	and steam	generators	3						
Total	4,071	137	-	92	801	-	-	220	-	-	2,821	-
Male	3,784	137	-	92	801	-	-	220	-	-	2,534	-
Female	287	_	-	-	-	_	-	-	-	-	287	_
Manufacture of weapons a		ion										
Total	2,631	-	_	-	2,297	-	-	-	-	-	334	-
Male	1,028	-	_	-	1,028	_	-	-	-	-	-	_
Female	1,603	-	-	-	1,269	-	-	-	-	-	334	-

Table 5.10 Cont'd

Minor Industrial Division						Reg	gion					
Minor Industrial Divisions							Benishangul-				Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Manufacture of other fabric	ated metal _l	oroducts;	metal wor	king service	activities							
Total	18,766	3,526	328	2,834	2,452	272	383	1,336	40	233	6,966	395
Male	17,777	3,436	328	2,834	2,452	272	383	1,336	40	233	6,069	395
Female	988	90	-	-	-	-	-	-	-	-	898	-
DIVISION 26 MANUFACTU	RE OF COM	IPUTER,	ELECTRO	ONIC AND	OPTICAL P	RODUCTS						
Total	5,104	543	16	124	1,183	73	401	301	-	49	2,415	-
Male	3,993	543	16	124	786	73	29	301	-	49	2,072	-
Female	1,110	-	-	-	396	-	372	-	-	-	343	-
Manufacture of electronic of	components	and board	ds									
Total	1,120	272	-	-	166	-	-	-	-	-	682	-
Male	1,037	272	-	-	83	-	-	-	-	-	682	-
Female	83	-	-	-	83	-	-	-	-	-	-	-
Manufacture of computers	and periphe	ral equipn	nent									
Total	1,380	-	-	62	436	-	209	-	-	-	673	-
Male	515	-	-	62	122	-	-	-	-	-	331	-
Female	865	-	-	-	313	-	209	-	-	-	343	-
Manufacture of communica	ation equipm	ent										
Total	402	271	-	-	57	73	-	-	-	-	-	-
Male	402	271	-	-	57	73	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of consumer e	electronics											
Total	1,250	-	-	-	524	-	-	-	-	-	726	-
Male	1,250	-	-	-	524	-	-	-	-	-	726	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of measuring,	testing, nav	igating ar	nd control	equipment;	watches ar	nd clocks						
Total	461	-	16	63	-	-	-	-	-	49	333	-
Male	461	-	16	63	-	-	-	-	-	49	333	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of irradiation,	electro med	lical and e	lectrother	apeutic equ	ipment							
Total	163	-	-	-	-	-	163	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	163	-	-	-	-	-	163	-	-	-	-	-
Manufacture of optical instr	ruments and	d photogra	phic equip	oment								
Total	330	-	-	-	-	-	29	301	-	-	-	-
Male	330	-	-	-	-	-	29	301	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-

Table 5.10 Cont'd

Minor Industrial Divisions						Re	gion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Manufacture of magnetic a	nd optical m	nedia										
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
DIVISION 27 MANUFACTU			EQUIPM	ENT								
Total	10,689	267	-	843	3,361	-	-	453	-	42	5,723	-
Male	8,583	267	-	285	2,577	-	-	453	-	42	4,959	-
Female	2,106	-	-	558	784	-	-	-	-	-	763	-
Manufacture of electric mo		ators, trans	sformers a	and electricit	y distribution	n						
Total	151	-	-	-	-	-	-	151	-	-	-	-
Male	151	-	-	-	-	-	-	151	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of batteries ar	nd accumula	ators										
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of wiring and	wiring devic	es										
Total	3,737	-	-	538	1,100	-	-	-	-	-	2,099	-
Male	2,814	-	-	90	1,100	-	-	-	-	-	1,623	-
Female	924	-	-	447	-	-	-	-	-	-	476	-
Manufacture of electric ligh	iting equipm	nent										
Total	2,827	-	-	93	1,826	-	-	-	-	-	908	-
Male	1,756	-	-	93	1,042	-	-	-	-	-	620	-
Female	1,071	-	-	-	784	-	-	-	-	-	287	-
Manufacture of domestic a	ppliances											
Total	944	267	-	212	121	-	-	301	-	42	-	-
Male	833	267	-	101	121	-	-	301	-	42	-	-
Female	111	-	-	111	-	-	-	-	-	-	-	-
Manufacture of other electr	rical equipm	ent										
Total	3,030	-	-	-	314	-	-	-	-	-	2,716	-
Male	3,030	-	-	-	314	-	-	-	-	-	2,716	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
DIVISION 28 MANUFACTU	RE OF MAG	CHINERY	AND EQU	JIPMENT N	OT ELSEW	HERE CL	ASSIFIED					
Total	7,021	-	110	1,370	2,770	206	41	1,095	22	48	1,360	-
Male	5,710	-	-	1,370	2,157	40	41	672	22	48	1,360	-
Female	1,311	-	110	-	613	166	-	423	-	-	-	-

Table 5.10 Cont'd

Minor Industrial Divisions							gion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Manufacture of general-put		nery										
Total	3,801	-	110	1,370	1,047	-	41	482	22	48	681	-
Male	3,691	-	-	1,370	1,047	-	41	482	22	48	681	-
Female	110	-	110	-	-	-	-	-	-	-	-	-
Manufacture of special-pur	-	nery										
Total	3,220	-	-	-	1,723	206	-	613	-	-	679	-
Male	2,019	-	-	-	1,110	40	-	189	-	-	679	-
Female	1,201	-	-	-	613	166	-	423	-	-	-	-
DIVISION 29 MANUFACTU	RE OF MOT	OR VEHI	CLES, TF	RAILERS AI	ND SEMI-T	RAILERS						
Total	13,350	1,104	10	1,253	2,281	92	-	-	-	-	8,288	322
Male	11,880	755	-	1,253	2,201	92	-	-	-	-	7,256	322
Female	1,471	349	10	-	80	-	-	-	-	-	1,032	-
Manufacture of motor vehice	cles											
Total	3,412	-	-	-	741	52	-	-	-	-	2,619	-
Male	3,076	-	-	-	741	52	-	-	-	-	2,282	-
Female	336	-	-	-	-	-	-	-	-	-	336	-
Manufacture of bodies (coach	work) for mo	otor vehicle	es; manu	facture of tra	ailers and	semi-trailers	3					
Total	7,823	1,104	10	567	685	40	-	-	-	-	5,095	322
Male	6,689	755	-	567	605	40	-	-	-	-	4,399	322
Female	1,134	349	10	-	80	-	-	-	-	-	696	-
Manufacture of parts and ac	cessories fo	r motor ve	ehicles									
Total	2,115	-	-	686	855	-	-	-	-	-	574	-
Male	2,115	-	-	686	855	-	-	-	-	-	574	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
DIVISION 30 MANUFACTU	RE OF OTH	IER TRAN	ISPORT I	EQUIPMEN	Т							
Total	2,722	405	16	239	1,722	-	-	-	-	-	341	-
Male	2,319	405	16	239	1,659	-	-	-	-	-	-	-
Female	404	-	-	-	62	-	-	-	-	-	341	-
Building of ships and boats	;											
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of railway loco	omotives and	d rolling st	ock									
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	_

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Manufacture of air and spa-	cecraft and	related m	achinery									
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of military figh	ting vehicle	S										
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of transport ed	quipment no	ot elsewhe	re classifi	ied								
Total	2,722	405	16	239	1,722	-	-	-	-	-	341	-
Male	2,319	405	16	239	1,659	-	-	-	-	-	-	-
Female	404	-	-	-	62	-	-	-	-	-	341	-
DIVISION 31 MANUFACTUR	RE OF FUR	NITURE										
Total	25,102	619	-	10,062	4,131	312	259	2,733	362	182	6,009	433
Male	22,932	528	-	10,062	2,974	312	143	2,305	317	182	5,676	433
Female	2,170	90	-	-	1,158	-	116	427	45	-	333	-
Manufacture of furniture												
Total	25,102	619	-	10,062	4,131	312	259	2,733	362	182	6,009	433
Male	22,932	528	-	10,062	2,974	312	143	2,305	317	182	5,676	433
Female	2,170	90	-	-	1,158	-	116	427	45	-	333	-
DIVISION 32 OTHER MANU	JFACTURIN	١G										
Total	18,438	874	-	7,688	6,147	121	266	416	-	147	2,780	-
Male	7,631	135	-	988	2,966	81	266	416	-	-	2,780	-
Female	10,807	738	-	6,700	3,181	40	-	-	-	147	-	-
Manufacture of jewellery, b	ijouterie an	d related a	articles									
Total	5,493	269	-	1,482	1,428	81	266	416	-	-	1,553	-
Male	4,082	135	-	890	782	40	266	416	-	-	1,553	-
Female	1,412	133	-	592	646	40	-	-	-	-	-	-
Manufacture of musical inst	truments											
Total	120	-	-	-	120	-	-	-	-	-	-	-
Male	120	-	-	-	120	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of sports good	ds											
Total	604	-	-	-	261	-	-	-	-	-	343	-
Male	343	-	-	-	-	-	-	-	-	-	343	-
Female	261	-	-	-	261	-	-	-	-	-	-	-

Table 5.10 Cont'd

Table 3.10 Cont u						D.	egion					
Minor Industrial Divisions		ı				I	Benishangul-				Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Manufacture of games and	toys				•	•			•			
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of medical an	d dental inst	truments a	and suppli	ies								
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Other manufacturing not el	sewhere cla	ssified										
Total	12,221	605	-	6,206	4,338	40	-	-	-	147	884	-
Male	3,087	-	-	98	2,064	40	-	-	-	-	884	-
Female	9,133	605	-	6,108	2,274	-	-	-	-	147	-	-
DIVISION 33 REPAIR AND	INSTALLAT	TON OF N	MACHINE	RY AND EC	QUIPMENT							
Total	31,893	5,402	159	4,243	6,536		137	3,523	70	48	10,989	494
Male	28,435	5,402	159	4,243	6,015	293	137	3,024	70	48	8,637	407
Female	3,459	-	-	-	521	-	-	499	-	-	2,352	87
Repair of fabricated metal	products, ma	achinery a	ınd equipi	ment								
Total	31,462	5,402	159	4,243	6,478	293	137	3,523	70	48	10,702	407
Male	28,090	5,402	159	4,243	5,958	293	137	3,024	70	48	8,350	407
Female	3,372	-	-	-	521	-	-	499	-	-	2,352	-
Installation of industrial ma	chinery and	equipmer	nt									
Total	431	-	-	-	57	-	-	-	-	-	287	87
Male	345	-	-	-	57	-	-	-	-	-	287	-
Female	87	-	-	-	-	-	-	-	-	-	-	87
SECTION D: ELECTRICITY,	GAS, STE	AM AND A	AIR CONI	DITIONING	SUPPLY							
Total	21,693	2,916	282	5,154	5,889	152	68	1,982	114	91	4,103	942
Male	17,435	2,300	282	4,685	4,781	101	68	1,117	22	49	3,088	942
Female	4,258	616	-	469	1,108	50	-	865	93	42	1,015	-
DIVISION 35 ELECTRICITY	[′] , GAS, STE	AM AND	AIR CON	DITIONING	SUPPLY							
Total	21,693	2,916	282	5,154	5,889	152	68	1,982	114	91	4,103	942
Male	17,435	2,300	282	4,685	4,781	101	68	1,117	22	49	3,088	942
Female	4,258	616	-	469	1,108	50	-	865	93	42	1,015	-
Electric power generation,	transmissio	n and disti	ribution									
Total	18,643	2,916	282	5,052	4,266	101	68	1,741	93	91	3,091	942
Male	15,734	2,300	282	4,685	3,680	101	68	875	-	49	2,750	942
Female	2,909	616	-	367	585	-	-	865	93	42	341	-

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Manufacture of gas; distrib	ution of gas	eous fuels	through	mains								<u></u>
Total	1,802	-	-	101	1,100	-	-	242	22	-	338	-
Male	1,178	-	-	-	577	-	-	242	22	-	338	-
Female	624	-	-	101	523	-	-	-	-	-	-	-
Steam and air conditioning	supply											
Total	1,248	-	-	-	524	50	-	-	-	-	674	-
Male	524	-	-	-	524	-	-	-	-	-	-	-
Female	725	-	-	-	-	50	-	-	-	-	674	-
SECTION E: WATER SUPPL	Y; SEWER	AGE, WAS			AND REMI		ACTIVITIES					
Total	55,487	2,099	3,988	14,085	14,387	3,072	143	9,941	394	453	6,018	907
Male	34,784	1,646	1,959	10,438	8,900	1,899	107	5,281	180	354	3,838	182
Female	20,703	453	2,029	3,647	5,487	1,173	36	4,660	213	99	2,181	725
DIVISION 36 WATER COLL												
Total	39,473	2,099	3,988	7,759	9,767	3,032	143	8,957	331	453	2,610	335
Male	26,550	1,646	1,959	6,887	6,826	1,899	107	4,580	180	354	1,929	182
Female	12,924	453	2,029	872	2,941	1,133	36	4,377	151	99	681	153
Water collection, treatment												
Total	39,473	2,099	3,988	7,759	9,767	3,032	143	8,957	331	453	2,610	335
Male	26,550	1,646	1,959	6,887	6,826	1,899	107	4,580	180	354	1,929	182
Female	12,924	453	2,029	872	2,941	1,133	36	4,377	151	99	681	153
DIVISION 37 SEWERAGE												
Total	2,803	-	-	800	933	-	-	-	-	-	916	153
Male	942	-	-	314	-	-	-	-	-	-	629	-
Female	1,861	-	-	487	933	-	-	-	-	-	287	153
Sewerage												
Total	2,803	-	-	800	933	-	-	-	-	-	916	153
Male	942	-	-	314	-	-	-	-	-	-	629	-
Female	1,861	-	-	487	933	-	-	-	-	-	287	153
DIVISION 38 WASTE COLL	ECTION, TI	REATMEN	NT AND D	ISPOSAL A	ACTIVITIES	; MATERI	ALS RECOVE	RY				
Total	10,791	-	-	5,526	2,157	-	-	590	40	-	2,149	330
Male	6,373	-	-	3,237	1,456	-	-	400	-	-	1,280	-
Female	4,418	-	-	2,288	700	-	-	190	40	-	869	330
Waste collection												
Total	6,262	-	-	3,085	970	-	-	400	-	-	1,477	330
Male	2,535	-	-	797	392	-	-	400	-	-	946	-
Female	3,727	-	-	2,288	577	-	-	-	-	-	531	330

Table 5.10 Cont'd

Maste treatment and disposal	Minor Industrial Divisions						Re	gion					
Waste treatment and disposal													
Total			Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Male 541 - 541 - 541 - - 123 - 190 40 - - - - 123 - 190 40 - 3.38 - - 190 40 - 338 - - 10 40 - 338 - - 10 40 - 334 - - 10 334 - - 10 334 - - 10 334 - - 10 334 - - 10 334 - <	Waste treatment and dispo												
Female 691	Total		-	-	-		-	-	190	40	-	338	-
Materials recovery	Male		-	-	-	541	-	-	-	-	-	-	-
Total 3,297 - 2,441 523 - 5 - 6 - 6 - 6 - 334 Female 3,297 - 2,441 523 - 5 - 6 - 6 - 7 - 7 - 334 Female 3,297 - 2,441 523 - 5 - 7 - 7 - 7 - 334 Female 3,297 - 2,441 523 - 7 - 7 - 7 - 7 - 334 Female 5 - 7 - 7 - 7 - 7 - 7 - 7 - 7 - 7 - 7 -		691	-	-	-	123	-	-	190	40	-	338	-
Male													
Female	Total		-	-	,			-	-	-	-		-
DIVISION 39 REMEDIATION ACTIVITIES AND OTHER WASTE MANAGEMENT SERVICES Total	Male	3,297	-	-	2,441	523	-	-	-	-	-	334	-
Total 2,419 1,530 40 - 395 22 - 343 89 Male 919 618 301 - 30 22 - 343 89 Remediation activities and other waste management services Total 2,419 1,530 40 - 395 22 - 343 89 Remediation activities and other waste management services Total 2,419 1,530 40 - 395 22 - 343 89 Male 919 1,530 40 - 395 22 - 343 89 Male 919 1,530 40 - 395 22 - 343 89 Female 1,500 912 40 - 395 22 - 343 89 SECTION F: CONSTRUCTION Total 507,021 47,379 8,642 107,696 148,626 5,881 7,385 64,098 2,401 4,027 101,468 9,419 Male 423,634 39,474 7,104 85,168 128,706 5,634 6,491 56,496 2,041 3,589 80,907 8,024 Female 83,387 7,905 1,538 22,528 19,920 247 894 7,602 359 438 20,561 1,395 DIVISION 41 CONSTRUCTION OF BUILDINGS Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 Construction of buildings Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 DIVISION 42 CIVIL ENGINEERING Total 81,035 6,653 676 15,821 22,523 1,287 2,272 10,762 185 207 18,145 2,504 Male 62,823 4,249 566 12,779 18,662 1,247 1,879 8,104 133 207 12,934 2,063 Female 18,212 2,405 110 3,042 3,861 40 393 2,658 53 - 5,521 145 40 Construction of roads and railway Total 46,995 3,209 218 11,028 15,506 978 2,272 5,521 145 167 6,308 1,644 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204		-	-	-	-	-	-	-	-	-	-	-	-
Male 919 - - - 618 - - 301 -	DIVISION 39 REMEDIATION	N ACTIVITIE	ES AND C	THER W	ASTE MAN	AGEMENT	SERVICES	S					
Female Remediation activities and other waster management services - - 912 40 - 93 22 - 343 89 Remediation activities and other waster management services Total 2,419 - - - 618 - - 301 - <td< td=""><td></td><td></td><td>-</td><td>-</td><td>-</td><td></td><td></td><td>-</td><td></td><td>22</td><td>-</td><td>343</td><td>89</td></td<>			-	-	-			-		22	-	343	89
Remediation activities and other waste management services Total 2,419 - - - - 1,530 40 - 395 22 - 343 89	Male		-	-	-	618		-	301	-	-	-	-
Total 2,419 1,530 40 - 395 22 - 343 89 Male 919 618 301 - 301 301 305 Female 1,500 912 40 301 301 343 89 SECTION F: CONSTRUCTION Total 507,021 47,379 8,642 107,696 148,626 5,881 7,385 64,098 2,401 4,027 101,468 9,419 Male 423,634 39,474 7,104 85,168 128,706 5,634 6,491 56,496 2,041 3,589 80,907 8,024 Female 83,387 7,905 1,538 22,528 19,920 247 894 7,602 359 438 20,561 1,395 DIVISION 41 CONSTRUCTION OF BUILDINGS Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 Construction of buildings Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 Construction of buildings Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 DIVISION 42 CIVIL ENGINEERING Total 81,035 6,653 676 15,821 22,523 1,287 2,272 10,762 185 207 18,145 2,504 Male 62,823 4,249 566 12,779 18,662 1,247 1,879 8,104 133 207 12,934 2,063 Female 18,212 2,405 110 3,042 3,861 40 393 2,658 53 - 5,211 440 Construction of roads and railways Total 46,995 3,209 218 11,028 15,506 978 2,272 5,521 145 167 6,308 1,644 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204			-	-	-	912	40	-	93	22	-	343	89
Male 919 - - - 618 - - 301 - - - 343 89 SECTION F: CONSTRUCTION Total 507,021 47,379 8,642 107,696 148,626 5,881 7,385 64,098 2,401 4,027 101,468 9,419 Male 423,634 39,474 7,104 85,168 128,706 5,634 6,491 56,496 2,041 3,589 80,907 8,024 Female 83,387 7,905 1,538 22,528 19,920 247 894 7,602 359 438 20,561 1,395 DIVISION 41 CONSTRUCTION OF BUILLDINGS Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568	Remediation activities and	other waste	managen	nent servi	ces								
Female	Total		-	-	-			-		22	-	343	89
Total 507,021 47,379 8,642 107,696 148,626 5,881 7,385 64,098 2,401 4,027 101,468 9,419 Male 423,634 39,474 7,104 85,168 128,706 5,634 6,491 56,496 2,041 3,589 80,907 8,024 Female 83,387 7,905 1,538 22,528 19,920 247 894 7,602 359 438 20,561 1,395 DIVISION 41 CONSTRUCTION OF BUILDINGS Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 Construction of buildings Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 DIVISION 42 CIVIL ENGINEERING Total 81,035 6,653 676 15,821 22,523 1,287 2,272 10,762 185 207 18,145 2,504 Male 62,823 4,249 566 12,779 18,662 1,247 1,879 8,104 133 207 12,934 2,063 Female 18,212 2,405 110 3,042 3,861 40 393 2,658 53 - 5,211 440 Construction of roads and railways Total 46,995 3,209 218 11,028 15,506 978 2,272 5,521 145 167 6,308 1,644 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204 Male 35,764 2,051 108	Male	919	-	-	-	618	-	-	301	-	-	-	-
Total 507,021 47,379 8,642 107,696 148,626 5,881 7,385 64,098 2,401 4,027 101,468 9,419 Male 423,634 39,474 7,104 85,168 128,706 5,634 6,491 56,496 2,041 3,589 80,907 8,024 Female 83,387 7,905 1,538 22,528 19,920 247 894 7,602 359 438 20,561 1,395 DIVISION 41 CONSTRUCTION OF BUILDINGS 1,538 22,528 19,920 247 894 7,602 359 438 20,561 1,395 DIVISION 41 CONSTRUCTION OF BUILDINGS 1,541 2,119 69,194 4,263 4,041 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 </td <td></td> <td></td> <td>-</td> <td>-</td> <td>-</td> <td>912</td> <td>40</td> <td>-</td> <td>93</td> <td>22</td> <td>-</td> <td>343</td> <td>89</td>			-	-	-	912	40	-	93	22	-	343	89
Male 423,634 39,474 7,104 85,168 128,706 5,634 6,491 56,496 2,041 3,589 80,907 8,024 Female 83,387 7,905 1,538 22,528 19,920 247 894 7,602 359 438 20,561 1,395 DIVISION 41 CONSTRUCTION OF BUILDINGS 70 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 Construction of buildings 70 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 </td <td>SECTION F: CONSTRUCTION</td> <td>N</td> <td></td>	SECTION F: CONSTRUCTION	N											
Female 83,387 7,905 1,538 22,528 19,920 247 894 7,602 359 438 20,561 1,395 DIVISION 41 CONSTRUCTION OF BUILDINGS Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 Construction of buildings Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568	Total	507,021	47,379	8,642	107,696	148,626	5,881	7,385	64,098	2,401	4,027	101,468	9,419
DIVISION 41 CONSTRUCTION OF BUILDINGS Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 Construction of buildings Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256	Male	423,634	39,474	,	85,168	128,706	5,634	6,491	56,496	2,041	3,589	80,907	8,024
Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 Construction of buildings Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 DIVISION 42 CIVIL ENGINEERING Total 81,035 6,653 676 15,821 22,523 1,287 2,272 10,762 185 207 18,145 2,504 Male 62,823 4,249 566 12,779 18,662 1,247 1,879 8,104 133 207 12,934 2,063 Female 18,212 2,405 110 3,042 3,861 40 393 2,658 53 - 5,211 440 Construction of roads and railways Total 46,995 3,209 218 11,028 15,506 978 2,272 5,521 145 167 6,308 1,644 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204				1,538	22,528	19,920	247	894	7,602	359	438	20,561	1,395
Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 Construction of buildings Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 DIVISION 42 CIVIL ENGINEERING Total 81,035 6,653 676 15,821 22,523 1,287 2,272 10,762 185 207 18,145 2,504	DIVISION 41 CONSTRUCT	ION OF BUI	LDINGS										
Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 Construction of buildings Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 DIVISION 42 CIVIL ENGINEERING Total 81,035 6,653 676 15,821 22,523 1,287 2,272 10,762 185 207 18,145 2,504 Male 62,823 4,249 566 12,779 18,662 1,247 1,879 8,104 133 207 12,934 2,063 Fema	Total	306,972	33,178	6,276	64,599	82,861						69,194	4,263
Construction of buildings Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 DIVISION 42 CIVIL ENGINEERING Total 81,035 6,653 676 15,821 22,523 1,287 2,272 10,762 185 207 18,145 2,504 Male 62,823 4,249 566 12,779 18,662 1,247 1,879 8,104 133 207 12,934 2,063 Female 18,212 2,405 110 3,042 3,861 40 393 2,658 53 - 5,211 440 Construction of roads and railways Total 46,995 3,209 218 11,028 15,506 978 2,272 5,521 145 167 6,308 1,644 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204	Male			4,967		72,181	4,038	3,204		1,267	1,764		3,568
Total 306,972 33,178 6,276 64,599 82,861 4,090 3,553 35,398 1,441 2,119 69,194 4,263 Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 DIVISION 42 CIVIL ENGINEERING Total 81,035 6,653 676 15,821 22,523 1,287 2,272 10,762 185 207 18,145 2,504 Male 62,823 4,249 566 12,779 18,662 1,247 1,879 8,104 133 207 12,934 2,063 Female 18,212 2,405 110 3,042 3,861 40 393 2,658 53 - 5,211 440 Construction of roads and railways Total 46,995 3,209 218 11,028 15,506 978 2,272 5,521 145 167 6,308 1,644 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204		47,737	3,645	1,308	15,652	10,681	52	349	3,571	173	355	11,256	695
Male 259,235 29,533 4,967 48,947 72,181 4,038 3,204 31,827 1,267 1,764 57,938 3,568 Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 DIVISION 42 CIVIL ENGINEERING Total 81,035 6,653 676 15,821 22,523 1,287 2,272 10,762 185 207 18,145 2,504 Male 62,823 4,249 566 12,779 18,662 1,247 1,879 8,104 133 207 12,934 2,063 Female 18,212 2,405 110 3,042 3,861 40 393 2,658 53 - 5,211 440 Construction of roads and railways Total 46,995 3,209 218 11,028 15,506 978 2,272 5,521 145 167 6,308 1,644 Male 35,764	Construction of buildings												
Female 47,737 3,645 1,308 15,652 10,681 52 349 3,571 173 355 11,256 695 DIVISION 42 CIVIL ENGINEERING Total 81,035 6,653 676 15,821 22,523 1,287 2,272 10,762 185 207 18,145 2,504 Male 62,823 4,249 566 12,779 18,662 1,247 1,879 8,104 133 207 12,934 2,063 Female 18,212 2,405 110 3,042 3,861 40 393 2,658 53 - 5,211 440 Construction of roads and railways Total 46,995 3,209 218 11,028 15,506 978 2,272 5,521 145 167 6,308 1,644 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204	Total	306,972	33,178	6,276	64,599	82,861	4,090	3,553	35,398	1,441	2,119	69,194	4,263
DIVISION 42 CIVIL ENGINEERING Total 81,035 6,653 676 15,821 22,523 1,287 2,272 10,762 185 207 18,145 2,504 Male 62,823 4,249 566 12,779 18,662 1,247 1,879 8,104 133 207 12,934 2,063 Female 18,212 2,405 110 3,042 3,861 40 393 2,658 53 - 5,211 440 Construction of roads and railways Total 46,995 3,209 218 11,028 15,506 978 2,272 5,521 145 167 6,308 1,644 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204	Male		29,533	4,967	48,947	72,181	4,038	3,204	31,827	1,267	1,764	57,938	3,568
Total 81,035 6,653 676 15,821 22,523 1,287 2,272 10,762 185 207 18,145 2,504 Male 62,823 4,249 566 12,779 18,662 1,247 1,879 8,104 133 207 12,934 2,063 Female 18,212 2,405 110 3,042 3,861 40 393 2,658 53 - 5,211 440 Construction of roads and railways Total 46,995 3,209 218 11,028 15,506 978 2,272 5,521 145 167 6,308 1,644 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204	Female	47,737	3,645	1,308	15,652	10,681	52	349	3,571	173	355	11,256	695
Male 62,823 4,249 566 12,779 18,662 1,247 1,879 8,104 133 207 12,934 2,063 Female 18,212 2,405 110 3,042 3,861 40 393 2,658 53 - 5,211 440 Construction of roads and railways Total 46,995 3,209 218 11,028 15,506 978 2,272 5,521 145 167 6,308 1,644 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204	DIVISION 42 CIVIL ENGINE	ERING											
Female 18,212 2,405 110 3,042 3,861 40 393 2,658 53 - 5,211 440 Construction of roads and railways Total 46,995 3,209 218 11,028 15,506 978 2,272 5,521 145 167 6,308 1,644 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204	Total	81,035	6,653	676	15,821	22,523	1,287	2,272	10,762	185	207	18,145	2,504
Construction of roads and railways Total 46,995 3,209 218 11,028 15,506 978 2,272 5,521 145 167 6,308 1,644 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204	Male				,						207		2,063
Total 46,995 3,209 218 11,028 15,506 978 2,272 5,521 145 167 6,308 1,644 Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204	Female	18,212	2,405	110	3,042	3,861	40	393	2,658	53	-	5,211	440
Male 35,764 2,051 108 8,608 11,827 937 1,879 4,825 93 167 4,065 1,204	Construction of roads and i	railways											
	Total	46,995	3,209	218	11,028	15,506	978		5,521	145	167	6,308	1,644
Female 11,232 1,157 110 2,420 3,679 40 393 696 53 - 2,243 440	Male	35,764	2,051	108	8,608	11,827	937	1,879	4,825	93	167	4,065	1,204
	Female	11,232	1,157	110	2,420	3,679	40	393	696	53	-	2,243	440

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Construction of utility proje	cts											
Total	20,261	832	244	4,242	4,682	259	-	2,544	-	-	6,931	527
Male	16,489	832	244	3,621	4,500	259	-	2,544	-	-	3,962	527
Female	3,772	-	-	621	182	-	-	-	-	-	2,969	-
Construction of other civil e	engineering	projects										
Total	13,779	2,613	214	551	2,334	50	-	2,697	40	41	4,907	332
Male	10,570	1,366	214	551	2,334	50	-	736	40	41	4,907	332
Female	3,209	1,247	-	-	-	-	-	1,962	-	-	-	-
DIVISION 43 SPECIALIZED	CONSTRU	ICTION A	CTIVITIES	S								
Total	119,014	7,547	1,691	27,277	43,242	503	1,560	17,938	775	1,701	14,129	2,652
Male	101,577	5,692	1,571	23,442	37,864	349	1,408	16,565	642	1,618	10,035	2,392
Female	17,437	1,855	120	3,835	5,378	154	152	1,373	133	83	4,094	260
Demolition and site prepara	ation											
Total	11,627	880	88	2,568	2,581	309	43	1,680	257	49	3,085	87
Male	8,131	446	79	2,468	2,059	154	43	1,245	186	49	1,313	87
Female	3,496	434	10	101	522	154	-	434	71	-	1,771	-
Electrical, plumbing and ot	her construc	tion instal	lation acti	ivities								
Total	17,599	546	442	2,870	4,896	40	-	2,535	22	49	5,954	245
Male	14,101	546	442	1,642	4,187	40	-	2,294	-	49	4,657	245
Female	3,498	-	-	1,229	709	-	-	241	22	-	1,297	-
Building completion and fin	ishing											
Total	9,467	577	331	1,488	3,479	40	78	1,085	104	215	2,070	-
Male	8,624	577	331	1,387	3,167	40	78	1,085	104	132	1,723	-
Female	843	-	-	101	313	-	-	-	-	83	347	-
Other specialized construct	tion activities	S										
Total	80,321	5,544	830	20,351	32,285	114	1,439	12,638	392	1,388	3,021	2,320
Male	70,721	4,123	720	17,945	28,451	114	1,287	11,940	351	1,388	2,342	2,060
Female	9,601	1,421	110	2,405	3,835	-	152	698	40	-	679	260
SECTION G: WHOLESALE	AND RETAI	L TRADE	; REPAIR	OF MOTO	R VEHICLE	ES AND N	OTORCYCLE	S				
Total	1,465,772	90,787	26,116	269,813	480,106	35,862	12,011	231,455	8,292	17,121	262,031	32,179
Male	706,136	33,308	11,002	122,190	235,570	13,579	5,195	113,143	4,280	6,207	148,267	13,396
Female	759,636	57,479	15,114	147,622	244,536	22,283	6,816	118,312	4,012	10,914	113,764	18,784
DIVISION 45 WHOLESALE		IL TRADE		PAIR OF M			ND MOTORCY	CLES				
Total	49,567	1,614	350	7,043	9,672	451	287	5,283	63	391	22,238	2,174
Male	45,670	1,344	350	6,061	9,387	451	287	4,849	63	391	20,583	1,904
Female	3,897	271	-	982	284	-	-	434	-	-	1,655	271

Table 5.10 Cont'd

Minor Industrial Divisions							gion					
and sex					_		Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Sale of motor vehicles												
Total	4,558	485	-	714	578	-	-	-	-	-	2,688	92
Male	2,772	485	-	-	578	-	-	-	-	-	1,708	-
Female	1,786	-	-	714	-	-	-	-	-	-	980	92
Maintenance and repair of I	motor vehic	les										
Total	38,016	858	-	5,418	7,814	401	58	3,847	63	301	17,869	1,387
Male	37,647	858	-	5,254	7,608	401	58	3,847	63	301	17,869	1,387
Female	369	-	-	164	206	-	-	-	-	-	-	-
Sale of motor vehicle parts	and access	ories										
Total	2,408	135	-	208	113	50	-	824	-	41	343	695
Male	1,134	-	-	104	34	50	-	390	-	41	-	516
Female	1,274	135	-	104	79	-	-	434	-	-	343	179
Sale, maintenance and repa		•	•									
Total	4,584	135	350	703	1,167	-	229	612	-	50	1,338	-
Male	4,117	-	350	703	1,167	-	229	612	-	50	1,007	-
Female	467	135	-	-	-	-	-	-	-	-	332	-
DIVISION 46 WHOLESALE	TRADE, EX	CEPT OF	MOTOR	VEHICLES	AND MOT	ORCYCLE	S					
Total	115,963	4,058	3,213	18,513	35,844	2,339	193	10,584	386	2,278	32,943	5,613
Male	79,272	3,429	2,372	14,451	27,547	1,757	193	4,798	333	1,209	19,785	3,396
Female	36,691	629	841	4,062	8,296	581	-	5,786	53	1,068	13,158	2,217
Wholesale on a fee or conti	ract basis											
Total	7,020	706	-	1,106	2,012	-	-	95	104	50	2,947	-
Male	5,123	706	-	862	1,339	-	-	95	104	50	1,967	-
Female	1,897	-	-	244	673	-	-	-	-	-	980	-
Wholesale of agricultural ra	w materials	and live a	nimals									
Total	30,049	886	1,603	2,944	14,454	843	77	1,969	49	327	4,637	2,260
Male	21,636	886	1,257	2,692	11,029	637	77	702	49	237	2,986	1,085
Female	8,413	-	347	252	3,425	206	-	1,267	-	90	1,651	1,175
Wholesale of food, beverag	es and toba	acco										
Total	24,408	1,218	494	6,254	6,428	776	-	2,108	46	646	4,853	1,584
Male	18,157	950	234	5,044	4,724	529	-	1,105	46	517	4,183	825
Female	6,251	268	260	1,210	1,703	247	-	1,003	-	129	670	760
Wholesale of household go	ods											
Total	29,654	467	571	2,874	6,440	242	-	2,596	53	924	15,006	482
Male	16,864	377	336	2,021	5,035	114	-	96	-	165	8,418	302
Female	12,791	90	235	852	1,405	128	-	2,500	53	759	6,589	180

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Wholesale of machinery, e	quipment an	d supplies	3									
Total	3,023	-	-	155	-	-	-	-	-	289	2,579	-
Male	564	-	-	94	-	-	-	-	-	241	230	-
Female	2,459	-	-	62	-	-	-	-	-	48	2,349	-
Other specialized wholesal	е											
Total	12,316	647	545	2,154	4,647	-	116	613	88	-	2,577	927
Male	9,552	510	545	1,447	3,557	-	116	462	88	-	2,001	825
Female	2,764	137	-	707	1,090	-	-	151	-	-	576	102
Non-specialized wholesale	trade											
Total	9,493	133	-	3,025	1,864	478	-	3,203	46	42	343	359
Male	7,376	-	-	2,291	1,864	478	-	2,338	46	-	-	359
Female	2,117	133	-	734	-	-	-	865	-	42	343	-
DIVISION 47 RETAIL TRAD	E, EXCEPT	OF MOT	OR VEHI	CLES AND	MOTORCY	'CLES						
Total	1,300,242	85,114	22,552	244,257	434,590	33,072	11,531	215,588	7,843	14,451	206,851	24,392
Male	581,194	28,535	8,280	101,678	198,635	11,371	4,715	103,496	3,885	4,606	107,899	8,096
Female	719,048	56,579	14,272	142,579	235,955	21,702	6,816	112,092	3,959	9,845	98,952	16,296
Retail sale in non-specialize	ed stores											
Total	522,998	31,782	5,884	107,943	156,041	14,011	6,536	85,583	1,690	3,218	101,583	8,726
Male	241,923	9,323	2,782	48,285	70,718	6,421	2,918	43,973	976	1,097	51,362	4,071
Female	281,075	22,459	3,102	59,658	85,324	7,591	3,618	41,610	715	2,121	50,221	4,656
Retail sale of food, beverage	ges and toba	acco in sp	ecialized	stores								
Total	143,789	13,035	3,539	41,576	43,128	5,008	1,520	23,167	1,797	814	8,570	1,634
Male	37,566	2,080	431	12,765	10,450	1,388	221	6,722	512	403	2,400	194
Female	106,223	10,955	3,108	28,811	32,677	3,620	1,299	16,445	1,285	411	6,170	1,441
Retail sale of automotive fu	•	lized store										
Total	7,294	-	436	1,788	1,106	-	29	2,731	-	41	1,075	88
Male	5,189	-	219	1,194	1,106	-	-	1,466	-	41	1,075	88
Female	2,105	-	218	594	-	-	29	1,265	-	-	-	-
Retail sale of information a	nd communi	ications ed	quipment	in specialize	ed stores							
Total	10,272	1,280	-	934	2,649	50	-	95	-	219	4,971	73
Male	8,044	1,190	-	812	2,002	50	-	95	-	139	3,683	73
Female	2,228	90	-	123	646	-	-	-	-	81	1,288	-
Retail sale of other househ	old equipme	ent in spec	ialized st	ores								
Total	94,195	9,292	495	19,981	25,712	1,827	639	7,958	362	1,276	25,582	1,072
Male	56,549	5,054	64	9,944	18,109	796	345	6,468	257	611	14,195	707
Female	37,646	4,237	431	10,037	7,603	1,031	293	1,490	105	665	11,387	365

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Retail sale of cultural and r	ecreation go	ods in sp	ecialized :	stores								
Total	7,145	243	-	61	1,125	-	261	1,544	-	148	3,763	-
Male	4,748	-	-	-	1,125	-	130	1,243	-	99	2,150	-
Female	2,397	243	-	61	-	-	130	301	-	49	1,613	-
Retail sale of other goods i	n specialize	d stores										
Total	127,105	9,242	427	18,374	43,259	716	341	19,875	2,629	2,208	27,085	2,951
Male	73,910	3,663	402	9,142	25,257	389	158	14,720	1,507	1,292	15,746	1,633
Female	53,195	5,579	26	9,232	18,001	326	182	5,154	1,121	916	11,339	1,318
Retail sale via stalls and m	arkets											
Total	176,775	7,707	10,169	31,492	66,892	4,935	1,111	26,442	972	4,430	16,534	6,094
Male	67,191	2,625	4,076	11,632	26,851	1,115	535	10,665	447	695	7,615	936
Female	109,584	5,082	6,093	19,860	40,041	3,820	576	15,777	524	3,735	8,918	5,157
Retail trade not in stores, s												
Total	210,668	12,533	1,602	22,108	94,679	6,525	1,095	48,194	394	2,097	17,689	3,753
Male	86,074	4,600	307	7,905	43,017	1,211	407	18,144	186	230	9,673	394
Female	124,594	7,933	1,295	14,203	51,662	5,313	688	30,050	208	1,867	8,016	3,359
SECTION H: TRANSPORTA	TION AND	STORAG	E									
Total	389,343	25,212	3,600	63,314	118,525	4,269	3,099	50,421	2,171	3,671	105,937	9,124
Male	368,724	23,333	3,373	60,105	114,599	4,229	3,022	48,595	2,067	3,312	97,414	8,674
Female	20,619	1,879	227	3,209	3,926	40	77	1,826	103	358	8,523	450
DIVISION 49 LAND TRANS	PORT AND	TRANSP	ORT VIA	PIPELINES								
Total	310,420	21,150	3,396	54,601	91,783	2,691	2,241	39,133	1,629	2,763	83,156	7,878
Male	300,275	20,804	3,169	51,944	89,976	2,691	2,165	38,338	1,607	2,714	79,229	7,640
Female	10,145	345	227	2,657	1,807	-	77	795	22	49	3,927	238
Transport via railways												
Total	10,423	698	147	1,726	3,245	51	-	746	-	-	3,496	315
Male	10,198	698	137	1,726	3,245	51	-	604	-	-	3,496	242
Female	225	-	10	-	-	-	-	143	-	-	-	73
Other land transport												
Total	297,656	20,452	3,249	52,875	87,313	2,640	2,241	37,963	1,629	2,763	78,968	7,563
Male	288,237	20,106	3,031	50,218	85,662	2,640	2,165	37,311	1,607	2,714	75,387	7,398
Female	9,418	345	218	2,657	1,652	-	77	653	22	49	3,581	165
Transport via pipeline												
Total	2,341	-	-	-	1,225	-	-	424	-	-	692	-
Male	1,839	-	-	-	1,070	-	-	424	-	-	346	-
Female	501	-	-	-	155	-	-	-	-	-	346	-

Table 5.10 Cont'd

Minor Industrial Divisions						R	egion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
DIVISION 50 WATER TRAN												
Total	144	-	-	-	57	-	-	-	-	-	-	86
Male	144	-	-	-	57	-	-	-	-	-	-	86
Female	-	-	-	-	-	-	-	-	-	-	-	-
Sea and coastal water tran	sport											
Total	86	-	-	-	-	-	-	-	-	-	-	86
Male	86	-	-	-	-	-	-	-	-	-	-	86
Female	-	-	-	-	-	-	-	-	-	-	-	-
Inland water transport												
Total	57	-	-	-	57	-	-	-	-	-	-	-
Male	57	-	-	-	57	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
DIVISION 51 AIR TRANSPO	ORT											
Total	5,249	497	-	329	579	-	-	241	81	49	3,169	304
Male	2,811	90	-	329	57	-	-	241	40	-	1,961	92
Female	2,439	407	-	-	522	-	-	-	40	49	1,208	212
Passenger air transport												
Total	4,296	407	-	329	57	-	-	241	81	49	2,828	304
Male	2,720	-	-	329	57	-	-	241	40	-	1,961	92
Female	1,576	407	-	-	-	-	-	-	40	49	867	212
Freight air transport												
Total	953	90	-	-	522	-	-	-	-	-	341	-
Male	90	90	-	-	-	-	-	-	-	-	-	-
Female	863	-	-	-	522	-	-	-	-	-	341	-
DIVISION 52 WAREHOUSI	NG AND SU	JPPORT A	CTIVITIE	S FOR TRA	ANSPORT <i>A</i>	ATION						
Total	69,986	3,475	205	7,417	25,639	1,579	716	11,047	461	859	17,911	677
Male	63,787	2,439	205	6,866	24,434	1,538	716	10,016	421	599	15,877	677
Female	6,199	1,036	_	552	1,205	40	-	1,030	41	260	2,034	-
Warehousing and storage	,	,			,			,			,	
Total	2,011	436	-	774	579	-	-	222	-	-	-	-
Male	1,474	256	-	417	579	-	-	222	-	-	-	-
Female	537	181	-	356	-	-	-	-	-	-	-	-
Support activities for transp	portation											
Total	67,974	3,039	205	6,644	25,060	1,579	716	10,824	461	859	17,911	677
Male	62,313	2,183	205	6,449	23,855	1,538	716	9,794		599	15,877	677
Female	5,662	856	_	195	1,205	40	-	1,030		260	2,034	-

Table 5.10 Cont'd

Table 5.10 Cont d						P.	egion					
Minor Industrial Divisions		J	1	1		170	Benishangul-			1	Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
DIVISION 53 POSTAL AND	COURIER		≣S	U.					· I		L.	
Total	3,545	90	-	967	467	-	141	-	-	-	1,701	178
Male	1,707	-	-	967	74	-	141	-	-	-	346	178
Female	1,837	90	-	-	392	-	-	-	-	-	1,355	-
Postal activities												
Total	2,051	-	-	967	467	-	92	-	-	-	346	178
Male	1,658	-	-	967	74	-	92	-	-	-	346	178
Female	392	-	-	-	392	-	-	-	-	-	-	-
Courier activities												
Total	1,494	90	-	-	-	-	49	-	-	-	1,355	-
Male	49	-	-	-	-	-	49	-	-	-	-	-
Female	1,445	90	-	-	-	-	-	-	-	-	1,355	-
SECTION I: ACCOMMODAT	ION AND FO	OOD SER	VICE AC	ΓΙVITIES								
Total	406,137	28,127	9,421	73,137	136,982	1,886	8,088	68,290	4,382	2,133	68,049	5,642
Male	115,886	4,186	1,534	13,389	43,137	761	2,516	26,670	1,423	460	20,943	865
Female	290,251	23,941	7,887	59,748	93,845	1,124	5,572	41,620	2,959	1,673	47,106	4,777
DIVISION 55 ACCOMMODA	NOITA											
Total	35,439	2,441	868	5,635	11,061	206	602	10,561	22	41	3,741	262
Male	13,666	724	-	1,789	3,783	154	182	6,070	-	-	874	91
Female	21,773	1,717	868	3,846	7,278	52	420	4,491	22	41	2,867	171
Short term accommodation	n activities											
Total	7,287	1,584	654	1,128	1,578	166	29	1,473	-	-	676	-
Male	1,952	362	-	594	610	114	-	272	-	-	-	-
Female	5,334	1,221	654	534	967	52	29	1,201	-	-	676	-
Camping grounds, recreati	onal vehicle	parks and	d trailer pa	ırks								
Total	464	-	-	-	48	-	-	416	-	-	-	-
Male	464	-	-	-	48	-	-	416	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Other accommodation												
Total	27,688	857	214	4,507	9,436	40	573	8,672	22	41	3,065	262
Male	11,250	361	-	1,195	3,125	40	182	5,382	-	-	874	91
Female	16,439	496	214	3,312	6,311	-	392	3,290	22	41	2,191	171
DIVISION 56 FOOD AND B	EVERAGE S	SERVICE	ACTIVITI	ES								
Total	370,698	25,686	8,554	67,502	125,921	1,679	7,486	57,729	4,361	2,092	64,308	5,380
Male	102,220	3,462	1,534	11,601	39,354	607	2,335	20,600	1,423	460	20,070	774
Female	268,478	22,224	7,020	55,902	86,567	1,072	5,151	37,129	2,937	1,632	44,239	4,606

Table 5.10 Cont'd

Minor Industrial Divisions						Re	gion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Restaurants and mobile for	od service a	ctivities										
Total	141,017	7,771	2,293	21,714	34,811	397	3,833	28,553	2,705	819	37,678	443
Male	43,335	362	765	6,146	12,760	40	1,130	9,217	1,031	90	11,620	175
Female	97,683	7,409	1,529	15,568	22,052	356	2,703	19,336	1,674	730	26,058	268
Event catering and other fo	ood service a	activities										
Total	131,119	5,621	2,004	12,531	57,802	713	2,609	21,652	857	1,103	21,738	4,488
Male	35,359	1,430	661	1,765	15,264	153	917	7,842	214	330	6,364	420
Female	95,759	4,191	1,343	10,766	42,538	561	1,692	13,810	643	772	15,374	4,068
Beverage serving activities	i											
Total	98,562	12,293	4,256	33,258	33,307	569	1,044	7,524	799	170	4,892	449
Male	23,526	1,670	108	3,690	11,330	414	288	3,542	179	40	2,086	179
Female	75,036	10,623	4,148	29,568	21,977	155	756	3,983	620	130	2,807	270
SECTION J: INFORMATION												
Total	73,483	3,061	836	9,174	11,578	422	649	12,865	289	518	33,221	871
Male	40,126	2,174	184	6,687	6,892	250	203	7,850	184	326	15,033	345
Female DIVISION 58 PUBLISHING	33,357 ACTIVITIES	887	653	2,487	4,686	172	446	5,015	105	192	18,188	526
Total	16,375	133	-	1,113	429	51	-	655	-	-	13,907	86
Male	5,434	-	-	1,113	275	-	-	368	-	-	3,678	-
Female	10,941	133	_	· -	154	51	-	287	_	_	10,229	86
Publishing of books, period	licals and ot	her publis	hing activ	ities							,	
Total	15,700	133	-	1,113	429	51	-	655	-	-	13,231	86
Male	5,434	-	-	1,113	275	-	-	368	-	-	3,678	-
Female	10,266	133	-	-	154	51	-	287	-	-	9,554	86
Software publishing												
Total	675	-	-	-	-	-	-	-	-	-	675	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	675	-	-	-	-	-	-	-	-	-	675	-
DIVISION 59 MOTION PICT PRODUCTION, SOUND RE												
Total	6,395	537	_	128	1,080	-	-	3,285	_	_	1,365	-
Male	3,898	537	-	60	454	_	-	1,826	_	-	1,020	-
Female	2,497	-	-	67	626	_	-	1,459	-	-	345	-
Motion picture, video and to	,	ogram acti	vities	-				,				
Total	5,507	447	-	67	1,080	-	-	2,884	_	-	1,028	-
Male	3,009	447	-	-	454	-	-	1,425	_	_	683	_
Female	2,497	-	-	67	626	-	-	1,459	_	-	345	_

Table 5.10 Cont'd

Minor Industrial Divisions						R	egion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Sound recording and music	c publishing	activit										
Total	889	90	-	60	-	-	-	401	-	-	336	-
Male	889	90	-	60	-	-	-	401	-	-	336	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
DIVISION 60 PROGRAMMI		COADCAS	TING AC	TIVITIES								
Total	3,978	270	-	289	955	84	-	95	-	137	1,964	184
Male	3,170	137	-	221	836	84	-	-	-	89	1,711	92
Female	808	133	-	67	119	-	-	95	-	48	253	92
Radio broadcasting												
Total	2,712	133	-	160	955	-	-	95	-	137	1,231	-
Male	2,063	-	-	160	836	-	-	-	-	89	978	-
Female	648	133	-	-	119	-	-	95	-	48	253	-
Television programming ar		-	ies									
Total	1,266	137	-	129	-	84	-	-	-	-	733	184
Male	1,107	137	-	62	-	84	-	-	-	-	733	92
Female	159	-	-	67	-	-	-	-	-	-	-	92
DIVISION 61 TELECOMMU	NICATIONS	3										
Total	19,071	1,284	333	4,621	3,451	173	105	3,257	122	49	5,229	448
Male	14,193	935	16	4,098	2,906	92	77	2,664	122	49	2,982	253
Female	4,878	349	317	523	545	81	29	592	-	-	2,248	194
Wired telecommunications	activities											
Total	4,928	530	110	3,082	475	-	-	-	-	49	683	-
Male	3,774	181	-	2,724	475	-	-	-	-	49	345	-
Female	1,154	349	110	358	-	-	-	-	-	-	338	-
Wireless Telecommunication	ons activities	3										
Total	3,071	-	207	447	67	52	-	316	-	-	1,695	286
Male	1,893	-	-	447	67	52	-	222	-	-	1,011	92
Female	1,179	-	207	-	-	-	-	93	-	-	683	194
Satellite Telecommunication	ns activities											
Total	735	360	-	-	205	-	-	-	82	-	-	88
Male	651	360	-	-	121	-	-	-	82	-	-	88
Female	84	-	-	-	84	-	-	-	-	-	-	-
Other telecommunications	activities											
Total	10,337	394	16	1,091	2,704	121	105	2,941	40	-	2,852	73
Male	7,876	394	16	926	2,243	40	77	2,442	40	-	1,625	73
Female	2,461	-	-	166	461	81	29	499	-	-	1,227	-

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
DIVISION 62 COMPUTER F	PROGRAMI	/ING, CO	NSULTAN	ICY AND R	ELATED A	CTIVITIES	3					
Total	13,301	523	-	1,293	3,849	-	209	1,943	-	146	5,338	-
Male	6,189	432	-	604	1,472	-	-	281	-	49	3,351	-
Female	7,112	90	-	689	2,377	-	209	1,662	-	97	1,987	-
Computer programming, co	onsultancy a	nd related	d activities									
Total	13,301	523	-	1,293	3,849	-	209	1,943	-	146	5,338	-
Male	6,189	432	-	604	1,472	-	-	281	-	49	3,351	-
Female	7,112	90	-	689	2,377	-	209	1,662	-	97	1,987	-
DIVISION 63 INFORMATIO	N SERVICE	ACTIVIT	IES									
Total	14,364	314	503	1,730	1,813	114	335	3,630	167	187	5,418	153
Male	7,243	133	168	590	948	73	126	2,711	62	139	2,292	-
Female	7,122	181	336	1,140	865	40	209	919	105	48	3,126	153
Data processing, hosting a	nd related a	ctivities; v	veb portal	5								
Total	6,177	-	472	1,082	675	73	-	871	-	-	2,850	153
Male	1,824	-	137	158	62	73	-	717	-	-	676	-
Female	4,353	-	336	924	613	-	-	154	-	-	2,174	153
Other information service a	activities											
Total	8,187	314	31	648	1,137	40	335	2,759	167	187	2,568	-
Male	5,418	133	31	432	885	-	126	1,993	62	139	1,616	-
Female	2,769	181	-	216	252	40	209	765	105	48	952	-
SECTION K: FINANCIAL AN	ID INSURAI	NCE ACT	IVITIES									
Total	171,923	9,289	921	30,152	42,446	3,008	1,713	25,059	1,448	659	55,125	2,102
Male	108,035	3,680	646	19,252	30,941	2,269	1,362	16,071	923	339	31,129	1,421
Female	63,887	5,609	274	10,900	11,505	739	351	8,988	525	320	23,996	681
DIVISION 64 FINANCIAL SI			EXCEPT		CE AND PE		JNDING					
Total	138,752	7,415	889	24,257	35,698	2,390	1,067	21,397	1,082	578	42,251	1,728
Male	87,255	3,271	630	15,997	26,101	1,744	936	14,382	764	258	22,126	1,047
Female	51,497	4,145	259	8,260	9,598	647	130	7,015	318	320	20,124	681
Monetary intermediation												
Total	75,369	5,346	599	14,451	17,286	1,552	586	7,727	655	388	25,805	973
Male	49,347	2,628	372	9,493	13,865	956	456	5,115	574	117	15,153	617
Female	26,022	2,717	227	4,958	3,421	596	130	2,612	81	271	10,652	357
Activities of holding compa												
Total	1,368	-	-	-	1,134	-	-	191	-	43	-	-
Male	1,252	-	-	-	1,018	-	-	191	-	43	-	-
Female	116	-	-	-	116	-	-	-	-	-	-	-

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Trusts, funds and similar fir	nancial entit	ies										
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Other financial service active	vities, excep	ot insuranc	e and per	nsion fundir	ng activities							
Total	62,015	2,070	290	9,806	17,278	839	480	13,479	427	147	16,446	754
Male	36,656	642	258	6,503	11,218	788	480	9,076	190	98	6,973	430
Female	25,359	1,427	32	3,303	6,061	51	-	4,403	237	49	9,472	324
DIVISION 65 INSURANCE,			PENSIO	N FUNDING	•	COMPUL	SORY SOCIAL	L SECURIT	Υ			
Total	3,831	971	-	933	506	-	42	-	82	-	1,012	286
Male	1,458	137	-	154	506	-	42	-	-	-	333	286
Female	2,373	834	-	779	-	-	-	-	82	-	678	-
Insurance												
Total	1,361	834	-	-	-	-	-	-	-	-	333	194
Male	663	137	-	-	-	-	-	-	-	-	333	194
Female	698	698	-	-	-	-	-	-	-	-	-	-
Reinsurance												
Total	996	-	-	183	-	-	42	-	-	-	678	92
Male	134	-	-	-	-	-	42	-	-	-	-	92
Female	861	-	-	183	-	-	-	-	-	-	678	-
Pension funding												
Total	1,474	137	-	750	506	-	-	-	82	-	-	-
Male	660	-	-	154	506	-	-	-	-	-	-	-
Female	814	137	-	596	-	-	-	-	82	-	-	-
DIVISION 66 ACTIVITIES A												
Total	29,340	903	32	4,962	6,242	618	604	3,663	285	81	11,863	89
Male	19,323	273	16	3,102	4,335	526	384	1,690	160	81	8,669	89
Female	10,017	630	16	1,860	1,907	92	220	1,973	125	-	3,193	-
Activities auxiliary to financ			•	urance and	•	iding						
Total	12,422	635	16	2,660	3,170	-	446	1,741	143	41	3,571	-
Male	4,872	273	-	800	1,263	-	226	633	40	41	1,598	-
Female	7,550	363	16	1,860	1,907	-	220	1,108	103	-	1,973	-
Activities auxiliary to insura			-									
Total	15,884	267	16	2,302	3,072	618	158	1,922	141	41	7,258	89
Male	13,761	-	16	2,302	3,072	526	158	1,057	119	41	6,382	89
Female	2,123	267	-	-	-	92	-	865	22	-	876	-

Table 5.10 Cont'd

Minor Industrial Divisions						R	egion					
and sex							Benishangul-				Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Fund management activitie	s											
Total	1,033	-	-	-	-	-	-	-	-	-	1,033	-
Male	690	-	-	-	-	-	-	-	-	-	690	-
Female	344	-	-	-	-	-	-	-	-	-	344	-
SECTION L: REAL ESTATE	ACTIVITIES	3										
Total	5,339	226	-	654	-	-	-	241	-	-	4,218	-
Male	2,909	90	-	-	-	-	-	241	-	-	2,578	-
Female	2,430	135	-	654	-	-	-	-	-	-	1,640	-
DIVISION 68 REAL ESTATE	E ACTIVITIE	S										
Total	5,339	226	-	654	-	-	-	241	-	-	4,218	-
Male	2,909	90	-	-	-	-	-	241	-	-	2,578	-
Female	2,430	135	-	654	-	-	-	-	-	-	1,640	-
Real estate activities with o	wn or lease		′									
Total	776	90	-	-	-	-	-	-	-	-	685	-
Male	90	90	-	-	-	-	-	-	-	-	-	-
Female	685	-	-	-	-	-	-	-	-	-	685	-
Real estate activities on a f	ee or contra	ct basis										
Total	4,563	135	-	654	-	-	-	241	-	-	3,532	-
Male	2,819	-	-	-	-	-	-	241	-	-	2,578	-
Female	1,744	135	-	654	-	-	-	-	-	-	954	-
SECTION M: PROFESSIONA	AL, SCIENT	IFIC AND	TECHNI	CAL ACTIV	ITIES							
Total	126,086	5,109	1,511	13,047	32,310		2,917	20,538	1,891	542	43,792	1,706
Male	85,419	3,815	561	9,772	23,614	2,368	1,900	14,675	1,008	259	25,991	1,459
Female	40,667	1,295	950	3,275	8,697	355	1,017	5,863	884	283	17,801	248
DIVISION 69 LEGAL AND A												
Total	74,523	4,220	887	7,622	19,299		1,081	9,627	1,425	90	27,489	737
Male	48,765	3,059	503	5,501	13,529	,	506	6,300	697	41	16,237	650
Female	25,758	1,161	384	2,121	5,770	304	576	3,327	728	49	11,252	87
Legal activities												
Total	40,148	2,598	887	5,780	11,853		409	5,061	1,271	90	10,180	737
Male	28,498	1,663	503	3,779	9,192		328	2,918	612	41	7,743	650
Female	11,650	935	384	2,002	2,661	213	80	2,143	660	49	2,437	87
Accounting, bookkeeping a	_		tax consu	-								
Total	34,375	1,622	-	1,842	7,446		672	4,566	154	-	17,310	-
Male	20,267	1,396	-	1,722	4,337		177	3,382	85	-	8,494	-
Female	14,108	226	-	119	3,109	91	495	1,184	68	-	8,816	-

Table 5.10 Cont'd

Minor Industrial Divisions						R	egion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
DIVISION 70 ACTIVITIES C	F HEAD OF	FICES; N	1ANAGEN	IENT CON	SULTANCY	' ACTIVIT	IES					
Total	15,694	-	326	2,757	1,027	40	615	5,691	63	-	5,014	161
Male	10,922	-	-	2,757	301	40	472	4,901	63	-	2,389	-
Female	4,772	-	326	-	727	-	143	790	-	-	2,626	161
Activities of head offices												
Total	8,717	-	326	711	149	40	443	4,830	22	-	2,033	161
Male	6,575	-	-	711	-	40	300	4,136	22	-	1,365	-
Female	2,142	-	326	-	149	-	143	694	-	-	669	161
Management consultancy	activities											
Total	6,977	-	-	2,045	878	-	172	861	40	-	2,981	-
Male	4,347	-	-	2,045	301	-	172	765	40	-	1,024	_
Female	2,630	-	_	-	577	-	-	95	-	-	1,957	-
DIVISION 71 ARCHITECTU	JRAL AND E	NGINEER	RING ACT	IVITIES; TI	ECHNICAL	TESTING	AND ANALYS	IS				
Total	4,022	133	_	-	-	-	-	125	40	-	3,723	-
Male	3,555	-	_	-	-	-	-	125	40	-	3,389	-
Female	467	133	-	-	-	-	-	-	-	-	334	_
Architectural and engineer	ing activities	and relate	ed technic	al consulta	ncy							
Total	1,963	133	_	-	-	-	-	125	-	-	1,705	-
Male	1,496	_	_	-	_	_	-	125	_	_	1,371	_
Female	467	133	_	-	-	-	-	-	-	-	334	-
Technical testing and anal	ysis											
Total	2,059	-	_	-	-	_	-	-	40	_	2,018	_
Male	2,059	-	-	-	-	-	-	-	40	-	2,018	-
Female	-	_	_	-	_	_	-	-	_	_	_	-
DIVISION 72 SCIENTIFIC F	RESEARCH	AND DEV	/ELOPME	NT								
Total	7,953	136	16	1,695	2,609	81	377	2,145	127	_	679	86
Male	5,511	136	16	895	1,640	81	328	1,544	105	_	679	86
Female	2,441	_	_	800	970	_	49	601	22	_	_	-
Research and experimenta		ent on nat	ural scien									
Total	5,799	136	_	1,603	1,462	81	163	1,904	105	_	345	-
Male	4,319	136	_	803	1,405	81	163	1,303	84	_	345	_
Female	1,480	-	-	800	57	-	-	601	22	_	-	-
Research and experimenta		ent on soc	ial science									
Total	2,153	-	16	92	1,148	_	214	241	22	_	334	86
Male	1,192	-	16	92	235	_	165	241	22	_	334	86
Female	961	-	-	-	912	_	49	-	_	_	-	-

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
DIVISION 73 ADVERTISING	AND MAR	KET RES	EARCH									
Total	4,635	-	-	202	1,309	-	-	904	122	122	1,975	-
Male	2,749	-	-	202	1,100	-	-	602	40	122	682	-
Female	1,886	-	-	-	209	-	-	302	82	-	1,293	-
Advertising												
Total	1,256	-	-	202	-	-	-	-	-	122	932	-
Male	657	-	-	202	-	-	-	-	-	122	333	-
Female	599	-	-	-	-	-	-	-	-	-	599	-
Market research and public	opinion pol	ling										
Total	3,378	-	-	-	1,309	-	-	904	122	-	1,043	-
Male	2,092	-	-	-	1,100	-	-	602	40	-	350	-
Female	1,287	-	-	-	209	-	-	302	82	-	693	-
DIVISION 74 OTHER PROF		, SCIENT	IFIC AND	TECHNIC/	AL ACTIVIT	IES						
Total	10,726	227	-	771	4,427	225	438	301	-	330	3,286	722
Male	7,685	227	-	417	3,518	174	308	301	-	95	1,923	722
Female	3,042	-	-	354	909	51	129	-	-	234	1,363	-
Specialized design activitie	s											
Total	3,989	-	-	-	2,457	51	-	301	-	234	946	-
Male	2,848	-	-	-	1,937	-	-	301	-	-	610	-
Female	1,141	-	-	-	520	51	-	-	-	234	335	-
Photographic activities												
Total	3,671	227	-	417	1,058	40	388	-	-	95	971	473
Male	3,204	227	-	417	1,058	40	259	-	-	95	633	473
Female	467	-	-	-	-	-	129	-	-	-	338	-
Other professional, scientifi	ic and techn	ical activi	ties not el	sewhere cla	ssifiedn							
Total	3,066	-	-	354	912	133	49	-	-	-	1,369	249
Male	1,633	-	-	-	523	133	49	-	-	-	679	249
Female	1,433	-	-	354	389	-	-	-	-	-	690	-
DIVISION 75 VETERINARY	ACTIVITIES	S										
Total	8,534	393	282	-	3,639	330	406	1,745	115	-	1,625	-
Male	6,233	393	42	-	3,527	330	286	901	62	-	692	-
Female	2,301	-	240	-	112	-	120	843	53	-	933	-
Veterinary activities												
Total	8,534	393	282	-	3,639	330	406	1,745	115	-	1,625	-
Male	6,233	393	42	-	3,527	330	286	901	62	-	692	-
Female	2,301	-	240	-	112	-	120	843	53	-	933	-

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
Minor Industrial Divisions and sex							Benishangul-				Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
SECTION N: ADMINISTRAT	IVE AND SI	JPPORT	SERVICE	ACTIVITIE	S							
Total	146,604	7,264	734	17,841	39,897	2,350	859	30,027	1,053	648	44,865	1,065
Male	87,366	5,399	369	10,181	25,545	1,589	545	15,971	620	139	26,313	696
Female	59,238	1,865	365	7,660	14,352	761	314	14,057	434	509	18,552	369
DIVISION 77 RENTAL AND			S									
Total	3,673	224	-	1,660	675	-	77	192	40	41	674	92
Male	2,866	133	-	1,079	675	-	77	96	40	-	674	92
Female	807	90	-	581	-	-	-	96	-	41	-	-
Renting and leasing of mot												
Total	1,121	-	-	93	510	-	-	96	-	-	331	92
Male	1,121	-	-	93	510	-	-	96	-	-	331	92
Female	-	-	-	-	-	-	-	-	-	-	-	-
Renting and leasing of pers		ousehold (goods									
Total	2,507	224	-	1,522	166	-	77	96	40	41	343	-
Male	1,700	133	-	941	166	-	77	-	40	-	343	-
Female	807	90	-	581	-	-	-	96	-	41	-	-
Renting and leasing of other	-	y, equipm	ent and ta	ingible good	ds							
Total	45	-	-	45	-	-	-	-	-	-	-	-
Male	45	-	-	45	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Leasing of intellectual prop	erty and sin	nilar produ	ıcts, exce _l	ot copyright	ed works							
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
DIVISION 78 EMPLOYMEN												
Total	23,943	762	98	1,483	6,702	391	209	2,838	249	49	10,899	263
Male	17,838	581	23	1,407	4,231	341	209	2,115	134	49	8,485	263
Female	6,105	181	75	75	2,471	50	-	722	116	-	2,414	-
Activities of employment pl	acement ag	encies										
Total	3,624	267	-	597	640	-	-	241	-	-	1,708	171
Male	3,624	267	-	597	640	-	-	241	-	-	1,708	171
Female	-	-	-	-	-	-	-	-	-	-	-	-
Temporary employment ag	ency activiti	es										
Total	4,727	133	-	94	522	-	-	95	134	-	3,750	-
Male	3,286	133	-	94	-	-	-	95	93	-	2,871	-
Female	1,441	-	-	-	522	-	-	-	40	-	879	-

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
Minor Industrial Divisions	I						Benishangul-				Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Other human resources pro	ovision			•		•						
Total	15,593	361	98	792	5,541	391	209	2,502	116	49	5,442	92
Male	10,929	181	23	717	3,591	341	209	1,780	40	49	3,906	92
Female	4,664	181	75	75	1,950	50	-	722	75	-	1,535	-
DIVISION 79 TRAVEL AGE	NCY, TOUR	OPERAT	TOR, RES	SERVATION	I SERVICE	AND REL	ATED ACTIVIT	ΓIES				
Total	7,090	918	-	206	1,573	73	-	721	-	-	3,507	92
Male	4,711	918	-	104	346	73	-	625	-	-	2,644	-
Female	2,380	-	-	102	1,227	-	-	96	-	-	863	92
Travel agency and tour ope		ies										
Total	5,801	765	-	206	1,494	73	-	625	-	-	2,637	-
Male	3,798	765	-	104	267	73	-	625	-	-	1,964	-
Female	2,003	-	-	102	1,227	-	-	-	-	-	674	-
Other reservation service a												
Total	1,290	153	-	-	79	-	-	96	-	-	870	92
Male	913	153	-	-	79	-	-	-	-	-	681	-
Female	377	-	-	-	-	-	-	96	-	-	189	92
DIVISION 80 SECURITY AN		GATION A	ACTIVITIE	≣S								
Total	21,742	1,162	-	4,531	5,544	202	-	3,876	104	-	6,177	146
Male	17,863	802	-	3,136	4,783	151	-	2,974	104	-	5,840	73
Female	3,879	360	-	1,395	761	50	-	902	-	-	337	73
Private security activities												
Total	2,143	-	-	-	229	-	-	287	-	-	1,627	-
Male	2,026	-	-	-	112	-	-	287	-	-	1,627	-
Female	116	-	-	-	116	-	-	-	-	-	-	-
Security systems service a												
Total	16,729	802	-	4,282	3,498	-	-	3,349	104	-	4,550	146
Male	13,899	802	-	2,888	3,374	-	-	2,447	104	-	4,212	73
Female	2,831	-	-	1,395	124	-	-	902	-	-	337	73
Investigation activities												
Total	2,870	360	-	249	1,818	202	-	241	-	-	-	-
Male	1,938	-	-	249	1,297	151	-	241	-	-	-	-
Female	932	360	-	-	521	50	-	-	-	-	-	-
DIVISION 81 SERVICES TO												
Total	20,172	713	224	3,327	2,089	284	-	2,939	22	229	10,344	-
Male	6,507	215	-	1,417	1,312		-	520	22	49	2,971	-
Female	13,665	498	224	1,910	777	284	-	2,419	-	180	7,373	-

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Combined facilities support	activities											
Total	3,332	-	-	1,490	522	-	-	-	22	49	1,248	-
Male	2,505	-	-	894	522	-	-	-	22	49	1,017	-
Female	826	-	-	596	-	-	-	-	-	-	230	-
Cleaning activities												
Total	15,060	498	224	1,837	1,567	284	-	1,793	-	180	8,677	-
Male	3,363	-	-	523	790	-	-	96	-	-	1,954	-
Female	11,697	498	224	1,314	777	284	-	1,696	-	180	6,723	-
Landscape care and mainte	enance serv	ice activiti	es									
Total	1,781	215	-	-	-	-	-	1,146	-	-	419	-
Male	639	215	-	-	-	-	-	424	-	-	-	-
Female	1,142	-	-	-	-	-	-	723	-	-	419	-
DIVISION 82 OFFICE ADMI	NISTRATIV	E, OFFIC	E SUPPC	RT AND O	THER BUS	INESS SL	JPPORT ACTI\	/ITIES				
Total	69,983	3,485	412	6,634	23,315	1,400	574	19,462	638	329	13,264	472
Male	37,581	2,749	347	3,037	14,199	1,022	259	9,639	320	41	5,700	268
Female	32,402	736	65	3,597	9,116	377	314	9,822	318	288	7,565	204
Office administrative and s	upport activi	ties										
Total	52,279	1,217	347	3,857	17,066	1,239	275	17,906	443	230	9,407	292
Male	26,956	719	347	2,303	8,909	862	143	9,083	187	41	4,274	88
Female	25,323	498	-	1,554	8,157	377	132	8,823	256	189	5,133	204
Activities of call centers												
Total	2,273	-	-	101	891	-	-	603	40	-	637	-
Male	522	-	-	-	522	-	-	-	-	-	-	-
Female	1,751	-	-	101	370	-	-	603	40	-	637	-
Organization of convention	s and trade	shows										
Total	1,195	-	-	719	-	-	143	-	80	-	253	-
Male	353	-	-	273	-	-	-	-	80	-	-	-
Female	842	-	-	446	-	-	143	-	-	-	253	-
Business support service a	ctivities not	elsewhere	e classifie	d								
Total	14,237	2,269	65	1,957	5,357	160	155	953	74	99	2,968	180
Male	9,751	2,030	-	461	4,768	160	116	556	53	-	1,426	180
Female	4,486	238	65	1,496	589	-	39	397	22	99	1,542	-
SECTION O: PUBLIC ADMIN	NISTRATIO	N AND DE	FENCE;	COMPULS	ORY SOCI	AL SECU	RI					
Total	349,400	27,395	6,243	82,625	81,495	6,371	10,615	63,604	3,733	7,190	54,729	5,399
Male	235,776	18,775	4,298	60,109	55,346	4,915	7,119	41,426	2,223	4,963	32,925	3,677
Female	113,623	8,620	1,945	22,516	26,149	1,456	3,497	22,178	1,510	2,227	21,804	1,722

5

Table 5.10 Cont'd

Table 5.10 Cont d						D.	ogion					
Minor Industrial Divisions	1	Г	1	1		K	egion Banjahangul			I	۸ ddia	Diro
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
DIVISION 84 PUBLIC ADMI								51414111	Janibola	· iaiaii	, waba	Dava
Total	349,400	27,395	6,243	82,625	81,495	6,371	10,615	63,604	3,733	7,190	54,729	5,399
Male	235,776	18,775	4,298	60,109	55,346	4,915	7,119	41,426	2,223	4,963	32,925	3,677
Female	113,623	8,620	1,945	22,516	26,149	1,456	3,497	22,178	1,510	2,227	21,804	1,722
Administration of the State						1,100	0, 107	22,170	1,010	2,221	21,001	1,722
Total	219,830	12,858	4,120	54,528	54,301	2,912	8,025	39,637	1,692	3,470	35,219	3,068
Male	135,114	7,374	2,861	38,459	33,052	1,908	5,077	23,727	993	2,071	18,005	1,586
Female	84,717	5,484	1,259	16,070	21,250	1,004	2,947	15,910	699	1,399	17,214	1,481
Provision of services to the	,			,	,	,	,	,		,	,	,
Total	119,369	14,538	1,866	25,083	24,231	3,342	2,353	22,615	1,793	3,452	17,766	2,331
Male	94,095	11,401	1,405	20,061	19,911	2,889	2,041	16,936	1,079	2,721	13,560	2,091
Female	25,273	3,137	461	5,022	4,319	453	312	5,679	714	731	4,206	240
Compulsory social security	activities											
Total	10,201	-	257	3,013	2,962	118	238	1,352	248	269	1,745	-
Male	6,567	-	32	1,589	2,383	118	-	763	151	171	1,361	-
Female	3,634	-	224	1,424	579	-	238	589	97	98	384	-
SECTION P: EDUCATION												
Total	469,184	37,627	5,085	111,552	120,381	6,296	7,812	98,077	4,681	4,977	66,485	6,211
Male	254,406	17,996	4,043	61,365	66,927	4,518	4,879	54,063	2,989	2,364	32,080	3,182
Female	214,778	19,631	1,042	50,187	53,454	1,778	2,932	44,015	1,692	2,613	34,405	3,029
DIVISION 85 EDUCATION												
Total	469,184	37,627	5,085	111,552	120,381	6,296	7,812	98,077	4,681	4,977	66,485	6,211
Male	254,406	17,996	4,043	61,365	66,927	4,518	4,879	54,063	2,989	2,364	32,080	3,182
Female	214,778	19,631	1,042	50,187	53,454	1,778	2,932	44,015	1,692	2,613	34,405	3,029
Pre-primary and primary ed	ducation											
Total	183,073	11,849	2,083	46,569	46,727	2,257	2,853	37,806	985	2,341	27,206	2,397
Male	83,267	2,924	1,285	20,433	22,434	1,563	1,675	18,994	405	1,206	11,234	1,114
Female	99,807	8,925	798	26,136	24,293	693	1,178	18,812	581	1,135	15,972	1,284
Secondary education												
Total	137,096	13,074	2,512	27,601	38,237	2,055	2,630	26,096	2,522	877	20,241	1,251
Male	94,383	8,716	2,333	20,069	26,529	1,517	1,579	18,407	1,991	595	11,747	900
Female	42,713	4,357	179	7,532	11,708	538	1,051	7,689	532	282	8,494	351
Higher education												
Total	107,029	9,684	235	26,217	23,238	1,018	1,005	30,579	755	1,424	11,318	1,556
Male	53,127	4,138	180	14,574	11,278	604	708	14,902	367	465	5,660	253
Female	53,903	5,546	55	11,643	11,960	414	297	15,678	389	959	5,659	1,303

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Other education												
Total	19,143	1,876	214	3,435	4,797	753	815	2,478	216	98	4,174	285
Male	13,134	1,723	214	3,360	2,596	713	553	1,123	134	49	2,383	285
Female	6,009	153	-	75	2,201	40	262	1,355	82	49	1,792	-
Educational support activiti	es											
Total	22,843	1,144	41	7,730	7,382	214	508	1,118	202	237	3,546	721
Male	10,496	494	32	2,929	4,090	121	365	637	93	49	1,057	630
Female	12,346	650	10	4,801	3,291	92	143	481	109	188	2,489	91
SECTION Q: HUMAN HEAL	TH AND SO	CIAL WO	RK ACTI	VITIES								
Total	189,014	15,351	3,147	31,631	54,186	2,762	5,353	28,964	1,537	3,093	39,654	3,335
Male	78,247	6,514	1,040	15,923	21,624	1,592	2,483	13,130	673	1,318	12,593	1,355
Female	110,766	8,837	2,107	15,708	32,561	1,170	2,870	15,834	864	1,774	27,060	1,980
DIVISION 86 HUMAN HEAL	TH ACTIVI	TIES										
Total	179,401	15,198	3,147	31,252	50,037	2,713	5,283	28,869	1,381	2,983	35,385	3,153
Male	74,445	6,361	1,040	15,862	19,826	1,542	2,483	13,130	539	1,208	11,098	1,355
Female	104,956	8,837	2,107	15,391	30,211	1,170	2,800	15,739	842	1,774	24,287	1,797
Hospital activities												
Total	90,209	9,989	2,278	12,572	27,365	1,289	2,338	13,390	605	1,794	16,814	1,776
Male	34,418	3,914	765	7,579	9,654	712	1,038	4,432	83	782	4,706	753
Female	55,791	6,075	1,513	4,993	17,710	577	1,300	8,958	521	1,012	12,108	1,023
Medical and dental practice	e activities											
Total	4,201	90	-	316	357	-	163	-	-	49	3,046	179
Male	2,352	-	-	101	118	-	-	-	-	49	2,083	-
Female	1,849	90	-	215	239	-	163	-	-	-	962	179
Other human health activiti	es scription											
Total	84,991	5,118	869	18,364	22,316	1,424	2,782	15,479	776	1,140	15,526	1,197
Male	37,676	2,447	276	8,181	10,054	831	1,445	8,698	456	377	4,309	603
Female	47,315	2,671	593	10,183	12,262	593	1,337	6,781	321	762	11,217	595
DIVISION 87 RESIDENTIAL	CARE ACT	ΓIVITIES										
Total	4,227	-	-	181	1,359	50	-	96	22	-	2,337	183
Male	772	-	-	62	120	50	-	-	-	-	541	-
Female	3,455	-	-	119	1,239	-	-	96	22	-	1,796	183
Residential nursing care fa	cilities											
Total	1,170	-	-	181	596	50	-	-	-	-	344	-
Male	183	-	-	62	72	50	-	-	-	-	-	-
Female	987	-	-	119	524	-	-	-	-	-	344	-

Table 5.10 Cont'd

Minor Industrial Divisions						R	egion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Residential care activities f	or mental re	tardation,	mental he	ealth and si	ubstance al	ouse						
Total	342	-	-	-	-	-	-	-	-	-	342	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	342	-	-	-	-	-	-	-	-	-	342	-
Residential care activities f	or the elderl	y and disa	abled									
Total	92	-	-	-	-	-	-	-	-	-	-	92
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	92	-	-	-	-	-	-	-	-	-	-	92
Other residential care active	vities .											
Total	2,622	-	-	-	763	-	-	96	22	-	1,650	91
Male	589	-	-	-	48	-	-	-	-	-	541	-
Female	2,033	-	-	-	715	-	-	96	22	-	1,109	91
DIVISION 88 SOCIAL WOR			OUT ACC	OMMODAT	ΓΙΟΝ							
Total	5,386	153	-	198	2,790	-	70	-	134	110	1,932	-
Male	3,030	153	-	-	1,679	-	-	-	134	110	954	-
Female	2,356	-	-	198	1,111	-	70	-	-	-	978	-
Social work activities withou	out accommo	dation for	the elder	ly and disal	bled							
Total	214	-	-	104	-	-	-	-	-	110	-	-
Male	110	-	-	-	-	-	-	-	-	110	-	-
Female	104	-	-	104	-	-	-	-	-	-	-	-
Other social work activities	without acc	ommodati	ion									
Total	5,172	153	-	94	2,790	-	70	-	134	-	1,932	-
Male	2,920	153	-	-	1,679	-	-	-	134	-	954	-
Female	2,252	-	-	94	1,111	-	70	-	-	-	978	-
SECTION R: ARTS, ENTER												
Total	49,289	3,960	673	12,003	9,490		405	6,238		287	15,350	521
Male	28,878	2,655	460	4,685	7,678	173	253	3,357	144	81	8,957	435
Female	20,411	1,305	214	7,318	1,811	-	152	2,881	45	206	6,393	86
DIVISION 90 CREATIVE, A	RTS AND E	NTERTAL	NMENT A	CTIVITIES	;							
Total	3,027	272	-	103	93	-	-	546	-	-	2,013	-
Male	2,039	137	-	103	47	-	-	403	-	-	1,350	-
Female	988	135	-	-	47	-	-	143	-	-	663	-
Creative, arts and entertain	nment activit	ies										
Total	3,027	272	-	103	93	-	-	546	-	-	2,013	-
Male	2,039	137	-	103	47	-	-	403	-	-	1,350	-
Female	988	135	-	-	47	-	-	143	-	-	663	-

Table 5.10 Cont'd

						R	egion					
Minor Industrial Divisions						1	Benishangul-				Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
DIVISION 91 LIBRARIES, A	RCHIVES, I		S AND O						I	l l	L	
Total	11,561	1,316	246	1,739	1,365	-	-	2,918	-	125	3,689	163
Male	5,160	453	246	577	1,188	-	-	1,190	-	41	1,303	163
Female	6,401	863	-	1,163	177	-	-	1,728	-	84	2,387	-
Libraries, archives, museur	ms and othe	r cultural a	activities									
Total	11,561	1,316	246	1,739	1,365	-	-	2,918	-	125	3,689	163
Male	5,160	453	246	577	1,188	-	-	1,190	-	41	1,303	163
Female	6,401	863	-	1,163	177	-	-	1,728	-	84	2,387	-
DIVISION 92 GAMBLING A	ND BETTIN	G ACTIVI	ΓIES									
Total	3,619	259	-	2,756	301	52	129	-	-	122	-	-
Male	531	259	-	91	-	52	129	-	-	-	-	-
Female	3,088	-	-	2,664	301	-	-	-	-	122	-	-
Gambling and betting activ	ities											
Total	3,619	259	-	2,756	301	52	129	-	-	122	-	-
Male	531	259	-	91	-	52	129	-	-	-	-	-
Female	3,088	-	-	2,664	301	-	-	-	-	122	-	-
DIVISION 93 SPORTS ACT												
Total	31,082	2,113	428	7,405	7,730	121	275	2,774	189	41	9,648	358
Male	21,148	1,807	214	3,914	6,444	121	123	1,764	144	41	6,305	272
Female	9,934	306	214	3,491	1,286	-	152	1,011	45	-	3,343	86
Sports activities												
Total	15,539	287	428	1,508	4,390	-	84	996	122	-	7,494	232
Male	10,720	133	214	731	4,307	-	84	301	122	-	4,682	146
Female	4,820	153	214	776	83	-	-	696	-	-	2,811	86
Other amusement and recr												
Total	15,542	1,826	-	5,897	3,340	121	191	1,778	67	41	2,154	126
Male	10,428	1,673	-	3,183	2,137	121	39	1,463	22	41	1,623	126
Female	5,114	153	-	2,715	1,203	-	152	315	45	-	532	-
SECTION S: OTHER SERVI												
Total	440,377	23,070	3,010	93,769	147,951	7,028	3,052	69,200	4,878	3,357	77,147	7,916
Male .	288,385	9,291	2,086	60,685	104,997	6,040	2,225	45,740	2,330	2,652	46,549	5,792
Female	151,992	13,780	924	33,084	42,954	988	827	23,460	2,548	705	30,598	2,124
DIVISION 94 ACTIVITIES C					05.01-						04.5==	
Total	94,442	4,112	658	21,015	35,642	302	376	9,431	469	223	21,372	843
Male	72,542	3,452	658	15,502	28,393	211	376	7,612	235	223	15,218	662
Female	21,899	659	-	5,513	7,249	90	-	1,819	235	-	6,154	181

Table 5.10 Cont'd

AA' I I I I I I I						Re	gion					
Minor Industrial Divisions							Benishangul-				Addis	Dire
and sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Activities of business, empl	loyers and p	profession	al membe	rship organ	izations					•		
Total	14,354	783	-	3,545	5,376	-	-	603	190	-	3,683	175
Male	7,397	349	-	2,503	2,398	-	-	603	93	-	1,365	86
Female	6,957	434	-	1,042	2,978	-	-	-	97	-	2,319	88
Activities of trade unions												
Total	20,278	975	331	3,204	11,196	50	-	768	22	-	3,560	173
Male	13,146	975	331	1,647	8,256	-	-	674	22	-	1,068	173
Female	7,132	-	-	1,557	2,940	50	-	93	-	-	2,492	-
Activities of other members												
Total	59,810	2,354	327	14,266	19,070	252	376	8,061	258	223	14,129	495
Male	52,000	2,128	327	11,352	17,739	211	376	6,335	120	223	12,785	403
Female	7,810	226	-	2,913	1,331	40	-	1,726	138	-	1,343	92
DIVISION 95 REPAIR OF C												
Total	32,353	1,409	227	5,256	15,404	314	86	2,984	382	270	5,842	178
Male .	29,311	1,276	227	4,378	14,323	234	86	2,984	214	270	5,227	92
Female	3,042	133		879	1,081	80	-	-	167	-	615	86
Repair of computers and co												
Total	2,483	191	10	473	1,184	52	-	93	-	50	344	86
Male	2,397	191	10	473	1,184	52	-	93	-	50	344	-
Female	86	-	-	-	-	-	-	-	-	-	-	86
Repair of personal and hou	•											
Total	29,870	1,218	218	4,783	14,221	262	86	2,891	382	221	5,498	92
Male	26,915	1,085	218	3,905	13,139	182	86	2,891	214	221	4,883	92
Female	2,956	133	-	879	1,081	80	-	-	167	-	615	-
DIVISION 96 OTHER PERS	ONAL SER	VICE ACT	TIVITIES									
Total	313,583	17,549	2,125	67,498	96,905	6,412	2,589	56,784	4,027	2,864	49,933	6,895
Male	186,531	4,562	1,201	40,805	62,281	5,595	1,763	35,143	1,881	2,159	26,104	5,038
Female	127,051	12,987	924	26,693	34,624	818	827	21,641	2,146	705	23,830	1,857
Other personal service acti	vities											
Total	313,583	17,549	2,125	67,498	96,905	6,412	2,589	56,784	4,027	2,864	49,933	6,895
Male	186,531	4,562	1,201	40,805	62,281	5,595	1,763	35,143	1,881	2,159	26,104	5,038
Female	127,051	12,987	924	26,693	34,624	818	827	21,641	2,146	705	23,830	1,857
SECTION T: ACTIVITIES OF SERVICES-PRODUCINGAC						ATED GOO	DDS AND					
Total	606,525	34,959	1,938	130,076	163,314	15,118	9,662	89,617	11,671	4,642	134,455	11,072
Male	139,157	7,274	343	20,687	45,301	4,224	3,879	20,964	2,567	679	31,296	1,943
Female	467,368	27,685	1,595	109,389	118,014	10,894	5,783	68,653	9,104	3,963	103,158	9,129
i Giliaic	401,300	21,000	1,090	103,303	110,014	10,034	5,705	00,000	3,104	5,505	103,130	3,123

30

Table 5.10 Cont'd

Minor Industrial Divisions						Re	egion					
and sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
DIVISION 97 ACTIVITIES O	F HOUSEH	OLDS AS	EMPLO'	ERS OF D	OMESTIC I	PERSON	NEL					
Total	444,881	27,034	1,684	86,669	106,437	9,416	3,304	74,643	7,437	4,413	114,591	9,252
Male	107,960	6,779	309	18,295	29,764	2,092	811	18,461	1,607	581	27,405	1,856
Female	336,921	20,255	1,375	68,375	76,673	7,324	2,493	56,183	5,830	3,831	87,187	7,396
Activities of households as	employers of	of domest	ic person	nel								
Total	444,881	27,034	1,684	86,669	106,437	9,416	3,304	74,643	7,437	4,413	114,591	9,252
Male	107,960	6,779	309	18,295	29,764	2,092	811	18,461	1,607	581	27,405	1,856
Female	336,921	20,255	1,375	68,375	76,673	7,324	2,493	56,183	5,830	3,831	87,187	7,396
DIVISION 98 UNDIFFERENTIA	ATED GOO	DS- AND	SERVICE	S-PRODUC	CING ACTIV	VITIES OF	PRIVATE HO	USEHOLD:	SFOR OWI	N USE		
Total	161,644	7,925	254	43,407	56,877	5,702	6,358	14,974	4,235	229	19,863	1,820
Male	31,197	495	34	2,392	15,537	2,132	3,068	2,504	960	98	3,891	86
Female	130,447	7,430	220	41,014	41,340	3,570	3,290	12,470	3,275	131	15,972	1,733
Undifferentiated goods-pro	ducing activ	ities of pri	vate hous	eholds for o	own use							
Total	27,452	650	-	-	17,513	40	3,213	4,205	343	-	1,361	126
Male	6,841	133	-	-	5,490	40	1,107	-	70	-	-	-
Female	20,611	517	-	-	12,023	-	2,106	4,205	272	-	1,361	126
Undifferentiated service-pro	oducing acti	vities of p	rivate hou	seholds for	own							
Total	134,192	7,274	254	43,407	39,364	5,662	3,145	10,769	3,892	229	18,502	1,694
Male	24,356	361	34	2,392	10,047	2,092	1,961	2,504	890	98	3,891	86
Female	109,836	6,913	220	41,014	29,318	3,570	1,184	8,265	3,002	131	14,611	1,607
SECTION U: ACTIVITIES OF	EXTRATE	RRITORI	AL ORGA	NIZATION	S AND BO	DIESDES	CRIPTION					
Total	78,337	6,757	1,122	8,574	19,391	2,351	742	11,353	1,990	514	24,828	716
Male	47,522	4,891	543	5,221	10,780	1,943	534	7,846	1,512	343	13,477	432
Female	30,815	1,866	579	3,353	8,611	408	209	3,507	478	171	11,351	284
DIVISION 99 ACTIVITIES O	F EXTRATE	ERRITOR	IAL ORG	ANIZATION	S AND BOI	DIES						
Total	78,337	6,757	1,122	8,574	19,391	2,351	742	11,353	1,990	514	24,828	716
Male	47,522	4,891	543	5,221	10,780	1,943	534	7,846	1,512	343	13,477	432
Female	30,815	1,866	579	3,353	8,611	408	209	3,507	478	171	11,351	284
Activities of extraterritorial	organization	s and boo	dies									
Total	78,337	6,757	1,122	8,574	19,391	2,351	742	11,353	1,990	514	24,828	716
Male	47,522	4,891	543	5,221	10,780	1,943	534	7,846	1,512	343	13,477	432
Female	30,815	1,866	579	3,353	8,611	408	209	3,507	478	171	11,351	284

Table 5.11 Employed Population of Urban Areas Aged Ten Years and Above by Status in Employment, Age Group and Sex - COUNTRY TOTAL: 2015

		CI IOIAL				Statu	s in Employ	ment						
				Paid Em	nlovees	0.00	=							
Age group and sex	Total Employed Persons	Total paid Employees	Employee - Government	Employee -Government Development organization	Employee -Private organization	Employee -NGO (including International organization)	Employee Domestic	Other Employee	Member of Co-Operatives	Self Employed	Unpaid Family Worker	Employer	Apprentice	Others
All Ages	•						•							
Total	7,083,094	3,442,585	1,052,822	481,824	1,382,519	76,161	375,369	73,889	54,547	2,873,646	591,681	58,825	9,716	52,093
Male	3,963,143	2,024,497	612,062	314,601	901,795	45,744	89,678	60,618	37,978	1,602,692	217,021	38,412	7,077	35,466
Female	3,119,950	1,418,088	440,761	167,223	480,724	30,418	285,691	13,271	16,568	1,270,954	374,660	20,413	2,639	16,627
10-14														
Total	194,969	31,323	-	-	6,196	-	24,130	997	153	25,616	137,310	-	-	567
Male	100,707	7,173	-	-	3,996	-	2,490	687	-	16,947	76,020	-	-	567
Female 15-19	94,262	24,150	-	-	2,200	-	21,640	311	153	8,669	61,290	-	-	-
Total	565,103	271,943	10,011	9,880	125,919	856	121,351	3,927	4,659	136,095	142,065	479	5,285	4,578
Male	224,956	78,855	1,846	3,720	57,242	63	12,794	3,190	2,538	72,818	63,527	479	4,532	2,206
Female	340,148	193,088	8,165	6,159	68,677	793	108,557	736	2,121	63,277	78,538	-	753	2,371
20-24														
Total	1,107,918	618,034	137,690	64,759	312,511	8,135	84,388	10,550	12,407	375,318	86,819	4,073	884	10,382
Male	546,853	290,460	55,919	29,918	175,466	3,742	17,798	7,617	8,102	197,870	39,058	1,990	843	8,532
Female 25-29	561,064	327,574	81,771	34,841	137,045	4,394	66,590	2,933	4,306	177,449	47,761	2,083	41	1,851
Total	1,455,213	821,567	279,994	108,424	332,864	16,230	69,802	14,253	12,821	523,593	75,949	7,748	2,835	10,701
Male	827,711	483,809	150,073	66,691	227,034	7,291	22,092	10,628	9,817	300,932	19,069	4,849	1,340	7,895
Female 30-34	627,502	337,759	129,922	41,733	105,829	8,939	47,710	3,625	3,003	222,661	56,880	2,898	1,495	2,806
Total	985,745	518,861	179,580	79,150	213,171	13,025	24,029	9,908	6,219	407,167	34,620	11,283	-	7,595
Male	599,578	338,461	110,719	53,694	148,382	7,457	8,809	9,400	5,189	238,006	5,505	7,489	-	4,928
Female 35-39	386,166	180,400	68,861	25,455	64,789	5,567	15,220	508	1,030	169,160	29,114	3,794	-	2,667
Total	935,320	422,019	160,117	69,849	149,346	9,727	22,598	10,382	7,568	441,107	48,285	6,403	350	9,589
Male	518,080	258,251	88,180	46,314	100,386	4,751	9,859	8,760	4,780	238,989	4,776	4,993	-	6,291
Female	417,240	163,767	71,936	23,535	48,960	4,976	12,739	1,622	2,788	202,118	43,508	1,411	350	3,298
40-44	F70 200	05/ 140	05.000	47.000	05 500	0.504	0.754	7.050	0.775	200 752	47.054	0.454	212	1.040
Total	578,390	256,149	95,988	47,383	85,580	9,584	9,754	7,859	3,775	289,758	17,051	9,451	362	1,843

Male	365,839	184,705	64,321	35,009	64,485	8,019	5,105	7,767	2,535	167,815	3,028	6,584	362	811
Female	212,551	71,444	31,666	12,374	21,096	1,565	4,649	92	1,240	121,943	14,024	2,868	-	1,032

Table 5.11 Cont'd

		•				Status	s in Employn	nent		•				•
				Paid Emp	oloyees									
Age group	Total Employed	Total paid	Employee-	Employee- Government Development	Employee -Private	Employee -NGO (including International	Employee	Other	Member of	Self	Unpaid Family			
and sex	Persons	Employees	Government	organization	organization		Domestic	Employee	Co-Operatives		Worker	Employer	Apprentice	Others
45-49				ı . g	1 · g			1				1	1 12 2	,
Total	421,698	187,196	79,953	40,464	50,219	6,609	5,087	4,864	2,654	207,030	16,922	5,163	-	2,732
Male	252,303	134,721	56,874	33,353	34,043	4,679	1,887	3,884	1,626	109,701	2,489	2,680	-	1,087
Female	169,394	52,475	23,079	7,110	16,175	1,930	3,200	980	1,029	97,329	14,433	2,483	-	1,645
50-54														
Total	288,013	131,463	51,579	33,933	32,353	5,817	4,348	3,432	1,203	139,754	8,932	4,691	-	1,970
Male	173,544	98,103	39,722	22,838	26,726	4,194	2,075	2,548	845	69,965	313	2,742	-	1,575
Female	114,470	33,360	11,857	11,095	5,627	1,624	2,273	884	358	69,789	8,618	1,949	-	395
55-59														
Total	210,275	86,590	35,648	14,121	28,465	2,837	2,152	3,366	1,209	110,514	8,322		-	579
Male	128,640	64,415	25,412	10,095	21,836	2,550	1,498	3,024	697	60,488	687	1,991	-	363
Female	81,635	22,174	10,236	4,026	6,629	287	654	342	512	50,026	7,636	1,071	-	215
60-64														
Total	145,598	44,508	11,254	5,965	23,077	1,172	2,305	734	826	91,544	6,445		-	261
Male	90,332	39,872	9,964	5,943	20,312	830	2,213	610	826	47,469	637	1,268	-	261
Female	55,266	4,636	1,290	22	2,766	342	92	124	-	44,075	5,808	746	-	-
65+														
Total	194,851	52,932	11,008	7,896	22,819	2,169	5,424	3,617	1,052	126,151	8,961	4,457	-	1,297
Male	134,599	45,672	9,031	7,024	21,888	2,169	3,057	2,503	1,024	81,693	1,910	3,348	-	951
Female	60,253	7,260	1,977	871	931	-	2,366	1,114	28	44,458	7,051	1,109	-	346

Table 5.12 Employed Population of Urban Areas Aged Ten Years and Above by Status in Employment, Region and Sex: 2015

Table 5.12 Emp	pioyed Po	pulation (of Urban Are	as Aged I	en Years a				pioyment, F	region a	nd Sex	: 2015		1
		1	r	T	I		in Employm	ent		1	1	1	1	
				Employee-		Employee- NGO								
Region				Government	Employee-	(including			Member of		Unpaid			
and Sex	.	Total paid	Employee-	Development	Private	International	Employee	Other	Co-	Self	Family			011
	Total	Employees	Government	organization	organization	organization)	Domestic	Employee	Operatives	Employed	worker	Employer	Apprentice	Others
COUNTRY TOTAL														
Total	7,083,094	3,442,585	1,052,822	481,824	1,382,519	76,161	375,369	73,889		2,873,646		58,825		52,093
Male	3,963,143	2,024,497	612,062	314,601	901,795	45,744	89,678	60,618		1,602,692		38,412		35,466
Female	3,119,950	1,418,088	440,761	167,223	480,724	30,418	285,691	13,271	16,568	1,270,954	374,660	20,413	2,639	16,627
TIGRAY REGION														
Total	477,813	235,334	80,184	26,477	89,058	5,903	30,096	3,617	11,934	209,094	14,224	3,457	1,608	2,161
Male	253,254	126,241	46,044	18,025	48,857	3,778	6,642	2,895	5,930	110,567	5,006	2,573	1,353	1,583
Female	224,559	109,093	34,140	8,452	40,201	2,124	23,454	722	6,004	98,527	9,218	884	255	578
AFFAR REGION														
Total	131,439	64,297	15,876	21,743	21,252	1,098	1,600	2,727	347	46,719	18,843	795	-	438
Male	75,452	42,510	10,415	14,452	14,095	543	278	2,727	331	25,892	6,166	328	-	224
Female	55,987	21,786	5,460	7,292	7,157	555	1,322	-	16	20,828	12,677	466	-	214
AMHARA REGION														
Total	1,449,091	628,384	235,757	82,165	214,524	8,937	69,374	17,628	8,703		159,512	10,283	2,316	9,522
Male	795,183	378,890	147,677	56,118	138,603	5,577	17,158	13,757	6,966	336,491	59,228	7,529	2,215	3,865
Female	653,908	249,494	88,080	26,047	75,921	3,360	52,216	3,871	1,737	293,880	100,284	2,754	101	5,657
OROMIA REGION														
Total	2,102,417	944,915	261,869	126,786	409,567	19,026	98,012	29,656	18,350	910,290		12,457		19,591
Male	1,213,723	589,684	156,565	87,530	281,621	10,449	26,906	26,612	13,022	511,517		8,960	2,276	15,029
Female	888,694	355,231	105,304	39,256	127,945	8,577	71,105	3,044	5,329	398,774	120,641	3,498	660	4,562
SOMALI REGION														
Total	119,082	42,140	16,655	7,621	10,288	2,190	5,141	245	1,000	57,726	16,582	967	81	586
Male	67,114	28,855	11,378	6,312	7,783	1,783	1,394	205	778	31,322	5,091	624	81	364
Female	51,967	13,284	5,277	1,308	2,504	408	3,747	40	222	26,405	11,491	343	-	223
BENISHANGUL-GUMU	Z REGION													
Total	107,053	45,856	24,301	6,943	11,478	703	2,203	228	208	39,196	20,664	880	29	220
Male	58,730	29,216	14,603	4,815	8,662	495	442	199	93	21,479	7,248	443	29	220
Female	48,323	16,640	9,698	2,128	2,816	209	1,761	29	114	17,717	13,415	437	-	-
S.N.N.P. REGION														
Total	1,096,887	475,132	213,944	61,548	143,598	12,166	38,105	5,770	6,009	474,911	120,258	7,773	1,322	11,484
Male	603,653	285,604	121,452	41,555	100,061	8,248	9,909	4,378	4,593	259,723	40,154	3,902	600	9,078
Female	493,233	189,528	92,492	19,993	43,537	3,918	28,196	1,392	1,416	215,188	80,104	3,870	722	2,406

=

Table 5.12 Cont'd

Table 3.12 Cont						01 :								
						Status	in Employm	ent			_	•		
						Employee-								
				Employee-		NGO								
Region				Government	Employee-	(including			Member of		Unpaid			
and Sex		Total paid	Employee-	Development	Private	International	Employee		Co-	Self	Family			
	Total	Employees	Government	organization	organization	organization)	Domestic	Employee	Operatives	Employed	Worker	Employer	Apprentice	Others
GAMBELLA REGION														
Total	61,809	25,235	10,754	5,666	4,652	1,990	1,848	325	535	20,030	15,557	274	179	-
Male	31,771	15,667	6,241	3,511	3,557	1,512	636	209	428	9,408	6,006	199	63	-
Female	30,038	9,568	4,513	2,155	1,095	478	1,211	115	107	10,622	9,551	75	116	-
HARARI REGION														
Total	59,109	30,567	15,461	1,836	9,992	514	2,454	311	190	24,990	2,679	584	49	50
Male	29,900	16,901	8,635	988	6,319	343	355	262	190	11,259	947	503	49	50
Female	29,210	13,666	6,826	848	3,673	171	2,100	49	-	13,731	1,731	81	-	-
ADDIS ABABA CITY ADM	MINISTRATION	ON												
Total	1,362,069	891,613	163,305	136,166	439,174	23,080	117,730	12,158	6,505	416,703	19,218	19,539	1,123	7,369
Male	772,367	477,298	80,647	77,808	273,339	12,745	24,276	8,484	5,007	263,516	9,815	11,922	338	4,470
Female	589,703	414,315	82,658	58,358	165,836	10,335	93,454	3,674	1,498	153,187	9,403	7,617	785	2,898
DIRE DAWA ADMINISTE	RATION													
Total	116,325	59,114	14,718	4,873	28,936	555	8,807	1,224	766	43,616	10,268	1,816	73	673
Male	61,997	33,631	8,404	3,486	18,898	271	1,681	890	640	21,519	4,124	1,427	73	583
Female	54,327	25,483	6,314	1,388	10,038	284	7,126	333	126	22,097	6,144	388	-	90

Table 5.13 Employed Population of Major Towns Aged Ten Years and Above by Status in Employment and Sex: 2015

1 able 5.13	Lilipioye	u ropulati	On Or Wajo	r Towns Ag	eu ren re				inployine	it and Se	X. 2013	1		
				Do	id Employees		tatus in Emp	loyment					1	I
				Pa	id Employees		Ι							
				F I		Employee-								
Major Town	Total			Employee-	-	NGO			N4					
and Sex	Employed	T. 1. 1	-	Government	Employee-	(including	-	Oll	Member of	Cult	Unpaid			
	Persons	Total paid	Employee	Development	Private	International	Employee	Other	Co-	Self	Family	F	A 12	Otto
MA IOD TOWN TO		Employees	-Government	organization	organization	organization)	Domestic	Employee	Operatives	Employed	Worker	Employer	Apprentice	Others
MAJOR TOWN TO		1 507 740	220.07	207 771	714 410	25.774	10/ 010	22.017	14.040	052.070	120.270	22.21/	2 (20	10.004
Total	2,560,286	1,506,742	339,967	206,671	714,412	35,764	186,912	23,016	14,048	853,060	139,369	32,216	2,629	
Male	1,443,652	830,971	180,519	126,365	446,889	21,779	38,676	16,742	10,937	512,713	57,658	21,212	1,604	8,558
Female	1,116,634	675,771	159,448	80,307	267,522	13,985	148,236	6,274	3,110	340,347	81,711	11,004	1,025	3,666
MEKELE TOWN	101 / 41	77.000	22.247	10.057	20 (5)	2.105	10.070	070	1 700	45.570	4.007	1.045	2//	00
Total	131,641	77,093	22,247	10,057	30,656	2,185	10,978	970	1,790	45,560	4,896	1,945	266	90
Male	70,849	41,976	11,652	7,310	18,232	1,534	2,505	745	1,067	24,596	1,793	1,061	266	90
Female	60,792	35,116	10,595	2,747	12,425	650	8,473	226	724	20,965	3,103	884	-	-
ASAYITA TOWN														
Total	12,819	6,361	2,167	2,445	1,134	107	508	-	16	5,711	697	34	-	-
Male	7,356	4,262	1,305	1,813	977	107	60	-	-	2,811	283	-	-	-
Female	5,463	2,099	862	632	157	-	448	-	16	2,899	415	34	-	-
GONDER TOWN														
Total	162,404	48,119	9,519	6,377	24,581	844	5,324	1,474	1,666	49,741	61,381	330	252	915
Male	85,621	27,382	5,056	4,402	15,203	535	915	1,271	1,562	31,568	23,820	330	151	810
Female	76,782	20,737	4,463	1,975	9,378	309	4,410	202	103	18,173	37,561	-	101	105
DESSIE TOWN	74 500	0, 0,0	40 ===		45.005									.=0
Total	71,530	36,949	12,775	3,403	15,907	232	4,182	450	61	29,895	3,106	955	87	478
Male	43,161	22,574	8,484	2,580	10,624	171	506	209	61	17,640	1,587	893	87	319
Female	28,369	14,374	4,291	823	5,283	61	3,676	241	-	12,255	1,518	62	-	160
BAHIR DAR TOWI														
Total	124,310	64,797	17,383	5,564	33,908	1,239	4,627	2,075	-	30,786	24,174	3,338	215	1,000
Male	66,389	34,591	9,619	3,836	18,970	735	93	1,338	-	17,875	10,447	2,420	215	840
Female	57,921	30,206	7,764	1,728	14,938	505	4,534	738	-	12,911	13,727	917	-	159
BISHOFTU														
TOWN		05 555			04 504							4.0		
Total	57,994	35,577	4,957	5,566	21,536	426	2,639	452	481	21,514	213	168	41	-
Male	33,733	21,461	3,295	3,614	12,585	228	1,287	452	268	11,778	58	168	-	-
Female	24,261	14,116	1,662	1,953	8,950	198	1,352	-	213	9,736	155	-	41	-
ADAMA TOWN	447.000	. =		7.000										=0.
Total	117,828	67,983	9,606	7,923	41,746	898	6,661	1,149	484	46,847	844	814	121	736
Male	69,953	38,823	6,035	4,920	23,788	732	2,443	905	365	29,223	357	450	121	615
Female	47,874	29,159	3,571	3,002	17,958	166	4,218	244	120	17,624	486	364	-	121
JIMMA TOWN	75.00-		4= 00=		A= A	,				0	0.00:			
Total	75,289	44,173	15,882	4,386	15,061	640	7,713	492	428	26,479	3,331	503	291	84
Male	42,862	24,135	7,769	2,811	11,320	246	1,621	367	344	16,210	1,453	441	194	84
Female	32,428	20,038	8,112	1,574	3,741	394	6,092	124	84	10,270	1,877	62	97	-

 $\frac{1}{2}$

Table 5.13 Cont'd

1 4510 0.10						S	Status in Emp	loyment							
				Pa	d Employees										
Major Town and Sex	Total Employed Persons	Total paid Employees	Employee -Government	Employee- Government Development	Employee- Private organization	Employee- NGO (including International organization)	Employee Domestic	Other Employee	C	iber of Co- ratives	Self Employed	Unpaid Family Worker	Employer	Approptice	Others
SHASHEMEN		Employees	-Government	organization	organization	organization)	Domestic	Employee	Opei	alives	Employed	worker	Employei	Apprentice	Others
Total	54,374	22,927	4,146	2,912	11,245	427	3,570		628	202	29,880	512	288	_	564
Male	32,171	14,806		2,912	8,344	427	3,570 896		561	202	16,228	304	148	-	483
iviale Female	22,203	8,121	2,398 1,748	731	2,901	427	2,673		67	202	13,652	207	148	-	483 82
JIJIGA TOWN		0,121	1,740	731	2,901	-	2,073		07	-	13,032	207	141	-	02
Total	40,815	19,031	8,038	2,796	4,028	1,103	2,935		131	477	18,784	1,651	569	81	222
Male	25,947	12,568		2,770	3,288	860	709		91	428	11,641	674	384	81	171
Female	14,868	6,463	2,493	720	740	243	2,226		40	50	7,143	976	185	-	51
ASOSA TOW		0,403	2,473	720	740	243	2,220		40	30	7,143	770	103	_	31
Total	21,112	12,604	6,901	950	3,698	277	550		228	208	7,210	906	113	29	43
Male	12,786	8,104	4,209	735	2,816	145	-		199	93	4,118	285	113	29	43
Female	8,326	4,500		215	881	132	550		29	114	3,091	621	-	-	-
HAWASSA TO		1,000	2,072	2.10	001	102	000		-/		0,071	021			
Total	124,702	71,945	25,528	6,859	30,030	2,060	6,518		950	491	47,923	3,344	1,000	_	_
Male	73,264	41,344	13,368	4,697	20,019	1,721	779		759	491	29,329	1,348	752	-	-
Female	51,439	30,601	12,160	2,162	10,011	339	5,738		190	_	18,593	1,996	248	-	-
GAMBELLA T															
Total	27,965	17,891	7,335	4,558	2,780	1,179	1,714		325	282	7,421	2,151	221	-	-
Male	15,295	11,113	4,098	3,108	2,168	981	549		209	219	3,402	362	199	-	-
Female	12,670	6,778	3,237	1,450	612	198	1,165		115	62	4,019	1,789	22	-	-
HARAR TOW	'N														
Total	59,109	30,567	15,461	1,836	9,992	514	2,454		311	190	24,990	2,679	584	49	50
Male	29,900	16,901	8,635	988	6,319	343	355		262	190	11,259	947	503	49	50
Female	29,210	13,666	6,826	848	3,673	171	2,100		49	-	13,731	1,731	81	-	-
ADDIS ABAB															
Total	1,362,069	891,613		136,166	439,174	23,080	117,730		,158	6,505	416,703	19,218	19,539	1,123	7,369
Male	772,367	477,298		77,808	273,339	12,745	24,276		,484	5,007	263,516	9,815	11,922	338	4,470
Female	589,703	414,315	82,658	58,358	165,836	10,335	93,454	3	,674	1,498	153,187	9,403	7,617	785	2,898
DIRE DAWA															
Total	116,325	59,114	14,718	4,873	28,936	555	8,807	1	,224	766	43,616	10,268	1,816	73	673
Male	61,997	33,631	8,404	3,486	18,898	271	1,681		890	640	21,519	4,124	1,427	73	583
Female	54,327	25,483	6,314	1,388	10,038	284	7,126		333	126	22,097	6,144	388	-	90

1/4

Table 5.14 Employed Population of Urban Areas Aged Ten Years and Above by Major Occupation Groups, Major Industrial Divisions and Sex - COUNTRY TOTAL: 2015

		1411(1 1017			Majo	or Occupation	nal Groups				
Major Industrial							Skilled				
Divisions				Technicians			agricultural,	Craft and			
and Sex				and	Clerical	Service	forestry	related	Plant and machine		
				associate	support	and sales	and fishery	trades	operators, and	Elementary	
L	Total	Managers	Professionals	professionals	workers	workers	workers	workers	assemblers	occupations	Others
Total Employed Pop											
Total	7,083,094	153,505	558,734		258,416	2,162,111	387,801	1,089,180	458,982	1,440,420	9,032
Male	3,963,143	117,550	369,274	344,949	90,129	912,282		696,336	418,261	736,433	7,254
Female	3,119,950	35,956	189,460	219,962	168,287	1,249,829	117,126	392,845	40,721	703,988	1,777
Agriculture, forestr		•									
Total	567,067	6,171	6,943	10,249	2,682	19,095	361,136	5,240	6,905	148,123	524
Male	382,749	5,059	4,847	7,719	641	9,733	250,637	3,507	5,803	94,803	-
Female	184,319	1,112	2,096	2,531	2,041	9,362	110,499	1,733	1,102	53,320	524
Mining and quarry	-										
Total	35,165	189	2,180	1,433	189	1,938	111	1,860	3,335	23,931	-
Male	28,221	189	1,729	693	-	1,938	111	1,860	3,277	18,423	-
Female	6,944	-	450	739	189	-	-	-	57	5,508	-
Manufacturing											
Total	929,847	11,646	10,919	21,234	9,789	178,728	5,826	477,301	101,946	112,335	123
Male	489,346	9,201	7,463	13,312	2,167	37,475	3,857	276,629	77,093	62,026	123
Female	440,501	2,445	3,456	7,922	7,622	141,253	1,969	200,672	24,853	50,309	-
Electricity, gas, ste	eam and air o	conditioning s	supply								
Total	21,693	90	2,131	2,822	1,822	1,345	-	7,122	2,123	4,238	-
Male	17,435	90	1,266	2,779	731	822	-	6,250	2,123	3,373	-
Female	4,258	-	865	42	1,091	523	-	872	-	865	-
Water supply; sew	erage, waste	e manageme	nt and remediati	on activities							
Total	55,487	3,688	2,995	4,514	1,921	3,299	594	6,445	4,315	27,716	-
Male	34,784	3,205	2,394	3,473	523	2,712	594	5,137	4,315	12,432	-
Female	20,703	483	601	1,041	1,398	587	-	1,308	-	15,283	-
Construction											
Total	507,021	4,955	16,745	20,475	5,982	7,941	-	275,429	20,314	155,181	-
Male	423,634	4,590	15,211	16,365	1,767	6,927	-	249,063	20,079	109,633	-
Female	83,387	366	1,534	4,110	4,215	1,015	-	26,367	234	45,547	-
Wholesale and ret	ail trade; rep	air of motor v	ehicles and mo	torcycles							
Total	1,465,772	14,065	6,918	12,463	5,824	1,177,480	10,233	122,013	12,090	104,686	-
Male	706,136	9,296	3,296	7,521	1,539	556,167	8,504	61,902	8,857	49,055	-
Female	759,636	4,769	3,622	4,942	4,285	621,313	1,729	60,111	3,234	55,632	-

Table 5.14 Cont'd

1 able 5.14 CO	ii u				Maio	or Occupation	nal Groups				
Major Industrial					,	-1	Skilled				
Divisions				Technicians			agricultural,	Craft and			
and Sex				and	Clerical	Service	forestry	related	Plant and machine		
5				associate	support	and sales	and fishery	trades	operators, and	Elementary	0.1
-	Total	Managers	Professionals	professionals	workers	workers	workers	workers	assemblers	occupations	Others
Transportation and	•										
Total	389,343	2,671	1,860	7,831	9,423	21,963	-	4,893	246,302	94,402	-
Male	368,724	1,763	1,249	5,873	5,571	18,388	-	4,331	241,554	89,995	-
Female	20,619	908	611	1,958	3,852	3,575	-	561	4,747	4,407	-
Accommodation ar											
Total	406,137	19,272	1,329	6,200	13,719	229,292	523	52,836	1,772	81,195	-
Male	115,886	11,664	1,226	1,748	5,593	63,776	-	13,698	1,674	16,509	-
Female	290,251	7,607	103	4,452	8,126	165,516	523	39,138	98	64,687	-
Information and co											
Total	73,483	331	17,537	15,088	17,096	7,009	313	9,023	3,278	3,809	-
Male	40,126	331	9,747	8,189	4,501	6,022	313	5,507	2,629	2,889	-
Female	33,357	-	7,790	6,899	12,595	987	-	3,517	650	920	-
Financial and insur											
Total	171,923	12,965	37,003	35,793	51,831	12,506	-	22	1,538	20,264	-
Male	108,035	11,124	23,431	23,654	26,039	9,566	-	22	1,538	12,661	-
Female	63,887	1,841	13,572	12,140	25,791	2,940	-	-	-	7,603	-
Real estate activitie	es										
Total	5,339	1,570	343	336	197	2,461	-	-	-	433	-
Male	2,909	1,570	-	336	-	914	-	-	-	90	-
Female	2,430	-	343	-	197	1,547	-	-	-	343	-
Professional, scien	tific and tec	hnical activiti	es								
Total	126,086	5,237	52,546	37,420	15,437	4,076	642	4,776	1,138	4,513	301
Male	85,419	3,963	39,037	24,593	7,744	2,843	-	3,597	800	2,541	301
Female	40,667	1,274	13,509	12,827	7,693	1,233	642	1,179	338	1,972	-
Administrative and	support ser	vice activities	3								
Total	146,604	8,484	21,527	27,295	23,847	28,965	790	3,012	1,951	30,447	287
Male	87,366	6,871	15,237	18,651	7,201	21,000	790	2,955	1,242	13,132	287
Female	59,238	1,612	6,290	8,644	16,646	7,965	-	57	709	17,315	-
Public administration	on and defe	nse; compuls	ory social secur	ity							
Total	349,400	32,938	78,716	75,267	38,696	72,648	935	2,722	10,839	29,632	7,006
Male	235,776	24,200	56,986	49,869	13,048	60,749	804	2,144	9,871	11,958	6,148
Female	113,623	8,738	21,731	25,398	25,649	11,899	131	578	967	17,674	858

Table 5.14 Cont'd

					Majo	or Occupation	nal Groups				
Major Industrial Divisions and Sex	Total	Managers	Professionals	Technicians and associate	Clerical support workers	Service and sales workers	Skilled agricultural, forestry and fishery workers	Craft and related trades workers	Plant and machine operators, and assemblers	Elementary occupations	Others
Education	rotai	Wanagoro	1 1010001011010	proroccionaio	WOINGIO	WOINGIO	WOTKOTO	WOINGIG	documento	оссиранопо	011010
Total	469,184	10,356	202,401	152,981	25,832	24,053	205	7,713	3,328	42,316	-
Male	254,406	8,390	129,892	82,271	5,028	10,810	-	5,404	2,575	10,036	-
Female	214,778	1,965	72,509	70,710	20,804	13,243	205	2,309	753	32,280	-
Human health and	social work	activities									
Total	189,014	5,196	56,019	71,519	11,901	14,603	-	319	3,110	26,348	-
Male	78,247	4,755	28,258	28,887	1,092	7,073	-	111	2,406	5,667	-
Female	110,766	441	27,761	42,632	10,810	7,530	-	207	704	20,681	-
Arts, entertainmen	t and recrea	tion									
Total	49,289	2,513	3,969	15,052	4,745	5,744		1,715	314	14,418	695
Male	28,878	2,513	2,876	10,492	566	3,513	125	1,715	314	6,417	348
Female	20,411	-	1,093	4,560	4,179	2,231	-	-	-	8,001	348
Other service activ	ities										
Total	440,377	4,965	18,586		7,317	135,772		46,238	9,302	188,890	96
Male	288,385	4,199	13,794	26,125	2,859	64,836		28,848	8,098	138,655	48
Female	151,992	765	4,792	2,164	4,458	70,936		17,390	1,204	50,235	48
Activities of housel		ployers; undi	fferentiated goo	ds- and service	s-producir	ng activities o	of				
households for own											
Total	606,525	-	341	801	2,526	201,106		59,713	19,935	316,655	-
Male	139,157	-	341	-	810	22,047	4,020	23,452	18,864	69,623	-
Female	467,368	-		801	1,716	179,059	1,428	36,261	1,072	247,032	-
Activities of extrate	_		-								
Total	78,337	6,206	17,727	17,851	7,641	12,087	-	787	5,150	10,888	-
Male	47,522	4,577	10,996	12,402	2,710	4,972		204	5,150	6,513	-
Female	30,815	1,629	6,731	5,449	4,931	7,115	-	583	-	4,375	

Table 5.15 Employed Population Aged Ten Years and Above by Status in Employment and Major Industrial Divisions and Sex-COUNTRY TOTAL: 2015

Γ		30011	IKI IOIAL.				Status in Er	mployment						
					Paid Emplo									
	Major Industrial			Employee-		Employee- NGO								
	Divisions	Total		Government	Employee-	(including		0.1	Member of	0.16	Unpaid			
	and Sex	Employed	Employee-	Development		International		Other	Co- Operatives	Self	Family Worker	Employer	Approptice	Othoro
L	Total Employe	Persons	Government	organization	organization	organization)	Domestic	Employee	Operatives	Employed	worker	Employer	Apprentice	Others
		•		404 004	4 202 540	70.404	275 200	72.000	E 4 E 4 7	0.070.040	E04 C04	E0 00E	0.740	F2 002
	Total	7,083,094	1,052,822	481,824	1,382,519	76,161	375,369	73,889		2,873,646		58,825		52,093
	Male	3,963,143	612,062	314,601	901,795	45,744	89,678	60,618	•	1,602,692		38,412		35,466
	Female	3,119,950	440,761	167,223	480,724	30,418	285,691	13,271	16,568	1,270,954	374,660	20,413	2,639	16,627
	Agriculture, for	-	isning											
	Total	567,067	-	46,948	59,696	3,088	-	1,281	3,071	,	164,485	2,895	116	3,018
	Male	382,749	-	30,896	34,908	1,830	-	834	1,864	214,409	93,728	2,242	63	1,976
	Female	184,319	-	16,051	24,788	1,258	-	447	1,207	68,061	70,757	653	53	1,043
	Mining and q													
	Total	35,165	-	4,166	11,431	133	-	392	2,852	15,264	507	419	-	-
	Male	28,221	-	3,119	9,518	133	-	392	2,811	11,322	507	419	-	-
17	Female	6,944	-	1,047	1,913	-	-	-	41	3,943	-	-	-	-
∞	Manufacturin	g												
	Total	929,847	-	52,713	281,165	-	-	7,047	6,814	487,039	72,388	10,101	4,215	8,364
	Male	489,346	-	34,430	187,958	-	-	5,468	4,815	217,631	23,715	6,739	4,215	4,373
	Female	440,501	-	18,283	93,207	-	-	1,578	1,999	269,408	48,673	3,362	-	3,991
	Electricity, gas	s, steam and	d air conditionin	ng supply										
	Total	21,693	13,100	4,076	2,782	-	-	-	-	1,212	523	-	-	-
	Male	17,435	11,135	3,327	1,823	-	-	-	-	1,150	-	-	-	-
	Female	4,258	1,965	749	959	-	-	-	-	62	523	-	-	-
	Water supply;	sewerage,	waste manage	ment and reme	diation activiti	es								
	Total	55,487	27,780	4,108	4,937	-	-	1,145	57	8,268	8,862	-	-	331
	Male	34,784	20,185	2,813	1,738	-	-	352	-	7,416	1,949	-	-	331
	Female	20,703	7,595	1,295	3,199	-	-	793	57	852	6,913	-	-	-
	Construction	,	,	,	,						, -			
	Total	507,021	_	44,987	222,689	-	-	18,281	16,666	178,285	5,515	4,520	486	15,591
	Male	423,634	-	34,082	180,798	-	-	16,505	11,712	159,304	4,565	3,975	486	12,206
	Female	83,387	_	10,905	41,891	-	-	1,776	4,954	18,981	951	544	-	
		,		,	,50.			.,	.,	,				-,-50

Table 5.15 Cont'd

Table 3.13	Cont a					Status in Er	mployment						
				Paid Emplo			1 -7						
Major Industrial			Employee-		Employee- NGO								
Divisions	Total		Government	Employee-	(including			Member of		Unpaid			
and Sex	Employed	Employee-	Development	Private	International		Other	Co-	Self	Family	l		0.1
	Persons	Government			organization)	Domestic	Employee	Operatives	Employed	Worker	Employer	Apprentice	Others
		e; repair of mo	tor vehicles and	-									
Total	1,465,772	-	8,803	148,392	-	-	5,000		1,151,370	,	15,842	1,764	5,008
Male	706,136	-	5,805	91,700	-	-	2,197	2,887	545,139	43,236	11,245	1,464	2,463
Female	759,636	-	2,997	56,692	-	-	2,803	4,436	606,231	79,035	4,596	300	2,545
•	on and storaç	ge											
Total	389,343	-	47,583	171,091	-	-	10,256	5,937	145,113	3,587	1,629	49	4,098
Male	368,724	-	37,706	166,029	-	-	9,100	5,937	142,252	2,727	825	49	4,098
Female	20,619	-	9,877	5,062	-	-	1,156	-	2,860	860	804	-	-
		l service activit											
Total	406,137	-	8,876	167,875	-	-	1,433	1,503	158,088	49,547	17,267	300	1,248
Male	115,886	-	2,616	55,442	-	-	590	463	36,182	10,889	9,016	300	388
Female	290,251	-	6,259	112,433	-	-	843	1,040	121,907	38,659	8,251	-	860
Information a	and communi	cation											
Total	73,483	34,797	7,611	17,281	-	-	246	-	11,432	1,645	331	-	141
Male	40,126	18,620	5,310	9,375	-	-	-	-	5,671	730	331	-	90
Female	33,357	16,177	2,300	7,907	-	-	246	-	5,761	915	-	-	51
Financial an	d insurance a	activities											
Total	171,923	-	111,319	46,284	-	-	-	376	13,848	95	-	-	-
Male	108,035	-	67,402	27,654	-	-	-	376	12,508	95	-	-	-
Female	63,887	-	43,917	18,630	-	-	-	-	1,340	-	-	-	-
Real estate	activities												
Total	5,339	-	685	870	-	-	-	-	2,213	-	1,570	-	-
Male	2,909	-	-	673	-	-	-	-	666	-	1,570	-	-
Female	2,430	-	685	197	-	-	-	-	1,547	-	-	-	-
Professional	l, scientific an	d technical act	tivities										
Total	126,086	-	93,041	17,245	-	-	-	120	15,021	587	73	-	-
Male	85,419	-	62,263	9,711	-	-	-	120	12,959	293	73	-	-
Female	40,667	-	30,778	7,534	-	-	-	-	2,061	293	-	-	-
	, -		, -	,					, =				

1/9

Table 5.15 Cont'd

							Status in Er	mployment						
					Paid Emplo									
	Major Industrial			Employee-		Employee- NGO								
	Divisions	Total		Government	Employee-	(including		0.1	Member of	0.16	Unpaid			
	and Sex	Employed Persons	Employee- Government	Development organization	Private	International organization)		Other Employee	Co- Operatives	Self Employed	Family	Employer	Apprentice	Others
Щ.	Administrative		t service activit		organization	organization	Domestic	Linployee	Орегануез	Linployed	WORKE	Linbioyo	трргонцос	Others
	Total	146,604	89,602	5,198	27,672	_	-	-	1,796	20,285	613	194	521	724
	Male	87,366	50,362	4,002	14,535	-	-	-	1,203	16,387	62	92	-	724
	Female	59,238	39,240	1,196	13,137	-	-	-	593	3,897	551	102	521	-
	Public admini	istration and	defense; comp	oulsory social s	ecurity									
	Total	349,400	348,465	935	-	-	-	-	-	-	-	-	-	-
	Male	235,776	234,972	804	-	-	-	-	-	-	-	-	-	-
	Female	113,623	113,492	131	-	-	-	-	-	-	-	-	-	-
	Education													
	Total	469,184	383,258	11,101	64,866	-	-	153	585	4,486	133	1,287	544	
	Male	254,406	214,393	4,276	31,047	-	-	-	306	3,047	-	633	97	608
	Female	214,778	168,866	6,825	33,819	-	-	153	279	1,439	133	655	447	2,164
100			work activities											
)	Total	189,014	135,188	5,515	31,436	-	-	96	421	13,289	1,553	1,094	422	-
	Male	78,247	51,180	1,949	13,696	-	-	-	392	10,568	45	418	-	-
	Female	110,766	84,008	3,566	17,740	-	-	96	29	2,721	1,508	676	422	-
	Arts, entertair													
	Total	49,289	20,633	1,089	11,325	-	-	-	336	12,557	1,498	123	476	1,252
	Male	28,878	11,216	187	7,226	-	-	-	336	8,415	83	123	41	1,252
	Female	20,411	9,417	902	4,099	-	-	-	-	4,142	1,415	-	436	-
	Other service													
	Total	440,377	-	22,210	80,902	-	-	28,559	6,689	279,389	11,731	1,481	760	8,656
	Male	288,385	-	12,776	52,451	-	-	25,180	4,756	180,603	4,587	711	362	6,958
	Female	151,992	-	9,434	28,451	-	-	3,379	1,933	98,786	7,143	770	398	1,699
			is employers; u	indifferentiated	•	•	•	s of nousend	lds for own use					
	Total	606,525	-	-	11,324	-	375,369	-	-		146,142	-	63	892
	Male .	139,157	-	-	3,574	-	89,678	-	-	16,094	,	-	-	-
	Female	467,368	- 	- 	7,750	-	285,691	-	-	56,640	116,332	-	63	892
			ı organizations	and bodies De	-	70.015				4 00 :				
	Total	78,337	-	859	3,256	72,940	-	-	-	1,281	-	-	-	-
	Male	47,522	-	835	1,939	43,781	-	-	-	967	-	-	-	-
	Female	30,815	-	24	1,317	29,160	-	-	-	314	-	-	-	

80

Table 5.16 Employed Population Aged Ten Years and Above by Status in Employment and Major Occupational Groups and Sex-COUNTRY TOTAL: 2015

	OUNTER I	01AL. 2013											
					S	tatus in Emp	oloyment						
	_			Paid Emplo							[1
					Employee-								
Major			Employee-		NGO								
Occupational	Total	Employee	Government	Employee-	(including	Employee	Other	Member of Co-	Self	Unpaid			1
Groups and Se	Employed Persons	Employee- Government	Development organization		International organization)		Employee			Family Worker	Employer	Apprentice	Others
Total Employed		Covernment	Organization	organization	organization)	DOMESTIC	Lilipioyee	Operatives	Lilipioyeu	VVOINGI	Lilibiosei	Applemine	Outers
Total	7,083,094	1,052,822	481,824	1,382,519	76,161	375,369	73,889	54 547	2,873,646	591 681	58,825	9 716	52,093
Male	3,963,143	612,062		901,795	45,744	89,678	60,618		1,602,692		38,412		35,466
Female	3,119,950	440,761	167,223	480,724	30,418	285,691	13,271	•	1,270,954		20,413		16,627
Managers	5,115,550	440,701	107,220	400,12 4	55,410	200,001	10,211	10,000	1,210,004	<i>51</i> 1 ,000	20,710	2,000	10,021
Total	153,505	58,501	27,924	21,377	5,047	_	1,815	607	291	2,010	35,047	_	888
Male	117,550	46,554		15,754	3,811	_	1,774	607	291	1,140	23,176	_	888
Female	35,956	11,947	4,369	5,623	1,235	-	40	-		870	11,871	_	-
Professionals		, 5	.,500	5,320	.,_00					0.0	, 5		
Total	558,734	329,543	98,052	83,128	17,480	-	4,465	120	20,182	1,177	2,215	422	1,951
Male	369,274	212,045		53,650	10,748	-	3,583	120	16,245	,	1,532	-	1,951
⇒ Female	189,460	117,498	•	29,478	6,731	-	882	-	3,937	223	683	422	-
Technicians a	,		•	•	•								
Total	564,911	292,646	87,266	86,080	17,925	-	12,600	3,272	55,230	3,089	1,966	826	4,012
Male	344,949	163,392	56,072	49,379	12,402	-	12,165	2,369	44,802	1,302	678	41	2,348
Female	219,962	129,254	31,193	36,701	5,523	-	434	903	10,428	1,787	1,289	785	1,664
Clerical suppo													
Total	258,416	100,197		58,130	7,243		430	1,472	19,347	1,508	255	521	92
Male	90,129	25,401	28,768	20,244	2,710	810	-	1,127	10,331	494	153	-	92
Female	168,287	74,796	37,928	37,886	4,533	1,716	430	344	9,016	1,014	102	521	-
Service and s													
Total	2,162,111	117,884		255,366	10,609	186,621	10,085	8,523	, ,		8,269	1,297	6,876
Male	912,282	89,828	,	127,175	4,149	19,076	6,740	3,737	569,291	56,498	3,791	600	4,469
Female	1,249,829	28,057	11,836	128,192	6,460	167,546	3,344	4,786	772,585	119,439	4,478	698	2,407
•		and fishery wo											
Total	387,801	-	14,313	12,967	1,221	4,968	-	2,091	253,889		2,819	116	812
Male	270,676	-	8,483	5,306	699	4,020	-	1,143	197,579		2,219	63	224
Female	117,126	-	5,830	7,661	522	949	-	948	56,310	43,666	600	53	588
Craft and rela			0.5.00	000.000		00.000	40.070	44.000	==0 000	07.00:		0.000	10.015
Total	1,089,180	19,767	35,320	298,696	787	39,636	12,250	14,686	573,880	•	4,417		16,018
Male	696,336	15,037	26,545	229,857	204	17,893	11,306	10,640	342,136	•	4,417	6,090	
Female	392,845	4,730	8,775	68,839	583	21,743	945	4,046	231,744	46,467	-	-	4,973

 $\overline{\alpha}$

Table 5.16 Cont'd

						S	tatus in Emp	oloyment						
					Paid Emplo	oyees								
(Major Occupational Groups and Sex	Total Employed Population	Employee- Government	Employee- Government Development organization	Employee- Private organization	Employee- NGO (including International organization)	Employee Domestic		Member of Co- Operatives	Self Employed	Unpaid Family Worker	Employer	Apprentice	Others
	Plant and mach	ine operator	s, and assemb	lers										
	Total	458,982	20,415	36,111	224,758	5,150	19,935	7,106	2,358	135,214	4,914	1,251	381	1,389
	Male	418,261	18,580	33,959	197,791	5,150	18,864	6,332	1,867	129,275	3,518	1,251	284	1,389
	Female	40,721	1,835	2,151	26,968	-	1,072	774	491	5,938	1,396	-	97	-
	Elementary occ	upations												
	Total	1,440,420	105,881	77,030	341,894	10,700	121,683	25,139	21,418	473,167	240,805	2,586	63	20,055
	Male	736,433	34,442	41,496	202,517	5,870	29,016	18,717	16,368	292,743	81,008	1,195	-	13,060
	Female	703,988	71,439	35,533	139,377	4,830	92,667	6,422	5,050	180,424	159,798	1,390	63	6,995
	Others													
	Total	9,032	7,989	349	123	-	-	-	-	572	-	-	-	-
182	Male	7,254	6,783	349	123	-	-	-	-	-	-	-	-	-
	Female	1,777	1,206	-	-	-	-	-	-	572	-	-	-	-

Table 5.17 Employed Population of Urban areas Aged Ten Years and Above by Number of Hours Worked During the Last Seven Days, Age Group and Sex – COUNTRY TOTAL: 2015

	, Age Glo	<u>up u o (</u>		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		f Hours Wo	rked Per M	/eek				Mean No.
Age group	I											of Hours
and Sex												Worked
	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Per week
All Ages												
Total	7,083,094	382,173	64,527	360,141	531,600	599,887	789,854	1,625,664	1,060,511	1,018,278	650,458	43.4
Male	3,963,143	197,014	23,811	139,776	229,385	287,592	434,689	967,233	640,362	652,941	390,340	45.9
Female	3,119,950	185,159	40,716	220,366	302,215	312,295	355,165	658,431	420,148	365,338	260,118	40.3
10-14												
Total	194,969	4,197	14,609	42,745	33,363	28,658	21,381	19,013	11,605	8,739	10,660	27.7
Male	100,707	3,279	5,067	25,168	15,871	16,301	13,288	9,726	5,862	3,972	2,173	25.6
Female	94,262	918	9,543	17,577	17,491	12,356	8,093	9,287	5,743	4,767	8,487	30.0
15-19												
Total	565,103	17,681	16,774	55,041	54,590	53,196	45,168	82,081	79,228	87,747	73,597	43.2
Male	224,956	6,316	7,117	23,532	25,806	22,310	16,467	31,953	32,714	36,461	22,279	41.4
Female	340,148	11,366	9,656	31,509	28,783	30,886	28,701	50,128	46,514	51,286	51,318	44.4
20-24												
Total	1,107,918	51,186	12,450	43,815	77,500	80,479	110,253	249,285	174,608	179,575	128,766	46.1
Male	546,853	17,596	5,544	17,422	36,131	35,499	48,829	118,071	95,784	99,013	72,966	48.7
Female	561,064	33,590	6,906	26,394	41,369	44,981	61,424	131,214	78,824	80,562	55,800	43.5
25-29												
Total	1,455,213	85,296	5,242	49,946	92,537	100,202	175,254	379,480	222,002	215,494	129,760	44.5
Male	827,711	37,159	1,269	17,492	36,465	49,840	96,307	221,713	135,259	148,053	84,152	47.6
Female	627,502	48,137	3,973	32,454	56,072	50,362	78,947	157,767	86,743	67,441	45,608	40.5
30-34												
Total	985,745	51,478	3,824	35,398	60,753	75,425	117,505	259,632	165,887	145,810	70,032	44.2
Male	599,578	30,335	1,059	12,708	25,881	40,520	68,420	166,042	101,822	104,546	48,244	46.6
Female	386,166	21,143	2,765	22,690	34,871	34,905	49,085	93,589	64,065	41,264	21,789	40.4
35-39												
Total	935,320	49,057	3,363	39,754	65,271	86,769	106,641	221,009	140,610	136,619	86,228	44.0
Male	518,080	26,650	487	12,044	22,890	36,321	55,671	131,054	83,905	92,453	56,605	47.7
Female	417,240	22,407	2,876	27,709	42,381	50,448	50,971	89,955	56,705	44,166	29,623	39.4
40-44												
Total	578,390	30,335	1,406	24,846	40,270	43,088	65,703	142,321	90,798	84,727	54,897	44.6
Male	365,839	20,851	766	7,716	20,281	23,927	42,008	99,909	60,687	55,361	34,334	45.8
Female	212,551	9,483	639	17,131	19,988	19,161	23,696	42,412	30,112	29,367	20,563	42.4

ŏ

Table 5.17 Cont'd

					Number o	f Hours Wo	orked per W	/eek				Mean No.
Age group and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	of Hours Worked Per week
45-49		•	•			•	•					
Total	421,698	25,317	1,048	17,986	34,386	42,184	51,707	107,109	56,847	56,075	29,039	41.9
Male	252,303	14,016	971	5,422	12,418	18,608	31,560	70,557	37,229	40,060	21,462	45.4
Female	169,394	11,301	77	12,565	21,967	23,576	20,148	36,552	19,618	16,015	7,577	36.6
50-54												
Total	288,013	15,158	1,161	13,388	23,320	31,365	40,724	61,448	45,236	32,823	23,390	42.1
Male	173,544	9,478	150	6,008	10,270	16,116	23,553	41,474	31,363	21,102	14,029	44.0
Female	114,470	5,681	1,011	7,379	13,050	15,250	17,170	19,974	13,873	11,721	9,362	39.4
55-59												
Total	210,275	16,291	1,560	11,747	15,778	23,041	23,167	43,762	29,294	28,781	16,856	41.4
Male	128,640	9,333	1,166	3,642	7,209	10,172	14,427	31,572	19,305	18,221	13,594	45.1
Female	81,635	6,958	394	8,105	8,568	12,868	8,740	12,191	9,989	10,560	3,262	35.7
60-64												
Total	145,598	16,587	2,670	10,022	11,047	12,501	14,809	27,628	18,702	19,873	11,759	38.9
Male	90,332	8,703	-	2,195	3,502	4,296	11,841	20,317	14,734	15,802	8,943	45.4
Female	55,266	7,884	2,670	7,827	7,545	8,206	2,968	7,312	3,967	4,071	2,817	28.3
65+												
Total	194,851	19,590	422	15,454	22,787	22,979	17,542	32,896	25,694	22,015	15,474	37.7
Male	134,599	13,298	214	6,426	12,660	13,682	12,319	24,845	21,698	17,896	11,560	40.8
Female	60,253	6,292	207	9,027	10,127	9,297	5,222	8,051	3,996	4,119	3,914	30.6

Table 5.18 Employed Population of Urban Areas Aged Ten Years and Above by Number of Hours Worked Per Week, Region and Sex - COUNTRY TOTAL: 2015

					Jumber of	Hours W	orked Per	Week				Mean no. of Hours
Region					Variabel of	l louis vv		VVCCK				Worked
and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Per week
COUNTRY TOTAL	•		•					•	•	•		•
Total	7,083,094	382,173	64,527	360,141	531,600	599,887	789,854	1,625,664	1,060,511	1,018,278	650,458	43.4
Male	3,963,143	197,014	23,811	139,776	229,385	287,592	434,689	967,233	640,362	652,941	390,340	45.9
Female	3,119,950	185,159	40,716	220,366	302,215	312,295	355,165	658,431	420,148	365,338	260,118	40.3
TIGRAY REGION												
Total	477,813	40,843	2,161	17,626	21,362	32,386	36,957	102,779	71,892	88,496	63,309	46.9
Male	253,254	21,392	1,081	6,760	8,476	15,351	18,513	54,708	41,323	53,110	32,540	48.3
Female	224,559	19,452	1,081	10,866	12,886	17,035	18,443	48,071	30,569	35,387	30,769	45.4
AFFAR REGION												
Total	131,439	12,667	1,251	8,225	9,083	13,711	17,422	23,507	21,139	14,652	9,781	39.2
Male	75,452	6,394	207	2,912	4,496	7,350	10,675	16,477	14,238	8,661	4,042	40.6
Female	55,987	6,273	1,044	5,313	4,587	6,362	6,747	7,030	6,901	5,991	5,739	37.2
AMHARA REGION												
Total	1,449,091	103,563	7,059	99,342	137,988	143,567	176,248	274,222	164,102	196,099	146,902	41.1
Male	795,183	52,340	2,509	32,861	54,185	66,109	105,643	168,779	98,121	125,900	88,734	44.3
Female	653,908	51,223	4,550	66,481	83,803	77,458	70,604	105,443	65,980	70,198	58,168	37.1
OROMIA REGION												
Total	2,102,417	115,414	20,128	116,758	162,824	196,648	182,396	484,143	309,446	313,207	201,455	43.4
Male	1,213,723	62,932	8,382	53,665	75,960	99,447	97,010	287,602	194,510	214,209	120,006	45.5
Female	888,694	52,482	11,746	63,093	86,864	97,201	85,386	196,541	114,936	98,997	81,449	40.4
SOMALI REGION												
Total	119,082	5,718	1,387	7,933	9,372	7,900	17,782	20,872	15,523	20,598	11,996	43.9
Male	67,114	2,411	367	2,265	3,878	3,852	11,262		10,102	12,246	7,007	47.2
Female	51,967	3,307	1,020	5,668	5,494	4,048	6,520	7,148	5,422	8,352	4,989	39.5
BENISHANGUL-GUI	MUZ REGIC	N										
Total	107,053	6,990	1,913	12,528	12,733	12,132	15,565	20,162	9,714	9,299	6,018	34.8
Male	58,730	4,060	305	6,511	5,195	5,429	8,949	12,440	6,764	6,134	2,942	37.0
Female	48,323	2,929	1,607	6,018	7,538	6,703	6,616	7,722	2,950	3,165	3,076	32.2
S.N.N.P REGION												
Total	1,096,887	44,834	26,199		106,228	- , -	139,977	237,573	124,279	•	104,052	41.8
Male	603,653	20,007	8,812	22,731	46,746	45,580	75,696	148,387	74,329	93,157	68,209	45.7
Female	493,233	24,827	17,387	46,554	59,482	51,907	64,281	89,187	49,950	53,815	35,843	37.1

25

Table 5.18 Cont'd

				N	lumber of	Hours Wo	orked Per	Week				Mean no. of Hours
Region and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Worked Per week
GAMBELLA REGION												
Total	61,809	2,754	1,482	6,776	8,098	7,822	7,557	8,880	5,543	5,568	7,330	37.9
Male	31,771	1,395	803	2,148	3,558	4,035	3,718	5,079	2,630	3,472	4,933	41.8
Female	30,038	1,360	679	4,628	4,540	3,786	3,838	3,801	2,913	2,096	2,397	33.7
HARARI REGION												
Total	59,109	2,648	229	1,793	3,507	5,972	8,865	14,496	9,314	8,655	3,629	43.2
Male	29,900	1,222	229	680	1,144	2,065	4,881	7,961	4,839	4,917	1,960	45.3
Female	29,210	1,426	-	1,112	2,362	3,907	3,984	6,535	4,475	3,739	1,669	41.0
ADDIS ABABA CITY A	OMINISTRA	TION										
Total	1,362,069	40,870	1,638	14,215	51,756	73,303	175,283	413,987	310,530	195,391	85,097	47.4
Male	772,367	22,275	1,023	7,122	22,147	34,220	92,045	236,958	182,841	119,924	53,813	48.7
Female	589,703	18,595	615	7,093	29,609	39,083	83,238	177,028	127,689	75,467	31,285	45.8
DIRE DAWA ADMINIST	TRATION											
Total	116,325	5,871	1,079	5,660	8,650	8,959	11,803	25,044	19,029	19,342	10,889	44.0
Male	61,997	2,585	92	2,121	3,601	4,154	6,296	15,118	10,665	11,211	6,154	46.5
Female	54,327	3,285	988	3,539	5,049	4,805	5,507	9,926	8,364	8,131	4,735	41.2

Table 5.19 Employed Population of Urban Areas Aged Ten Years and Above by Number of Hours Worked Per Week, Major Industrial Divisions and Sex, COUNTRY- TOTAL: 2015

Divisions and Ge	ж, ооон		17(2. 20									Mean No.
Major Industrial				N	lumber of	Hours W	orked Per	Week				of Hours
Division												Worked
and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Per Week
Total Employed Population												
								1,625,664	1,060,511	1,018,278	650,458	43.4
Male	3,963,143	197,014	23,811	139,776	229,385	287,592	434,689	967,233	640,362	652,941	390,340	45.9
Female	3,119,950	185,159	40,716	220,366	302,215	312,295	355,165	658,431	420,148	365,338	260,118	40.3
Agriculture, forestry and fishing												
Total	567,067	61,437	4,763	52,112	75,578	78,495	70,379	85,291	61,634	63,475	13,904	32.8
Male	382,749	45,986	2,812	33,476	51,090	47,769	49,010	56,961	41,212	43,579	10,853	32.9
Female	184,319	15,451	1,951	18,636	24,488	30,726	21,369	28,330	20,422	19,896	3,051	32.7
Mining and quarrying												
Total	35,165	9,601	-	315	392	2,381	4,911	7,477	4,848	3,833	1,406	34.6
Male	28,221	5,686	-	315	363	2,025	4,431	5,491	4,848	3,656	1,406	38.6
Female	6,944	3,915	-	-	29	356	481	1,986	-	177	-	18.2
Manufacturing												
Total	929,847	60,950	5,634	55,833	75,565	85,950	77,716	212,016	151,923	137,124	67,137	42.1
Male	489,346	26,071	532	12,899	25,894	29,148	35,672	128,448	91,356	99,501	39,827	47.3
Female	440,501	34,879	5,102	42,934	49,671	56,801	42,044	83,569	60,567	37,623	27,310	36.2
Electricity, gas, steam and air con	nditioning su	upply										
Total	21,693	133	-	-	555	979	3,132	8,713	3,265	3,494	1,422	48.4
Male	17,435	90	-	-	123	939	3,039	6,473	2,821	3,112	837	48.5
Female	4,258	42	-	-	432	40	93	2,240	444	382	585	48.1
Water supply; sewerage, waste m	nanagemen	t and rem	ediation	activities								
Total	55,487	4,195	1,044	6,358	3,243	3,351	7,546	14,155	8,537	5,523	1,537	36.7
Male	34,784	1,639	207	1,881	388	2,495	5,069	11,857	5,197	4,629	1,423	43.3
Female	20,703	2,555	836	4,477	2,855	856	2,477	2,298	3,340	894	114	25.7
Construction												
Total	507,021	38,176	1,374	21,470	25,822	38,784	52,612	153,697	81,225	71,990	21,872	41.9
Male	423,634	28,791	1,374	17,430	21,405	33,479	45,376	129,122	70,651	57,383	18,624	42.2
Female	83,387	9,386	-	4,041	4,416	5,305	7,235	24,575	10,574	14,607	3,248	40.5
Wholesale and retail trade; repair	of motor ve	ehicles an	d motorc	cycles								
Total	1,465,772	70,400	8,045	74,366	140,182	130,849	117,737	248,769	232,435	280,929	162,058	45.5
Male	706,136	24,535	4,459	26,462	44,919	50,397	51,876	125,515	126,076	161,294	90,604	49.4
Female					,	00,00.	0.,0.0	120,010	120,010	,	00,00.	

8

Table 5.19 Cont'd

Major Industrial				N	umber of	Hours Wo	orked Per	Week				Mean No. of Hours
Division	T		4.0	4.40	10.01	22.22	0.4.00	40.40	40.55	50.74		Worked
and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Per Week
Transportation and storage												
Total	389,343	17,630	901	9,691	12,840	16,098	21,346	65,994	70,198	97,725	76,920	54.5
Male	368,724	15,014	901	9,251	11,867	15,024	19,577	60,630	67,053	94,843	74,565	55.0
Female	20,619	2,616	-	440	973	1,074	1,769	5,364	3,146	2,883	2,355	45.4
Accommodation and food service												
Total	406,137	16,856	315	13,364	16,421	33,668	29,484	58,855	64,924		100,712	53.9
Male	115,886	2,747	315	1,885	1,829	7,946	6,755	12,447	19,907	21,853	40,200	60.9
Female	290,251	14,109	-	11,479	14,592	25,722	22,728	46,408	45,017	49,684	60,512	51.0
Information and communication												
Total	73,483	2,739	53	1,266	1,050	966	10,017	32,417	12,634	7,496	4,846	47.1
Male	40,126	240	-	301	602	179	4,991	18,139	6,986	4,891	3,798	51.4
Female	33,357	2,499	53	965	448	786	5,027	14,278	5,647	2,606	1,049	41.9
Financial and insurance activities	;											
Total	171,923	7,032	-	741	2,364	7,906	29,344	74,607	28,872	14,949	6,110	44.5
Male	108,035	5,680	-	576	1,463	4,789	16,770	44,749	17,536	11,137	5,335	45.1
Female	63,887	1,352	-	164	901	3,117	12,574	29,858	11,335	3,812	776	43.5
Real estate activities												
Total	5,339	343	_	_	_	593	343	1,773	954	718	615	47.2
Male	2,909	_	_	_	_	_	_	757	954	583	615	59.6
Female	2,430	343	_	_	_	593	343	1,016	_	135	_	32.5
Professional, scientific and techn	ical activities	8						·				
Total	126,086	2,824	_	1,533	2,119	3,735	30,061	53,911	20,449	7,953	3,501	43.3
Male	85,419	1,261	_	1,419	966	2,518	19,210	36,209	13,950	6,830	3,055	44.4
Female	40,667	1,563	_	113	1,153	1,217	10,851	17,702	6,498	1,123	446	41.0
Administrative and support service		.,			.,	-,	,	,	0,100	.,		
Total	146.604	3,520	801	354	4,184	5.872	25.337	55,914	23,677	17.410	9,535	46.8
Male	87,366	2,316	-	-	1,587	2.994	14.135	28,349	15,966	13.369	8,651	50.3
Female	59,238	1,204	801	354	2,597	2,877	11,201	27,566	7,711	4,042	885	41.7
Public administration and defense				00 1	2,001	2,011	,	27,000	.,	1,0 12	000	
Total	349,400	16,221	313	2,917	7,454	11,736	75,938	131,977	56,064	26,623	20,157	44.5
Male	235,776	10,046	-	361	3,076	7,294	43,633	85,042	42,292	25,257	18,774	47.4
Female	113,623	6,174	313	2,556	4,378	4,441	32,305	46,935	13,773	1,366	1,383	38.4
i Gillale	113,023	0,174	313	2,550	4,570	, 1	32,303	40,933	13,773	1,500	1,505	50.4

Table 5.19 Cont'd

Major Industrial				N	lumber of	Hours Wo	orked Per	Week				Mean No. of Hours
Division												Worked
and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Per Week
Education												
Total	469,184	13,447	176	5,451	52,636	56,810	103,605	161,869	44,098	20,868	10,225	37.8
Male	254,406	4,423	176	2,852	26,254	30,482	54,439	93,313	23,724	12,181	6,562	39.0
Female	214,778	9,024	-	2,599	26,382	26,328	49,167	68,556	20,374	8,686	3,663	36.5
Human health and social work ac	tivities											
Total	189,014	7,770	927	3,096	2,142	7,376	29,001	67,288	39,686	21,053	10,676	46.0
Male	78,247	1,937	865	964	752	2,847	11,456	23,718	16,249	11,864	7,595	49.7
Female	110,766	5,832	62	2,131	1,390	4,529	17,545	43,570	23,437	9,189	3,081	43.4
Arts, entertainment and recreation	n											
Total	49,289	2,293	-	850	4,887	2,683	4,549	13,310	10,686	4,411	5,619	45.9
Male	28,878	1,507	-	504	2,492	1,882	2,525	8,154	5,361	3,504	2,949	45.5
Female	20,411	786	-	346	2,396	801	2,024	5,156	5,324	907	2,670	46.4
Other service activities												
Total	440,377	27,716	5,660	23,221	38,841	45,144	44,170	78,516	61,539	71,623	43,948	42.9
Male	288,385	12,169	3,156	12,520	22,426	28,040	28,753	54,371	43,663	49,984	33,303	45.7
Female	151,992	15,547	2,504	10,701	16,415	17,104	15,417	24,145	17,876	21,639	10,645	37.7
Activities of households as emplo	•		•	and								
services-producing activities of he	ouseholds fo	or own use	Э									
Total	606,525	15,043	33,551	86,467	60,536	60,672	42,742	70,151	71,161	82,201	84,001	40.2
Male	139,157	5,097	9,013	16,318	10,429	13,348	11,464	21,646	16,528	17,576	17,738	40.1
Female	467,368		24,538	70,149	50,107	47,325	31,278	48,505	54,633	64,625	66,263	40.2
Activities of extraterritorial organized	zations and	bodies De	escription	1								
Total	78,337	3,852	971	738	4,789	5,838	9,885	28,963	11,701	7,344	4,256	42.7
Male	47,522	1,778	-	362	1,462	3,994	6,509	15,842	8,032	5,916	3,626	46.3
Female	30,815	2,074	971	376	3,327	1,844	3,376	13,121	3,669	1,427	630	37.2

Table 5.20 Employed Population Aged Ten Years and Above by Number of Hours Worked Per Week, Major Occupational Groups and Sex - COUNTRY TOTAL: 2015

				١	Number of	Hours Wo	orked Per	Week				Mean No. of Hours
	Total											Worked
	Employed											Per Week
and Sex	Persons	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	
Total Employed Person												
	7,083,094	,	,	,	,	,	,		, ,	1,018,278	,	43.4
	3,963,143	•				•	•	967,233	640,362	652,941		45.9
Female	3,119,950	185,159	40,716	220,366	302,215	312,295	355,165	658,431	420,148	365,338	260,118	40.3
Managers												
Total	153,505	4,778	41	1,161	3,472	5,176	21,323	51,989	33,789	16,232	15,544	48.5
Male	117,550	3,511	41	1,161	2,752	3,131	15,508	42,367	25,116	12,839	11,123	48.3
Female	35,956	1,267	-	-	720	2,045	5,815	9,622	8,673	3,393	4,420	49.3
Professionals												
Total	558,734	16,186	1,000	3,408	31,779	36,512	115,057	226,616	83,856	34,733	9,587	41.3
Male	369,274	8,782	1,000	1,953	17,282	26,093	75,180	147,109	55,142	29,112	7,621	42.3
Female	189,460	7,404	-	1,455	14,498	10,419	39,877	79,507	28,714	5,621	1,966	39.2
Technicians and associate pr	rofessionals	3										
Total	564,911	28,024	383	8,604	33,717	40,293	106,487	201,835	87,590	38,676	19,303	41.3
Male	344,949	13,445	383	6,232	18,732	22,960	65,670	119,191	55,140	28,286	14,910	42.6
Female	219,962	14,579	-	2,372	14,984	17,333	40,817	82,644	32,450	10,390	4,393	39.2
Clerical support workers												
Total	258,416	11,096	722	1,327	3,173	6,279	53,467	114,819	37,761	17,850	11,922	44.2
Male	90,129	2,686	-	353	788	2,555	15,143	41,070	14,836	7,183	5,515	46.6
Female	168,287	8,410	722	974	2,384	3,724	38,324	73,749	22,925	10,668	6,407	43.0
Service and sales workers												
Total	2,162,111	105,267	11,305	95,220	170,122	176,440	177,764	354,158	323,590	407,102	341,141	48.2
Male	912,282	30,448	4,280	23,966	45,464	58,065	70,086	161,334	153,957	201,421	163,261	52.4
Female	1.249.829	74.819	7,026	71,254	124.658	118,375	107.679	192,824	169,633	•	177,880	45.1
Skilled agricultural, forestry a	nd fishery v	vorkers	,	,	,	,	•	,	•	,	,	
Total	387,801	53,416	3,913	37,953	61,662	56,999	44,284	46,915	36,334	39,020	7,304	29.9
Male	270,676	41,079	1,819	23,845	43,080	35,641	33,272	32,342	25,052	29,322	5,222	30.1
Female	117,126	12,337	2,094	14,108	18,582	21,358	11.012	14.572	11,283	9.698	2,082	29.5

Table 5.20 Cont'd

				١	lumber of	Hours W	orked Per	Week				Mean No. of Hours
Major Occupational Groups and Sex	Total Employed Persons	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Worked Per Week
Craft and related trades wor		<u> </u>	1-3	4-12	13-21	22-30	31-39	40-40	49-31	30-74	75+	
Total	1,089,180	59,228	5,705	60,918	80,597	108,859	109,328	259,651	188,662	162,116	54,117	41.8
Male	696,336	37,217	1,325	27,406	37,738	54,389	68,727	183,897	133,849	120,636	31,152	44.0
Female	392,845	22,011	4,379	33,512	42,859	54,470	40,601	75,755	54,813	41,480	22,965	37.8
Plant and machine operator	s, and assem	nblers										
Total	458,982	24,287	227	4,945	8,214	12,861	26,051	104,079	88,844	113,258	76,217	54.1
Male	418,261	22,509	227	4,164	6,828	10,940	23,581	90,965	79,224	105,938	73,886	54.7
Female	40,721	1,778	-	780	1,386	1,921	2,469	13,115	9,620	7,320	2,331	48.0
Elementary occupations												
Total	1,440,420	79,490	41,232	146,606	138,864	156,421	135,066	261,337	177,471	189,205	114,730	38.8
Male	736,433	36,935	14,737	50,696	56,721	73,818	66,496	146,038	95,817	118,118	77,057	43.2
Female	703,988	42,554	26,495	95,910	82,144	82,603	68,570	115,299	81,653	71,087	37,673	34.2
Others												
Total	9,032	401	-	-	-	48	1,026	4,265	2,613	86	593	47.2
Male	7,254	401	-	-	-	-	1,026	2,919	2,229	86	593	47.8
Female	1,777	-	-	-	-	48	-	1,345	384	-	-	44.9

Table 5.21 Employed Population of Urban Areas Aged Ten Years and Above by Number of Hours Worked Per Week, Status in Employment and Sex - COUNTRY TOTAL: 2015

Employment an	T SEX - CC	ONTINI	IOIAL		lumbar of	Houre W	orked Per	Mook				
					NUTTIDET OF	Tiouis W	JINEU FEI	VVCCN				Mean No. of Hours
_Status in												Worked Per
Employment	T		4.0	4.40	40.04	00.00	04.00	40.40	40.57	50.74	75	Week
and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	
Total Employed Population	7 000 004	000 470	0.4.505	000 444	=04.000		700 054	4 00= 004	4 000 544	4 0 4 0 0 7 0	050 450	40.4
Total								1,625,664		1,018,278		43.4
Male	3,963,143							967,233	640,362		390,340	45.9
Female	3,119,950	185,159	40,716	220,366	302,215	312,295	355,165	658,431	420,148	365,338	260,118	40.3
Employee-Government												
Total	1,052,822	40,128	1,231	8,991	61,694	70,432	217,864	387,794	147,919	73,159	43,611	42.1
Male	612,062	17,800	865	2,700	28,443	39,761	118,012	221,875	88,118	57,587	36,901	44.6
Female	440,761	22,328	365	6,292	33,251	30,671	99,852	165,919	59,801	15,571	6,710	38.6
Employee-Government Develop	oment organ	ization										
Total	481,824	18,151	315	2,968	8,313	15,251	90,373	192,566	88,901	42,362	22,624	45.1
Male	314,601	9,458	315	2,187	3,783	9,396	52,028	125,104	61,586	31,828	18,914	46.8
Female	167,223	8,693	-	781	4,530	5,855	38,345	67,462	27,314	10,533	3,710	41.9
Employee-Private organization												
Total	1,382,519	53,256	2,518	22,523	37,878	75,205	115,508	381,465	285,092	235,589	173,485	50.4
Male	901,795	32,918	2,518	15,182	27,929	40,159	68,403	231,355	180,028	174,435	128,868	51.8
Female	480,724	20,338	-	7,340	9,949	35,047	47,105	150,111	105,064	61,154	44,617	47.9
Employee-NGO (including Inter	national org	anization)										
Total	76,161	3,852	971	376	4,808	5,576	8,960	28,372	11,381	7,525	4,340	43.1
Male	45,744	1,778	_	_	1,071	3,732	5,517	15,016	8,507	6,412	3,710	47.8
Female	30,418	2,074	971	376	3,737	1,844	3,443	13,356	2,874	1,113	630	36.1
Employee Domestic	,	, -			-, -	, -	-, -	-,	,-	, -		
Total	375,369	12,176	1,435	10,623	16,615	27,270	31.138	57,884	62,445	76,815	78,969	53.3
Male	89,678	4,835	-	2,075	2,314	6,865	9,638	19,127	13,674	15,882	15,268	50.3
Female	285,691	7,341	1,435	8,547	14,302	20,405	21,499	38,757	48,771	60,933		54.2
Other Employee	200,001	7,011	1,100	0,011	,002	20, 100	21,100	00,707	10,111	00,000	00,701	02
Total	73,889	3.161	383	3,174	8,786	8,094	4.978	16,219	11,721	11,256	6,116	43.3
Male	60,618	2,580	383	2,982	3,830	7,277	4,548	12,944	10,581	9,513		45.4
Female	13,271	581	-	193	4,957	817	430	3,275	1,141	1,742	135	34.0
Member of Co-Operatives	10,271	301		133	7,337	017	730	3,273	1,171	1,7 42	100	34.0
Total	54,547	2,167	391	2,728	3,521	4,873	3,235	15,480	6,145	11,765	4,242	45.2
Male			391									45.2 45.5
	37,978	1,574		1,003	2,258	3,147	2,375	13,286	4,655	5,882	3,408	45.5 44.6
Female	16,568	593	-	1,725	1,263	1,726	860	2,194	1,490	5,883	834	44.6

Table 5.21 Cont'd

												Mean No.
Status in					Number	of Hours \	Norked Per	r Week				of Hours
Employment			4.0	4.40	40.04	00.00	04.00	40.40	40.57	50.74	7.5	Worked
and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Per Week
Self Employed												
Total	2,873,646	224,501	19,428	180,261	274,377	306,471	254,508	459,927	392,967	489,551	271,655	41.9
Male	1,602,692	115,599	6,506	69,577	122,655	144,465	144,455	279,939	247,332	315,197	156,967	44.6
Female	1,270,954	108,902	12,922	110,684	151,723	162,006	110,053	179,988	145,634	174,355	114,687	38.4
Unpaid Family Worker												
Total	591,681	17,766	37,266	121,519	107,867	80,428	51,210	55,594	39,548	53,797	26,687	28.9
Male	217,021	6,157	12,243	39,829	33,065	29,148	21,865	25,613	15,400	25,393	8,309	31.1
Female	374,660	11,610	25,023	81,690	74,802	51,280	29,345	29,980	24,148	28,404	18,378	27.6
Employer												
Total	58,825	3,200	-	1,536	2,506	1,025	2,502	13,811	9,149	11,734	13,361	54.1
Male	38,412	2,472	-	1,014	1,381	212	1,338	9,611	6,007	7,434	8,943	53.9
Female	20,413	728	-	523	1,124	813	1,164	4,200	3,143	4,300	4,418	54.5
Apprentice												
Total	9,716	523	-	846	1,529	75	907	3,179	898	708	1,050	40.2
Male	7,077	419	-	749	794	75	752	1,886	898	453	1,050	43.1
Female	2,639	104	-	97	736	-	154	1,293	-	255	-	32.4
Others												
Total	52,093	3,292	590	4,597	3,705	5,186	8,671	13,372	4,345	4,018	4,317	38.0
Male	35,466	1,424	590	2,479	1,864	3,354	5,757	11,478	3,576	2,924	2,019	39.2
Female	16,627	1,868	-	2,118	1,841	1,832	2,915	1,894	769	1,094	2,298	35.4

Table 5.22 Paid Employees of Urban Population Aged Ten Years and Above by Region, Types of Payment and Sex: 2015

Region & Types of Payment	Total	Paid Employ	/ees	Employe	ee-Govern	ment	Governi	Employee- ment Devel organizatior			mployee- e organiza	tion	1 2	ee-NGO (onal orga		Emplo	oyee Don	nestic		Other Pa Employe	-
i ayınıcını	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female
COUNTRY	TOTAL																				
Total	3,442,585	2,024,497			612,062	440,761	481,824	314,601		1,382,519	901,795	480,724	76,161	45,744	30,418	375,369	89,678		73,889	60,618	- ,
In cash	3,339,437	, ,	1,370,871	1,035,161	600,365	434,796	470,080	306,319	163,761	1,334,891	874,611	460,280	74,966	44,649	30,317	353,708	83,978	269,730	70,631	58,644	11,987
In kind	26,349	12,248	14,101	2,885	1,477	1,408	2,032	571	1,461	12,541	5,228	7,313	403	403	-	6,614	3,979	2,635	1,874	590	1,284
Both	76,799	43,683	33,116	14,776	10,220	4,557	9,712	7,711	2,001	35,087	21,956	13,131	793	691	101	15,047	1,721	13,326	1,384	1,384	-
TIGRAY R																					
Total	235,334	126,241	109,093	80,184	46,044	34,140	26,477	18,025	8,452	89,058	48,857	40,201	5,903	3,778	2,124	30,096	6,642	23,454	3,617	2,895	722
In cash	227,518	123,084	104,434	78,876	45,428	33,448	25,966	17,514	8,452	84,604	47,442	37,162	5,903	3,778	2,124	28,897	6,027	22,870	3,274	2,895	379
In kind	3,460	1,093	2,367	-	-	-	-	-	-	2,364	478	1,886	-	-	-	752	615	137	343	-	343
Both	4,356	2,064	2,293	1,308	616	692	511	511	-	2,090	936	1,154	-	-	-	447	-	447	-	-	-
AFFAR RE																					
Total	64,297	42,510	21,786	15,876	10,415	5,460	21,743	14,452	7,292	21,252	14,095	7,157	1,098	543	555	1,600	278	1,322	2,727	2,727	-
In cash	62,373	41,155	21,218	14,985	9,968	5,018	20,726	13,544	7,182	21,252	14,095	7,157	1,098	543	555	1,584	278	1,307	2,727	2,727	-
In kind	16	-	16	-	-	-	-	-	-	-	-	-	-	-	-	16	-	16	-	-	-
Both	1,908	1,355	552	890	448	443	1,017	908	110	-	-	-	-	-	-	-	-	-	-	-	-
7 AMHARA F																					
Total 4	628,384	378,890	249,494	235,757	147,677	88,080	82,165	56,118	26,047	214,524	138,603	75,921	8,937	5,577	3,360		17,158		17,628	13,757	3,871
In cash	614,268	369,693	244,576	233,083	145,453	87,630	82,165	56,118	26,047	208,985	135,291	73,694	8,172	4,914	3,258	65,507	14,841	50,666	16,356	13,075	3,281
In kind	5,613	3,726	1,886	446	352	94	-	-	-	1,602	756	846	67	67	-	2,317	1,961	356	1,181	590	590
Both	8,503	5,471	3,032	2,228	1,871	356	-	-	-	3,937	2,556	1,381	697	596	101	1,550	356	1,194	91	91	-
OROMIA R																					
Total	944,915	589,684	355,231	261,869	156,565	105,304	126,786	87,530	39,256	409,567	281,621	127,945	19,026	10,449	8,577		26,906	,	,	26,612	3,044
In cash	904,683	567,639	337,044	252,651	149,986	102,665	121,035	83,085	37,950	393,230	272,787	120,443	19,026	10,449	8,577	90,004	25,639	64,365	28,738	25,694	3,044
In kind	4,698	1,750	2,948	237	237	-	93	-	93	2,452	651	1,801	-	-	-	1,917	862	1,055	-	-	-
Both	35,533	20,295	15,239	8,982	6,342	2,639	5,658	4,446	1,213	13,884	8,183	5,702	-	-	-	6,091	406	5,685	918	918	-
SOMALI R																					
Total	42,140	28,855	13,284	16,655	11,378	5,277	7,621	6,312	1,308	10,288	7,783	2,504	2,190	1,783	408	5,141	1,394	3,747	245	205	40
In cash	40,962	27,678	13,284	16,605	11,328	5,277	7,330	6,021	1,308	9,451	6,947	2,504	2,190	1,783	408	5,141	1,394	3,747	245	205	40
In kind	133	133	-	-	-	-	133	133	-	-	-	-	-	-	-	-	-	-	-	-	-
Both	1,044	1,044	-	50	50	-	158	158	-	836	836	-	-	-	-	-	-	-	-	-	-

Table 5.22 Cont'd

Region &Types of	Total	Paid Empl	oyees	Employ	ee-Goverr	nment		mployee overnme nent orga	nt		Employee ate organiz		(includ	ployee-N ing Intern ganizatio	ational	Emplo	yee Dom	nestic		ther Pa Imploye	-
Payment	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female
	NGUL-GU	MUZ																			
REGION																					
Total	45,856	29,216	16,640	24,301	14,603	9,698	6,943	4,815	2,128	11,478	8,662	2,816	703	495	209	2,203	442	1,761	228	199	29
In cash	42,789	27,485	15,304	23,972	14,367	9,605	6,273	4,222	2,051	9,542	7,760	1,781	703	495	209	2,071	442	1,629	228	199	29
In kind	245	77	169	92	-	92	-	-	-	153	77	77	-	-	-	-	-	-	-	-	-
Both	2,821	1,654	1,167	236	236	-	670	593	77	1,783	825	958	-	-	-	132	-	132	-	-	-
S.N.N.P F																					
Total	475,132	285,604	189,528	213,944		92,492		41,555	19,993	143,598	100,061		12,166	8,248	3,918	38,105	9,909	28,196	5,770		1,392
In cash	459,312	276,651	182,661	212,896	120,796	92,100	,	41,460	19,296	133,288	92,580	40,708	12,071	8,153	3,918	34,532	9,285	25,247	5,770	4,378	1,392
In kind	1,589	495	1,094	301		301	191	96	96	400	400	-	-	-	-	697	-	697	-	-	-
Both	14,230	8,457	5,773	747	656	91	601	-	601	9,910	7,081	2,829	95	95	-	2,877	625	2,252	-	-	-
	LA REGIO		0 = 10	40.754		. =									.=0						
Total	25,235	15,667	9,568	10,754	6,241	4,513	5,666	3,511	2,155	4,652	3,557	1,095	1,990	1,512	478	1,848	636	1,211	325	209	115
In cash	24,436	15,211	9,225	10,590	6,077	4,513	5,532	3,430	2,102	4,395	3,387	1,008	1,990	1,512	478	1,645	636	1,009	285	169	115
In kind	257	164	93	164	164	-	53	-	53	-	-	-	-	-	-	40	-	40	-	-	-
Both	541	292	249	-	-	-	82	82	-	257	171	86	-	-	-	162	-	162	40	40	-
HARARI I				4- 444	0.405							0 (70									
Total	30,567	16,901	13,666	15,461	8,635	6,826	1,836	988	848	9,992	6,319	3,673	514	343	171	2,454	355	2,100	311	262	49
In cash	30,429	16,763	13,666	15,411	8,585	6,826	1,836	988	848	9,951	6,278	3,673	514	343	171	2,454	355	2,100	263	214	49
In kind	91	91	-	50	50	-	-	-	-	41	41	-	-	-	-	-	-	-	-	-	-
Both	48	48	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	48	48	-
	BABA CITY			440.005	00 / 17	00 (50	40/4//	77.000	E0 0E0	100 171	070 000	4 / 5 00 /	00 000	40 745	40.005	447.700	04.07/	00.454	40.450	0.404	0 (74
Total		477,298		163,305	80,647	82,658	136,166	,	,	439,174	-,	165,836	- /	12,745	.,	117,730	.,	93,454	12,158		3,674
In cash	874,360	469,577		161,374	79,973	81,401		76,452		431,256	269,145	162,110		12,409	10,335	113,876	- /		11,521		3,325
In kind	10,246	4,718	5,528	1,595	674	921	1,562	342	1,220	5,529	2,825	2,704	336	336	-	874	541	333	350	-	350
Both	7,007	3,003	4,003	336	-	336	1,014	1,014	-	2,390	1,368	1,022	-	-	-	2,980	334	2,646	287	287	-
	WA ADMIN								4 000		40.000	40.000						= 407			
Total	59,114	33,631	25,483	14,718	8,404	6,314	4,873	3,486	1,388	28,936	18,898	10,038	555	271	284	8,807	1,681	7,126	1,224	890	333
In cash	58,305	33,631	24,675	14,718	8,404	6,314	4,873	3,486	1,388	28,936	18,898	10,038	555	271	284	7,999	1,681	6,318	1,224	890	333
In kind	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Both	808	-	808	-	-	-	-	-	-	-	-	-	-	-	-	808	-	808	-	-	

Table 5.23 Paid Employees of Urban Population Aged Ten Years and Above by Period of Payment and Sex- COUNTRY TOTAL: 2015

	,	OX COCITIII	Y TOTAL: 201				1
				aid Employees			
			Employee		Employee-NGO		
Period of	Total		Government	Employee-	(including		C
Payment	Paid	Employee-	Development	Private	International	Employee	Other Paid
and Sex	Employees	Government	organization	organization	organization)	Domestic	Employee
Total Paid Employ			,				
Total	3,442,585	1,052,822	481,824	1,382,519	76,161	375,369	73,889
Male	2,024,497	612,062	314,601	901,795	45,744	89,678	60,618
Female	1,418,088	440,761	167,223	480,724	30,418	285,691	13,271
Hourly							
Total	14,095	-	90	11,415	342	1,663	584
Male	10,850	-	90	8,617	342	1,317	483
Female	3,244	-	-	2,798	-	346	101
Daily							
Total	169,627	1,180	3,202	119,484	470	25,059	20,232
Male	139,448	344	2,347	101,443	470	16,995	17,850
Female	30,179	836	855	18,041	-	8,065	2,382
Weekly	•			•		•	
Total	114,366	358	1,694	98,779	-	7,515	6,020
Male	90,222	358	394	79,438	-	5,033	4,998
Female	24,145	-	1,300	19,341	-	2,481	1,022
Half Month	,		,	•		•	•
Total	79,947	4,543	6,430	60,956	1,086	3,890	3,041
Male	57,576	4,240	4,438	43,276	605	2,189	2,827
Female	22,371	303	1,992	17,680	481	1,701	214
Monthly	,		,	,	_	, -	
Total	3,025,631	1,044,239	467,011	1,070,089	73,885	332,047	38,360
Male	1,699,708	605,919	304,558	652,437	44,206	61,651	30,936
Female	1,325,923	438,320	162,453	417,652	29,678	270,396	7,424
Yearly	, -,-	,	- ,	,- 3—	-,	-,	,
Total	12,540	2,142	1,426	3,961	-	4,420	590
Male	7,490	1,200	993	2,969	_	2,328	-
Female	5,049	942	433	992	-	2,092	590
Other/specify	3,310	3.2	100	302		_,502	300
Total	26,093	360	1,683	17,834	379	775	5,062
Male	18,916	-	1,494	13,614	121	164	3,524
Female	7,176	360	189	4,220	259	611	1,537
Not Stated	7,170	300	103	7,220	239	011	1,007
Total	287	-	287	-	-	-	-
Male	287	-	287	-	-	-	-
Female	-	-	-	-	-	-	-

Table 5.24 Paid Employees of Urban Population Aged Ten Years and Over by Period of Payment, Sex and Amount of Total Payment Per Month – COUNTRY TOTAL: 2015

	Per Mon	th – COL	JNTRY T	OTAL: 2	2015												
						Amount	of Total P	ayment P	er Month	(In Birr)							Mean
Period of																	Amount
Payment																	of Total
and Sex																	Payment
and Sex			500-	1000-	1500-	2000-	2500-	3000-	3500-	4000-	4500-	5000-	5500-	6000-	6500-		Per
	Total	<500	999	1499	1999	2499	2999	3499	3999	4499	4999	5499	5999	6499	6999	7000+	Month
Total Paid Em	ployees																
Total	3,442,585	436,903	762,048	431,524	441,671	365,150	248,661	248,293	140,194	101,199	58,082	64,901	23,044	29,352	8,595	82,967	1,981.9
Male	2,024,497												20,890				2,354.1
Female	1,418,088										11,695		2,155	6,283		11,881	1,450.6
Hourly		,	,	,	,	,	,	,	·	,	·	,	,	,		·	·
Total	14,095	4,364	1,555	2,705	1,682	958	235	957	89	-	350	287	-	-	-	912	2,048.6
Male	10,850	2,532	1,136	2,095	1,682	958	235	573	89	-	350	287	-	-	-	912	2,410.8
Female	3,244	1,832	419	610	-	-	-	384	-	-	-	-	-	-	-	-	837.3
Daily																	
Total	169,627	40,109	46,427	19,072	25,802	13,285	4,710	8,823	1,526	2,283	2,859	374	144	3,176	-	1,037	1,382.6
Male	139,448	20,764	37,691	18,633	25,163	12,758	4,710	8,421	1,526	2,192	2,859	374	144	3,176	-	1,037	1,575.3
Female	30,179	19,346	8,735	440	640	527	-	402	-	90	-	-	-	-	-	-	491.9
Weekly																	
Total	114,366	23,299	29,991	25,865	14,123	9,344	2,885	2,946	2,108	1,744	798	341	109	764	-	49	1,272.1
Male	90,222	13,607	21,571	22,930	12,884	8,585	2,393	2,337	2,108	1,744	798	341	109	764	-	49	1,402.2
Female	24,145	9,692	8,419	2,935	1,239	759	493	608	-	-	-	-	-	-	-	-	786.3
Half Month																	
Total	79,947	12,826	26,442	12,841	10,375	8,118	384	3,533	226	1,589	1,017	535	-	435	630	994	1,422.6
Male	57,576	7,517	14,341	10,933	8,559	7,786	384	3,533	226	1,246	686	535	-	435	520	874	1,596.7
Female	22,371	5,309	12,101	1,908	1,816	332	-	-	-	343	332	-	-	-	110	121	974.7
Monthly																	
Total	3,025,631									,			22,791				2,062.6
Male	1,699,708					210,420			98,255				20,636				2,503.5
Female	1,325,923	261,125	376,661	159,032	156,405	118,863	77,002	72,166	36,704	23,125	11,014	13,243	2,155	6,283	386	11,760	1,497.3
Yearly																	
Total	12,540	3,809	1,810	1,743	392	1,792	116	658	689	786	349	334	-	-	-	62	1,675.4
Male	7,490	2,337	1,119	1,430	392	221	116	343	349	786		334	-	-	-	62	1,689.9
Female	5,049	1,472	690	313	-	1,571	-	314	341	-	349	-	-	-	-	-	1,653.8
Other/specify																	
Total	26,093	8,101	4,498	4,677	2,625	2,370	794	1,151	596	417	39	161	-	-	-	661	1,434.0
Male	18,916	4,423	2,430	4,239	2,224	2,038	794	1,151	338	417	39	161	-	-	-	661	1,710.6
Female	7,176	3,678	2,068	439	401	332	-	-	259	-	-	-	-	-	-	-	705.0
Not Stated																	4 000 5
Total	287	-	-	-	-	-	-	-	-	-	287	-	-	-	-	-	4,800.0
Male	287	-	-	-	-	-	-	-	-	-	287	-	-	-	-	-	4,800.0
Female	-	-	-	-	-	-	-	-	-	-		-	-	-	-	-	

Table 5.25 Paid Employees of Urban Population Aged Ten Years and Above by Amount of Payment per Month, Major Industrial Divisions and Sex -COUNTRY TOTAL: 2015

	1510115 4110	2 OOK 0			<u> </u>		ount of To	tal Paymer	nt Per Mont	h							Mean
																	Amount of
Major Industrial																	Total
Divisions	T.1.1	F00	500-	1000-	1500-	2000-	2500-	3000-	3500-	4000-	4500-	5000-	5500-	6000-	6500-	7000	Payment
and Sex	Total	<500	999	1499	1999	2499	2999	3499	3999	4499	4999	5499	5999	6499	6999	7000+	Per Month
Total Employed Popul		40 / 000	7/0.040	104 504		0/5 450	0.10 / / 4	0.40.000	440404	404 400	F0 000		00.044	00.050	0.505	00.017	4 004 0
Total	3,442,585	436,903	762,048	431,524	441,671	365,150	248,661	248,293	140,194	101,199	58,082	64,901	23,044	29,352	8,595	82,967	1,981.9
Male	2,024,497	134,451	352,954	265,848	281,171	242,767	171,166	174,418	102,891	77,641	46,387	51,658	20,890	23,069	8,099	71,086	2,354.1
Female	1,418,088	302,452	409,094	165,676	160,500	122,384	77,495	73,875	37,303	23,558	11,695	13,243	2,155	6,283	496	11,881	1,450.6
Agriculture, forestry ar	•																
Total	111,013	27,857	35,058	11,056	7,205	6,477	3,560	5,862	3,460	5,325	800	901	53	342	82	2,977	1,575.3
Male	68,468	13,836	18,202	8,005	4,993	5,446	2,568	4,324	2,567	3,719	454	901	53	342	82	2,977	1,900.3
Female	42,545	14,021	16,856	3,051	2,212	1,031	992	1,538	892	1,606	346	-	-	-	-	-	1,052.2
Mining and quarrying																	
Total	16,123	2,117	3,115	2,238	2,674	2,002	232	1,819	711	1,003	-	92	-	120	-	-	1,757.3
Male	13,162	915	2,735	2,238	2,344	1,574	232	1,412	711	1,003	-	-	-	-	-	-	1,793.3
Female	2,960	1,202	380	-	331	429	-	408	-	-	-	92	-	120	-	-	1,597.5
Manufacturing																	
Total	340,925	36,533	106,198	62,228	48,485	31,095	17,180	14,901	4,840	4,799	2,969	4,269	1,621	599	110	5,096	1,526.4
Male	227,857	19,986	57,255	44,559	36,717	25,790	11,380	10,676	4,076	4,347	2,882	3,474	1,530	497	-	4,688	1,696.1
Female	113,068	16,547	48,944	17,669	11,768	5,306	5,801	4,225	764	452	87	795	92	102	110	408	1,184.6
Electricity, gas, steam	and air condi	tioning sup _l															
Total	19,958	349	3,189	2,510	1,247	5,441	2,642	1,952	404	636	502	660	-	-	92	332	2,244.0
Male	16,285	349	2,091	1,987	1,247	5,003	1,028	1,952	404	636	502	660	-	-	92	332	2,345.3
Female	3,673	-	1,099	523	-	437	1,614	-	-	-	-	-	-	-	-	-	1,794.6
Water supply; sewerag	ge, waste mar		ind remedia	ation activi	ties												
Total	37,969	2,267	7,779	6,948	5,676	5,252	1,997	3,276	380	1,789	-	862	461	98	-	1,185	2,127.4
Male	25,088	1,078	3,434	4,746	3,861	5,161	1,323	1,336	287	1,748	-	410	461	98	-	1,144	2,400.4
Female	12,881	1,189	4,345	2,202	1,814	92	674	1,940	92	40	-	452	-	-	-	40	1,595.7
Construction																	
Total	285,958	21,877	65,261	50,778	34,435	24,785	18,300	24,303	10,145	10,785	5,050	6,255	1,791	3,806	1,068	7,320	2,162.5
Male	231,386	12,150	44,827	42,369	29,989	21,682	16,380	22,015	9,650	9,796	3,959	5,731	1,022	3,428	1,068	7,320	2,358.9
Female	54,572	9,727	20,434	8,409	4,446	3,103	1,920	2,287	495	989	1,091	523	769	378	-	-	1,330.2
Wholesale and retail tr	rade; repair of	motor vehi	cles and m	notorcycles													
Total	162,195	25,013	57,043	25,812	17,876	14,320	7,822	4,839	1,453	756	1,218	2,407	-	1,962	-	1,674	1,370.2
Male	99,702	15,467	29,436	15,855	13,898	9,976	4,186	3,901	1,043	756	886	1,108	-	1,616	-	1,576	1,498.9
Female	62,492	9,546	27,607	9,957	3,979	4,344	3,637	939	410	-	332	1,299	-	346	-	98	1,164.9

2

Table 5.25 Cont'd

l able 5.25	Contra					Λ	a.m.t af T-1	al Dayme - :-	t Don Maint	h							Moon
	<u> </u>			ı		AM	ount of Tot	ai Paymer.	it Per Iviont	П		ı			l		Mean Amount of
Major Industrial																	Total
Divisions			500-	1000-	1500-	2000-	2500-	3000-	3500-	4000-	4500-	5000-	5500-	6000-	6500-		Payment
and Sex	Total	< 500	999	1499	1999	2499	2999	3499	3999	4499	4999	5499	5999	6499	6999	7000+	Per Month
Transportation and	d storage																
Total	228,930	8,915	38,491	34,921	45,184	28,783	15,277	23,258	7,158	7,378	3,600	3,978	380	3,378	1,432	6,799	2,222.7
Male	212,835	8,282	34,106	32,404	41,914	27,604	13,355	22,468	6,824	6,910	3,559	3,978	380	3,287	1,432	6,330	2,246.9
Female	16,095	634	4,385	2,516	3,269	1,178	1,922	790	334	468	41	-	-	90	-	469	1,902.9
Accommodation a	and food service a	ctivities															
Total	178,183	43,388	81,320	27,708	11,936	5,611	3,569	1,421	860	405	345	523	-	362	405	331	902.0
Male	58,648	8,558	27,274	10,641	5,468	1,845	1,894	912	453	57	287	523	-	-	405	331	1,109.5
Female	119,536	34,829	54,046	17,067	6,468	3,765	1,675	509	408	348	58	-	-	362	-	-	800.3
Information and co	ommunication																
Total	59,935	369	8,464	10,816	7,602	8,572	4,157	5,362	3,978	4,617	775	1,901	1,081	783	98	1,358	2,469.8
Male	33,305	62	3,534	6,778	3,411	4,884	2,052	1,971	3,888	2,448	775	1,410	461	334	-	1,297	2,688.5
Female	26,630	307	4,930	4,038	4,192	3,687	2,105	3,392	90	2,169	-	491	621	448	98	61	2,196.3
Financial and insu	rance activities																
Total	157,603	1,877	18,030	17,268	12,323	16,891	14,054	19,448	14,080	10,996	7,441	8,370	1,107	5,090	1,177	9,451	3,155.8
Male	95,056	831	8,447	11,381	5,924	9,213	8,312	11,334	7,570	6,146	5,715	5,610	1,018	3,948	1,121	8,487	3,471.0
Female	62,548	1,047	9,583	5,887	6,400	7,677	5,742	8,114	6,510	4,850	1,726	2,760	89	1,142	56	964	2,676.8
Real estate activiti	ies																
Total	1,556	-	987	90	-	-	135	343	-	-	-	-	-	-	-	-	1,416.0
Male	673	-	583	90	-	-	-	-	-	-	-	-	-	-	-	-	719.4
Female	883	-	405	-	-	-	135	343	-	-	-	-	-	-	-	-	1,947.3
Professional, scien	ntific and technica	al activities															
Total	110,286	1,392	8,112	5,881	13,114	18,297	14,182	13,547	10,234	6,119	3,586	6,268	1,285	2,653	334	5,281	3,123.3
Male	71,974	757	3,943	1,898	6,908	11,898	10,941	8,571	7,649	4,971	2,364	3,735	1,285	1,787	334	4,935	3,425.3
Female	38,312	636	4,170	3,982	6,206	6,399	3,241	4,977	2,585	1,149	1,221	2,533	-	866	-	347	2,555.9
Administrative and	d support service a	activities															
Total	122,472	3,135	24,660	14,697	20,273	16,721	9,948	12,155	5,940	4,906	2,586	3,142	1,388	881	521	1,519	2,264.3
Male	68,899	1,083	9,015	7,622	12,523	8,775	7,595	6,378	4,130	4,116	2,236	2,556	1,388	360	521	602	2,492.8
Female	53,573	2,052	15,644	7,075	7,750	7,946	2,354	5,777	1,810	791	350	586	-	521	-	917	1,970.4
Public administrati	ion and defense;	compulsory	social seco	urity													
Total	349,400	8,044	41,357	38,030	57,981	52,022	36,432	40,210	23,647	14,236	7,262	13,620	6,261	1,734	1,177	7,387	2,524.7
Male	235,776	5,263	24,033	24,054	31,865	29,460	28,056	29,064	19,889	11,757	5,772	12,249	5,922	1,601	1,177	5,614	2,738.6
Female	113,623	2,781	17,325	13,976	26,116	22,562	8,376	11,146	3,759	2,479	1,490	1,371	338	133	_	1,773	2,080.8

Table 5.25 Cont'd

Table 5.25 C	ont a					Λ	ount of Tot	al Daymaan	t Dor Mont	<u> </u>							Moon
						Am	ount of Tot	ai Paymen	it Per Monti	1							Mean Amount of
Major Industrial																	Total
Divisions			500-	1000-	1500-	2000-	2500-	3000-	3500-	4000-	4500-	5000-	5500-	6000-	6500-		Payment
and Sex	Total	<500	999	1499	1999	2499	2999	3499	3999	4499	4999	5499	5999	6499	6999	7000+	Per Month
Education		Į.	· ·			<u> </u>	Į.	· ·	u.	<u> </u>	L. L.		ı. I		l l		
Total	459,378	11,208	58,288	45,947	74,925	70,484	65,849	40,059	37,951	16,402	12,105	6,543	3,204	2,002	1,255	13,155	2,468.3
Male	249,715	5,217	15,451	16,041	35,049	40,633	43,313	24,388	24,477	13,102	9,432	5,767	3,050	1,292	1,134	11,369	2,875.6
Female	209,662	5,991	42,837	29,906	39,876	29,851	22,536	15,671	13,474	3,300	2,672	776	154	710	122	1,786	1,983.2
Human health and so	cial work activi	ities															
Total	172,235	5,315	30,595	19,575	31,020	26,190	19,128	15,758	4,646	3,344	5,180	1,717	2,542	1,969	543	4,711	2,305.0
Male	66,825	1,280	8,195	3,023	8,340	12,103	8,923	8,925	2,018	1,804	3,364	1,443	2,449	1,391	433	3,134	2,924.0
Female	105,410	4,036	22,400	16,552	22,680	14,086	10,206	6,833	2,628	1,540	1,816	275	92	579	110	1,577	1,912.6
Arts, entertainment ar	nd recreation																
Total	33,047	3,484	4,331	3,797	6,380	5,428	3,116	4,296	545	522	91	713	-	-	-	343	1,922.6
Male	18,628	2,200	1,773	378	3,372	4,016	1,599	3,541	403	522	-	482	-	-	-	343	2,227.1
Female	14,418	1,284	2,559	3,419	3,008	1,412	1,516	756	143	-	91	231	-	-	-	-	1,529.3
Other service activitie	S																
Total	131,671	22,571	41,542	21,162	19,067	10,393	3,823	4,772	792	1,642	2,047	675	904	1,087	-	1,195	1,385.8
Male	90,408	12,109	28,478	15,838	14,451	7,528	3,014	2,839	401	876	2,047	-	904	1,087	-	835	1,456.8
Female	41,264	10,462	13,064	5,324	4,616	2,864	808	1,933	391	766	-	675	-	-	-	360	1,230.4
Activities of household	ds as employe	rs; undiffer	entiated go	ods- and s	ervices-pro	ducing act	tivities of h	ouseholds	for own use	9							
Total	386,694	207,000	118,241	20,844	14,505	8,791	2,419	4,202	4,952	1,282	1,320	164	272	802	-	1,900	686.7
Male	93,252	23,103	26,787	11,564	12,110	6,758	2,358	3,127	3,836	856	1,291	164	272	551	-	476	1,290.6
Female	293,442	183,898	91,454	9,280	2,395	2,034	61	1,074	1,116	427	29	-	-	251	-	1,423	494.7
Activities of extraterrit	orial organizat	ions and bo	odies Desci	ription													
Total	77,056	4,190	9,986	9,218	9,762	7,595	4,838	6,510	4,018	4,255	1,207	1,842	697	1,685	301	10,953	3,499.8
Male	46,555	1,926	3,357	4,377	6,788	3,417	2,658	5,286	2,616	2,071	861	1,457	697	1,449	301	9,296	4,248.8
Female	30,501	2,265	6,629	4,841	2,974	4,178	2,181	1,224	1,402	2,184	346	385	-	235	-	1,657	2,356.6

Table 5.26 Employed Population of Urban areas of Regions Aged Ten Years and above Who Were Not at Work by Reason for Absence, Region and Sex: 2015

	L AL VYOIK D	y Neason		n ce, Regio eason for Ab		1. 2013	
Region	 	Sick	Holiday/	Training/	Off Off	No work	Other/
and Sex	Total	Injury	Vacation	Education	Season	temporarily	Specify
COUNTRY TOTAL	i Ulai	irijury	vacalion	Luucalion	Jeasun	temporaniy	Ореспу
Total	382,173	83,657	69,921	16,813	58,521	62,341	90,920
Male	197,014	41,672	29,344	12,779	43,117		32,497
Female	185,159	41,986	40,576	4,035	15,404		58,423
	100,109	41,500	40,576	4,035	15,404	24,130	50,423
TIGRAY REGION	40,843	6 755	7 11 1	2 200	0 050	0.470	G OFF
Total		6,755 2,992	7,414 3,078	2,288	8,053 6,301	9,479 4,320	6,855 2,823
Male Female	21,392 19,452	2,992 3,763	4,336	1,879 409	1,752		4,032
AFFAR REGION	13,40Z	3,703	4,330	409	1,732	5,159	4,032
	12 667	2 267	2 204	150	2 002	1 560	2 202
Total	12,667	2,267	3,204	159	2,093		3,383
Male	6,394 6,373	426	2,040	150	1,413 680		1,382
Female	6,273	1,841	1,165	159	სგი	428	2,001
AMHARA REGION	100 500	22.254	24.044	0 575	10 744	16.750	40.400
Total Male	103,563	23,354	21,941	3,575	18,744		19,196
Male Female	52,340 51,223	14,116	5,943 15,998	1,682 1,894	14,090	•	5,049
Female	51,223	9,238	15,998	1,894	4,654	5,292	14,148
OROMIA REGION	115 114	20.240	44700	6 505	24 202	16 100	20.225
Total Malo	115,414	28,343	14,726	6,565	21,303		28,295
Male Female	62,932 52,482	13,406	7,562	6,023	15,879	8,482 7,700	11,580 16,715
Female SOMALI REGION	52,482	14,936	7,164	542	5,424	7,700	16,715
	E 740	4 757	4.004	0.40	000	CEO.	4.000
Total Malo	5,718	1,757	1,304	246	668		1,092
Male Female	2,411 3,307	840	590	240	365 303		312 780
Female	3,307	917	714	246	303	346	780
BENISHANGUL-GUMUZ		4 000	000	200	0.700	E4.4	4 000
Total Male	6,990 4,060	1,330	939 510	206 177	2,763		1,238
Male Female	4,060	1,151	519 420	177	1,667		383 855
Female	2,929	178	420	29	1,097	351	855
S.N.N.P REGION	44.004	0.000	7.000	0.004	0.000	0.054	40,000
Total	44,834	9,069	7,608	2,321	2,303		16,882
Male Female	20,007	3,190 5,870	3,755	2,321	1,606		5,341
Female	24,827	5,879	3,853	-	697	2,857	11,541
GAMBELLA REGION	0.754	070	070	4.40	040	202	445
Total Male	2,754	973 461	670	146 53	240	280	445
Male Female	1,395	461 512	240 430	53	131	280	230
Female	1,360	512	430	93	109	-	215
HARARI REGION	0.040	000	004	40	00	057	750
Total Male	2,648	600	891 346	49 40	99	257 257	752 351
Male	1,222	219	346 545	49	-	257	351
Female	1,426	381 N	545	-	99	-	400
ADDIS ABABA CITY ADN			0.000	4.004	0.004	0.000	40.000
Total Male	40,870	7,540	9,668	1,004	2,001	9,669	10,988
Male Female	22,275 18 505	4,152	4,632 5,036	342 663	1,665	7,066 2,603	4,418 6,570
Female	18,595	3,388	5,036	663	336	2,603	6,570
DIRE DAWA ADMINISTR		4.074	4 554	054	050	0.47	4 700
Total	5,871	1,671	1,554	254	253	347	1,793
Male	2,585	719	639	254	-	347	627
Female	3,285	952	915		253		1,166

Table 5.27 Employed Populations of Urban Areas Aged Ten Years and Above Whether Available and Ready to Work Additional Hours by Percent Available and Ready to Work Additional Hours, Age Group and Sex, - COUNTRY TOTAL: 2015

10 11011			Ready to Work Addition	nal Houre
Age Group	Total	Seeking or	Not seeking/	Percent available
and Sex	Employed	Available to Work	Available to Work	and Ready to work
and ook	Persons	additional Hours	additional Hours	Additional Hours
All ages	. 5.55115			
Total	7,083,094	2,832,969	4,250,124	40.0
Male	3,963,143	1,632,783	2,330,360	41.2
Female	3,119,950	1,200,186	1,919,764	38.5
10-14	, -,-30	,,- - -	, -,· - -	
Total	194,969	22,696	172,273	11.6
Male	100,707	14,614	86,093	14.5
Female	94,262	8,082	86,180	8.6
15-19				
Total	565,103	179,737	385,367	31.8
Male	224,956	75,041	149,914	33.4
Female	340,148	104,695	235,452	30.8
20-24	4 407 040	E00.050	E0E E00	47.4
Total	1,107,918	522,358	585,560	47.1 52.0
Male Female	546,853 561,064	284,469	262,385 323,175	52.0 42.4
	561,064	237,889	323,175	42.4
25-29 Total	1,455,213	652,100	803,114	44.8
Male	827,711	373,619	454,092	44.6 45.1
Female	627,502	278,481	349,022	44.4
30-34	5,502	0, .0 !	J.0,022	
Total	985,745	414,107	571,638	42.0
Male	599,578	256,444	343,134	42.8
Female	386,166	157,663	228,503	40.8
35-39				
Total	935,320	391,551	543,769	41.9
Male	518,080	207,950	310,130	40.1
Female	417,240	183,601	233,639	44.0
40-44		0.40 == :	00= 0:=	
Total	578,390	240,774	337,616	41.6
Male	365,839	152,399	213,441	41.7
Female 45-49	212,551	88,375	124,175	41.6
45-49 Total	421,698	161,363	260,335	38.3
notai Male	421,698 252,303	100,665	260,335 151,638	38.3 39.9
Female	252,303 169,394	60,698	108,697	35.8
50-54	100,004	55,050	100,001	50.0
Total	288,013	90,212	197,801	31.3
Male	173,544	55,120	118,424	31.8
Female	114,470	35,092	79,378	30.7
55-59	, -			
Total	210,275	64,642	145,633	30.7
Male	128,640	43,489	85,151	33.8
Female	81,635	21,152	60,483	25.9
60-64				
Total	145,598	43,227	102,371	29.7
Male	90,332	29,388	60,945	32.5
Female	55,266	13,840	41,426	25.0
65+		= = 0.00		
Total	194,851	50,204	144,648	25.8
Male	134,599	39,584	95,014	29.4
Female	60,253	10,619	49,633	17.6

Table 5.28 Employed Populations of Urban Areas Aged Ten Years and Above Who Sought or Available to Work Additional Hours by Number of Hours Worked Per Week, Age Group and Sex - COUNTRY TOTAL: 2015

						ours Worke						Mean No.
Age Group				110		Jaio Worke						Hours
and Sex	Total	^	4.2	4 40	12.21	22.20	24.20	40-48	40 F7	E0 74	75 .	Worked
A II. A	Total	0	1-3	4-12	13-21	22-30	31-39	40-40	49-57	58-74	75+	Per Week
All Ages	0.000.000	440 504	47.000	4== 000	007 500	000 470	044.045	505.000	400.000	004 040	0.4.0.000	44.0
Total	2,832,969	140,561	17,996	157,966	267,539	308,478	341,345	585,220	409,262	391,343	213,260	41.6
Male	1,632,783	77,979	4,926	63,299	127,973	154,525	195,868	359,199	253,767	256,085	139,162	44.1
Female	1,200,186	62,582	13,070	94,667	139,566	153,953	145,477	226,021	155,494	135,258	74,098	38.3
10-14												
Total	22,696	-	1,296	2,872	3,106	4,166	2,539	2,572	2,942	2,192	1,011	34.7
Male	14,614	-	40	2,728	2,415	2,122	1,695	1,401	1,981	1,694	538	35.1
Female	8,082	-	1,256	143	691	2,045	845	1,171	960	497	474	34.1
15-19												
Total	179,737	4,220	2,527	16,016	18,410	22,146	14,178	27,658	28,820	28,992	16,771	42.5
Male	75,041	1,353	207	5,596	9,253	6,041	5,248	11,364	13,597	13,963	8,419	45.6
Female	104,695	2,866	2,319	10,420	9,157	16,105	8,930	16,294	15,222	15,028	8,353	40.2
20-24												
Total	522,358	17,803	6,195	24,553	51,311	51,448	54,965	108,885	79,402	82,302	45,492	43.6
Male	284,469	8,287	2,693	11,787	24,984	24,658	28,671	58,239	46,065	49,576	29,509	45.8
Female	237,889	9,516	3,502	12,766	26,327	26,790	26,294	50,646	33,338	32,727	15,983	41.0
25-29	•	,	,	,	,	,	,	,	,	,	,	
Total	652,100	35,523	3,794	28,562	58,169	62,589	86,738	145,203	96,256	92,218	43,047	41.9
Male	373,619	18,165	1,038	11,450	26,684	31,946	45,128	87,079	57,494	65,614	29,021	44.7
Female	278,481	17,358	2,756	17,112	31,486	30,643	41,610	58,125	38,762	26,604	14,026	38.1
30-34	_,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	,	_,	,	.,	,	,	,	,		,	
Total	414,107	20,665	982	21,728	37,502	43,991	47,254	93,329	66,053	56,054	26,548	41.6
Male	256,444	13,301	135	8,659	17,911	23,831	31,809	61,445	42,267	38,269	18,817	43.9
Female	157,663	7,364	846	13,069	19,591	20,160	15,445	31,884	23,786	17,786	7,732	38.0
35-39	107,000	7,001	010	10,000	10,001	20,100	10,110	01,001	20,700	17,700	1,102	00.0
Total	391,551	22,283	1,054	23,795	35,591	49,084	52,426	74,019	51,839	50,086	31,375	41.0
Male	207,950	11,948	101	6,465	13,289	22,936	27,861	42,933	30,338	32,808	19,272	44.4
Female	183,601	10,335	953	17,330	22,302	26,147	24,565	31,086	21,501	17,279	12,103	37.1
40-44	100,001	10,333	300	17,550	22,302	20,147	24,303	31,000	21,501	17,273	12,103	37.1
	240 774	0.269		17 506	24 049	22 002	20.200	E1 2E7	20.720	27 020	10.072	42.2
Total	240,774	9,268	-	17,526	21,918	23,803	30,280	51,257	29,720	37,928	19,073	42.3
Male	152,399	6,353	-	6,483	11,102	15,367	20,646	37,378	19,235	23,257	12,577	43.5
Female	88,375	2,915	-	11,043	10,816	8,437	9,634	13,879	10,485	14,671	6,495	40.3

Table 5.28 Cont'd

				Nu	mber of Ho	urs Worke	d Per Weel	<				Mean No.
Age Group and Sex	Total	0	4.2	4.42	12.21	22.20	24.20	40.49	40 F7	E0 74	75.	Hours Worked Per
45.40	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Week
45-49												
Total	161,363	8,756	62	8,405	14,150	18,057	22,050	40,969	20,307	16,199	12,409	40.7
Male	100,665	4,414	62	2,760	5,933	9,674	14,068	27,506	14,252	12,982	9,013	44.2
Female	60,698	4,342	-	5,644	8,216	8,383	7,982	13,463	6,055	3,217	3,396	35.0
50-54												
Total	90,212	6,135	505	4,192	11,348	12,079	14,264	13,548	11,980	9,276	6,885	39.3
Male	55,120	4,813	150	2,858	7,028	7,040	7,151	8,852	8,484	5,713	3,030	38.2
Female	35,092	1,322	355	1,334	4,319	5,039	7,112	4,696	3,496	3,563	3,855	41.0
55-59												
Total	64,642	6,885	783	3,901	5,229	10,372	7,104	10,735	6,527	7,233	5,873	37.9
Male	43,489	4,038	389	1,937	2,371	5,208	5,928	8,582	5,737	4,654	4,645	41.5
Female	21,152	2,847	394	1,964	2,858	5,163	1,177	2,153	790	2,578	1,229	30.5
60-64												
Total	43,227	3,512	480	2,031	4,217	4,870	5,566	8,369	6,885	4,543	2,755	38.2
Male	29,388	1,581	-	803	1,459	1,767	4,465	6,628	6,828	3,458	2,398	44.3
Female	13,840	1,931	480	1,228	2,758	3,103	1,100	1,740	57	1,086	357	25.2
65+												
Total	50,204	5,511	318	4,384	6,590	5,873	3,980	8,677	8,532	4,320	2,019	34.6
Male	39,584	3,726	110	1,771	5,544	3,934	3,199	7,792	7,488	4,097	1,923	37.9
Female	10,619	1,785	207	2,614	1,045	1,938	781	885	1,044	223	96	22.2

Table 5.29 Employed Population of Urban Areas of Regions Aged Ten Years and above Who Were Willing and Available to Work Additional Hours by Type of Work Sought or Available for and Sex: 2015

	T and Ocx. 2015	vpe of Work	Sought or Available	for
Region		At the	Other job in	Other job with more
and Sex		present	addition to	hours to replace the
	Total	main job	the present	present job
COUNTRY TOTAL		,	'	, ,
Total	2,832,969	694,510	1,466,820	671,639
Male	1,632,783	438,624	856,316	337,844
Female	1,200,186	255,886	610,504	333,795
TIGRAY REGION				
Total	167,712	49,492	73,810	44,409
Male	89,871	30,011	36,041	23,819
Female	77,840	19,481	37,769	20,590
AFFAR REGION				
Total	73,845	28,911	35,324	9,609
Male	44,700	15,206	23,958	5,536
Female	29,145	13,706	11,366	4,073
AMHARA REGION				
Total	549,726	101,040	275,365	173,322
Male	305,073	67,983	157,195	79,895
Female	244,653	33,057	118,169	93,427
OROMIA REGION				
Total	961,608	278,905	527,667	155,036
Male	570,738	175,287	312,038	83,413
Female	390,870	103,618	215,629	71,623
SOMALI REGION				
Total	28,520	9,904	9,947	8,669
Male	17,390	7,358	5,676	4,356
Female	11,130	2,546	4,271	4,313
BENISHANGUL-GUMUZ F	REGION			
Total	38,538	13,552	20,960	4,026
Male	22,951	8,676	12,403	1,872
Female	15,587	4,876	8,556	2,154
S.N.N.P REGION				
Total	542,795	100,146	305,924	136,726
Male	307,011	62,734	180,054	64,223
Female	235,784	37,412	125,870	72,503
GAMBELLA REGION				
Total	23,468	4,896	15,228	3,344
Male	13,879	3,321	8,732	1,826
Female	9,589	1,575	6,495	1,518
HARARI REGION				
Total	31,392	11,531	16,334	3,526
Male .	16,268	7,197	7,337	1,735
Female	15,124	4,334	8,998	1,792
ADDIS ABABA CITY ADMI				
Total	366,809	86,990	169,117	110,702
Male .	219,012	56,262	102,698	60,053
Female	147,797	30,728	66,419	50,649
DIRE DAWA ADMINISTRA		.		
Total	48,558	9,143	17,146	22,269
Male	25,890	4,590	10,184	11,116
Female	22,668	4,553	6,962	11,154

Table 5.30 Employed Population of Urban Areas Aged Ten Years and Above by Sector of Economy, Percentage Share of Informal, Age Group and Sex – COUNTRY TOTAL: 2015

Ago Group		Sector of	Economy		Porcont
Age Group and Sex		Formal	Informal	Not	Percent Informal
and Sex	Total	Job	Job	Identified	IIIIOIIIIai
All Ages					
Total	6,020,229	4,228,119	1,675,501	116,609	27.8
Male	3,520,231	2,682,435	763,659	74,137	21.7
Female	2,499,998	1,545,684	911,842	42,472	36.5
10-14					
Total	84,445	31,520	51,324	1,601	60.8
Male	51,041	19,946	29,959	1,135	58.7
Female	33,404	11,574	21,365	465	64.0
15-19	050 005	007.000	4.40.404	7 400	20.0
Total	356,985	207,089	142,404	7,492	39.9
Male	174,088	97,566	73,055	3,467	42.0
Female	182,897	109,523	69,349	4,024	37.9
20-24 Total	027 404	604 760	220.072	15 750	24.5
Total	937,491	691,760	229,972	15,759	24.5
Male	484,176	363,410 328,350	114,292	6,473	23.6
Female 25-29	453,315	320,330	115,679	9,286	25.5
Total	1,290,360	962,132	300,972	27,256	23.3
Male	758,751	596,669	144,492	17,589	19.0
Female	531,609	365,463	156,480	9,666	29.4
30-34	001,000	000,400	100,400	5,000	20.4
Total	879,859	652,119	206,222	21,518	23.4
Male	545,260	434,685	95,864	14,710	17.6
Female	334,599	217,435	110,357	6,808	33.0
35-39	,	_,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		-,	
Total	837,909	588,364	235,316	14,229	28.1
Male	474,103	373,570	90,139	10,394	19.0
Female	363,806	214,794	145,177	3,835	39.9
40-44					
Total	518,708	369,037	141,653	8,018	27.3
Male	329,555	258,415	65,687	5,453	19.9
Female	189,153	110,621	75,966	2,565	40.2
45-49					
Total	386,553	268,078	112,772	5,702	29.2
Male	235,020	191,104	40,649	3,266	17.3
Female	151,533	76,974	72,123	2,436	47.6
50-54					
Total	260,586	169,247	85,976	5,363	33.0
Male	161,494	121,312	36,349	3,833	22.5
Female	99,092	47,936	49,627	1,530	50.1
55-59	407.050	105 510	E0 0EE	2.000	24.0
Total	187,253	125,510	58,055	3,688	31.0
Male Female	115,394	89,322	23,717	2,355	20.6
60-64	71,859	36,188	34,337	1,333	47.8
Total	122,315	72,386	47,669	2,260	39.0
Male	79,977	59,219	18,872	1,886	23.6
Female	42,338	13,167	28,797	373	68.0
65+	72,330	13, 107	20,131	3/3	00.0
Total	157,767	90,876	63,166	3,724	40.0
Male	111,373	77,217	30,582	3,574	27.5
Female	46,394	13,660	32,584	150	70.2
	.5,551	. 0,000	32,001		

Table 5.31 Employed Population of Urban Areas Aged Ten Years and Above by Sector of Economy, Percentage Share of Informal, Region and Sex: 2015

Economy, F	rercentage S		rmal, Region Economy	anu Sex. 20	13
Region				Not	Percent
and Sex	Total	Formal	Informal		Informal
	Total	Job	Job	Identified	
COUNTRY TOTAL	0.000.000	4 000 440	4 075 504	440.000	07.0
Total	6,020,229	4,228,119	1,675,501	116,609	27.8
Male .	3,520,231	2,682,435	763,659	74,137	21.7
Female	2,499,998	1,545,684	911,842	42,472	36.5
TIGRAY REGION					
Total	422,736	315,144	106,604	988	25.2
Male	231,365	188,210	42,258	898	18.3
Female	191,371	126,934	64,346	90	33.6
AFFAR REGION					
Total	119,651	77,580	41,529	541	34.7
Male	68,597	52,355	15,701	541	22.9
Female	51,054	25,226	25,828	-	50.6
AMHARA REGION					
Total	1,221,815	895,706	323,111	2,998	26.4
Male	711,170	577,921	131,579	1,670	18.5
Female	510,645	317,785	191,532	1,328	37.5
OROMIA REGION					
Total	1,804,898	1,166,868	611,476	26,554	33.9
Male	1,072,238	760,823	292,831	18,585	27.3
Female	732,660	406,045	318,645	7,970	43.5
SOMALI REGION					
Total	98,090	56,126	41,536	428	42.3
Male	58,907	39,336	19,143	428	32.5
Female	39,184	16,791	22,393	-	57.1
BENISHANGUL-GUMUZ	REGION				
Total	85,026	62,167	22,743	116	26.7
Male	48,469	40,147	8,206	116	16.9
Female	36,557	22,020	14,537	-	39.8
S.N.N.P REGION	ŕ	,	,		
Total	918,814	599,675	317,294	1,844	34.5
Male	527,209	383,109	143,060	1,041	27.1
Female	391,605	216,566	174,235	804	44.5
GAMBELLA REGION	,	,	,		
Total	46,243	33,333	12,116	795	26.2
Male	26,438	21,146	5,003	288	18.9
Female	19,805	12,186	7,113	506	35.9
HARARI REGION	-,	,	, -		
Total	53,729	35,560	17,337	833	32.3
Male	28,612	21,288	6,812	513	23.8
Female	25,117	14,271	10,525	320	41.9
ADDIS ABABA CITY ADN			. 0,020	0_0	
Total	1,146,136	922,147	148,803	75,186	13.0
Male	688,572	556,656	85,189	46,726	12.4
Female	457,564	365,490	63,614	28,459	13.9
DIRE DAWA ADMINISTR		000,400	00,014	20,700	10.9
Total	103,091	63,814	32,952	6,325	32.0
Male	58,654	41,445	13,878	3,331	23.7
			·	·	
Female	44,437	22,369	19,074	2,994	42.9

Table 5.32 Employed Populations of Major Towns Aged Ten Years and Above by Sector of Economy, Percentage Share of Informal and Sex: 2015

Sector of	Economy,	Percentage S	nare of into	ormai and S	ex: 2015
Major Towns		Sector of E	Economy		Percent
and Sex		Formal	Informal	Not	Informal
and ocx	Total	Job	Job	Identified	
COUNTRY TOTAL		•		· '	
Total	2,160,342	1,681,043	392,330	86,969	18.2
Male	1,289,283	1,037,348	198,531	53,404	15.4
Female	871,059	643,694	193,799	33,566	22.2
MEKELE TOWN					
Total	113,465	87,867	24,978	620	22.0
Male	65,152	53,166	11,456	530	17.6
Female	48,313	34,700	13,522	90	28.0
ASAYITA TOWN					
Total	11,418	7,690	3,628	99	31.8
Male	6,821	5,201	1,521	99	22.3
Female	4,596	2,489	2,107	-	45.8
GONDER TOWN	o=:	= 0.555		4.4.5	
Total	95,564	73,288	21,866	410	22.9
Male	58,323	48,779	9,236	308	15.8
Female	37,241	24,509	12,630	102	33.9
DESSIE TOWN	04.455	40.000	40.047	444	05.0
Total	64,155	48,028	16,017	111	25.0
Male	40,931	32,782	8,149	- 111	19.9
Female BAHIR DAR TOWN	23,224	15,245	7,868	111	33.9
Total	116,025	97,003	18,122	900	15.6
Male	64,390	97,003 56,142	7,970	900 279	12.4
Female	54,390 51,635	40,861	7,970 10,152	622	19.7
BISHOFTU TOWN	01,000	-10,00 1	10,102	022	10.7
Total	53,277	43,640	9,522	115	17.9
Male	30,816	26,779	3,980	57	12.9
Female	22,461	16,861	5,542	58	24.7
ADAMA TOWN	,	-,	- ,	- -	
Total	99,665	79,601	20,064	-	20.1
Male	60,469	49,939	10,530	-	17.4
Female	39,196	29,662	9,534	-	24.3
JIMMA TOWN	•	-			
Total	65,983	47,637	17,809	538	27.0
Male	40,357	30,167	9,736	454	24.1
Female	25,627	17,470	8,073	84	31.5
SHASHEMENE					
TOWN					
Total	43,735	29,551	13,765	420	31.5
Male	26,416	20,160	5,960	296	22.6
Female	17,319	9,391	7,805	123	45.1
JIJIGA TOWN	05.07.	04 == 4	40.000	400	07.1
Total	35,274	21,754	13,092	428	37.1
Male	23,056	15,814	6,815	428	29.6
Female	12,217	5,941	6,276	-	51.4
ASOSA TOWN	10 510	16 707	2 044		4.4.4
Total Mole	19,519	16,707	2,811	-	14.4
Male	12,218	11,269	949	-	7.8 25.5
Female	7,301	5,438	1,863	-	25.5

Table 5.32 Cont'd

Table 5.32 Cont d					
Major Towns		Sector of I	Economy		Percent
Major Towns and Sex		Formal	Informal	Not	Informal
and Sex	Total	Job	Job	Identified	
HAWASSA TOWN					
Total	115,715	88,416	27,107	191	23.4
Male	71,026	55,904	15,027	96	21.2
Female	44,689	32,513	12,080	95	27.0
GAMBELLA TOWN					
Total	23,592	18,340	4,458	795	18.9
Male	13,469	11,856	1,324	288	9.8
Female	10,123	6,483	3,133	506	31.0
HARAR TOWN					
Total	53,729	35,560	17,337	833	32.3
Male	28,612	21,288	6,812	513	23.8
Female	25,117	14,271	10,525	320	41.9
ADDIS ABABA CITY AI	OMINISTRA	TION			
Total	1,146,136	922,147	148,803	75,186	13.0
Male	688,572	556,656	85,189	46,726	12.4
Female	457,564	365,490	63,614	28,459	13.9
DIRE DAWA TOWN					
Total	103,091	63,814	32,952	6,325	32.0
Male	58,654	41,445	13,878	3,331	23.7
Female	44,437	22,369	19,074	2,994	42.9

Table 5.33 Distribution of Urban Population Aged Ten years and above Working in the Informal Sector by Major Occupational Groups, Region and Sex: 2015

Major Occupational Groups												
		T		Major	Occupation	nal Group		·	T			
				 			Skilled	Craft	Plant and			
D				Technicians		Service	agricultural,	and	machine			
Region and				and	Clerical	and	forestry and		operators,			
Sex	Total	Managara	Drofessionals	associate	support	sales	fishery	trades	and	Elementary		
COLINITRY TOTAL	TOTAL	Managers	Professionals	professionals	workers	workers	workers	workers	assemblers	occupations		
COUNTRY TOTAL	4 075 504	F 000	40.047	04.540	4 7 40	705.404	450 450	440.000	07.000	000 400		
Total	1,675,501	5,836	10,017	24,510	•	725,191	•	410,328	37,928	303,489		
Male .	763,659	3,030	6,659	19,235	•	200,856		202,692	34,548	189,919		
Female	911,842	2,806	3,358	5,274	1,025	524,336	50,458	207,637	3,380	113,569		
TIGRAY REGION												
Total	106,604	610	685	1,249	357	38,191	11,574	26,516	7,245	20,177		
Male	42,258	269	329	921	223	5,666	6,966	11,177	4,952	11,755		
Female	64,346	342	357	328	133	32,525	4,608	15,339	2,293	8,422		
AFFAR REGION												
Total	41,529	214	-	-	-	20,326	7,350	5,532	568	7,539		
Male	15,701	-	-	-	-	4,947	5,398	1,667	568	3,122		
Female	25,828	214	-	-	-	15,379	1,952	3,865	-	4,418		
AMHARA REGION												
Total	323,111	1,037	103	1,997	-	129,778	44,336	80,695	2,654	62,510		
Male	131,579	592	103	1,937	-	24,641	27,130	36,283	2,654	38,239		
Female	191,532	445	-	60	-	105,138	17,206	44,412	-	24,271		
OROMIA REGION												
Total	611,476	2,080	2,508	6,328	1,319	281,839	45,540	165,974	14,993	90,894		
Male	292,831	1,068	2,475	4,743	1,109	83,192	36,515	82,862	14,877	65,991		
Female	318,645	1,012	34	1,585	211	198,648	9,025	83,112	116	24,904		
SOMALI REGION												
Total	41,536	81	-	1,596	-	22,495	4,340	5,243	832	6,949		
Male	19,143	-	-	1,246	-	6,969	1,841	2,871	668	5,548		
Female	22,393	81	-	350	-	15,526	2,499	2,372	165	1,400		
BENISHANGUL-GUMUZ R						,	,	•		•		
Total	22,743	_	-	29	_	9,104	5,289	3,491	-	4,830		
Male	8,206	_	-	29	_	691	2,879	1,876	-	2,730		
Female	14,537	_	_	-	_	8,413	2,410	1,615	_	2,100		
S.N.N.P. REGION	.,					-,	_,	, •		_,		
Total	317,294	_	1,299	4,454	1.544	140,790	26,872	73,365	3,656	65,315		
Male	143,060	_	696	1,845	1,544	44,159	17,109	39,458	3,355	34,894		
Female	174,235	-	603	2,609	-,	96,631	9,763	33,907	301	30,421		

 $\frac{1}{2}$

Table 5.33 Cont'd

Table 5.55 Cont a				Major	Occupation	nal Group)S			
				,			Skilled	Craft	Plant and	
				Technicians		Service	agricultural,	and	machine	
Region and				and	Clerical	and	forestry and	related	operators,	
Sex				associate	support	sales	fishery	trades	and	Elementary
	Total	Managers	Professionals	professionals	workers	workers	workers	workers	assemblers	occupations
GAMBELLA REGION										
Total	12,116	-	-	116	239	5,255	1,381	2,141	155	2,829
Male	5,003	-	-	116	164	1,293	983	662	133	1,653
Female	7,113	-	-	-	75	3,962	399	1,479	22	1,176
HARARI REGION										
Total	17,337	81	98	351	88	7,792	166	5,605	220	2,935
Male	6,812	41	98	351	-	1,085	117	2,726	220	2,174
Female	10,525	41	-	-	88	6,707	49	2,879	-	761
ADDIS ABABA CITY ADM	IINISTRATIC	N								
Total	148,803	1,733	5,237	8,045	1,195	56,685	3,733	33,298	7,261	31,616
Male	85,189	1,061	2,873	7,702	678	25,235	2,401	19,750	6,778	18,712
Female	63,614	672	2,365	343	517	31,450	1,332	13,549	483	12,904
DIRE DAWA ADMINISTR	ATION									
Total	32,952	-	86	346	-	12,936	2,876	8,468	344	7,895
Male	13,878	-	86	346	-	2,979	1,660	3,360	344	5,102
Female	19,074	-	-	-	-	9,958	1,215	5,107	-	2,793

Table 5.34: Distribution of Urban Population of Regions Aged Ten Years and above Who were Working in the Informal Sector by Major Branches of Industry Region and Sex: 2015

Region	and Sex: 2		sector Of Econo	mγ	
		Manufacturing,			Agriculture,
Region		Construction,	Whole Sale	Other	Hunting,
and Sex		Mining &	and Retail	Service	Forestry &
	Total	Quarrying	Trade	Sectors	Fishing
COUNTRY TOTAL	•	, ,			
	1,675,50				
Total	1	503,077	599,969	383,233	189,222
Male	763,659	236,723	201,746	201,335	123,855
Female	911,842	266,354	398,223	181,897	65,367
TIGRAY REGION					
Total	106,604	33,462	33,353	23,659	16,130
Male	42,258	19,510	6,557	6,619	9,572
Female	64,346	13,952	26,796	17,040	6,559
AFFAR REGION					
Total	41,529	7,682	16,634	8,416	8,797
Male	15,701	2,196	4,430	2,253	6,822
Female	25,828	5,486	12,204	6,163	1,975
AMHARA REGION	000 444	440.400	00.074		
Total	323,111	112,139	96,654	57,566	56,751
Male	131,579	42,684	27,176	26,184	35,534
Female	191,532	69,455	69,478	31,382	21,218
OROMIA REGION	044 470	400 454	004.000	407.000	50,000
Total	611,476	188,454	231,866	137,223	53,933
Male	292,831	94,815	84,168	71,548	42,300
Female	318,645	93,639	147,698	65,675	11,633
SOMALI REGION Total	41,536	2 902	22 556	9 622	5 166
Male	19,143	3,892 3,296	23,556 6,519	8,622 7,010	5,466 2,319
Female	22,393	597	17,037	1,612	3,147
BENISHANGUL-GUI			17,037	1,012	5,147
Total	22,743	7,225	4,992	4,426	6,100
Male	8,206	1,801	1,156	2,008	3,242
Female	14,537	5,424	3,836	2,419	2,858
S.N.N.P REGION	14,007	0,424	0,000	2,410	2,000
Total	317,294	96,306	112,079	73,752	35,157
Male	143,060	41,359	39,094	42,710	19,897
Female	174,235	54,947	72,985	31,042	15,260
GAMBELLA REGION		2 1,2 11	-,	,	,
Total	12,116	2,406	4,030	4,509	1,170
Male	5,003	729	1,396	1,896	983
Female	7,113	1,678	2,634	2,613	188
HARARI REGION					
Total	17,337	4,377	8,682	4,112	166
Male	6,812	2,432	1,603	2,660	117
Female	10,525	1,945	7,079	1,453	49
ADDIS ABABA CITY	ADMINIST	RATION			
Total	148,803	39,852	54,989	51,536	2,427
Male	85,189	24,039	26,755	32,986	1,409
Female	63,614	15,813	28,234	18,549	1,018
DIRE DAWA ADMIN					
Total	32,952	7,282	13,135	9,411	3,123
Male	13,878	3,863	2,893	5,461	1,660
Female	19,074	3,419	10,242	3,950	1,463

Table 5.35 Distribution of Urban Population of Regions Aged ten years and above Working in the Informal Sector by Highest Grade Completed, Region and Sex: 2015

•						Highes	st Grade Com	pleted					Never
Region					_			10+2/		Diploma or		_	Attended
and Sex		_			Secondary			Certificate	0 .:6 .	Degree	D: 1	Degree	Formal/
	Total	Pre	Non-	Grades	not	Secondary	Duan aratam.	Not	Certificate	Not	Diploma	and above	Informal Education
COLINITRY TOTAL	Total	SCHOOL	Formal	1-8	completed	Completed	Preparatory	Completed	Completed	Completed	Completed	Completed	Education
COUNTRY TOTAL	4 075 504	E 044	70.040	740 770	400 470	400.005	47.544	4.050	40.445	44 700	00.707	40.470	440.504
Total	1,675,501			740,773	108,470	199,385	17,544	1,859	10,415	11,786	38,797	12,179	449,534
Male	763,659			385,694	60,466	109,655	12,329	656	6,006	7,012	21,818	9,053	120,290
Female	911,842	5,131	48,948	355,079	48,004	89,730	5,216	1,203	4,409	4,774	16,979	3,126	329,244
TIGRAY REGION													
Total	106,604	-	0,1 .0	47,882	3,813	15,700	882	-	710	415	4,311	1,646	24,495
Male	42,258	-	2,390	20,307	1,796	7,727	314	-	361	191	2,395	751	6,025
Female	64,346	-	4,359	27,576	2,017	7,973	569	-	349	224	1,915	895	18,470
AFFAR REGION													
Total	41,529	-	.,	18,546	2,014	1,306	252	16	24	16	846	140	16,463
Male	15,701	-	214	8,908	491	858	34	-	-	-	290	140	4,766
Female	25,828	-	1,694	9,638	1,523	448	218	16	24	16	556	-	11,697
AMHARA REGION													
Total	323,111			113,960	16,598	32,573	3,789	506	2,753	2,395	3,450	288	107,329
Male	131,579		14,294	57,372	8,619	14,411	3,027	445	1,471	752	1,841	221	28,931
Female	191,532	553	24,428	56,587	7,979	18,163	762	61	1,282	1,643	1,609	67	78,398
OROMIA REGION													
Total	611,476	3,022	16,228	282,093	48,296	74,130	5,386	577	2,901	4,750	11,890	5,799	156,404
Male	292,831	456	8,553	151,255	26,355	40,994	3,891	-	1,545	3,079	6,823	4,969	44,909
Female	318,645	2,566	7,674	130,838	21,941	33,136	1,495	577	1,356	1,671	5,067	830	111,495
SOMALI REGION													
Total	41,536	274	1,562	11,988	1,810	1,495	350	-	-	82	372	315	23,288
Male	19,143	51	900	6,718	1,098	1,056	233	-	-	82	372	233	8,401
Female	22,393	223	662	5,269	713	439	118	-	-	-	-	82	14,887
BENISHANGUL-GUMUZ	REGION												
Total	22,743	246	2,634	9,004	672	1,830	-	-	160	199	490	28	7,481
Male	8,206	77	979	4,097	229	944	-	-	-	-	29	28	1,823
Female	14,537	169	1,655	4,907	443	885	-	-	160	199	462	-	5,657
S.N.N.P. REGION													
Total	317,294	804	5,544	163,241	14,496	40,473	2,994	562	2,926	2,375	7,094	1,968	74,818
Male	143,060	-	799	83,220	8,333	21,434	2,231	162	2,203	1,641	4,891	1,382	16,763
Female	174,235	804	4,744	80,021	6,162	19,038	764	401	723	734	2,203	586	58,054

Table 5.35 Cont'd

	Highest Grade Completed												
Region								10+2/		Diploma or			Attended
and Sex					Secondary			Certificate		Degree		Degree	Formal/
		Pre	_Non-	Grades	not	Secondary		Not	Certificate	Not	Diploma	and above	Informal
	Total	school	Formal	1-8	completed	Completed	tory	Completed	Completed	Completed	Completed	Completed	Education
GAMBELLA REGION													
Total	12,116	53	193	5,784	602	1,651	299	62	91	128	341	-	2,912
Male	5,003	-	98	2,703	129	927	135	-	45	128	226	-	612
Female	7,113	53	95	3,080	473	724	164	62	46	-	116	-	2,300
HARARI REGION													
Total	17,337	-	307	8,496	1,687	2,025	98	49	144	48	417	89	3,976
Male	6,812	-	132	3,801	898	1,238	49	49	48	48	149	89	311
Female	10,525	-	176	4,695	789	787	49	-	97	-	268	-	3,665
ADDIS ABABA CITY ADM	IINISTRATI	ON											
Total	148,803	678	2,361	64,126	17,043	25,118	3,164	-	620	1,166	9,013	1,905	23,608
Male	85,189	-	610	40,146	11,785	17,816	2,086	-	333	879	4,404	1,239	5,891
Female	63,614	678	1,751	23,980	5,259	7,302	1,078	-	287	287	4,610	666	17,717
DIRE DAWA ADMINISTRA	ATION												
Total	32,952	86	2,640	15,654	1,438	3,085	330	86	86	212	572	-	8,761
Male	13,878	-	931	7,167	733	2,249	330	-	-	212	399	-	1,857
Female	19,074	86	1,709	8,487	705	836	-	86	86	-	173	-	6,904

Table 5.36 Distribution of Urban Population of Regions Aged Ten Years and Above Working in the Informal Sector by Type of Working Place, Region and Sex: 2015

							Types of V	Vorking P	lace					
Region							Farm			Any	in	Where		
and Sex		Business		At	On	Open	area		Quarrying	whereas		customer	Lakes/	Other
	Total	house	Office	home	street	market	/field	Factory	/mining	found	site	available	river	/specify/
COUNTRY TOTAL														
Total	1,675,501			486,048		312,480	127,472	4,351	4,589	,	58,862	66,638	6,155	49,529
Male	763,659		20,285	97,446	93,611	99,186	95,150	2,092	2,859	143,318	48,402	42,765	4,459	33,163
Female	911,842	93,808	13,013	388,601	94,400	213,294	32,322	2,259	1,730	20,020	10,460	23,873	1,697	16,366
TIGRAY REGION														
Total	106,604	13,555	3,597	41,645	7,967	8,091	8,631	1,021	-	10,921	7,011	2,703	723	740
Male	42,258	5,578	1,797	4,714	2,693	1,385	7,196	511	-	9,419	6,227	1,757	723	259
Female	64,346	7,977	1,799	36,931	5,274	6,706	1,434	510	-	1,502	785	946	-	482
AFFAR REGION														
Total	41,529	3,591	676	13,086	8,601	2,545	5,113	35	437	796	375	2,196	218	3,860
Male	15,701	751	-	2,340	2,400	577	3,480	-	437	780	167	1,753	218	2,797
Female	25,828	2,840	676	10,746	6,201	1,968	1,633	35	-	16	207	443	-	1,063
AMHARA REGION														
Total	323,111	26,429		112,905	22,400	49,028	46,202	-	1,540	32,008	11,380	10,007	592	7,797
Male	131,579	12,400	1,372	15,948	12,469	14,056	32,470	-	950	24,056	9,137	4,682	592	3,447
Female	191,532	14,029	1,452	96,956	9,931	34,972	13,732	-	590	7,952	2,243	5,325	-	4,349
OROMIA REGION														
Total	611,476	56,628	7,631	182,707	71,684	140,514	39,588	1,661	1,441	48,720	19,001	19,563	2,877	19,461
Male	292,831	24,362	4,912	52,512	36,648	47,721	32,683	756	1,047	46,508	16,130	13,844	1,816	13,893
Female	318,645	32,265	2,718	130,196	35,036	92,794	6,905	905	394	2,213	2,872	5,719	1,061	5,569
SOMALI REGION														
Total	41,536	8,532	589	3,122	3,963	9,010	5,422	-	-	5,431	1,062	2,868	-	1,537
Male	19,143	3,425	589	630	1,122	2,861	2,189	-	-	4,512	1,062	2,215	-	537
Female	22,393	5,107	-	2,492	2,841	6,149	3,232	-	-	919	-	653	-	1,000
BENISHANGUL-GUMUZ	REGION													
Total	22,743	1,977	259	9,403	509	2,138	3,606	-	-	2,495	424	261	379	1,290
Male	8,206	732	-	1,000	106	308	2,487	-	-	2,233	293	232	286	528
Female	14,537	1,245	259	8,403	403	1,830	1,120	-	-	262	132	29	92	762
S.N.N.P. REGION														
Total	317,294	33,768	5,564	78,615	39,095	76,157	13,121	863	662	33,869	8,018	19,416	1,019	7,127
Male	143,060	17,762	3,706	8,968	21,348	23,427	11,018	563	302	32,221	6,489	10,766	521	5,968
Female	174,235	16,006	1,858	69,647	17,748	52,729	2,104	301	360	1,648	1,528	8,650	498	1,158

Table 5.36 Cont'd

							Types of V	Vorking P	lace					
Region							Farm			Any	in	Where		
and Sex		Business		At	On	Open	area		Quarrying	whereas	construction	customer	Lakes/	Other
	Total	house	Office	home	street	market	/field	Factory	/mining	found	site	available	river	/specify/
GAMBELLA REGION														
Total	12,116	478	162	5,394	1,506	1,236	1,171	49	-	1,305	43	297	349	127
Male	5,003	197	40	709	483	620	1,063	49	-	1,183	43	231	304	81
Female	7,113	281	121	4,685	1,023	616	108	-	-	122	-	66	45	46
HARARI REGION														
Total	17,337	1,434	392	3,490	4,354	2,739	117	-	165	1,622	1,550	1,424	-	49
Male	6,812	399	256	336	1,199	298	117	-	123	1,540	1,462	1,032	-	49
Female	10,525	1,035	137	3,154	3,154	2,441	-	-	42	82	88	392	-	-
ADDIS ABABA CITY ADI	<i>I</i> INISTRATI	ON												
Total	148,803	25,883	11,279	29,184	20,664	16,789	1,893	348	344	21,968	7,432	6,574	-	6,446
Male	85,189	14,274	7,287	9,596	13,447	6,799	873	-	-	17,020	5,533	5,333	-	5,028
Female	63,614	11,609	3,993	19,588	7,216	9,990	1,020	348	344	4,948	1,899	1,241	-	1,418
DIRE DAWA ADMINISTI	RATION													
Total	32,952	2,458	325	6,497	7,268	4,233	2,608	374	-	4,200	2,564	1,328	-	1,095
Male	13,878	1,043	325	694	1,696	1,134	1,574	213	-	3,845	1,859	918	-	577
Female	19,074	1,415	-	5,803	5,572	3,100	1,034	161	-	355	706	409	-	518

Employed Population Urban Areas Aged Ten Years and Above by Mean Number of Hours Per Week in Major Branches of Industry, Region and Sex: 2015 **Table 5.37**

Region ai					
	Me	an Number of Ho			k in Major
		Bran	ches of Indu	ustry	
Region		Manufacturing,	Whole	O41	Agriculture,
and Sex		Construction,	Sale and	Other	Hunting,
	Total	Mining &	Retail	Service	Forestry &
		Quarrying	Trade	Sectors	Fishing
COUNTRY TOTAL		Qualitying	Haue		Fishing
COUNTRY TOTAL		a= a	~~ =	40.0	0.4.0
Total	39.3	37.9	39.7	43.8	31.6
Male	41.9	42.7	42.7	45.0	32.8
Female	37.1	33.6	38.2	42.4	29.5
TIGRAY REGION					
Total	43.5	42.2	42.5	51.5	35.2
Male	46.4	46.0	43.1	59.7	37.7
Female	41.7	36.7	42.4	48.2	32.4
AFFAR REGION	71.7	30.7	72.7	40.2	J2.4
	44 5	20.4	44.0	40.0	44.4
Total	41.5	38.1	44.0	40.0	41.1
Male .	44.2	48.9	47.6	36.1	42.2
Female	39.9	32.7	42.6	41.1	37.4
AMHARA REGION					
Total	34.8	33.4	32.2	44.6	31.8
Male	37.2	38.0	32.7	43.3	34.7
Female	33.2	30.6	32.0	45.8	27.1
OROMIA REGION	00.2	00.0	02.0	10.0	
Total	39.1	39.0	38.8	43.0	29.1
Male	41.1	42.3	40.9	45.2	29.1
				-	
Female	37.3	35.6	37.6	40.4	28.9
SOMALI REGION					
Total	49.2	49.4	50.2	51.2	41.7
Male	50.4	50.1	52.2	50.3	45.5
Female	48.2	45.5	49.3	55.2	39.0
BENISHANGUL-GUMU	Z REGI	ON			
Total	30.3	27.2	29.0	40.4	27.0
Male	34.1	36.9	35.4	33.8	31.5
Female	28.4	24.3	27.0	45.4	23.6
S.N.N.P REGION	20.7	24.0	27.0	40.4	20.0
Total	38.9	36.7	39.9	44.2	20.5
					30.5
Male .	42.5	43.6	44.0	45.5	30.8
Female	35.9	31.2	37.8	42.2	30.2
GAMBELLA REGION					
Total	38.1	30.4	41.8	43.3	21.5
Male	39.4	36.9	51.3	41.3	21.3
Female	37.2	27.5	36.9	44.8	22.5
HARARI REGION					
Total	39.6	39.4	38.2	41.6	70.0
Male	41.6	41.3	39.5	40.7	95.0
Female	38.4	37.3	37.9	43.3	10.0
			31.8	43.3	10.0
ADDIS ABABA CITY AD			40.0	44.0	07.0
Total	45.4	45.6	48.8	41.9	37.9
Male	47.1	47.5	51.7	43.5	36.7
Female	43.0	42.5	46.0	38.9	39.6
DIRE DAWA ADMINIST	RATIO	V			
Total	38.9	38.7	43.4	37.5	24.8
Male	42.1	43.2	51.7	39.9	29.3
Female	36.6	34.0	41.0	34.1	20.3
- Ciriaio	55.5	51.0		<u> </u>	

CHAPTER VI

UNEMPLOYEMENT IN URBAN AREAS

6.1. Introduction

Unemployment is one of the indicators that have been used to measure the extent of unutilized human resource in the economy at a given area or region in a certain period of time. It indicates the quantity and quality of the human labour a nation or a region would have, and should be used to promote its economic development, but not actually used. The unutilized human resource as a result of unemployment has its own social and economic implication. Therefore, the availability of reliable and timely statistical data on unemployment is vital to formulate and design new strategies as well as to monitor the existing employment policies.

A wide range of information regarding unemployment and the characteristics of unemployed population were collected in the 2014 Urban Employment and Unemployment Survey. This information is useful to estimate the number of unemployed population and rate of unemployment at various reporting levels.

This chapter presents the size of unemployed population and rate of unemployment at national, regional and major urban levels. In addition, the characteristics of unemployed population such as age, sex, education, training status, marital status, previous work experience, duration of unemployment, type of jobs looking for, problems in establishing own business or enterprise, actions taken to seek work ... etc., are presented in detail. These information were collected from persons aged ten years and above by using the two known measurement approaches, namely current and usual approaches. But, the results presented in this chapter refer only the current activity status approach.

During the data collection period, respondents in the sampled household were asked detailed labour force questions using structured questionnaires. These includes:-

- a. Whether they were seeking for employment, if not reason for not seeking.
- b. Whether available for work during the coming one month, if not reason for not being available.
- c. type of job looking for /i.e. self employment or paid employment or other/
- d. type of problems faced to establish own business.
- e. Previous work experience and duration of unemployment.

Based on the responses obtained from the respondents aged ten years and above, the measurement of unemployment was done using various methods.

6.2 Measurement of Unemployment

The measurement of unemployment is based on the following three criteria that must be satisfied simultaneously: "without work", "currently available for work" 'and "seeking work" (ILO, 1983). The standard definition of unemployment is based on the "seeking work" criterion that can be interpreted as activities or efforts that non-working persons performed to look for jobs during a specified reference period or prior to it (i.e., paid or self employment). The specific steps include registration at a public or private exchange; applying to employers; checking at worksites, farms, factory gates, market or other assembly places; placing or answering news paper advertisements; seeking assistance of friends or relatives; visit vacancy boards; looking for land, building, machinery or equipment to establish own enterprise, and arranging financial resources; applying for work permits and licenses...etc.

However, the standard definition of unemployment with its emphasis on seeking work criterion might be restrictive and might not fully capture the prevailing employment situations in many developing countries including Ethiopia, where the labour market is largely unorganized or of limited scope, where labour absorption is inadequate or where the labour force is largely self-employed.

Hence, the International standards introduced provisions, which allows for the relaxation of the seeking work criterion in certain situations. The provisions are two types, namely, partial relaxation and complete relaxation. Following the recommendations of the International standard and considering the prevailing national situations, the employment and unemployment and related labour force surveys used the above provisions to measure the different forms of unemployment. The definition of the two provisions is described as follows.

Under partial relaxation, the definition of unemployment includes discouraged job seekers, future start and layoffs in addition to persons satisfying the standard definition. Discouraged job seekers are persons who want a job but did not take any active step to search for work because they believe that they cannot find one. Future starts are those persons without work who have made arrangements to take up paid employment or to undertake self-employment activity at a date subsequent to the reference period. Layoff is a person whose contract of employment or activity has been suspended by the employer for a specified or unspecified period at the end of which the person concerned has a recognized right or recognized expectation to cover employment with the employer (ILO, 1990).

The completely relaxed definition of unemployment includes persons without work and those who are available for work, including those who were or were not seeking work. That is, the seeking work criterion is completely relaxed and unemployment is based on the "without work" and "availability" criterion only. The availability in this situation is tested by asking the willingness to take up work for wage or salary in locally prevailing terms, or readiness to undertake self-employment activity; given the necessary resources and facilities. It should be noted that fulltime students are considered as available if and only if they are ready to withdraw from their studies in order to accept a job.

This survey collected unemployment data in the standard, partially relaxed and completely relaxed options of measurements. After thorough evaluation and assessment of the results obtained using the three alternative and complementary measures; the rates obtained using the completely relaxed definition was found most

plausible and hence selected for reporting. In this survey, those persons aged ten years and above who had no work or did not work at least four hours or did not have job to return to, were asked to respond whether they were available or willing to work if job was found during the coming one month. Those who responded "Yes" answer to this question were further tested whether they were ready to take a job under prevailing conditions. By doing so, those persons who had no job but available to work are considered as unemployed at the time of the survey.

6.3. Size and Rate of Urban Unemployment

The proportion of unemployed population aged ten years and above is described using unemployment rate. Unemployment rate is computed as percentage of the unemployed population over the total number of economically active population. The rate indicates that the number of persons who are ready to participate or engaged in the production of goods and services. A higher rate of unemployment implies the loss of individual contributions to the growth of Growth Domestic Product (GDP) as well as the national economy, and hence a waste of human resource.

The distribution of unemployed population aged ten years and above and unemployment rate of urban areas of the country by region, age group and sex is presented in Table 6.1. The survey result reveals that the unemployment rate at national urban level was 16.8 percent. The corresponding rates for males and females were 10.4 percent and 23.8 percent, respectively (Table 6.1). Unemployment rate in the regions illustrates that the highest rate was recorded in Addis Ababa 21.2 percent and Amhara region 19.0 percent, while the lowest rate as compared to the highest one was found in Gambella (7.9 percent) and Benishangul-Gumuz (8.3 percent) regions. But in the other regions of the country, the rate was found in the range between 12 – 18 percent (Table 6.1).

Table 6.2 shows the distribution of unemployed population of total major towns and sixteen selected major towns classified by age group and sex. Among the major towns, the highest unemployment rate above 24.8 percent was recorded in Dessie,

Adama, Bishftu, Addis Ababa towns, in that order. The lowest unemployment rate as compared to the highest one was found in Asayita town (9.2 percent).

With regard to literacy status and educational attainment of the unemployed population with age group and sex as well as the level unemployed by education for each region is shown in Table 6.3 and 6.4, while the training status and types of training of the unemployed illustrated in Table 6.5. The distribution of unemployed persons disaggregated by marital status, age group and sex is presented in Table 6.6. Table 6.7 shows similar information for each region classified by sex.

Those persons aged ten years and above who were looking for work during the three months prior to date of interview were asked to report what steps they have taken to seek work and type of work they look for during this period. Accordingly, Table 6.8 presents the distribution of unemployed persons who were and not looking for work during the three months prior to the survey date by sex and step taken to seek work for urban areas of each region. Table 6.9 also provides data on type of job they were looking for by sex and region.

The size of unemployed persons of urban areas of the country categorized by age group sex, previous work experience and duration of unemployment is presented in Table 6.10. The corresponding figure by sex and region is presented in Table 6.11. The distribution of unemployed persons according to the type of problems they faced to start own business by region and sex is shown in Table 6.12.

Table 6.1. Unemployed Population of Regions Aged Ten Years and Above by Unemployment Rate, Age Group and Sex: 2015

	Rate, Age								
Age Group	Economic	ally Active P	opulation	Unempl	oyed Popu	lation	Unempl	oyment	Rate
-9- 3.00P	Total	Male	Female	Total	Male	Female	Total	Male	Female
			COU	INTRY TOTA	NL				
All Ages	8,517,320	4,425,363	4,091,957	1,434,227	462,220	972,007	16.8	10.4	23.8
10-14	219,359	109,853	109,506	24,390	9,146	15,244	11.1	8.3	13.9
15-19	769,650	296,881	472,768	204,546	71,926	132,621	26.6	24.2	28.1
20-24	1,526,378	684,020	842,358	418,460	137,166	281,294	27.4	20.1	33.4
25-29	1,762,753	932,297	830,456	307,539	104,586	202,953	17.4	11.2	24.4
30-34	1,137,781	642,517	495,264	152,036	42,939	109,097	13.4	6.7	22.0
35-39	1,051,546	544,699	506,847	116,225	26,618	89,607	11.1	4.9	17.7
40-44	634,192	383,827	250,365	55,802	17,988	37,815	8.8	4.7	15.1
45-49	476,342	266,347	209,995	54,645	14,044	40,601	11.5	5.3	19.3
50-54	318,954	181,517	137,437	30,941	7,974	22,967	9.7	4.4	16.7
55-59	242,179	139,108	103,071	31,904	10,468	21,436	13.2	7.5	20.8
60-64	160,439	96,996	63,442	14,841	6,664	8,177	9.3	6.9	12.9
65+	217,748	147,300	70,448	22,897	12,702	10,195	10.5	8.6	14.5
			TIG	RAY REGIO	N				
All Ages	587,134	280,472	306,662	109,321	27,218	82,103	18.6	9.7	26.8
10-14	7,535	2,804	4,730	1,253	591	662	16.6	21.1	14.0
15-19	61,189	24,673	36,515	18,958	8,108	10,850	31.0	32.9	29.7
20-24	112,587	35,446	77,142	37,978	7,766	30,212	33.7	21.9	39.2
25-29	126,379	59,845	66,534	18,047	3,801	14,246	14.3	6.4	21.4
30-34	90,203	51,430	38,772	15,205	4,045	11,160	16.9	7.9	28.8
35-39	60,016	28,716	31,299	5,930	90	5,839	9.9	0.3	18.7
40-44	33,629	21,705	11,924	4,162	1,544	2,619	12.4	7.1	22.0
45-49	29,785	16,346	13,439	3,722	153	3,569	12.5	0.9	26.6
50-54	22,771	15,676	7,095	1,260	135	1,125	5.5	0.9	15.9
55-59	13,990	6,211	7,779	959	229	730	6.9	3.7	9.4
60-64	10,077	6,637	3,439	363	-	363	3.6	_	10.6
65+	18,974	10,982	7,993	1,482	756	726	7.8	6.9	9.1
			AFI	FAR REGION	N				
All Ages	151,415	82,346	69,069	19,977	6,894	13,083	13.2	8.4	18.9
10-14	4,283	2,050	2,233	224	224	-	5.2	10.9	-
15-19	11,232	4,775	6,457	1,970	1,083	887	17.5	22.7	13.7
20-24	24,827	14,517	10,310	6,704	3,230	3,475	27.0	22.2	33.7
25-29	31,837	14,580	17,257	4,748	754	3,994	14.9	5.2	23.1
30-34	19,863	11,722	8,141	1,799	10	1,790	9.1	0.1	22.0
35-39	20,207	12,029	8,177	1,402	544	858	6.9	4.5	10.5
40-44	11,674	6,660	5,014	1,078	370	708	9.2	5.6	14.1
45-49	10,020	5,086	4,934	725	377	347	7.2	7.4	7.0
50-54	6,571	4,338	2,233	256	-	256	3.9	-	11.5
55-59	4,575	1,753	2,822	806	39	768	17.6	2.2	27.2
60-64	3,821	2,831	990	110	110	-	2.9	3.9	_,
65+	2,506	2,004	502	154	154	-	6.1	7.7	-

Table 6.1. Cont'd

	Economi	cally Active F	opulation	Unem	ployed Popu	ulation	Unemployment Rate		
Age Group	Total	Male	Female	Tota	l Male	Female	Total	Male	Female
			AMH	IARA REGI	ON				
All Ages	1,788,246	895,283	892,963	339,154	100,100	239,055	19.0	11.2	26.8
10-14	46,262	23,207	23,055	1,674	431	1,243	3.6	1.9	5.4
15-19	152,567	56,747	95,820	46,092	14,065	32,027	30.2	24.8	33.4
20-24	342,654	141,773	200,881	109,704	31,790	77,914	32.0	22.4	38.8
25-29	370,018	198,954	171,064	71,007	22,335	48,672	19.2	11.2	28.5
30-34	226,062	126,119	99,943	35,281	9,702	25,579	15.6	7.7	25.6
35-39	207,355	103,841	103,514	25,317	6,353	18,963	12.2	6.1	18.3
40-44	120,913	72,528	48,385	12,778	3,272	9,506	10.6	4.5	19.6
45-49	97,318	48,489	48,828	10,106	2,193	7,913	10.4	4.5	16.2
50-54	74,320	32,840	41,480	12,287	2,671	9,616	16.5	8.1	23.2
55-59	57,490	32,740	24,750	6,735	1,628	5,108	11.7	5.0	20.6
60-64	38,235	22,604	15,631	3,993	1,675	2,318	10.4	7.4	14.8
65+	55,052	35,441	19,611	4,180	3,984	196	7.6	11.2	1.0
			ORG	OMIA REGIO	ON				
All Ages	2,489,389	1,341,888	1,147,501	386,972	128,166	258,806	15.5	9.6	22.6
10-14	75,552	44,773	30,778	7,817	2,873	4,944	10.3	6.4	16.1
15-19	250,525	106,242	144,283	64,916	23,927	40,990	25.9	22.5	28.4
20-24	460,295	220,141	240,154	114,084	36,756	77,328	24.8	16.7	32.2
25-29	522,240	287,404	234,837	87,835	29,632	58,202	16.8	10.3	24.8
30-34	314,812	183,894	130,918	34,057	9,880	24,177	10.8	5.4	18.5
35-39	303,041	157,128	145,913	30,808	8,129	22,679	10.2	5.2	15.5
40-44	184,229	117,938	66,291	12,518	5,956	6,562	6.8	5.1	9.9
45-49	135,063	76,086	58,977	14,673	3,167	11,507	10.9	4.2	19.5
50-54	77,332	47,348	29,984	4,851	1,690	3,160	6.3	3.6	10.5
55-59	62,598	36,510	26,088	5,137	1,700	3,437	8.2	4.7	13.2
60-64	48,470	28,927	19,544	4,162	1,749	2,413	8.6	6.0	12.3
65+	55,232	35,497	19,734	6,116	2,707	3,409	11.1	7.6	17.3
			SO	MALI REGIO	ON				
All Ages	139,928	75,181	64,747	20,846	8,066	12,780	14.9	10.7	19.7
10-14	7,278	2,923	4,356	1,043	227	816	14.3	7.8	18.7
15-19	12,343	4,600	7,743	1,838	650	1,188	14.9	14.1	15.3
20-24	19,289	9,274	10,015	5,952	2,415	3,538	30.9	26.0	35.3
25-29	24,742	14,094	10,649	3,479	836	2,643	14.1	5.9	24.8
30-34	17,905	10,585	7,320	2,731	750	1,981	15.3	7.1	27.1
35-39	17,619	9,072	8,547	1,830	664	1,165	10.4	7.3	13.6
40-44	15,523	8,410	7,113	1,451	492	959	9.3	5.9	13.5
45-49	10,662	6,582	4,080	793	742	51	7.4	11.3	1.2
50-54	6,288	4,200	2,088	968	631	336	15.4	15.0	16.1
55-59	1,556	1,198	359	226	174	52	14.5	14.5	14.5
60-64	3,301	2,065	1,236	210	210	-	6.3	10.1	-
65+	3,420	2,179	1,242	326	276	51	9.5	12.6	4.1

Table 6.1. Cont'd

Age Group	Economic	cally Active P	opulation	Unemp	loyed Popu	lation	Unemplo	oyment	Rate
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
			BENISHAN	GUL-GUMUZ	REGION				
All Ages	116,784	61,559	55,224	9,731	2,830	6,901	8.3	4.6	12.5
10-14	7,430	2,802	4,628	43	43	-	0.6	1.5	-
15-19	10,982	5,236	5,745	1,012	348	664	9.2	6.6	11.6
20-24	22,861	9,971	12,890	4,000	1,208	2,792	17.5	12.1	21.7
25-29	25,440	14,760	10,680	1,450	394	1,056	5.7	2.7	9.9
30-34	13,941	7,513	6,428	876	135	741	6.3	1.8	11.5
35-39	12,218	6,825	5,393	532	-	532	4.4	-	9.9
40-44	7,235	3,468	3,767	605	77	529	8.4	2.2	14.0
45-49	6,340	4,169	2,171	253	119	134	4.0	2.9	6.2
50-54	4,319	2,842	1,477	215	172	43	5.0	6.0	2.9
55-59	2,643	1,804	839	349	172	177	13.2	9.5	21.1
60-64	1,140	759	381	163	163	-	14.3	21.4	-
65+	2,234	1,408	826	233	-	233	10.4	-	28.2
			S.N	.N.P REGIO	N				
All Ages	1,239,834	649,662	590,172	142,947	46,009	96,939	11.5	7.1	16.4
10-14	44,912	19,802	25,110	5,106	1,174	3,932	11.4	5.9	15.7
15-19	135,714	55,436	80,278	27,595	10,016	17,579	20.3	18.1	21.9
20-24	216,166	104,712	111,454	36,141	12,436	23,705	16.7	11.9	21.3
25-29	255,250	134,744	120,506	32,126	11,276	20,850	12.6	8.4	17.3
30-34	139,813	78,742	61,071	12,676	3,612	9,064	9.1	4.6	14.8
35-39	150,232	74,343	75,889	8,956	1,310	7,646	6.0	1.8	10.1
40-44	100,167	60,560	39,607	4,809	1,229	3,581	4.8	2.0	9.0
45-49	68,948	41,526	27,423	4,554	661	3,894	6.6	1.6	14.2
50-54	46,438	27,599	18,840	4,162	1,258	2,904	9.0	4.6	15.4
55-59	30,991	20,932	10,059	2,497	1,460	1,037	8.1	7.0	10.3
60-64	20,293	8,027	12,266	2,052	891	1,161	10.1	11.1	9.5
65+	30,908	23,240	7,668	2,272	687	1,586	7.4	3.0	20.7
			GAME	BELLA REG	ION				
All Ages	66,910	33,282	33,628	5,101	1,511	3,590	7.6	4.5	10.7
10-14	4,965	2,081	2,884	121	40	81	2.4	1.9	2.8
15-19	7,488	3,413	4,075	713	162	551	9.5	4.7	13.5
20-24	14,037	5,900	8,137	1,921	536	1,384	13.7	9.1	17.0
25-29	14,802	7,487	7,315	1,123	359	764	7.6	4.8	10.4
30-34	9,384	4,940	4,444	455	-	455	4.8	-	10.2
35-39	6,190	3,880	2,310	134	41	93	2.2	1.0	4.0
40-44	3,663	2,069	1,594	174	93	81	4.7	4.5	5.1
45-49	3,016	1,323	1,693	364	204	159	12.1	15.4	9.4
50-54	1,995	1,180	816	-	-	-	-	-	-
55-59	703	657	46	75	75	-	10.7	11.4	-
60-64	341	177	165	-	-	-	-	-	-
65+	326	176	150	22	-	22	6.6	-	14.4

Table 6.1. Cont'd

Age Group	Economic	cally Active P	opulation	Unem	oloyed Popu	ulation	Unempl	oyment	Rate
Age Gloup	Total	Male	Female	Tota	l Male	Female	Total	Male	Female
			HAI	RARI REGIO	N				
All Ages	67,319	32,616	34,703	8,209	2,716	5,493	12.2	8.3	15.8
10-14	663	229	434	97	49	48	14.6	21.3	11.0
15-19	2,941	1,046	1,894	759	317	442	25.8	30.3	23.4
20-24	8,446	3,368	5,078	1,611	528	1,083	19.1	15.7	21.3
25-29	12,798	6,708	6,090	1,863	376	1,487	14.6	5.6	24.4
30-34	9,256	4,691	4,565	697	229	468	7.5	4.9	10.2
35-39	10,459	4,015	6,444	1,292	454	839	12.4	11.3	13.0
40-44	6,995	3,861	3,134	586	262	324	8.4	6.8	10.3
45-49	5,620	3,201	2,420	447	132	315	8.0	4.1	13.0
50-54	4,294	2,326	1,968	237	99	138	5.5	4.2	7.0
55-59	2,163	1,257	906	365	186	179	16.9	14.8	19.8
60-64	1,888	1,138	751	41	-	41	2.1	-	5.4
65+	1,795	777	1,019	215	86	129	12.0	11.0	12.7
		AD	DIS ABABA	CITY ADM	NISTRATIO	ON			
All Ages	1,728,296	902,490	825,806	366,227	130,123	236,103	21.2	14.4	28.6
10-14	17,786	7,733	10,053	6,677	3,339	3,338	37.5	43.2	33.2
15-19	111,430	30,002	81,428	37,991	12,287	25,704	34.1	41.0	31.6
20-24	283,945	128,729	155,216	94,289	38,828	55,460	33.2	30.2	35.7
25-29	353,807	181,270	172,537	81,166	33,267	47,900	22.9	18.4	27.8
30-34	280,552	153,982	126,570	45,908	13,786	32,122	16.4	9.0	25.4
35-39	246,248	136,113	110,135	36,758	8,211	28,547	14.9	6.0	25.9
40-44	138,833	80,932	57,901	16,532	4,269	12,263	11.9	5.3	21.2
45-49	94,830	55,954	38,876	17,061	5,629	11,432	18.0	10.1	29.4
50-54	68,221	40,358	27,863	5,526	974	4,552	8.1	2.4	16.3
55-59	59,178	32,299	26,878	13,729	4,338	9,391	23.2	13.4	34.9
60-64	29,680	21,829	7,851	3,394	1,691	1,704	11.4	7.7	21.7
65+	43,785	33,288	10,497	7,195	3,504	3,691	16.4	10.5	35.2
			DIRE DAW	'A ADMINIS	TRATION				
All Ages	142,066	70,585	71,482	25,742	8,587	17,154	18.1	12.2	24.0
10-14	2,693	1,448	1,245	334	153	181	12.4	10.6	14.5
15-19	13,240	4,712	8,528	2,702	965	1,738	20.4	20.5	20.4
20-24	21,270	10,188	11,081	6,077	1,673	4,404	28.6	16.4	39.7
25-29	25,438	12,451	12,987	4,695	1,555	3,140	18.5	12.5	24.2
30-34	15,990	8,899	7,091	2,352	790	1,562	14.7	8.9	22.0
35-39	17,962	8,736	9,225	3,266	822	2,444	18.2	9.4	26.5
40-44	11,331	5,696	5,636	1,108	424	684	9.8	7.4	12.1
45-49	14,740	7,586	7,155	1,947	668	1,279	13.2	8.8	17.9
50-54	6,404	2,811	3,594	1,179	344	835	18.4	12.3	23.2
55-59	6,292	3,748	2,544	1,026	468	558	16.3	12.5	21.9
60-64	3,193	2,002	1,190	353	177	177	11.1	8.8	14.9
65+	3,515	2,308	1,207	702	549	153	20.0	23.8	12.7

Table 6.2 Unemployed Population of Major Towns Aged Ten Years and Above by Unemployment Rate, Age Group and Sex: 2015

		_		Lucad Davidation							
	Economic	cally Active P	opulation	Unemp	oloyed Pop	ulation	Unemp	oloyment	Rate		
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female		
			MAJO	R TOWNS	TOTAL						
All Ages	3,162,287	1,645,930	1,516,358	602,001	202,278	399,723	19.0	12.3	26.4		
10-14	64,450	31,224	33,226	10,167	4,580	5,586	15.8	14.7	16.8		
15-19	243,232	80,385	162,847	65,208	21,286	43,923	26.8	26.5	27.0		
20-24	532,299	236,841	295,459	158,838	57,368	101,470	29.8	24.2	34.3		
25-29	656,234	338,251	317,983	137,617	49,464	88,154	21.0	14.6	27.7		
30-34	472,739	262,969	209,770	71,866	21,028	50,839	15.2	8.0	24.2		
35-39	415,504	227,529	187,975	58,226	13,777	44,448	14.0	6.1	23.6		
40-44	242,721	142,180	100,541	27,187	7,929	19,258	11.2	5.6	19.2		
45-49	179,210	104,708	74,501	25,868	8,269	17,599	14.4	7.9	23.6		
50-54	123,094	70,591	52,502	11,931	3,076	8,855	9.7	4.4	16.9		
55-59	101,058	57,771	43,286	18,770	6,379	12,391	18.6	11.0	28.6		
60-64	56,456	39,226	17,230	5,433	2,659	2,774	9.6	6.8	16.1		
65+	75,290	54,254	21,036	10,891	6,463	4,428	14.5	11.9	21.0		
	MEKELE TOWN										
All Ages	156,214	77,892	78,323	24,574	7,043	17,531	15.7	9.0	22.4		
10-14	3,225	1,430	1,795	391	-	391	12.1	-	21.8		
15-19	13,523	4,669	8,854	1,871	788	1,083	13.8	16.9	12.2		
20-24	27,729	10,114	17,615	7,419	2,138	5,281	26.8	21.1	30.0		
25-29	36,514	16,758	19,756	7,040	2,255	4,785	19.3	13.5	24.2		
30-34	26,625	14,932	11,693	4,085	896	3,189	15.3	6.0	27.3		
35-39	17,318	10,235	7,083	1,636	90	1,545	9.4	0.9	21.8		
40-44	9,948	6,135	3,812	837	481	356	8.4	7.8	9.3		
45-49	6,360	3,986	2,374	492	-	492	7.7	-	20.7		
50-54	4,868	3,074	1,793	362	135	227	7.4	4.4	12.7		
55-59	4,095	2,459	1,636	90	-	90	2.2	-	5.5		
60-64	2,764	1,969	795	-	-	-	-	-	-		
65+	3,246	2,130	1,116	349	259	90	10.7	12.1	8.1		
			AS	AYITA TO	WN						
All Ages	14,118	7,769	6,350	1,299	412	887	9.2	5.3	14.0		
10-14	119	34	85	-	-	-	-	-	-		
15-19	929	335	595	211	94	117	22.7				
20-24	2,288	1,126	1,162	333	47	285	14.5	4.2			
25-29	3,328	1,749	1,579	358	92	266	10.8	5.2	16.9		
30-34	1,661	973	688	42	10	32	2.5	1.0	4.7		
35-39	1,778	1,144	633	82	-	82	4.6	-	13.0		
40-44	1,253	699	555	94	40	54	7.5	5.7	9.7		
45-49	1,001	611	390	64	48	16	6.4	7.8	4.1		
50-54	528	270	258	34	-	34	6.4	-	13.2		
55-59	440	242	198	39	39	-	8.8	16.0	-		
60-64	339 456	212 375	127	-	- 11	-	9.6	- 11 G	-		
65+	456	375	80	44	44	-	9.6	11.6	-		

Table 6.2 Cont'd

Table 6.2 C								Unemployment Rate		
Age Group	Economica	Ily Active	Population	Unemp	oloyed Popu	lation	Unemp	oloyment F	Rate	
go 3.00p	Total	Male	Female	Total	Male	Female	Total	Male	Female	
				GONDER T	OWN					
All Ages	182,670	91,7			6,108	14,158	11.1	6.7	15.6	
10-14	23,059	11,9	63 11,09	96 102	-	102	0.4	-	0.9	
15-19	23,403	10,2	09 13,1	94 3,040	815	2,225	13.0	8.0	16.9	
20-24	27,908	13,78		22 5,299	2,026	3,273	19.0	14.7	23.2	
25-29	29,876	14,4	30 15,4	46 5,449	1,450	3,999	18.2	10.0	25.9	
30-34	19,617	10,3	24 9,2	93 1,915	208	1,707	9.8	2.0	18.4	
35-39	16,526	6,2			151	1,449	9.7	2.4	14.1	
40-44	11,695	7,6	31 4,0	54 590	386	204	5.0	5.1	5.0	
45-49	9,712	5,7	35 3,9	78 760	253	507	7.8	4.4	12.8	
50-54	5,196	2,19	91 3,0	590	205	385	11.4	9.4	12.8	
55-59	5,725	5,725 2,998 2,		27 307	102	205	5.4	3.4	7.5	
60-64	4,191	2,68	39 1,5	02 411	309	102	9.8	11.5	6.8	
65+	5,763	3,4	99 2,2	64 203	203	-	3.5	5.8	-	
				DESSIE TO	NWC					
All Ages	95,083	50,3	96 44,6	37 23,553	7,235	16,318	24.8	14.4	36.5	
10-14	1,167	28	32 8	35 520	158	362	44.6	56.1	40.9	
15-19	7,296	3,2	60 4,0	37 2,450	1,327	1,123	33.6	40.7	27.8	
20-24	16,404	7,6	49 8,7	55 5,982	1,623	4,358	36.5	21.2	49.8	
25-29	19,122	10,2			1,063	3,904	26.0	10.3	44.1	
30-34	12,085	7,10			545	1,559	17.4		31.3	
35-39	11,759	6,19			834	1,878	23.1	13.5	33.7	
40-44	7,963	3,7			658	1,602	28.4	17.6	37.8	
45-49	5,351	3,4			219	305	9.8		16.3	
50-54	5,121	2,6			184	644	16.2		25.9	
55-59	3,078	1,8			395	416	26.3		33.1	
60-64	2,625	1,70		61 124	61	63	4.7		7.3	
65+	3,111	2,2		08 272	167	105	8.7		11.6	
				BAHIR DAR	TOWN					
All Ages	146,126	73,9	47 72,1°	79 21,816	7,557	14,259	14.9	10.2	19.8	
10-14	7,520	5,2			-	185	2.5		8.0	
15-19	16,625	7,5			1,258	1,126	14.3	16.7	12.4	
20-24	29,080	9,7					26.1	21.8	28.4	
25-29	31,066	16,40				3,279	17.4		22.4	
30-34	19,186	9,9				1,463	12.2		15.9	
35-39	14,783	8,4			277	680	6.5		10.8	
40-44	7,799	4,4			160	663	10.6		19.5	
45-49	7,163	4,7			67	252	4.5		10.7	
50-54	3,657	1,3			-	469	12.8		20.4	
55-59	4,103	2,0			93	324	10.2		15.8	
60-64	1,850	1,2		00 254	-	254	13.7		42.2	
65+	3,295	2,6		00 673	582	90	20.4		15.1	

Table 6.2 Cont'd

Age Group	Economica	lly Active	Population	Unemp	loyed Popu	lation	Unemp	oloyment F	Rate
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
				BISHOFTU	TOWN				
All Ages	75,309	38,30	06 37,00	03 17,315	4,574	12,741	23.0	11.9	34.4
10-14	251	19	94 !	57 80	80	-	31.9	41.4	-
15-19	4,248	1,29			728	1,338	48.6		45.3
20-24	13,786	5,93			968	2,704	26.6	16.3	34.4
25-29	17,230	8,74			740	3,514	24.7		41.4
30-34	12,551	6,3			527	1,939	19.6		31.1
35-39	9,686	5,6	54 4,03	32 1,860	621	1,239	19.2		30.7
40-44	5,294	2,75			84	782	16.4		30.7
45-49	4,268	2,3			84	374	10.7		19.5
50-54	2,862	1,81			343	316	23.0		30.1
55-59	2,389	1,3			171	439	25.6		41.1
60-64	1,229		75 25		56	99	12.6		38.8
65+	1,515	95	57 5	58 171	171	-	11.3	17.9	-
				ADAMA T	OWN				
All Ages	154,552	81,66			11,707	25,017	23.8		34.3
10-14	1,578	9	12 66		477	122	37.9	52.2	18.3
15-19	12,778	3,52	27 9,2	51 4,844	400	4,444	37.9	11.4	48.0
20-24	28,739	12,78	32 15,9	57 9,280	3,184	6,096	32.3	24.9	38.2
25-29	36,138	19,42			3,410	5,660	25.1	17.6	33.9
30-34	18,690	10,93	38 7,7	52 3,443	971	2,472	18.4	8.9	31.9
35-39	19,966	11,84	41 8,12	25 4,057	1,320	2,737	20.3	11.1	33.7
40-44	10,797	6,50	06 4,29	91 1,372	364	1,007	12.7	5.6	23.5
45-49	8,202	4,23	32 3,97	70 1,866	493	1,373	22.7	11.6	34.6
50-54	6,533	3,94	45 2,58	38 923	470	452	14.1	11.9	17.5
55-59	4,836	3,28	38 1,54	18 660	248	412	13.6	7.5	26.6
60-64	3,460	2,26	31 1,20	00 242	-	242	7.0	-	20.2
65+	2,835	1,99	99 83	370	370	-	13.0	18.5	-
				JIMMA TO	NWN				
All Ages	88,590	47,24	46 41,34	14 13,301	4,385	8,916	15.0	9.3	21.6
10-14	2,345	52	29 1,8	16 216	93	123	9.2	17.5	6.8
15-19	9,948	3,77			930	1,233	21.7		20.0
20-24	16,916	8,30			1,223	3,112	25.6		36.1
25-29	19,810	11,49			878	2,160	15.3		26.0
30-34	11,010	6,89	95 4,1°		509	873	12.6	7.4	21.2
35-39	8,679	4,03			274	513	9.1	6.8	11.0
40-44	5,974	3,29			60	141	3.4		5.3
45-49	5,011	3,12			121	335	9.1		17.8
50-54	3,440	2,24			93	121	6.2		10.1
55-59	2,603	1,78		22 181	-	181	7.0		22.1
60-64	1,354		37 66		203	60	19.5		9.0
65+	1,502	1,09		06 62	-	62	4.2		15.4

Table 6.2 Cont'd

Age Group	Economica	Ily Active	Population	Unemp	loyed Popu	lation	Unemp	oloyment F	Rate
Age Group	Total	Male Female		Total	Male	Female	Total	Male	Female
			S	HASHEMEN	E TOWN				
All Ages	63,556	34,32	28 29,22	28 9,182	2,157	7,025	14.4	6.3	24.0
10-14	778	32		57 116	-	116	15.0	-	25.4
15-19	5,005	2,15		1,063	426	637	21.2	19.8	22.3
20-24	9,908	4,12	26 5,78	32 2,554	418	2,136	25.8	10.1	36.9
25-29	16,004	8,35	51 7,6	53 2,257	366	1,891	14.1	4.4	24.7
30-34	9,896	5,84	19 4,04	1,148	277	872	11.6	4.7	21.5
35-39	9,439	6,07	76 3,36	889	183	706	9.4	3.0	21.0
40-44	4,650	3,15	51 1,50	00 142	108	34	3.0	3.4	2.2
45-49	2,847	1,47	78 1,37	70 350	141	209	12.3	9.6	15.2
50-54	1,688	70		310	_	310	18.3	-	31.5
55-59	1,419	1,01			34	48	5.8		12.0
60-64	812	55		34	-	34	4.2		12.9
65+	1,110	55		30 238	204	34	21.4		6.0
				JIJIGA TO	WN				
All Ages	49,033	29,5	15 19,5°	18 8,218	3,568	4,650	16.8	12.1	23.8
10-14	866	13			-	443	51.1	_	60.2
15-19	4,208	1,84			305	557	20.5	16.5	23.6
20-24	7,704	4,16	•		1,056	1,279	30.3		36.2
25-29	9,486	5,86			472	973	15.2		26.9
30-34	6,669	4,56			582	415	15.0		19.7
35-39	6,309	3,38			308	365	10.7		12.5
40-44	5,505	3,5			304	351	11.9		17.7
45-49	3,274	2,55			274	51	9.9		7.0
50-54					187	112	11.4		15.6
	2,623 599	1,90 35			40	52	15.4		21.3
55-59 60.64	936	65 65		32 -	40	52			21.3
60-64 65+	851	58			40	51	- 10.7		18.9
				ASOSA TO	OWN				
All Ages	24,199	13,79	98 10,40	3,087	1,012	2,075	12.8	7.3	20.0
10-14	234	14		91 43	43	_,0.0	18.2		
15-19	1,762	85)4 457	117	340	26.0		37.7
20-24	5,044	2,09			273	647	18.2		21.9
25-29	6,600	3,96			317	431	11.3		16.3
30-34	3,188	2,16			135	278	13.0		27.1
35-39	2,739	1,80		35 157	-	157	5.7		16.8
40-44	1,684	83		15 121	_	121	7.2		14.4
45-49	1,004	81)5 99	42	57	8.1		14.1
50-54	832	59		12 85	42	43	10.3		17.8
55-59	422	26		62 43	43	43	10.3		17.0
					43	-			-
60-64 65+	201 271	17 10		<u>29</u> - 71 -	-	-	-	-	-
UUT	211	10	, I	-	-	-	-	-	-

Table 6.2 Cont'd

Age Group	Economica	lly Active	Population	Unemp	loyed Popu	lation	Unemp	loyment F	Rate
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
				HAWASSA	TOWN				
All Ages	143,138	77,15	51 65,98	37 18,436	3,887	14,549	12.9	5.0	22.0
10-14	1,643	46	50 1,18	33 242	148	95	14.8	32.1	8.0
15-19	12,779	3,88	82 8,89	7 1,633	367	1,266	12.8	9.5	14.2
20-24	26,001	11,73	37 14,26	5,526	722	4,805	21.3	6.1	33.7
25-29	31,925	16,74	l9 15,17	76 5,130	838	4,292	16.1	5.0	28.3
30-34	20,796	12,83	39 7,9	56 2,202	682	1,520	10.6	5.3	19.1
35-39	17,959	10,98	35 6,9 ⁷	74 1,418	191	1,226	7.9	1.7	17.6
40-44	11,085	7,79	92 3,29	93 879	287	592	7.9	3.7	18.0
45-49	8,032	3,92	26 4,10	06 501	59	441	6.2	1.5	10.8
50-54	5,919	3,82	20 2,10	00 217	-	217	3.7	-	10.3
55-59	3,200	2,3	57 84	13 242	148	95	7.6	6.3	11.2
60-64	1,687	99	93 69	94 162	162	-	9.6	16.3	-
65+	2,112	1,6	2 50	00 284	284	-	13.5	17.6	-
				GAMBELLA	TOWN				
All Ages	32,018	16,50)2 15,5°	15 4,052	1,207	2,846	12.7	7.3	18.3
10-14	522	20)4 3	17 121	40	81	23.3	19.8	25.5
15-19	3,119	1,26	67 1,8	52 713	162	551	22.8	12.8	29.7
20-24	7,131	2,96	65 4,16	66 1,607	536	1,071	22.5	18.1	25.7
25-29	7,092	3,6	5 3,47	77 746	272	474	10.5	7.5	13.6
30-34	4,969	2,54	13 2,42	26 367	-	367	7.4	-	15.1
35-39	3,896	2,5	54 1,34	12 81	41	40	2.1	1.6	3.0
40-44	1,916	1,24	10 67	76 121	40	81	6.3	3.2	11.9
45-49	1,575	88	91 68	35 200	40	159	12.7	4.5	23.3
50-54	907	54	11 36	- 66	-	-	-	-	-
55-59	517	5		- 75	75	-	14.5	14.5	-
60-64	246			65 -	-	-	-	-	-
65+	128	8	34 4	14 22	-	22	16.9	-	49.2
				HARAR TO	OWN				
All Ages	67,319	32,6	6 34,70	3 8,209	2,716	5,493	12.2	8.3	15.8
10-14	663	22		97	49	48	14.6	21.3	11.0
15-19	2,941	1,04			317	442	25.8	30.3	23.4
20-24	8,446	3,36	5,0	78 1,611	528	1,083	19.1	15.7	21.3
25-29	12,798	6,70	08 6,09	90 1,863	376	1,487	14.6	5.6	24.4
30-34	9,256	4,69			229	468	7.5	4.9	10.2
35-39	10,459	4,01			454	839	12.4	11.3	13.0
40-44	6,995	3,86			262	324	8.4	6.8	10.3
45-49	5,620	3,20			132	315	8.0		13.0
50-54	4,294	2,32			99	138	5.5		7.0
55-59	2,163	1,2		06 365	186	179	16.9	14.8	19.8
60-64	1,888	1,13		51 41	-	41	2.1	-	5.4
65+	1,795	77	77 1,0	19 215	86	129	12.0	11.0	12.7

Table 6.2 Cont'd

Ago Croup	Economica	lly Active	Population	Unemp	oloyed Popu	lation	Unemp	loyment F	Rate
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
				ADDIS ABAE	BA CITY				
All Ages	1,728,296	902,49	90 825,80	06 366,227	130,123	236,103	21.2	14.4	28.6
10-14	17,786	7,73	33 10,0	6,677	3,339	3,338	37.5	43.2	33.2
15-19	111,430	30,00	02 81,42	28 37,991	12,287	25,704	34.1	41.0	31.6
20-24	283,945	128,72	29 155,2°	16 94,289	38,828	55,460	33.2	30.2	35.7
25-29	353,807	181,27	70 172,5	81,166	33,267	47,900	22.9	18.4	27.8
30-34	280,552	153,98	32 126,5°	70 45,908	13,786	32,122	16.4	9.0	25.4
35-39	246,248	136,11	13 110,1	36,758	8,211	28,547	14.9	6.0	25.9
40-44	138,833	80,93	32 57,90	16,532	4,269	12,263	11.9	5.3	21.2
45-49	94,830	55,95	54 38,8°	76 17,061	5,629	11,432	18.0	10.1	29.4
50-54	68,221	40,35	58 27,80	5,526	974	4,552	8.1	2.4	16.3
55-59	59,178	32,29	99 26,8	78 13,729	4,338	9,391	23.2	13.4	34.9
60-64	29,680	21,82	29 7,8	51 3,394	1,691	1,704	11.4	7.7	21.7
65+	43,785	33,28	38 10,49	7,195	3,504	3,691	16.4	10.5	35.2
				DIRE DAWA	TOWN				
All Ages	142,066	70,58	35 71,48	32 25,742	8,587	17,154	18.1	12.2	24.0
10-14	2,693	1,44	1,24	15 334	153	181	12.4	10.6	14.5
15-19	13,240	4,71	12 8,5	28 2,702	965	1,738	20.4	20.5	20.4
20-24	21,270	10,18	38 11,0	31 6,077	1,673	4,404	28.6	16.4	39.7
25-29	25,438	12,45	51 12,98	37 4,695	1,555	3,140	18.5	12.5	24.2
30-34	15,990	8,89	99 7,09	91 2,352	790	1,562	14.7	8.9	22.0
35-39	17,962	8,73	36 9,22	25 3,266	822	2,444	18.2	9.4	26.5
40-44	11,331	5,69	96 5,63	36 1,108	424	684	9.8	7.4	12.1
45-49	14,740	7,58	36 7,1	55 1,947	668	1,279	13.2	8.8	17.9
50-54	6,404	2,81	11 3,59	94 1,179	344	835	18.4	12.3	23.2
55-59	6,292	3,74	18 2,5	1,026	468	558	16.3	12.5	21.9
60-64	3,193	2,00			177	177	11.1	8.8	14.9
65+	3,515	2,30	08 1,20	702	549	153	20.0	23.8	12.7

Table 6.3 Unemployed Populations of Urban Areas Aged Ten Years and Above by Literacy Status, Educational Attainment, Age Group and Sex - COUNTRY TOTAL: 2015

	and Sex			AL. 2013			All Litera	ıto.					Illiterate
	Total						All Litera		<u> </u>	Dinlomal			IIIIlerale
Age Group	Unemployed				Casandani			10+2/		Diploma/		Damas	
and Sex	Persons	Nam		Cradas	Secondary	Casandani		Certificate	Contidionto	Degree	Dinlama	Degree	Navion
	1 0130113	Non-	Dl	Grades	not	Secondary	D	Not	Certificate	Not	Diploma	and above	Never
		Formal	Preschool	1-8	completed	Completed	Preparatory	Completed	Completed	Completed	Completed	Completed	Attended
All Ages													
Total	1,434,227	29,986	1,928	493,928	143,919	294,207	49,001	6,047	27,462	41,857	121,121	46,319	178,451
Male	462,220	8,274	800	146,293	52,078	107,715	22,368	2,783	9,160	19,761	42,502	24,747	25,739
Female	972,007		1,127	347,635	91,841	186,492	26,633	3,264	18,303	22,096	78,619	21,572	•
10-14													
Total	24,390	160	-	18,389	993	391	-	-	-	-	-	-	4,456
Male	9,146	-	-	8,128	-	-	-	-	-	-	-	-	1,017
Female	15,244	160	-	10,260	993	391	-	-	-	-	-	-	3,439
15-19													
Total	204,546	683	-	80,082	17,968	65,399	7,869	2,308	4,942	6,539	7,329	-	11,428
Male	71,926	592	-	26,101	7,382	24,246	3,324	1,112	2,050	2,341	2,340	-	2,438
Female	132,621	91	-	53,982	10,586	41,153	4,545	1,195	2,892	4,198	4,989	-	8,990
20-24													
Total	418,460	644	-	121,460	50,669	90,806	20,909	1,753	12,650	20,616	54,540	20,164	24,249
Male	137,166	-	-	31,066	18,964	32,766	10,030	818	4,366	9,607	18,486	8,767	2,296
Female	281,294	644	-	90,394	31,705	58,040	10,878	935	8,284	11,010	36,053	11,398	21,954
25-29													
Total	307,539	3,118	-	94,047	30,871	64,348	12,187	1,542	5,804	10,655	33,711	16,719	34,538
Male	104,586	1,783	-	23,952	13,201	24,605	6,081	812	1,675	6,287	12,607	9,886	3,697
Female	202,953	1,334	-	70,095	17,671	39,743	6,107	730	4,129	4,368	21,104	6,833	30,841
30-34													
Total	152,036	2,279	987	59,980	13,429	28,841	4,600	344	1,931	1,042	9,113	3,054	26,437
Male	42,939	759	541	17,698	3,685	9,365	1,443	-	241	208	3,226	3,054	2,721
Female	109,097	1,521	445	42,282	9,745	19,477	3,156	344	1,690	834	5,887	-	23,716
35-39													
Total	116,225	4,036	156	42,957	15,603	17,865	2,497	-	177	1,544	4,453	1,954	24,984
Male	26,618	283		10,702	2,819	4,788	1,326	-	-	883	1,469	962	3,385
Female	89,607	3,753	156	32,255	12,784	13,077	1,171	-	177	660	2,984	992	21,599

234

Table 6.3 Cont'd

Table 0.5							All Litera	ite					Illiterate
Age Group								10+2/		Diploma/			
and Sex	Total				Secondary			Certificate		Degree		Degree	
and Scx	Unemployed	Non-		Grades	not	Secondary		Not	Certificate	Not	Diploma	and above	Never
	Persons	Formal	Preschool	1-8	completed	Completed	Preparatory	Completed	Completed	Completed	Completed	Completed	Attended
40-44													
Total	55,802	2,718	74	25,528	3,880	9,767	346	40	1,175	346	2,547	1,257	8,124
Male	17,988	114	74	9,022	2,484	3,067	-	40	465	-	789	82	1,849
Female	37,815	2,603	-	16,506	1,396	6,700	346	-	710	346	1,758	1,175	6,275
45-49													
Total	54,645	3,774	185	19,520	5,155	6,923	553	61	359	1,022	3,058	182	13,853
Male	14,044	839	185	4,881	2,124	2,330	164	-	-	342	1,200	126	1,853
Female	40,601	2,936	-	14,638	3,031	4,592	389	61	359	680	1,858	56	12,000
50-54													
Total	30,941	3,888	-	11,408	1,713	2,349	-	-	61	-	448	374	10,699
Male	7,974	419	-	3,933	235	912	-	-	-	-	114	374	1,987
Female	22,967	3,469	-	7,475	1,478	1,438	-	-	61	-	334	-	8,712
55-59													
Total	31,904	2,964	-	10,758	1,686	3,684	41	-	-	93	3,278	895	8,503
Male	10,468	678	-	5,197	546	2,415	-	-	-	93	581	112	845
Female	21,436	2,287	-	5,561	1,140	1,269	41	-	-	-	2,697	783	7,658
60-64													
Total	14,841	1,583	525	4,502	456	1,722	-	-	364	-	490	860	4,340
Male	6,664	757	-	1,870	-	1,108	-	-	364	-	490	860	1,216
Female	8,177	826	525	2,632	456	614	-	-	-	-	-	-	3,124
65+													
Total	22,897	4,140	-	5,296	1,495	2,112	-	-	-	-	2,155	860	6,839
Male	12,702	2,051	-	3,740	639	2,112	-	-	-	-	1,201	524	2,434
Female	10,195	2,089	-	1,556	855	-	-	-	-	-	954	336	4,405

235

Table 6.4 Unemployed Population of Urban Areas Aged Ten Years and Above by Educational Attainment, Region and Sex: 2015

	-						All Litera	•			<u> </u>		Illiterate
Region and Sex	Total Unemployed Persons	Non- Formal	Preschool	Grades 1-8	Secondary not completed	Secondary Completed	Preparatory	10+2/ Certificate Not Completed	Certificate Completed	Diploma or Degree Not Completed	Diploma Completed	Degree and above Completed	Never
COUNTRY T	TOTAL												
Total	1,434,227	29 986	1,928	493,928	143.919	294,207	49,001	6,047	27,462	41,857	121,121	46,319	178,451
Male	462,220		800	146,293	52,078	107,715	22,368	2,783	9,160	19,761	42,502	24,747	25,739
Female	972,007		1,127	347,635	91,841	186,492	26,633	3,264		22,096	78,619	21,572	
TIGRAY REC	,	,	.,	o,ooo	0.,0	.00,.02	_0,000	0,20	. 0,000	,000	. 5,5 . 5	,	.02,0
Total	109,321	1,861	_	39,245	4,132	28,098	3,543	271	1,460	2,167	12,055	3,249	13,241
Male	27,218	259	_	8,965	1,745	8,610	973	271	90	90	3,073	1,652	1,490
Female	82,103	1,603	-	30,280	2,387	19,487	2,570	_	1,369	2,077	8,982	1,596	11,751
AFFAR REG	,	,		,	•	,	,		•	•	•	,	,
Total	19,977	936	-	7,469	1,280	2,388	111	110	76	488	1,703	269	5,147
Male	6,894	136	-	2,386	324	1,528	-	-	34	488	755	24	1,218
Female	13,083	800	-	5,083	955	860	111	110	42	-	948	245	3,929
AMHARA RE	EGION												
Total	339,154	15,004	429	106,129	19,496	69,437	13,027	1,520	8,107	6,628	34,262	10,747	54,367
Male	100,100	5,886	-	27,014	5,822	21,000	5,269	905	2,446	4,016	11,385	6,757	9,598
Female	239,055	9,119	429	79,115	13,673	48,437	7,759	614	5,662	2,612	22,876	3,990	44,769
OROMIA RE	GION												
Total	386,972	4,186	1,326	144,119	44,668	85,452	8,883	2,053	6,736	9,404	26,395	10,254	43,495
Male	128,166	-	800	45,011	13,797	35,213	5,069	837	1,831	4,958	9,848	5,651	5,152
Female	258,806	4,186	525	99,108	30,871	50,240	3,814	1,216	4,905	4,447	16,548	4,603	38,343
SOMALI REC	GION												
Total	20,846	906	-	4,783	1,104	1,249	1,272	-	-	336	1,206	789	9,200
Male	8,066	379	-	1,992	808	491	855	-	-	121	717	737	1,965
Female	12,780	527	-	2,791	296	758	417	-	-	215	488	52	7,235
BENISHANG	SUL-GUMUZ F	REGION											
Total	9,731	77	77	3,892	278	2,239	218	-	223	77	637	29	1,985
Male	2,830	-	-	967	219	1,061	49	-	-	-	243	-	291
Female	6,901	77	77	2,925	58	1,178	169	-	223	77	394	29	1,695

Table 6.4 Cont'd

Table 6.4 C	Total						All Litera	te					Illiterate
Region and Sex	1	Non- Formal	Preschool	Grades 1-8	Secondary not completed	Secondary Completed	Preparatory	10+2/ Certificate Not Completed	Certificate Completed	Diploma or Degree Not Completed	Diploma Completed	Degree and above Completed	Never
S.N.N.P RE	GION												
Total	142,948	2,325	96	54,568	6,607	30,057	5,656	400	4,519	5,687	15,899	4,809	12,325
Male	46,009	2,020	-	16,238	3,232	11,709	2,521	400	1,587	1,998	4,143	2,342	1,838
Female	96,939	2,325	96	38,330	3,375	18,348	3,135		2,932	3,689	11,756	2,467	10,486
GAMBELLA	•	2,020	30	00,000	0,070	10,040	0,100		2,502	0,000	11,700	2,407	10,400
Total	5,101	_	_	1,872	302	854	457	161	_	273	505	_	677
Male	1,511	_	_	581	115	103	279	161	_	122	149	_	-
Female	3,590	_	_	1,291	186	751	177	-	_	151	356	_	677
HARARI RE				1,201	100	751	177			101	330		011
Total	8,209	229	_	2,617	545	2,164	164	48	247	181	1,024	90	901
Male	2,716	50	_	643	221	755	123	48		92	609	90	85
Female	5,493	179	_	1,974		1,409	40	-	247	89	415	-	816
	BA CITY ADM		TION	.,	<u></u>	.,							0.0
Total	366,227	2,593	-	120.887	64,158	67,740	14,949	1,324	5,758	15,972	24,981	15,235	32,629
Male	130,123	957	_	40,467	24,854	25,523	6,774	-	3,099	7,569	10,516	6,898	3,466
Female	236,103	1,636	_	80,420	39,304	42,217	8,175	1,324	2,658	8,403	14,465	8,337	29,163
DIRE DAWA	A ADMINISTRA			, -	-,	,	-,	,-	,	-,	,	-,	,
Total	25,742	1,869	_	8,347	1,349	4,528	722	161	336	642	2,455	848	4,484
Male	8,587	608	_	2,028	939	1,722	456	161	73	306	1,064	596	635
Female	17,154	1,261	_	6,319	410	2,806	266	_	264	336	1,392	251	3,849

Table 6.5 Unemployed Population of Urban Areas of Regions Aged Ten Years and Above by Training Status, Types of Training and Sex: 2015

Training Status, Types	and OCA. 20	-				Re	gion					
of Training and Sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
COUNTRY TOTAL												
TRAINING STATUS												
Total	1,434,227	109,321	19,977	339,154	386,972	20,846	9,731	142,947	5,101	8,209	366,227	25,742
Male	462,220	27,218	6,894	100,100	128,166	8,066	2,830	46,009	1,511	2,716	130,123	8,587
Female	972,007	82,103	13,083	239,055	258,806	12,780	6,901	96,939	3,590	5,493	236,103	17,154
Trained												
Total	340,509	26,205	3,248	83,228	78,156	2,759	1,649	38,311	785	2,828	95,498	7,842
Male	143,957	7,934	1,190	35,632	33,649	1,892	793	15,021	279	1,234	42,197	4,137
Female	196,552	18,271	2,058	47,597	44,507	867	856	23,290	507	1,594	53,301	3,705
Not Trained												
Total	1,093,718	83,116	16,729	255,926	308,816	18,087	8,082	104,636	4,316	5,381	270,729	17,900
Male	318,263	19,284	5,704	64,468	94,517	6,174	2,036	30,988	1,233	1,482	87,926	4,450
Female	775,455	63,832	11,025	191,458	214,299	11,913	6,046	73,649	3,083	3,899	182,802	13,449
DETAILED TYPES (OF TRAINII	NG										
ALL TRAINED												
Total	340,509	26,205	3,248	83,228	78,156	2,759	1,649	38,311	785	2,828	95,498	7,842
Male	143,957	7,934	1,190	35,632	33,649	1,892	793	15,021	279	1,234	42,197	4,137
Female	196,552	18,271	2,058	47,597	44,507	867	856	23,290	507	1,594	53,301	3,705
SOCIAL SCIENCE												
Total	67,704	5,379	208	14,559	14,955	1,148	211	7,422	101	375	22,095	1,250
Male	26,605	2,740	58	6,220	5,605	792	77	2,760	40	133	7,470	709
Female	41,100	2,639	150	8,339	9,350	356	134	4,662	62	242	14,625	541
Accounting & Fir												
Total	23,256	1,167	164	4,053	6,945	161	57	1,550	62	144	8,506	447
Male	9,352	591	24	1,125	3,651	121	-	243	-	-	3,424	174
Female	13,904	576	141	2,929	3,294	40	57	1,307	62	144	5,082	273
Agricultural Ecor												
Total	106	-	-	-	62	-	-	-	-	43	-	-
Male	106	-	-	-	62	-	-	-	-	43	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Archeology & He	-				· ·							
Total	160	-	-	98	61	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	160	-	-	98	61	-	-	-	-	-	-	-

Table 6.5 Cont'd

aining Status, Types of						Re	egion					
Training and Sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Banking &Insurance												
Total	1,666	-	-	-	-	-	-	-	-	-	1,666	
Male	253	-	-	-	-	-	-	-	-	-	253	
Female	1,413	-	-	-	-	-	-	-	-	-	1,413	
Business Administration	n											
Total	1,081	153	-	-	180	-	77	-	-	-	671	
Male	77	-	-	-	-	-	77	-	-	-	-	
Female	1,005	153	-	-	180	-	-	-	-	-	671	
Civics & Ethics												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Cooperatives Organiza	tion & Manag	ement										
Total	191	-	-	103	-	88	-	-	-	-	-	
Male	191	-	-	103	-	88	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Demography												
Total	62	-	-	-	62	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	62	_	-	_	62	_	-	-	_	-	_	
Economics												
Total	2,455	-	-	815	152	186	-	279	-	-	1,022	
Male	2,008	_	-	705	-	186	-	96	_	_	1,022	
Female	447	_	-		152	_	-	184	_	-	· -	
Education												
Total	3,231	583	-	702	572	_	-	601	-	-	679	
Male	1,849	583	_	=00	121	_	-	301	_	_	336	
Female	1,381	-	_	404	452	_	-	300	-	_	343	,
Fine & Theatre Arts	.,							230			2.0	
Total	167	90	_	_	_	_	77	_	_	_	_	
Male	90	90	_	_	_	_	-	_	_	_	_	
Female	77	30	_	_	_	_	77	_	_	_	_	

Table 6.5 Cont'd

raining Status, Types of						Re	egion					
Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Folklore & Literature												
Total	394	-	-	-	394	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	394	-	-	-	394	-	-	-	-	-	-	
Gender & Developmer	nt											
Total	189	-	-	-	-	-	-	189	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	189	-	-	-	-	-	-	189	-	-	-	
Geography												
Total	3,489	723	-	1,570	771	92	-	-	-	-	334	
Male	1,702	362	-	1,094	246	-	-	-	-	-	-	
Female	1,787	360	-	476	525	92	-	-	-	-	334	
Governance & Public	Administration											
Total	5,370	574	-	1,540	836	245	-	1,543	-	49	337	24
Male	2,653	226	-	1,253	-	73	-	942	-	-	-	15
Female	2,717	349	-	288	836	172	-	601	-	49	337	8
History												
Total	1,348	-	-	-	394	-	-	-	-	-	954	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	1,348	-	-	-	394	-	-	-	-	-	954	
Journalism & Broadca	asting											
Total	1,706	-	-	772	58	-	-	184	-	-	691	
Male	1,049	-	-	644	58	-	-	-	-	-	348	
Female	656	-	-	129	-	-	-	184	-	-	344	
Language												
Total	4,252	-	44	1,403	654	-	-	-	-	98	1,964	8
Male	540	-	34	-	123	-	-	-	-	49	334	
Female	3,711	-	10	1,403	531	-	-	-	-	49	1,630	8
Law												
Total	3,143	750	-	331	302	-	-	1,034	-	41	685	
Male	1,595	750	-	229	179	-	-	396	-	41	-	
Female	1,549	-	-	102	123	-	-	638	-	-	685	
Library & Information S												
Total	241	-	-	-	-	-	-	241	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	241	-	-	_	-	_	-	241	-	_	-	

Table 6.5 Cont'd

raining Status, Types of						Re	egion					
Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Management		<u> </u>									'	
Total	2,210	-	-	424	371	41	-	305	-	-	1,069	
Male	1,544	-	-	67	62	41	-	305	-	-	1,069	
Female	666	-	-	356	309	-	-	-	-	-	-	
Management Informati	on System											
Total	999	-	-	594	405	-	-	-	-	-	-	
Male	405	-	-	-	405	-	-	-	-	-	-	
Female	594	-	-	594	-	-	-	-	-	-	-	
Marketing												
Total	1,676	-	-	-	-	-	-	241	-	-	1,435	
Male	241	_	-	_	_	_	-	241	_	_	· -	
Female	1,435	-	-	-	-	-	-	-	-	-	1,435	
Philosophy	,										,	
Total	436	-	-	93	-	-	-	-	-	-	343	
Male	343	-	-	-	-	-	-	-	-	-	343	
Female	93	-	-	93	_	-	-	-	-	-	-	
Political Science												
Total	343	-	-	-	-	-	-	-	-	-	343	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	343	-	-	-	-	-	-	-	-	-	343	
Procurement & Supply	Management											
Total	595	-	-	-	159	-	-	93	-	-	343	
Male	436	_	-	_	_	_	-	93	_	_	343	
Female	159	-	-	-	159	-	-	-	-	-	-	
Psychology												
Total	374	-	-	-	-	-	-	238	-	-	136	
Male	143	-	-	-	-	-	-	143	-	-	-	
Female	232	-	-	-	-	-	-	95	-	-	136	
Purchasing												
Total	599	259	-	_	_	118	-	95	40	_	-	
Male	246	-	-	_	_	118	-	-	40	_	-	
Female	353	259	-	-	-	-	-	95	-	-	-	
Rural Development												
Total	174	_	-	92	-	82	_	-	-	_	-	
Male	82	_	_	-	_	82	-	_	_	_	_	
Female	92	_	_	92	_	-	_	_	_	_	_	

Table 6.5 Cont'd

aining Status, Types of						Re	gion					
Training and Sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Social Work												
Total	261	-	-	-	261	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	261	-	-	-	261	-	-	-	-	-	-	
Sociology & Anthropol	ogy											
Total	1,929	137	-	45	1,205	52	-	402	-	-	-	8
Male	793	137	-	45	523	-	-	-	-	-	-	8
Female	1,136	-	-	-	682	52	-	402	-	-	-	
Special Needs educati	ion											
Total	84	-	-	-	-	84	-	-	-	-	-	
Male	84	-	-	-	-	84	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Tourism & Hotel Mana	gement											
Total	2,489	273	-	775	210	-	-	242	-	-	917	7
Male	249	-	-	-	176	-	-	-	-	-	-	7
Female	2,240	273	-	775	34	-	-	242	-	-	917	
Other Social Science F	Fields											
Total	3,028	670	-	1,147	901	-	-	184	-	-	-	12
Male	573	-	-	447	-	-	-	-	-	-	_	12
Female	2,455	670	-	700	901	-	-	184	-	-	-	
NATURAL SCIENCE												
Total	47,464	3,774	292	18,273	11,037	141	264	4,726	151	139	8,400	26
Male	11,318	362	-	4,555	2,247	90	-	564	-	-	3,499	
Female	36,146	3,412	292	13,718	8,790	51	264	4,162	151	139	4,901	26
Biology												
Total	2,778	153	32	1,528	705	-	-	360	-	-	-	
Male	311	-	-	254	57	-	-	-	-	-	_	
Female	2,467	153	32	1,275	648	-	-	360	-	-	_	
Biotechnology												
Total	403	-	-	-	56	-	-	-	-	-	347	
Male	347	-	-	-	-	_	-	_	-	-	347	
Female	56	_	-	-	56	_	_	_	-	-	-	
Chemistry												
Total	219	_	-	143	-	-	77	_	-	-	_	
Male	75	_	-	75	-	-	-	_	-	-	_	
Female	144	_	_	67	-	_	77	_	_	_	_	

Table 6.5 Cont'd

ning Status, Types of	<u>.</u>					Re	egion					
Training and Sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Computer Science												
Total	24,038	3,105	244	9,595	3,684	90	187	3,104	151	139	3,564	17
Male	4,299	362	-	2,596	367	90	-	-	-	-	883	
Female	19,739	2,742	244	6,999	3,318	-	187	3,104	151	139	2,681	17
Earth Science												
Total	822	-	-	-	520	-	-	301	-	-	-	
Male	400	-	-	-	400	-	-	-	-	-	-	
Female	421	-	-	-	120	-	-	301	-	-	-	
Environmental Science												
Total	1,695	-	16	596	646	-	-	96	-	-	341	
Male	404	-	-	-	62	-	-	-	-	-	341	
Female	1,292	-	16	596	584	-	-	96	-	-	-	
Food Science												
Total	1,336	-	-	-	116	-	-	-	-	-	1,220	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	1,336	-	-	-	116	-	-	-	-	-	1,220	
Geology												
Total	444	-	-	444	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	444	-	-	444	-	-	-	-	-	-	-	
Information Systems												
Total	4,866	-	-	2,012	2,119	-	-	301	-	-	343	Ç
Male	722	-	-	-	379	-	-	-	-	-	343	
Female Mathematics	4,144	-	-	2,012	1,739	-	-	301	-	-	-	(
Total	2,861	_	_	2,122	_	_	_	121	_	_	618	
Male	1,641	_	_	1,233	_	_	_	121	_	_	287	
Female	1,220	_	_	890	_	_	-	-	_	_	331	
Meteorology	,											
Total	_	_	_	_	_	_	-	_	_	_	_	
Male	_	_	_	_	_	_	-	_	_	_	_	
Female	_	_	_	_	_	_	_	_	_	_	_	
Physics												
Total	700	_	_	300	57	_	-	_	_	_	344	
Male	400	_	_	-	57	_	_	_	_	_	344	
Female	300	_	_	300	-	_	_	_	_	_	-	

Table 6.5 Cont'd

aining Status, Types of						Re	egion					
Training and Sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Sport Science												
Total	3,478	-	-	690	1,449	51	-	-	-	-	1,288	
Male	1,972	-	-	94	924	-	-	-	-	-	954	
Female	1,505	-	-	596	525	51	-	-	-	-	333	
Statistics												
Total	362	-	-	62	-	-	-	301	-	-	-	
Male	301	-	-	-	-	-	-	301	-	-	-	
Female	62	-	-	62	-	-	-	-	-	-	-	
Other Natural Science												
Total	3,462	516	-	782	1,685	-	-	143	-	-	336	
Male	446	-	-	303	-	-	-	143	-	-	-	
Female	3,016	516	-	479	1,685	-	-	-	-	-	336	
ENGINEERING & ALI	LIED SERVI											
Total	52,015	3,413	586	12,718	11,271	73	86	6,100	81	443	16,388	85
Male	27,178	1,188	-	5,943	5,146	73	57	3,709	-	171	10,605	28
Female	24,837	2,225	586	6,774	6,125	-	29	2,391	81	272	5,784	5
Agricultural Engineerin	g											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Architecture												
Total	932	-	-	92	523	-	-	-	-	49	267	
Male	141	-	-	92	-	-	-	-	-	49	-	
Female	790	-	-	-	523	-	-	-	-	-	267	
Cadastre & mapping S												
Total	88	-	-	-	88	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	88	-	-	-	88	-	-	-	-	-	-	
Chemical Engineering												
Total	604	-	-	-	262	-	-	-	-	-	342	
Male	262	-	-	-	262	-	-	-	-	-	-	
Female	342	-	-	-	-	-	-	-	-	-	342	
Civil Engineering												
Total	3,376	153	-	169	680	-	-	121	-	122	2,039	(
Male	2,089	-	-	-	124	-	-	-	-	122	1,751	(
Female	1,287	153	-	169	556	-	-	121	-	-	287	

Table 6.5 Cont'd

aining Status, Types of						Re	gion					
Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Computer Engineering	<u> </u>					•				•	•	
Total	4,240	133	-	1,046	541	-	-	797	-	-	1,722	
Male	2,060	-	-	892	-	-	-	401	-	-	767	
Female	2,180	133	-	155	541	-	-	396	-	-	954	
Communication Engine	eering											
Total	194	-	-	-	-	-	-	-	-	-	-	19
Male	194	-	-	-	-	-	-	-	-	-	-	19
Female	-	-	-	-	-	-	-	-	-	-	-	
Construction Technolo	gy											
Total	4,023	-	-	190	837	-	29	1,285	-	-	1,682	
Male	2,229	-	-	103	-	-	-	444	-	-	1,682	
Female	1,794	-	-	87	837	-	29	841	-	-	-	
Drafting												
Total	838	-	-	728	-	-	-	-	-	110	-	
Male	75	-	-	75	-	-	-	-	-	-	-	
Female	763	-	-	653	-	-	-	-	-	110	-	
Electrical engineering												
Total	7,437	515	32	1,783	1,606	-	-	882	81	-	2,447	g
Male	4,754	153	-	748	1,168	-	-	582	-	-	2,102	
Female	2,683	362	32	1,035	439	-	-	300	81	-	344	ç
Esthetics												
Total	238	238	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female Food Processing	238	238	-	-	-	-	-	-	-	-	-	
Total	6,464	-	-	2,041	636	-	-	-	-	162	3,464	16
Male	1,718	-	-	142	-	-	-	-	-	-	1,576	
Female Industrial Engineering	4,746	-	-	1,899	636	-	-	-	-	162	1,888	16
Total	1,716	223	_	709	_	_	_	693	_	_	_	ç
Male	825	223	_	-	_	_	_	602	_	_	_	
Female	891	-	_	709	_	_	- -	91	_	_	_	ç
Manufacturing Technol				700				51				
Total	177	90	-	87	-	_	-	-	_	_	_	
Male	90	90	-	-	-	_	-	-	_	_	_	
Female	87	-	_	87	_	_	_	_	_	_	_	

Table 6.5 Cont'd

aining Status, Types of						Re	egion					
Training and Sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Mechanical Engineering	ng											
Total	3,668	90	-	645	1,062	-	-	396	-	-	1,475	
Male	3,336	90	-	645	1,062	-	-	396	-	-	1,142	
Female	333	-	-	-	-	-	-	-	-	-	333	
Metal Engineering												
Total	809	-	-	-	522	-	-	-	-	-	287	
Male	809	-	-	-	522	-	-	-	-	-	287	
Female	-	-	-	-	-	-	-	-	-	-	-	
Petroleum Engineering	9											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Surveying												
Total	9,771	1,020	224	4,688	1,577	-	29	1,189	-	-	973	
Male	5,208	133	-	2,798	1,013	-	29	948	-	-	287	
Female	4,563	887	224	1,889	564	-	-	241	-	-	685	
Textile Engineering												
Total	360	360	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	360	360	-	-	-	-	-	-	-	-	-	
Urban Land Developm	ent Service											
Total	59	-	-	-	-	-	-	59	-	-	-	
Male	59	-	-	-	-	-	-	59	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Urban Planning & Man	-											
Total	363	363	-	-	-	-	-	-	-	-	-	
Male	363	363	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Water Resources Engi	neering											
Total	1,644	90	331	92	933	73	-	125	-	-	-	
Male	832	-	-	92	541	73	-	125	-	-	-	
Female	813	90	331	-	392	-	-	-	-	-	-	
Wood Science Techno	logy											
Total	2,679	135	-	355	523	-	-	151	-	-	1,361	1
Male	1,321	135	-	355	-	-	-	151	-	-	679	
Female	1,358	-	-	-	523	-	-	-	-	-	682	1

Table 6.5 Cont'd

Fraining Status, Types of						Re	egion					
Training and Sex		_		_	_	_	Benishangul-	_			Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Other Engineering Field												
Total	2,335	-	-	93	1,481	-	28	401	-	-	331	
Male	813	-	-	-	454	-	28	-	-	-	331	
Female HEALTH SCIENCE	1,522	-	-	93	1,027	-	-	401	-	-	-	
Total	31,724	2,262	219	6,890	9,600	495	170	5,526	70	523	5,519	45
Male	9,857	-	-	1,607	4,206	238	-	796	-	386	2,623	
Female	21,868	2,262	219	5,283	5,394	257	170	4,729	70	137	2,896	45
Anesthesia												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Animal Health Care												
Total	993	-	-	92	785	80	36	-	-	-	-	
Male	865	-	-	-	785	80	-	-	-	-	-	
Female	128	-	-	92	-	-	36	-	-	-	-	
Dentistry												
Total	344	-	-	-	-	-	-	-	-	-	344	
Male	344	-	-	-	-	-	-	-	-	-	344	
Female	-	-	-	-	-	-	-	-	-	-	-	
Dental Hygiene												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female Food Hygiene	-	-	-	-	-	-	-	-	-	-	-	
Total	334	-	-	-	-	-	-	-	-	-	334	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	334	-	-	-	-	-	-	-	-	-	334	
Health Extension												
Total	1,647	-	-	599	61	-	77	-	-	-	910	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	1,647	-	-	599	61	-	77	-	-	-	910	
Health Officer												
Total	1,340	-	-	-	581	-	-	96	-	41	624	
Male	1,188	-	-	-	524	-	-	-	-	41	624	
Female	153	-	-	-	57	-	-	96	-	-	-	

Table 6.5 Cont'd

aining Status, Types of						Re	egion					
Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Daw
Massage & Physiothera	ру											
Total	809	-	-	75	-	-	-	-	-	-	733	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	809	-	-	75	-	-	-	-	-	-	733	
Medical Doctor												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Medical Laboratory												
Total	941	-	-	105	121	-	-	323	-	48	344	
Male	209	-	-	-	-	-	-	161	-	48	-	
Female	732	-	-	105	121	-	-	161	-	-	344	
Midwifery												
Total	2,507	1,087	-	724	523	-	-	-	-	174	-	
Male	133		-			-	-	-	-	133	-	
Female	2,374	1,087	-	724	523	-	-	-	-	41	-	
Nursing												
Total	16,315	1,175	-	0,000	5,141	415	57	4,101	49	214	1,219	
Male	4,121	-	-	.,	1,382	158	-	635	-	117	645	
Female	12,195	1,175	-	2,399	3,759	257	57	3,465	49	97	574	
Nursing Assistance	0.007		0.40								007	
Total	2,027	-	219	-	1,141	-	-	-	-	-	667	
Male	1,288	-	-	-	621	-	-	-	-	-	667	
Female Optometry	739	-	219	-	520	-	-	-	-	-	-	
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Pharmacy												
Total	2,067	-	-	424	948	-	-	305	-	48	343	
Male	1,652	-	-	424	838	-	-	-	-	48	343	
Female	415	-	-	-	110	-	-	305	-	-	-	
Psychiatry												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	

Table 6.5 Cont'd

Training Status, Types of						Re	egion					
Training and Sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Public/Environmental												
Total	358	-	-	-	57	-	-	301	-	-	-	
Male	57	-	-	-	57	-	-	-	-	-	-	
Female	301	-	-	-	-	-	-	301	-	-	-	
Radiography												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Retail Pharmacy Assi	stance											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Veterinary Medicine												
Total	1,689	-	-	1,288	-	-	-	401	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	1,689	-	-	1,288	-	-	-	401	-	-	-	
Other Health Fields												
Total	353	-	-	-	243	-	-	-	22	-	-	8
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	353	-	-	-	243	-	-	-	22	-	-	8
AGRICULTURAL SC												
Total	7,933	1,579	969	1,349	2,227	299	99	618	253	92	-	77
Male	3,884	-	755	445	1,467	299	70	285	109	92	-	36
Female	4,049	1,579	214	905	760	-	29	333	143	-	-	8
Agricultural Extension												
Total	2,969	360	-	356	1,558	133	-	242	88	50	-	18
Male	1,284	-	-	-	832	133	-	-	88	50	-	18
Female	1,685	360	-	356	726	-	-	242	-	-	-	
Animal Production & I	Management											
Total	650	-	428	-	-	73	-	-	62	-	-	8
Male	523	-	428	-	-	73	-	-	22	-	-	
Female	127	-	-	-	-	-	-	-	40	-	-	8
Animal Science												
Total	560	259	-	-	118	92	29	-	63	-	-	
Male	210	-	-	-	118	92	-	-	-	-	-	
Female	350	259	-	-	-	-	29	-	63	-	-	

Table 6.5 Cont'd

aining Status, Types of						Re	egion					
Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Artificial insemination				•				•	•			•
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Bamboo Harvesting												
Total	93	-	-	93	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	93	-	-	93	-	-	-	-	-	-	-	
Beekeeping												
Total	343	343	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	343	343	-	-	-	-	-	-	-	-	-	
Dairy Production												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Field Crops Production												
Total	214	-	214	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	214	-	214	-	-	-	-	-	-	-	-	
Forestry												
Total	176	-	-	-	34	-	-	143	-	-	-	
Male	143	-	-	-	-	-	-	143	-	-	-	
Female	34	-	-	-	34	-	-	-	-	-	-	
Horticulture												
Total	164	-	-	-	123	-	-	-	40	-	-	
Male	123	-	-	-	123	-	-	-	-	-	-	
Female	40	-	-	-	-	-	-	-	40	-	-	
Leather Basic Production	on											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Natural Resource Mana	-											
Total	1,114	-	327	445	-	-	70		-	-	-	180
Male	1,023	-	327	445	-	-	70		-	-	-	180
Female	91	-	-	-	-	-	-	91	-	-	-	

Table 6.5 Cont'd

ining Status, Types of						Re	egion					
Training and Sex	Total	Tigrov	Affor	Ambara	Oromic	Complia	Benishangul-	CNNDD	Combols	Horori	Addis	Dire
Plant Science	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female Post Harvest Manager	- ment	-	-	-	-	-	-	-	-	-	-	
Total	-	_	_	_	_	_	_	_	_	_	_	
Male	_	_	_	_	_	_	_	_	_	_	_	
Female	-	_	-	_	-	_	-	-	_	_	-	
Poultry Production	-	-	-	_	-	-	-	-	_	_	-	
Total	607	255	_	352								
Male	007	200	-	332	-	-	-	-	-	-	-	
Female	607	- 255	-	352	-	-	-	-	-	-	-	
Range Science	007	200	-	332	-	-	-	-	-	-	-	
Total												
Male	-	-	-	-	-	-	-	-	-	-	-	
rivale Female	-	-	-	-	-	-	-	-	-	-	-	
	- raduation	-	-	-	-	-	-	-	-	-	-	
Stimulants & Spices P	roduction											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Sugar Crops Production	on											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female Vegetables, Fruits, Flo	- owers Production	n -	-	-	-	-	-	-	-	-	-	
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	_	-	_	-	-	-	-	-	-	-	
Female	-	_	-	_	-	-	-	-	-	-	-	
Other Agricultural Field	ds											
Total	1,043	361	-	103	394	_	-	143	_	42	_	
Male	579	_	-	-	394	_	-	143	_	42	_	
Female	464	361	-	103	_	_	-	_	_	-	_	
CONSTRUCTION INI												
Total	23,473	2,830	126	9,817	4,480	50	111	1,160	-	350	3,674	8
Male	14,741	1,580	-		2,059	-	39	842	_	139	3,339	55
Female	8,732	1,251	126		2,421	50	72		_	211	335	32

Table 6.5 Cont'd

ining Status, Types of						Re	gion					
Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Bar Bending & Concre	ting				•					•	•	
Total	690	-	-	355	-	-	-	-	-	-	335	
Male	355	-	-	355	-	-	-	-	-	-	-	
Female	335	-	-	-	-	-	-	-	-	-	335	
Building Construction												
Total	4,650	1,081	16	2,268	34	-	75	842	-	-	334	
Male	2,611	718	-	644	34	-	39	842	-	-	334	
Female	2,039	362	16	1,624	-	-	36	-	-	-	-	
Building Electrical Insta	allation											
Total	1,584	256	-	538	394	50	-	-	-	-	347	
Male	520	-	-	93	80	-	-	-	-	-	347	
Female	1,064	256	-	445	314	50	-	-	-	-	-	
Building Metal construc	ction											
Total	122	-	-	-	-	-	-	-	-	122	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	122	-	-	-	-	-	-	-	-	122	-	
Carpentry & Joinery												
Total	2,236	361	-	1,875	-	-	-	-	-	-	-	
Male	1,965	90	-	1,875	-	-	-	-	-	-	-	
Female	271	271	-	-	-	-	-	-	-	-	-	
Coble Stone Carving												
Total	1,448	-	-	611	599	-	36	-	-	49	-	15
Male	439	-	-	75	210	-	-	-	-	-	-	15
Female	1,009	-	-	536	389	-	36	-	-	49	-	
Coble Stone Road Cor	nstruction											
Total	2,532	-	-	928	1,382	-	-	125	-	97	-	
Male	697	-	-	322	278	-	-	-	-	97	-	
Female	1,835	-	-	606	1,104	-	-	125	-	-	-	
Concreting & Reinforci	ng Works											
Total	617	-	-	-	522	-	-	96	-	-	-	
Male	522	-	-	-	522	_	-	_	_	-	-	
Female	96	-	-	-	-	-	-	96	-	-	-	
Construction Equipmen												
Total	420	_	_	-	-	-	-	_	-	-	332	8
Male	420	-	-	-	-	-	-	-	-	-	332	8
Female	-	_	_	_	_	_	-	_	_	_		Ū

Table 6.5 Cont'd

aining Status, Types of						Re	egion					
Training and Sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Construction Manager	ment											
Total	1,925	227	-	185	784	-	-	-	-	41	689	
Male	1,702	137	-	93	784	-	-	-	-	-	689	
Female	223	90	-	92	-	-	-	-	-	41	-	
Finishing Construction												
Total	417	-	-	92	-	-	-	-	-	-	-	32
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	417	-	-	92	-	-	-	-	-	-	-	32
Foundry Works												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Geo-Technical Drilling												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Masonry Works												
Total	93	-	-	93	-	-	-	-	-	-	-	
Male	93	-	-	93	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Metal Machining												
Total	160	-	-	-	67	-	-	-	-	-	-	ç
Male	126	-	-	-	34	-	-	-	-	-	-	9
Female	34	-	-	-	34	-	-	-	-	-	-	
Metal Works												
Total	2,855	153	-	1,886	-	-	-	-	-	42	683	ç
Male	2,855	153	-	1,886	-	-	-	-	-	42	683	ç
Female	-	-	-	-	-	-	-	-	-	-	-	
Plastering & Paints Wo	orks											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Road Construction Mai	ntenance											
Total	1,656	-	-	523	178	-	-	-	-	-	954	
Male	1,595	-	-	523	118	-	-	-	-	-	954	
Female	60	-	-	-	60	-	-	-	-	-	-	

Table 6.5 Cont'd

raining Status, Types of						Re	egion					
Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Road Surfacing		3 - 7						_				
Total	_	_	_	_	_	_	_	_	_	_	_	
Male	_	_	_	-	_	-	-	_	_	_	_	
Female	-	_	-	-	_	_	-	-	_	-	_	
Tiling Work												
Total	-	-	-	-	-	-	-	_	-	-	_	
Male	-	-	-	-	-	-	-	-	-	-	_	
Female	-	-	-	-	-	-	-	-	-	-	-	
Tunnel Construction												
Total	440	349	-	91	-	-	-	-	-	-	-	
Male	440	349	-	91	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Water Well Drilling & C	onstruction											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Welding												
Total	404	404	-	-	-	-	-	-	-	-	-	
Male	133	133	-	-	-	-	-	-	-	-	-	
Female	272	272	-	-	-	-	-	-	-	-	-	
Other Construction Fiel	lds											
Total	1,224	-	110		520	-	-	96	-	-	-	12
Male	269	-	-	142	-	-	-	-	-	-	-	12
Female	956	-	110	229	520	-	-	96	-	-	-	
TRANSPORT OPERAT												
Total	36,960	1,332	251	8,158	8,495	299	431	5,364	130	272	10,606	1,62
Male	33,325	1,332	251	7,571	7,462	299	431	4,821	130	231	9,176	1,62
Female	3,635	-	-	587	1,033	-	-	544	-	41	1,431	
Aircraft Mechanic												
Total	67	-	-	-	67	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	67	-	-	-	67	-	-	-	-	-	-	
Automotive Driving												
Total	2,095	-	-	725	1,045	-	143	-	-	-	-	18
Male	1,571	-	-	725	521	-	143	-	-	-	-	18
Female	524	-	-	-	524	-	-	-	-	-	-	

Table 6.5 Cont'd

aining Status, Types of						Re	gion					
Training and Sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Automotive Electrical/I	Electronic Serv	vicing										
Total	6,275	-	-	1,933	796	80	-	1,276	-	41	2,060	8
Male	4,939	-	-	1,488	679	80	-	733	-	-	1,871	8
Female	1,336	-	-	445	118	-	-	544	-	41	189	
Automotive Mechanic												
Total	8,212	893	31	2,824	894	-	-	541	-	-	3,029	
Male	8,091	893	31	2,824	774	-	-	541	-	-	3,029	
Female	121	-	-	-	121	-	-	-	-	-	-	
Automotive Painting												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Automotive Servicing I	-											
Total	854	-	-	-	123	-	-	-	-	-	645	8
Male	854	-	-	-	123	-	-	-	-	-	645	8
Female	-	-	-	-	-	-	-	-	-	-	-	
Automotive Technolog												
Total	817	-	-	355	120	-	-	-	-	-	343	
Male	698	-	-	355	-	-	-	-	-	-	343	
Female	120	-	-	-	120	-	-	-	-	-	-	
City Bus Driving												
Total	441	-	-	-	-	-	-	360	40	41	-	
Male	441	-	-	-	-	-	-	360	40	41	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Taxi Driving												
Total	6,137	-	-	944	1,666	219	68	-	41	91	1,988	1,12
Male	5,994	-	-	801	1,666	219	68	-	41	91	1,988	1,12
Female	142	-	-	142	-	-	-	-	-	-	-	
Other Transport Service	ce Fields											
Total	12,062	439	220	1,378	3,784	-	219	3,187	49	99	2,542	14
Male	10,737	439	220	1,378	3,700	-	219	3,187	49	99	1,300	14
Female	1,326	-	-	-	84	-	-	-	-	-	1,242	
INFRASTRUCTURE	SERVICES											
Total	4,953	238	112	1,580	114	-	29	995	-	-	1,705	18
Male	2,745	-	-	593	114	-	-	241	-	-	1,705	9
Female	2,209	238	112	986	-	-	29	754	-	-	-	8

Table 6.5 Cont'd

raining Status, Types of						Re	egion					
Training and Sex							Benishangul-				Addis	Dire
3	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Biomedical Equipment	t Servicing					•						
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Home/Office Elect/Ele	ctronics Equip	. Servicing										
Total	488	-	-	-	57	-	-	-	-	-	342	88
Male	399	-	-	-	57	-	-	-	-	-	342	-
Female	88	-	-	-	-	-	-	-	-	-	-	88
Hydrometric Monitoring	g											
Total	238	238	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	238	238	-	-	-	-	-	-	-	-	-	
Infrastructure operation	ns											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
IT Service Manageme	nt											
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
IT support Service												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Mechatronics & Instrur		vicing										
Total	316	-	-	-	-	-	29	-	-	-	287	-
Male	287	-	-	-	-	-	-	-	-	-	287	
Female	29	-	-	-	-	-	29	-	-	-	-	-
Production Machine In	nstallation											
Total	342	-	-	-	-	-	-	-	-	-	342	
Male	342	-	-	-	-	-	-	-	-	-	342	
Female	-	-	-	-	-	-	-	-	-	-	-	
Sanitary Installation												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	

Table 6.5 Cont'd

aining Status, Types of						Re	egion					
Training and Sex							Benishangul-				Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Daw
Telecommunication W	orks											
Total	733	-	-	-	-	-	-	-	-	-	733	
Male	733	-	-	-	-	-	-	-	-	-	733	
Female	-	-	-	-	-	-	-	-	-	-	-	
Water Supply & Distrib		ons										
Total	876	-	112	93	-	-	-	579	-	-	-	
Male	332	-	-	-	-	-	-	241	-	-	-	
Female	544	-	112	93	-	-	-	339	-	-	-	
Web & Multimedia Des	signing											
Total	446	-	-	446	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	446	-	-	446	-	-	-	-	-	-	-	
Other Infrastructure Se	ervice Fields											
Total	1,513	-	-	1,041	57	-	-	416	-	-	-	
Male	650	-	-	593	57	-	-	-	-	-	-	
Female	863	-	-	447	-	-	-	416	-	-	-	
PRODUCTION OPERA	ATIONS											
Total	8,721	1,408	235	467	1,640	-	-	813	-	49	3,932	
Male	1,022	191	16	158	60	-	-	-	-	-	596	
Female	7,699	1,217	219	308	1,580	-	-	813	-	49	3,336	
Animal By-Product Pro	ocessing											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Plant By-Product Proc	essing											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Hide & Skin Processin	g											
Total	780	-	-	94	-	-	-	-	-	-	686	
Male	343	-	-	-	-	-	-	-	-	-	343	
Female	437	-	-	94	-	-	-	-	-	-	343	
Textile Chemical Proce	essing											
Total	253	-	-	-	-	-	-	-	-	-	253	
Male	253	-	-	-	-	-	-	-	-	-	253	
Female	-	_	-	_	_	_	-	_	_	_	-	

Table 6.5 Cont'd

aining Status, Types of						Re	egion					
Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Daw
CNC Machine Operation	on	<u> </u>				ı		ı			Į.	
Total	_	_	_	-	-	_	-	_	_	_	_	
Male	_	_	_	-	-	_	-	_	_	_	_	
Female	-	-	-	_	-	-	-	_	-	-	-	
Tool & Die Making												
Total	534	191	-	_	-	-	-	_	-	-	342	
Male	191	191	_	-	-	_	-	-	_	_	-	
Female	342	-	-	-	-	-	-	-	-	-	342	
Dry Food & Spice Pro												
Total	706	-	-	154	-	-	-	212	-	-	341	
Male	_	_	_	-	-	_	-	-	_	_	-	
Female	706	_	-	154	-	-	-	212	_	_	341	
Bamboo Derivatives Pr	roduction											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Bamboo Furniture Mak	king											
Total	343	-	-	-	-	-	-	-	-	-	343	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	343	-	-	-	-	-	-	-	-	-	343	
Sugar & Ethanol Produ	uction											
Total	16	-	16	-	-	-	-	-	-	-	-	
Male	16	-	16	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Garment production												
Total	343	-	-	-	-	-	-	-	-	-	343	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	343	-	-	-	-	-	-	-	-	-	343	
Leather Goods Making	I											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Metal Machining												
Total	86	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	86	-	-	-	-	-	-	-	-	-	-	

Table 6.5 Cont'd

aining Status, Types of						Re	egion					
Training and Sex			A. 65				Benishangul-	0111:55			Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Millwright												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Root Crops Processing	g											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Production Technology	y											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Furniture Making												
Total	893	674	-	158	60	-	-	-	-	-	-	
Male	219	-	-	158	60	-	-	-	-	-	-	
Female	674	674	-	-	-	-	-	-	-	-	-	
Dressmaking & Tailorii	ng											
Total	4,677	452	219	61	1,580	-	-	601	-	49	1,624	Ş
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	4,677	452	219	61	1,580	-	-	601	-	49	1,624	Ç
Cement Production Eq	quipment Oper	ation										
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Other Production Oper	rations											
Total	90	90	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	90	90	-	-	-	-	-	-	-	-	-	
OTHER SOCIAL SERV	VICES											
Total	59,561	3,990	249	9,418	14,338	254	249	5,587	-	584	23,178	1,7
Male	13,282	540	110	2,346	5,283	101	119	1,002	-	81	3,184	51
Female	46,278	3,449	139	7,072	9,055	153	129	4,585	-	503	19,994	1,19
Ambulance Service &	Emergency Ca	are										
Total	865	-	-	-	-	-	-	865	-	-	_	
Male	-	-	-	-	-	-	-	-	-	-	_	
Female	865	-	_	-	-	-	-	865	-	-	_	

Table 6.5 Cont'd

raining Status, Types of						Re	egion					
Training and Sex	Total	Tigray	Affar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Domestic Care giving	· · · · · · · · · · · · · · · · · · ·			•					•			•
Total	688	-	-	-	521	-	-	-	-	41	-	126
Male	521	-	-	-	521	-	-	-	-	-	-	
Female	167	-	-	-	-	-	-	-	-	41	-	126
Domestic Service												
Total	-	-	-	-	-	-	-	-	-	-	-	,
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Embalming Service												
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Embroidery												
Total	1,762	-	-	676	583	-	-	393	-	110	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	1,762	-	-	676	583	-	-	393	-	110	-	-
Fashion Designing												
Total	3,139	90	-	266	303	-	-	-	-	122	2,270	88
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	3,139	90	-	266	303	-	-	-	-	122	2,270	88
Hairdressing Beauty C	are											
Total	17,497	2,405	-	2,485	5,197	153	-	734	-	-	6,181	343
Male	1,390	191	-	-	520	-	-	-	-	-	679	-
Female	16,108	2,213	-	2,485	4,677	153	-	734	-	-	5,503	343
Housekeeping & Laund	dry Service											
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Mortuary & Cemetery S	Service											
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Music												
Total	635	-	-	446	-	-	-	-	-	-	189	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	635	-	-	446	-	-	-	-	-	-	189	-

Table 6.5 Cont'd

Training Status, Types of						Re	egion					
Training and Sex	_				_		Benishangul-		_		Addis	Dire
	Total	Tigray	Affar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Police												
Total	4,526	-	110	1,391	771	50	43	400	-	-	1,518	24
Male	4,373	-	110	1,391	771	50	43	400	-	-	1,518	g
Female	153	-	-	-	-	-	-	-	-	-	-	15
Pottery												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Records & Archives Se	ervices											
Total	92	-	-	92	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	92	-	-	92	-	-	-	-	-	-	-	
Secretarial Service												
Total	11,939	615	139	1,654	1,959	-	129	1,126	-	219	5,433	66
Male	704	-	-	446	-	-	-	-	-	81	-	1
Female	11,235	615	139	1,208	1,959	-	129	1,126	-	138	5,433	48
Tour & Travel Service												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Video & Photography												
Total	524	-	-	-	524	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	524	-	-	-	524	-	-	-	-	-	-	
Waste Handling & Was	ste Diversion											
Total	466	-	-	-	93	-	-	-	-	42	332	
Male	_	_	_	_	-	_	-	-	_	-	_	
Female	466	-	-	-	93	-	-	-	-	42	332	
Weaving												
Total	761	_	-	577	185	-	-	_	-	-	_	
Male	185	_	-	-	185	-	-	-	-	-	-	
Female	577	_	_	577	-	-	-	_	_	-	-	
Other Social Services												
Total	16,664	880	-	1,830	4,203	51	77	2,070	-	50	7,256	2
Male	6,110	349	-	509	3,286	51	77	603	-	-	987	2
Female	10,555	531	_	1,321	917	-	-	1,467	_	50	6,268	_

Table 6.6 Unemployed Populations of Urban Areas Aged Ten Years and Above by Marital Status, Age Group and Sex - COUNTRY TOTAL: 2015

	Status, Age	Oroup and		Marital Statu			
Age Group		Never		viai itai Statu	15		Live
and Sex	Total	Married	Married	Divorced	Widowed	Separated	together
All Ages	Total	Marrica	Marrica	Divolcca	vvidovcd	Ocparated	together
Total	1,434,227	641,074	630,660	77,424	53,288	21,230	10,551
Male	462,220	323,033	114,173	14,365	4,054	3,469	3,127
Female	972,007	318,041	516,488	63,059	49,235	17,760	7,424
10-14	372,007	310,041	310,400	00,000	45,255	17,700	7,727
Total	24,390	23,267	1,123				
Male	9,146	8,472	674	_	_	_	_
Female	15,244	14,795	449	-	-	-	-
15-19	15,244	14,795	449	-	-	-	-
Total	204 546	163,781	34,964	1 760		1 020	2 002
	204,546			1,769	-	1,039	2,993
Male	71,926	70,078	1,402	103	-	4 000	343
Female	132,621	93,704	33,562	1,666	-	1,039	2,650
20-24	440.400	220 000	450.004	40.040	4 470	4 700	2.050
Total	418,460	238,860	159,631	12,912	1,472	1,728	3,858
Male	137,166	125,027	11,287	529	4 470	134	189
Female	281,294	113,833	148,344	12,382	1,472	1,594	3,669
25-29	007.500	405.074	4.40.504	45.074	4 005	0.550	4.550
Total	307,539	135,671	149,521	15,974	1,265	3,558	1,550
Male .	104,586	80,419	19,422	3,116	112	167	1,352
Female	202,953	55,253	130,100	12,858	1,154	3,391	198
30-34							
Total	152,036	41,177	97,911	5,785	2,812	3,876	475
Male	42,939	21,941	17,798	1,907	303	898	93
Female	109,097	19,237	80,114	3,878	2,510	2,978	382
35-39							
Total	116,225	22,784	69,814	11,868	8,634	2,590	536
Male .	26,618	10,987	11,171	3,353	406	618	84
Female	89,607	11,797	58,643	8,515	8,227	1,972	452
40-44							
Total	55,802	8,632	32,420	7,722	4,678	1,571	779
Male	17,988	3,684	10,110	2,573	523	391	707
Female	37,815	4,947	22,310	5,149	4,155	1,180	73
45-49							
Total	54,645	2,902	35,246	6,649	6,923	2,564	360
Male	14,044	968	12,123	238	-	356	360
Female	40,601	1,934	23,123	6,412	6,923	2,209	-
50-54							
Total	30,941	673	16,075	5,713	6,639	1,842	-
Male	7,974	317	7,137	111	129	280	-
Female	22,967	356	8,938	5,602	6,509	1,562	-
55-59							
Total	31,904	1,783	16,936	3,645	8,232	1,308	-
Male	10,468	806	8,504	464	115	578	-
Female	21,436	978	8,432	3,181	8,116	730	-
60-64							
Total	14,841	-	7,341	2,041	5,298	161	-
Male	6,664	-	5,745	-	919	-	-
Female	8,177	-	1,596	2,041	4,379	161	-
65+							
Total	22,897	1,543	9,678	3,346	7,337	993	-
Male	12,702	336	8,801	1,971	1,546	48	-
Female	10,195	1,207	878	1,375	5,790	945	-

Table 6.7 Unemployed Population of Urban Areas Aged Ten Years and Above by Marital Status, Region and Sex: 2015

Status, Region and Sex: 2015 Marital Status Region and Sex Never													
		<u>.</u>	Ma	rital Status									
Region and Sex		Never					Live						
	Total	Married	Married	Divorced	Widowed	Separated	together						
COUNTRY TOTAL													
Total	1,434,227	641,074	630,660	77,424	53,288	21,230	10,551						
Male	462,220	323,033	114,173	14,365	4,054	3,469	3,127						
Female	972,007	318,041	516,488	63,059	49,235	17,760	7,424						
TIGRAY REGION													
Total	109,321	43,881	50,381	9,582	1,508	2,471	1,498						
Male	27,218	19,655	5,351	817	90	135	1,169						
Female	82,103	24,225	45,030	8,765	1,418	2,336	329						
AFFAR REGION													
Total	19,977	5,567	11,977	1,001	935	435	62						
Male	6,894	4,647	1,670	443	134	-	_						
Female	13,083	919	10,306	559	801	435	62						
AMHARA REGION	-,		-,				_						
Total	339,154	124,623	171,305	28,078	11,062	1,866	2,221						
Male	100,100	65,411	27,898	5,739	898	61	93						
Female	239,055	59,212	143,407	22,339	10,164	1,805	2,128						
OROMIA REGION	200,000	00,212	0, .0.	22,000	10,101	1,000	2,120						
Total	386,972	162,916	179,931	18,335	15,638	8,176	1,977						
Male	128,166	89,268	31,890	3,149	643	2,596	620						
Female	258,806	73,648	148,041	15,186	14,994	5,580	1,356						
SOMALI REGION	200,000	70,010	1 10,0 11	10,100	1 1,00 1	0,000	1,000						
Total	20,846	6,832	12,076	974	383	288	292						
Male	8,066	3,575	4,184	308	-	200	202						
Female	12,780	3,257	7,892	667	383	288	292						
BENISHANGUL-GUM		3,237	7,032	007	303	200	232						
Total	9,731	2,436	5,487	931	586	262	29						
Male	2,830	1,744	525	268	292	202	29						
Female	6,901	693	4,961	663	292	262	29						
S.N.N.P REGION	6,901	093	4,901	003	294	202	29						
Total	142,947	67.011	64,306	2 700	E E16	1,211	1,105						
		67,011		3,799	5,516		1,105						
Male	46,009	34,312	10,309	1,199	96	93	4 405						
Female	96,939	32,698	53,997	2,601	5,421	1,117	1,105						
GAMBELLA REGION	E 404	4 740	0.744	75	400	200							
Total	5,101	1,748	2,741	75 50	138	399	-						
Male	1,511	915	481	53	400	62	-						
Female	3,590	833	2,260	22	138	337	-						
HARARI REGION	0.000	0.700	0.040		4.40	450	000						
Total	8,209	2,700	3,349	635	448	450	628						
Male	2,716	1,673	589	141	49	90	175						
Female	5,493	1,027	2,760	494	399	360	453						
ADDIS ABABA CITY A													
Total	366,227	212,847	117,454	12,479	15,897	5,332	2,218						
Male	130,123	96,483	28,617	2,096	1,680	343	905						
Female	236,103	116,364	88,837	10,383	14,217	4,989	1,313						
DIRE DAWA ADMINIS													
Total	25,742	10,514	11,655	1,533	1,177	341	521						
Male	8,587	5,351	2,659	153	171	88	164						
Female	17,154	5,163	8,996	1,380	1,006	253	357						

264

Table 6.8 Unemployed Populations of Urban Areas Aged Ten Years and Above Who Were Seeking and Not Seeking Work by Steps Taken to Seek Work, Region and Sex: 2015

Takon	TO OCCION TO	roik, itegioi	i dila ook								1
				Steps T	aken to Sear	ch for Work					
Danian and Cau			Through		Seeking						Not
Region and Sex		Searching	news		assistance	Trying to	Direct				Seeking
		vacancy	paper,	I have	of friends,	establish	application	Checking			Job
		advertising	Radio	unemployment	relatives,	own	to	at work		Not	
	Total	boards	and TV	card	etc	enterprise	employee	sites	Others	Stated	
COUNTRY TOTAL											
Total	1,434,227	256,863	54,039	34,949	227,313	199,520	27,060	110,961	15,233	14,848	493,442
Male	462,220	106,223	24,818	13,041	86,138	62,783	11,263	52,606	5,608	4,088	95,651
Female	972,007	150,640	29,220	21,908	141,175	136,736	15,797	58,355	9,626	10,760	397,791
TIGRAY REGION											
Total	109,321	25,234	318	881	9,411	9,616	800	11,633	90	-	51,337
Male	27,218	7,771	228	643	2,232	3,123	-	4,404	90	-	8,728
Female	82,103	17,463	90	238	7,179	6,493	800	7,229	-	-	42,609
AFFAR REGION											
Total	19,977	1,055	111	-	3,979	2,772	1,371	3,617	224	-	6,848
Male	6,894	308	-	-	1,581	799	823	1,602	-	-	1,782
Female	13,083	747	111	-	2,398	1,973	548	2,015	224	-	5,066
AMHARA REGION											
Total	339,154	66,334	7,071	1,030	39,332	65,289	5,062	19,739	444	75	134,780
Male	100,100	28,589	2,949	416	13,195	16,594	2,173	10,217	444	-	25,523
Female	239,055	37,745	4,122	613	26,137	48,695	2,888	9,522	-	75	109,257
OROMIA REGION											
Total	386,972	55,970	7,086	2,994	87,001	56,671	1,287	31,352	-	3,087	141,524
Male	128,166	25,887	3,504	1,339	36,147	20,581	986	16,136	-	1,305	22,281
Female	258,806	30,083	3,582	1,655	50,854	36,090	301	15,217	-	1,782	119,243
SOMALI REGION											
Total	20,846	1,463	363	719	4,592	2,321	530	1,664	253	171	8,770
Male	8,066	904	133	300	2,492	964	274	938	-	-	2,063
Female	12,780	559	230	419	2,100	1,357	257	726	253	171	6,707
BENISHANGUL-GUMUZ	REGION										
Total	9,731	1,283	-	119	744	2,039	72	948	-	-	4,526
Male	2,830	407	-	119	274	668	-	376	-	-	985
Female	6,901	876	-	-	470	1,370	72	572	-	-	3,541

Table 6.8 Cont'd

				Steps T	aken to Seard	ch for Work					
Region and Sex		Searching	Through news		Seeking assistance	Trying to	Direct				Not Seeking
		vacancy	paper,	I have	of friends,	establish	application	Checking			Job
		advertising	Radio	unemployment	relatives,	own	to	at work		Not	
	Total	boards	and TV	card	etc	enterprise	employee	sites	Others	Stated	
S.N.N.P REGION											
Total	142,948	40,643	4,645	3,997	17,986	21,725	4,483	13,588	3,089	1,844	30,946
Male	46,009	18,181	2,305	1,262	5,611	4,951	786	5,533	1,426	601	5,351
Female	96,939	22,462	2,340	2,736	12,375	16,774	3,697	8,055	1,663	1,243	25,594
GAMBELLA REGION											
Total	5,101	657	22	40	609	741	116	201	-	75	2,640
Male	1,511	249	22	40	40	267	-	121	-	-	773
Female	3,590	408	-	-	569	475	116	81	-	75	1,867
HARARI REGION											
Total	8,209	1,265	275	-	923	1,773	129	1,590	-	43	2,210
Male	2,716	595	187	-	371	300	129	702	-	-	432
Female	5,493	670	88	-	552	1,473	-	888	-	43	1,778
ADDIS ABABA CITY ADMIN	NISTRATION										
Total	366,227	58,976	33,913	24,865	59,234	32,256	12,801	24,340	10,846	9,307	99,689
Male	130,123	21,131	15,419	8,618	22,652	13,810	5,841	11,280	3,487	2,182	25,703
Female	236,103	37,845	18,494	16,247	36,582	18,445	6,960	13,060	7,359	7,125	73,986
DIRE DAWA ADMINISTRAT	ΓΙΟΝ										
Total	25,742	3,983	235	304	3,501	4,317	411	2,287	287	245	10,172
Male	8,587	2,202	73	304	1,543	726	251	1,298	161	-	2,029
Female	17,154	1,781	163	-	1,958	3,590	159	989	126	245	8,143

Table 6.9 Unemployed Population of Urban Areas Aged ten Years and Above by Types of Job Looking for, Region and Sex: 2015

Бу Турос	OI JOD LOOP		egion and Sex			
			e of Work Seeki	ng or Available f	or	
Region and Sex		To				
		Establish	Paid	Paid	Any	
	Total	OWN	Employment-	Employment-	Available Work	Others
COUNTRY TOTAL	Total	Business	Private	Government	VVOIK	Others
Total	1 424 227	471 467	111 167	171 101	670.016	10 155
	1,434,227	471,467	111,167	171,421	670,016	10,155
Male Female	462,220 972,007	110,823 360,644	49,634 61,532	62,116	236,180	3,466 6,689
TIGRAY REGION	972,007	300,044	01,332	109,305	433,836	0,009
Total	109,321	39,748	4,596	10,659	53,840	477
Male	27,218	7,399	1,052	2,122	16,454	191
Female	82,103	32,350	3,543	8,537	37,387	286
AFFAR REGION	02,103	32,330	3,343	0,337	37,367	200
Total	19,977	7,952	487	4,009	7,418	111
Male	6,894	2,495	361	1,273	2,655	111
Female	13,083	5,457	126	2,736	4,763	
AMHARA REGION	10,000	J, 4 J1	120	2,730	7,103	-
Total	339,154	133,133	14,362	34,201	155,728	1,731
Male	100,100	31,995	7,055	11,979	48,479	591
Female	239,055	101,138	7,306	22,221	107,250	1,139
OROMIA REGION	200,000	101,100	7,000	22,221	107,200	1,100
Total	386,972	134,826	19,324	37,362	193,615	1,844
Male	128,166	29,708	9,735	12,171	76,015	536
Female	258,806	105,118	9,589	25,192	117,600	1,308
SOMALI REGION	200,000	100,110	3,303	20,102	117,000	1,000
Total	20,846	6,348	1,760	3,096	9,642	_
Male	8,066	1,286	562	1,513	4,706	_
Female	12,780	5,062	1,198	1,583	4,937	_
BENISHANGUL-GUMU		0,002	1,100	1,000	1,001	
Total	9,731	4,533	541	988	3,668	_
Male	2,830	1,192	214	402	1,021	_
Female	6,901	3,342	327	586	2,647	_
S.N.N.P REGION	2,001	-,- :-			_,-,-	
Total	142,947	35,894	9,618	28,390	68,504	541
Male	46,009	8,080	3,756	11,537	22,335	300
Female	96,939	27,814	5,862	16,853	46,169	241
GAMBELLA REGION	,	,-	-,	-,	-,	
Total	5,101	937	237	862	3,013	53
Male	1,511	320	80	386	725	_
Female	3,590	617	156	476	2,287	53
HARARI REGION						
Total	8,209	3,050	301	1,083	3,732	43
Male	2,716	580	132	329	1,631	43
Female	5,493	2,470	170	754	2,100	-
ADDIS ABABA CITY A	DMINISTRATION	NC				
Total	366,227	95,326	57,964	48,242	160,249	4,446
Male	130,123	26,108	25,499	19,755	57,221	1,540
Female	236,103	69,218	32,464	28,487	103,028	2,906
DIRE DAWA ADMINIS		•	•			•
Total	25,742	9,719	1,978	2,528	10,607	910
Male	8,587	1,660	1,188	649	4,938	153
Female	17,154	8,059	790	1,880	5,669	756

Table 6.10 Unemployed Population of Urban Areas Aged Ten Years and Above by Previous Work Experience, Duration of Unemployment, Age Group and Sex - COUNTRY TOTAL: 2015

		Previous Wo	rk Experience		1017		Dura	ition of Une	emplovme	nt (In Moi	nths)				
A O	Total	Unemployed	_,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,				2 4.0			(
Age Group	Unemployed	Without	Unemployed												
and Sex	Persons	Work	With Work												
		Experience	Experience	Total	< 1	1-6	7-12	13-24	25-36	37-48	49-60	61-72	73-84	85-95	96+
Λ.ΙΙ. Λ.σ.σ.σ.															
All Ages	4 40 4 007	707.440	007.004	4 404 007	445.000	057.500	005 404	404.000	00.000	00.077	04.050	40.400	5 4 4 4	4.000	000 444
Total	1,434,227	737,146	697,081	1,434,227	115,628	357,568	,	161,330	89,233	,	21,059	,	,	4,283	398,444
Male .	462,220	234,318	227,902	462,220	35,333	141,978	82,919	43,962	24,417	6,972	7,241	4,240	935	691	113,532
Female	972,007	502,827	469,180	972,007	80,295	215,590	152,512	117,368	64,815	25,705	13,819	9,193	4,206	3,592	284,912
10-14															
Total	24,390	19,497	4,893	24,390	2,658	14,743	1,571	1,658	62	61	-	-	-	-	3,636
Male .	9,146	6,804	2,341	9,146	1,077	4,735	1,204	533	-	-	-	-	-	-	1,597
Female	15,244	12,692	2,552	15,244	1,581	10,009	368	1,125	62	61	-	-	-	-	2,039
15-19															
Total	204,546	145,744	58,802	204,546	19,220	63,279	41,203	22,374	9,538	650	780	2,164	416	688	44,234
Male	71,926	50,984	20,942	71,926	3,920	29,254	13,784	5,212	3,509	211	438	-	102	-	15,497
Female	132,621	94,760	37,860	132,621	15,300	34,025	27,419	17,162	6,029	439	342	2,164	314	688	28,737
20-24															
Total	418,460	246,146	172,314	418,460	29,458	109,681	81,339	49,136	25,794	7,632	5,053	2,288	1,752	1,023	105,306
Male	137,166	87,006	50,161	137,166	9,324	40,416	27,358	14,225	6,793	1,846	1,151	1,777	521	-	33,755
Female	281,294	159,140	122,154	281,294	20,134	69,265	53,981	34,911	19,001	5,786	3,903	511	1,231	1,023	71,550
25-29															
Total	307,539	150,703	156,837	307,539	17,169	69,248	52,411	38,418	22,057	11,403	5,651	3,517	515	1,108	86,044
Male	104,586	55,545	49,041	104,586	6,666	25,413	18,719	11,054	7,851	1,258	3,831	343	40	336	29,075
Female	202,953	95,158	107,796	202,953	10,503	43,836	33,691	27,364	14,206	10,145	1,820	3,174	474	772	56,969
30-34															
Total	152,036	65,583	86,453	152,036	14,583	32,016	21,127	19,183	9,034	4,012	4,061	2,222	866	355	44,577
Male	42,939	14,673	28,266	42,939	4,875	13,236	9,715	3,182	1,742	617	706	1,778	73	355	6,660
Female	109,097	50,910	58,187	109,097	9,708	18,780	11,412	16,002	7,292	3,395	3,355	444	793	-	37,916
35-39															
Total	116,225	43,768	72,458	116,225	11,279	20,812	13,920	12,087	10,101	2,684	2,466	1,180	73	342	41,281
Male	26,618	7,413	19,205	26,618	2,295	7,530	1,986	3,820	1,217	507	450	343	73	-	8,398
Female	89,607	36,355	53,252	89,607	8,984	13,282	11,934	8,267	8,884	2,178	2,016	838	-	342	32,884

Table 6.10 Cont'd

Table 0.10	Conta	Drovious Mo	ous Work Experience Duration of Unemployment (In Months)												
	Total		rk Experience	T	1		Duia	ation of One	employme	iii (iii ivioi	11115)	1	ı		
Age Group	Unemployed	Unemployed Without	Unemployed												
and Sex	Persons	Work	With Work												
	. 0.000	Experience	Experience	Total	< 1	1-6	7-12	13-24	25-36	37-48	49-60	61-72	73-84	85-95	96+
40-44	•	•			•	•			'			•			<u>'</u>
Total	55,802	18,952	36,850	55,802	4,658	14,363	5,746	4,541	3,048	2,183	936	98	709	142	19,378
Male	17,988	3,778	14,210	17,988	1,358	5,973	2,636	754	1,287	322	313	-	-	-	5,346
Female	37,815	15,174	22,640	37,815	3,300	8,391	3,110	3,786	1,761	1,861	624	98	709	142	14,033
45-49															
Total	54,645	13,660	40,985	54,645	5,183	10,509	5,254	5,177	4,607	1,595	720	952	515	625	19,509
Male	14,044	1,512	12,532	14,044	1,143	5,721	2,458	1,808	59	536	118	-	-	-	2,203
Female	40,601	12,147	28,453	40,601	4,041	4,787	2,797	3,369	4,548	1,059	602	952	515	625	17,307
50-54															
Total	30,941	8,458	22,483	30,941	3,311	4,854	5,022	4,231	1,542	428	754	343	171	-	10,285
Male	7,974	715	7,258	7,974	2,317	1,431	610	1,812	1,140	187	61	-	-	-	415
Female	22,967	7,742	15,225	22,967	993	3,423	4,412	2,418	402	241	693	343	171	-	9,870
55-59															
Total	31,904	11,675	20,229	31,904	3,077	7,152	4,669	1,665	1,548	814	638	669	-	-	11,673
Male	10,468	1,629	8,839	10,468	642	2,574	1,787	989	102	393	174	-	-	-	3,805
Female	21,436	10,046	11,390	21,436	2,435	4,578	2,882	675	1,445	421	464	669	-	-	7,867
60-64															
Total	14,841	4,087	10,754	14,841	2,187	2,767	1,165	1,260	992	120	-	-	-	-	6,351
Male	6,664	752	5,912	6,664	1,020	1,311	700	418	628	-	-	-	-	-	2,586
Female	8,177	3,335	4,841	8,177	1,167	1,456	465	842	363	120	-	-	-	-	3,764
65+															
Total	22,897	8,873	14,024	22,897	2,847	8,144	2,004	1,601	910	1,096	-	-	125	-	6,171
Male	12,702	3,506	9,195	12,702	695	4,385	1,963	153	88	1,096	-	-	125	-	4,196
Female	10,195	5,367	4,828	10,195	2,151	3,759	41	1,448	822	-	-	-	-	-	1,975

Table 6.11 Unemployed Populations of Urban Areas Aged Ten Years and Above by Previous Work Experience, Duration of Unemployment, Region and Sex - COUNTRY TOTAL: 2015

	Total	Previous Wo	Previous Work Experience Duration of Unemployment (In Months)												
Region and Sex	Unemployed Persons	Unemployed Without Work	Unemployed With Work	-	_		7.10	40.04	25.00	07.45	40.00	04.76	70.0	05.05	
		Experience	Experience	Total	< 1	1-6	7-12	13-24	25-36	37-48	49-60	61-72	73-84	85-95	96
COUNTRY TOTAL															
Total	1,434,227	737,146	•	1,434,227	•	357,568	235,431	161,330	89,233	32,677	21,059	13,433	5,141	4,283	1
Male	462,220	234,318	227,902	462,220	35,333	141,978	82,919	43,962	24,417	6,972	7,241	4,240	935	691	113,
Female	972,007	502,827	469,180	972,007	80,295	215,590	152,512	117,368	64,815	25,705	13,819	9,193	4,206	3,592	284,
TIGRAY REGION															
Total	109,321	54,649	54,672	109,321	8,657	23,362	26,709	17,368	6,717	2,702	329	929	90	-	22,
Male	27,218	13,446	13,772	27,218	1,647	8,217	6,483	3,255	1,545	1,049	-	343	-	-	4,
Female	82,103	41,203	40,900	82,103	7,010	15,144	20,226	14,113	5,172	1,653	329	585	90	-	17,
AFFAR REGION															
Total	19,977	9,501	10,476	19,977	2,512	5,538	3,097	3,671	2,803	352	142	-	-	-	1,
Male	6,894	3,682	3,212	6,894	681	1,997	1,212	1,845	881	24	-	-	-	-	
Female	13,083	5,819	7,264	13,083	1,831	3,541	1,885	1,826	1,922	329	142	-	-	-	1,
AMHARA REGION															
Total	339,154	165,509	173,645	339,154	16,594	76,251	69,168	46,675	23,833	10,106	5,905	4,554	1,729	1,954	82,
Male	100,100	45,711	54,389	100,100	6,099	31,039	21,945	9,674	3,597	481	2,200	750	102	355	23,
Female	239,055	119,798	119,257	239,055	10,494	45,212	47,222	37,001	20,236	9,625	3,705	3,804	1,627	1,599	58,
OROMIA REGION															
Total	386,972	186,012	200,960	386,972	23,056	92,299	58,766	40,718	17,980	9,819	5,994	3,042	1,479	523	133,
Male	128,166	56,935	71,231	128,166	5,453	37,369	22,525	11,915	5,628	2,162	3,661	837	646	-	37,
Female	258,806	129,077	129,729	258,806	17,603	54,930	36,241	28,803	12,351	7,657	2,333	2,206	833	523	95,
SOMALI REGION															
Total	20,846	11,693	9,153	20,846	2,981	7,211	2,561	1,845	771	203	528	363	228	-	4.
Male	8,066	3,987	4,079	8,066	804	2,672	1,279	639	670	153	118	73	187	-	1,
Female	12,780	7,706	5,074	12,780	2,177	4,539	1,282	1,206	101	50	411	289	40	-	2,
BENISHANGUL-GU	IMUZ REGION														
Total	9,731	3,876	5,855	9,731	446	3,279	1,700	1,303	371	222	105	29	-	-	2,
Male	2,830	1,113	1,716	2,830	369	1,281	406	277	116	129	-	-	-	-	
Female	6,901	2,763	4,138	6,901	77	1,998	1,294	1,026	255	92	105	29	-	-	2,

Table 6.11 Cont'd

	Total	Previous Wo	rk Experience				Dura	ation of U	nemploy	ment (In	Months)				
Region and Sex	Unemployed Persons	Unemployed Without Work Experience	Unemployed With Work Experience	Total	< 1	1-6	7-12	13-24	25-36	37-48	49-60	61-72	73-84	85-95	96+
S.N.N.P. REGION					ı		ı								
Total	142,947	80,945	62,002	142,947	12,236	39,733	23,841	11,930	9,628	2,543	2,607	1,122	455	95	38,758
Male	46,009	26,104	19,904	46,009	3,839	14,023	10,233	3,468	2,449	651	96	1,027	-	-	10,222
Female	96,939	54,841	42,098	96,939	8,397	25,710	13,608	8,461	7,179	1,892	2,512	95	455	95	28,536
GAMBELLA REGIO	N														
Total	5,101	2,579	2,522	5,101	579	2,038	438	151	70	88	49	-	40	-	1,649
Male	1,511	657	855	1,511	41	765	136	41	-	-	-	-	-	-	529
Female	3,590	1,923	1,667	3,590	538	1,274	301	110	70	88	49	-	40	-	1,120
HARARI REGION															
Total	8,209	2,996	5,213	8,209	1,109	1,665	2,187	1,420	325	293	-	99	-	-	1,112
Male	2,716	939	1,777	2,716	279	712	417	618	196	152	-	-	-	-	342
Female	5,493	2,058	3,436	5,493	830	953	1,770	802	129	141	-	99	-	-	770
ADDIS ABABA CITY	' ADMINISTRATION	ON													
Total	366,227	211,112	155,114	366,227	42,864	99,015	42,462	33,903	25,025	4,998	4,944	3,113	673	1,711	107,519
Male	130,123	78,565	51,559	130,123	15,426	40,791	16,294	11,279	8,736	1,905	1,074	1,210	-	336	33,073
Female	236,103	132,548	103,556	236,103	27,438	58,225	26,167	22,625	16,289	3,093	3,870	1,903	673	1,375	74,446
DIRE DAWA ADMIN	IISTRATION														
Total	25,742	8,273	17,469	25,742	4,594	7,177	4,503	2,346	1,710	1,351	455	183	447	-	2,975
Male	8,587	3,179	5,408	8,587	694	3,111	1,988	950	598	266	93	-	-	-	888
Female	17,154	5,094	12,061	17,154	3,901	4,066	2,515	1,396	1,112	1,086	363	183	447	-	2,087

Dus	 	gion and	1 36x - CO	<u> </u>			o otort ::	ur bugingss				
					Proble	m raced t		ur business		I		
							Lack of					
Region and sex			01 .			Lack of	working		.			
3		I have	Shortage		lack of	finance	place	Lack	Shortage/			
		no	of	Lack of	work	and	and	/absence	absence of	Don't	Lack of	
	Total	problem	finance	training	place	training	finance	of license	equipment	know	information	Others
COUNTRY TOTAL												
Total	471,467	34,420	230,685	4,654	61,714	17,361	77,909	4,832	895	870		33,462
Male	110,823	9,307	56,879		10,317	7,231	17,405	1,359	355	242	711	5,179
Female	360,644	25,113	173,807	2,815	51,397	10,130	60,504	3,473	540	628	3,954	28,283
TIGRAY REGION												
Total	39,748	4,081	26,514	288	3,328	-	1,817	256	-	-	-	3,464
Male	7,399	831	4,660	135	-	-	813	256	-	-	-	704
Female	32,350	3,250	21,854	153	3,328	-	1,004	-	-	-	-	2,760
AFFAR REGION												
Total	7,952	671	4,643	127	725	359	339	110	-	110	110	758
Male	2,495	218	1,717	111	119	331	-	-	-	-	-	-
Female	5,457	454	2,926	16	606	28	339	110	-	110	110	758
AMHARA REGION	,		,									
Total	133,133	11,651	57,947	614	20,682	1,351	27,094	2,419	710	-	1,155	9,510
Male	31,995	4,681	15,926	614	3,727	352	3,938	446	355	-	-	1,955
Female	101,138	6,970	42,021	-	16,955	999	23,155	1,972	355	-	1,155	7,555
OROMIA REGION	,	,	•		,		,	,			,	,
Total	134,826	7,535	70,410	2.171	14,968	5,016	24,198	301	185	-	116	9,927
Male	29,708	1,434	14,808	604	2,979	2,259	7,031	-	-	_	_	594
Female	105,118	6,101	55,602		11,989	2,757	17,167	301	185	_	116	9,332
SOMALI REGION	,	-,	,	1,000	,	_,	,					-,
Total	6,348	235	5,417	_	40	_	233	_	_	156	81	186
Male	1,286		1,182	-	-	-		_	_	104	-	-
Female	5,062	235	4,235	_	40	_	233	_	_	52	81	186
BENISHANGUL-GU			.,200				200			02	01	, 00
Total	4,533	359	2,192	68	452	352	209	42	_	116	_	743
Male	1,192	129	828	39	77	-	77	42	_		_	5
Female	3,342	229	1,365	29	376	352	132	-	_	116	_	743
i ciliale	5,542	223	1,505	23	570	552	132	_	_	110	_	173

Table 6.12 Cont'd

Table 6.12 Cont d												
					Proble	m faced t	o start yo	ur business				
							Lack of					
Dogion and say						Lack of	working					
Region and sex		I have	Shortage		lack of	finance	place	Lack	Shortage/a		Lack of	
		no	of	Lack of	work	and	and	/absence	bsence of	Don't	informatio	Other
	Total	problem	finance	training	place	training	finance	of license	equipment	know	n	S
S.N.N.P. REGION												
Total	35,894	1,688	18,516	-	4,427	3,466	2,280	803	-	301	2,080	2,333
Male	8,080	-	5,375	-	374	1,744	492	-	-	-	96	-
Female	27,814	1,688	13,142	-	4,053	1,722	1,788	803	-	301	1,985	2,333
GAMBELLA REGIO	N											
Total	937	97	388	-	-	121	75	-	-	49	164	43
Male	320	-	204	-	-	40	75	-	-	-	-	-
Female	617	97	184	-	-	81	-	-	-	49	164	43
HARARI REGION												
Total	3,050	490	1,307	-	513	-	434	-	-	49	-	257
Male	580	83	179	-	97	-	83	-	-	49	-	90
Female	2,470	408	1,127	-	417	-	351	-	-	-	-	167
ADDIS ABABA CITY	ADMINI:	STRATIO	N									
Total	95,326	6,632	38,099	1,386	15,223	6,330	20,040	902	-	-	958	5,755
Male	26,108	1,931	11,012	336	2,689	2,414	4,804	615	-	-	615	1,691
Female	69,218	4,701	27,087	1,050	12,534	3,916	15,236	287	-	-	343	4,064
DIRE DAWA ADMIN	NSTRATION NECESTRATION NECESTRA	ON										
Total	9,719	982	5,252	-	1,356	365	1,191	-	-	88	-	486
Male	1,660	-	988	-	255	90	93	-	-	88	-	146
Female	8,059	982	4,263	-	1,100	275	1,099	-	-	-	-	340

Annex I- III

Annex I Survey Questionnaire

Annex II Estimation Procedures of Total Ratio and Sampling Errors

Annex III Estimates of CV's for selected Tables

CENTRAL STATISTICAL AGENCY URBAN EMPLOYMENT UNEMPLOYMENT SURVEY - 2015

				105 Keftegna/Wereda/							112 Is there continued questionnaire?
101 Kill	102 Zone	103 Wereda	104 Town	K/Ketema	106 Kebele	107 Enumeration Area code	108 Household selection Ser.No	109 Name of Head of HH	110 Branch Office	111- Serial No of Enumeration Area	
											1. Yes 2. No

	HOUR	N	AINUT	E
INTERVIEW BEGIN				
INTERVIEW END				

CONFIRMATION

	Name	Signature	Date	Code
Enumerator			/ /	
supervisor			/ /	
Statistician			/ /	

THE CENTRAL STATISTICAL AGENCY URBAN EMPLOYMENT - UNEMPLOYMENT SURVEY - 2015

SECTION 1. AREA IDENTIFICATION OF THE SELECTED HOUSEHOLD

(For Office Use Only)

101 Kilil	102 Zone	103 Wereda	104 Town	105 Keftegna/wereda/ K/Ketema	106 Kebele	107 Enumeration Area code	110 - Branch Office	111- Serial No of Enumeration Area

For office work only (Branch Office)

	NAME	SIGNATURE	DATE	CO	DE
Enumerator					
supervisor					
Statistician					
Branch Head					

For office work only (Head Office)

	NAME	SIGNATURE	DATE	CO	DE
Editor					
Editor-verifier					
Encoder					
Encoder-verifier					

THE CENTRAL STATISTICAL AGENCY URBAN EMPLOYMENT - UNEMPLOYMENT SURVEY - 2015

ECTION											
1 Kilil	102 Zone	103 Wereda	104 Town	105 Keftegna/wereda/ K/Ketema	106 Kebele	107 Enumeration Area code	108 Household selection ser.no	109 Name of Head of HH	110 Branch Office	111- Serial No of Enumeration Area	112 Is there c
											1. Yes 2
-1			1 1						•		•
CPT ON					0.5.5						
CTION	1. AREA IDENTI	IFICATION O	F THE SELE	CTED HOUSEH	OLD					page 4 of 5 pages	
CTION	1. AREA IDENTI	IFICATION O	F THE SELE		OLD		T	T	T	page 4 of 5 pages	7
				105 Keftegna/wereda/		107 Enumeration Area code	108 Household selection set no	100 Name of Head of HH	110 Branch Office	112- Serial No of	7
	1. AREA IDENTI	IFICATION O	F THE SELE		OLD 106 Kebele	107 Enumeration Area code	108 Household selection ser.no	109 Name of Head of HH	110 Branch Office]
				105 Keftegna/wereda/		107 Enumeration Area code	108 Household selection ser.no	109 Name of Head of HH	110 Branch Office	112- Serial No of	
				105 Keftegna/wereda/		107 Enumeration Area code	108 Household selection ser.no	109 Name of Head of HH	110 Branch Office	112- Serial No of	
				105 Keftegna/wereda/		107 Enumeration Area code	108 Household selection ser.no	109 Name of Head of HH	110 Branch Office	112- Serial No of	
				105 Keftegna/wereda/		107 Enumeration Area code	108 Household selection ser.no	109 Name of Head of HH	110 Branch Office	112- Serial No of	
			104 Town	105 Keftegna/wereda/ K/Ketema	106 Kebele			109 Name of Head of HH		112- Serial No of Enumeration Area	
				105 Keftegna/wereda/ K/Ketema		107 Enumeration Area code	108 Household selection ser.no	109 Name of Head of HH	110 Branch Office	112- Serial No of Enumeration Area	
Kilil			104 Town	105 Keftegna/wereda/ K/Ketema	106 Kebele			109 Name of Head of HH		112- Serial No of Enumeration Area	
Kilil			104 Town	105 Keftegna/wereda/ K/Ketema	106 Kebele			109 Name of Head of HH		112- Serial No of Enumeration Area	
Kilil			104 Town	105 Keftegna/wereda/ K/Ketema	106 Kebele			109 Name of Head of HH		112- Serial No of Enumeration Area	
Kilil	102 Zone	103 Wereda	104 Town	105 Keftegna/wereda/ K/Ketema	106 Kebele			109 Name of Head of HH		112- Serial No of Enumeration Area	
Kilil		103 Wereda	104 Town	105 Keftegna/wereda/ K/Ketema	106 Kebele			109 Name of Head of HH		112- Serial No of Enumeration Area	
Kilil	102 Zone	103 Wereda	104 Town	105 Keftegna/wereda/ K/Ketema	106 Kebele			109 Name of Head of HH		112- Serial No of Enumeration Area	
CTION	102 Zone 1. AREA IDENTI	IFICATION O	F THE SELE	105 Keftegna/wereda/ K/Ketema CCTED HOUSEHO 105 Keftegna/wereda/	106 Kebele					page 5 of 5 pages 112- Serial No of Enumeration Area	
Kilil	102 Zone	IFICATION O	104 Town	105 Keftegna/wereda/ K/Ketema	106 Kebele			109 Name of Head of HH 109 Name of Head of HH		page 5 of 5 pages	

page 1 of 5 pages

SECTION 2: SOCIO-DEMOGRAPHIC CHARACTERSTICS OF HOUSEHOLD

RELATIONSHIP

What is Your

0= Head

1= Spouse

2= Son/daughter of

head and spouse

3= Son/daughter of

4= Son/daughter of

5= Mother/father of

head/spouse

6= Sister/brother of

7 = Paid Domestic

head/spouse

Workers

8= Other relatives

9= Non-relatives

spouse

Relationship to the

Head of Household?

For all memebers of the Household

203

201

Serial

202

Full name of usual household

members

-Head

-Spouse

according

who live

List names of usual

household members

in the following order:

-Never married children

to seniority in age

-Married children and their

household members

in this housing unit

-Other relatives

- Non relatives

204 205

SEX AGE

208

What was the

Hghiest Grade

the list of answers and codes are given in the

You have

Completed

209

Have you

received any

professional

or Vocational/

Technical

training?

1= Yes

2= No

Go to Col. 211

For members aged 10 years

TRAINING

In what type

of profession

(Vocational/

technical) are

you trained?

list of

and codes

are provided

in the manual

answers

and over

207

For members aged 5

Have you ever

Attended

Formal/

informal

Education?

1 = Yes, Iam

Attending

This Year

2 . ሥስቋ = ህ

befor this

Year

3 = Never

Go to col. 209

attended

attended

years and above

EDUCATIONAL STATUS

Can you

read and

write?

1= Yes

2= No

SECTION 3: ECONOMIC ACTIVITY DURING THE LAST 7 DAYS (For members aged 5 years and above) Last 7 days refers to the Seven Days Prior to the Date of Interview. 303 304 305 306 307 301 ECONOMIC ACTIVITY STATUS If code 1 -During the last 7 days were you engaged in

During the last 7 days were you engaged in							Enumerator:-	<u>:</u> I	Even though you w	vere	<i>col.30</i>
any kind of productive activities such as							Check in	r	not working during	g the	What was the main
work for payment,for family gain or	sqo				job		Col.302 total		last 7 days, did yo	ou	reason for not
profit, produce for own consumption?	at all jobs				additional job		hours	ł	nave a job, busines	ss	working during thε
Enumerator:- Ask respondents "What they have	work a						worked in the	C	or holding which y	⁄ou	last 7 days?
been doing in the last 7 days" then	you w				is no		last 7 days?	V	will return to?		
mark code "1" if the respondent was					If there is						1. Sick, Injury
engaged in at least one of the activities mentioned below.	hours did		job?				1= Less than				2. Holiday/ vaca
Mark code 2 if participated in none of the activities.	nany		nary.		al job		4 hours or not	1	I= Yes, I have own		3. Training/
- Work as paid employee for Government/ gov't -	how 1		main/primary job?	1	additional job/s?		asked in col.302		business/enterpris	se	Education
parastatal, domestic, NGO, Private organization?	al for		mair		ur adı			2	2= Yes, paid for dur	ration	4. Off season
- Run a business, big or small, for self or with	in total for ho me)		your		in your		2= 4 hours		of absence		5. No work
partners (e.g barber, shoe shining, tradeetc.)?	time,		spend in your		spend		and above	7	3= Yes, with assurance	ice	temporarily
- Work in Agriculture privately/salaried (e.g	luding lunch and journey time, in to	0	eds r		you			/	or agreement for		6. Other/Specify
ploughing and Cattle rearing, poultry,etc.)?	nd jou		id you		did			′	returning to worl	k.	
- Produce goods for sale (e.g 'injera','Tella"Sefied'etc.)?	nch an		nany Hours did you		Hours "98"		Go to	4	4= No	7	
- Produce permanent goods for own family?	ng lun		y Hor		many code		Col. 308		•		
- Help in family farm or buisnes without payment ?	scludi the la		2		How fill in			(Go to Col.4	01	
eg. In agriculture,trade or in other enterprises.	Exc]		How								
- Engage in construction or major repair for own/family use?											
							1				
-other productive activity not mentioned above	Week's Hou	rs in	Week's Hou	ırs	Week's H	ours					
	Week's Hou all Producti		Week's Hou		Week's H						
-other productive activity not mentioned above		ve									
-other productive activity not mentioned above for own or family gain?	all Producti	ve	in primary/		in additio						
-other productive activity not mentioned above for own or family gain?	all Producti	ve	in primary/		in additio						
-other productive activity not mentioned above for own or family gain?	all Producti	ve	in primary/		in additio						
-other productive activity not mentioned above for own or family gain?	all Producti	ve	in primary/		in additio						
-other productive activity not mentioned above for own or family gain?	all Producti	ve	in primary/		in additio						
-other productive activity not mentioned above for own or family gain?	all Producti	ve	in primary/		in additio						
-other productive activity not mentioned above for own or family gain?	all Producti	ve	in primary/		in additio						
-other productive activity not mentioned above for own or family gain?	all Producti	ve	in primary/		in additio						
-other productive activity not mentioned above for own or family gain?	all Producti	ve	in primary/		in additio						
-other productive activity not mentioned above for own or family gain?	all Producti	ve	in primary/		in additio						
-other productive activity not mentioned above for own or family gain?	all Producti	ve	in primary/		in additio						
-other productive activity not mentioned above for own or family gain?	all Producti	ve	in primary/		in additio						
-other productive activity not mentioned above for own or family gain?	all Producti	ve	in primary/		in additio						
-other productive activity not mentioned above for own or family gain?	all Producti	ve	in primary/		in additio						
-other productive activity not mentioned above for own or family gain?	all Producti	ve	in primary/		in additio						

211

MARITAL

STATUS

1=Never

married

2=Married

3= Divorced

4= Widowed

5=Separated

210

3 in 6

tion

______06
______07
______08
______09
______10
_____11

202	201	318	· · · · · · · · · · · · · · · · · · ·	319	320	401	402	403	404	CTERISTICS OF UNI 405	406	407	408	409	410	41
202	201	Informal Sector of F	conomy	319	320	1	What step have	For those who	If opportunities	Are you willing to take up	Why are you not	For those who answered	For those who	Have you	For how	How do you
		If coded 03, 06, 08- 12	· · ·	UNDEREMPLOYM	ENT	For those coded 4 In Col. 306	you taken mainly	answered	to work exist	work for wage or salary on	available for	Code ''2 - 6'' in Col. 405	answered	ever done	many months	support
		The organization's /S	ector's	Were you seeking and available to work	work sought or available for?	for work or try to	of work or to start your own business?	in Col. 401	one month are you willing	to undertake self-employment activity given the necessary		What type of job are you looking for?	What are/were the	past for pay or profit?	you been unemployed	your self?
'ull Name		X 5	 	for additional	available for .	establish your	your own business.	What was the	and ready to	resources and facility? If the	0= Home maker	die you looking for.	problems you faced	pront.	ployed?	
		_		hours in the last	1 =More hours at	own business		reason that	work for	answer is "Yes" where do you	1= Pregnancy/	Enumerator:	to establish your		project	1= Casual Worl
nsfer from page1, Col.202)		n you		7 days?	present job 2 =Other job(s) in	during the last 3 months?	Alternative	you didn't seek or try to	income/ earning?	prefer the place of job to be?	delivery 2= Student	Read the alternative answers if need	own Business/ Enterprise?	1. No		2= Family Supp 3=Support from
ousehold members are		izatio	nce?	For those not worked Were you seeking	addition to the present job	10	(answers are indicated	establish your own		1. Not available	3= Disabled 4= Illness/ Injury	arises. 1. Self employment		2. Yes, as employee3. Yes, for family		Relatives& Frie 4=Saving
ferred in this column		organ ket? ook a	a lice	and available to work	3= Other job(s) with	1= Yes	below	business?	1= Yes	1. Not available	5= Too young	1. Sen employment		4. Yes, for own		5= Remittance
Col. 401 - 507 shall be	ıber	f the c	have	in the last 7 days?	more hours to					2. Within this dwelling.	6=Remmitance	2. Paid employment-private				6=Begeging
ked "" for members I below 10 years.	unu e	in for in for in for ise ha	orise l		replace present one	2= No >		Alternative	2= No 1	3. Within residence village/	7= Old age/ Pensioned	3. Paid employment-Gov't4. Any available work	See alternatives answer below	If more than one,refers to		7= Others
i below 10 years.	Line	t/Serviged	enter					answers are	\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \	town only 4. Only in Urban areas of this	8= Others	5. Others	answer below	the recent one.		
		engs engs	the 6	1=Yes	Go to	Go to		indicated) 	country	(Specify)					
		he pr	Does		Col.501	Col. 403	All go to	below	Go to Col.406	5. Any where in the country		Skip to Col. 409				
		Does 1		2- No. 5	Consor		col. 404	55.01	3010 001.400							
		-		2= No			101. 101			6. Overseas only						
				」						skip to Col. 407	Co to Col.501					Go to Col.
		1. Yes		Go to												
		2. No		Col.502												
	- 	3. Don't know														
	$\begin{vmatrix} & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & & \\ & & & & & \\ & & & & & \\ & & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & & \\ & \\ & \\ & \\ & & \\ & $	A B	С		_											
	0 1															
	, , <u>-</u>															
	0 3															
	0 4															
	0 5															
															_	
	0 6															
	0 7															
	0 /															
	0 8															
	0 9															
	1 0															
	1 1															
								Colu	mn 403				Column 408			
					Column 402		01. Pregnancy/deliver	у				have no problem		07. Shortage/abs	ence of	
				 Searching vacancy adv Through News paper, 			02. Illness/Injury03. Personal/family re	enoncihility	previous work 09. Thought no w			hortage of finance ack of training		equipment 08. Don't know		
				2. I nrough News paper,3. I have unemployment			03. Personal/tamily re 04. Responsibility of h			ork avanable te work, I thought there will		ack of training roblem of working place/la	and	09. Lack of info		
				4. Seeking assistance of			05. Old age/Pension	-		f money, raw materialetc.	04. L	ack of finance and training		10. Others /Spec	ify/	
				5. Trying to establish own	_		06. Education/Trainin		11. Too young			ack of working place & fin	ance			
				6. Direct application to e7. Checking at work sites			07. Already found/ma arrangement for we		12. Remittance	ing that it is the role of Men	[06. L	ack/absence of License				

SECTION 5: ECONOMIC ACTIVITY DURING THE LAST 6 MONTHS (For members age 5 years and above)

202	201	501	502	503	504	505	506	507
				ECONOMIC AC	TIVITY STATUS			
		Enumerator:- Ask respondents the following questions		LCONOMIC AC				
Full Name		separatelyand mark code "1" and list the activity (ies)	Durig the last	ç·	orking?	ally	<u>Interviewer :-</u>	If code 1 is filled in Col.506
		if the respondent was engaged in at least one of the	6 months have	king	WOI	mic	Check in Col. 505	What was the main
Transfer from page1, Col.202)		activities.Mark code 2 if participated in 'none of the activity	have you ever	working?	not	Conc	total number of weeks	reason for not working
		During the last 6 months, did you engage in any work	looked for work	you y	noń	on e	economically not active is	or not being availabe
		or in any work for pay or profit or family gain?	or been	re y	ere	e yo		for work for
All household members are			available to work	8	8 83	wer	1. 12 weeks or more	most of the
transferred in this column		- Did you work as an employee for Government/	in some	eeks	veek	eks		last 6 months?
but Col. 401 - 507 shall be		Private enterprise?	productive	8 8	n śu	, we		
marked "_" for members		- Did you work as merchant? (including petty trade)	activity?	any	ma	nany	2.Less than 12 weeks	0= Pregnancy/
aged below 10 years.		- Did you work as service giving agent be it private or		∃ 3	WOL	ğ ü	1 1	delivery
		salaried (such as barber, shoe shining,etc.)?		· ho	for 1	r ho	1 1	1. Home maker
		- Did you work in Agriculture privately/salaried (such as	1= Yes	for	hs	s fo	♥	2. Student
				months	months	nth	END	
		plowing and, Cattle rearing, poultry,etc.)?	, , , , , , , , , , , , , , , , , , ,			6 mc		3. Disablity
		- Did you produce goods for sale (such as "injera", "Tella"etc)	Go to Col.504	st 6	last 6	ts.		4. Illness
		- Did you produce permanent goods for your family?		e la	le la	ie la		5. Too young
		- Did you engage in productive activity for your family	2=No	During the	During the	During the I not active?		6. Remittance
		without payment?	1 1	urir	uri	urin ot ac		7. Pensioned/old age
		- Other productive activity not mentioned above?	<i> </i>	Ω		D Di		8. Other /Specify
			•		1		Ⅎ	
		1.Yes Co to Col.503	Go to Col.505	Weeks	Weeks	Weeks		END
		2 .No		Worked	Unemployed	not active		
	01							
	02							
	03							
	0.5	<u> </u>						
		l —						
	04							
		l —						
	05							
	0.0							
	06							
	07							
	<u> </u>							
	08							
		<u> </u>						
				_	_	_		
	09							
	10							
]			<u> </u>			
]						
	1 11	- 1						

(For members aged 5 years and above)

202	201	308	309	310	311	312	313	314	315	316	317
Full Name Transfer from page1		For those who worked		code 2 in col. 305) or for those to(code 1-3 in col.306)	who did not work but had	For those who Terms of	answered co	ode 01 - 06 in	If code 1 - 4	is filled in Col.314	If code 01 - 06 in Col.311
, Col. 202)	Col.201)	OCCUPATION, IN	NDUSTRY, EMPLOYME	NT STATUS AND WORK PLACE	E FOR THE MAIN JOB	Employment		EARN	INGS FROM	PAID EMPLOYM	ENT
All household members are transferred in this column; but Col.308 - 320	Line number (Transfer from page 1, Col	What was your main occupation? Note:- If a person was not	Where did you engage your work? 00= Business house 01= Office 02= At home 03= On street	What was the major product or service of this organization? Note:- If a person was not	parastatal 03= Employee - Private Orgn. 04= Employee- NGO or	What are your terms of employment in your main			What was the number of () you	Based on Col.314 what was the amountpaid in your main	What was the total amount paid in your main occupation during the last month?
shall be marked '''' for members aged below 10 years.	rT)	engaged at work in the last 7 days, then the occupation before that period will be asked	04= 'Gulit'/Open market 05= Farm area/field 06= Factory 07= Quarrying/mining 08=Any where as found 09= Other /specify/	engaged at work in the last 7 days, then the product/service of the organization before that period will be asked.	(Int.Organization) 05=Employee - domestic 06= Other employees 07= Member of	occupation? 1. Permanent employee 2. Temporary employee 3. Contoployee 4. Casual worker 5. Other /specify	your main paid emplo- yment? 1= In cash 2= In kind 3= Both	job? 1= Hourly 2= Daily 3= Weekly 4= Half month 5= Monthly 6= Yearly 7= Other/ Specify	were paid for during the last month?	occupation during that last pay period?	
					12=Others(specify) Go to Col.318			Go to Col.317			
	0 1										
	0 2										
	0 3										
	0 4										
	0 5										
	0 6										
	0 7										
	08										
	0 9										
	10										
	1 1										

For office work only (Branc	h Office)		
Number of Questionnaire			

	Name	Signature	Date
Enumerator			
supervisor			
Statistician			

For office work only (Head Office) Number of Questionnaire

	Name	Signature	Date
Editor		-	
Editor-verifyer			
Encoder			
Encoder-verifyer			

Date

ANNEX II

Estimation Procedures of Total, Ratio and Sampling Errors

The following formulas were used to estimate the required variables by reporting levels.

1. Estimate of Total \hat{Y}_h in Major Urban Domain (Category I)

$$\hat{Y}_h = \sum_{i=1}^{n_h} \sum_{i=1}^{h_{hi}} W_{hi} Y_{hij}$$

Where,

$$W_{hi} = \frac{M_h H_{hi}}{n_h M_{hi} h_{hi}}$$
 is the basic sampling weight

2. Estimate of Total \hat{Y}_h in Other Urban Domain (Category II)

$$\hat{Y}_{h} = \sum_{i=1}^{n_{h}} \frac{M_{h}}{n_{h}n_{h}} \sum_{i=1}^{n_{hii}} \frac{H_{hij}}{M_{h} h_{h}} \sum_{k=1}^{n_{hij}} Y_{hijk}$$

$$= \sum_{i=1}^{n_h} \sum_{j=1}^{n_{hi}} \sum_{k=1}^{h_{hij}} W_{hij} Y_{hijk}$$

Where,

$$W_{\rm hij} = \frac{M_{\rm h} H_{\rm hij}}{n_{\rm h} n_{\rm hi} M_{\rm hij} h_{\rm hij}}$$
 is basic sampling weight

The following notations were used in the formula:

 M_h = Total number of households in stratum h obtained from the sampling frame.

 M_{hi} = Total number of households in EA/PSU i for major urban domain or in urban center/PSU i for other urban domain, stratum h obtained from the sampling frame.

 H_{hi} = Total number of households obtained from the survey listing in sample EA/PSU i stratum h for major urban domain.

- h_{hi} = Total number of households successfully covered in EA/PSU i stratum h for major urban domain.
- $M_h = \text{Total number of households of the domain in stratum h obtained from the sampling frame}$
- n_h = Number of successfully covered urban centers for other urban domain / covered EAs for major urban domain in stratum h.
- $M_{\it hi} =$ Total number of households in EA/PSUi in stratum h obtained from the sampling frame
- $h_{\it hi=}$ = Total number of sampled and covered households in sampled EA/PSU i , and stratum h
- \hat{Y}_{hij} = The observed value of characteristic y for household j, in EA/PSUi, stratum h of Major Urban Domain
- M_{hij} = Total number of households in EA/SSU j, urban centers/PSU i and stratum h obtained from the sampling frame for other urban center domain.
- n_{hi} = Number of sample EAs successfully covered in urban center/PSU i and stratum h for other urban center domain.
- H_{hij} = Total number of households obtained from the survey listing in EA/SSU j, urban center/PSU i and stratum h for other urban center domain
- h_{hij} = Number of sample households successfully covered in EA /SSU j, urban center/PSU i and stratum h for other urban center domain.
- Y_{hijk} = The observed value of a characteristic y for household k in EA/SSUj, urban center/PSU i and stratum h for other urban domain.
- Note: Estimate of total at country level, \hat{Y} , is obtained by summing up stratum/domain total estimates.

$$\hat{Y} = \sum_{h} \hat{Y}_{h}$$

3. Sampling Variance of the Estimates:

Sampling variance of estimate of stratum total are given by the following formulas: The variance of domain or reporting total estimate is:

$$Var(\hat{Y}_h) = (1 - f_h) \frac{n_h}{n_h - 1} \sum_{i=1}^{n_h} \left(\hat{Y}_{hi} - \frac{\hat{Y}_h}{n_h} \right)^2 + f_h \sum_{i=1}^{n_h} (1 - f_{hi}) \left(\frac{h_{hi}}{h_{hi} - 1} \right) \sum_{j=1}^{h_{hi}} \left(\hat{Y}_{hij} - \frac{\hat{Y}_{hi}}{h_{hi}} \right)^2$$

in which $\hat{Y}_{hi} = W_{hi} \sum_{i=1}^{h_{hi}} Y_{hij}$ for major urban centers domains,

and $\hat{Y}_{hi} = \sum_{j=1}^{n_{hi}} W_{hij} \sum_{k=1}^{n_{hij}} y_{hijk}$ for other urban center domain.

$$V(\hat{Y}) = \sum_{h} V(\hat{Y}_{h})$$

$$SE(\hat{Y}_h) = \sqrt{Var(\hat{Y}_h)}$$

4. Coefficient of Variation (CV) and Confidence Interval (CI)

The following formulas were used to calculate CV and CI of the domain (reporting level) total.

The coefficient of variation (CV) of domain total in percentage is:

$$CV\left(\hat{Y}_{h}\right) = \frac{\sqrt{VAR(\hat{Y}_{h})}}{\hat{Y}_{h}} *100$$

and Ninety-five percent confidence interval (CI) of domain total:

$$\hat{Y}_h \pm 1.96 * SE(\hat{Y}_h)$$

5. Ratio Estimates:

$$\hat{R}_h = \frac{\hat{Y}_h}{\hat{X}_h}$$
 and $\hat{R} = \frac{\hat{Y}}{\hat{X}}$

Where the numerator and the denominator are estimates of domain totals of characteristic y and x, respectively.

$$Var\left(\hat{R}_{h}\right) = \frac{1}{\hat{X}_{h}^{2}} \left[Var\left(\hat{Y}_{h}\right) + \hat{R}_{h}^{2} Var\left(\hat{X}_{h}\right) - 2\hat{R}_{h} Cov\left(\hat{Y}_{h}, \hat{X}_{h}\right) \right]$$

In which

$$Cov(\hat{Y}_{h}, \hat{X}_{h}) = (1 - f_{h}) \frac{n_{h}}{n_{h} - 1} \sum_{i=1}^{n_{h}} \left(\hat{Y}_{hi} - \frac{\hat{Y}_{h}}{n_{h}} \right) \left(\hat{X}_{hi} - \frac{\hat{X}_{h}}{n_{h}} \right) + f_{h} \sum_{i=1}^{n_{h}} (1 - f_{hi}) \left(\frac{h_{hi}}{h_{hi} - 1} \right) \sum_{j=1}^{n_{hi}} \left(\hat{Y}_{hij} - \frac{\hat{Y}_{hi}}{h_{hi}} \right) \left(\hat{X}_{hij} - \frac{\hat{X}_{hi}}{h_{hi}} \right)$$

ANNEX - III

ANNEX TABLE 1. Urban Population Aged Ten Years And Over By Age Group, Sex and Activity Status During the Last Seven Days (Current Status Approach) – Country Total: 2012

Age Group and Sex		During the Last Sev	ren Days (Current Status Ap	proacrij –	Country rotal. 20	712
Total 1,110,653 1,141 6,944,691 1,51 4,165,962 1,89 Male 5,186,014 1,57 3,612,423 1,64 1,575,923,72 1,99 10-14		All Person	CV	Active	CV	Non-Active	CV
Male 5,186,014 1.57 3,612,423 1.64 1,573,590 2.28 Female 5,924,639 1.42 3,332,267 1.71 2,592,372 1.99 10-14 1 2 2.28 105,974 8.65 648,689 2.79 Male 754,662 2.83 105,974 8.65 648,689 2.79 Female 837,500 2.69 114,211 8.54 723,289 2.71 15-19 Total 1,752,372 2.04 628,140 3.16 1,124,233 2.48 Male 754,648 2.83 256,911 4.36 497,736 3.37 Female 997,725 2.32 371,228 3.97 626,496 2.79 20-24 Total 1,707,114 2.02 1,263,796 2.23 443,318 3.52 Total 1,707,114 2.02 1,263,796 2.23 443,318 3.52 25-29 Total 1,615,015 1.67 1,397,273							
Female	Total						
Total 1,592,162 2.33 20,185 7.64 1,371,978 2.28 Male 754,662 2.83 105,974 8.65 648,689 2.79 Female 837,500 2.69 114,211 8.54 723,289 2.71 15-19 Total 1,752,372 2.04 628,140 3.16 1,124,233 2.48 Male 754,648 2.83 256,911 4.36 497,736 3.37 Female 997,725 2.32 371,228 3.97 626,496 2.79 20-24 Total 1,707,114 2.02 1,263,796 2.23 443,318 3.52 Male 743,495 2.62 584,861 2.98 158,634 4.99 Female 963,619 2.32 678,935 2.74 284,684 4.19 25-29 Total 1,615,015 1.67 1,397,273 1.78 217,742 4.89 Male 776,270 2.34 733,144 2.41 43,126 9.61 Female 838,745 2.01 664,129 2.29 174,616 5.25 30-34 Total 1,006,702 2.08 895,639 2.17 110,63 5.49 Male 507,720 2.8 488,015 2.78 19,705 0.99 Female 498,982 2.73 407,624 3.1 91,357 6.07 35-39 Total 945,870 2.46 840,501 2.62 105,369 6.21 Male 465,729 3.33 450,158 3.36 15,571 0.73 Female 480,140 2.73 390,343 3.17 89,797 6.72 40-44 Total 56,878 2.6 503,733 2.74 65,145 7.46 40-44 Total 299,009 3.28 290,065 3.31 8,944 0.49 Female 299,009 3.28 290,065 3.31 8,944 0.49 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-54 Total 381,120 3.01 302,251 3.48 78,870 0.43 50-56 Total 381,120 3.01 302,251 3.48 78,870 0.45 50-56 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 176,523 4.21 166,568 4.34 9.995 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 176,523 4.21 166,568 4.34 9.995 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Female 151,640 4.52 70,910 6.45 80,730 6.35 60-64 239,219,4 4.09 132181.2 5.56	Male	5,186,014		3,612,423	1.64	1,573,590	
Total 1,592,162 2.33 220,185 7.64 1,371,978 2.28 Malae 754,662 2.83 105,974 8.65 648,669 2.79 Female 837,500 2.69 114,211 8.54 723,289 2.71 15-19 Total 1,752,372 2.04 628,140 3.16 1,124,233 2.48 Male 754,648 2.83 256,911 4.36 497,736 3.37 Female 997,725 2.32 371,228 3.97 626,496 2.79 20-24 Total 1,707,114 2.02 1,263,796 2.23 443,318 3.52 Male 743,495 2.62 584,861 2.98 158,634 4.96 Female 963,619 2.32 678,935 2.74 284,684 4.19 25-29 Total 1,615,015 1.67 1,397,273 1.78 217,742 4.89 Male 776,270 2.34 733,144 2.41 43,126 9.61 Female 838,745 2.01 664,129 2.29 174,616 5.25 30-34 Total 1,006,702 2.08 895,639 2.17 111,063 5.40 Male 507,720 2.8 488,015 2.78 19,705 0.99 Female 498,982 2.73 407,624 3.1 91,357 6.07 35-39 Total 945,870 2.46 840,501 2.62 105,369 6.21 Male 465,729 3.33 450,158 3.36 15,571 0.73 Female 480,140 2.73 390,343 3.17 89,797 6.72 40-44 Total 56,878 2.6 503,733 2.74 65,145 7.72 40-44 Total 299,009 3.28 290,065 3.31 8,944 0.49 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-54 Total 381,120 3.01 302,251 3.48 78,870 6.47 50-55 Total 181,565 5.21 106,145 5.65 12,210 0.57 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 176,523 4.21 166,568 4.34 9.955 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 176,523 4.21 166,568 4.34 9.955 0.59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 176,523 4.21 166,568 4.34 9.955 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 176,523 4.21 166,568 4.34 9.955 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 176,523 4.21 166,568 4.34 9.955 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 176,523 4.21 166,568 4.34	Female	5,924,639	1.42		1.71	2,592,372	1.99
Male 754,662 2.83 105,974 8.65 648,689 2.79 Female 837,500 2.69 114,211 8.54 723,289 2.71 15-19 Total 1,752,372 2.04 628,140 3.16 1,124,233 2.48 Male 754,648 2.83 256,911 4.36 497,736 3.37 Female 997,725 2.32 371,228 3.97 626,496 2.79 20-24 Total 1,707,114 2.02 1,263,796 2.23 443,318 3.52 Male 743,495 2.62 584,861 2.98 158,634 4.96 Female 963,619 2.32 678,935 2.74 284,684 4.19 25-29 Total 1,615,015 1.67 1,397,273 1.78 217,742 4.89 Male 776,270 2.34 733,144 2.41 43,126 9.61 Female 498,982 2.01 664,129 2.29							
Female	Total	1,592,162		220,185	7.64	1,371,978	2.28
Total 1,752,372 2.04 628,140 3.16 1,124,233 2.48 Male 754,648 2.83 256,911 4.36 497,736 3.37 Female 997,725 2.32 371,228 3.97 626,496 2.79 20-24 Total 1,707,114 2.02 1,263,796 2.23 443,318 3.52 Male 743,495 2.62 584,861 2.98 158,634 4.96 Female 963,619 2.32 678,935 2.74 284,684 4.19 25-29 Total 1,615,015 1.67 1,397,273 1.78 217,742 4.89 Male 776,270 2.34 733,144 2.41 43,126 9.61 Female 838,745 2.01 664,129 2.29 174,616 5.25 30-34 Total 1,006,702 2.8 895,639 2.17 111,063 5.46 Male 507,720 2.8 488,015 2.78 19,705 0.99 Female 498,982 2.73 407,624 3.1 91,357 6.07 35-39 Total 9,458,70 2.46 840,501 2.62 105,369 6.21 Male 465,729 3.33 450,158 3.36 15,571 0.73 Female 480,140 2.73 390,343 3.17 89,797 6.72 40-44 Total 568,878 2.6 503,733 2.74 65,145 7.46 Male 299,009 3.28 290,065 3.31 8,944 0.49 Female 269,869 3.66 213,668 4.13 56,201 8.13 45-49 Total 31,102 3.01 302,251 3.48 78,870 6.47 Male 233,108 4.1 224,308 4.2 8,799 0.43 45-49 Total 31,120 3.01 302,251 3.48 78,870 6.47 Male 176,523 4.21 166,568 4.34 9,955 0.59 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-55 9 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 176,523 4.21 166,568 4.34 9,955 0.59 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-55 9 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 118,356 5.21 100,145 5.65 12,210 0.57 Female 151,640 4.52 70,910 6.45 80,730 6.35 60-64 239,194 4.09 132181.2 5.56 107038.3 5.85 Total Male 118,356 5.21 100,145 5.65 12,210 0.57 Female 151,640 4.52 70,910 6.45 80,730 6.35 60-64 239,194 4.09 132181.2 5.56 107038.3 5.85 Total Male 1315,416 5.38 83,244 6.5 5 20,560 1.01 Female 135,416 5.38 48,937 9.15 86,479 6.19	Male	754,662	2.83	105,974	8.65	648,689	2.79
Total 1,752,372 2,04 628,140 3.16 1,124,233 2,48 Male 754,648 2.83 256,911 4.36 497,736 3.37 Female 97,775 2.32 371,228 3.97 626,496 2.79 20-24	Female	837,500	2.69	114,211	8.54	723,289	2.71
Male 754,648 2.83 256,911 4.36 497,736 3.37 Female 997,725 2.32 371,228 3.97 626,496 2.79 20-24 Total 1,707,114 2.02 1,263,796 2.23 443,318 3.52 Male 743,495 2.62 584,861 2.98 158,634 4.96 Female 963,619 2.32 678,935 2.74 284,684 4.19 25-29 Total 1,615,015 1.67 1,397,273 1.78 217,742 4.89 Male 776,270 2.34 733,144 2.41 43,126 9.61 Female 838,745 2.01 664,129 2.29 174,616 5.25 30-34 Total 1,006,702 2.08 895,639 2.17 111,063 5.46 Male 507,720 2.8 488,015 2.78 19,705 0.99 Female 498,982 2.73 407,624 3.1 9	15-19						
Female 20-24 997,725 2.32 371,228 3.97 626,496 2.79 20-24 Total 1,707,114 2.02 1,263,796 2.23 443,318 3.52 Male 743,495 2.62 584,861 2.98 158,634 4,96 Female 963,619 2.32 678,935 2.74 284,684 4.19 25-29 Total 1,615,015 1.67 1,397,273 1,78 217,742 4.89 Male 776,270 2.34 733,144 2.41 43,126 9.61 Female 838,745 2.01 664,129 2.29 174,616 5.25 30-34 Total 1,006,702 2.08 895,639 2.17 111,063 5.46 Male 507,720 2.8 488,015 2.78 19,705 0.99 Female 4945,870 2.46 840,501 2.62 105,369 6.21 Male 265,729 3.33 450,158 3.36	Total						
Total		754,648					
Total Male 1,707,114 2.02 1,263,796 2.23 443,318 3.52 Male 743,495 2.62 584,861 2.98 158,634 4.96 Female 963,619 2.32 678,935 2.74 284,684 4.19 25-29 Total 1,615,015 1.67 1,397,273 1.78 217,742 4.89 Male 776,270 2.34 733,144 2.41 43,126 9.61 Female 838,745 2.01 664,129 2.29 174,616 5.25 30-34 Total 1,006,702 2.08 895,639 2.17 111,063 5.46 Male 507,720 2.8 488,015 2.78 19,705 0.99 Female 498,982 2.73 407,624 3.1 91,357 6.07 35-39 Total 945,870 2.46 840,501 2.62 105,369 6.21 Male 456,729 3.33 450,158 3.36 <t< td=""><td>Female</td><td>997,725</td><td>2.32</td><td>371,228</td><td>3.97</td><td>626,496</td><td>2.79</td></t<>	Female	997,725	2.32	371,228	3.97	626,496	2.79
Male Female 743,495 2.62 584,861 2.98 158,634 4.96 Female 963,619 2.32 678,935 2.74 284,684 4.19 25-29 Total 1,615,015 1.67 1,397,273 1.78 217,742 4.89 Male 776,270 2.34 733,144 2.41 43,126 9.61 Female 838,745 2.01 664,129 2.29 174,616 5.25 30-34 Total 1,006,702 2.08 895,639 2.17 111,063 5.46 Male 507,720 2.8 488,015 2.78 197,055 6.07 Female 498,982 2.73 407,624 3.1 91,355 6.07 35-39 Total 945,870 2.46 840,515 2.62 105,369 6.21 Male 465,729 3.33 450,158 3.36 15,571 0.73 Female 480,140 2.73 390,343 3.17 <td< td=""><td>20-24</td><td></td><td></td><td></td><td></td><td></td><td></td></td<>	20-24						
Female 25-29 963,619 2.32 678,935 2.74 284,684 4.19 25-29 Total 1,615,015 1.67 1,397,273 1.78 217,742 4.89 Male 776,270 2.34 733,144 2.41 43,126 9.61 Female 838,745 2.01 664,129 2.29 174,616 5.25 30-34 Total 1,006,702 2.08 895,639 2.17 111,063 5.46 Male 507,720 2.8 488,015 2.78 19,705 0.99 Female 498,982 2.73 407,624 3.1 91,357 6.07 35-39 Total 945,870 2.46 840,501 2.62 105,369 6.21 Male 465,729 3.33 450,158 3.36 15,571 0.73 Female 480,140 2.73 390,343 3.17 89,797 6.72 40-44 Total 568,878 2.6 503,733 2	Total	1,707,114	2.02	1,263,796	2.23	443,318	3.52
Total 1,615,015 1.67 1,397,273 1.78 217,742 4.89 Male 776,270 2.34 733,144 2.41 43,126 9.61 Female 838,745 2.01 664,129 2.29 174,616 5.25 30-34 Total 1,006,702 2.08 895,639 2.17 111,063 5.46 Male 507,720 2.8 488,015 2.78 19,705 0.99 Female 498,982 2.73 407,624 3.1 91,357 6.07 35-39 Total 945,870 2.46 840,501 2.62 105,369 6.21 Male 465,729 3.33 450,158 3.36 15,571 0.73 Female 480,140 2.73 390,343 3.17 89,797 6.72 40-44 Total 568,878 2.6 503,733 2.74 65,145 7.46 Male 299,009 3.28 290,065 3.31 8,944 0.49 Female 269,869 3.66 213,668 4.13 56,201 8.13 45-49 Total 477,925 2.74 404,873 3.02 73,052 6.84 Male 233,108 4.1 224,308 4.2 8,799 0.43 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-54 Total 381,120 3.01 302,251 3.48 78,870 6.47 Male 176,523 4.21 166,568 4.34 9,955 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 118,356 5.21 106,145 5.65 12,210 0.57 Female 151,640 4.52 70,910 6.45 80,730 6.35 60-64 239219.4 4.09 132181.2 5.56 107038.3 5.85 Total Male 103,804 5.65 83,244 6.5 20,560 1.01 Female 103,804 5.65 83,244 6.5 20,560 1.01	Male	743,495	2.62	584,861	2.98	158,634	4.96
Total Male 1,615,015 (2.70) 1.67 (2.70) 1,397,273 (2.34) 1.78 (2.41) 217,742 (4.89) Male 776,270 (2.34) 733,144 (2.41) 43,126 (9.61) 9.61 Female 838,745 (2.01) 664,129 (2.29) 174,616 (5.25) 30-34 Total 1,006,702 (2.08) 895,639 (2.17) 111,063 (5.46) Male 507,720 (2.8) (48,8015 (2.78) 19,705 (0.99) 0.99 Female 498,982 (2.73) (407,624) (3.1) (91,357) 6.07 35-39 Total 945,870 (2.46) (840,501) (2.62) (105,369) (6.21) Male 465,729 (3.33) (450,158) (3.36) (15,571) (0.73) Female 480,140 (2.73) (3.90,343) (3.17) (89,797) (6.72) 40-44 Total (480,140) (2.73) (3.90,343) (3.17) (89,797) (6.72) 40-44 Total (568,878) (2.6) (503,733) (2.74) (55,145) (7.64) Male (2.99,009) (3.28) (2.90,065) (3.31) (89,44) (4.99) Female (2.69,869) (3.66) (2.13,668) (4.13) (56,201) (8.13) 45-49 Total (4.77,925) (2.74) (404,873) (3.02) (7.30,52) (6.84) Male (2.33,108) (4.1) (2.24,308) (4.2) (8.79) (4.24) (4.25) (6.94) 50-54 Total (381,120) (3.01) (3.02,251) (3.48) (4.34) (9.955) (6.94)	Female	963,619	2.32	678,935	2.74	284,684	4.19
Male Female 776,270 2.34 733,144 2.41 43,126 9.61 Female 838,745 2.01 664,129 2.29 174,616 5.25 30-34 1,006,702 2.08 895,639 2.17 111,063 5.46 Male 507,720 2.8 488,015 2.78 19,705 0.99 Female 498,982 2.73 407,624 3.1 91,357 6.07 35-39 Total 945,870 2.46 840,501 2.62 105,369 6.21 Male 465,729 3.33 450,158 3.36 15,571 0.73 Female 480,140 2.73 390,343 3.17 89,797 6.72 40-44 Total 568,878 2.6 503,733 2.74 65,145 7.46 Male 299,009 3.28 290,065 3.31 8,944 0.49 Female 269,869 3.66 213,668 4.13 56,201 8.13 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>							
Female 30-34 838,745 2.01 664,129 2.29 174,616 5.25 30-34 1,006,702 2.08 895,639 2.17 111,063 5.46 Male 507,720 2.8 488,015 2.78 19,705 0.99 Female 498,982 2.73 407,624 3.1 91,357 6.07 35-39 Total 945,870 2.46 840,501 2.62 105,369 6.21 Male 465,729 3.33 450,158 3.36 15,571 0.73 Female 480,140 2.73 390,343 3.17 89,797 6.72 40-44 Total 568,878 2.6 503,733 2.74 65,145 7.46 Male 299,009 3.28 290,065 3.31 8,944 0.49 Female 269,869 3.66 213,668 4.13 56,201 8.13 45-49 Total 477,925 2.74 404,873 3.02 73,052 <td>Total</td> <td>1,615,015</td> <td></td> <td>1,397,273</td> <td>1.78</td> <td>217,742</td> <td></td>	Total	1,615,015		1,397,273	1.78	217,742	
Total 1,006,702 2.08 895,639 2.17 111,063 5.46 Male 507,720 2.8 488,015 2.78 19,705 0.99 Female 498,982 2.73 407,624 3.1 91,357 6.07 35-39	Male	776,270	2.34	733,144	2.41	43,126	9.61
Total Male 1,006,702 2.08 895,639 2.17 111,063 5.46 Male 507,720 2.8 488,015 2.78 19,705 0.99 Female 498,982 2.73 407,624 3.1 91,357 6.07 35-39 Total 945,870 2.46 840,501 2.62 105,369 6.21 Male 465,729 3.33 450,158 3.36 15,571 0.73 Female 480,140 2.73 390,343 3.17 89,797 6.72 40-44 Total 568,878 2.6 503,733 2.74 65,145 7.46 Male 299,009 3.28 290,065 3.31 8,944 0.49 Female 269,869 3.66 213,668 4.13 56,201 8.13 45-49 Total 477,925 2.74 404,873 3.02 73,052 6.84 Male 233,108 4.1 224,308	Female	838,745	2.01	664,129	2.29	174,616	5.25
Male 507,720 2.8 488,015 2.78 19,705 0.99 Female 498,982 2.73 407,624 3.1 91,357 6.07 35-39 Total 945,870 2.46 840,501 2.62 105,369 6.21 Male 465,729 3.33 450,158 3.36 15,571 0.73 Female 480,140 2.73 390,343 3.17 89,797 6.72 40-44 Total 568,878 2.6 503,733 2.74 65,145 7.46 Male 299,009 3.28 290,065 3.31 8,944 0.49 Female 269,869 3.66 213,668 4.13 56,201 8.13 45-49 Total 477,925 2.74 404,873 3.02 73,052 6.84 Male 233,108 4.1 224,308 4.2 8,799 0.43 Female 244,817 3.38 180,565 4.04 64,252 <td< td=""><td>30-34</td><td></td><td></td><td></td><td></td><td></td><td></td></td<>	30-34						
Female 35-39 498,982 2.73 407,624 3.1 91,357 6.07 Total 35-39 945,870 2.46 840,501 2.62 105,369 6.21 Male 465,729 3.33 450,158 3.36 15,571 0.73 Female 480,140 2.73 390,343 3.17 89,797 6.72 40-44 70tal 568,878 2.6 503,733 2.74 65,145 7.46 Male 299,009 3.28 290,065 3.31 8,944 0.49 Female 269,869 3.66 213,668 4.13 56,201 8.13 45-49 Total 477,925 2.74 404,873 3.02 73,052 6.84 Male 233,108 4.1 224,308 4.2 8,799 0.43 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-54 Total 381,120 3.01 302,251 3.48 78,870 6.47 Male 176,523 4.21	Total	1,006,702		895,639	2.17	111,063	5.46
Total 945,870 2.46 840,501 2.62 105,369 6.21 Male 465,729 3.33 450,158 3.36 15,571 0.73 Female 480,140 2.73 390,343 3.17 89,797 6.72 40-44 Total 568,878 2.6 503,733 2.74 65,145 7.46 Male 299,009 3.28 290,065 3.31 8,944 0.49 Female 269,869 3.66 213,668 4.13 56,201 8.13 45-49 Total 477,925 2.74 404,873 3.02 73,052 6.84 Male 233,108 4.1 224,308 4.2 8,799 0.43 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-54 Total 381,120 3.01 302,251 3.48 78,870 6.47 Male 176,523 4.21 166,568 4.34 9,955 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 118,356 5.21 106,145 5.65 12,210 0.57 Female 151,640 4.52 70,910 6.45 80,730 6.35 60-64 239219.4 4.09 132181.2 5.56 107038.3 5.85 Total Male 151,640 4.52 70,910 6.45 80,730 6.35 60-64 239219.4 4.09 132181.2 5.56 107038.3 5.85 Total Male 103,804 5.65 83,244 6.5 20,560 1.01 Female 135,416 5.38 48,937 9.15 86,479 6.19	Male	507,720	2.8	488,015	2.78	19,705	0.99
Total 945,870 2.46 840,501 2.62 105,369 6.21 Male 465,729 3.33 450,158 3.36 15,571 0.73 Female 480,140 2.73 390,343 3.17 89,797 6.72 40-44 Total 568,878 2.6 503,733 2.74 65,145 7.46 Male 299,009 3.28 290,065 3.31 8,944 0.49 Female 269,869 3.66 213,668 4.13 56,201 8.13 45-49 Total 477,925 2.74 404,873 3.02 73,052 6.84 Male 233,108 4.1 224,308 4.2 8,799 0.43 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-54 Total 381,120 3.01 302,251 3.48 78,870 6.47 Male 176,523 4.21 166,568 4.34 9,955 <td< td=""><td>Female</td><td>498,982</td><td>2.73</td><td>407,624</td><td>3.1</td><td>91,357</td><td>6.07</td></td<>	Female	498,982	2.73	407,624	3.1	91,357	6.07
Male 465,729 3.33 450,158 3.36 15,571 0.73 Female 480,140 2.73 390,343 3.17 89,797 6.72 40-44 70 (4) 568,878 2.6 503,733 2.74 65,145 7.46 Male 299,009 3.28 290,065 3.31 8,944 0.49 Female 269,869 3.66 213,668 4.13 56,201 8.13 45-49 Total 477,925 2.74 404,873 3.02 73,052 6.84 Male 233,108 4.1 224,308 4.2 8,799 0.43 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-54 Total 381,120 3.01 302,251 3.48 78,870 6.47 Male 176,523 4.21 166,568 4.34 9,955 0.59 Female 204,597 4.11 135,682 5.08 68,915 <t< td=""><td>35-39</td><td></td><td></td><td></td><td></td><td></td><td></td></t<>	35-39						
Female 40-44 480,140 2.73 390,343 3.17 89,797 6.72 40-44 Total 568,878 2.6 503,733 2.74 65,145 7.46 Male 299,009 3.28 290,065 3.31 8,944 0.49 Female 269,869 3.66 213,668 4.13 56,201 8.13 45-49 Total 477,925 2.74 404,873 3.02 73,052 6.84 Male 233,108 4.1 224,308 4.2 8,799 0.43 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-54 Total 381,120 3.01 302,251 3.48 78,870 6.47 Male 176,523 4.21 166,568 4.34 9,955 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Total 269,995 3.46 177,055 4.27				840,501			
40-44 Total 568,878 2.6 503,733 2.74 65,145 7.46 Male 299,009 3.28 290,065 3.31 8,944 0.49 Female 269,869 3.66 213,668 4.13 56,201 8.13 45-49 Total 477,925 2.74 404,873 3.02 73,052 6.84 Male 233,108 4.1 224,308 4.2 8,799 0.43 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-54 Total 381,120 3.01 302,251 3.48 78,870 6.47 Male 176,523 4.21 166,568 4.34 9,955 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 118,356 5.21 106,145 5.65 12,210 0.57 Female 151,640 4.	Male	465,729		450,158			
Total 568,878 2.6 503,733 2.74 65,145 7.46 Male 299,009 3.28 290,065 3.31 8,944 0.49 Female 269,869 3.66 213,668 4.13 56,201 8.13 45-49 Total 477,925 2.74 404,873 3.02 73,052 6.84 Male 233,108 4.1 224,308 4.2 8,799 0.43 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-54 Total 381,120 3.01 302,251 3.48 78,870 6.47 Male 176,523 4.21 166,568 4.34 9,955 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 118,356 5.21 106,145 5.65 12,210		480,140	2.73	390,343	3.17	89,797	6.72
Male 299,009 3.28 290,065 3.31 8,944 0.49 Female 269,869 3.66 213,668 4.13 56,201 8.13 45-49 Total 477,925 2.74 404,873 3.02 73,052 6.84 Male 233,108 4.1 224,308 4.2 8,799 0.43 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-54 50-55	40-44						
Female 45-49 269,869 3.66 213,668 4.13 56,201 8.13 Total 477,925 2.74 404,873 3.02 73,052 6.84 Male 233,108 4.1 224,308 4.2 8,799 0.43 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-54 50-59 50-							
Total 477,925 2.74 404,873 3.02 73,052 6.84 Male 233,108 4.1 224,308 4.2 8,799 0.43 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-54 Total 381,120 3.01 302,251 3.48 78,870 6.47 Male 176,523 4.21 166,568 4.34 9,955 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 118,356 5.21 106,145 5.65 12,210 0.57 Female 151,640 4.52 70,910 6.45 80,730 6.35 60-64 239219.4 4.09 132181.2 5.56 107038.3 5.85 Total Male 103,804 5.65 83,244 6.5 20,560 1.01 Female 135,416 5.38 48,937 9.15 86,479 6.19				290,065			
Total 477,925 2.74 404,873 3.02 73,052 6.84 Male 233,108 4.1 224,308 4.2 8,799 0.43 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-54 Total 381,120 3.01 302,251 3.48 78,870 6.47 Male 176,523 4.21 166,568 4.34 9,955 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 118,356 5.21 106,145 5.65 12,210 0.57 Female 151,640 4.52 70,910 6.45 80,730 6.35 60-64 239219.4 4.09 132181.2 5.56 107038.3 5.85 Total Male 103,804 5.65 83,244 6.5 20,560 <		269,869	3.66	213,668	4.13	56,201	8.13
Male 233,108 4.1 224,308 4.2 8,799 0.43 Female 244,817 3.38 180,565 4.04 64,252 6.94 50-54 50-54 50-54 50-54 50-54 50-54 50-54 50-54 50-54 50-54 6.47 6.47 6.47 6.47 6.47 6.47 6.47 6.47 6.47 6.47 6.47 6.43 9,955 0.59 6.59 6.59 6.59 6.59 6.59 6.90 6.59 6.59 6.59 6.59 6.59 6.59 6.59 6.59 6.59 6.59 6.59 6.59 6.59 6.59 6.59 6.91 7.07 8.09 8.09 7.07 7.07 7.07 7.07 7.07 7.07 7.07 7.07 7.07 7.07 7.07 7.07 7.07 7.07 7.07							
Female 50-54 244,817 3.38 180,565 4.04 64,252 6.94 Total 70tal 70ta							
50-54 Total 381,120 3.01 302,251 3.48 78,870 6.47 Male 176,523 4.21 166,568 4.34 9,955 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 700 <td></td> <td>233,108</td> <td></td> <td></td> <td>4.2</td> <td>8,799</td> <td></td>		233,108			4.2	8,799	
Total 381,120 3.01 302,251 3.48 78,870 6.47 Male 176,523 4.21 166,568 4.34 9,955 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 118,356 5.21 106,145 5.65 12,210 0.57 Female 151,640 4.52 70,910 6.45 80,730 6.35 60-64 239219.4 4.09 132181.2 5.56 107038.3 5.85 Total Male 103,804 5.65 83,244 6.5 20,560 1.01 Female 135,416 5.38 48,937 9.15 86,479 6.19		244,817	3.38	180,565	4.04	64,252	6.94
Male 176,523 4.21 166,568 4.34 9,955 0.59 Female 204,597 4.11 135,682 5.08 68,915 7.07 55-59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 118,356 5.21 106,145 5.65 12,210 0.57 Female 151,640 4.52 70,910 6.45 80,730 6.35 60-64 239219.4 4.09 132181.2 5.56 107038.3 5.85 Total Male 103,804 5.65 83,244 6.5 20,560 1.01 Female 135,416 5.38 48,937 9.15 86,479 6.19							_
Female 55-59 204,597 4.11 135,682 5.08 68,915 7.07 Total 70-55-59 269,995 3.46 177,055 4.27 92,940 5.96 Male 118,356 5.21 106,145 5.65 12,210 0.57 Female 151,640 4.52 70,910 6.45 80,730 6.35 60-64 239219.4 4.09 132181.2 5.56 107038.3 5.85 Total Male Male 103,804 5.65 83,244 6.5 20,560 1.01 Female 135,416 5.38 48,937 9.15 86,479 6.19							
55-59 Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 118,356 5.21 106,145 5.65 12,210 0.57 Female 151,640 4.52 70,910 6.45 80,730 6.35 60-64 239219.4 4.09 132181.2 5.56 107038.3 5.85 Total Male 103,804 5.65 83,244 6.5 20,560 1.01 Female 135,416 5.38 48,937 9.15 86,479 6.19							
Total 269,995 3.46 177,055 4.27 92,940 5.96 Male 118,356 5.21 106,145 5.65 12,210 0.57 Female 151,640 4.52 70,910 6.45 80,730 6.35 60-64 239219.4 4.09 132181.2 5.56 107038.3 5.85 Total Male 103,804 5.65 83,244 6.5 20,560 1.01 Female 135,416 5.38 48,937 9.15 86,479 6.19		204,597	4.11	135,682	5.08	68,915	7.07
Male 118,356 5.21 106,145 5.65 12,210 0.57 Female 151,640 4.52 70,910 6.45 80,730 6.35 60-64 239219.4 4.09 132181.2 5.56 107038.3 5.85 Total Male 103,804 5.65 83,244 6.5 20,560 1.01 Female 135,416 5.38 48,937 9.15 86,479 6.19			_				
Female 151,640 4.52 70,910 6.45 80,730 6.35 60-64 239219.4 4.09 132181.2 5.56 107038.3 5.85 Total Nale 103,804 5.65 83,244 6.5 20,560 1.01 Female 135,416 5.38 48,937 9.15 86,479 6.19							
60-64 239219.4 4.09 132181.2 5.56 107038.3 5.85 Total Male 103,804 5.65 83,244 6.5 20,560 1.01 Female 135,416 5.38 48,937 9.15 86,479 6.19							
Total Male 103,804 5.65 83,244 6.5 20,560 1.01 Female 135,416 5.38 48,937 9.15 86,479 6.19	Female						
Total Male 103,804 5.65 83,244 6.5 20,560 1.01 Female 135,416 5.38 48,937 9.15 86,479 6.19	60-64	239219.4	4.09	132181.2	5.56	107038.3	5.85
Female 135,416 5.38 48,937 9.15 86,479 6.19							
Female 135,416 5.38 48,937 9.15 86,479 6.19	Male	103,804	5.65	83,244	6.5	20,560	1.01
	65+						
Total 554,280 3.22 179,064 4.89 375,215 3.81		554,280	3.22	179,064	4.89	375,215	3.81
Male 252,690 4.15 123,030 5.65 129,661 5.55						129,661	
Female 301,589 3.77 56,035 8.28 245,555 4.19		301,589		56,035			4.19

ANNEX TABLE 2 Economically Active Urban Population of Major Towns Aged
Ten Years And Over By Age Group, Sex, Whether Employed or
Not During the Last Seven Days (Current Status Approach): 2012

Age Group	Economically	o Last	Days (cuit	one orac	45 / Ipprodony: 20	
	, and the second	CV	Employed	CV	Unomployed	CV
and Sex	Active	CV	Employed	CV	Unemployed	CV
All Ages	4 044 401	1 51	E 704 114	1 41	1 210 574	2 71
Total	6,944,691	1.51	5,726,116	1.61 1.7	1,218,574	2.71 3.96
Male	3,612,423	1.64	3,202,042		410,381	
Female	3,332,267	1.71	2,524,074	1.96	808,193	2.98
10-14	220 105	7 4 1	105 547	7.00	24 (10	17 OE
Total	220,185 105,974	7.64 8.65	195,567 94,187	7.82 9.02	24,618 11,787	17.85
Male		8.54	101,380			21.31
Female	114,211	8.34	101,380	8.73	12,831	21.63
15-19 Total	628,140	3.16	492,071	3.75	136,069	5.37
	256,911	4.36	205,515	5.06	51,396	8.52
Male Female	371,228	3.97	286,556	4.58	84,673	6.75
20-24	3/1,220	3.97	200,000	4.30	04,073	0.75
Total	1,263,796	2.23	902,304	2.54	361,492	3.64
Male	584,861	2.23	457,454	3.29	127,407	5.76
Female	678,935	2.74	444,850	3.33	234,085	4.15
25-29	070,733	2.74	444,000	3.33	234,003	4.13
Total	1,397,273	1.78	1,128,487	2	268,787	3.94
Male	733,144	2.41	653,525	2.63	79,619	7.03
Female	664,129	2.29	474,962	2.77	189,168	4.43
30-34	004,127	2.27	474,702	2.11	107,100	4.43
Total	895,639	2.17	767,262	2.38	128,378	5.02
Male	488,015	2.78	448,530	2.93	39,484	8.39
Female	407,624	3.1	318,731	3.45	88,893	6.54
35-39	407,024	5.1	310,731	3.43	00,073	0.54
Total	840,501	2.62	742,001	2.73	98,500	5.8
Male	450,158	3.36	423,955	3.48	26,203	11.98
Female	390,343	3.17	318,046	3.49	72,297	6.66
40-44	370,343	3.17	310,040	5.47	12,271	0.00
Total	503,733	2.74	445,655	2.92	58,078	7.74
Male	290,065	3.31	273,476	3.43	16,589	13.09
Female	213,668	4.13	172,179	4.47	41,489	9.37
45-49	210,000	1.10	1,2,1,,	,	11/107	7.07
Total	404,873	3.02	361,687	3.25	43,186	8.89
Male	224,308	4.2	211,592	4.34	12,716	15.73
Female	180,565	4.04	150,094	4.56	30,470	10.2
50-54	.00/000		.00,07.		33/1/3	
Total	302,251	3.48	266,025	3.66	36,225	9.95
Male	166,568	4.34	153,598	4.53	12,971	17.94
Female	135,682	5.08	112,428	5.43	23,254	11.46
55-59			,		-,	
Total	177,055	4.27	152,584	4.67	24,471	10.67
Male	106,145	5.65	96,469	6.03	9,677	17.27
Female	70,910	6.45	56,115	7.39	14,794	14.27
60-64						
Total	132,181	5.56	113,615	6.04	18,566	13.55
Male	83,244	6.5	75,221	7	8,023	19.25
Female	48,937	9.15	38,393	10.02	10,544	17.36
65+						
Total	179,064	4.89	158,859	5.14	20,205	12.59
Male	123,030	5.65	108,519	5.89	14,510	14.89
Female	56,035	8.28	50,340	8.8	5,694	23.04

ANNEX TABLE 3 Paid Employees of Urban Population Urban Areas Aged Ten Years and Over by Sex, Period of Payment and Amount of Total Payment – Country Total: 2012

								Amo	unt of Total	Pavmen	t(in Birr)							
								7 11 110	01 10101	. ajmon								
	Total Paid																	
Doriod of Daymont	Employees	CV	.EO	CV	E0 00	CV	100 100	CV	200 200	CV	400 400	CV	700 000	CV	1000.	CV	Not Ctatad	CV
Period of Payment	Employees	CV	<50	CV	50-99	CV	100-199	CV	200-399	CV	400-699	CV	700-999	CV	1000+	CV	Not Stated	CV
Total	2,867,785	1.87	14,939	16.6	29,218	10.79	108,926	6.23	388,948	3.72	548,049	3.51	385,789	3.47	1,368,088	2.79	23,827	15.62
Male	1,682,365	2	1,658	42.13	5,576	23.27	24,032	12.85	118,389	6.18	283,880	4.45	250,551	4.12	984,180	2.77	14,099	16.55
Female	1,185,420	2.5	13,280	17.93	23,642	12.11	84,894	6.74	270,559	4.4	264,169	4.73	135,239	5.51	383,908	4.03	9,728	23.14
Hourly																		
Total	20,972	15.92	1042.6	61.23	173	68.58	1,290	54.36	2,901	33.51	3,777	33.76	2,642	43.38	8,496	26.77	650.01	65.65
Male	14,847	17.45	489.6	83.92	0	0	385	74.08	1,715	41.72	2,760	41.22	2,541	44.97	6,652	27.2	303.93	81.61
Female	6,125	28.45	553	88.35	173.42	68.58	904	70.81	1186	55.48	1,017	56.64	101.27	99.51	1843.5	49.14	346.08	0.01
Daily																		
Total	124,396	7.52	2,000	48.14	3,322	34.49	7,271	20.75	15,658	15.41	27,827	12.22	18,203	14.23	48,694	11.77	1419.25	52.14
Male	95,866	7.86	59.22	98.5	1,159	56.63	3,639	26.55	8,907	18.98	20,838	13.2	16,400	14.66	43,890	11.77	972.53	57.95
Female	28,530	13.41	1,941	49.52	2163.06	43.47	3,632	32.16	6,752	23.86	6,988	24.34	1,803	43.04	4804.38	33.16	446.72	0.01
Weekly																		
Zer Total Male	119,978	9.71	1,125	54.79	1217.3	55.98	6,682	42.59	14,542	15.56	32,214	15.57	25,680	18.06	38,237	11.11	280.52	99.82
	91,241	9.4	209	96.11	642.13	74.72	3,887	33.72	7,491	23.02	20,783	18.56	21,180	14.94	36,769	11.2	280.52	99.82
Female	28,737	19.97	916	63.62	575.17	84.14	2,795	67.34	7,051	20.51	11,431	23.02	4,500	49.47	1468.08	41.24	0	0
Fortnight	00 /7/	10.07	FF0 40	70.04	4 407	F 4 77	1.040	FF 00	11 004	10.50	20.4/2	47.50	1 (00 (15.01	05.040	4/4	202 70	70.74
Total	89,676	10.07	552.43	73.04	1,187	54.77	1,043	55.33	11,084	19.59	33,463	17.53	16,824	15.01	25,319	16.1	203.78	72.74
Male	55,420	10.48	346.08	0.01	0 1,187	0	222.46	0.03	969	0.17	2,133	0.78	2,371	0.85	4,025 1504	1.37 40.95	79.66	0.01
Female Monthly	34,255	18.06	206.35	99.76	1,107	54.77	727	73.27	7,953	23.47	19,813	25.66	2,742	30.79	1304	40.95	123.53	0.01
Total	2,459,248	2.03	9,379	18.78	22,089	12.52	88,594	6.32	335,315	4.06	436,670	3.38	316,180	3.75	1,234,931	3	16,090	20.13
Male	1,395,219	2.03	554.41	71.06	3,439	28.49	14,804	16.98	92,971	6.99	220,076	3.30 4.74	191,074	4.65	863,134	2.99	9,167	19.96
Female	1,393,219	2.17	8,825	19.46	18,651	13.23	73,790	6.56	242,344	4.68	216,594	4.74	125,106	5.58	371,798	4.07	6,923	29.13
Yearly	1,004,027	2.02	0,023	17.40	10,001	13.23	13,170	0.50	242,344	4.00	210,374	4.47	123,100	3.30	371,770	4.07	0,723	27.13
Total	12.056	23.49	633.46	58.4	802	50.71	1,403	51.07	2,062	38.17	2283.22	49.05	654.15	67.7	4,219	41.4	0	0
Male	5,656	25.65	0	0	336	69.47	813.33	71.68	1,058	53.81	403.07	99.95	654.15	67.7	2,392	43.15	0	0
Female	6,400	29.73	633.46	58.4	465.24	72.32	589.6	70.63	1004.27	54.79	1880.15	55.58	001.10	0	1827.18	51.28	0	0
Other/specify	0,100	270	000110	00	.00.2	, 2.02	00710	7 0.00	100 1127	0 ,	.0000	00.00		ŭ	.027110	01.20	· ·	· ·
Total	37,495	21.45	206	99.76	427	74.84	2.643	34.49	7,386	31.08	11,784	40.28	5,606	24.5	7,866	19.53	1576.88	43.44
Male	20,940	15.97	0	0	0	0	187	71.09	3,117	31.66	5,338	34.92	4,619	25.13	7,225	20.65	453.12	69.14
Female	16,555	38.99	206	99.76	427	74.84	2,456	36.72	4,270	40.49	6445.73	68.84	987.19	74.34	640.31	58.05	1123.76	54.21
Not Stated	,								,									
Total	3,964	37.69	0	0	0	0	0	0	0	0	31.41	83.53	0	0	326.25	86.47	3,607	40.68
Male	3,176	42.9	0	0	0	0	0	0	0	0	31.41	83.53	0	0	303.31	92.49	2,841	46.92
Female	788.26	69.05	0	0	0	0	0	0	0	0	0	0	0	0	22.94	84.21	765.32	71.08

REFERENCES

ILO (1990).	Survey of Economically Active population, Employment, Unemployment and Under Employment: An ILO Manual on Concepts and methods, Geneva.
ILO (1996).	Child Labor Survey: Results of Methodological experiments in four Countries 1992/1993. Geneva.
ILO (1998).	Sixteenth International Conference of Labour Statistical report of the Conference, Geneva.
ILO (1999).	Key Indicators of the Labor Market country profile, Geneva.
ILO (1999).	Unemployment and Underemployment in South Africa.

Shryock, H. S. and Siegel J.S. (1976). The Methods and Materials of Demography (Condensed Edition) San Diego, California.

ILO (2000) Current International Recommendations on Labour Statistics, 2nd edition, Geneva.

UN (2008), International Standard Industrial Classification of All Economic Activities, **Rev.4**. New York